# ESA Biological Evaluation for Clean Water Act Section 404 Assumption by the State of Florida

July 2023



# ESA Biological Evaluation for Clean Water Act Section 404 Assumption by the State of Florida

July 2023



U.S. Environmental Protection AgencyRegion 461 Forsyth St SWAtlanta, GA 30303

## **Executive Summary**

In accordance with the Endangered Species Act of 1973 (ESA) and its implementing regulations, the purpose of this Biological Evaluation (BE) is to evaluate the possible effects of the U.S. Environmental Protection Agency's (EPA) approval of the State of Florida's assumption of administration of a Clean Water Act (CWA) Section 404 program on ESA-listed species, proposed species, designated critical habitat, and proposed critical habitat (Title 50 of the *Code of Federal Regulations* [CFR] § 402.12). The BE considers whether the EPA's approval of the assumption request (Action) might affect listed species or designated critical habitat and if any effects are likely to adversely affect any such species or habitat in the Action Area, including in retained waters downstream of waters assumed by the State. The EPA is the federal agency charged with approving or denying the State's request, pursuant to the CWA implementing regulations (40 CFR Part 233). The Florida Department of Environmental Protection (FDEP) is the State agency that assumed administration of a CWA Section 404 Program on December 17, 2020 (assumption).

Florida's assumed CWA Section 404 program includes only those waters of the United States not retained by the U.S. Army Corps of Engineers (USACE), referred to as assumed waters or Stateassumed waters. The USACE retained permitting responsibility for the discharge of dredged or fill material in those waters defined as "retained waters." This definition can be found in FDEP's *State 404 Program Applicant's Handbook* in Chapter 2.0, and the process to determine whether a project is in retained or assumed waters is described in Chapter 4.1 of the Handbook (FDEP 2020a). In addition, Appendix A of FDEP's State 404 Handbook includes the Retained Waters List maintained by the USACE. The administrative boundary demarcating the adjacent wetlands over which jurisdiction is retained by the USACE is a 300-foot guideline established from the ordinary high-water mark or mean high tide line of the retained water. In the case of a project that involves discharges of dredged or fill material both waterward and landward of the 300-foot guideline, the USACE retains jurisdiction to the landward boundary of the project for the purposes of that project only. The USACE also retains permitting authority for projects within "Indian country" as that term is defined at 18 USC § 1151.

On July 17, 2019, FDEP sent a request to the EPA that sought designation, pursuant to 50 CFR § 402.08, to serve as a non-federal representative to prepare a Biological Assessment (BA) for possible ESA Section 7 consultation. The EPA designated FDEP as the non-federal representative to prepare a BA in a letter dated December 12, 2019, consistent with 50 CFR § 402.08; accordingly, FDEP drafted a BA, which the EPA adapted into the 2020 BE. In a separate letter to the U.S. Department of Commerce and the Department of the Interior dated December 13, 2019, the EPA provided notification that it would voluntarily engage in informal consultation with the Services (the U.S. Fish and Wildlife Service [USFWS] and the National Oceanic and Atmospheric Administration National Marine Fisheries Service [NMFS]).

FDEP requested input on a draft list of threatened and endangered species and designated critical habitat from the USFWS and the NMFS on November 22, 2019. On April 15, 2020, the NMFS responded to FDEP that ESA-listed species under NMFS's jurisdiction do not occur in waters that are assumable by the State (see Appendix A). NMFS stated that based on its review, while there is shared jurisdiction for the Gulf sturgeon between NMFS and USFWS, the USFWS is responsible for consultations regarding sturgeon and critical habitat in riverine habitat units. NMFS further stated that the riverine critical habitat units and river areas where Gulf sturgeon are known to occur are all listed by the USACE as retained waters. Based on NMFS's determination, the EPA, therefore, concluded

that approval by the EPA of FDEP's assumption of the Section 404 program would have no effect on NMFS jurisdictional species within assumed waters. Nevertheless, the EPA has prepared this BE to consider whether the EPA's approval Action might affect listed species or designated critical habitat in the Action Area, including whether the Action is likely to adversely affect any such species or habitat in retained waters downstream of waters assumed by the State.

The State's assumption of the dredge and fill permitting program under Section 404 of the CWA for waters assumable by the State is implemented by processes and procedures described in State regulations (Chapters 62-330 and 62-331, Florida Administrative Code [F.A.C.]), Memoranda of Agreement with the EPA and with the USACE, and a Memorandum of Understanding with the Florida Fish and Wildlife Conservation Commission (FWC) and the USFWS. The EPA is initiating consultation with NMFS for the EPA's Action, the approval of Florida's request to assume administration of a CWA Section 404 program, and is requesting concurrence with its effects determinations for the Shortnose Sturgeon, the Atlantic Sturgeon, and the Smalltooth Sawfish. In accordance with ESA Section 7 regulations, this BE evaluates effects only on ESA species of concern, which includes ESA-listed species and their critical habitats, species proposed to be listed and their proposed critical habitat, candidate species, petitioned species, and species otherwise under review for potential ESA listing. For the sake of clarity in understanding the State's implementation of its program, the State 404 program would address species that may not be listed under the ESA but are protected under other federal or state laws. Examples of these types of laws include the Bald and Golden Eagle Protection Act (Title 16 United States Code [USC] § 668) and the Migratory Bird Treaty Act of 1918 (16 USC §§ 703-712 (§ 709 omitted)). Also, all NMFS ESA proposed, candidate, and petitioned plant and animal species with ranges within Florida are discussed in this BE, to provide a thorough review of Florida's potentially at-risk species, and because the lists of threatened and endangered species may be modified to include some or all the petitioned species, depending upon the outcome of those status reviews. This BE evaluates the effects that the EPA's approval of the State of Florida's assumption of the CWA Section 404 program could have on listed species in present time and analyzes effects into the future. All species currently listed, proposed to be listed, or might be listed in the future are discussed, with evaluations of effects on respective guilds. This review evaluates the effects the State's 404 program could have on any species currently listed or those for which there is reason to believe they could be listed in the future because implementation of the State's 404 program would be the direct effect of the EPA's approval of Florida's assumption of the CWA Section 404 program.

This BE analyzes a total of 28 animal species in the State of Florida that are federally listed as endangered (15), threatened (12) or proposed for consideration (one). Species that must be addressed pursuant to Section 404 of the CWA and Chapter 62-331, F.A.C. for the State 404 program are federally listed species or species proposed to be listed under the ESA. If these species have designated critical habitats or critical habitats proposed to be designated, the adverse effects and impacts on those habitats also must be analyzed and addressed in the State 404 program. <u>See</u> Table 3-1 for the list of species discussed in this BE.

Because the State 404 program is implemented through State-issued 404 permits authorizing activities, a programmatic BE is appropriate to describe the regulatory process by which the State of Florida makes determinations on the issuance of State 404 permits. This BE addresses how the process considers effects on listed endangered and threatened species or species proposed to be listed so that the effects of this Action might be analyzed. The BE describes the effects of the Action, which include a broad array of activities likely to be authorized and a general description of the ESA-listed species

and their habitat ranges and critical habitats likely to be affected by such activities. These descriptions include baseline discussions; historical perspectives based on previous permitting by the USACE; an estimate of the physical, chemical, and biotic stressors to species that are likely to be produced; and a description of the processes and mechanisms to avoid and minimize the adverse effects of these activities on ESA-listed species and designated critical habitats.

In implementing the State 404 program, FDEP sends copies of all permit applications and its preliminary site-specific determinations of potential effects on listed species to the NMFS for review and comment (see Chapter 7 of this document for details). This coordination is to ensure that any permit issued by FDEP is not likely to jeopardize the continued existence of any listed or proposed species or to adversely modify or destroy designated critical habitat (pursuant to 40 CFR § 233.20(a)). Upon coordination, FDEP will consider any information that NMFS provides as technical assistance and will include all species protection measures that the NMFS may recommend as permit conditions or deny the request for a permit, whether the permit conditions are designed to avoid jeopardizing an ESA-listed species or adverse modification of designated critical habitat, or whether permit conditions are designed to avoid or minimize the amount of take when the take or other effects would not be likely to jeopardize a species or adversely modify designated critical habitat. This exchange of information between NMFS and FDEP falls within the broad scope of "technical assistance" as described in the ESA's implementing regulations and the NMFS's Interagency Consultation Handbook.

FWC intends to partner with FDEP to assist in the coordination of federally listed species reviews with the NMFS, expanding FWC's current review of impacts on federally listed and State listed species during the State's Environmental Resource Permit (ERP) process (Chapter 62-330, F.A.C.). The interactions proposed to take place among agencies will inform applicants of the importance of implementing impact avoidance and minimization measures on listed species in order to be eligible to receive authorization of their proposed activities.

The State 404 program rule (Rules 62-331.053(3)(a)4, 62-331.201(3)(k), and 62-331-248(3)(k), F.A.C.) and EPA regulations (40 CFR §§ 233.20, 230.10(b)(3)) prohibit issuance of a permit that will likely jeopardize the continued existence of endangered or threatened species or result in the likely destruction or adverse modification of habitat designated as critical for these species as determined by NMFS, and confirm that NMFS's conclusions about the effects of State 404 permits on listed species are determinative (40 CFR § 230.30(c)). FDEP will monitor adverse effect determinations on listed species and critical habitat by incorporating information into their permit tracking database, similar to the information collected by the USACE. This data collection will assist in facilitating compliance with permit conditions designed to avoid jeopardy or adverse modification of designated critical habitat would ultimately result in the State either denying the permit or the State informing the EPA that it will neither issue nor deny the permit, which would result in the EPA transferring the permit to the USACE for processing in accordance with 40 CFR § 233.50(j).

FDEP has indicated that it intends to resolve all disputes and objections with NMFS prior to the EPA's review during the public noticing process. Because of this process and the State's rule stipulating that NMFS's recommendations and conditions are determinative, it is unlikely that issues would need to be elevated to the EPA. During its review, the EPA might comment upon, object to, or make recommendations with respect to a permit pursuant to 40 CFR § 233.50. As stated in 40 CFR § 233.50(j), in the event a state that has assumed CWA Section 404 responsibilities neither satisfies

EPA's objections or requirements for a permit condition nor denies the permit, the EPA will transfer the permit to the USACE, and the USACE shall process the permit application.

During the analysis of existing Florida CWA Section 404 permitting data provided by the USACE for fiscal years 2014 through 2018, approximately seven percent of past USACE Section 404 permit application reviews involved formal consultations. It is reasonable to anticipate that the past number of applications submitted for that five-year period provides an approximation of the number and types of proposed permitting activities that might occur over the next five-year period. Further analysis of this data shows that a small proportion of the total number of ESA-listed species accounted for the majority of consultations. Many of the species subject to frequent ESA consultation have programmatic consultation keys or programmatic biological opinions, which can help guide or inform future reviews and impact assessments by FDEP, FWC, and NMFS. See Chapter 4.2.5 in this document regarding the limitations of the USACE queried data used in the analyses of this BE and interpretation using the USACE database and shapefile.

Historical CWA Section 404 permitting by the USACE resulted in issuance of permits for one or more projects that adversely affected one or more listed species and their critical habitats. The USACE Section 404 process, in accordance with Section 7 of the ESA, employed various conservation measures to avoid and minimize adverse effects. Similar to the USACE Section 404 program, however, the State 404 program might result in the issuance of one or more permits with projects that may affect but are not likely to adversely affect an ESA-listed species and/or designated critical habitat. Any adverse effects would be addressed during the technical assistance process between FDEP and NMFS.

This BE is provided to assist in the review of potential effects on ESA-listed endangered and threatened species anticipated with the State of Florida's assumption of Section 404 of the CWA in the Action Area, including in retained waters downstream of waters assumed by the State. EPA is requesting concurrence by NMFS on the effects determinations for the Atlantic sturgeon, Shortnose sturgeon, and smalltooth sawfish, set forth within this BE with respect to EPA's approval of the State's Section 404 program.

## Contents

Ex	ecutive Sun	1mary	i		
Gl	ossary		xi		
1	Purpose of	the Biological Evaluation	1		
	1.1 Objectives of Action				
	1.1.1	The Programmatic Biological Evaluation	2		
	1.2 The I	History of Florida's Request for Assumption	3		
	1.3 Orga	nization of the Document	4		
2	Action and	Action Area	5		
	2.1 Desc	ription of Action	5		
	2.2 Type	s of Activities Authorized by Florida's 404 Program	6		
	2.2.1	Proposed Activities Authorized or Exempt Under Program	7		
	2.3 Desc	ription of Action Area	13		
	2.4 Ecos	ystems and Habitats Located within the Action Area	18		
	2.4.1	Aquatic Ecosystems and Habitats	18		
	2.4.2	River and Stream Ecosystems and Habitats	18		
	2.4.3	Estuarine Ecosystems and Habitats	19		
	2.4.4	Marine Ecosystem and Habitat	19		
3	ESA-Cons	idered Species Potentially Affected by the Action	19		
	3.1 ESA-	Considered Species and Critical Habitat in the Action Area	21		
	3.2 NMF	S-Listed Species	22		
	3.3 State	-Listed Species	22		
4	Baseline C	onditions	22		
	4.1 Regu	latory Baseline	22		
	4.1.1	Federal Wetland Regulations	22		
	4.1.2	Individual Permits	23		
	4.1.3	General Permits	24		
	4.1.4	Nationwide Permits	24		
	4.1.5	Regional General Permits	24		
	4.1.6	Programmatic General Permits	25		
	4.1.7	Florida Wetlands Regulations	25		
	4.2 Envir	ronmental Baseline	27		
	4.2.1	Coastal Florida	27		
	4.2.2	Wetlands	27		
	4.2.3	Central and Southern Florida Project	27		

	4.2.4	Everglades Restoration			
	4.2.5	Historical Federal Permitting: Habitat	29		
	4.2.6	Freshwater Non-forested Wetlands—Palustrine Wetlands including Pa Emergent and Palustrine Shrub Scrub	lustrine 32		
	4.2.7	Freshwater Forested Wetlands			
	4.2.8	Lakes			
	4.2.9	Rivers and Streams			
	4.2.10	Marine			
	4.2.11	Uplands			
	4.2.12	Inventories and Surveys for Habitat Types and Quantities			
	4.2.13	Historical Federal Permitting: ESA Consultations			
5	Potential E	ffects on ESA-Considered Species	37		
	5.1 Type	s of Effects			
	5.1.1	Biotic Stressors			
	5.1.2	Physical Stressors			
	5.1.3	Physicochemical Water Quality Stressors	40		
	5.2 Potential Effects				
	5.2.1 Mammals				
	5.2.2	Reptiles	41		
	5.2.3	Fish	41		
	5.2.4	Coral	41		
	5.2.5	Mollusks	41		
6	Cumulative	e Effects	41		
	6.1 Effec	ts of Non-Federal Activities	42		
	6.1.1	Watershed Development	42		
	6.1.2	Increased Water Use	45		
	6.1.3	Climate Change	47		
	6.1.4	Summary of Cumulative Effects of Non-Federal Activities	48		
7	State 404 F	Program NMFS Coordination	48		
	7.1 Technical Assistance between the State and NMFS				
	7.1.1	Protected Resources Division	52		
	7.2 State	404 Program Application Review Process	52		
	7.3 Revie	ew for Impacts to NMFS Species	54		
	7.3.1	NMFS Programmatic Biological Opinions	54		
	7.3.2	Case by Case Assessments When Opinions Are Not Available	55		
	7.3.3	Impact/Effect Determinations and Protective Measures	56		

	7	7.3.4 Key Commitments for the NMFS-FWC-DEP Coordination Process	58
	7.4	EPA Oversight and Review	60
	7.5	Elevation of Issues and Dispute Resolution between FDEP, NMFS, and EPA	62
8	Cone	clusions	63
	8.1	Effect of the Action on Tribal Resources or Interests	64
	8.2	Uncertainty Associated with the Effects Determination	65
	8.3	Overall Conclusions	65
9	Liter	rature Cited	66

## Figures

Figure 2-1.	Example of activities authorized by the USACE 404 program1	5
Figure 2-2.	Example of activities authorized by USACE and FDEP1	6
Figure 2-3.	Example of a linear project1	7
Figure 4-1	Historic freshwater flows compared to freshwater flows after C&SF Project (Source USACE and SFWMD 2007)	: 8
Figure 4-2.	Future ecosystem conditions based on CERP (Source: USACE and SFWMD 2007).	29
Figure 4-3.	Acreage of authorized fill, by wetland type	2
Figure 6-1	Comparison of projected 2010-2070 population change in four Florida regions (Source: 1000 Friends of Florida et al. 2017)4	3
Figure 6-2	A comparison of the State development scenarios (Source: 1000 Friends of Florida e al. 2016)4	et 4
Figure 6-3.	State water scenarios (Source: 1000 Friends of Florida et al. 2016)4	6
Figure 7-1.	Species Coordination Process Flowchart	51
Figure 7-1.	Species coordination overview	52

## Tables

Table 4-1. A	Authorized Areas and Acreages by Wetland Type, 2014–2018
Table 4-2a.	Consultation Types for Federal Actions for NMFS Species; Totals, FY 2014 – 2018 
Table 4-2b.	Consultation Types for Federal Actions by Year for NMFS Species, FY 2014 – 2018 
Table 4-3a.	ESA Consultations for NMFS Species without Programmatic, By Species, 2014-2018
Table 4-3b.	ESA Consultations for NMFS Species with Programmatic, 2014-2018
Table 6-1	Historic Water Use in Florida (millions of gallons per day)46

## Appendices

Appendix A April 15, 2020, Letter from NMFS to FDEP Regarding NMFS Jurisdiction	70
Appendix B Species Accounts	73
Appendix C Effects of the Action on ESA-Listed Species	92
Appendix D Figures 1 and 2	99
Appendix E Dredge and Fill Activities as Defined in the Corps Database	102
Appendix F USACE Jacksonville District Programmatic Biological Opinion	.118

## Acronyms

•

°F	degrees Fahrenheit
BA	Biological Assessment
BE	Biological Evaluation
BMP	Best Management Practice
BiOp	Biological Opinion
C&SF	Central and Southern Florida
CERP	Comprehensive Everglades Restoration Plan
CFR	Code of Federal Regulations
CWA	Clean Water Act
DPS	Distinct Population Segment
EPA	U.S. Environmental Protection Agency
ERP	Environmental Resource Permit
ESA	Endangered Species Act of 1973
F.A.C.	Florida Administrative Code
F.S.	Florida Statutes
FDEP	Florida Department of Environmental Protection
FEMA	Federal Emergency Management Agency
FLCCS	Florida Land Cover Classification System
FNAI	Florida Natural Areas Inventory
FWC	Florida Fish and Wildlife Conservation Commission
GIS	Geographic Information Systems
НСР	Habitat Conservation Plan
ITS	Incidental Take Statement
km	kilometers
m	meters
MMPA	Marine Mammal Protection Act
MOA	Memorandum of Agreement

MOU	Memorandum of Understanding
NMFS	National Marine Fisheries Service
NOAA	National Oceanic and Atmospheric Administration
NWP	Nationwide Permit
RAI	Request for Additional Information
USACE	U.S. Army Corps of Engineers
USC	United States Code
WMD	Water Management District
WOTUS	Waters of the United States

### Glossary

Action means all activities or programs of any kind authorized, funded, or carried out, in whole or in part, by federal agencies in the United States or upon high seas (50 CFR § 402.02). For the purposes of this document, the Action is the EPA's approval of the State of Florida's request for the assumption of the administration and permitting of a CWA Section 404 program.

Action Area means all areas to be affected directly or indirectly by the Federal Action and not merely the immediate area involved in the Action (50 CFR § 402.02).

Activity for the purposes of the State 404 program only, means "discharge of dredged material" and/or "discharge of fill material" as those terms are defined in 40 CFR § 232.2 (Rule 62-331.030, F.A.C.).

Administratively complete means an application that contains all the items required under the public noticing requirements of Rule 62-331.060, F.A.C.

Affect/effect as a verb, to "affect" means to bring about a change. The "effect" (usually a noun) is the result of a change. "Affect" appears in Section 7 of the ESA (16 USC § 1536) and "Effect" appears throughout ESA Section 7 regulations (50 CFR § 402 et seq) and guidance documents (ESA Section 7 Consultation Handbook).

Assumed waters (or State-assumed waters) State-assumed waters are all waters of the United States that are not retained waters (as identified in Chapter 62-331, F.A.C., and *FDEP's State 404 Program Applicant's Handbook* (FDEP 2020a)).

**Assumption** means a state has requested and the EPA has approved a state dredge and fill permitting program to be administered in lieu of the Section 404 program administered by the USACE for discharges into assumed waters.

**Avoidance** means mitigating a resource impact by selecting the least damaging project type, spatial location, and extent compatible with achieving the purpose of the project. Avoidance is achieved through an analysis of appropriate and practicable alternatives and a consideration of impact footprint.

**Best available data** means data to ensure the quality of the science used to establish official positions, decisions, and actions taken by the State of Florida during the review of State 404 program permit applications and the quality of the biological, ecological, technical, and other relevant information that is used will be only that which is reliable, credible, and represents the best data available. Under Section 7(a)(2) of the ESA, the NMFS Incidental Take Statement (ITS) is required to use the best available science. In the context of the ESA, the USFWS and NMFS have a policy statement that further describes best available data (see Notice of Interagency Cooperative Policy on Information Standards Under the Endangered Species Act 1994).

**Biological opinion** means a document that includes (1) the opinion of the USFWS or the NMFS as to whether or not a federal action is likely to jeopardize the continued existence of listed species or result in the destruction or adverse modification of designated critical habitat; (2) a summary of information on which the opinion is based; and (3) a detailed discussion of the effects of the action on listed species or designated critical habitat (50 CFR §§ 402.02, 402.14(h)).

**Candidate species** means plant and animal taxa considered for possible addition to the list of endangered and threatened species. These are taxa for which NMFS has on file sufficient information on biological vulnerability and threat(s) to support issuance of a proposal to list, but issuance of a

proposed rule is currently precluded by higher priority listing actions (see 50 CFR § 424.02, ESA Section 7 Consultation Handbook).

**CWA** means the Clean Water Act (also known as the Federal Water Pollution Control Act or FWPCA) Pub. L. 92–500, as amended by Pub. L. 95–217, 33 USC 1251 *et seq.* (Rule 62-331.030, F.A.C., FDEP's State 404 Handbook section 2.0(b)1).

**Conservation** means to use all methods and procedures necessary to bring any endangered species or threatened species to the point at which protective measures are no longer necessary. Such methods and procedures include, but are not limited to, all activities associated with scientific resources management, such as research, census, law enforcement, habitat acquisition and maintenance, propagation, live trapping, and transplantation, and, in the extraordinary case in which population pressures within a given ecosystem cannot be otherwise relieved, may include regulated taking (ESA Section 7; 16 USC § 1532(3)).

**Critical habitat** for a threatened or endangered species means (1) the specific areas within the geographical area occupied by the species, at the time it is listed in accordance with the provisions of 16 CFR Part 1533, on which are found the physical or biological features essential to the conservation of the species and which may require special management considerations or protection; and (2) specific areas outside the geographical area occupied by the species at the time it is listed in accordance with the provisions of 16 CFR Part 1533, upon a determination by the Secretary of the Interior or the Secretary of Commerce that such areas are essential for the conservation of the species (ESA Section 7; 16 USC § 1532(5)).

**Cumulative effects** are those effects of future State or private activities, not involving federal activities, that are reasonably certain to occur within the action area of the action subject to coordination. For the purposes of this document, this definition applies only to ESA Section 7 analyses (50 CFR § 402.02).

**Destruction or adverse modification of critical habitat** means a direct or indirect alteration that appreciably diminishes the value of critical habitat as a whole for the conservation of a listed species (50 CFR § 402.02).

**Effects of the action** are all consequences to listed species or critical habitat that are caused by the proposed action, including the consequences of other activities caused by the proposed action. A consequence is caused by the proposed action if it would not occur but for the proposed action and it is reasonably certain to occur. Effects of the action may occur later in time and may include consequences occurring outside the immediate area involved in the action (50 CFR § 402.02).

**Endangered species** means any species in danger of extinction throughout all or a significant portion of its range other than a species of Class Insecta determined by the Secretary to constitute a pest whose protection under the provisions of the ESA would present an overwhelming and overriding risk to man (ESA Section 7; 16 USC § 1532(6))

**Endangered or threatened species** in this document refers to those animal species identified as federally endangered or threatened by the USFWS or NMFS and those animal species identified as State-listed by the FWC; it also means plant species identified as endangered or threatened by the NMFS or by the Florida Department of Agriculture and Consumer Services when such plants are located in a wetland or other surface water (Rule 62-330.021, F.A.C.).

**Environmental baseline** refers to the condition of the listed species or its designated critical habitat in the action area, without the consequences to the listed species or designated critical habitat caused by the proposed action. The environmental baseline includes the past and present impacts of all federal, State, or private actions and other human activities in the action area; the anticipated impacts of all proposed federal projects in the action area that have already undergone formal or early ESA Section 7 consultation; and the impact of State or private actions that are contemporaneous with the consultation in process. The consequences to listed species or designated critical habitat from ongoing agency activities or existing agency facilities that are not within the agency's discretion to modify are part of the environmental baseline (50 CFR § 402.02).

**ESA-species of concern** refers to species that are either listed, proposed, petitioned, or considered candidates for listing under the ESA.

**Fish or wildlife** means any member of the animal kingdom, including without limitation any mammal, fish, bird (including any migratory, nonmigratory, or endangered bird for which protection is also afforded by treaty or other international agreement), amphibian, reptile, mollusk, crustacean, arthropod, or other invertebrate, and includes any part, product, egg, or offspring thereof, or the dead body or parts thereof. (ESA Section 7; 16 USC § 1532(8)).

**Incidental take** refers to takings that result from, but are not the purpose of, carrying out an otherwise lawful activity conducted by a federal agency or applicant (50 CFR § 402.02).

**Jeopardize the continued existence of** means to engage in an action that reasonably would be expected, directly or indirectly, to reduce appreciably the likelihood of both survival and recovery of a listed species in the wild by reducing the reproduction, numbers, or distribution of that species (50 CFR § 404.02).

**Listed species** refers to any species of fish, wildlife, or plant determined to be endangered or threatened under ESA Section 4 or under Chapter 68A-27, F.A.C.

**May affect** is the appropriate conclusion made by a federal action agency in the context of the ESA or by the State in the context of the State 404 program when a proposed activity is reasonably certain to affect any ESA-listed species or designated critical habitat.

**Minimization** means mitigating an aquatic resource impact by managing the severity of a project's impact on resources at the selected site. Minimization is achieved through the incorporation of appropriate and practicable design and risk avoidance measures.

**No effect** is the appropriate conclusion when the action agency determines its action will not affect a listed species or designated critical habitat per the ESA.

**Practicable** means available and capable of being done after taking into consideration cost, existing technology, and logistics considering overall project purposes (Rule 62-331.030, F.A.C.).

**Programmatic consultation** (under ESA implementing regulations in 50 CFR § 402.02) is a consultation addressing an agency's multiple actions on a program, region, or other basis. Programmatic consultations allow the NMFS to consult on the effects of programmatic actions such as:

• Multiple similar, frequently occurring, or routine actions expected to be implemented in particular geographic areas; and

• A proposed program, plan, policy, or regulation providing a framework for future proposed actions.

**Project area** or **Project site** means a portion of the State-assumed waters where specific dredging or filling activities are permitted and consists of a bottom surface area, any overlying volume of water, and any mixing zones. In the case of wetlands on which surface water is not present, the project area consists of the wetland surface area (Rule 62-331.030, F.A.C.). In the context of the review of State 404 permit applications for endangered and threatened species, also includes those areas outside the immediate area of activity that may affect listed species using those areas.

**Proposed species** means any species of fish, wildlife or plant that is proposed in the *Federal Register* to be listed under ESA Section 4 (50 CFR § 402.02).

**Protection measures** means those avoidance and minimization measures to address adverse impacts on listed species and critical habitat under the State 404 program. Protection measures recommended by the NMFS are incorporated as conditions into the State 404 permit. Examples of protection measures include, but are not limited to, project design changes and operational restrictions for the protection of species (i.e., seasonal restrictions for construction work) (as used in Chapter 62-331, F.A.C.).

**Reasonable potential to affect** for the purposes of this document and for the State of Florida's 404 program refers to a project that has a reasonable potential for affecting endangered or threatened species (40 CFR § 233.51(b)(2)) when it has been determined during the species coordination process that the project may affect federally listed species or their critical habitat.

**Retained waters** means those waters that are presently used or are susceptible to use in their natural condition or by reasonable improvement as a means to transport interstate or foreign commerce shoreward to their ordinary high-water mark, including all waters subject to the ebb and flow of the tide shoreward to their mean high-water mark, such wetlands adjacent thereto. The USACE will retain responsibility for permitting for the discharge of dredged or fill material in those waters identified in the Retained Waters List (Appendix A of FDEP's State 404 Handbook) as well as all waters subject to the ebb and flow of the tide shoreward to their mean high-water mark that are not specifically listed in the Retained Waters List, including wetlands adjacent thereto landward to the administrative boundary. The administrative boundary demarcating the adjacent wetlands over which jurisdiction is retained by the USACE is a 300-foot guideline established from the ordinary high-water mark or mean high tide line of the retained water. In the case of a project that involves discharges of dredged or fill material both waterward and landward of the 300-foot guideline, the USACE will retain jurisdiction to the landward boundary of the project for the purposes of that project only (Rule 62-331.030, F.A.C.).

Section 7 consultation refers to ESA Section 7(a)(2) that requires federal agencies to use their authorities to further the conservation of listed species, including the requirement to consult with the NMFS to ensure that they are not undertaking, funding, permitting, or authorizing actions likely to jeopardize the continued existence of listed species or destroy or adversely modify designated critical habitat.

Section 404 is a section of the federal CWA that establishes a program to regulate the discharge of dredged and fill material into the WOTUS.

Services(s) describes the USFWS and/or the NMFS.

**Species coordination** is a process to address potential adverse effects on threatened or endangered species, ensuring compliance with Chapter 62-331, F.A.C. and the ESA. This process includes coordination between FDEP, FWC, and NMFS during the review of submitted State 404 permit applications. Recommendations for avoiding and minimizing the effects of a project on federally listed species and their critical habitat is provided by technical assistance from the NMFS.

**Species coordination lead or State species lead** is the designated FDEP or FWC staff person who coordinates with the NMFS and is the point of contact for all ESA species issues during the review of a specific State 404 application. FDEP and FWC staff will decide on a project-by-project basis which agency will act as the State's species lead on the project when coordinating with the NMFS. Factors that will be considered in this decision include the complexity of the coordination and relative workloads.

**State 404 program** is the FDEP program (Chapter 62-331, F.A.C.) that fulfills the requirements of CWA Section 404 in a similar manner as the USACE 404 program, post-assumption, if approved by EPA.

**Stream** means any river, creek, slough, or natural watercourse in which water usually flows in a defined bed or channel. It is not essential that the flowing be uniform or uninterrupted. The fact that some part of the bed or channel shall have been dredged or improved does not prevent the watercourse from being a stream (§ 373.019(20), F.S.).

**Stressors** are any physical, chemical, or biological alteration of resources (i.e., increase, decrease, or introduction) that can induce an adverse organism response. Stressors can act directly on an individual or indirectly through impacts on resources.

**Surface water** means water upon the surface of the Earth, whether contained in bounds created naturally or artificially or diffused. Water from natural springs shall be classified as surface water when it exits from the spring onto the Earth's surface (§ 373.019(21), F.S.).

**Take** means to harass, harm, pursue, hunt, shoot, wound, kill, trap, capture, or collect or attempt to engage in any such conduct. (ESA Section 7; 16 USC § 1532(19)).

- **Harass** is further defined as actions that create the likelihood of injury to listed species to such an extent as to significantly disrupt normal behavior patterns that include, but are not limited to, breeding, feeding, and sheltering.
- **Harm** is further defined to include significant habitat modification or degradation that results in death or injury to listed species by significantly impairing behavioral patterns, such as breeding, feeding, or sheltering (ESA implementing regulations in 50 CFR § 17.3).

**Technical assistance** refers to a coordination process described in the ESA Section 7 Consultation Handbook (1998) that outlines a variety of ways in which the NMFS provides expertise and guidance on an individual project basis. In the context of State 404 permit application reviews, the NMFS assists the State of Florida by providing information, reviews, and recommendations on effect determinations and protective measures to ensure compliance with the ESA and Chapter 62-331, F.A.C.

**Technically complete** means a State 404 application on which each item is sufficiently complete to enable FDEP to determine if the proposed project complies with Chapter 62-331, F.A.C. If a project requires both an ERP and a State 404 program authorization, the State 404 program review shall not

be considered complete until the ERP review is complete. This is to satisfy the requirement for reasonable assurance that State water quality standards and coastal zone consistency requirements will be met (Rule 62-331.030, F.A.C.).

**Threatened species** means any species that is likely to become an endangered species within the foreseeable future throughout all or a significant portion of its range (ESA Section 7; 16 USC § 1532(20)).

**Uplands** means areas that are not wetlands or other surface waters, as delineated pursuant to Rules 62-340.100 through 62-340.550, F.A.C., as ratified by § 373.4211, F.S.

**USACE 404 program** refers to the administration by and permitting responsibilities of the USACE for CWA Section 404 prior to any assumption by the State of Florida.

Waters of the State are as defined in § 403.031(13), F.S.

**Waters of the United States (WOTUS)** means those waters defined in the regulations in 40 CFR § 120.2.

**Wetlands** means areas inundated or saturated by surface or ground water at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions. Wetlands generally include swamps, marshes, bogs, and similar areas (40 CFR § 120.2).

Wildlife Plan refers to Florida's State Wildlife Action Plan (FWC 2019) prepared by FWC.

**Works** means all artificial structures, including, but not limited to, ditches, canals, conduits, channels, culverts, pipes, and other construction that connects to, draws water from, drains water into, or is placed in or across the waters in the State (§ 373.403(5), F.S.) and includes all types of dredging and filling to create, remove, or locate structures in, on, or over wetlands or other surface waters (Rule 62-330.021, F.A.C.).

## 1 Purpose of the Biological Evaluation

The purpose of this Biological Evaluation (BE) is to evaluate the effects of the approval by the U.S. Environmental Protection Agency (EPA) of the State of Florida's request to assume administration of a Clean Water Act (CWA) Section 404 program. The BE considers whether EPA's approval of the assumption request (Action) might affect listed species or designated critical habitat within the Action Area, including whether the Action is likely to adversely affect species listed or proposed to be listed under the Endangered Species Act of 1973 (ESA) or designated critical habitat and proposed critical habitat for those species in retained waters downstream of waters assumed by the State (Title 50 *Code of Federal Regulations* [CFR] § 402.12). The EPA is the federal agency that approved the State's request, pursuant to the CWA implementing regulations (40 CFR Part 233). The Florida Department of Environmental Protection (FDEP) is the state agency that requested administration of a CWA Section 404 program (assumption).

Florida's assumption of a CWA Section 404 program includes only those WOTUS referred to as "assumed waters" or "State-assumed waters," not those retained by the U.S. Army Corps of Engineers (USACE). The USACE retains permitting responsibility for the discharges of dredged or fill material into those waters defined as "Retained Waters." The definition of retained waters can be found in FDEP's *State 404 Program Applicant's Handbook* in Chapter 2.0, and the process to determine whether a project is located in retained or assumed waters is described in Chapter 4.1 of the Handbook (FDEP 2020a). In addition, Appendix A of the Handbook includes the Retained Waters List maintained by the USACE (FDEP 2020a).

### 1.1 Objectives of Action

By assuming responsibilities under Section 404 of the CWA, the State of Florida is able to have an active, major role in the implementation of the CWA within its jurisdiction. The State can address regional issues by permitting discharges of dredged or fill material through the Florida Environmental Resource Permit (ERP) program per Chapter 62-330, Florida Administrative Code (F.A.C.) and State 404 program per Chapter 62-331, F.A.C. authorizations.

For Florida to assume administration of a CWA Section 404 program, pursuant to the provisions of 40 CFR § 233.10, the State submitted a request to assume the program to the EPA Region 4 Administrator which included the following items:

- Letter from the Governor of Florida requesting program approval;
- Complete program description as set forth in 40 CFR § 233.11;
- Attorney General's statement, as set forth in 40 CFR § 233.12;
- Memorandum of Agreement (MOA) with the EPA Regional Administrator, as set forth in 40 CFR § 233.13;
- MOA with the Secretary of the Army, as set forth in 40 CFR § 233.14; and
- Copies of all applicable State statutes and regulations, including those governing applicable State administrative procedures.

The State's assumption of the dredge and fill permitting program under Section 404 of the CWA in waters assumed by the State is implemented by processes and procedures described in State regulations (Chapters 62-330 and 62-331, F.A.C.), MOAs with the EPA and the USACE, and an agreement among Florida Fish and Wildlife Conservation Commission (FWC), and the U.S. Fish and Wildlife Service, and an anticipated agreement among FDEP, FWC, and the National Marine Fisheries Service (NMFS). The EPA is requesting initiation of consultation with NMFS on the EPA's approval of Florida's request to assume administration of the CWA Section 404 program and seeks NMFS's concurrence on its effects determinations for the Atlantic Sturgeon, Shortnose Sturgeon, and smalltooth sawfish, set forth in this BE.

#### 1.1.1 The Programmatic Biological Evaluation

In accordance with ESA Section 7 regulations, this BE evaluates effects only on ESA-listed species and species proposed to be listed and their critical habitat or proposed critical habitat. For the sake of clarity in understanding the State's implementation of the State 404 program and the ERP program, the State 404 program addresses species that might not be listed under the ESA but are protected under other federal or State laws. Examples of these laws include the Bald and Golden Eagle Protection Act (Title 16 *United States Code* [USC] § 668 *et seq.*) and the Migratory Bird Treaty Act of 1918 (16 USC §§ 703–712 (§ 709 omitted)).

Additionally, ESA proposed, candidate, and petitioned plant and animal species with ranges within Florida are discussed in this BE to provide a thorough review of Florida's potentially atrisk species, and because the lists of species that are listed as threatened may be modified to include some or all of these petitioned species, depending on the outcome of the status review. This BE analyzes a total of 28 animal species in the State of Florida that are federally listed as endangered (15), threatened (12), or proposed for consideration (one). The effects of the approval by the EPA for the State of Florida assuming administration of a CWA Section 404 program on listed species are discussed for present time and estimated for the future. All species that are currently listed, proposed to be listed, or might be listed in the future are discussed, with evaluations of effects on respective guilds. This review helps evaluate the effects of the State 404 program on any species currently listed or those EPA is aware of that might become listed in the next 10–20 years.

Because the State 404 program is implemented through State-issued 404 permit actions, a programmatic BE is appropriate to describe the regulatory process by which the State of Florida will make determinations on the issuance of State 404 permits. The BE addresses how that process considers effects on listed endangered and threatened species or species proposed to be listed so that the effects of this Action may be analyzed.

This BE addresses the programmatic impacts and effects of the State of Florida's 404 permitting program. More specifically, the components of the State 404 program establish the process and responsibilities for FDEP to follow to effectively implement the State 404 program. The programmatic consultation the EPA is requesting would examine whether and to what degree the EPA's approval of the State 404 program ensures that implementation of the program is not likely to jeopardize the continued existence of endangered or threatened species or result in the destruction or adverse modification of critical habitat.

As recently recognized by the USFWS, "various forms of programmatic consultations have been successfully implemented for many years now" (84 CFR Part 44996).

Assumption under CWA Section 404 is guided by regulations in 40 CFR Part 233, which contain specific requirements regarding the issuance of permits by a State 404 program. The State of Florida's 404 program (Chapter 62-331, F.A.C.) has a structure and technical assistance process that requires a review of species listed by the ESA at the time permit applications are received. This approach ensures that, if a new species is listed or a listed species' status changes between this ESA consultation and the time of permit application, the effects of any State 404 permit on the new species or its status change would be reviewed by the NMFS. In addition, by taking a broad look at endangered, threatened, proposed, petitioned, and candidate species, this BE (and any subsequent Biological Opinion (BiOp) that NMFS may issue) will cover a broad range of species and habitat that might be listed or designated in the future, making for a more efficient permit review process. If the State 404 program's processes are proposed to be changed significantly in the future, those changes would be submitted to the EPA for review before implementation and might trigger reinitiation of consultation with the NMFS.

### 1.2 The History of Florida's Request for Assumption

On March 23, 2018, the Florida Governor signed into law Rule 2018-88, Laws of Florida, granting FDEP the power and authority to adopt rules to assume and implement the Section 404 dredge and fill program (creating § 373.4146, Florida Statutes [F.S.]). FDEP then prepared Chapter 62-331, F.A.C. to implement the State 404 program and MOA with the EPA and the USACE, as required by 40 CFR § 233.13. In addition, a Memorandum of Understanding (MOU) was prepared among FDEP, FWC, and USFWS to better facilitate post-assumption implementation.

On July 17, 2019, FDEP sent a request to the EPA that sought designation, pursuant to 50 CFR § 402.08, to serve as a non-federal representative to prepare a Biological Assessment (BA) for possible ESA Section 7 consultation. The EPA designated FDEP as the non-federal representative to prepare a BA in a letter dated December 12, 2019, consistent with 50 CFR § 402.08. This designation allowed FDEP to consult informally with the USFWS and the NMFS, to coordinate on the list of species that might be affected and on potential processes for the review of State 404 permits.

A brief sequence of recent events and actions by the State of Florida pertaining to the assumption are as follows:

- On March 23, 2018, the Governor signed into law Rule 2018-88, Laws of Florida, which created § 373.4146, F.S., granting FDEP with the power and authority to adopt rules to assume and implement a CWA Section 404 dredge and fill program.
- In 2018, FDEP began work with both the EPA and the USACE to draft separate MOAs that describe the commitments and responsibilities of each agency, should the assumption be approved by the EPA. FDEP also began assembly of other required components that would constitute a complete assumption request package per 40 CFR § 233.10-14(b).
- In May 2018, FDEP published a Notice of Rule Development to implement the State 404 program and held three rulemaking workshops to collect public comment on the draft rule, Chapter 62-331, F.A.C. The workshops were held on May 5, May 31, and June 1,

2018. This rule has been created to implement the State 404 program and to include federal requirements that are not currently covered under the ERP program.

- On July 17, 2019, FDEP sent a request to the EPA that sought designation, pursuant to 50 CFR § 402.08, to serve as a non-federal representative for ESA Section 7 consultation to prepare a BA.
- On November 22, 2019, FDEP sent a request to USFWS and NMFS to review a preliminary list of affected species for the BA.
- On December 12, 2019, FDEP Secretary received a response from the EPA Region 4 Administrator approving the requested non-federal representative designation, allowing FDEP to move forward with the development of a BA. The letter indicated that, at the request of FDEP, the EPA would voluntarily engage in consultation with the Services on approving Florida's program.
- In February 2020, FDEP published a Notice of Proposed Rule to implement the State 404 program and in April, held five rulemaking hearings to collect public comment on the draft rule, Chapter 62-331, F.A.C. The hearings occurred on April 2, April 6, April 10, April 24, and April 27, 2020.
- In February 2020, FDEP, FWC, and USFWS began drafting an MOU to describe commitments, roles, and responsibilities in the State 404 program's species coordination process, should the assumption be approved by the EPA.
- Additional public hearings were held in March and April 2020 regarding rulemaking for Chapter 62-331, F.A.C.
- On April 15, 2020, NMFS responded to FDEP's November 22, 2019, letter with the conclusion that ESA-listed species under NMFS's jurisdiction do not occur in waters that are assumable by the State.
- On June 8, 2020, the Final Notice of Proposed Rule Change for Chapter 62-331, F.A.C. was published.
- On Dec. 22, 2020, the EPA published its approval of Florida's State 404 Program in the Federal Register.
- On March 31, 2021, the EPA received 60-Day Notice of Intent to Sue for Violations of the Endangered Species Act in Connection with EPA's Approval of Florida's Assumption of a Clean Water Act Section 404 Program.
- On December 20, 2021, the EPA responded to the NOI stating, "the EPA intends to initiate an ESA consultation [with NMFS] as appropriate."

### **1.3 Organization of the Document**

This BE is organized as follows:

- **Chapter 1** provides an overview of the objectives of the Action, whereby the State of Florida proposed to assume the administration of Section 404 of the CWA in those waters not retained by the USACE and offers a history (timeline) of actions leading up to FDEP's request for assumption.
- Chapter 2 describes the request for assumption and the Action Area. It also identifies the ecosystems and habitats evaluated in the remainder of the document.

- **Chapter 3** identifies ESA-listed species evaluated in the remainder of the document and describes sources of information for the ESA-listed, NMFS-listed, and State-listed species and critical habitat within the Action Area.
- **Chapter 4** develops the baseline environmental conditions prior to the Action. Further, it provides a brief history of the regulatory framework prior to (and leading up to) the Action.
- Chapter 5 analyzes the potential effects of the Action on ESA-listed species and designated critical habitat.
- Chapter 6 considers the cumulative effect of non-federal actions that are reasonably certain to occur within the Action Area in the foreseeable future.
- **Chapter 7** discusses species coordination, avoidance, and minimization measures and obtaining NMFS technical assistance as well as various tools and guidelines for species review and decision-making.
- Chapter 8 provides an effects determination for ESA-listed species and their critical habitat.
- Chapter 9 provides a listing of the references used to prepare this document.

The April 15, 2020, letter from NMFS to FDEP stating that it does not have assets (ESA-listed species) that might be affected by this Action is provided in Appendix A. Species accounts on those ESA-listed, candidate, or under review species in Florida potentially affected by the Section 404 assumption are contained in Appendix B. Additional information pertaining to the effects of the Action on ESA-listed species, map figures, and a list of dredge and fill activities as described in the USACE database are contained in appendices C, D, and E, respectively.

## 2 Action and Action Area

### 2.1 Description of Action

This BE analyzes the EPA Action, which is the approval of Florida's assumption of administration of a CWA Section 404 program in State-assumed waters. The State of Florida's request for assumption was approved; thus, FDEP assumed regulatory responsibility over all dredging and filling activities in WOTUS not retained by USACE pursuant to 33 USC § 1344(g) (see Chapter 2.3 of this document).

Per CWA Section 404(g) (33 USC § 1344(g)), a state, with approval from the EPA, may be authorized to administer its own permit program for the discharge of dredge or fill material into certain WOTUS in lieu of the permitting program implemented by the USACE. The EPA has promulgated regulations in 40 CFR Part 233 outlining, among other things, its requirements for approving a State 404 program.

The Secretary of FDEP is the State official charged with administering the State 404 program in accordance with 40 CFR Part 233 and has authority to issue permits pursuant to Part IV of § 373, F.S. In accordance with § 373.4146, F.S., FDEP has the power and authority to issue permits for regulated activities conducted in State-assumed waters.

FDEP, the water management districts (WMDs), and certain local governments delegated by FDEP jointly implement Florida's ERP program. The agencies' responsibilities are divided

according to the Operating and Delegation Agreement (operating agreements) and the geographic regions of the WMDs. At this time, the administration and implementation of the State 404 program is limited to FDEP. It is anticipated that the State 404 program might be delegated to the WMDs by FDEP in the future. If this is considered, approval from the EPA will be required prior to implementation. Upon any such approval, these agencies will be required to follow the same procedures outlined for the State 404 program.

Florida's State 404 program will regulate discharges of dredge or fill material in those areas where USACE does not retain jurisdiction (assumed waters). The EPA's review of the proposed State 404 program included Chapter 62-331, F.A.C., which includes requirements of federal law that are not addressed in the currently effective State regulations for dredge and fill permitting (e.g., noticing provisions, alternatives analysis, and the federal mitigation hierarchy). The State 404 rule, Chapter 62-331, F.A.C., also includes definitions, procedures for review and agency action on exemption requests, processes for individual permits, public notice requirements, procedures regarding mitigation banking, and procedures and descriptions for general permits created to correspond to the federal nationwide permits (NWPs) and appropriate regional general permits administered by the USACE. The EPA's review of this program also included FDEP's State 404 Handbook and new forms, incorporated by rule reference.

### 2.2 Types of Activities Authorized by Florida's 404 Program

This subchapter discusses the different types of future proposed activities that might be authorized by State 404 program permits. It includes a list of the types of future activities that might be authorized under the Section 404 assumption. A more expansive list is available in Appendix E, which was derived from the USACE project information database. All the activities in the Appendix E list, however, may not be applicable to the State 404 program.

Dredging and filling activities include, but are not limited to, the following:

- Discharge of fill material
- Dredging
- Ecological restoration
- Discharge of dredged material
- Excavation associated with the discharge of dredged or fill material
- Other (aquaculture, work, aerial or submarine cable crossings)
- Conversion of waters type (forested wetland to emergent wetland, stream to lake)
- Commercial developments
- Residential developments
- Single-family residence

- Agriculture
- Utilities
- Roadways
- Airports
- Marinas
- Docks
- Piers
- Boat Ramps
- Dams
- Levees
- Mining activities
- Mitigation
- Restoration

The focus of this BE is on potential impacts to NMFS species located within the Action Area, including within retained waters downstream of the assumed waters. These waters with respect to marine and estuarine species include coastal waters, inlets, bays, and other areas where marine and freshwaters mix. The extent of freshwater habitats that are evaluated here are only for three

listed marine anadromous species: the shortnose sturgeon, Atlantic sturgeon, and smalltooth sawfish. These three species may inhabit certain freshwaters in Florida during certain times of the year and certain phases of their life cycles.

### 2.2.1 Proposed Activities Authorized or Exempt Under Program

Types of activities the State 404 program will regulate include all dredge and fill activities within the State-assumed waters. Construction activities, including excavation and filling of wetlands, might impact ESA-listed species and habitat occurring in those areas. Other effects associated with dredge and fill construction include, but are not limited to, turbidity, decreased water quality, noise, pollution (including exhaust and new sources of light), short- or long-term hydrodynamic changes in the area and its surroundings, and changes in wet and dry periodicity. Permitted projects might have adverse impacts, and those impacts will be taken into consideration during the permitting process, with protection measures identified and implemented to avoid or reduce those impacts. The following proposed activities and exemptions are excerpts from 40 CFR Part 232 and select definitions from 40 CFR § 232.2, also summarized in Appendix B of FDEP's State 404 Handbook (FDEP 2020a):

#### Discharge of dredged material.

- (1) Except as provided below in paragraph (2), the term "discharge of dredged material" means any addition of dredged material into, including redeposit of dredged material other than incidental fallback within, the waters of the United States. The term includes, but is not limited to, the following:
  - (i) The addition of dredged material to a specified discharge site located in waters of the United States;
  - (ii) The runoff or overflow, associated with a dredging operation, from a contained land or water disposal area; and
  - (iii)Any addition, including redeposit other than incidental fallback, of dredged material, including excavated material, into waters of the United States which is incidental to any activity, including mechanized land clearing, ditching, channelization, or other excavation.
- (2) The term discharge of dredged material does not include the following:
  - (i) Discharges of pollutants into waters of the United States resulting from the onshore subsequent processing of dredged material that is extracted for any commercial use (other than fill). These discharges are subject to Section 402 of the Clean Water Act even though the extraction and deposit of such material may require a permit from the Corps or applicable state.
  - (ii) Activities that involve only the cutting or removing of vegetation above the ground (e.g., mowing, rotary cutting, and chain sawing) where the activity neither substantially disturbs the root system nor involves mechanized pushing, dragging, or other similar activities that redeposit excavated soil material.
  - (iii)Incidental fallback.
- (3) Section 404 authorization is not required for the following:
  - (i) Any incidental addition, including redeposit, of dredged material associated with any activity that does not have or would not have the effect of destroying or degrading an area of waters of the U.S. as defined in paragraphs (4) and (5) of this definition;

however, this exception does not apply to any person preparing to undertake mechanized land clearing, ditching, channelization and other excavation activity in a water of the United States, which would result in a redeposit of dredged material, unless the person demonstrates to the satisfaction of the Corps, or the EPA as appropriate, prior to commencing the activity involving the discharge, that the activity would not have the effect of destroying or degrading any area of waters of the United States, as defined in paragraphs (4) and (5) of this definition. The person proposing to undertake mechanized land clearing, ditching, channelization, or other excavation activity bears the burden of demonstrating that such activity would not destroy or degrade any area of waters of the United States.

- (ii) Incidental movement of dredged material occurring during normal dredging operations, defined as dredging for navigation in navigable waters of the United States, as that term is defined in 33 CFR part 329, with proper authorization from the Congress or the Corps pursuant to 33 CFR part 322; however, this exception is not applicable to dredging activities in wetlands, as that term is defined at §232.2® of this chapter.
- (iii)Certain discharges, such as those associated with normal farming, silviculture, and ranching activities, are not prohibited by or otherwise subject to regulation under Section 404. See 40 CFR § 232.3 for discharges that do not require permits.
- (4) For purposes of this section, an activity associated with a discharge of dredged material destroys an area of waters of the United States if it alters the area in such a way that it would no longer be a water of the United States.

Note: Unauthorized discharges into waters of the United States do not eliminate Clean Water Act jurisdiction, even where such unauthorized discharges have the effect of destroying waters of the United States.

(5) For purposes of this section, an activity associated with a discharge of dredged material degrades an area of waters of the United States if it has more than a de minimis (i.e., inconsequential) effect on the area by causing an identifiable individual or cumulative adverse effect on any aquatic function.

#### Discharge of fill material.

(1) The term "discharge of fill material" means the addition of fill material into waters of the United States. The term generally includes, without limitation, the following activities: Placement of fill that is necessary for the construction of any structure or infrastructure in a water of the United States; the building of any structure, infrastructure, or impoundment requiring rock, sand, dirt, or other material for its construction; site-development fills for recreational, industrial, commercial, residential, or other uses; causeways or road fills; dams and dikes; artificial islands; property protection and/or reclamation devices such as riprap, groins, seawalls, breakwaters, and revetments; beach nourishment; levees; fill for structures such as sewage treatment facilities, intake and outfall pipes associated with power plants and subaqueous utility lines; placement of fill material for construction or maintenance of any liner, berm, or other infrastructure associated with solid waste landfills; placement of overburden, slurry, or tailings or similar mining-related materials;" after the words "utility lines; and artificial reefs.

- (2) In addition, placement of pilings in waters of the United States constitutes a discharge of fill material and requires a Section 404 permit when such placement has or would have the effect of a discharge of fill material. Examples of such activities that have the effect of a discharge of fill material include, but are not limited to, the following: Projects where the pilings are so closely spaced that sedimentation rates would be increased; projects in which the pilings themselves effectively would replace the bottom of a waterbody; projects involving the placement of pilings that would reduce the reach or impair the flow or circulation of waters of the United States; and projects involving the placement of pilings which would result in the adverse alteration or elimination of aquatic functions.
  - (i) Placement of pilings in waters of the United States that does not have or would not have the effect of a discharge of fill material shall not require a Section 404 permit. Placement of pilings for linear projects, such as bridges, elevated walkways, and powerline structures, generally does not have the effect of a discharge of fill material. Furthermore, placement of pilings in waters of the United States for piers, wharves, and an individual house on stilts generally does not have the effect of a discharge of fill material. All pilings, however, placed in the navigable waters of the United States, as that term is defined in 33 CFR part 329, require authorization under Section 10 of the Rivers and Harbors Act of 1899 (see 33 CFR part 322).

#### 40 CFR § 232.3 Activities not requiring permits.

Except as specified in paragraphs (a) and (b) of this section, any discharge of dredged or fill material that may result from any of the activities described in paragraph (c) of this section is not prohibited by or otherwise subject to regulation under this part.

- (a) If any discharge of dredged or fill material resulting from the activities listed in paragraph (c) of this section contains any toxic pollutant listed under Section 307 of the Act, such discharge shall be subject to any applicable toxic effluent standard or prohibition and shall require a Section 404 permit.
- (b) Any discharge of dredged or fill material into waters of the United States incidental to any of the activities identified in paragraph (c) of this section must have a permit if it is part of an activity whose purpose is to convert an area of the waters of the United States into a use to which it was not previously subject, where the flow or circulation of waters of the United States may be impaired or the reach of such waters reduced. Where the proposed discharge will result in significant discernable alterations to flow or circulation, the presumption is that flow or circulation may be impaired by such alteration.

Note: For example, a permit will be required for the conversion of a cypress swamp to some other use or the conversion of a wetland from silvicultural to agricultural use when there is a discharge of dredged or fill material into waters of the United States in conjunction with construction of dikes, drainage ditches or other works or structures used to affect such conversion. A conversion of Section 404 wetland to a non-wetland is a change in use of an area of waters of the U.S. A discharge which elevates the bottom of waters of the United States without converting it to dry land does not thereby reduce the reach of, but may alter the flow or circulation of, waters of the United States.

- (c) The following activities are exempt from Section 404 permit requirements, except as specified in paragraphs (a) and (b) of this section:
  - (1)(i) Normal farming, silviculture, and ranching activities such as plowing, seeding, cultivating, minor drainage, and harvesting for the production of food, fiber, and forest products, or upland soil and water conservation practices, as defined in paragraph (d) of this section.

(ii) (A) To fall under this exemption, the activities specified in paragraph (c)(1) of this section must be part of an established (i.e., ongoing) farming, silviculture, or ranching operation, and must be in accordance with definitions in paragraph (d) of this section. Activities on areas lying fallow as part of a conventional rotational cycle are part of an established operation.

(B) Activities which bring an area into farming, silviculture or ranching use are not part of an established operation. An operation ceases to be established when the area in which it was conducted has been converted to another use or has lain idle so long that modifications to the hydrological regime are necessary to resume operation. If an activity takes place outside the waters of the United States, or if it does not involve a discharge, it does not need a Section 404 permit whether or not it was part of an established farming, silviculture or ranching operation.

- (2) Maintenance, including emergency reconstruction of recently damaged parts, of currently serviceable structures such as dikes, dams, levees, groins, riprap, breakwaters, causeways, bridge abutments or approaches, and transportation structures. Maintenance does not include any modification that changes the character, scope, or size of the original fill design. Emergency reconstruction must occur within a reasonable period of time after damage occurs in order to qualify for this exemption.
- (3) Construction or maintenance of farm or stock ponds or irrigation ditches or the maintenance (but not construction) of drainage ditches. Discharge associated with siphons, pumps, headgates, wingwalls, weirs, diversion structures, and such other facilities as are appurtenant and functionally related to irrigation ditches are included in this exemption.
- (4) Construction of temporary sedimentation basins on a construction site which does not include placement of fill material into waters of the United States. The term "construction site" refers to any site involving the erection of buildings, roads, and other discrete structures and the installation of support facilities necessary for construction and utilization of such structures. The term also includes any other land areas which involve land-disturbing excavation activities, including quarrying or other mining activities, where an increase in the runoff of sediment is controlled through the use of temporary sedimentation basins.
- (5) Any activity with respect to which a State has an approved program under Section 208(b)(4) of the Act which meets the requirements of Section 208(b)(4)(B) and (C).
- (6) Construction or maintenance of farm roads, forest roads, or temporary roads for moving mining equipment, where such roads are constructed and maintained in accordance with best management practices (BMPs) to assure that flow and circulation patterns and chemical and biological characteristics of waters of the United States are not impaired,

that the reach of the waters of the United States is not reduced, and that any adverse effect on the aquatic environment will be otherwise minimized. The BMPs which must be applied to satisfy this provision include the following baseline provisions:

- Permanent roads (for farming or forestry activities), temporary access roads (for mining, forestry, or farm purposes) and skid trails (for logging) in waters of the United States shall be held to the minimum feasible number, width, and total length consistent with the purpose of specific farming, silvicultural or mining operations, and local topographic and climatic conditions;
- (ii) All roads, temporary or permanent, shall be located sufficiently far from streams or other water bodies (except for portions of such roads which must cross water bodies) to minimize discharges of dredged or fill material into waters of the United States;
- (iii) The road fill shall be bridged, culverted, or otherwise designed to prevent the restriction of expected flood flows;
- (iv) The fill shall be properly stabilized and maintained to prevent erosion during and following construction;
- (v) Discharges of dredged or fill material into waters of the United States to construct a road fill shall be made in a manner that minimizes the encroachment of trucks, tractors, bulldozers, or other heavy equipment within the waters of the United States (including adjacent wetlands) that lie outside the lateral boundaries of the fill itself;
- (vi) In designing, constructing, and maintaining roads, vegetative disturbance in the waters of the United States shall be kept to a minimum;
- (vii) The design, construction and maintenance of the road crossing shall not disrupt the migration or other movement of those species of aquatic life inhabiting the water body;
- (viii) Borrow material shall be taken from upland sources whenever feasible;
- (ix) The discharge shall not take, or jeopardize the continued existence of, a threatened or endangered species as defined under the Endangered Species Act, or adversely modify or destroy the critical habitat of such species;
- (x) Discharges into breeding and nesting areas for migratory waterfowl, spawning areas, and wetlands shall be avoided if practical alternatives exist;
- (xi) The discharge shall not be located in the proximity of a public water supply intake;
- (xii) The discharge shall not occur in areas of concentrated shellfish production;
- (xiii) The discharge shall not occur in a component of the National Wild and Scenic Rivers System;
- (xiv) The discharge of material shall consist of suitable material free from toxic pollutants in toxic amounts; and
- (xv) All temporary fills shall be removed in their entirety and the area restored to its original elevation.
- (d) For purpose of paragraph (c)(1) of this section, cultivating, harvesting, minor drainage, plowing, and seeding are defined as follows:

(1) Cultivating means physical methods of soil treatment employed within established farming, ranching and silviculture lands on farm, ranch, or forest crops to aid and improve their growth, quality, or yield.

(2) Harvesting means physical measures employed directly upon farm, forest, or ranch crops within established agricultural and silvicultural lands to bring about their removal from farm, forest, or ranch land, but does not include the construction of farm, forest, or ranch roads.

(3)(i) Minor drainage means:

(A) The discharge of dredged or fill material incidental to connecting upland drainage facilities to waters of the United States, adequate to affect the removal of excess soil moisture from upland croplands. Construction and maintenance of upland (dryland) facilities, such as ditching and tiling, incidental to the planting, cultivating, protecting, or harvesting of crops, involve no discharge of dredged or fill material into waters of the United States, and as such never require a Section 404 permit;

(B) The discharge of dredged or fill material for the purpose of installing ditching or other water control facilities incidental to planting, cultivating, protecting, or harvesting of rice, cranberries, or other wetland crop species, where these activities and the discharge occur in waters of the United States which are in established use for such agricultural and silvicultural wetland crop production;

(C) The discharge of dredged or fill material for the purpose of manipulating the water levels of, or regulating the flow or distribution of water within, existing impoundments which have been constructed in accordance with applicable requirements of the Act, and which are in established use for the production or rice, cranberries, or other wetland crop species.

Note: The provisions of paragraphs (d)(3)(i) (B) and (C) of this section apply to areas that are in established use exclusively for wetland crop production as well as areas in established use for conventional wetland/non-wetland crop rotation (e.g., the rotations of rice and soybeans) where such rotation results in the cyclical or intermittent temporary dewatering of such areas.

(D) The discharge of dredged or fill material incidental to the emergency removal of sandbars, gravel bars, or other similar blockages which are formed during flood flows or other events, where such blockages close or constrict previously existing drainageways and, if not promptly removed, would result in damage to or loss of existing crops or would impair or prevent the plowing, seeding, harvesting, or cultivating of crops on land in established use for crop production. Such removal does not include enlarging or extending the dimensions of, or changing the bottom elevations of, the affected drainageway as it existed prior to the formation of the blockage. Removal must be accomplished within one year after such blockages are discovered in order to be eligible for exemption.

(ii) Minor drainage in waters of the United States is limited to drainage within areas that are part of an established farming or silviculture operation. It does not include drainage associated with the immediate or gradual conversion of a wetland to a non-wetland (e.g., wetland species to upland species not typically adequate to life in saturated soil conditions), or conversion from one wetland use to another (for example, silviculture to farming). In addition, minor drainage does not include the construction of any canal, ditch, dike or other waterway or structure which drains or otherwise significantly modifies a stream, lake, swamp, bog or any other wetland or aquatic area constituting waters of the United States. Any discharge of dredged or fill material into the waters of the United States incidental to the construction of any such structure or waterway requires a permit.

(4) Plowing means all forms of primary tillage, including moldboard, chisel, or wide-blade plowing, discing, harrowing, and similar physical means used on farm, forest, or ranch land for the breaking up, cutting, turning over, or stirring of soil to prepare it for the planting of crops. Plowing does not include the redistribution of soil, rock, sand, or other surficial materials in a manner which changes any area of the waters of the United States to dryland. For example, the redistribution of surface materials by blading, grading, or other means to fill in wetland areas is not plowing. Rock crushing activities which result in the loss of natural drainage characteristics, the reduction of water storage and recharge capabilities, or the overburden of natural water filtration capacities do not constitute plowing. Plowing, as described above, will never involve a discharge of dredged or fill material.

(5) Seeding means the sowing of seed and placement of seedlings to produce farm, ranch, or forest crops and includes the placement of soil beds for seeds or seedlings on established farm and forest lands.

(e) Federal projects which qualify under the criteria contained in Section 404® of the Act are exempt from Section 404 permit requirements but may be subject to other State or Federal requirements.

### 2.3 Description of Action Area

The Action Area is defined as "all areas to be affected directly or indirectly by the Federal Action and not merely the immediate area involved in the Action" (50 CFR § 402.02; Subsection 62-331.010(2), F.A.C.). For the review of this Action, the Action Area encompasses the geographic extent of the FDEP's assumption of Section 404 permitting within the entire State of Florida. The Action Area consists of, and is limited to, the State-assumed waters and areas affected directly or indirectly by the Federal Action. This also includes retained waters downstream of waters assumed by the State.

Florida's assumption to administer a CWA Section 404 program only includes those waters of the United States (WOTUS) not retained by the USACE; these waters administered by the State Section 404 program are referred to as assumed waters or State-assumed waters. The USACE retains permitting responsibility for the discharge of dredged or fill material in those waters defined as "Retained Waters." This definition can be found in FDEP's *State 404 Program Applicant's Handbook* in Chapter 2.0 and the process to determine whether a project is located in retained or assumed waters, is described in Chapter 4.1 of the Handbook (FDEP 2020a). In addition, Appendix A of the 404 Handbook includes the Retained Waters List maintained by the USACE. The administrative boundary demarcating the adjacent wetlands over which jurisdiction is retained by the USACE is a 300-foot guideline established from the ordinary high-water mark or mean high tide line of the retained water. In the case of a project that involves discharges of dredged or fill material both waterward and landward of the 300-foot guideline, the USACE will retain jurisdiction to the landward boundary of the project for the purposes of that project only.

The USACE also retains permitting authority for projects within "Indian country" as that term is defined at 18 USC § 1151 (provided below):

Except as otherwise provided in Sections 1154 and 1156 of this title, the term "Indian country," as used in this chapter, means

all land within the limits of any Indian reservation under the jurisdiction of the United States Government, notwithstanding the issuance of any patent, and, including rights-of-way running through the reservation,

all dependent Indian communities within the borders of the United States whether within the original or subsequently acquired territory thereof, and whether within or without the limits of a state, and

all Indian allotments, the Indian titles to which have not been extinguished, including rights-of-way running through the same.

A list of "Indian country" can be found online in the USACE Jacksonville District Regulatory Division Sourcebook.

The boundary of a mitigation bank, excluding the service area, shall be considered the project boundary, even if only a portion of the bank requires a dredge and fill permit under Section 404 of the CWA.

Federal- and State-approved dredge and fill activities might occur in isolation or adjacent to one another. In the case of a project that involves discharges of dredged or fill material both waterward and landward of the 300-foot guideline, the USACE will retain jurisdiction to the landward boundary of the project for the purposes of that project only (Figure 2-1). Projects that fall entirely outside of the 300-foot buffer of the retained waters will be permitted by the State 404 program (Project 3 in Figure 2-2). Linear projects that have some portion of dredge and fill activities in retained waters will be entirely authorized and permitted by the USACE, even if dredge and fill activities occur in wetlands landward of the 300-foot buffer (Figure 2-3).



Figure 2-1. Example of activities authorized by the USACE 404 program.



Figure 2-2. Example of activities authorized by USACE and FDEP.

Projects 1, 2, and 3 show the difference between retained and assumed waters responsibilities. Project 3 would be authorized by FDEP.



Figure 2-3. Example of a linear project.

Linear projects may sometimes be miles long, but if there are dredge or fill activities waterward of the 300-foot guideline within the project boundary, the project is considered within retained waters and will be processed by the USACE.

### 2.4 Ecosystems and Habitats Located within the Action Area

The State of Florida has several widely used land cover classification systems that define habitat and land use across the entire state. Each classification plays an integral role in *Florida's State Wildlife Action Plan* (Wildlife Plan) published by FWC. The Wildlife Plan was the main resource for descriptions of ecosystems within the proposed Action Area. The key systems referenced for this BE are the Natural Communities Classification developed by the Florida Natural Areas Inventory (FNAI) and the Florida Land Cover Classification System (FLCCS) developed by FWC. FLCCS is a combination of several systems that includes the FNAI Natural Communities Classification. This discussion summarizes major aquatic ecosystem types within Florida, including marine ecosystems; freshwater non-forested wetlands; freshwater forested wetlands, lakes, rivers and streams; and estuarine wetlands. Each of the major types includes a variety of more specific sub-types (Tables 2-1 and 2-2).

#### 2.4.1 Aquatic Ecosystems and Habitats

Florida's aquatic resources are made up of 88,250 kilometers (km) of streams and rivers, 127,239 km of ditches and canals, and 45,838 square kilometers (km<sup>2</sup>) of freshwater and tidal wetlands (Appendix D, Figure 1). Florida's coastline ranks second in length only to that of Alaska (NOAA 2017b) and includes nearly 2,000 miles of coastline and marine habitats. These coastline and aquatic systems provide economic and ecological benefits vital to the economy and environment of the state. Resources such as oyster reefs, seagrass beds, marshes, and mangroves provide habitat for various species integral to Florida's economy.

#### 2.4.2 River and Stream Ecosystems and Habitats

Florida's rivers and streams are characterized as natural, flowing waters from their sources to the limits of tidal influence and bounded by channel banks (Appendix D, Figure 1) (FNAI 2010). Of the 1,700 rivers that flow through the state, 23 are considered major rivers. Species such as the shortnose sturgeon, Atlantic sturgeon, and smalltooth sawfish are found in very few of these waterways. The St. Mary's River and Cape Coral are two of the few areas that these species may access inland rivers and streams. The canals in Cape Coral are disconnected habitat that might no longer be accessible to these species. These canals might be accessible during times of high waters (i.e., flooding). Florida contains an abundance of springs that originate from underground aquifers. These springs are noted for their high water clarity, low sedimentation, stable channels, and openings that are less than 40 feet wide. Table 2-1 provides several habitat classes of rivers and streams in Florida.

Habitat Class	Includes
River	Alluvial river
Stream	Blackwater stream, calcareous stream, seepage stream, spring-run stream, and tidally influenced stream
Springs	Major springs – 32 count

Tahle	2-1	River	and	Stream	Habitat	Classes
lane	2-1.	RIVEI	anu	Sueam	Πανιιαι	0103363

Source: FWC 2019.
#### 2.4.3 Estuarine Ecosystems and Habitats

The estuarine ecosystem is the tidally influenced zone landward to the point at which seawater becomes significantly diluted with freshwater inflow from the land (Appendix D, Figure 2) (FNAI 2010). This ecosystem occurs in the intertidal or supratidal zones, is dominated by herbaceous or woody halophytic vascular plants, and experiences salinity levels of 0.5 parts per thousand and higher. Species diversity is low because of the extreme physiological stressors in these habitats.

Soils in mangrove swamps and salt marshes are typically muck and sand or limestone substrate and are inundated by daily tides with saltwater creating anoxic conditions. Both systems are found in relatively flat, low-wave energy areas. Salt marshes are dominated by vascular plants (e.g., saltmarsh cordgrass, needle rush, and saltwort). Mangrove swamps are usually stands of one of the three species present in Florida: red, white, and black mangroves (Table 2-2).

#### Table 2-2. Estuarine Area Ecosystems and Habitats

Major Type	Acres
Mangrove swamp	614,097
Salt marsh	378,677

Source: FWC 2019.

#### 2.4.4 Marine Ecosystem and Habitat

The marine ecosystem is the subtidal, intertidal, and supratidal zone of the sea, landward to the point at which seawater becomes significantly diluted with freshwater inflows from the land (FNAI 2010). The marine ecosystem is classified as mineral based (e.g., consolidated or unconsolidated), faunal based (e.g., coral reef, mollusk reef, octocoral bed, sponge bed, or worm reef), floral based (e.g., algal bed or seagrass bed), or a composite (e.g., community elements from more than one natural community). Florida has over 1,350 miles of coastline and is second only to Alaska for amount of marine ecosystem available.

# 3 ESA-Considered Species Potentially Affected by the Action

Twenty-nine species—28 species are listed under the ESA and one species is considered a candidate for listing under the ESA—occur or could occur within the Action Area and might potentially be affected by the Action. Table 3-1 lists these species, which form the basis of the analysis of the effects in Chapter 5 of this document. Species for which NMFS has been petitioned for listing are included because some of these species may be added to the federal list in the future. The programmatic nature of this BE and the proposed structure of the species coordination process address future listing status changes for listed species and any new species that might be listed that may not be included in this BE. Any changes in listing status for affected species will be addressed during species coordination and technical assistance with the NMFS.

Common Name	Scientific Name	Federal Status	Critical Habitat Designated	Are assumed waters regularly utilized by the species?
	MAMM	ALS		<u>I</u>
Blue whale	Balaenoptera musculus	Endangered	No	No
Byrde's whale <sup>a</sup>	Balaenoptera edeni [unnamed subspecies]	Endangered	No	No
Fin whale*a	Balaenoptera physalus	Endangered	No	No
Humpback whale <sup>a</sup>	Megaptera novaeangliae	Endangered–4 DPSs Threatened–1 DPS State-listed–1 DPS	No	No
North Atlantic right whale <sup>a</sup>	Eubalaena glacialis	Endangered	Yes <sup>c</sup>	No
Rice's whale	Balaenoptera ricei	Endangered	No	No
Sei whale <sup>a</sup>	Balaenoptera borealis	Endangered	No	No
Sperm whale <sup>a</sup>	Physeter macrocephalus	Endangered	No	No
	REPTI	LES		
Kemp's ridley sea turtle <sup>a</sup>	Lepidochelys kempii	Endangered	No	No
Leatherback sea turtle <sup>a</sup>	Dermochelys coriacea	Endangered	No	No
Loggerhead sea turtle <sup>a</sup>	Caretta	Endangered–5 DPS Threatened–4 DPS	Noc	No
Hawksbill sea turtle <sup>a</sup>	Eretmochelys imbricata	Endangered	No	No
Green sea turtle <sup>a</sup>	Chelonia mydas	Threatened	No	No
	COR	AL		
Elkhorn coral <sup>a</sup>	Acropora palmata	Threatened	Yes	No
Staghorn coral <sup>a</sup>	Acropora cervicornis	Threatened	Yes	No
Boulder star coral <sup>a</sup>	Orbicella franksi	Threatened	No <sup>d</sup>	No
Mountainous star coral <sup>a</sup>	Orbicella faveolate	Threatened	No <sup>d</sup>	No
Lobed star coral <sup>a</sup>	Orbicella annularis	Threatened	No <sup>d</sup>	No
Rough cactus coral <sup>a</sup>	Mycetophyllia ferox	Threatened	No <sup>d</sup>	No
Pillar coral <sup>a</sup>	Dendrogyra cylindrus	Threatened	No <sup>d</sup>	No
	FISI	4		
Shortnose sturgeon <sup>a</sup>	Acipenser brevirostrum	Endangered	No	Yes (streams)
Gulf sturgeon <sup>a</sup>	Acipenser oxyrinchus [=oxyrhynchus] desotoi	Threatened	Yes	Yes (streams)
Atlantic sturgeon <sup>a</sup>	Acipenser oxyrinchus	Endangered–South Atlantic DPS (occurs in Florida	Yes	Yes (streams)
Smalltooth sawfish <sup>a</sup>	Pristis pectinate	Endangered–U.S. and Bahaman DPS	Yes	Yes (streams)
Nassau grouper <sup>a</sup>	Epinephelus striatus	Threatened	No	No

#### Table 3-1. ESA-Considered Species Potentially Affected by the Action

Common Name	Scientific Name	Federal Status	Critical Habitat Designated	Are assumed waters regularly utilized by the species?		
Oceanic whitetip shark	Carcharhinus longimanus	Threatened	No	No		
Giant manta ray <sup>a</sup>	Mobula birostris	Threatened	No	No		
MOLLUSKS						
Queen conch	Alger gigas	Proposed threatened	No	No		

Notes: DPS = distinct population segment.

<sup>a</sup> Species is on Florida's Endangered and Threatened Species List.

<sup>b</sup> Listed by USFWS and the State of Florida only.

° Critical habitat has been designated by USFWS in terrestrial areas for nesting.

<sup>d</sup> There is a proposed rule to designate critical habitat for this coral species: Proposed Rule to Designate Critical Habitat for the Threatened Caribbean Corals | NOAA Fisheries.

## 3.1 ESA-Considered Species and Critical Habitat in the Action Area

This BE discusses all ESA-listed, proposed, candidate, and petitioned plant and animal species under NMFS jurisdiction and with ranges within the Action Area. The status of listed species can change from year to year, and species for which NMFS has been petitioned for ESA listing might be listed in the future. Including a comprehensive analysis in this BE for the species listed in Table 3-1 helps evaluate the effects of the State 404 program on any species that were listed at the time this BE was prepared or that were under review and might become listed in the future.

Information for the species analyzed in this document was gathered from the NOAA Endangered Species Directory, the NMFS National ESA Critical Habitat Mapper, the FWC's Species Profiles, and documents identified through those sources.

This BE also discusses critical habitat for ESA-listed, proposed, candidate, and petitioned plant and animal species under NMFS jurisdiction and with ranges within the Action Area. As the precise locations of future permit applications are not known at this time, no attempt was made to generate detailed range maps or analyze species distributions relative to anticipated impacts. Without knowledge of future permit locations, it is impractical to determine the amount of overlap between project sites and species at this time.

Similarly, critical habitat was not mapped because the locations of future permit applications are not precisely known at this time. Conservation tools are available online, including geographic information system (GIS) downloads for critical habitats and habitat ranges and the National ESA Critical Habitat Mapper (NOAA 2022r). Table 3-1 indicates whether critical habitat has been designated for each ESA-listed species. For species with designated critical habitat, future State 404 permit review will evaluate whether the project location is within a critical habitat unit.

Each State 404 permit application within assumed waters will identify the location and extent of proposed activities, allowing identification of ESA-listed species present or potentially present, and presence or absence of designated critical habitat. This data will be used to determine potential effects on ESA-listed species on a project-specific basis.

# 3.2 NMFS-Listed Species

FDEP requested from the USFWS and the NMFS input on a draft species list for Florida's application package for assumption on November 22, 2019. On April 15, 2020, the NMFS responded to FDEP that ESA-listed species under NMFS's jurisdiction do not occur in waters that are assumable by the State (see Appendix A). NMFS stated that for species for which jurisdiction is shared—for example, the NMFS and the USFWS share jurisdiction for Gulf sturgeon—the USFWS is responsible for consultations regarding sturgeon and critical habitat in riverine habitat units. The NMFS further stated that the riverine critical habitat units and river areas where Gulf sturgeon are known to occur are all listed by the USACE as retained waters.. Based on the NMFS's determination, the EPA, therefore, concluded in 2020 that the EPA's approval of FDEP's assumption of the Section 404 program would have no effect on NMFS jurisdictional species. Today's BE considers whether the EPA's approval of Florida's assumption request (Action) may affect listed species or designated critical habitat under NMFS jurisdiction in the Action Area, including whether the Action is likely to adversely affect any of those species or habitats in retained waters downstream of waters assumed by the State.

# 3.3 State-Listed Species

Although this BE focuses on federal ESA-listed, proposed, or under review species, there is considerable overlap with Florida's State-listed species (FWC 2018). Chapter 68A-27, F.A.C. provides a current listing status of all Florida's federal- and State-listed species. It is noteworthy that Florida's definition of "take" is identical to the federal definition (Chapter 68A-27.003, F.A.C.). See Chapter 7.1 of this document for additional information regarding State-listed species.

# 4 Baseline Conditions

This chapter identifies and describes all known natural and anthropogenic sources of impact on ESA-considered species and the condition of their habitats in the Action Area, except those caused by the Action. The purpose of the environmental baseline is to provide the context for the impacts of the Action with regard to the impacts of all the other human activities that are also affecting the ESA-considered species.

This environmental baseline describes wetlands status and trends within Florida, and implications for ESA-considered species. The baseline assessment focuses on factors influencing habitat and those which affect the distribution and abundance of ESA-considered species. The baseline also includes past and present impacts of federal, State, and private activities within the Action Area, and activities underway at present (i.e., coincident with assumption). The Action Area was previously defined and described in Chapter 2 of this document.

# 4.1 Regulatory Baseline

#### 4.1.1 Federal Wetland Regulations

Section 404 of the CWA (33 USC § 1344) establishes a program to regulate the discharge of dredged or fill material into WOTUS. The Administrator of the EPA, in conjunction with the

Secretary of the Army, acting through the Chief of Engineers, established guidelines for regulating such discharges under Section 404(b)(1) of the CWA (40 CFR Part 230). The EPA and the USACE jointly implement the regulation and permitting of such proposed activities, except where a State or Tribe has assumed administration of a CWA Section 404 program. In Florida, the USACE Jacksonville District acts as the regulatory agency that issues Section 404 dredge and fill permits in retained waters.

#### 4.1.2 Individual Permits

To receive a dredge and fill permit authorization from the USACE, an applicant must demonstrate the following under 40 CFR § 230.10:

- No practicable alternative to the proposed activity exists that would have less adverse impact on the aquatic ecosystem;
- The proposed activity will not:
  - violate State water quality standards,
  - violate any applicable toxic effluent standard or prohibition,
  - jeopardize the continued existence of endangered and threatened species, or result in the likelihood of destruction or adverse modification of critical habitat
  - violate any requirement imposed to protect a marine sanctuary;
- The proposed activity will not cause or contribute to significant degradation of WOTUS; and
- The applicant has taken appropriate and practicable steps that will minimize potential adverse impacts of the discharge on the aquatic ecosystem.

The USACE will request any additional information required to deem an application complete, typically within 15 days of receipt of the application. Once the agency deems the application to be complete, the USACE publishes a public notice within 15 days to receive comments from interested and/or affected parties on the proposed action.

Following receipt of an application and evaluation as to the completeness of the application, the USACE is charged with evaluating, among other considerations, the effects of the proposed action on ESA-listed species or designated critical habitat. Where a proposed action may affect a listed species or critical habitat, USACE coordinates and/or consults with the Services prior to issuing any permit. If the USACE determines the proposed activity may affect any endangered or threatened species or their critical habitat, beneficially or adversely, the USACE District Engineer will initiate consultation with the Services. If the USACE District Engineer makes a determination from the submitted application that the proposed activity would not affect ESA-listed species or their critical habitat, the public notice will contain a statement attesting to such and consultation with the Services is not required.

The comment period is typically 30 days; upon receipt of comments, the USACE evaluates the comments received, provides them to the applicant, and determines whether a public hearing is required. Following the comment period (and a public hearing if conducted), the USACE makes a determination as to whether the Section 404 permit should be issued. This determination is based on applicable regulations governing the activity as well as comments received as part of the record. The USACE District Engineer will either prepare a Statement of Findings or, where an Environmental Impact Statement has been prepared, a Record of Decision on all permit decisions. The final action of the USACE is either the signature of the issuing official on the authorizing document (a USACE permit) or a signature on a letter notifying the applicant of the work; a denial will contain written documentation of the reason(s) for the denial.

#### 4.1.3 General Permits

A general permit is issued for structures, work, or discharges that will result in only minimal adverse effects. General permits are issued on a nationwide, regional, or state basis for particular categories of activities. There are three types of general permits: NWPs, Regional General Permits, and Programmatic General Permits. General permits (which are reviewed by the Services) are usually valid for five years and may be reauthorized by the USACE. (The Services will review the proposed reauthorizations).

#### 4.1.4 Nationwide Permits

On a five-year basis, the USACE issues NWPs pursuant to Section 404(e) of the CWA (33 USC § 1344) and Section 10 of the Rivers and Harbors Act of 1899 (33 USC § 401 *et seq.*). As of January 6, 2017, there were a total of 52 NWPs. The NWPs streamline the requirements of the CWA and are informed by extensive feedback from the public and other key stakeholders. NWPs provide expedited review of projects that have minimal impact on the aquatic environment. Categories of activities that may be covered under the NWPs include, but are not limited to, linear transportation projects, bank stabilization activities, aquatic habitat restoration, residential development, commercial and industrial developments, aids to navigation, and certain maintenance activities.

In 2017, the USACE added two new NWPs in addition to the 50 that were in place in 2012. One addition provides a mechanism for an efficient authorization process for the removal of low-head dams to restore streams and enhance public safety; the second addition covers the construction and maintenance of living shorelines to control erosion in coastal areas.

#### 4.1.5 Regional General Permits

As of February 2020, the USACE Jacksonville District has issued 18 general permits. Each regional general permit has specific terms and conditions, all of which must be met for project-specific actions to be verified as compliant with and covered by the respective general permit.

#### 4.1.6 Programmatic General Permits

Programmatic General Permits are based on an existing state, local, or other federal programs and designed to avoid duplication of those programs. The USACE Jacksonville District lists 12 programmatic general permits, one of which is only applicable to Puerto Rico.

#### 4.1.7 Florida Regulation of Wetlands Through the Environmental Resource Permit (ERP) Program

Part IV of § 373, F.S. regulates dredging and filling in wetlands and other surface waters, such as the construction, alteration, operation, maintenance, abandonment, and removal of stormwater management systems, dams, impoundments, reservoirs, works (including, but not limited to, ditches, canals, conduits, channels, culverts, pipes, and other artificial structures), and appurtenant works (artificial improvements to a dam).

This statute authorizes FDEP and the five WMDs in the state to jointly implement Florida's ERP program. The responsibilities of the agencies are divided according to Operating Agreements between FDEP and the particular WMD. Provisions in the statute allow for FDEP to approve local government programs to implement the ERP program on behalf of FDEP and the WMDs. As of January 2020, full delegation has been given to Broward County and minor works delegated to the Environmental Protection Commission for Hillsborough County.

The ERP program operates in addition to the USACE Section 404 program that regulates activities in WOTUS. All state, local, and regional governments in Florida delineate wetlands in accordance with State methodology (Chapter 62-340, F.A.C.) instead of the federal wetland delineation method (Section 404 of the CWA and the Federal Manual for Identifying and Delineating Jurisdictional Wetlands). While the ERP application is issued, withdrawn, or denied in accordance with State statutory and rule criteria (briefly summarized below), agency action on the ERP application also constitutes any needed water quality certification under Section 401 of the CWA, and coastal zone consistency concurrence statements with Florida's federally approved Coastal Zone Management program under Section 307 (Coastal Zone Management Act). The USACE must take separate action to issue or deny any needed federal permit under Section 404 of the CWA and/or Section 10 of the Rivers and Harbors Act of 1899.

To receive an ERP, an applicant must demonstrate that the proposed activity will not be harmful to the water resources of the State and will not be inconsistent with the overall objectives of Florida rules and statutes. The applicant must provide reasonable assurance that the activity will not violate the applicable State water quality standards and that the activity is not contrary to the public interest for all waters that are not designated as Aquatic Preservers or Outstanding Florida Waters. For activities in those designated waters, the applicant must provide reasonable assurance that the proposed activity will be clearly in the public interest. Surface water quality standards are published in Chapter 62-302, F.A.C. In addition, FDEP provides policy guidance on anti-degradation in Rule 62-4.242, F.A.C., and in Rule 62-302.300, F.A.C, which allows for the protection of water quality above the minimum required for classification. Further, FDEP administers the Impaired Waters Rule (Chapter 62-303, F.A.C.), and has established Total Maximum Daily Load criteria (Chapter 62-304, F.A.C.) (FDEP 2020b). It is the intent of FDEP and the WMDs that these criteria are implemented in a manner that achieves a programmatic goal and a project-permitting goal of no net loss in wetlands or other surface water functions.

To determine whether an activity is not contrary to the public interest or is clearly in the public interest, FDEP must consider and balance the following criteria:

- Whether the activity will adversely affect the public health, safety, or welfare or the property of others;
- Whether the activity will adversely affect the conservation of fish and wildlife, including endangered or threatened species, or their habitats;
- Whether the activity will adversely affect navigation or the flow of water or cause harmful erosion or shoaling;
- Whether the activity will adversely affect the fishing or recreational values or marine productivity in the vicinity of the activity;
- Whether the activity will be of a temporary or permanent nature;
- Whether the activity will adversely affect or will enhance significant historical and archaeological resources; and
- The current condition and relative value of functions being performed by areas affected by the proposed activity.

FDEP provides a copy of all notices of ERP applications for individual permits that propose regulated activities in, on, or over wetlands or other surface waters to FWC for review and comment. The FDEP and FWC frequently work together on non-regulatory as well as regulatory issues. Examples of the many collaboration efforts include habitat restoration projects, management of State parks and management of species on other State-owned easements and property.

In accordance with the provisions of § 373.4141, F.S., FDEP shall review the ERP application to determine if it is complete. If the application is incomplete, FDEP must request additional information within 30 days. The applicant must respond to such requests within 90 days. Within 30 days after receipt of such additional information, FDEP must review the submitted material for completeness. The WMD processing procedures for ERP vary somewhat to accommodate the requirements of their specific Governing Boards.

In accordance with § 120, F.S., FDEP must decide whether it should issue or deny an ERP within 60 days after receipt of the original application, the last item of timely requested additional material, or the applicant's written request to begin processing the permit application. Application completeness is determined by whether the applicant has submitted all materials required for review as specified by rule and statute. The WMDs also are subject to this requirement, but their ERP processing procedures vary by each district to accommodate the requirements of their different Governing Boards. Pursuant to § 120.60(1), F.S., any application that FDEP or the WMD does not approve or deny within 60 days is considered approved by default.

Once issued, ERP permits are valid for the life of the system (which includes all structures and works authorized for construction or land alteration). The ERP permit does not automatically expire after the construction phase (typically five years) of a project but continues to cover operation (use) of the system in perpetuity.

# 4.2 Environmental Baseline

#### 4.2.1 Coastal Florida

There are 88,250 km of streams and rivers, 127,239 km of ditches and canals, and 45,838 km<sup>2</sup> of freshwater and tidal wetlands in Florida (NOAA 2017b). Flather et al. (1998) identified habitat loss and alien species as the two most widespread threats to endangered species, affecting more than 95 percent and 35 percent of listed species, respectively. These effects are largely the result of human-altered hydrology reducing the spatial extent and quality of available habitat as well as the connectivity of rivers and streams for migrating species. Polluted runoff has been identified as a major stressor affecting Florida's coastal waters, contributing to coral reef degradation, fish kills, seagrass bed declines, and algal blooms, including blooms of toxic algae (USEPA 2012a).

Climate change also is projected to have substantial direct effects on individuals, populations, species, and the community structure and function of marine, coastal, and terrestrial ecosystems in the foreseeable future (McCarty 2001; IPCC 2002; Parry et al. 2007; IPCC 2013). Increasing temperatures can cause shifts in the latitudinal distribution of freshwater and marine species and can result in the available habitat favoring non-native species over native species (NOAA 2107). Ocean acidification, as a result of increased atmospheric carbon dioxide, also can interfere with numerous biological processes in corals, including fertilization, larval development, settlement success, and secretion of skeletons (Albright et al. 2010).

#### 4.2.2 Wetlands

In 1845, the State of Florida contained an estimated 20.3 million acres of wetlands (Dahl 2005). By 1996, only about half of the original wetlands remained (USFWS 1996). From the mid-1950s through the mid-1970s, prior to the CWA, the rate of wetland loss has been estimated at 72,000 acres per year (Hefner 1986). In the following decade, wetland loss decreased to an estimated 23,700 acres per year (Hefner et al. 1994).

As of 1996, an estimated 11.4 million acres of wetlands covered about 29 percent of the surface area of Florida, more than any other state in America at that time. Of these wetlands, 90 percent or about 10.2 million acres were freshwater wetlands. The average annual net loss of wetlands from 1985 through 1996 was 4,740 acres, and freshwater forested wetlands exhibited a net gain. During the 1985-1996 interval, 72 percent of wetland loss was attributed to development and 28 percent to agriculture (Dahl 2005).

#### 4.2.3 Central and Southern Florida Project

The Central and Southern Florida (C&SF) Project for flood control and other purposes was authorized by the Flood Control Act of 1948 as an improvement plan for flood control, drainage, and other purposes over an 18,000-square-mile area of central and south Florida. This project authorized the diversion of water to the Atlantic Ocean and the Gulf of Mexico through canals and the diversion of water southwest through the Everglades. The project provided benefits for human populations that were able to build, grow, and develop on these new lands. The Everglades Agricultural Area was developed for the production of food, and areas further east became densely populated cities, including Miami, Ft. Lauderdale, and West Palm Beach. Unfortunately, the project also resulted in the modification and loss of 2,400 square miles of freshwater wetlands, including in the Everglades (USACE 2019a). See Figure 4-1 for a comparison of historic freshwater flows compared to water flows today.



Figure 4-1 Historic freshwater flows compared to freshwater flows after C&SF Project (*Source*: USACE and SFWMD 2007).

#### 4.2.4 Everglades Restoration

Florida's Everglades are twice the size of New Jersey, composed of a mixture of dense forests and open prairies, sunny croplands and shady swamps, rural areas and cities. The South Florida Ecosystem Restoration Program, a partnership between the federal government and the State of Florida, consists of a suite of projects that focus on the C&SF system. In 2000, congressional authorization created the Comprehensive Everglades Restoration Plan (CERP), which is the single largest of the program's efforts. CERP is a 50/50 partnership between the federal government and the State of Florida. It is a program to restore, protect, and preserve water resources in C&SF, including the Everglades. The USACE is the lead federal agency, and the South Florida Water Management District is the lead State agency in this effort.

For 20 years, the CERP program has been designing, planning, and constructing multiple components of the South Florida Ecosystem Restoration Program. The goal of these efforts is to eventually improve 2.4 million acres of south Florida's wetlands ecosystems (including Everglades National Park), by reducing high volume discharges from Lake Okeechobee to the estuaries and improve water delivery to the Florida and Biscayne Bays, as well as enhance the freshwater supply (USACE 2019b). See Figure 4-2 for a depiction of the Everglades restoration project and the associated improvements to future ecosystem conditions.



Figure 4-2. Future ecosystem conditions based on CERP (Source: USACE and SFWMD 2007).

#### 4.2.5 Historical Federal Permitting: Habitat

The following discussion is based on data provided by the USACE and characterizes Section 404 permits issued from 2014 through 2018. The habitat types are from the Wildlife Plan and Guide to the Natural Communities of Florida (FNAI 2010), and each includes multiple subtypes. These are cross walked with Cowardin types used for USACE reporting; similar definitions may not match entirely.

Table 4-1 summarizes wetland permits issued in Florida by the USACE from FY 2014 through FY 2018. Permitted wetland fill is presented by type of wetland and acreage; the dataset is limited to wetlands mostly within the Action Area covered by the current document and does not include all wetlands in the State. For the first four years of the period, permitted wetland fill ranged from 1,587 to 2,417 acres per year, then more than doubled to 4,363 acres in FY 2018 reportedly because of a confluence of multiple larger one-time projects being permitted in that year. The mean annual permitted acreage for the period was 2,423 acres per year. This suggests that annual wetland loss has remained at or below levels identified in previous decades. The numbers in Table 4-1 do not include offsets from wetland mitigation, wetland dredge, or grantfunded restoration. Based on national data, mitigation and restoration likely result in lower net loss of wetlands and even net gains for some habitat types in some years (1998-2004), although in 2004-2009 there was a continued national net loss (Dahl 2011).

As indicated in Table 4-1, wetland fill impacts within the Action Area were primarily to palustrine wetlands. Impacts on other wetland types were considerably less in comparison. Palustrine wetlands provide habitat for numerous ESA-listed species.

#### **NOTES regarding Interpretation of USACE Data:**

A single 404 permit may authorize activities that affect multiple wetlands or surface water areas within the project boundaries. The total number of areas will be larger than the total number of permits issued.

The USACE data assigns an ID number for each WOTUS that is delineated on a site, and that number is in Table 4-1 under the column "Number of Areas.". The acreages for these areas under the column "Total Acres" represent the entire footprint of the fill authorized for each area, not just the amount of WOTUS that was adversely impacted. (R. Barron, USACE, personal communication, April 16, 2020).

At the time of this analysis, the USACE was developing a GIS shapefile to generally depict the retained waters. The draft shapefile (February 2019) did not include all of the waters noted in the USACE list of retained water (particularly canal systems in southwest and south Florida). Because it was the best available information at the time, FDEP queried the USACE project data by overlaying it with the draft retained water shapefile. This was performed in order to estimate the number and types of projects in State-assumed waters, outside of boundaries of the draft USACE retained waters file. Due to the limitations of the geographically referenced shapefile, the USACE permit-related data in this BE does not accurately represent projects/permits in only assumed waters. Please note that this data was used to analyze and depict general data trends for this BE, and not serve as a source for definitive quantitative data. Tables and figures in this BE that use this data likely overestimate the number of projects that will be in assumed waters but is presented here as best available information. The data provided in Table 4-1 and Figure 4-3 likely inadvertently include some retained waters and may include values for some Section 10 projects.

FY 14		14	FY 15		FY 16		FY 17		FY 18	
Cowardin Type	Number <sup>a</sup> of Areas	Total Acres	Number of Areas	Total Acres	Number of Areas	Total Acres	Number of Areas	Total Acres	Number of Areas	Total Acres
Estuarine	29	2.71	39	6.85	22	5.55	55	17.3	112	10.82
Lacustrine	29	10.39	26	48.86	35	9.29	24	17.45	32	12.46
Marine	6	0.46	2	0.03	4	3.08	11	59.91	15	10.58
Palustrine	740	1,491.65	788	1,452.29	786	2,162.99	990	1,851.78	1,234	4,141.19
Riverine	130	432.94	124	78.95	169	235.75	141	44.34	238	188.06
Total	934	1,938.15	979	1,586.98	1,016	2,416.66	1,221	1,990.78	1,631	4,363.11

#### Table 4-1. Authorized Areas and Acreages by Wetland Type, 2014–2018

Source: USACE Jacksonville District.

Note: <sup>a</sup> The Number of Areas represents the number of WOTUS areas with authorized activities. A single 404 permit may authorize activities in multiple WOTUS areas, so the number of areas will exceed the number of permits issued. Also see Notes subsection under Section 4.2.5 regarding areas and acreages. FY = Federal Fiscal Year (October through September).



Figure 4-3. Acreage of authorized fill, by wetland type. *Source*: USACE Jacksonville District.

#### 4.2.6 Freshwater Non-forested Wetlands—Palustrine Wetlands including Palustrine Emergent and Palustrine Shrub Scrub

As shown in Table 4-1, freshwater non-forested wetlands (assumed to be roughly equivalent to combined palustrine emergent and palustrine shrub scrub Cowardin Types) are associated with a considerable number of the permitted activities in Florida from FY 2014 through 2018; the number of WOTUS areas authorized ranged from 263 to 542, and the acreage ranged from 320 to 2,069 annually. Although freshwater non-forested wetlands are among the most extensive in Florida (about 5.4 million acres) (FWC 2019), Table 4-1 suggests that Section 404 permitted fill activities disproportionately affected this habitat type.

#### 4.2.7 Freshwater Forested Wetlands

Freshwater forested wetlands (assumed to be roughly equivalent to palustrine forested Cowardin Types) accounted for an even greater number of Corps-permitted fill activities from FY 2014 through 2018; the number of WOTUS areas authorized ranged from 354 to 622, and the acreage ranged from 727 to 1,980.

Freshwater forested wetlands include about 4.2 million acres or about 10 percent of Florida's land area. Table 4-1 suggests that Section 404 permitted fill activities disproportionately affected this habitat type.

#### 4.2.8 Lakes

Lakes (assumed to be roughly equivalent to lacustrine Cowardin Types) accounted for a relatively small proportion of permitted activities: 24 to 32 WOTUS areas authorized by the Corps from FY 2014 through 2018, and nine to 48 acres. Lakes cover almost 1.3 million acres in Florida, with much of the surface area in public ownership.

#### 4.2.9 Rivers and Streams

Rivers and streams (assumed to be roughly equivalent to riverine Cowardin Types) accounted for 124 to 238 WOTUS areas authorized by the Corps from FY 2014 through 2018, and 79 to 432 acres with considerable year to year variation. Estuarine Cowardin Type wetlands may also fall in this category, with a small area included in assumed waters; 22 to 112 WOTUS areas associated with permits were issued for estuarine wetlands, including three to 17 acres of impacts.

#### 4.2.10 Marine

Relatively small areas of marine habitat (assumed to be equivalent to marine Cowardin Types) are within assumed waters; WOTUS areas authorized by the Corps ranged from two to 15 per year, including less than one to about 10 acres of impacts per year.

#### 4.2.11 Uplands

While uplands are not regulated under State wetland regulations, if uplands include listed species which can be adversely affected as a result of the Action, those impacts/effects should be addressed. ESA-listed species could be affected by Actions on uplands that are associated with wetlands permits such as construction of access roads or staging areas, and many species utilize both wetland and upland habitat. Thus, such features are often included as part of the species coordination process for a State 404 permit application review.

#### 4.2.12 Inventories and Surveys for Habitat Types and Quantities

Information on Florida habitat types is summarized in the Wildlife Plan and is also available as GIS layers from the FNAI. While this information is in some cases based on site-specific inventories or surveys, it is presented at the statewide level. A statewide approach is believed to be appropriate for this statewide Programmatic BE. Available GIS layers can be used to map specific wetlands, at a scale that can be presented in a statewide view or mapped on a finer scale when needed.

#### 4.2.13 Historical Federal Permitting: ESA Consultations

Baseline conditions were derived from the USACE permit database provided by the Regulatory Division of the USACE Jacksonville District, which encompasses federal fiscal years 2014 through 2018. This database includes all temporary and permanent permitted wetland impacts by Cowardin code, permit authorization type, dredge or fill acreage approved, and a project site coordinate. The database also includes all ESA consultations by type, agency, closure method,

ESA-listed species potentially affected, and a corresponding object identification number that links the ESA consultations with the permitted wetland impacts.

Please see the Notes subsection in 4.2.5 for details regarding interpretation of the data in this BE. Tables 4-2a and 4-2b show the types of consultations for ESA-listed species for the USACE consultation data involving permanent and temporary dredge and fill actions in Florida from 2014 through 2018. Note that these tables include the West Indian Manatee, which is state listed and was also included in the USFWS BiOp. EPA estimates that out of those past Section 404 permit applications reviewed, 8.1 percent of reviews were reasonably certain to result in take (n = -37) permit reviews out of 459). This percentage was calculated using the USACE Jacksonville District permitting database's "Formal" choice in the "Consultation Type" field, since most of the formal consultations can be assumed to be associated with activities that may cause incidental take. The percentages of all types of consultations are shown for the five-year period is shown in Table 4.2a. In Table 2b, this data is shown annually for years 2014-2018. Approximately 22.8 percent resulted in "Informal" type consultations. As shown in Table 4.2, around 69 percent of findings over the period 2014-2018 were covered by existing programmatic consultations or decision tools. For the USACE permit data collected during the years 2014-2018, a small proportion of the total number of ESA-listed species accounted for the majority of consultations. During this time, many of the species subject to frequent consultation had existing decisions tools such as consultation keys or programmatic BiOps. For those species, such decision tools can help guide future consultations and assist FDEP when assessing potential effects of proposed actions.

Of the 28 listed and one (1) proposed species in the Action Area, 19 have been the subject of ESA consultations in the past five years. One species (West Indian Manatee) accounted for 55.1 percent of species-level consultations, and just eight (8) species accounted for the majority of all species-level consultations (see Table 4-3a and Table 4-3b; note that most consultations include more than one species, therefore these numbers may represent a smaller number of consultations). Recent consultations are not distributed evenly within the Action Area and can be especially dense in areas of rapid growth and development.

Discrepancies in the numbers between Table 4-2 and Table 4-3 are explained as follows. Each permit may have several habitat types and species associated with it that may require input from either the USFWS or NMFS, sometimes both. Due to the availability of tools such as programmatic consultations and consultation keys, some species for a given permit may be covered under a programmatic consultation while others may require a formal or informal consultation. This results in the possibility of multiple species in the USACE database being associated with one USACE-assigned ESA Action Identification number based on their closure method and consultation type while other species associated with the permit fall into a different ESA Action Identification number.

Table 4-2a.	Consultation	Types for	Federal	Actions	for NMFS	Species;
٦	otals, FY 2014	– 2018				

ESA Determination	Total # (2014-2018)	% of Total
Formal	37	8.1
Informal	105	22.8
Programmatic	317	69.1

ESA Determination	Total # (2014-2018)	% of Total
Total	459	100

Source: USACE Jacksonville District.

# Table 4-3b. Consultation Types for Federal Actions by Year for NMFS Species, FY 2014 – 2018

ESA Determination	Total # 2014	Total # 2015	Total # 2016	Total # 2017	Total # 2018	% of Total
Formal	5	8	14	8	2	8.1
Informal	34	24	14	18	15	22.8
Programmatic	70	54	46	65	82	69.1
Total	109	86	74	91	99	100

Source: USACE Jacksonville District.

# Table 4-4a. ESA Consultations for NMFS Species without Programmatic, By Species, 2014-2018

Species	# of Consultations	Species	# of Consultations			
	Mam	mals				
West Indian Manatee ( <i>Trichechus manatus</i> ) <sup>1</sup>	45	Rice's Whale (Balaenoptera ricei)	0			
Fin Whale (Balaenoptera physalus)	0	North Atlantic Right Whale ( <i>Eubalaena glacialis</i> )	6			
Sperm Whale ( <i>Physeter</i> macrocephalus)	0	Byrde's Whale (Balaenopter edeni)	0			
Sei Whale (Balaenoptera borealis)	0	Humpback Whale ( <i>Megaptera novaeangilae</i> )	0			
Blue Whale (Balaenoptera musculus)	0	-	-			
	Rep	tiles				
Kemp's Ridley Sea Turtle (Lepidochelys kempii)	78	Hawksbill Sea Turtle ( <i>Eretmochelys imbricata</i> )	68			
Leatherback Sea Turtle ( <i>Dermochelys coriacea</i> )	72	Green Sea Turtle (Chelonia mydas)	84			
Loggerhead Sea Turtle ( <i>Caretta</i> caretta)	83	-	-			
	Shar	k/Ray				
Ocean Whitetip Shark (Carcharhinus longimanus)						
Fish						
Shortnose Sturgeon ( <i>Acipenser</i> brevirostrum)	2	Atlantic Sturgeon ( <i>Acipenser</i> oxyrinchus oxyrinchus)	3			

<sup>1</sup> The West Indian Manatee is state-listed and was included in the USFWS BiOp.

Species	# of Consultations	Species	# of Consultations		
Smalltooth Sawfish (Pristis pectinata)	61	Gulf Sturgeon (Acipenser oxyrinchus)	13		
Nassau Grouper ( <i>Epinephulus</i> <i>striatus</i> )	1				
Coral/Conch					
Elkhorn Coral (Acropora palmata)	2	Staghorn Coral ( <i>Acropora</i> cervicornis)	3		
Boulder Star Coral (Orbicella franksi)	1	Mountainous Star Coral ( <i>Orbicella faveolate</i> )	1		
Lobed Star Coral (Orbicella annularis)	1	Rough Cactus Coral ( <i>Mycetophyllia ferox</i> )	1		
Pillar Cactus (Dendrogyra cylindrus)	1	Queen Conch (Aliger gigas)	0		

Source: USACE Jacksonville District.

#### Table 4-5b. ESA Consultations for NMFS Species with Programmatic, 2014-2018

Species	# of Consultations	Species	# of Consultations				
	Mam	imals					
West Indian Manatee ( <i>Trichechus manatus</i> ) <sup>2</sup>	253	Rice's Whale (Balaenoptera ricei)	0				
Fin Whale (Balaenoptera physalus)	0	North Atlantic Right Whale ( <i>Eubalaena glacialis</i> )	2				
Sperm Whale ( <i>Physeter</i> macrocephalus)	0	Byrde's Whale ( <i>Balaenopter edeni</i> )	0				
Sei Whale (Balaenoptera borealis)	0	Humpback Whale ( <i>Megaptera novaeangilae</i> )	0				
Blue Whale (Balaenoptera musculus)	0	-	-				
	Rep	tiles					
Kemp's Ridley Sea Turtle (Lepidochelys kempii)	52	Hawksbill Sea Turtle ( <i>Eretmochelys imbricata</i> )	37				
Leatherback Sea Turtle (Dermochelys coriacea)	33	Green Sea Turtle (Chelonia mydas)	54				
Loggerhead Sea Turtle ( <i>Caretta</i> caretta)	53	-	-				
	Shar	k/Ray					
Ocean Whitetip Shark ( <i>Carcharhinus longimanus</i> )	0	Giant Manta Ray (Mobula birostris)	0				
	Fish						
Shortnose Sturgeon ( <i>Acipenser</i> brevirostrum).	6	Atlantic Sturgeon ( <i>Acipenser</i> oxyrinchus oxyrinchus)	6				

<sup>2</sup> The West Indian Manatee is state-listed and was included in the USFWS BiOp.

Species	# of Consultations	Species	# of Consultations		
Smalltooth Sawfish (Pristis pectinata)	53	Gulf Sturgeon (Acipenser oxyrinchus)	54		
Nassau Grouper ( <i>Epinephulus</i> <i>striatus</i> )	0	-	-		
Coral					
Elkhorn Coral (Acropora palmata)	0	Staghorn Coral (Acropora cervicornis)	0		
Boulder Star Coral (Orbicella franksi)	0	Mountainous Star Coral (Orbicella faveolate)	0		
Lobed Star Coral (Orbicella annularis)	0	Rough Cactus Coral ( <i>Mycetophyllia ferox</i> )	0		
Pillar Cactus (Dendrogyra cylindrus)	0	Queen Conch (Aliger gigas)	0		

Source: USACE Jacksonville District.

# 5 Potential Effects on ESA-Considered Species

Effects of an action are all consequences to listed species or critical habitat that are caused by the action, including the consequences of other activities that are caused by the action. A consequence is caused by the action if it would not occur but for the action and it is reasonably certain to occur. Effects of the action may occur later in time and may include consequences occurring outside the immediate area involved in the action (50 CFR § 402.02). The Action studied in this BE is the EPA's approval of the State's assumption of the dredge and fill permitting program under Section 404 of the CWA for waters assumable by the State. A direct consequence of the Action is that an applicant must go through the State 404 permitting process for a permit in waters assumed by the State and may not go directly to the USACE to get a 404 permit to place fill in State-assumed waters. Therefore, the effects of the EPA approving the program assumption is that FDEP is the permitting authority in assumed waters. The impact of FDEP issuing Section 404 permits is the placement of dredged or fill material into WOTUS, altering the environment where ESA-considered species might exist or designated critical habitat might exist. Since states are not required to consult under ESA Section 7 because there is no federal agency action, the lack of ESA Section 7 consultation on State-issued permits is an effect of the EPA's approval of the State's CWA Section 404 program.

The Action is not expected to substantially increase or decrease the number of Section 404-type authorized activities post-assumption as compared to before assumption, since the only expected change will be the entity processing the permits. The State 404 program's processes and structure will be consistent with CWA requirements and Section 404 analysis of species and aquatic resources and potential effects of proposed fill or dredge activities. It is reasonable to expect that permitting history available from the USACE for the interval of 2014–2018 approximates the number and types of permitting activities that might be anticipated for the next five-year interval. See Chapter 4.2.1 in this document regarding the limitations of the data used in the analyses of this BE and regarding the interpretations of the USACE database and shapefile.

The precise number and locations of future Section 404 permit applications are unknown. Some can be predicted well in advance, such as large public works projects with long lead times and planning horizons. Most applications, however, especially from individuals or private entities, are less predictable. While precise locations are mostly unknown, consultations related to Section 404 applications tend to cluster in rapidly developing regions of the State.

Based on the best available data on Corps permitting activities (FY 2014-2018s) as a baseline (see Chapter 4 of this document), as well as ecological theory, professional judgment, and the information included in the species accounts in Appendix B, this BE includes a qualitative assessment of the potential effects of the Action to species and designated critical habitat within the Action Area, including potential effects to species and designated critical habitat in retained waters downstream of waters assumed by the State. The assessment does not distinguish direct and indirect effects based on 2019 updates to the Section 7 implementing regulations of the ESA (50 CFR Part 402); however, both immediate impacts and those delayed in time were considered.

Table C.1.a in Appendix C summarizes the potential impacts of dredge and fill activities on ESA-listed and proposed-to-listed species. This stressor table is intended to identify potential effects and facilitate analysis of which effects are reasonably likely to occur as a result of activities authorized through the State's assumed Section 404 program. A detailed analysis of potential effects in the future is not possible, because, as stated above, the exact locations, amounts, and types of impacts are not yet known. This chapter briefly summarizes major categories of impacts, followed by a discussion of potential effects on major taxa.

# 5.1 Types of Effects

This discussion is not exhaustive but includes some of the more common influencing factors and consequences to species that might result from dredge and fill activities. There is a spectrum of effects to consider, including beneficial effects, but for the purposes of this BE, the focus will be on potential adverse effects. For example, there is a level of disturbance without a detectable sign of effect (e.g., a sea turtle cannot return to its nesting site but finds a suitable site nearby). There also is disturbance with a detectable sign of effect (e.g., a sea turtle cannot find any suitable site nearby, which affects its ability to create a nest for its eggs). Heightened levels of effect include injury that is observable or detectable, such as a failure to reproduce because of physiological or ecological effects of the action or the death of one or more individuals.

There also is a spectrum of likelihoods that an effect might occur from a theoretical or conceptual standpoint. The likelihood that an effect might occur can be described by how confident a reviewer is that something is actually going to occur. These levels of likelihood that something will occur are potential, unlikely, possible, likely, more than likely, and reasonably certain. Potential stressors or effects on the species and to the essential physical and biological features of any critical habitat are discussed in this section and summarized in Appendix C, but also must be assessed during the review of State 404 permit applications.

#### 5.1.1 Biotic Stressors

Biotic stress is stress that results from damage done to an organism by other living organisms, such as bacteria, viruses, fungi, parasites, beneficial and harmful insects, weeds, and cultivated

or native plants. Dredge and fill activities can alter competitive balances, change predator/prey relationships, or encourage the establishment of invasive plants or animals, which can alter habitat structure. Loss or decrease, or colonization or increase, of a species can cascade through multiple trophic levels and, in some cases, even contribute to habitat alteration (e.g., monotypic stands of invasive plants).

#### 5.1.2 Physical Stressors

Physical stressors should be assessed individually and cumulatively, when assessing potential biological impacts. An example of a physical stressor is the physical diversion of a stream to another location, which might adversely impact the habitat for the flora and fauna using the original location of the stream. Other examples include the construction of new dams, weirs, and culverts, improvements to the structures, and even the removal of the structures; the removal of wetlands and surface waters by filling with materials; and the creation of surface waters by dredging and excavation. Physical stressors include direct mortality burying or trapping individual animals by fill or equipment placing fill, especially for smaller, less mobile species, which also can result in loss of habitat and displacement of more mobile species able to escape the immediate effect. Fill of part of a wetland can contribute to loss of function even if most of the wetland remains intact.

Dredging can result in temporary or permanent loss of aquatic bottom communities, and sedimentation can reduce visibility, clog gills of aquatic species, and bury immobile organisms. Recovery from dredge effects might require hours or years, depending on the habitat, substrate type, and extent of the disturbance.

Fragmentation can affect species that migrate seasonally between habitat types (e.g., sturgeon that migrate upstream to spawn before moving back downstream to feed and rest) or that have large home ranges and frequently move among resource types. It also can disrupt metapopulations, especially for short-lived invertebrates dependent on stochastic environments, reducing the frequency of recolonization of otherwise suitable habitat.

Changes to hydrologic regime can include lowering of ground water levels, increased runoff, or altered hydroperiod. Changes that result in early drying of ponds or wetlands might result in mortality to pond breeding amphibian larvae or small fish, while conversion of a seasonal wetland to a permanent pond might allow colonization by large predators, including stocking of game fish.

Some construction activities (e.g., pile driving or dredging) can result in air or underwater noise and vibration effects. Analysis of and attempting to reduce these effects has become more common in recent years. Construction activities such as noise, vibration, and visual disturbance might alter the behavior of ESA-listed species.

Even measures intended to reduce effects can sometimes have unintended consequences. Some projects have relocated ESA-listed species, but little follow-up monitoring has been done to document success or failure of the translocations.

#### 5.1.3 Physicochemical Water Quality Stressors

Physicochemical stress results from environmental factors such as food/nutrition, toxins, metabolic disorders, infections, and inflammation. Water quality changes are commonly associated with dredge and fill activities, although they are not always easy to describe. Effects can include changes in water temperature, gas or oil dripping from construction equipment or generators to fill a wetland, eliminating or reducing natural filtration of sediment and pollutants, and resulting in degradation of downstream habitat. Dredging also might resuspend environmental contaminants in sediment (common in industrial areas). Changes to nutrient cycling or exchange are even less obvious and might result from new activities occurring on or adjacent to the filled areas.

## 5.2 Potential Effects

This topic is discussed by major taxa and the potential effects of the Action from future activities that might be authorized by the State 404 program on guilds of species with similar habitat needs or life-history traits. These effects are similar to the effects of the USACE Section 404 process for granting or denying permits, only it will be performed by the State 404 program process as a consequence of the Action. It is assumed that overall, the future number and/or any rate of increase in the number of State 404 applications and the general types of activities and overall dredge or fill quantities will be similar to those permitted in similar jurisdictional waters as past requests for permits by the USACE and would not change because of an approval of the State's request for assumption.

#### 5.2.1 Mammals

Nine mammalian species/subspecies included in Appendix C, Table C.1.b. are all aquatic mammals. Impact analysis on the species level (that takes into consideration species'/subspecies' microhabitat preferences) is presented in Table C.1.b.

In general, all mammalian species under consideration in this BE that occupy or frequent WOTUS or adjacent habitats during any stage of their life histories and behaviors may be disturbed by activities during periods when authorized activity-related noise and vibration exceed baseline levels. Construction activities also might disturb the natural behavior of ESA-considered species due to the resulting visual disturbance.

The blue whale, Byrde's whale, fin whale, humpback whale, North Atlantic right whale, Rice's whale, sei whale, and sperm whale are the only ESA- or State-listed aquatic marine mammals considered in this BE. The West Indian manatee is also state listed, but the species is managed by the USFWS and is covered in the USFWS BiOp. Threats to these species include vessel strikes, entanglement in fishing gear, ocean noise, ocean debris, oil spills, and climate change. Future activities authorized by the State 404 program will not affect these species including dredge and fill activities, noise pollution, increases in vessel storage and operation, and impacts on habitat, including changes in hydrologic regimes, water quality, water control structures, and habitat fragmentation due to the lack of these species in retained waters.

#### 5.2.2 Reptiles

Five reptiles included in Appendix C, Table C.1.b. are all marine reptiles and include green sea turtle, hawksbill sea turtle, Kemp's ridley sea turtle, leatherback sea turtle, and loggerhead sea turtle. Impact analysis on the species level (that takes into consideration species/subspecies microhabitat preferences) is presented in Appendix C, Table C.1.b.

The permits issued under the State 404 program will have no effect on reptiles in the action area. Threats to the species are bycatch in fishing gear, direct harvest of turtles and eggs, loss and degradation of nesting habitat (e.g., structural modifications, sand nourishment, and sand extraction), predation of eggs and hatchlings, vessel strikes, pollution and marine debris, and climate change. While the beta Section 7 Mapper shows overlap of sea turtle layers and assumed waters, these layers are in beta form and extend beyond expected nesting habitat. The state will coordinate wherever overlap occurs; however, because nesting habitat is found in areas that are retained by the Corps and are not reasonably expected to have downstream impacts from assumed waters, no effects are expected.

#### 5.2.3 Fish

Seven species/subspecies of fish included in Appendix C, Table C.1.b are all marine fish. They include the Atlantic sturgeon, giant manta ray, Gulf sturgeon, Nassau grouper, oceanic whitetip shark, shortnose sturgeon, and smalltooth sawfish. Marine species are less likely to be affected by small amounts of dredge or fill because they tend to occur in larger and more contiguous habitats, although sedimentation, water quality degradation, and, to a lesser extent, direct habitat loss are potential effects. As fish are, by definition, fully aquatic, they are frequently affected by CWA activities. To the extent that any of these species have the potential to be impacted by the Action, coordination with the Services and important safeguards might need to be identified during permit review, such as implementing BMPs and avoidance and minimization measures.

#### 5.2.4 Coral

Seven coral species included in Appendix C, Table C.1.b. are boulder star coral, elkhorn coral, lobed star coral, mountainous star coral, pillar coral, rough cactus coral, and staghorn coral. Threats to coral species include climate change (including ocean warming and ocean acidification), diseases, habitat degradation, and land-based sources of pollution (e.g., sedimentation that affects water clarity).

#### 5.2.5 Mollusks

One mollusk is included in Appendix C, Table C.1.b.: the queen conch. Threats to this species include climate change (e.g., ocean warming and ocean acidification), diseases, habitat degradation, and land-based sources of pollution (e.g., sedimentation that affects water clarity).

# 6 Cumulative Effects

Cumulative effects are defined as the effects of future non-federal activities, including Tribal, State, local, and private actions on ESA-considered species or their critical habitat that are reasonably certain to occur in the Action Area for this BE.

## 6.1 Effects of Non-Federal Activities

ESA regulations (50 CFR §§ 402.02, 402.14) require the action agency to evaluate all effects of a proposed non-federal activity. Effects of an activity may occur later in time and may include consequences occurring outside the immediate area involved. Cumulative effects are those effects of future State or private activities, not involving federal activities, that are reasonably certain to occur within the Action Area of the federal Action subject to consultation 50 CFR § 402.02.

There are existing anthropogenic stressors currently impacting ESA-listed species. In addition, there are national-scale, non-federal activities that are reasonably certain to occur. These include watershed development, increased water use, and climate change (USEPA 2013). The "What is Your Vision for Florida's Future? Florida 2070 and Water 2070 Joint Project," prepared by 1000 Friends of Florida, University of Florida Geoplan Center, and Florida Department of Agriculture and Consumer Services (2017), addresses the current anthropogenic stressors that will impact development and water use in Florida in the future.

#### 6.1.1 Watershed Development

Florida's population continues to grow rapidly, and it is expected that 33.7 million people will reside in Florida by 2070—nearly 15 million more than were in residence in 2010 (1000 Friends of Florida et al. 2017). The greatest projected population growth is observed in the Central region of Florida (Figure 6-1). If growth continues as projected, nearly one-third of Florida's land will be developed, and development-related water demand will more than double. Due to the significant increase of projected development, especially in the Central region of Florida, agricultural lands will be consumed by residential, commercial, or industrial activity; therefore, agricultural irrigation development is expected to decrease except for the Southern region of Florida.



Figure 6-1 Comparison of projected 2010-2070 population change in four Florida regions (Source: 1000 Friends of Florida et al. 2017).

Projected development scenarios for the State of Florida depict a significant increase in land use between 2010 and 2070 (Figure 6-2). It is expected that by 2070, developed land in Florida will increase from 6,275,000 acres to 11,648,000 acres (1000 Friends of Florida et al. 2017). As a result of increased civil and industrial development, land usage for agricultural purposes in the State of Florida is expected to decrease from 7,586,000 acres to 5,422,000 acres by 2070. However, increased development has the potential for various adverse effects in terms of water quality.

Alterations to Florida's coastline can cause female sea turtles to look for other beaches for their nests or build their nests closer to the shoreline where they are more susceptible to drowning or getting washed away. Increased stormwater runoff can bring pollutants into the nesting areas and unnatural lighting can cause newly hatched turtles to crawl away from the ocean towards the development.

Developmental activity associated with new construction, as well as the overall growth of urban areas, directly affect the stormwater quality and can result in detrimental effects for aquatic communities. Not only does the increased flow from additional stormwater drainage systems affect receiving water bodies, but the stormwater runoff quality that is impacted by developmental activities such as litter, chemicals, metals, nutrients, pesticides, bacteria, sediment loads, and organic matter can harm biological health. These various pollutants and how they can impact water quality are discussed below.



Figure 6-2 A comparison of the State development scenarios (Source: 1000 Friends of Florida et al. 2016).

Increased sedimentation loads in water systems can result from construction and agricultural activities. There are significant differences in suspended solid concentrations between urban and non-urban stream systems. For example, a study comparison in South Carolina documented a higher total suspended solid concentration value of 2.7 kilograms per square meter per year in an urbanized coastal stream compared to a nearby forested stream that reported a value of 1.6 kilograms per square meter per year (Nagy et al. 2011). Both suspended and accumulated sediment can have adverse impacts. Sedimentation can infill porous areas in a stream/riverbed, eliminating niche areas that multiple aquatic organisms use for egg protection and/or attachment surfaces. Suspended sedimentation can lead to increased turbidity, which not only decreases dissolved oxygen levels (one of the most critical components of water quality) but also can interfere with benthic organisms and their feeding abilities. Makepeace et al. (1995) found that turbid water can be detrimental to aquatic biota. The authors found that total suspended solids concentration between 25 and 100 milligrams per liter "could reduce a river's primary biological productivity by 13 percent to 50 percent" (Makepeace et al. 1995). Increased sedimentation and nutrients are also detrimental to coral species that thrive in clear, low-nutrient water.

Pollutants, including chemicals, bacteria, metals, oils, and pesticides are commonly found in stormwater runoff. These substances and increased concentration levels in urban stormwater can be correlated with construction, developmental, and agricultural activities. Dependent on various factors, these chemicals and substances can result in either acute or chronic detrimental effects to biological life. Stream hydrologic, microbial, and physiochemical data collected in watershed areas in Florida with impervious surfaces ranging from 0-15 percent displayed higher pH, specific conductivity, elevated temperature, higher loads of nutrients (Cl<sup>-</sup>, NO<sup>3-</sup>, SO4<sup>2-</sup>, Na<sup>+</sup>, K<sup>+</sup>, Mg<sup>2+</sup>, Ca<sup>2+</sup>, and total phosphorus), higher bacterial concentrations (fecal coliform and *Escherichia coli*,) and increased hydrologic flashiness in areas with greater impervious surfaces (Nagy et al. 2011). The expected trend of exponential developmental growth of approximately 15.55 percent by 2070 for the State of Florida will increase impervious surface areas; conversely, stormwater quality will decrease while flow rates will demonstrate hydrologic flashiness. The

combined hydrologic flashes and influx of toxic substances in stormwater will negatively influence biological communities.

Heavy nutrient pollution is deemed one of America's most challenging environmental problems and is mainly influenced by nitrogen and phosphorus loading. Sources of heavy nutrient loading include but are not limited to agriculture, urban stormwater runoff, wastewater, power generation, and private fertilizer usage. Heavy concentrations of dissolved nitrogen and phosphorus in freshwater systems stimulate plant growth, especially algal growth, which ultimately can harm biological life due to the lack of available dissolved oxygen. A two-year monitoring study on reefs in Florida found that harmful algal blooms alter entire fish assemblages and can reduce both abundance and species richness (Baumberger 2008). With development in Florida increasing in the future, increased construction and higher rates of urban stormwater will consequently increase heavy nutrient loading in aquatic systems and reduce biotic diversity.

Another issue specific to Florida, related to watershed developmental concerns, is the potential for saltwater intrusion into confined aquifer systems. Historical evidence since the 1930s documents the concerns and issues related to saltwater intrusion in Florida as a result of encroaching development. In many areas around Florida, draining efforts were initiated to provide dry land for both developmental and agricultural purposes. There are severe and current concerns for the drinking water of nearly 2.5 million residents of Miami-Dade County, due to saltwater intrusion of the Biscayne aquifer stem (a consequence of the draining of the Everglades to allow for development) (Prinos 2014). As of 2011, Prinos determined that 463 square miles (mi<sup>2</sup>) of the Biscayne aquifer had been intruded with saltwater. Further pump-out of wetland or lake areas in Florida to create additional areas for development could impact groundwater quality by saltwater intrusion mechanisms.

#### 6.1.2 Increased Water Use

An increase in population size will place higher demands on the supply of water. Table 6-1 provides the historical water use data in Florida from 1975-2000. A large source of freshwater for the State of Florida is from underground aquifer systems. With growing pressures of climate change, influxes of anthropogenic demands, and unstable weather patterns, aquifer systems will be unable to meet recharge rates to suffice these conditions. Projected data report that by 2070, development-related freshwater demands will increase as much as 100 percent compared to 2010 (Figure 6-3). The Florida statewide demand for water in 2010 was 5,269,311,481 gallons per day; however, by 2070, the demand is expected to increase to 8,094,862,839 gallons per day (1000 Friends of Florida et al. 2017). It is estimated that approximately 90 percent of the consumed water in the State of Florida is sourced from groundwater, and only 10 percent is from surface water (Holt 2005).

		•	•	• • • •		
Category	1975	1980	1985	1990	1995	2000
Public Supply	1124	1406	1685	1925	2079	2437
Domestic self-supplied	228	243	259	299	297	199
Commercial – industrial mining	883	700	709	770	692	563
Agricultural irrigation <sup>a</sup>	2930	3026	2798	3495	3244	3923
Recreational irrigation <sup>b</sup>	n/a	n/a	182	310	281	411
Power generation	1608	1326	680	784	637	659
Total	6773	6701	6313	7583	7230	8192

#### Table 6-1 Historic Water Use in Florida (millions of gallons per day)

n/a - Not available

a - Withdrawals for crops, livestock, and fish farming;

b - Withdrawals for turfgrass and landscaping;

Source: Holt 2005, reprinted from Bureau of Economic and Business Research, University of Florida, Table 8:40, Florida Statistical Abstracts 2003

The higher demands for groundwater result in the extensive pumping of aquifer systems, which can result in saltwater intrusion. Depleting existing aquifer storage potential and further, harming future aquifer water quality by saltwater intrusion is expected as a consequence of increased water demands for the State of Florida. Receiving an average of 4.6 feet of rainfall annually, Florida has the second-highest precipitation rate behind Louisiana. With climatic changes and differing seasonal patterns, precipitation is likely to be influenced in the future. Approximately 70 percent of annual rainfall in Florida is lost due to evaporation and the remaining 30 percent flows through pervious surfaces to aquifers, surface water bodies, or to impervious surfaces (Holt 2005). Rising temperatures associated with climate change will increase the evaporation rates of Florida precipitation, therefore reducing overall recharge rates of surface and groundwater systems.



Figure 6-3. State water scenarios (Source: 1000 Friends of Florida et al. 2016).

Impacts of increased water demand may include reduced depth, areal coverage, and habitat quality for many aquatic communities and aquatic threatened and endangered species. Anthropogenic water demands are also likely to be exacerbated by climate change (see below), leading to greater hydrological variability with an increased probability of drought in many regions.

#### 6.1.3 Climate Change

Florida is among the states in the United States most vulnerable to climate change. The extent of coastline in Florida, along with its low elevation and heavy development of coastal and inland areas, will result in large-scale impacts both to human development and biological habitat as climate changes and sea levels continue to rise. According to the Fourth National Climate Assessment (4<sup>th</sup> NCA), "The Southeast's diverse natural systems, which provide many benefits to society, will be transformed by climate change. Changing winter temperature extremes, wildfire patterns, sea levels, hurricanes, floods, droughts, and warming ocean temperatures are expected to redistribute species and greatly modify ecosystems. As a result, the ecological resources that people depend on for livelihood, protection, and well-being are increasingly at risk, and future generations can expect to experience and interact with natural systems that are much different than those that we see today."

Annual average temperatures throughout the Southeast United States have been increasing since the 1970s. The 2010-2019 decade through 2017 was warmer than any previous decade, with much of this warming being experienced as higher nighttime temperatures and longer freeze-free season lengths (4th NCA). Statistically significant warming is projected for all parts of the United States through the next century, although, in the Southeast, that trend will be somewhat mitigated by latent heat release from increases in evapotranspiration. Average annual temperatures are projected to increase by 3.4 degrees Fahrenheit (°F) to 4.3 °F by the middle of the 2000s, and between 4.4 °F and 7.7 °F by 2100 (Vose et al. 2017).

Warmer winter temperatures in Florida will drive some ecosystem changes, especially when combined with sea level rise. Mangroves are being documented in more northerly locations than previously, thanks to more winters without freeze events, and are beginning to replace *Spartina* marshes. As sea level rises, mangroves and other tidal marsh species may be blocked from migrating up-gradient by sea walls, roads, ditches, or other natural or manmade landscape features, and may drown in place. In addition, warmer ocean water can cause coral to release the algae that live in their tissue and provide them food, which will cause the coral to die.

Landfall of tropical storms and hurricanes has occurred more frequently in Florida than any other state and increasing sea surface temperatures are expected to increase the frequency of high-intensity tropical cyclones (Mendelsohn et al. 2012). Additionally, increasing sea surface temperature has already negatively impacted Florida's coral reefs. As the ocean becomes more acidic due to the uptake of atmospheric carbon, it will become increasingly difficult for a variety of marine invertebrates to produce calcium carbonate shells and skeletons. This will impact biodiversity and the ecosystem services these species provide. These include shellfisheries production, wave and storm surge attenuation, water filtration, tourism, and transfer of energy via trophic processes to recreationally and commercially important finfish species.

Localized changes in precipitation regimes are also expected to occur due to climate change and will likely have a negative impact on a variety of habitats and species. Periods of unstable precipitation patterns of severe storms followed by longer drought periods pose not only risks for threatened and endangered species but also humans. Abiotic factors such as hydroperiod, water table height, salinity gradient, and surface water depth are likely to be disrupted as precipitation patterns change. Southeast Florida has already experienced hydroclimate variability, which has increased drought conditions causing decreased surface water levels, decreased groundwater recharge rates, lower groundwater tables, and ultimately leads to higher risks of saltwater intrusion (Abiy 2019).

Climate change impacts associated with rising sea levels pose immediate and long-term risks for land management practices of Florida. Coastal degradation and land loss due to sea-level rise will result in land fragmentation and habitat loss for biological organisms. Changes in abiotic factors from a number of climate change-related drivers are likely to cause habitats and species to shift their ranges in response. In addition to biological habitat loss, rising sea levels will also reduce available land in Florida for development. A study of ESA-listed subspecies in Florida showed both high vulnerability and low adaptive capacity in response to rising sea levels and habitat fragmentation (Benscoter et al. 2013). This will be especially problematic for habitats that are unable to migrate because of natural or anthropogenic barriers, and for species that are rare, occur in isolated populations, and/or have poor dispersal capabilities. For example, sea turtles need certain climate conditions for incubation of their eggs, and changes in climate could cause different conditions for precipitation and temperature as well as sea-level rise (Wellock 2020). Sea-level rise could submerge about 80 percent of current habitat that would otherwise be suitable for future egg incubation (Wellock 2020). Sustainable and climate change specific planning should be at the forefront of developmental planning, conservation efforts, and land usage for the State of Florida. FWC has an adaptation guide, A Guide to Climate Change Adaptation for Conservation that can be found on their website at https://myfwc.com/media/5864/adaptation-guide.pdf. For more information about what FWC is doing to address the impacts of climate change on fish and wildlife in Florida, see their website at https://mvfwc.com/conservation/special-initiatives/climate-change/.

#### 6.1.4 Summary of Cumulative Effects of Non-Federal Activities

As compared to the regulatory baseline, the Action does not authorize any new activities or increased discharge of pollutants that would significantly increase the magnitude of adverse cumulative environmental impacts to ESA-listed species, and the implementation of the State 404 program by FDEP will ensure effects on ESA-listed species will be evaluated and addressed at the project level.

# 7 State 404 Program NMFS Coordination

The following chapter describes the proposed coordination procedure between FDEP, FWC, and NMFS to avoid and minimize adverse effects to NMFS' jurisdictional species and designated critical habitat. The State 404 program rule under Rules 62-331.053(3)(a)4, 62-331.201(3)(k), and 62-331-248(3)(k) Florida Administrative Code (F.A.C.) and the EPA regulations at 40 CFR § 233.20 and 40 CFR § 230 prohibit issuance of a permit that, "jeopardizes the continued

existence of a species listed as endangered or threatened... or results in likelihood of the destruction or adverse modification of... a critical habitat under the Endangered Species Act of 1973, as amended." Under the coordination procedure, NMFS will recommend protection measures to appropriate to avoid or minimize adverse impacts associated with proposed projects and those protection measures must be included in a State 404 Program permit (Chapter 62-331.053(3)(a)4, F.A.C.). This will be ensured by the coordination process among the FDEP, FWC, and NMFS, which is referred to as technical assistance. The word "impact" used in Chapters 62-330 and 62-331, F.A.C., describes effects similar to "may affect" and "adverse effects" under the ESA. The terms "effect" and "impact" are used interchangeably throughout this document.

The US Army Corps of Engineers retains permitting authority in all estuarine, marine, and tidal waters, as well as all navigable waters that are on the retained waters list provided by the Corps (Corps jurisdictional waters are referred to as "retained waters"). The State's jurisdictional waters are only fresh waters that are not on the retained waters list, and wetlands that are Waters of the United States and not within a 300-foot buffer from the mean high water mark of retained waters (Florida's jurisdictional waters are referred to as "state-assumed waters"). In the State 404 Program, 'Retained Waters' is defined as (State 404 Program Handbook, section 2):

"Retained Waters" means those waters which are presently used or are susceptible to use in their natural condition or by reasonable improvement as a means to transport interstate or foreign commerce shoreward to their ordinary high water mark, including all waters which are subject to the ebb and flow of the tide shoreward to their mean high water mark, including wetlands adjacent thereto. The Corps will retain responsibility for permitting for the discharge of dredged or fill material in those waters identified in the Retained Waters List (Appendix A), as well as all waters subject to the ebb and flow of the tide shoreward to their mean high water mark that are not specifically listed in the Retained Waters List, including wetlands adjacent thereto landward to the administrative boundary. The administrative boundary demarcating the adjacent wetlands over which jurisdiction is retained by the Corps is a 300-foot guide line established from the ordinary high water mark or mean high tide line of the retained water. In the case of a project that involves discharges of dredged or fill material both waterward and landward of the 300-foot guide line, the Corps will retain jurisdiction to the landward boundary of the project for the purposes of that project only.

The FDEP and USFWS coordination process outlined in the USFWS's Programmatic Biological Opinion for the U.S. Environmental Protection Agency's Approval of FDEP's Assumption of the Administration of the Dredge and Fill Permitting Program under Section 404 of the Clean Water Act (2020), and in EPA's original BE for Florida's Assumption that was prepared when EPA consulted with USFWS, is similar to the coordination process that is now being proposed among FDEP, FWC, and NMFS;, however there are some differences.

Based on historical permit application reviews by the Corps prior to Florida's assumption of the CWA Section 404 program, it is anticipated that the majority of the reviews for NMFS species in state-assumed waters will be addressed using existing NMFS programmatic opinions or other NMFS guidance documents. When such tools are not available, technical assistance with the NMFS-PRD will be initiated, as needed.

These procedures are in place to ensure that any FDEP authorization issued will not jeopardize the continued existence of a federally listed species and will not result in adverse modification or destruction of designated critical habitat. This process will also include monitoring and reporting the number of permits with reviews of NMFS ESA-listed species. FDEP commits to providing annual reports to NMFS so that they may easily monitor any adverse impacts that may occur.

While this BE discusses NMFS species and critical habitat review coordination, it is important to note that both the State 404 permit and the State ERP program require protections for both federally listed and State-listed species. The FWC provides reviews for both the 404 and ERP programs regarding State-listed species in Chapter 68A-27, F.A.C.

FDEP coordination with NMFS-PRD will include FWC as the 404 species lead reviewer and coordinator. The FWC has partnered with FDEP to assist with the coordination of federally listed species reviews, which would take place concurrently with their review of impacts to State-listed species (per Chapters 62-330, F.A.C. and 68A-27, F.A.C.) for the State 404 Program. Throughout the rest of this document, FDEP and/or FWC may be referred to generally as the "State," meaning one of these two agencies within the State of Florida.

The State 404 program (Chapter 62-331, F.A.C.) incorporates a technical assistance process to assess whether effects to listed species and their critical habitats under the jurisdiction of the USFWS have reasonable potential to occur, and if so, further determines whether those effects are "likely to be an adverse effect" or "not likely to be an adverse effect." If adverse effects/impacts may occur, conditions or measures to avoid and minimize the impacts would be included as permit conditions and implemented by the Permittee. This State 404 program species assessment is modeled after the federal processes for determining, avoiding, and minimizing effects to listed species and ensures compliance with the ESA and the CWA during State 404 permit application review and permit issuance.

The resulting species coordination processes are intended to fulfill the following criteria when reviewing future State 404 permit applications: (1) the scope of the action is adequately described; (2) the physical, chemical, or biotic stressors to species that are likely to be produced as a result of the action is estimated; (3) the adverse effects of such activities on ESA-listed species and designated critical habitat is minimized; (4) likely adverse effects on listed species and critical habitat are monitored and evaluated; (5) permit compliance is monitored and enforced; and (6) if new information becomes available (including inadequate protection for species or low levels of compliance), the action is re-evaluated and modified if warranted. For an example of a use of these criteria, see the Endangered Species Act Section 7 Consultation, Programmatic Biological Opinion on the U.S. Environmental Protection Agency's Issuance and Implementation of the Final Regulations, Section 316(b) of the Clean Water Act, dated May 19, 2014.

In some cases, an applicant may have a project that had undergone previous Section 7 consultation or Section 10 permitting with the NMFS and is covered by a valid BiOp or is covered by a valid Habitat Conservation Plan (HCP) and Incidental Take Permit (ITP). To obtain a permit from NMFS for such take, an applicant must develop an HCP that meets specific requirements identified in Section 10(a)(2)(A) of the ESA and its implementing regulations at 50 CFR § 17.22 and § 17.32 and 50 CFR § 222.25, § 222.27, and § 222.31. HCPs are planning documents that describe the anticipated effects of the proposed taking; how those impacts will be

minimized or mitigated; and how the HCP is to be funded. Additional information about Habitat Protection Planning can be found at: <u>https://www.fws.gov/endangered/what-we-do/hcp-overview.html</u>.

The Habitat Conservation Planning Handbook (December 2016) provides guidance regarding how approved HCPs or new HCPs being developed are integrated with the ESA Section 7(a)(2)consultations with other federal agencies. This guidance specifically addresses potential federal agency permit actions within an HCP area for activities where take has been authorized in an ESA Section 10 permit.

The guidance provides the State the ability to recognize the pre-existing HCP/ITP and use it to comply with the ESA. There is a streamlined process that can occur if the application's activities are also covered by the HCP within the HCP plan area, and the effects to all ESA-listed species and/or critical habitat were analyzed in the NMFS's biological opinion. However, the streamlined process may not be available if a proposed project's actions involve activities, species, or areas that were not covered in the biological opinion associated with the HCP, or the FDEP intends to authorize the action in a manner that is inconsistent with the biological opinion.

In accordance with Section 7.2.3 of the FDEP's State 404 Handbook, it is FDEP's intent to adopt this guidance for its species coordination process to insure consistent and adequate protections for species and regulatory certainty to the regulated community, satisfying the requirements in Chapters 62-331, and 62-330, F.A.C. After technical assistance with the NMFS to confirm that the activities in the proposed State 404 permit are consistent with the HCP and biological opinion, FDEP will incorporate by reference in the State 404 permit, the terms and conditions contained in the approved HCP or biological opinion. Compliance with the HCP requires project applicants to implement the applicable and appropriate avoidance and minimization measures, and any other applicable terms and conditions as contained in the HCP.

# 7.1 Technical Assistance between the State and NMFS

Similar to the USACE Section 404 permit review process, applicants submitting a State 404 permit application are required to submit information that allows State and Federal agencies to sufficiently assess potential adverse impacts of the proposed project on listed species and any designated critical habitats. To that end, at a minimum the FDEP may request the following information as part of the State 404 application [Refer to Applicant's Handbook Volume I Section 10.2.; Application Form Section C Part 2(1) and Section I(6):

- Description of the proposed activity
- Description of the specific areas affected by the activity
- Description of protected or listed species and designated critical habitat that are affected by the activity
- Description on the manner in which species may be affected by the activity
- Analysis of any cumulative effects, which are the effects of future State or private activities that are reasonably certain to occur within the project area
- Relevant information (e.g., species surveys, etc.)
- When needed, proposed project designs and proposed permit conditions that would avoid, minimize, and/or mitigate the expected impacts to listed species and their habitats.

If incomplete, additional information will be requested during the information gathering and review processes.

#### 7.1.1 Protected Resources Division

The coordination process is intended to ensure that any authorizations issued under the State 404 Program do not jeopardize the continued existence of a federally listed species under NMFS jurisdiction or will not result in adverse modification or destruction of their designated critical habitat. Using programmatic biological opinions, such as the U.S. Army Corps of Engineers Jacksonville District's Programmatic Biological Opinion (JAXBO) dated November 20, 2017 and included in Appendix F of this BE, and the NOAA Fisheries endangered species mapping application (Section 7 Mapper) tool with a specific coverage for reviewing State 404 Program applications, the State will identify projects that may affect listed or proposed to be listed species under NMFS jurisdiction. If the activity is not covered under a programmatic opinion, the State will initiate technical assistance with NMFS by forwarding a link of application materials, the applicant's assessment and proposal, draft questions for FDEP's Request for Additional Information (RAI), and preliminary draft assessment and proposed project design criteria (if/when available). If NMFS has additional questions or needs additional information, the State will request those questions or needed information from the applicant. NMFS may provide input on the State's preliminary reviews and/or provide project design criteria to be incorporated into FDEP permits. If jeopardy is anticipated by NMFS, FDEP will issue a Notice of Intent to Deny the permit.

# 7.2 State 404 Program Permit Application Review Process

Upon receipt, applications submitted to FDEP will be forwarded to FWC, which will review the proposed activities and locations to determine if NMFS listed species and/or critical habitat may be affected. If so, FWC will use any programmatic tools available to address potential impacts. If tools are not available, FWC will forward preliminary assessments and links to project information on FDEP's website to NMFS PRD.

Pursuant to Rule 62-331.052, F.A.C., FDEP will review the application within 30 days of receipt of an application for a permit, or receipt of any additional information provided by the applicant in response to FDEP's request for additional information, 1) for administrative and technical completeness; and 2) to request any additional information required to publish the public notice to determine if the proposed activity meets the conditions for issuance. The applicant may voluntarily submit a written waiver of the above 30-day timeclock requirement to allow more time for FDEP to determine if additional information is required. However, FDEP is not obligated to accept the waiver or to delay sending the request for additional information. FDEP may request additional information as necessary during its review of any information that FDEP receives during the public comment period, at a public meeting, or during federal review.

Pursuant to Chapter 62-331.051(4), F.A.C., FDEP must not consider applications administratively complete (and noticed to the public) until it includes, among other things, data and information necessary for purposes of reviewing impacts to state and federal listed species, including compliance with any applicable requirements resulting from consultation with, or technical assistance by, the FWC, the USFWS, and NMFS. In addition, the applicant is required to submit application information including acknowledgement of, and/or compliance with, any requirements from NMFS.

Within 10 days of FDEP determining that an application is administratively complete pursuant to subsection 62-331.060(1), F.A.C., FDEP will provide public notice as described in subsection 62-331.060(2), F.A.C. In addition, FDEP will send a copy of the public notice to the EPA for those projects that the EPA reviews, in accordance with section 5.2.5 of the FDEP's State 404 Handbook.

Pursuant to 62-331.060(1), FDEP will provide public notice within 10 days of the following: (1) FDEP determination that an application for an individual permit or major modification is administratively complete; (2) FDEP notification to a permittee of revocation or suspension of a permit; and (3) issuance of an emergency field authorization. The FDEP shall provide public notice 30 days prior to any scheduled public meeting for such projects.

From date of publication, interested parties may express their views concerning the permit application, modification, revocation, or suspension for a period of 30 days, or 15 days for the projects listed in 62-331.060(b)(3)(b). The public notice comment period shall automatically be extended to the close of any public meeting if one is held. The presiding officer may also extend the comment period at the public meeting.

Time frames are identified in State regulations; however, FDEP intends to adhere to some transitional targeted timelines between these timeframes. According to FDEP, estimates for the transitional timelines are outlined in the steps below.

- (1) FDEP receives a permit application. Within 5 days, FDEP sends a copy to FWC.
- (2) Within 30 days (preferably 28 to allow FDEP time to incorporate questions), FWC screens the applications for potential NMFS species (Smalltooth sawfish, Atlantic sturgeon, or Shortnose sturgeon) or Smalltooth sawfish critical habitat, and submits a request for additional information that might be needed.
- (3) Within 10-15 days of receipt of the requested application materials and information from the applicant, FWC may send additional questions or additional needed information to FDEP. Or, if information needed is complete, FWC will respond to FDEP stating that the application is under review.
- (4) During the RAI and review process, FWC will identify affected species, affected critical habitats, types of effects, and potential protective measures that will become project modifications or permit conditions. Coordination with NMFS may or may not occur, depending on the project and availability of programmatic opinions and guidance to determine if there are potential likely adverse effects that, according to the programmatic opinions and guidance, may need to be addressed during coordination with NMFS. See Figure 7-1. If coordination is required based on the information from the programmatic opinions and guidance, or on a case by case basis where federal species guidelines or decision tools are not available, NMFS provides technical assistance, and any recommended protective measures will be incorporated as permit conditions.
- (5) FWC recommendations for permit conditions or statements that no permit conditions are needed are sent to FDEP as final comments. These final comments are required before the application can be deemed administratively complete or the public notice posted.
- (6) After application is deemed complete, FDEP has 10 days to issue a public notice for applications for individual permits.

- (7) The public notice is provided to the EPA, FWC and NMFS.
- (8) The public notice comment period is 15 days for minor projects and 30 days for more complicated ones.
- (9) If EPA waives its federal review, then after the public comment period closes, FDEP has either 60 or 90 days to issue or deny the permit, respectively.
- (10) If EPA's federal review is not waived, then the 40 CFR § 233.50 regulations and timeframes govern the timeframes and process of permit review.
  - a) The EPA has 10 days to send a copy of the public notice to NMFS (along with the Corps and FWS).
  - b) NMFS has 15 days to notify EPA whether it intends to comment.
  - c) If NMFS has notified EPA of its intent to comment, NMFS has 50 days from receipt of public notice from the EPA to send comments to the EPA.
  - d) The EPA has 90 days from receipt of the public notice to send comments or objection to FDEP. The final decision to comment, object, or to require permit conditions will be made by the EPA.

## 7.3 Review for impacts to NMFS species

The first step in assessing potential adverse impacts to listed species or those proposed to be listed as endangered or threatened, as well as their habitats, is to define the project area (similar to the "action area" during a federal Section 7 consultation). The project area can be larger than the immediate area of activity, since it is an identification of all areas that may affect listed species and their habitats directly or indirectly by the project's activities. Downstream effects on species must also be considered during the review.

For species within the NMFS PRD jurisdiction, the NMFS Species Directory

(https://www.fisheries.noaa.gov/species-directory/threatened-endangered) is used to locate threatened and endangered species in the region. In addition the NMFS ESA Critical Habitat Mapper (https://www.fisheries.noaa.gov/resource/map/national-esa-critical-habitat-mapper) can be used to determine whether critical habitat is also located in the relevant action area. While critical habitat is a special designation under the ESA, during project reviews all occupied habitat within the species range that may be adversely affected will be considered if the project's activity may affect listed or proposed species, even that which is not designated as critical habitat under the ESA.

#### 7.3.1 NMFS Programmatic Biological Opinions

An existing programmatic biological opinion that covers 10 activity categories and the three (3) NMFS species to be addressed (i.e., Smalltooth sawfish, Atlantic sturgeon and Shortnose sturgeon) is the JAXBO, which is provided in Appendix F of this BE. The State intends to use the JAXBO, including any subsequent revisions to the JAXBO, to review a project's potential effects to these species if the proposed activities are addressed in this opinion. If an activity is not addressed by JAXBO, it would require coordination with NMFS if effects are anticipated by the State. The JAXBO provides project design criteria (PDC) to limit the effects of projects for which it covers. It also provides effects determinations informed by the type of activity category and the species being addressed. Based on those determinations the State will determine whether continued coordination in accordance with the flow chart below (Figure 7-1) is needed.
Previously completed programmatic consultations (such as those associated with HCPs/ITPs) and broader, activity-based programmatic biological opinions, when available, may help identify where impacts/effects will occur and whether technical assistance with NMFS is needed.

#### 7.3.2 Case by Case Assessments When Opinions Are Not Available

For those species or activities that do not have federal species guidelines or decision tools, a case-by-case assessment must be performed. A preliminary assessment with proposed project design criteria and proposed protection measures must be submitted to FDEP by the applicant. Applicants will need to provide all of the information necessary to perform an assessment of potential impacts to listed species and their habitats, as well as submit proposed protection measures to avoid and minimize the anticipated impacts.

The following factors should be considered when evaluating the impacts/effects of the activity:

- Proximity of the activity to the species and/or designated critical habitat;
- Location and extent of the area of disturbance;
- Timing (with regards to sensitive periods of a species lifecycle);
- Duration of the activity or impact;
- Disturbance frequency;
- Nature of the effect (elements of a lifecycle, population size, variability, or distribution, physical and biological features of habitat, etc.);
- Affects to species and critical habitat adjacent to the area of activity; and
- Any downstream effects to listed species and critical habitat as a result of the activity.

Federal species guidelines, NMFS specific species webpages, the results of species surveys, relevant scientific literature, and other available sources of information are reviewed to develop preliminary conclusions of impact/effect as well as develop any potential protection measures. Physical or Biological Features essential to the conservation of the species as identified in the final rules designating critical habitat for specific listed species should be considered to determine whether there may be adverse modification of critical habitat.

Adverse impacts to species include the potential for harm to members of the species, such as injury, death, or those that occur by loss of feeding, breeding, or sheltering resources due to a project's activities affecting habitat where members of the species exist. Adverse impacts also include the potential for jeopardizing the continued existence of a species, or adversely modifying critical habitat. Adverse effects/impacts, or types of harm to one or more individuals, can result directly from dredging and filling activities involved with construction or demolition proposed by the project, as well as secondary impacts caused by the ongoing operations of the project once constructed. Assessment of adverse cumulative impacts must be considered during the review of State 404 permit applications; the assessment of expected impacts to species that may be caused from past authorized projects, as well as those future projects that are reasonably certain to occur. Adverse impacts to habitats, particularly critical habitat. These characteristics are important to the listed species using the area, and harm to species may occur temporarily during construction or permanently during operation or by alteration of the habitat.

The State will use information from the applicant and assess all the data available to them to determine if there will be impacts to any listed species and the severity of adverse impacts to each species and habitats present in the project area. A preliminary review may be submitted to the NMFS with a request for technical assistance or may be developed in coordination with the NMFS. Once the species coordination review and technical assistance with the NMFS is completed, the species coordination process with the NMFS ends for those species, and the protection measures recommended by NMFS are incorporated as conditions to any State 404 permit that is issued. At any time during the review of State 404 permit applications, if modifications to the project are proposed after the species coordination process, the species review will be revisited with the NMFS. If modification occurs after the permit is granted, and the modification causes effects not previously considered by NMFS, the applicant may be liable for any take under Section 9 of the ESA.

#### 7.3.3 Impact/Effect Determinations and Protective Measures

The word "impact" used in Chapters 62-330 and 62-331, F.A.C. describes effects similar to "may affect" and "adverse effects" under ESA. Throughout this document, these terms are used interchangeably. The State 404 program has two standards of review regarding the protection of listed species: the species protections required under CWA and ESA, and the species protections required under State Chapters 62-330, and 62-331, F.A.C. Under State rules, the requirements of CWA and ESA are incorporated into the species review process for adverse impacts to listed species and their habitats. The State rules are broad; the ERP rule and State 404 program rule protect not only federally listed and State-listed species, but all fish and wildlife in Florida.

Under the state rules, information about a species should be cross-referenced with knowledge of the project's activities and the project area to help predict whether and how the species at any life stage will respond when exposed to the effects of the activities. Based on best available data, if any of the following occurs it will be determined that the project will adversely impact/effect the species, and technical assistance with the NMFS may be needed to determine possible protective measures. The following statements can be used as guidance for the determination of "may adversely impact":

- Data indicate the species may be exposed to the elements of the activity and respond deleteriously upon exposure to elements of the activity or to stressors produced by the activity; or
- Data indicate the proposed activity will cause changes to the physical and biological features of critical habitat and produces exposure or stressor to species.

As the species coordination process progresses during the review of a State 404 permit application, proposed activities will be assessed for potential effects to species. The State will ask NMFS to confirm whether adverse impacts are anticipated to occur, and how significant the impacts are expected to affect the species or critical habitats. These determinations can be categorized as follows:

#### 1. No Effect

#### 2. May Affect, Not Likely to Adversely Affect:

This determination is reached when there is reasonable certainty that a proposed activity

may affect a species or designated critical habitat, but the effect is not anticipated to cause harm to a member of the species, nor cause adverse impacts to critical habitat that would result in harm to a listed species dependent on that habitat. The impact expected may not require protective measures (if discountable, or insignificant) or it may be deemed as not likely to adversely affect via required protective measures to avoid and minimize the effects.

#### 3. May Affect, Likely to Adversely Affect

This preliminary determination is reached when there is reasonable certainty that a proposed activity may adversely affect a species or designated critical habitat, and the effect is anticipated to cause harm to a member of the species and/or cause adverse impacts to critical habitat that would result in harm to a listed species dependent on that habitat. Harm to an individual(s) means injury or death. The level of harm, however, may or may not result in jeopardizing the continued existence of the species or destroying or adversely modifying critical habitat. The conclusion or Endangered Species Act (ESA) conclusion may be one of the following:

Projects with "May Affect, Likely to Adversely Affect" will require further, extensive review by NMFS to determine if effects can be addressed. After this review, there may be two final conclusions pursuant to the ESA:

#### Not likely to Jeopardize the Continued Existence of the species or Destroy or Adversely Modify Critical Habitat

After additional review, if protective measures are available that sufficiently avoids, minimizes, or mitigates the expected adverse impacts, a conclusion of "not likely to jeopardize" or "not likely to adversely modify critical habitat" would be the final effect determination. Any anticipated incidental take would need to be addressed for this final conclusion.

#### Jeopardizes the Continued Existence of the Species, and/or Destroys or Adversely Modifies Critical Habitat

The State 404 program rule includes stipulations (Rules 62-331.053(3)(a)4, 62-331.201(3)(k), and 62-331-248(3)(k) F.A.C.) that prohibit issuance of a permit that will likely jeopardize the continued existence of endangered or threatened species or result in the likely destruction or adverse modification of habitat designated as critical for these species as determined by NMFS. Through technical assistance with the NMFS, the State will be informed of when a proposed project and its activities are anticipated to jeopardize the continued existence of the species, or if critical habitat is destroyed, or adversely modified. Under these circumstances, the State, NMFS, and the applicant will discuss to determine what, if any, protective measures may be appropriate, or if permit denial is warranted.

While many adverse effects can be avoided and minimized during the species coordination review process, impacts that are likely to adversely affect a species (e.g., likely to cause incidental take or jeopardy) or critical habitat (e.g., destroy or adversely modify critical habitat) must be specifically identified for NMFS to review and provide proposed protective measures to avoid or minimize the adverse effects. These types of projects will receive the most stringent review and be documented in the Public Notice as well as the FDEP database and project file.

The public notice, required by Rule 62-331.060, F.A.C., will include effect determination statements and include the proposed protection measures, such as project design criteria and/or permit conditions. The effect determination statements for listed species may include one of the following: "No Effect," "Not Likely to Adversely Affect," or "Not likely to Jeopardize the Continued Existence of the Species," or "Jeopardizes the Continued Existence of the Species." The effect determination statements for critical habitat may include one of the following: "No Effect" or "Likely to Adversely Modify or Destroy Critical Habitat." The public notice will also include information about how any anticipated effects will be addressed by the permit.

The EPA will receive copies of all public notices where at least one protected or ESA-listed species and/or critical habitat has been identified as potentially affected by the project (for determinations other than "No Effect"). This coordination with EPA is required if there is reasonable potential for affecting listed species and/or their habitats. This provides an opportunity for the EPA to monitor the effectiveness of the species coordination process and to provide oversight for CWA and ESA compliance.

#### 7.3.4 Key commitments for the NMFS-FWC-DEP coordination process

The State of Florida is required to incorporate all NMFS recommendations for protection measures as State 404 program permit conditions (Subsection 62-331.054(1), F.A.C.). FDEP will collect data for impacts to NMFS species and critical habitat in State 404 issued permits and will provide annual reports to NMFS. This data will continue the monitoring of adverse effects on listed species and critical habitat, facilitating the State's ability to do compliance.

As described elsewhere in this BE, the NMFS Protected Species Division coordination process may involve a NMFS review of State 404 permit applications and preliminary assessments if the action or activity is not addressed by existing tools such as programmatic opinions or guidance documents.

The State will make the decision regarding the determination of "No Effect," using available data and decision tools as well guidance in NMFS Procedure 02-110-20 Protected Resources Management "No Effect" Determinations. Such "No Effect" determinations do not require NMFS concurrence; however, the State may request technical assistance from NMFS while making a decision on the effect determination. Projects forwarded to NMFS will be those which have been determined cannot be addressed under programmatic biological opinions (such as JAXBO) or other any developed GIS layers, the NOAA fisheries Section 7 mapper along with a buffer to ensure all effects, including indirect and appropriate downstream effects, are considered. Key commitments between the State and the NMFS to ensure a successful coordination review process may include:

- State procedures to review State 404 applications will be similar to and will utilize NMFS-approved permitting guidance, programmatic opinions and follow procedures similar to NMFS procedures currently used by the USACE.
  - The State will be the lead in identifying and addressing potential effects to species in NMFS jurisdiction when reviewing State 404 applications. These species include the Smalltooth sawfish, Atlantic sturgeon and the Shortnose sturgeon, and this review

includes potential adverse modifications to designated Critical Habitat for the Smalltooth sawfish.

- The State will identify issues and preliminarily determine the potential adverse effects from proposed projects using existing or future NMFS tools or guidance. Information and proposals provided by the applicant will be considered by the State and provided to NMFS.
- If the State determines that no effects will result from a proposed project, no coordination or concurrence from NMFS is required. If needed, NMFS may be contacted for technical assistance while making a decision for "No Effect." NMFS will assist the State when possible.
- The State will use programmatic biological opinions (such as JAXBO) and/or any other NMFS decision tools (such as a GIS layer with a buffer, created by NMFS) in order to determine a project's effect on a species or critical habitat. If impacts can be addressed by programmatic opinions or other available NMFS tools or guidance, no coordination or concurrence with NMFS may be required (see Figure 7-1). FDEP will incorporate NMFS programmatic recommendations as permit conditions and enforce those requirements for the life of the permit.
- When a proposed action or activity is not addressed with a NMFS programmatic opinion or decision tool, technical assistance from NMFS may be requested. The State will include a link to the project materials on FDEP's website and may include preliminary assessments for anticipated direct and/or indirect effects to species found (onsite, offsite, or downstream), potential impacts to critical habitat, and proposed recommended permit conditions address adverse impacts. NMFS will assist the State for these types of instances.
- For the types of reviews that require coordination or concurrence from NMFS, once technical assistance with NMFS begins, FDEP will be responsible for all inquiries to NMFS regarding the status and timing of their review. This includes inquiries for the purposes of FDEP internal review as well as providing updates to applicant or other outside entity inquiries.
- Comments coordinated with NMFS (programmatically or technical assistance) will be added to FWC's final comment letters to FDEP. If NMFS has provided written responses regarding recommended requirements, FWC will forward that documentation to FDEP for the official file. If no State-listed species comments are needed, NMFS comments will be forwarded to FDEP by FWC. FDEP will incorporate the recommendations as permit conditions and enforce those requirements for the life of the permit.
- If the State finds during the review that the applicant for a State 404 permit is the holder of a valid and active biological opinion, or Habitat Conservation Plan Incidental Take Permit (HCP/ITP), or a similar binding agreement that is issued by the NMFS and the species and activities described in the State 404 permit application are covered in the BiOp, HCP/ITP (or similar agreement), then no additional avoidance and minimization measures would be required. Documentation should be provided to NMFS to confirm the project is exactly the same. A reference to the biological opinion/ITP should be included as a recommended permit condition to FDEP.

- Technical assistance by NMFS may be accomplished by individual NMFS staff or by NMFS-approved tools, such as programmatic biological opinions, guidance, or NMFS GIS layers.
- The State may coordinate on NMFS issues while participating in the State 404 Program's Wildlife Technical Team meetings, when needed. Staff from NMFS-Protected Resources Division (PRD) may be invited to, or can be included in these meetings, as needed. This Team will oversee the coordination process, including but not limited to assisting in training efforts; providing expertise for staff to elevate questions and decision-making, as needed; and assisting in refining coordination processes, procedures, and future improvements, as needed, related to State of Florida permitting under Chapter 62-331, F.A.C.
- State permit review staff will be trained in the new NMFS coordination procedures and processes.
- If NMFS concludes that a permit application is likely to jeopardize the continued existence of species or adversely modify designated critical habitat and no protection measures are available to reduce the risk to an acceptable level, FDEP will issue a Notice of Intent to Deny the permit.

#### 7.4 EPA Oversight and Review

In accordance with the FDEP-EPA MOA and governing federal regulations (40 CFR § 233), EPA will retain federal oversight authority on the issuance of State 404 permits, with the ability to review applications, review proposed protection measures/conditions, recommend additional protection measures if deemed prudent and practicable, and, if necessary, object to the permit issuance. The EPA's federal authority also allows the EPA to intervene in the application review process, if needed, where there may be disagreements or issues that need to be resolved (40 CFR § 233.50). FDEP can also request the EPA's assistance in the application review process for the same reasons, if needed.

Pursuant to 40 CFR § 233.51(b)(2), there are specific categories of discharge the EPA can waive federal review of State 404 permit applications. Discharges with reasonable potential for affecting endangered or threatened species, however, are not waived and must be reviewed. For those projects subject to the EPA review, FDEP will send a copy of the public notice to the EPA, in accordance with section 5.2.5 of the FDEP's State 404 Handbook (also 40 CFR § 233.50(a)(1)). Under the State 404 program, the State, in coordination with NMFS, determines when projects have a reasonable potential for affecting listed or proposed to be listed species. Details regarding the level of effects to species and their critical habitat as well as proposed protection measures will be included in the public notice.



<sup>1</sup>For specific activities in Smalltooth sawfish habitat



Within 30 days of receipt of notice, the EPA will notify FDEP of the intent to comment upon, object to, or make recommendations with respect to a permit application, draft general permit or FDEP's failure to accept the recommendations of an affected state pursuant to § 233.31(a). Within this time period, the EPA may also request information from FDEP if the information provided is inadequate to determine whether the permit application or draft permit meets the requirements of the CWA, 40 CFR § 233.50, and the 404(b)(1) Guidelines. The EPA may notify FDEP that there is no comment but reserves the right to object within 90 days of receipt, based on any new information brought out by the public during the comment period or at a hearing.

Pursuant to 40 CFR § 233.50(b), the EPA will provide a copy of each public notice, each draft general permit, and other information needed for review of the application to the USACE, USFWS, and NMFS, within 10 days of receipt. These agencies will notify the EPA within 15 days of their receipt if they wish to comment on the public notice or draft general permit. Such agencies should submit their evaluation and comments to the EPA within 50 days of such receipt. The final decision to comment, object, or to require permit conditions will be made by the EPA.

If the EPA has given notice to FDEP of the intent to comment, those comments shall be submitted within 90 days of the receipt of the public notice, draft general permit, or FDEP's failure to accept the recommendations of an affected State, in accordance with 40 CFR § 233.31(a), an FDEP permit shall not be issued until the comments are resolved. Within 90 days, the EPA will provide a written statement with comments, objections, or recommendations; and the actions that must be taken by FDEP in order to eliminate any objections (see 40 CFR § 233.50(e) for more details). FDEP shall not issue a permit until steps required by the EPA are completed to eliminate an objection or incorporate a requirement for a permit condition to a permit application or draft general permit. Within 90 days of FDEP receipt of EPA's comments, the EPA may hold a public hearing on the objection or requirement (see 40 CFR § 233.50(g) and (h) for more details).

If no public hearing is held, within 90 days of receipt of the EPA's comments, FDEP shall either issue the permit revised to satisfy the EPA's objections or notify the EPA of its intent to deny the permit. In the event FDEP neither satisfies the EPA's objections, requirement for a permit condition, nor denies the permit, the EPA shall transfer the permit application to the USACE for processing.

## 7.5 Elevation of Issues and Dispute Resolution between FDEP, NMFS, and EPA

State 404 permits (as with USACE 404 permits) must comply with the Section 404(b)(1) Guidelines, which include a stipulation that "[w]here consultation with the Secretary of the Interior occurs under Section 7 of the Endangered Species Act, the conclusions of the Secretary concerning the impact(s) of the discharge on threatened and endangered species and their habitat shall be considered final" (per 40 CFR § 230.30(c)). Furthermore, per the State 404 Chapter 62-331, F.A.C. and CWA 404(g) and the 404(b)(1) Guidelines: 404 permits may not be issued if doing so jeopardizes the continued existence of species of plants and animals listed as endangered or threatened under the Endangered Species Act of 1973, as amended, or results in likelihood of the destruction or adverse modification of a designated critical habitat. In the event FDEP questions or disagrees with a local NMFS office about the necessity of project-specific, species-specific NMFS recommended protection measures being added as permit conditions for the purpose of avoiding and minimizing incidental take to fish and wildlife, the following steps would guide the resolution process.

If there is a dispute that occurs between FDEP and NMFS prior to FDEP issuing a public notice for a permit application, the beginning of a dispute resolution process will involve the State and NMFS office. If needed, the Florida 404 Endangered Species technical team composed of FDEP, FWC, and NMFS staff will assist in resolving any disputes. The EPA will not be involved at this stage but may later be involved should it receive a copy of the public notice from FDEP. The EPA would engage per the requirements for oversight of permits (40 CFR § 233.50). In the event the dispute cannot be resolved between FDEP and NMFS, the local office of NMFS will coordinate with higher levels within NMFS. If the NMFS and FDEP positions stay the same, FDEP will prepare a public notice that reflects NMFS's position and permit conditions and submit that public notice to the EPA (per the EPA's regulations governing federal oversight of State 404 permits - 40 CFR § 233.50) along with a transmittal letter to the EPA that succinctly presents the FDEP's concerns, alternative conditions, and any other relevant facts.

Per 40 CFR § 233.50, the EPA will then transmit FDEP's public notice to NMFS within 10 days of receipt. NMFS will notify the EPA within 15 days of receipt on whether it will submit comments. NMFS will have 50 days from receipt to provide comments to the EPA. The EPA shall make final decisions to comment, object or to require permit conditions (per 40 CFR § 233.50) and shall consider NMFS conclusions concerning impacts to ESA-listed species to be final (per CFR 230.30(c). FDEP or any interested person can ask the EPA to hold a public hearing on the EPA's objection or required permit conditions. At any point and whether or not the EPA has a public hearing, FDEP may accept the EPA's conditions and issue the permit, deny the permit, or notify the EPA that it is neither issuing nor denying the permit. In the latter case, the EPA would transfer the permit application to the Corps for processing (per 40 CFR § 233.50).

## 8 Conclusions

This BE evaluates the effects of the approval by the EPA for the State of Florida assuming Section 404 of the CWA on listed species in present time and estimates effects into the future within the Action Area, including in retained waters downstream of waters assumed by the State. All species that are currently listed, proposed to be listed, or might be listed in the future are discussed, with evaluations of effects on respective guilds. This review helps evaluate the effects of the State 404 program on any species that are currently listed or may become listed in the future.

The State's assumption of the dredge and fill permitting program under Section 404 of the CWA for waters assumable by the State is implemented by processes and procedures described in State regulations (Chapters 62-330 and 62-331, F.A.C.), agreements with EPA and USACE, and potentially also through the proposed species coordination process outlined in this BE.

In implementing the State 404 program, FDEP proposes to request technical assistance from the NMFS in addressing federally listed species and critical habitat issues. This coordination is to ensure that any permit issued by FDEP is not likely to jeopardize the continued existence of any listed species or to adversely modify or destroy designated critical habitat (pursuant to 40 CFR § 233.20(a)). FDEP will consider all information provided by NMFS and will include all species protection measures recommended by NMFS as conditions to the State 404 permit.

The assumption of Section 404 of the CWA by the State of Florida allows the State to assess and regulate all activities that occur in assumed waters, integrating protections that were previously regulated by two different programs. Chapter 62-331, F.A.C., by referencing much of the existing State ERP Chapter 62-330, F.A.C., facilitates the State's implementation of the CWA requirements.

The State 404 program is designed to provide conservation of ESA-listed species that is equivalent to the USACE Section 404 program. The ability of the State of Florida to process applications with permitting authority derived from Section 404 of the CWA in conjunction with the State ERP program allows for addressing potential adverse effects on both federally listed and State-listed species comprehensively, consistently, and efficiently. Conservation strategies for some species will benefit multiple species that share the same habitat. Some State-listed species also are proposed to be federally listed and assessing the expected adverse impacts from proposed projects on all listed species at the same time is likely to improve conservation efforts.

Florida's State ERP program is jointly implemented by FDEP, the WMDs, and certain local governments delegated by FDEP. At this time, the administration and implementation of the State 404 program is limited to FDEP. It is anticipated, however, that the State could seek to delegate elements of the 404 program to the WMDs in the future. If this is considered, approval from the EPA will be required prior to implementation. Upon any such approval, these agencies will be required to follow the same procedures outlined for the State 404 program.

#### 8.1 Effect of the Action on Tribal Resources or Interests

Consultations on permit actions taking place on waters in Indian Country continue to be conducted by the USACE. The Section 404 assumption rules include a procedure to offer an opportunity for tribal comment by the Seminole Tribe of Florida and the Miccosukee Tribe of Indians of Florida on any applications within a buffer zone around these lands.

A public notice is sent to the Seminole Tribe of Florida Environmental Resource Management Department for any activity that is within six miles of the Seminole Tribe of Florida's Big Cypress or Brighton Reservations; within two miles of the Seminole Tribe of Florida's Immokalee, Lakeland, or Fort Pierce Reservations; within one mile of the Seminole Tribe of Florida's Tampa, Coconut Creek, or Hollywood Reservations; within the Seminole Tribe's reserved rights areas, including, but not limited to, within Big Cypress National Preserve; within Big Cypress National Preserve addition lands; within Everglades National Park; within Rotenberger Wildlife Management Area; or within Water Conservation Area 3-A.

A public notice is sent to the Miccosukee Tribe of Indians of Florida for any activity that is within two miles of the Miccosukee Federal Reservation; Miccosukee Reserve Area; Krome Avenue, Dade Corners, Cherry Ranch, or Sherrod Ranch Reservations; and Coral Way, Lambick, or Sema Trust Properties. Also, for any activity within the Miccosukee Tribe's reserved rights areas, including, but not limited to, within Big Cypress National Preserve; within Big Cypress National Preserve addition lands; within Everglades National Park; within Rotenberger Wildlife Management Area; or within Water Conservation Area 3-A. With these provisions, no substantial difference in effects on tribal resources would be expected under the Action.

#### 8.2 Uncertainty Associated with the Effects Determination

The precise locations and types of activities included in future permit applications are unknown at this time. The number of applications and impacts may vary over time and with economic fluctuations and would be greatest in areas of rapid growth where the number of proposed projects tends to be high. However, over shorter time frames, one or a few large projects (for example, a major infrastructure initiative) can result in a concentration of proposed impacts in a given area or a given set of habitat types, with effects on one or a few species. Not all individual project proposals are known or can be anticipated in advance.

Other uncertainties can include the introduction of new invasive species, the spread of new pathogens, or other biotic or abiotic factors that can affect ESA-listed species or populations. For example, chytrid fungus and white-nose syndrome have contributed to the loss of populations of amphibians and bats, respectively, in recent years. Unexpected changes in the status of an ESA-listed species can lead to the increased importance of future project-level decisions, which under baseline conditions would have little effect on the species. Similarly, recovery of species can lead to increased resilience and, ultimately, in some cases, to delisting. If there are uncertainties, FDEP will work closely with the NMFS on a project-by-project basis to address those uncertainties and any impacts to ESA-listed species or their critical habitat to ensure effects are avoided and minimized.

#### 8.3 Overall Conclusions

This BE analyzes a total of 28 animal species in the State of Florida that are federally listed as endangered (15), threatened (12), or otherwise ESA-considered species (1) that must be addressed pursuant to Section 7 of the ESA, Section 404 of the CWA, and Chapter 62-331, F.A.C. for the State 404 program, are federally listed species or species proposed to be listed under the ESA (a total of 141 species). If these listed species have designated critical habitats or critical habitats proposed to be listed, the adverse effects and impacts on those habitats also must be analyzed and addressed under the State 404 program. The results of this analysis indicated that of the 28 species evaluated, 25 resulted in a "No Effect" determination and three<sup>3</sup>, the Atlantic sturgeon, short-nosed sturgeon, and smalltooth sawfish, resulted in a "May Affect – Not Likely to Adversely Affect." EPA has determined that for the three species, EPA's action results in a MA – NLAA because: 1) adverse effects will be avoided or minimized through the anticipated coordination process to take place between FDEP, FWC, and NMFS; 2)there is not a high likelihood that these species will be in the Action Area, and even if they are, the

<sup>&</sup>lt;sup>3</sup> Gulf sturgeon has shared jurisdiction with USFWS covering them in freshwater under the USFWS BiOp. We expect no effect on the Gulf sturgeon for waters within NMFS jurisdiction.

coordination process will ensure that impacts to the species and designated critical habitat are addressed; and 3) EPA will exercise its oversight role to provide additional protections to the listed species and designated critical habitat (see Appendix C for expanded information).

Historical Section 404 permitting by the USACE resulted in issuance of permits for one or more projects that adversely affected one or more listed species and their critical habitats. The USACE Section 404 process, in accordance with Section 7 of the ESA, employed various conservation measures to avoid and minimize adverse effects. Similar to the USACE Section 404 program, the State 404 program can avoid and minimize significant adverse effects through the technical assistance process between FDEP, FWC and the NMFS, as demonstrated in the past through the USACE Section 404 review and permitting processes.

It is anticipated that with the proposed technical assistance to be set in place by agreement among FDEP, FWC, and NMFS, Chapter 62-331 F.A.C., and MOAs provided in FDEP's assumption application package, the Action will result in procedural and substantive protections that are at least as protective as the protections afforded by the USACE Section 404 program and through Section 7 consultation with the USACE under the ESA. The State 404 program is prohibited from issuing a permit that would jeopardize the continued existence of a species or adversely modify designated critical habitats because of the prohibition in the 404(b) Guidelines. The State's assumption of CWA Section 404 permit program is not expected to result in significant changes from the number, type, or location of permit applications and proposed projects typically issued by the USACE.

## 9 Literature Cited

- 1000 Friends of Florida, University of Florida Geoplan Center, and Florida Department of Agriculture and Consumer Services. 2017. *A Special Report–What is Your Vision for Florida's Future?* Florida 2070 and Water 2070 Joint Project. Accessed January 15,2020. https://1000friendsofflorida.org/florida2070.
- Abiy, A. Z., A. M. Melesse, W. Abtew, and D. Whitman. 2019. Rainfall trend and variability in Southeast Florida: Implications for freshwater availability in the Everglades. *PLOS ONE* 14(2): e0212008.
- Albright, R., B. Mason, M. Miller, and C. Langdon. 2010. Ocean acidification compromises recruitment success of the threatened Caribbean coral *Acropora palmata*. *Proceedings of the National Academy of Sciences* 107:20400-20404.
- Baumberger, R. E. 2008. *The Impacts of Harmful Algal Blooms on a Florida Reef Fish Community*. Master's thesis. Florida Atlantic University, Boca Raton, FL. https://fau.digital.flvc.org/islandora/object/fau%3A2853/datastream/OBJ/view/impacts\_o f harmful algal blooms on a Florida reef fish community.pdf.
- Benscoter, A. M., J. S. Reece, R. F. Noss, L.A. Brandt, F. J. Mazzotti, et al. 2013. Threatened and endangered subspecies with vulnerable ecological traits also have high susceptibility to sea level rise and habitat fragmentation. *PLOS ONE* 8(8): e70647.

- Dahl, T. E. 2005. Status and Trends of Wetlands in the Conterminous United States 1998 to 2004. U.S. Fish and Wildlife Service, Fisheries and habitat Conservation, Washington, DC.
- Dahl, T. E. 2011. *Status and Trends of Wetlands in the Conterminous United States 2004-2009*. U.S. Department of the Interior, Fish and Wildlife Service, Washington, D.C.
- FDEP (Florida Department of Environmental Protection). 2019. *Wetlands Mitigation*. Accessed January 27, 2020. https://floridadep.gov/water/submerged-lands-environmental-resources-coordination/content/mitigation.
- FDEP (Florida Department of Environmental Protection). 2020a. *State 404 Program Applicant's Handbook*. FDEP. Version 02/11/2020. Accessed February 24, 2020. https://www.flrules.org/gateway/reference.asp?No=Ref-12064.
- FDEP (Florida Department of Environmental Protection). 2020b. *Water Quality Standards*. FDEP. Accessed February 16, 2020. https://floridadep.gov/DEAR/Water-Quality-Standards.
- Flather, C. H., M. S. Knowles, and I. A. Kendall. 1998. Threatened and endangered species geography. *BioScience* 48:365-375.
- FNAI (Florida Natural Areas Inventory). 2010. Guide to the Natural Communities of Florida. 2010 edition. Tallahassee, FL. Accessed February 24, 2020. https://www.fnai.org/PDFs/Full\_FNAI-Natural-Community-Classification-Guide-2010\_20150218.pdf.
- FWC (Florida Fish and Wildlife Conservation Commission FWC). 2016. Florida's Imperiled Species Management Plan. November 2016, with amendments incorporated December 2018. Tallahassee, FL.
- FWC (Florida Fish and Wildlife Conservation Commission). 2018. *Florida's endangered and threatened species*. Updated December 2018. FWC, Tallahassee, FL.
- FWC (Florida Fish and Wildlife Conservation Commission). 2019. *Florida's Wildlife Legacy Initiative: Florida's State Wildlife Action Plan.* Tallahassee, FL.
- Florida Statutes (F.S.) Chapter 373. Water Resources.
- Hefner, J. M. 1986. Wetlands of Florida, 1950s to 1970s. Pp. 23-31 in: E. D. Estevez, J. Miller, J. Morris, and R. Hamman (eds.), *Managing Cumulative Effects in Florida Wetlands*. New College Environmental Studies Program Publication No. 37. Omnipress, Madison, WI.
- Holt, L. 2005. Avoiding a water crisis in Florida: how should water resources be managed in response to growth? *Florida Water Resources Journal* (October):16-22.
- IPCC (The Intergovernmental Panel on Climate Change). 2002. Climate Change and Biodiversity. IPCC Technical Paper V. Gitay, H., A. Suarez, R.T. Watson, and D.J. Dokken (editors). IPCC Geneva, Switzerland.

- IPCC (The Intergovernmental Panel on Climate Change). 2013. *Fifth Assessment Report of the Intergovernmental Panel on Climate Change*. Cambridge, United Kingdom, New York, NY.
- Makepeace, D. K., D. W. Smith, and S. J. Stanley. 1995. Urban stormwater quality: summary of contaminant data. *Critical Reviews in Environmental Science and Technology* 25:93-139.
- McCarty, J. P. 2001. Ecological consequences of recent climate change. *Conservation Biology* 15:320-331.
- Mendelsohn, R., K. Emauel, S. Chonabayshi, and L. Bakkenshen. 2012. The impacts of climate change on global tropical cyclone damage. *Nature Climate Change* 2: 205-209.
- Nagy, R. C., B. G. Lockaby, L. Kalin, and C. Anderson. 2011. Effects of Urbanization on Stream Hydrology and Water Quality: the Florida Gulf Coast. *Hydrological Processes* 26:2019-2030.
- NOAA (National Oceanic and Atmospheric Administration). 2017a. Biological Opinion on the Approval of Florida Estuary Specific Numeric Nutrient Criteria under section 303(c) of the Clean Water Act for Total Phosphorus, Total Nitrogen, and Chlorophyll-a in 42 Estuary Segments. FPR-2017-9187. Silver Spring, MD.
- NOAA (National Oceanic and Atmospheric Administration). 2017b. Shoreline Mileage of the United States. Accessed March 2023. https://coast.noaa.gov/data/docs/states/shorelines.pdf.
- NOAA Fisheries (National Oceanic and Atmospheric Administration Fisheries). 2020. *Smalltooth sawfish. Species directory*. Accessed January 7, 2020. https://www.fisheries.noaa.gov/species/smalltooth-sawfish#overview.
- North American Wetlands Conservation Council (Canada) (NAWCCC). 2000. *Wetland Mitigation in Canada. A Framework for Application. No. 2000-1.* Accessed January 28, 2020. http://nawcc.wetlandnetwork.ca/Wetland%20Mitigation%202000-1.pdf.
- Parry, M., O. Canziani, J. Palutikof, and P. J. van der Linden, editors. 2007. Climate Change 2001: Impacts, Adaptation, and Vulnerability. Contribution of working group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom.
- Prinos, S. T. 2014. Origins and delineation of saltwater intrusion in the Biscayne Aquifer and changes in the distribution of saltwater in Miami-Dade County, Florida. *Scientific Investigations Report*. doi:10.3133/sir20145025.
- USACE (U.S. Army Corps of Engineers). 2019a. *Central and Southern Florida (C&SF) Project* – *Fact Sheet*. USACE, Jacksonville District, Jacksonville, FL. Accessed February 24, 2020. https://www.saj.usace.army.mil/About/Congressional-Fact-Sheets-2019/C-SF-Project-C.
- USACE (U.S. Army Corps of Engineers). 2019b. *Ecosystem Restoration*. USACE, Jacksonville District, Jacksonville, FL. Accessed February 24, 2020. https://www.saj.usace.army.mil/Missions/Environmental/Ecosystem-Restoration/.

- USACE (U.S. Army Corps of Engineers) and South Florida Water Management District (SFWMD). 2007. Broward County Water Preserve Areas Comprehensive Everglades Restoration Plan (CERP) Final Project Implementation Report and EIS Briefing to the Civil Works Review Board. Accessed January 27, 2020. https://usace.contentdm.oclc.org/digital/collection/p16021coll7/id/5154.
- USEPA (U.S. Environmental Protection Agency). 2013. *ESA Biological Evaluation for CWA Section 316(B) Rulemaking*. USEPA, Office of Water Engineering and Analysis Division, Washington, DC.
- USEPA (U.S. Environmental Protection Agency). 2012. National Coastal Condition Report IV. EPA-842-R-10-003. USEPA, Office of Research and Development and Office of Water, Washington, DC. https://www.epa.gov/national-aquatic-resource-surveys/national-coastal-condition-reports.
- USFWS (U. S. Fish and Wildlife Service). 1996. *The South Florida ecosystem*. South Florida Ecological Services Field Office, Vero Beach, FL.
- Vose, R.S., D.R. Easterling, K.E. Kunkel, A.N. LeGrande, and M.F. Wehner, 2017: Temperature changes in the United States. In: *Climate Science Special Report: Fourth National Climate Assessment, Volume I*, ed. D.J. Wuebbles, D.W. Fahey, K.A. Hibbard, D.J. Dokken, B.C. Stewart, and T.K. Maycock. U.S. Global Change Research Program, Washington, DC, USA, pp. 185-206, doi: 10.7930/J0N29V45.
- Wellock, B. 2020. Coastal development, changing climate threaten sea turtle nesting habitat. *Florida State University News*. Florida State University. Tallahassee, FL. Accessed January 2023. https://news.fsu.edu/news/science-technology/2020/08/25/coastaldevelopment-changing-climate-threaten-sea-turtle-nesting-habitat/.

## Appendix A

April 15, 2020, Letter from NMFS to FDEP Regarding NMFS Jurisdiction



UNITED STATES DEPARTMENT OF COMMERCE National Oceanic and Atmospheric Administration NATIONAL MARINE FISHERIES SERVICE Silver Spring, MD 20910

April 15, 2020

Heather Mason, PWS Environmental Administrator Florida Department of Environmental Protection Submerged Lands and Environmental Resources Coordination 2600 Blair Stone Road, MS 2500 Tallahassee, FL 32399-2400

RE: National Marine Fisheries Service in the Clean Water Act Section 404 Assumption by the State of Florida

Dear Ms. Mason:

On November 22, 2019, the National Marine Fisheries Service (NMFS) received your request for input on a draft species list for Florida's assumption of Clean Water Act Section 404 permitting from the U.S. Army Corps of Engineers (USACE). Through subsequent communications with you and your staff, a review of state mapping products, NMFS'own mapping analysis, and a review of state-provided documents about the assumption, we conclude that Endangered Species Act (ESA)-listed species under NMFS' jurisdiction do not occur in waters that are assumable by the state.

We specifically analyzed the possible spatial overlap of the assumption with waters used by shortnose sturgeon, Atlantic sturgeon, smalltooth sawfish, and Gulf sturgeon. Based on that analysis, shortnose sturgeon and Atlantic sturgeon occur in the St. Marys and St. Johns Rivers, which are included on the USACE retained waters list. Smalltooth sawfish occur in waters "subject to the ebb and flow of the tide" which will also remain under the USACE's jurisdiction, per the draft State 404 Program Applicant's Handbook definition of "Retained Waters." Therefore, the USACE will retain ESA Section 7 responsibility for proposed Section 404 actions in the waterways where NMFS's trust resources are most likely to occur.

For Gulf sturgeon, which has shared jurisdiction between NMFS and the U.S. Fish and Wildlife Service, the U.S. Fish and Wildlife Service is responsible for all consultations regarding Gulf sturgeon and critical habitat in riverine habitat units (final rule designating critical habitat for the Gulf sturgeon – 68 FR 13370). Rivers in Florida that include riverine critical habitat units (i.e., Escambia, Yellow, Choctawhatchee, Apalachicola, and Suwannee rivers) and river areas where Gulf sturgeon are known to occur (e.g., lower Ochlockonee River) are all listed by the USACE as retained waters.

Based on this determination we also assume that the Environmental Protection Agency will make a "no effect" determination for NMFS' ESA-listed species that were originally identified as part of this proposed assumption.



We appreciate all the information and assistance you provided in making this determination. If you have any questions, please contact Dr. Pat Shaw-Allen at (301) 427-8473, or by e-mail at pat.shaw-allen@noaa.gov or me at (301) 427-8495 or by e-mail at cathy.tortorici@noaa.gov.

Sincerely, TORTORICI.CAT Digitally signed by TORTORICI.CATHRYN.E.13 HRYN.E.136582 6850 Date: 2020.04.15 13:25:25 -04'00'

Cathryn E. Tortorici Chief, ESA Interagency Cooperation Division Office of Protected Resources

cc: Karen Myers, US Fish and Wildlife Service Robert Tawes, US Fish and Wildlife Service David Bernhart, NMFS, Southeast Regional Office

# **Appendix B** Species Accounts

The primary source of information for these species accounts is the Species Directory developed by NOAA Fisheries. References for each species are provided at the end of this appendix.

#### **B.1 Vertebrates**

#### B.1.1 MAMMALS

#### Blue Whale (Balaenoptera musculus)

The blue whale is listed as endangered under the Endangered Species Act of 1973 (ESA) and as protected under the Marine Mammal Protection Act (MMPA). Critical habitat has not been designated.

Blue whales are found in all oceans except the Arctic Ocean. In general, distribution is driven largely by food availability—they occur in waters where krill are concentrated. In the North Atlantic Ocean, their range extends from the subtropics to the Greenland Sea. Along the West Coast of the United States, eastern North Pacific blue whales are believed to spend winters off the coasts of Mexico and Central America. Blue whales with young calves are regularly observed in the Gulf of California (Sea of Cortez) from December through March. In the northern Indian Ocean, there is a "resident" population. In the Southern Hemisphere, Antarctic blue whales occur mainly in relatively high latitude waters south of the Antarctic Convergence and close to the ice edge in summer. The primary diet of blue whales is krill, but fish and copepods may occasionally be included.

The primary threats to the blue whale are vessel strikes, entanglement in fishing gear, ocean noise, habitat degradation, pollution, and climate change.

#### Byrde's Whale, Gulf of Mexico (GOMx) (Balaenoptera edeni [unnamed subspecies])

The Bryde's (pronounced "broodus") whale is a member of the baleen whale family. It is considered one of the "great whales," or rorquals, which is a group that includes blue whales and humpback whales. Bryde's whales are found in warm, temperate oceans, including the Atlantic, Indian, and Pacific oceans. Some populations of Bryde's whales make short migratory movements with the seasons, while others do not migrate, making them unique among other migrating baleen whales. In 2019, the National Marine Fisheries Service (NMFS) listed the GOMx Byrde's whale as endangered under the ESA (84 FR 15446, April 15, 2019).

Research suggests that Bryde's whales spend most of the day within 50 feet of the water's surface. Their diet consists of krill, copepods, red crabs, and shrimp as well as a variety of schooling fish, such as herring, mackerel, pilchards, and sardines. Bryde's whales use different methods to feed in the water column, including skimming the surface, lunging, and creating bubble nets. The GOMx Bryde's whale is presently in danger of extinction (i.e., meets the definition of endangered) throughout its range because of its small population size and restricted range and the threats from energy exploration, development, and production; oil spills and oil spill response; vessel collision; fishing gear entanglement; and anthropogenic noise. Critical habitat is not determinable at this time but will be proposed in a future rulemaking.

#### Fin Whale (Balaenoptera physalus)

The fin whale is listed as endangered throughout its range under the ESA and as depleted throughout its range under the MMPA. Critical habitat has not been designated.

Fin whales are typically found in deep, offshore waters of all major oceans, primarily in temperate to polar latitudes. They are less common in the tropics. Most migrate from the Arctic and Antarctic feeding areas in the summer to tropical breeding and calving areas in the winter. The location of winter breeding grounds is not known. Fin whales travel in the open seas, away from the coast, so they are difficult to track. There are three named subspecies of fin whale: *B. physalus* in the North Atlantic and North Pacific, *B. physalus quoyi* in the Southern Ocean, and *B. physalus patachonica* in the mid-latitude Southern Ocean.

Fin whales are fast swimmers and are often found in social groups of two to seven individuals. In the North Atlantic, they are often seen feeding in large groups that include humpback whales, minke whales, and Atlantic white-sided dolphins. During the summer, fin whales feed on krill, small schooling fish, and squid. Fin whales fast in the winter while they migrate to warmer waters. Little is known about the social and mating systems of fin whales. For management purposes, they are split into four stocks to provide population counts: California/Oregon/Washington, Hawaii, Alaska, and Western North Atlantic.

Like all large whales, fin whales were hunted by commercial whalers, which severely lowered their populations. Currently their primary threats include vessel strikes, entanglement in fishing gear, ocean noise, and climate change.

#### Humpback Whale (Megaptera novaeangliae)

The humpback whale was listed by the National Oceanic and Atmospheric Administration (NOAA) as endangered under the Endangered Species Conservation Act of 1969 in 1970 and then under the ESA in 1973 because commercial whaling operations had severely reduced humpback whale numbers from historical levels. NOAA Fisheries worked worldwide to identify and implement protections for humpback whales. The International Whaling Commission's final whaling moratorium on commercial harvest, in effect since 1985, played a major role in the recovery of humpback whales. Currently, four out of the 14 distinct population segments (DPSs) are still protected as endangered by NMFS, and one is listed as threatened (81 FR 62259, September 2016). Three humpback whale stocks in U.S. waters are designated as depleted under the MMP.

During the warmer months (and occasionally into cooler months), humpback whales spend most of their time feeding and building up fat stores (blubber) to sustain them throughout the winter. Humpback whales filter-feed on small crustaceans (mostly krill) and small fish. They use several techniques to help them herd, corral, and disorient prey that can include using bubbles, sounds, the seafloor, and even their pectoral fins.

The primary threats to humpback whales include marine noise, climate change, vessel strikes, and entanglement in fishing gear (NMFS 2023).

#### North Atlantic Right Whale (Eubalaena glacialis)

The North Atlantic right whale is one of the world's most endangered large whale species. It has been listed as endangered under the ESA since 1970. Critical habitat was designated on February 26, 2016, and includes the coast of New England (foraging area) and the southeast U.S. coast from Cape Fear, North Carolina, to below Cape Canaveral, Florida (calving area).

North Atlantic right whales primarily occur in Atlantic coastal waters on the continental shelf, although they also are known to travel far offshore, over deep water. Right whales migrate seasonally and may travel alone or in small groups. In the spring and summer and into fall, many of these whales can be found in waters off New England and further north into Canadian waters, where they feed and mate. Each fall, some right whales travel more than 1,000 miles from these feeding grounds to the shallow, coastal waters of their calving grounds off South Carolina, Georgia, and northeastern Florida, although migration patterns vary.

Right whales feed by opening their mouths while swimming slowly through large patches of copepods and other zooplankton. Right whales feed anywhere from the water's surface to the bottom of the water column. Groups of right whales may be seen actively socializing at the water's surface, known as surface-active groups, or SAGs. Mating and socializing occur in SAGs, which are observed during all seasons and in all habitats. Right whales communicate using low-frequency moans, groans, and pulses, which may maintain contact between individuals, communicate threats, signal aggression, or be used for other social reasons.

The primary threats to the species are entanglement in fishing gear and vessel strikes. Other threats include climate change—which may alter their migratory patterns and feeding areas—and the impacts of ocean noise on their ability to communicate, find food, and navigate.

#### Rice's Whale (Balaenoptera ricei)

The Rice's whale is one of the most endangered species of whale in the world. In 2021, NOAA Fisheries revised the common and scientific name of the listed entity to Rice's whale, *Balaenoptera ricei*, and its classification to species to reflect the new scientifically accepted taxonomy and nomenclature of the species. Rice's whales are protected under the MMPA and were listed as endangered in 2019 under the ESA. Critical habitat has not been designated.

The Rice's whale is one of the few types of baleen whales to prefer warmer, tropical waters and that does not make long-distance migrations. They remain in the Gulf of Mexico year-round, along the continental shelf break between 100 meters (m) and about 400 m deep. The historical distribution of Rice's whales might once have encompassed the northern and southern Gulf of Mexico. Rice's whales are usually seen alone or in pairs, but may form larger, loose groups associated with feeding. Little is known about their foraging ecology and diet. Data from two Rice's whales, however, suggest they may mostly forage at or near the seafloor.

The Rice's whale's very small population size and limited distribution increase its vulnerability to threats. The most significant threats they face are from energy exploration and development, oil spills and spill response, vessel strikes, ocean noise, ocean debris, aquaculture, and entanglement in fishing gear.

#### Sei Whale (Balaenoptera borealis)

The sei whale is listed as endangered under the ESA and as depleted under the MMPA. Critical habitat has not been designated.

Sei whales have a wide distribution and live in subtropical, temperate, and subpolar waters around the world. They prefer temperate waters in the mid-latitudes and can be found in the Atlantic, Indian, and Pacific oceans. During the summer, they are commonly found in the Gulf of Maine and on Georges Bank and Stellwagen Bank off the U.S. coast in the western North Atlantic. The movement patterns of sei whales are not well known, but they are typically observed in deeper waters far from the coastline. Sei whales have an unpredictable distribution.

The species usually is observed alone or in small groups of two to five animals. An average sei whale eats about 2,000 pounds of food per day. They can dive for 5 to 20 minutes to feed on copepods, krill, small schooling fish, and cephalopods by both gulping and skimming. They prefer to feed at dawn and may exhibit unpredictable behavior while foraging and feeding on prey.

The global sei whale population was greatly decreased by historical commercial whaling during the 19th and 20th centuries. An estimated 300,000 sei whales were killed for their meat and oil. The primary threats they face today are vessel strikes, entanglement in fishing gear, ocean noise, and climate change.

#### Sperm Whale (*Physeter macrocephalus*)

The sperm whale is listed as endangered under the ESA and as depleted under the MMPA. Critical habitat has not been designated.

Sperm whales are the largest of the toothed whales and have one of the widest global distributions of any marine mammal species. They are found in all deep oceans, from the equator to the edge of the pack ice in the Arctic and Antarctic. Their distribution is dependent on their food source and suitable conditions for breeding and varies with the sex and age composition of the group. Some populations appear to have different migration patterns by life history status, with adult males making long oceanographic migrations into temperate waters and females and young staying in tropical waters year-round.

Sperm whale's diet consists of species such as squid, sharks, skates, and fish that also occupy deep ocean waters. Sperm whales can consume about 3 to 3.5 percent of their body weight per day. Most females will form lasting bonds with other females of their family, and, on average, 12 females and their young will form a social unit. While females generally stay with the same social unit in and around tropical waters their entire lives, young males will leave when they are between 4 and 21 years old and can be found in "bachelor schools," composed of other males approximately the same age and size. As males get older and larger, they begin migrating toward the poles. As a result, bachelor schools become smaller, and the largest males are often found alone.

Sperm whales are named after the waxy substance—spermaceti—found in their heads. The spermaceti is an oil sac that helps the whales focus sound. Spermaceti was used in oil lamps, lubricants, and candles. Sperm whales were a primary target of the commercial whaling industry from 1800 to 1987, which nearly decimated all sperm whale populations. While whaling is no longer a major threat, sperm whale populations are still recovering. Primary threats they face today include vessel strikes, entanglement in fishing gear, ocean noise, marine debris, climate change, and oil spills.

#### West Indian Manatee (Trichechus manatus)

The West Indian manatee was listed as endangered effective March 11, 1967, under the Endangered Species Preservation Act of 1966 (32 FR 4001). Listing was revised in 1970 to include the Caribbean subspecies and South American subspecies (*T. m. manatus*) (35 FR

18319). The species was down-listed to threatened on April 5, 2017 (82 FR 16668). Critical habitat was designated in 1976 (50 CFR § 17.95(a)). This species is currently listed by the U.S. Fish and Wildlife Service (USFWS) and the State of Florida but is not listed by the NMFS.

There are two subspecies of West Indian manatee, T. m. manatus and T. m. latirostris. The latter of these subspecies occurs in Florida and the Gulf of Mexico (USFWS 1999, 2007). There are four unique Florida subpopulations/management units (genetic exchange occurs among populations) that exhibit varying degrees of site fidelity: The Upper St. Johns River subpopulation, the Northwest subpopulation, the Southwest subpopulation, and the Atlantic Coast subpopulation (USFWS 2007). Florida manatees occur in freshwater, brackish, and marine environments, including coastal river estuaries, sloughs, canals, creeks, and lagoons (USFWS 1999). The species requires a source of freshwater for drinking (USFWS 2001). They are not cold-tolerant, prefer waters with temperatures above 68 °F, and remain in warm-water sites around the state during the winter (e.g., warm water springs and power plant outfall sites) (USFWS 2001, 2007). Breeding may occur year-round, although peak breeding season is suspected to occur from March through November. Manatees form "mating herds" (one female and multiple males) to breed. Manatees reach sexual maturity at around four years. Gestation is roughly 11 to 14 months, and females may give birth to one or two calves per litter. The species feeds on submerged and emergent floating vegetation, such as seagrass, hydrilla (Hydrilla verticillata), and smooth cordgrass (Spartina alterniflora) (USFWS 2001).

The most recent population estimate (2015), put the Florida population at 6,350 manatees (82 FR 16668). Recent analysis indicates this population is stable or increasing throughout much of the state (USFWS 2014). Threats to the species include human-caused mortality (watercraft collisions), interactions with commercial fishing gear, pollution, exposure to cold/loss of warm-water refugia, and red tides (*Gymnodinium breve*) (USFWS 2007, 2014).

#### **B.1.2 REPTILES**

#### Green Sea Turtle (Chelonia mydasis)

The green sea turtle is listed as threatened for the South Atlantic DPS (NOAA 2023). The green sea turtle is the largest sea turtle found in the United States. (NOAA 2023). During the day, green turtles occupy shallow flats and seagrass meadows, and, in the evening, they return to their sleeping quarters of rock ledges, oyster bars, and coral reefs (FWC 2023). Adult green turtles have a mostly vegetarian diet of seagrasses and algae. During the breeding season, late spring and early summer, male sea turtles migrate to offshore waters to mate with females. Nesting seasons vary in the different geographical areas of their range; however, the Florida population nests between the months of June and September (FWC 2023). Florida hosts one of the largest groupings of green turtle nests in the western Atlantic Ocean. More than 37,000 green sea turtle nests were documented in Florida in 2015, a record number (FWC 2023). In 1998, NOAA Fisheries designated critical habitat for green turtles in coastal waters around Culebra Island, Puerto Rico (63 FR 46693, September 2, 1998).

The primary threat to this species is entanglement in fishing gear (FWC 2023). In addition, increased beach development can cause degradation of the habitat and limit the amount of nesting sites available (FWC 2023; NOAA 2023). Coastal development also increases artificial lighting, which can cause hatchlings to migrate towards the lights instead of the ocean (FWC

2023). Other threats to this species are climate change, direct harvest of turtles and eggs, disease, ocean pollution/ marine debris, and vessel strikes (NOAA 2023).

#### Hawksbill Sea Turtle (Eretmochelys imbricata)

The hawksbill sea turtle is the most abundant species of sea turtle that nests in the United States. They are listed as endangered. The NMFS designated the waters surrounding Culebra, Mona, and Monito Islands, Puerto Rico, as critical habitat for this species (63 FR 46693, September 2, 1998). The species inhabits the tropical and subtropical waters of all the world's major oceans. It is the only species of sea turtle that can survive on a diet consisting mainly of sponges, but also will eat marine algae, corals, mollusks, tunicates, crustaceans, sea urchins, small fish, and jellyfish. Like other sea turtle species, hawksbills can migrate long distances between foraging areas and nesting beaches.

Hawksbill turtles use a variety of habitats during different stages of their life cycle, but largely inhabit nearshore foraging grounds, especially healthy coral reef habitats. In the Eastern Pacific, large hawksbill populations have been found in mangrove estuaries. Upon leaving their nesting beaches, most hawksbill hatchlings enter pelagic (open sea) habitat, where they take shelter in floating algal mats and drift lines of flotsam and jetsam for approximately 1 to 5 years. Eventually, juveniles migrate to shallower coastal feeding grounds, including their preferred coral reef habitats, where they mature to adulthood and spend the remainder of their lives. The ledges and caves of coral reefs provide shelter for resting hawksbills during the day and at night. Hawksbills also are found around rock formations, high-energy shoals (sand bars in shallow water), and estuaries that provide good habitat for sponge growth.

The primary threats to the species are bycatch in fishing gear, the direct harvest of turtles and eggs, loss and degradation of nesting habitat, predation of eggs and hatchlings, vessel strikes, pollution and marine debris, and climate change.

#### Kemp's Ridley Sea Turtle (Lepidochelys kempii)

The Kemp's ridley sea turtle is listed as endangered throughout its range under the ESA. Critical habitat has not been designated. Kemp's ridley sea turtles are primarily found in the Gulf of Mexico, but juveniles also are found in the Atlantic Ocean as far north as Nova Scotia and sometimes even occur in the eastern North Atlantic. Kemp's ridleys primarily occupy nearshore coastal (neritic) habitats in the Gulf of Mexico that include muddy or sandy bottoms where their preferred prey of small animals/insects and plants is found. The primary threats to the species are bycatch in fishing gear, the direct harvest of turtles and eggs, loss and degradation of nesting habitat, predation of eggs and hatchlings, vessel strikes, pollution and marine debris, and climate change.

#### Leatherback Sea Turtle (Dermochelys coriacea)

The leatherback sea turtle has been listed as endangered since1970, originally under the Endangered Species Conservation Act of 1969 throughout its range under the Endangered Species Act and then under the ESA. The NMFS determined critical habitat for this species in waters adjacent to Sandy Point Beach, St. Croix, U.S. Virgin Islands in 1979 (44 FR 17710, March 23, 1979). An additional 41,914 square miles in the Pacific Ocean were added as critical habitat in 2012 (77 FR 4169, January 26, 2012). The leatherback sea turtle has the widest global distribution of any reptile, with nesting occurring mainly on tropical or subtropical beaches.

Once prevalent in every ocean except the Arctic and Antarctic, the leatherback population is rapidly declining in many parts of the world. In terms of feeding, the leatherback sea turtles use pointed tooth-like cusps and sharp-edged jaws that are perfectly adapted for a diet of soft-bodied open ocean prey, such as jellyfish and salps. They face threats on both nesting beaches and in the marine environment the greatest of which worldwide include incidental capture in fishing gear (bycatch), hunting of turtles, and collection of eggs for human consumption. The Pacific leatherback turtle populations are most at-risk of extinction. Pacific leatherbacks are one of nine ESA-listed species identified in NOAA's Species in the Spotlight initiative. Through this initiative, NOAA Fisheries has made it a priority to focus recovery efforts on stabilizing and recovering Pacific leatherback populations to attempt to prevent their extinction.

#### Loggerhead Sea Turtle (Caretta caretta)

The loggerhead sea turtle is the most abundant species of sea turtle that nests in the United States. There are nine DPSs of loggerhead sea turtles. The North Pacific Ocean DPS, Mediterranean Sea DPS, Northeast Atlantic DPS, North Indian Ocean DPS, and South Pacific Ocean DPS are listed as endangered. The Northwest Atlantic Ocean DPS, South Atlantic Ocean DPS, Southeast Indo-Pacific Ocean DPS, and Southwest Indian Ocean DPS are listed as threatened. Critical habitat has not been designated in any marine areas, but the USFWS has designated specific areas in the terrestrial environment of the U.S. Atlantic Ocean DPS of the loggerhead sea turtle (79 FR 39755, July 10, 2014). Juvenile and adult loggerheads live in U.S. coastal waters, but many adults that nest on U.S. beaches migrate from neighboring nations like the Bahamas, Cuba, and Mexico. Loggerhead turtles are found worldwide with nine DPS listed under the ESA. In the United States, the Northwest Atlantic Ocean DPS of loggerhead turtle nests primarily along the Atlantic coasts of Florida, South Carolina, Georgia, and North Carolina and along the Florida and Alabama coasts in the Gulf of Mexico. Total estimated nesting in the United States is more than 100,000 nests per year.

Loggerhead sea turtles are carnivores, only occasionally consuming plant material. Juveniles and adults in coastal waters eat mostly bottom-dwelling invertebrates, such as whelks and other mollusks and horseshoe and other crabs. Their powerful jaws are designed to crush their prey. The largest threats to loggerhead populations include bycatch in fishing gear, loss and degradation of nesting habitat, vessel strikes, direct harvest of turtles and eggs, ocean pollution/marine debris, and climate change.

#### B.1.3 FISH

#### Atlantic Sturgeon (Acipenser oxyrinchus oxyrinchus)

The Atlantic sturgeon is managed across five DPSs all of which were federally listed in 2012: the Carolina DPS, Chesapeake Bay DPS, New York Bight DPS, and South Atlantic DPS are listed as endangered and the Gulf of Maine DPS is listed as threatened (77 FR 5913, 77 FR 5880). The South Atlantic DPS was listed as endangered effective April 6, 2012 (77 FR 5913). Critical habitat was designated for all DPSs in 2017 (82 FR 39160). Critical habitat for the South Atlantic DPS includes rivers in South Carolina, Georgia, and Florida: the Edisto, Combahee-Salkehatchie, Savannah, Ogeechee, Altamaha, Ocmulgee, Oconee, Satilla, and St. Mary's rivers (82 FR 39160).

The species' historical range spans riverine and coastal waters along the Atlantic Coast of North America from Florida to Canada. As an anadromous fish, adults spend their lives foraging in the Atlantic Ocean, with records of occurrences as far north as Iceland. Adults return to their natal freshwater rivers to spawn in springtime every one to five years. Adult sturgeons are characterized as bottom feeders, using their barbels to scan the benthos primarily in search of invertebrates, such as mollusks, worms, crustaceans, and fish (NOAA Fisheries 2020).

Extensive overfishing contributed to initial species declines (NOAA Fisheries 2020). Despite a moratorium on all Atlantic sturgeon fisheries that began in 1998, some populations have continued to decline (77 FR 5913). Current threats include fisheries bycatch, habitat degradation (e.g., dredging and poor water quality), habitat impediment (e.g., dams), and vessel strikes (FWC 2020c; NOAA Fisheries 2020).

#### Giant Manta Ray (Mobula birostris)

In 2018 the giant manta ray was listed as threatened under the ESA. Critical habitat has not yet been designated. The giant manta ray is found worldwide in tropical, subtropical, and temperate bodies of water. The giant manta ray is a migratory species and seasonal visitor along productive coastlines with regular upwelling, in oceanic island groups, and near offshore pinnacles and seamounts. The timing of these visits varies by region and seems to correspond with the movement of zooplankton, current circulation and tidal patterns, seasonal upwelling, seawater temperature, and possibly mating behavior. Manta rays primarily feed on planktonic organisms, such as euphausiids, copepods, mysids, decapod larvae, and shrimp, but some studies have noted their consumption of small and moderately sized fish as well. The species has also been observed in estuarine waters and oceanic inlets and within bays and intercoastal waterways. As such, giant manta rays can be found in cool water, as low as 19 °C, although temperature preference appears to vary by region.

The primary threats to the species are overfishing, bycatch, and harvest for international trade.

## Gulf Sturgeon (*Acipenser oxyrinchus desotoi*, formerly known as *Acipenser oxyrhynchus desotoi*)

The Gulf sturgeon was listed as threatened effective October 30, 1991 (56 FR 49653). Critical habitat was designated for the species on March 19, 2003 (68 FR 13370).

The Gulf sturgeon is primarily confined to the eastern Gulf of Mexico and classified as a subspecies of Atlantic sturgeon (*Acipenser oxyrinchus oxyrinchus*) (NatureServe 2020). This anadromous fish spends much of its adult life foraging in the Gulf during colder months of the year, returning to natal freshwater river systems to spawn from February to April (FNAI 2001). The species may migrate as far as 225 kilometers (km) upriver to spawn. Spawning typically occurs over substrates consisting of hard clay, rubble, and gravel (NatureServe 2020). Adults return to the Gulf of Mexico in late fall and young-of-the-year remain in their natal rivers for as long as 12 months before moving into the lower estuarine habitats (68 FR 13370). Adult sturgeons are characterized as bottom feeders, using their barbels to scan the benthos primarily in search of invertebrates, such as brachiopods, insect larvae, mollusks, worms, and crustaceans (56 FR 49653).

Initially, population declines were the product of overfishing for caviar, smoked fish, and isinglass. Damming and disconnection of spawning grounds from the Gulf of Mexico is one of

the current major factors contributing to the species' decline. Other factors such as habitat modification caused by dredging, navigation maintenance activities, and water pollution also pose potential threats (68 FR 13370). In Florida, the Gulf sturgeon's distribution extends from all major Panhandle river systems east to the Suwannee River (FNAI 2001a). The species has been seen as far southeast as Florida Bay, although these observations coincide with especially cold years and are rare (68 FR 13370).

#### Nassau Grouper (Epinephelus striatus)

The Nassau grouper has been listed as threatened under the ESA since 2016. It is a reef fish that tends to be associated with hard structures like reefs (both natural and artificial), rocks, and ledges. These fish are late-maturing, long-lived, top-level predators found in southern coastal Florida, the Florida Keys, Bermuda, the Yucatan, and the Caribbean Sea. Currently, all harvest of Nassau grouper is prohibited in the United States.

While the Nassau grouper is considered a reef fish, it transitions as it grows through a series of shifts in both habitat and diet. As larvae, they are planktonic and, as juveniles, they are found in nearshore shallow waters in macroalgal and seagrass habitats. They shift deeper as they grow, to predominantly reef habitat (forereef and reef crest). The main influences on where they live are not known, although water clarity, habitat, and benthos (the community of organisms in the seabed) seem to be important. The primary threats to the Nassau grouper are overfishing, a lack of effective regulations, and enforcement of those regulations that do exist.

#### **Oceanic Whitetip Shark (***Carcharhinus longimanus*)

In 2018 the oceanic whitetip shark was listed as threatened under the ESA. It also is listed under Appendix II of the Convention on International Trade in Endangered Species of Wild Fauna and Flora, Annex II of the Protocol Concerning Specially Protected Areas and Wildlife of the Cartagena Convention, and Annex I of the Sharks Memorandum of Understanding on the Convention on Migratory Species. Critical habitat has not yet been designated.

The main threat to oceanic whitetip sharks is bycatch in commercial fisheries combined with demand for its fins. They are frequently caught in pelagic longline, purse seine, and gillnet fisheries worldwide and their fins are highly valued in the international trade for shark products. As a result, their populations have declined throughout its global range. The species is found throughout the world in tropical and subtropical waters. It is a pelagic species, generally found offshore in the open ocean, on the Outer Continental Shelf, or around oceanic islands in deep water areas. Although it can make deep dives and has been recorded up to 1,082 m (3,549 feet) deep, it typically lives in the upper part of the water column, from the surface to at least 200 m (656 feet deep). Oceanic whitetip sharks have a strong preference for the surface mixed layer in warm waters above 20 °C and are, therefore, considered a surface-dwelling shark.

The primary threats to the species are commercial fishing and bycatch and harvest for international trade.

#### Shortnose Sturgeon (Acipenser brevirostrum)

The shortnose sturgeon was listed as endangered effective March 11, 1967 (32 FR 4001). Critical habitat has not been designated (USFWS 2020).

The shortnose sturgeon's historical range spans riverine and estuarine waters along the Atlantic Coast of North America from the Indian River in Florida to the St. John River in Canada. There are three metapopulations distributed across this area: the northern, mid-Atlantic, and southern metapopulations. The current range is disjunct, with a 402-km gap between the northern and mid-Atlantic metapopulations and the southern metapopulation (NOAA 2020). The species is managed across 19 DPSs (NMFS 1998).

As primarily an amphidromous fish, shortnose sturgeon are born in freshwater rivers in which they live and spend most of their adult lives, making short trips to marine waters for foraging (NOAA 2020). Some populations in the southern range exhibit estuarine anadromous life histories, but adults remain in estuaries rather than foraging offshore and only engage in short-distance migrations (NMFS 1998). In the southern portion of their range, spawning occurs from January to April (NOAA 2020). Spawning habitat typically includes the uppermost reach of a river (NMFS 1998). Adult sturgeons are characterized as benthic feeders, primarily upon mollusks and crustaceans (NOAA 2020).

Extensive overfishing contributed to the species' initial declines, and pollution and habitat impediments resulting from industrial development have prevented its recovery (NMFS 1998). Ongoing major threats include habitat impediment (e.g., dams), habitat degradation (e.g., dredging and poor water quality), and fisheries bycatch (NOAA 2020).

#### Smalltooth Sawfish (Pristis pectinate)

The U.S. DPS of smalltooth sawfish was listed as endangered on April 1, 2003 (68 FR 15674). The Bahamian DPS was listed as endangered in 2014 (79 FR 73977). In 2009, two areas along the southwestern coast of Florida were designated as critical habitat for the U.S. DPS (74 FR 45353).

This species has a circumtropical distribution, from Brazil through the Caribbean and Central America, the Gulf of Mexico, and the Atlantic Coast of the United States (NMFS 2009). The NMFS regulates the U.S. DPS in four regions of the eastern United States, including Florida. Peninsular Florida has the largest number of capture records on the East Coast (Wiley and Brame 2018). The species occurs off the southwest coast of Florida from about Charlotte Harbor through the Everglades (NOAA Fisheries 2020). Currently, smalltooth sawfish can be found with regularity only in south Florida between the Caloosahatchee River and the Florida Keys (NMFS 2009). Their habitat includes shallow coastal waters of most warm seas. The species is found very close to shore in muddy and sandy bottoms, seldom descending to depths greater than 10 m. They often are found in sheltered bays, on shallow banks, and in estuaries or river mouths (NMFS 2009). Juvenile smalltooth sawfish generally live in estuaries during their first two years (habitats fringed with vegetation, especially red mangroves (*Rhizophora mangle*)), and move into more coastal habitats upon reaching 2 m in length. Female smalltooth sawfish may have 7 to 14 pups at a time, gestate for 12 months, and give birth every other year (NOAA Fisheries 2020). This species generally subsists on small schooling fish, crustaceans, and other bottom-dwelling organisms (NMFS 2009) and use their rostra (or "saw") to slash through schools of fish and to find shrimp and crabs on the seafloor (NOAA Fisheries 2020).

The species has declined as a result of habitat loss (development of the Florida waterfront) and bycatch (NMFS 2009; NOAA Fisheries 2020; Wiley and Brame 2018). Historically, smalltooth sawfish were often accidentally caught in fishing nets and often killed rather than released

unharmed (NOAA Fisheries 2020). This threat has been reduced with the 1995 enactment of the Florida Net Ban Amendment and improved education reform (NMFS 2009; NOAA Fisheries 2020). It is likely that the population is currently at a level less than five percent of its size at the time of European settlement (NMFS 2009).

#### **B.2** Invertebrates

#### B.2.1 CORAL

#### Boulder Star Coral Orbicella franksi)

Boulder star (is a coral species native to shallow waters in the Caribbean Sea, the Gulf of Mexico, the Bahamas, Bermuda, and Florida. Colonies of boulder star coral usually form massive clumps with uneven surfaces, and sometimes they form plates. They are usually orange-brown, greenish-brown, or grayish brown, but the extremities of the lumps are often pale or white. Boulder star coral is listed as threatened under the ESA. Critical habitat has not yet been designated, but there is a proposition to do so. Threats to the species include climate change (e.g., ocean warming and ocean acidification), diseases, habitat degradation, land-based sources of pollution, and unsustainable fishing.

#### Elkhorn Coral (Acropora palmata)

Elkhorn coral is found typically in clear, shallow water (1–15 feet) on coral reefs throughout the Bahamas, Florida, and the Caribbean. The northern extent of its range in the Atlantic Ocean is Broward County, Florida, where it is relatively rare. Elkhorn coral lives in high-energy zones, with a lot of wave action. Too much wave action (major storms) can cause this branching coral to break. Fragmentation via branch breakage, however, is one method of reproduction for the species. Four critical areas of habitat have been designated off the coast of Florida and off the islands of Puerto Rico and the U.S. Virgin Islands. The species has been listed as threatened under the ESA since 2006.

Elkhorn coral used to be a dominant coral on Caribbean reefs and was so abundant that an entire reef zone is named for it. Beginning in the 1980s, the elkhorn coral population declined 97 percent from white band disease, which kills the coral's tissues. Currently, there are locations such as the U.S. Virgin Islands where populations of elkhorn coral appear stable at low abundance, and other locations such as the Florida Keys where population numbers are decreasing. Successful reproduction is very rare, so it is hard for elkhorn coral populations to increase.

Elkhorn coral get food from photosynthetic algae that live inside the coral's cells. They also feed by capturing plankton with their polyps' tentacles. Coral bleaching is the loss of the algae that live in coral tissue. This loss can lead to coral death through starvation or increased vulnerability to diseases. Because of their tree-like growth form, elkhorn corals provide complex habitat for fish and other coral reef organisms. When elkhorn corals are abundant, they provide shoreline protection from large waves and storms. The greatest threat to elkhorn coral is ocean warming, which causes the corals to release the algae that live in their tissues and provide them food, usually causing death. Other threats to elkhorn coral are ocean acidification that makes it harder for them to build their skeleton, unsustainable fishing practices that deplete the herbivores that keep the reef clean, and land-based sources of pollution that impacts the clear, low-nutrient waters in which they thrive.

#### Lobed Star Coral (Orbicella annularis)

The lobed star coral is a species of coral that lives in the western Atlantic Ocean. It is the most thoroughly studied and most abundant species of reef-building coral in the Caribbean to date. It grows into varying colony shapes (heads, columns, and plates) in response to differing light conditions. Lobed star coral is listed as threatened under the ESA. Critical habitat has not yet been designated. Threats to the species include climate change (e.g., ocean warming and ocean acidification), diseases, habitat degradation, and land-based sources of pollution.

#### Mountainous Star Coral (Orbicella faveolate)

Mountainous star coral is native to the Caribbean Sea and the Gulf of Mexico. Its color is usually pale brown but may be deep brown with fluorescent green highlights. Mountainous star coral is listed as threatened under the ESA. Critical habitat has not yet been designated. Threats to the species include climate change (e.g., ocean warming and ocean acidification), diseases, habitat degradation, and land-based sources of pollution.

#### Pillar Coral (Dendrogyra cylindrus)

Pillar coral is listed as threatened under the ESA. Critical habitat has not yet been designated. Pillar coral is a hard coral found in the western Atlantic Ocean and the Caribbean Sea. It often resembles fingers or a cluster of cigars, growing up from the sea floor.

#### Rough Cactus Coral (Mycetophyllia ferox)

Rough cactus coral is found in the Caribbean, southern Gulf of Mexico, Florida, and the Bahamas. It is listed as threatened under the ESA. Critical habitat has not yet been designated. Threats to the species include climate change (e.g., ocean warming and ocean acidification), diseases, habitat degradation, and land-based sources of pollution.

#### Staghorn Coral (Acropora cervicornis)

Staghorn coral is found typically in clear, shallow water (15–60 feet) on coral reefs throughout the Bahamas, Florida, and the Caribbean. The northern extent of the range in the Atlantic Ocean is Palm Beach County, Florida, where it is relatively rare. Staghorn coral lives in many coral reef habitats, including spur and groove, bank reef, patch reef, and transitional reef habitats as well as on limestone ridges, terraces, and hardbottom habitats. Four critical areas of habitat have been designated off the coast of Florida and off the islands of Puerto Rico and the U.S. Virgin Islands.

Staghorn coral get food from photosynthetic algae that live inside the coral's cells. They also feed by capturing plankton with their polyps' tentacles. Coral bleaching is the loss of the algae that live in coral tissue. Because of their bush-like growth form, staghorn corals provide complex habitat for fish and other coral reef organisms. When staghorn corals are abundant, they provide shoreline protections from large waves and storms.

Staghorn coral used to be a dominant coral on Caribbean reefs and was so abundant that an entire reef zone is named for it. Beginning in the 1980s, the staghorn coral population declined 97 percent from white band disease, which kills the coral's tissues. Populations appear to consist mostly of isolated colonies or small groups of colonies compared to the vast thickets once prominent throughout its range, with thickets still a prominent feature at only a handful of known

locations. Successful reproduction is very rare, so it is hard for staghorn coral populations to increase.

The greatest threat to staghorn coral is ocean warming, which causes the corals to release the algae that live in their tissues and provide them food, usually causing death. Other threats to staghorn coral are ocean acidification that makes it harder for them to build their skeleton, unsustainable fishing practices that deplete the herbivores that keep the reef clean, and land-based sources of pollution that impact the clear, low- nutrient waters in which they thrive.

#### **B.2.2 MOLLUSKS**

#### Queen Conch (Strombus gigas)

The queen conch is a large gastropod mollusk. On September 7, 2022, NOAA announced a proposed rule to list the queen conch as a threatened species.

The species is slow growing and late to mature, reaching up to 12 inches in length and living up to 30 years. The queen conch occurs throughout the Caribbean Sea, the Gulf of Mexico, the Florida Keys, and around Bermuda. They are benthic-grazing herbivores that feed on diatoms, seagrass detritus, and various types of algae and epiphytes. Adult queen conch prefer sandy algal flats but are also found on gravel, coral rubble, smooth hard coral, and beach rock bottom, while juveniles are primarily associated with seagrass beds.

Queen conch is highly sought after for its meat and is one of the most valuable species in the Caribbean. In Florida, the species may not be harvested or sold.

The primary threats to the queen conch are harvesting and habitat degradation (Horn et al. 2022). Additional threats include climate change, oil spills, and pollution (Horn et al. 2022). Threats such as coastal development, recreational access, and dredging may occur infrequently, have a narrow geographic scope, or have uncertain or indirect effects on the species (Horn et al. 2022).

#### **Literature Cited for Species Accounts**

- FWC (Florida Fish and Wildlife Conservation Commission). 2019. *Florida's Wildlife Legacy Initiative: Florida's State Wildlife Action Plan.* Tallahassee, Florida, USA.
- FWC (Florida Fish and Wildlife Conservation Commission). 2020. Atlantic Sturgeon. FWC, Tallahassee, FL. Accessed February 10, 2020. https://myfwc.com/wildlifehabitats/profiles/freshwater/atlantic-sturgeon/.
- FWC (Florida Fish and Wildlife Conservation Commission). 2023. Green Sea Turtle. FWC, Tallahassee, FL. Accessed January 2023. https://myfwc.com/wildlifehabitats/profiles/reptiles/sea-turtles/green-sea-turtle/.
- Florida Natural Areas Inventory (FNAI). 2001. *Field Guide to the Rare Animals of Florida*. Florida Resources and Environmental Analysis Center, Florida State University, Tallahassee, FL. Accessed January 15, 2020. www.fnai.org.
- Horn, C., M. Karnauskas, J.C. Doerr, M.H. Miller, M. Neuman, R. Hill, and K.J. McCarthy.
  2022. *Endangered Species Act Status Review Report: Queen Conch* (Aliger gigas).
  National Oceanic and Atmospheric Administration, National Marine Fisheries Service,
  Southeast Fisheries Science Center. Miami, FL.
- NMFS (National Marine Fisheries Service). 1998. *Final Recovery Plan for the Shortnose Sturgeon* Acipenser brevirostrum. NMFS, Silver Spring, MD.
- NMFS (National Marine Fisheries Service). 2009. *Recovery Plan for Smalltooth Sawfish (*Pristis pectinata). NMFS, Silver Spring, MD.
- Wiley, T., and A. Brame. 2018. Smalltooth sawfish (Pristis pectinata) 5-Year Review: Summary and Evaluation of United States Distinct Population Segment of Smalltooth Sawfish. NMFS, Southeast Regional Office, St. Petersburg, FL.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2020. Species Directory Website. NOAA Fisheries. Office of Protected Resources, Silver Spring, Maryland, USA. https://www.fisheries.noaa.gov/species.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022a. *Atlantic Sturgeon*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/atlantic-sturgeon.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022b. *Blue Whale*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/blue-whale.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022c. *Boulder Star Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/boulder-star-coral.

- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022d. *Byrde's Whale*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed January 2023. https://www.fisheries.noaa.gov/species/brydes-whale#overview.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022e. *Elkhorn Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/elkhorn-coral.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022f.*Fin Whale*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/fin-whale.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022g. *Giant Manta Ray*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD Accessed December 2022. https://www.fisheries.noaa.gov/species/giant-manta-ray.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022h. *Green Sea Turtle*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/green-turtle.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022i. *Gulf Sturgeon*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/gulf-sturgeon.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022j. *Hawksbill Sea Turtle*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/hawksbill-turtle.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022k. *Humpback Whale*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/humpback-whale.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 20221. *Kemp's Ridley Sea Turtle*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/kemps-ridley-turtle.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022m. *Leatherback Sea Turtle*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/leatherback-turtle.

- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022n. Lobed Star Coral. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/lobed-star-coral.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022o. Loggerhead Sea Turtle. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/loggerhead-turtle.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022p. *Mountainous Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/mountainous-star-coral.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022q. *Nassau Grouper*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/nassau-grouper.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022r. *National ESA Critical Habitat Mapper*. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD https://www.fisheries.noaa.gov/resource/map/national-esa-critical-habitat-mapper.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022s. North Atlantic Right Whale. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/whales#by-species.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022t. Oceanic Whitetip Shark. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/oceanic-whitetip-shark.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022u. *Pillar Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/pillar-coral.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022v. Queen Conch. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed January 2023. https://www.fisheries.noaa.gov/species/queen-conch.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022w. *Rice's Whale*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/rices-whale.

- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022x. *Rough Cactus Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/rough-cactus-coral.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022y. Sei Whale. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/sei-whale.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022z. Shortnose Sturgeon. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/shortnose-sturgeon.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022aa. Smalltooth Sawfish. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/smalltooth-sawfish.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022ab. Sperm Whale. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/whales#by-species.
- NOAA (National Oceanic and Atmospheric Administration Fisheries). 2022ac. *Staghorn Coral*. Species Directory web page on the NOAA Fisheries website. NOAA Fisheries, Office of Protected Resources, Silver Spring, MD. Accessed December 2022. https://www.fisheries.noaa.gov/species/staghorn-coral.
- NatureServe. 2020. NatureServe Explorer an online encyclopedia of life [web application]. Version 7.1. NatureServe, Arlington, VA. Accessed January 15, 2020. http://explorer.natureserve.org.
- USFWS (U.S. Fish and Wildlife Service). 1999. *South Florida Multi-Species Recovery Plan*. U.S. Department of the Interior, Fish and Wildlife Service, Southeast Regional Office, Atlanta, GA.
- USFWS (U.S. Fish and Wildlife Service). 2001. *Florida Manatee Recovery Plan*, (Trichechus manatus latirostris), *Third Revision*. U.S. Department of the Interior, Fish and Wildlife Service, Southeast Regional Office, Atlanta, GA.
- USFWS (U.S. Fish and Wildlife Service). 2007. *West Indian Manatee* (Trichechus manatus) 5-*Year Review: Summary and Evaluation*. U.S. Department of the Interior, Fish and Wildlife Service, North Florida Ecological Services Office, Jacksonville, FL.
- USFWS (U.S. Fish and Wildlife Service). 2014. *West Indian manatee (Trichechus manatus) Florida Stock (Florida Subspecies, Trichechus manatus latirostris), Stock Assessment Report.* U.S. Department of the Interior, Fish and Wildlife Service, North Florida Ecological Services Office, Jacksonville, FL.
USFWS (U.S. Fish and Wildlife Service). 2020. *Environmental Conservation Online System* (*ECOS*). U.S. Department of the Interior, Fish and Wildlife Service, Southeast Regional Office, Atlanta, GA. Accessed January 15, 2020. https://ecos.fws.gov.

**Appendix C** Effects of the Action on ESA-Listed Species

# Table C.1.a. ESA-Listed Species—Stressors

Cuild		EL O	CH3	Wetlende		Biotic			Physical			Chemical		
Guild		FLO	GHY	wetlands	Competition	Predation	Invasive Species	Fill	Dredging	Habitat Fragmentation	Hydro	Nutrients	WQ	
Mammal	Fin Whale ( <i>Balaenoptera</i> <i>physalus</i> )	E	N	Ν		х		х	х	Х	х	х	Х	
Mammal	Sperm Whale ( <i>Physeter</i> <i>macrocephalus</i> )	Е	N	Ν		х		х	х	Х	х	Х	Х	
Mammal	Sei Whale ( <i>Balaenoptera</i> borealis)	E	N	Ν		х		х	х	Х	х	х	Х	
Mammal	Blue whale ( <i>Balaenoptera musculus</i> )	Е	N	N		х		х	х	Х	х	х	Х	
Mammal	Rice's Whale ( <i>Balaenoptera ricei</i> )	E	Ν	Ν		Х		Х	Х	×	х	х	х	
Mammal	North Atlantic Right Whale ( <i>Eubalaena</i> <i>glacialis</i> )	E	Y	N		х		х	х	х	х	х	х	
Mammal	Humpback Whale ( <i>Megaptera</i> <i>novaeanglia</i> e)	NW (state- listed)	Y	N		х		х	х	Х	х	х	х	
Mammal	Byrde's Whale (Gulf of Mexico, or GOMx) (Balaenoptera edeni [unnamed subspecies])	E	N	N		х		х	х	Х	x	х	x	
Reptile	Kemp's Ridley Sea Turtle ( <i>Lepidochelys kempii</i> )	E	N	Y		х		х	х	Х	х	х	Х	
Reptile	Leatherback Sea Turtle ( <i>Dermochelys</i> <i>coriacea</i> )	E	Y	Y		х		х	х	х	х	x	х	

Guild		EI 9	CH3	Watlanda		Biotic		Physical			Chemical		
Guild		FL3	СПР	wettanus	Competition	Predation	Invasive Species	Fill	Dredging	Habitat Fragmentation	Hydro	Nutrients	WQ
Reptile	Loggerhead Sea Turtle ( <i>Caretta</i> <i>caretta</i> )	Е	Y	Y		х		х	х	Х	х	Х	х
Reptile	Hawksbill Sea Turtle ( <i>Eretmochelys</i> <i>imbricata</i> )	Е	Y	Y		х		х	х	Х	х	х	х
Reptile	Green Sea Turtle ( <i>Chelonia mydas</i> )	т	Ν	Y		х		Х	Х	×	Х	Х	х
Fish	Shortnose Sturgeon (Acipenser brevirostrum)	Е	N	Y (streams)		х		х	х	Х	х		х
Fish	Gulf Sturgeon (Acipenser oxyrinchus [=oxyrhynchus] desotoi)	т	Y	Y (streams)		х		х	х	Х	Х		x
Fish	Atlantic Sturgeon ( <i>Acipenser oxyrinchus</i> <i>oxyrinchus</i> )	E*	Y	Y (streams)		х							х
Fish	Smalltooth Sawfish ( <i>Pristis pectinate</i> )	E*	Y	Y (streams)		Х		Х	Х	х	Х	Х	х
Fish	Nassau Grouper ( <i>Epinephelus striatus</i> )	Ρ	Ν	Y		х		Х	х	×	х	х	х
Shark/Ray	Oceanic Whitetip shark ( <i>Carcharhinus</i> <i>longimanus</i> )	т	N	Ν		х							
Shark/Ray	Giant Manta Ray ( <i>Mobula birostris</i> )	т	Ν	Y		х		Х	Х	x	Х	Х	х
Coral	Elkhorn Coral ( <i>Acropora palmata</i> )	т	Y	Y		х		Х	х	×	х	х	х
Coral	Staghorn Coral (Acropora cervicornis)	т	Y	Y		х		Х	Х	x	Х	Х	х
Coral	Boulder Star Coral (Orbicella franksi)	т	Ρ	Y		х		х	х	x	х	х	х

Cuild		EI S	CH3	Watlanda		Biotic		Physical			Chemical		
Guild	CN (SN)	FLO	υΠŕ	wellanus	Competition	Predation	Invasive Species	Fill	Dredging	Habitat Fragmentation	Hydro	Nutrients	WQ
Coral	Mountainous Star Coral ( <i>Orbicella faveolate</i> )	т	Ν	Y		х		х	х	Х	х	x	х
Coral	Lobed Star Coral ( <i>Orbicella annularis</i> )	т	Ν	Y		х		Х	х	×	х	х	х
Coral	Rough Cactus Coral ( <i>Mycetophyllia ferox</i> )	т	Ν	Y		х		Х	Х	x	х	Х	х
Coral	Pillar Coral ( <i>Dendrogyra</i> <i>cylindrus</i> )	т	N	Y		х		х	х	х	х	x	х
Conch	Queen Conch ( <i>Aliger</i> gigas)	Р	Ν	Y		х		Х	Х	х	х	Х	х

Table C.1.b. Effects of the Action on ESA-Lis	sted Species—Findings
---	-----------------------

Guild	CN (SN)	FLS	CH?	Finding	Comments (ESA determinations made for all species, including proposed, to encompass possibility for future listings)
			•	Ν	/ammals
Whale	Fin Whale ( <i>Balaenoptera physalus</i> )	E	Ν	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, and climate change.
Whale	Sperm Whale ( <i>Physeter</i> macrocephalus)	E	Ν	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, marine debris, climate change, and oil spills.
Whale	Sei Whale ( <i>Balaenoptera borealis</i> )	E	Ν	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, and climate change.
Whale	Blue Whale ( <i>Balaenoptera musculus</i> )	E	N	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, habitat degradation, pollution, and climate change.
Whale	Rice's Whale ( <i>Balaenoptera ricei</i> )	E	Ν	No Effect	Inhabits Gulf of Mexico away from the coast so not expected in action area. Major threats include energy exploration and development, oil spills and spill response, vessel strikes, ocean noise, marine debris, aquaculture, and entanglement in fishing gear.
Whale	North Atlantic Right Whale ( <i>Eubalaena glacialis</i> )	E	N	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, and climate change.
Whale	Byrde's Whale Balaenoptera edeni [unnamed subspecies]	E	Ν	No Effect	Inhabits open ocean and Gulf of Mexico away from the coast so not expected in action area. Major threats include energy exploration and development, oil spills and spill response, vessel strikes, ocean noise, and entanglement in fishing gear.
Whale	Humpback Whale (Megaptera novaeangliae)	NW (state- listed)	Y	No Effect	Inhabits open ocean away from the coast so not expected in action area. Major threats include vessel strikes, entanglement in fishing gear, ocean noise, and climate change.
		-			Reptiles
Turtle	Kemp's Ridley Sea Turtle ( <i>Lepidochelys kempii</i> )	E	Ν	No Effect	Inhabits the Gulf of Mexico and Atlantic Ocean in near coastal habitats but are not expected in the action area. Major threats to the species are bycatch in fishing gear, the direct harvest of turtles and eggs, loss and degradation of nesting habitat, predation of eggs and hatchlings, vessel strikes, pollution and marine debris, and climate change.
Turtle	Leatherback Sea Turtle (Dermochelys coriacea)	E	Y	No Effect	Inhabits the Gulf of Mexico and Atlantic Ocean in near coastal habitats but are not expected in the action area. Major threats to the species are bycatch in fishing gear and the direct harvest of turtles and eggs,
Turtle	Loggerhead Sea Turtle ( <i>Caretta caretta</i> )	E	Y	No Effect	Inhabits the Gulf of Mexico and Atlantic Ocean in near coastal habitats but are not expected in the action area. Major threats to the species are bycatch in fishing gear, loss and degradation of nesting habitat, vessel strikes, direct harvest of turtles and eggs, pollution, marine debris, and climate change,
Turtle	Hawksbill Sea Turtle ( <i>Eretmochelys imbricata</i> )	E	Ν	No Effect	Inhabits the Gulf of Mexico and Atlantic Ocean in near coastal habitats but are not expected in the action area. Major threats to the species are bycatch in fishing gear, loss and degradation of nesting habitat, vessel strikes, direct harvest of turtles and eggs, pollution, marine debris, and climate change,

Guild	CN (SN)	FLS	CH?	Finding	Comments (ESA determinations made for all species, including proposed, to encompass possibility for future listings)
Turtle	Green Sea Turtle ( <i>Chelonia mydas</i> )	Т	Ν	No Effect	Inhabits the Gulf of Mexico and Atlantic Ocean in near coastal habitats but are not expected in the action area. Major threats to the species are loss and degradation of nesting habitat, vessel strikes, direct harvest of turtles and eggs, pollution, marine debris, and climate change,
				Inv	vertebrates
Coral	Elkhorn Coral ( <i>Acropora palmata</i> )	Т	Y	No Effect	Inhabits clear shallow water on coral reefs throughout Florida (northern extent is Broward County). Not expected to be present in the action area. Major threats to the species include ocean warming (climate change), ocean acidification, unsustainable fishing practices that impact the reef habitat, and land-based pollution.
Coral	Staghorn Coral ( <i>Acropora</i> cervicornis)	Т	Y	No Effect	Inhabits clear shallow water on coral reefs throughout Florida (northern extent is Palm Beach County). Not expected to be present in the action area. Major threats to the species include ocean warming (climate change), ocean acidification, unsustainable fishing practices that impact the reef habitat, and land-based pollution.
Coral	Boulder Star Coral ( <i>Orbicella franksi</i> )	Т	Ρ	No Effect	Inhabits clear shallow water on coral reefs throughout Florida. Not expected to be present in the action area. Major threats to the species include climate change (ocean warming and ocean acidification), unsustainable fishing practices that impact the reef habitat, diseases, habitat degradation, and land-based pollution.
Coral	Mountainous Star Coral ( <i>Orbicella faveolate</i> )	Т	Ν	No Effect	Inhabits the Gulf of Mexico. Not expected to be present in the action area. Major threats to the species include climate change (ocean warming and ocean acidification), unsustainable fishing practices that impact the reef habitat, diseases, habitat degradation, and land-based pollution.
Coral	Lobed Star Coral ( <i>Orbicella annularis</i> )	Т	Ν	No Effect	Inhabits the western Atlantic Ocean and Caribbean Sea. Not expected to be present in the action area. Major threats to the species include climate change (ocean warming and ocean acidification), diseases, habitat degradation, and land-based pollution.
Coral	Rough Cactus Coral ( <i>Mycetophyllia ferox</i> )	Т	Ν	No Effect	Inhabits the Caribbean Sea, Gulf of Mexico, and coastal waters of Florida. Not expected to be present in the action area. Major threats to the species include climate change (ocean warming and ocean acidification), diseases, habitat degradation, and land-based pollution.
Coral	Pillar Coral ( <i>Dendrogyra cylindrus</i> )	Т	Ν	No Effect	Inhabits the western Atlantic Ocean and Caribbean Sea. Not expected to be present in the action area. Major threats to the species include climate change (ocean warming and ocean acidification), diseases, habitat degradation, and land-based pollution.
Conch	Queen Conch ( <i>Aliger gigas</i> )	Ρ	Ν	No Effect	Inhabits the Caribbean Sea, Gulf of Mexico, and coastal waters of Florida. Not expected to be present in the action area. Major threats to the species include overharvesting, climate change, oil spills, pollution, as well as habitat loss due to impacts from coastal development, recreational access, and dredging.
			-		Fishes
Sturgeon	Shortnose Sturgeon (Acipenser brevirostrum)	E	N	May Affect – Not Likely to Adversely Affect	Inhabits rivers and estuaries. Major threats include habitat impediment (e.g., dams), habitat degradation (e.g., dredging and poor water quality), and fisheries bycatch.

Guild	CN (SN)	FLS	CH?	Finding	Comments (ESA determinations made for all species, including proposed, to encompass possibility for future listings)
Sturgeon	Gulf Sturgeon ( <i>Acipenser</i> oxyrinchus [=oxyrhynchus] desotoi)	Т	Y	No effect <sup>1</sup>	Inhabits rivers and Gulf of Mexico waters. Major threats include damming and disconnection of spawning grounds, as well as habitat modification due to dredging, navigation maintenance activities, and water pollution. Impacts to aquatic habitat, hydrology, and water quality.
Sturgeon	Atlantic Sturgeon ( <i>Acipenser</i> oxyrinchus oxyrinchus)	E*	Y	May Affect – Not Likely to Adversely Affect	Inhabits riverine and Atlantic coast waters. Major threats include fisheries bycatch, habitat degradation (e.g., dredging and poor water quality), habitat impediment (e.g., dams), and vessel strikes. Impacts to aquatic habitat, hydrology, and water quality.
Sawfish	Smalltooth Sawfish ( <i>Pristis pectinate</i> )	E*	Y	May Affect – Not Likely to Adversely Affect	Inhabits estuaries and coastal waters. Major threats include habitat loss and bycatch. Impacts to aquatic habitat, hydrology, and water quality.
Fish	Nassau Grouper ( <i>Epinephelus</i> <i>striatus</i> )	Ρ	Ν	No effect	Inhabits coastal areas of Florida and Florida Keys. Major threats include overfishing and a lack of effective regulations and enforcement of those regulations. Impacts to aquatic habitat, hydrology, and water quality.
Shark/Ray	Oceanic Whitetip Shark (Carcharhinus longimanus)	т	Ν	No effect	Inhabits offshore open ocean, on the outer continental shelf, or around oceanic islands in deep water areas. Major threats include commercial fishing and bycatch along with harvest for international trade.
Shark/Ray	Giant Manta Ray ( <i>Mobula birostris</i> )	Т	Ν	No effect	Inhabits offshore oceanic waters and in productive coastal areas, has been observed in estuarine waters, oceanic inlets, and within bays and intercoastal waterways but not expected in the action area. Major threats include overfishing and bycatch along with harvest for international trade.

FLS – Federal Listing Status under the Endangered Species Act (ESA) CH? – Has critical habitat been designated?

Table Text Key:

FLS – E (Endangered), T (Threatened), TS (Threatened due to Similarity of Appearance), C (Candidate), P (Proposed for listing), UR (Under Review), URS (Under Review, Substantial Finding), NW (Not Warranted Finding), \* (Listing status of DPS that occurs in Florida. Other DPSs may have different listing statuses in other regions).

CH – Y (Yes), N (No)

Finding – NLAA (May Affect Not Likely to Adversely Affect), LAA (May Affect Likely to Adversely Affect)

### Header Key:

CN (SN) - Common Name (Scientific Name)

<sup>1</sup> Gulf sturgeon has shared jurisdiction with USFWS covering them in freshwater, so they are covered in assumed waters under the USFWS BiOp

# **Appendix D** Figures 1 and 2



N:\US\Atlanta\Projects\565\11206416\GIS\Ma Print date: 28 May 2020 - 15:12



**Appendix E** Dredge and Fill Activities as Defined in the Corps Database

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx)
\ AGRICULTURE \ CONVERSION
\AGRICULTURE \ NON-EXEMPT
\ AQUACULTURE \ FINFISH
\ AQUACULTURE \ PLANTS
\ AQUACULTURE \ SHELLFISH
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ COMMERCIAL
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RECREATIONAL, \ DEVELOPMENT \ COMMERCIAL
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ COMMERCIAL
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ ENERGY GENERATION \ NUCLEAR
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ STRUCTURE \ INTAKE/OUTFALL
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ STRUCTURE \ UTILITY LINE OR STRUCTURE
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE)
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ TRANSPORTATION \ ROADS \ CULVERT
\ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ TRANSPORTATION \ UTILITY \ AERIAL
\ DEVELOPMENT \ COMMERCIAL
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ INDUSTRIAL
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ INDUSTRIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RECREATIONAL
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \ DEVELOPMENT \ COMMERCIAL, \ STRUCTURE \ BULKHEAD \ DEVELOPMENT \ COMMERCIAL. \ STRUCTURE \ INTAKE/OUTFALL \ DEVELOPMENT \ COMMERCIAL, \ TRANSPORTATION \ AIRPORT \ FACILITY \ DEVELOPMENT \ COMMERCIAL, \ TRANSPORTATION \ RAIL \ FACILITY \ DEVELOPMENT \ COMMERCIAL, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ STRUCTURE \ INTAKE/OUTFALL \ DEVELOPMENT \ COMMERCIAL, \ TRANSPORTATION \ ROADS \ CULVERT \ DEVELOPMENT \ INDUSTRIAL \ DEVELOPMENT \ INDUSTRIAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ COMMERCIAL \DEVELOPMENT \ INDUSTRIAL, \ DEVELOPMENT \ COMMERCIAL \ DEVELOPMENT \ RECREATIONAL \ DEVELOPMENT \ RECREATIONAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE \ DEVELOPMENT \ RECREATIONAL, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \DEVELOPMENT \ RECREATIONAL. \ MITIGATION \ RESTORATION \ WETLAND. \ MITIGATION \ ENHANCEMENT \ DEVELOPMENT \ RECREATIONAL, \ STRUCTURE \ BOAT RAMP \ DEVELOPMENT \ RECREATIONAL. \ STRUCTURE \ DOCK \ FIXED. \ STRUCTURE \ BREAKWATER. \ DEVELOPMENT \ RECREATIONAL, \ TRANSPORTATION \ ROADS \ CULVERT, \ STRUCTURE \ BULKHEAD \ DEVELOPMENT \ RECREATIONAL. \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ COMMERCIAL \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ COMMERCIAL \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ MITIGATION \ ENHANCEMENT, \ MITIGATION \ CREATION \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ STRUCTURE \ MARINA, \ DREDGING \ NAVIGATION \ PRIVATE

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ STRUCTURE \ MARINA, \ STRUCTURE \ BOAT RAMP \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ DEVELOPMENT \ COMMERCIAL \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY, \ TRANSPORTATION \ UTILITY \ BURIED \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ MULTI-FAMILY \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ COMMERCIAL \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ COMMERCIAL, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ RECREATIONAL \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY. \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY. \ DEVELOPMENT \ COMMERCIAL \DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \DREDGING \ CHANNELIZATION, \ STRUCTURE \ BOAT LIFT \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY. \ OTHER \ BANK STABILIZATION \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ OTHER \ INDIAN TRIBE OR STATE 404 PROGRAM \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ STRUCTURE \ BULKHEAD \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ STRUCTURE \ DOCK \ FIXED \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ STRUCTURE \ MARINA \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ STRUCTURE \ MISCELLANEOUS \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ ROADS \ CULVERT \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ TRANSPORTATION \ ROADS \ CULVERT \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ DREDGING \ BOAT SI IP \ DREDGING \ CHANNELIZATION \DREDGING \CHANNELIZATION. \STRUCTURE \WEIR \ DREDGING \ DISPOSAL \ DREDGING \ DISPOSAL, \ DREDGING \ MAINTENANCE, \ STRUCTURE \ INTAKE/OUTFALL

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx)
\ DREDGING \ GENERAL
\ DREDGING \ GENERAL, \ OTHER \ BANK STABILIZATION, \ MITIGATION \ ENHANCEMENT
\ DREDGING \ GENERAL, \ STRUCTURE \ BOAT RAMP, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS
\ DREDGING \ GENERAL, \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW), \ OTHER \ BANK STABILIZATION
\ DREDGING \ MAINTENANCE
\ DREDGING \ MAINTENANCE, \ DREDGING \ BOAT SLIP, \ STRUCTURE \ BOAT HOUSE
\ DREDGING \ MAINTENANCE, \ DREDGING \ CHANNELIZATION
\ DREDGING \ MAINTENANCE, \ DREDGING \ GENERAL
\ DREDGING \ MAINTENANCE, \ OTHER \ BANK STABILIZATION
\ DREDGING \ MAINTENANCE, \ OTHER \ INDIAN TRIBE OR STATE 404 PROGRAM
\ DREDGING \ MAINTENANCE, \ STRUCTURE \ BULKHEAD
\ DREDGING \ MAINTENANCE, \ STRUCTURE \ GABION
\ DREDGING \ MAINTENANCE, \ STRUCTURE \ WATER CONTROL
\ DREDGING \ MAINTENANCE, \ STRUCTURE \ WATER CONTROL, \ STRUCTURE \ WEIR
\ DREDGING \ MAINTENANCE, \ TRANSPORTATION \ PIPELINE \ MAINTENANCE
\ DREDGING \ NAVIGATION \ PRIVATE, \ TRANSPORTATION \ ROADS \ MAINTENANCE
\ ENERGY GENERATION \ NATURAL GAS
\ ENERGY GENERATION \ NATURAL GAS, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ ENERGY GENERATION \ NATURAL GAS, \ TRANSPORTATION \ UTILITY \ STRUCTURE
\ ENERGY GENERATION \ SOLAR
\ MINING AND DRILLING \ DRILLING \ ACCESS
\ MINING AND DRILLING \ DRILLING \ FACILITIES
\ MINING AND DRILLING \ MINING \ FACILITES
\ MINING AND DRILLING \ MINING \ GRAVEL, \ MINING AND DRILLING \ MINING \ SAND
\ MINING AND DRILLING \ MINING \ OTHER MINERAL
\ MINING AND DRILLING \ MINING \ PHOSPHATE
\ MINING AND DRILLING \ MINING \ ROCK
\ MINING AND DRILLING \ MINING \ SAND
\ MINING AND DRILLING \ MINING \ SAND, \ MINING AND DRILLING \ MINING \ ROCK

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx)
\ MITIGATION \ CREATION
\ MITIGATION \ CREATION, \ DREDGING \ MAINTENANCE
\ MITIGATION \ ENHANCEMENT
\ MITIGATION \ ENHANCEMENT, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ MITIGATION \ FISH/WILDLIFE \ CREATION
\ MITIGATION \ FISH/WILDLIFE \ CREATION, \ OTHER \ BANK STABILIZATION
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ DEVELOPMENT \ RECREATIONAL
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ MITIGATION \ FISH/WILDLIFE \ CREATION, \ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ MITIGATION \ RESTORATION \ WETLAND, \ MITIGATION \ FISH/WILDLIFE \ CREATION
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ OTHER \ DAMS \ LOW WATER, \ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT, \ TRANSPORTATION \ ROADS \ CULVERT
\ MITIGATION \ FISH/WILDLIFE \ PLANTING
\ MITIGATION \ FISH/WILDLIFE \ RESTORATION
\ MITIGATION \ MITIGATION BANK
\ MITIGATION \ MITIGATION BANK, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY
\ MITIGATION \ MITIGATION BANK, \ MITIGATION \ PRESERVATION, \ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ RESTORATION \ STREAM
\ MITIGATION \ RESTORATION \ STREAM, \ MITIGATION \ CREATION, \ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ RESTORATION \ STREAM, \ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ RESTORATION \ STREAM, \ MITIGATION \ RESTORATION \ WETLAND, \ OTHER \ DAMS \ REMOVAL
\ MITIGATION \ RESTORATION \ WETLAND
\ MITIGATION \ RESTORATION \ WETLAND, \ AGRICULTURE \ CONVERSION
\ MITIGATION \ RESTORATION \ WETLAND, \ DEVELOPMENT \ RECREATIONAL
\ MITIGATION \ RESTORATION \ WETLAND, \ MITIGATION \ CREATION, \ MITIGATION \ RESTORATION \ STREAM
\ MITIGATION \ RESTORATION \ WETLAND, \ MITIGATION \ RESTORATION \ STREAM
\ MITIGATION \ RESTORATION \ WETLAND, \ TRANSPORTATION \ ROADS \ CULVERT
\ MITIGATION \ WETLAND RECLAMATION

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \MITIGATION \WETLAND RECLAMATION, \DREDGING \MAINTENANCE **\OTHER \BANK STABILIZATION** \ OTHER \ BANK STABILIZATION, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \OTHER \ BANK STABILIZATION, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \OTHER \ BANK STABILIZATION, \ DREDGING \ GENERAL \OTHER \ BANK STABILIZATION, \ DREDGING \ MAINTENANCE \OTHER \ BANK STABILIZATION, \ DREDGING \ NAVIGATION \ PRIVATE \OTHER \ BANK STABILIZATION, \ MITIGATION \ ENHANCEMENT \OTHER \ BANK STABILIZATION, \ STRUCTURE \ BULKHEAD \OTHER \ BANK STABILIZATION, \ STRUCTURE \ GABION \OTHER \ BANK STABILIZATION, \ STRUCTURE \ INTAKE/OUTFALL \ OTHER \ BANK STABILIZATION, \ STRUCTURE \ INTAKE/OUTFALL, \ DREDGING \ GENERAL \OTHER \ BANK STABILIZATION, \ STRUCTURE \ MAINTENANCE \OTHER \ BANK STABILIZATION, \ STRUCTURE \ MARINA, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \OTHER \ BANK STABILIZATION, \ STRUCTURE \ WATER CONTROL \OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ BRIDGE \ MAINTENANCE \ OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ ROADS \ MAINTENANCE \OTHER \ CLEANUP HAZARDOUS OR TOXIC WASTES \OTHER \ DAMS \ COFFER \OTHER \ DAMS \ COFFER, \ OTHER \ BANK STABILIZATION \OTHER \ DAMS \ GENERAL \OTHER \ DAMS \ LOW WATER, \ OTHER \ DAMS \ REMOVAL \OTHER \ DAMS \ MAINTENANCE \OTHER \DAMS \ MAINTENANCE, \STRUCTURE \GABION, \STRUCTURE \WATER CONTROL \OTHER \ DAMS \ REMOVAL \OTHER \ DAMS \ REMOVAL, \ MITIGATION \ WETLAND RECLAMATION \OTHER \ DAMS \ RESERVOIR

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx)
\ OTHER \ DAMS \ WEIR
\ OTHER \ DAMS \ WEIR, \ OTHER \ DAMS \ MAINTENANCE
\ OTHER \ INDIAN TRIBE OR STATE 404 PROGRAM
\ OTHER \ RESTRICTED AREAS
\ OTHER \ SURVEY ACTIVITIES
\ STRUCTURE \ BOAT LIFT
\ STRUCTURE \ BOAT LIFT, \ STRUCTURE \ DOCK \ FIXED
\ STRUCTURE \ BOAT LIFT, \ STRUCTURE \ MISCELLANEOUS, \ STRUCTURE \ DOCK \ FIXED
\ STRUCTURE \ BOAT RAMP
\ STRUCTURE \ BOAT RAMP, \ DREDGING \ GENERAL, \ DEVELOPMENT \ COMMERCIAL
\ STRUCTURE \ BOAT RAMP, \ STRUCTURE \ DOCK \ FIXED, \ DEVELOPMENT \ RECREATIONAL
\ STRUCTURE \ BOAT RAMP, \ STRUCTURE \ MARINA
\ STRUCTURE \ BREAKWATER
\ STRUCTURE \ BREAKWATER, \ MITIGATION \ FISH/WILDLIFE \ CREATION
\ STRUCTURE \ BRIDGE/RELATED WORK
\ STRUCTURE \ BRIDGE/RELATED WORK, \ STRUCTURE \ INTAKE/OUTFALL
\ STRUCTURE \ BRIDGE/RELATED WORK, \ STRUCTURE \ MAINTENANCE
\ STRUCTURE \ BRIDGE/RELATED WORK, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT
\ STRUCTURE \ BRIDGE/RELATED WORK, \ TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW)
\ STRUCTURE \ BRIDGE/RELATED WORK, \ TRANSPORTATION \ ROADS \ CULVERT
\ STRUCTURE \ BRIDGE/RELATED WORK, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS
\ STRUCTURE \ BULKHEAD
\ STRUCTURE \ BULKHEAD, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY
\ STRUCTURE \ BULKHEAD, \ DREDGING \ BOAT SLIP
\ STRUCTURE \ BULKHEAD, \ DREDGING \ BOAT SLIP, \ STRUCTURE \ DOCK \ FIXED
\ STRUCTURE \ BULKHEAD, \ DREDGING \ MAINTENANCE, \ STRUCTURE \ DOCK \ FIXED
\ STRUCTURE \ BULKHEAD, \ OTHER \ BANK STABILIZATION
\ STRUCTURE \ BULKHEAD, \ STRUCTURE \ DOCK \ FIXED
\ STRUCTURE \ BULKHEAD, \ STRUCTURE \ DOCK \ FIXED, \ STRUCTURE \ BOAT LIFT

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \ STRUCTURE \ BULKHEAD, \ STRUCTURE \ INTAKE/OUTFALL \ STRUCTURE \ BULKHEAD, \ STRUCTURE \ MAINTENANCE \STRUCTURE \BULKHEAD, \STRUCTURE \MARINA **\STRUCTURE \BULKHEAD, \STRUCTURE \UTILITY LINE OR STRUCTURE** \STRUCTURE \BULKHEAD, \STRUCTURE \WATER CONTROL \ STRUCTURE \ DOCK \ FIXED \STRUCTURE \DOCK \FIXED, \DEVELOPMENT \COMMERCIAL \STRUCTURE \DOCK \FIXED, \DEVELOPMENT \RESIDENTIAL \SINGLE FAMILY \ STRUCTURE \ DOCK \ FIXED, \ OTHER \ BANK STABILIZATION, \ STRUCTURE \ BREAKWATER \ STRUCTURE \ DOCK \ FIXED, \ STRUCTURE \ BOAT LIFT \ STRUCTURE \ DOCK \ FIXED, \ STRUCTURE \ BULKHEAD \ STRUCTURE \ DOCK \ FIXED, \ TRANSPORTATION \ AIRPORT \ RUNWAY, \ STRUCTURE \ BOAT RAMP \ STRUCTURE \ DOCK \ FLOATING **\STRUCTURE \ELEV REC DECK \STRUCTURE \GABION** \STRUCTURE \GABION, \STRUCTURE \MAINTENANCE, \STRUCTURE \UTILITY LINE OR STRUCTURE **\STRUCTURE \GABION, \TRANSPORTATION \BRIDGE \REPLACEMENT \STRUCTURE \INTAKE/OUTFALL** \STRUCTURE \INTAKE/OUTFALL, \DEVELOPMENT \ASSOCIATED INFRASTRUCTURE \ STRUCTURE \ INTAKE/OUTFALL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ STRUCTURE \ MAINTENANCE \STRUCTURE \INTAKE/OUTFALL, \DEVELOPMENT \COMMERCIAL \ STRUCTURE \ INTAKE/OUTFALL, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \STRUCTURE \INTAKE/OUTFALL, \DREDGING \MAINTENANCE \ STRUCTURE \ INTAKE/OUTFALL, \ ENERGY GENERATION \ NATURAL GAS **\STRUCTURE \INTAKE/OUTFALL, \OTHER \BANK STABILIZATION** \STRUCTURE \INTAKE/OUTFALL, \OTHER \BANK STABILIZATION, \STRUCTURE \MAINTENANCE \STRUCTURE \INTAKE/OUTFALL, \OTHER \DAMS \WEIR \STRUCTURE \INTAKE/OUTFALL, \STRUCTURE \MAINTENANCE \ STRUCTURE \ INTAKE/OUTFALL, \ STRUCTURE \ UTILITY LINE OR STRUCTURE

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \STRUCTURE \INTAKE/OUTFALL, \STRUCTURE \WATER CONTROL **\STRUCTURE \INTAKE/OUTFALL, \TRANSPORTATION \ROADS \IMPROVEMENTS** \ STRUCTURE \ INTAKE/OUTFALL, \ TRANSPORTATION \ UTILITY \ MAINTENANCE **\STRUCTURE \MAINTENANCE \STRUCTURE \MAINTENANCE. \OTHER \BANK STABILIZATION** \ STRUCTURE \ MAINTENANCE, \ OTHER \ BANK STABILIZATION, \ STRUCTURE \ WATER CONTROL \STRUCTURE \MAINTENANCE, \STRUCTURE \INTAKE/OUTFALL \STRUCTURE \MAINTENANCE, \STRUCTURE \WATER CONTROL \STRUCTURE \ MAINTENANCE, \ STRUCTURE \ WATER CONTROL, \ STRUCTURE \ WEIR \STRUCTURE \ MAINTENANCE, \ STRUCTURE \ WEIR \STRUCTURE \MAINTENANCE, \TRANSPORTATION \ROADS \CULVERT **\STRUCTURE \MARINA** \ STRUCTURE \ MARINA, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \STRUCTURE \MARINA, \DREDGING \NAVIGATION \PRIVATE \ STRUCTURE \ MARINA, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS **\STRUCTURE \MISCELLANEOUS** \STRUCTURE \MISCELLANEOUS, \DEVELOPMENT \RECREATIONAL \ STRUCTURE \ MISCELLANEOUS, \ OTHER \ BANK STABILIZATION, \ STRUCTURE \ BULKHEAD \STRUCTURE \MISCELLANEOUS. \STRUCTURE \WATER CONTROL. \OTHER \DAMS \WEIR **\STRUCTURE \PIER \NON-RESIDENTIAL** \ STRUCTURE \ PILE/DOLPHIN, \ DREDGING \ BOAT SLIP \STRUCTURE \ RAMP \STRUCTURE \ RAMP, \ STRUCTURE \ RECREATIONAL **\STRUCTURE \MAINTENANCE, \TRANSPORTATION \ROADS \CULVERT \STRUCTURE \MARINA \STRUCTURE \RECREATIONAL** \STRUCTURE \ RECREATIONAL, \ DEVELOPMENT \ RECREATIONAL \STRUCTURE \ RECREATIONAL, \ STRUCTURE \ PIER \ NON-RESIDENTIAL, \ OTHER \ BANK STABILIZATION \STRUCTURE \ RECREATIONAL, \ STRUCTURE \ WATER CONTROL

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx)
\ STRUCTURE \ REMOVAL
\ STRUCTURE \ REMOVAL, \ TRANSPORTATION \ UTILITY \ MAINTENANCE
\ STRUCTURE \ SCIENTIFIC DEVICE
\ STRUCTURE \ UNSPECIFIED
\ STRUCTURE \ UNSPECIFIED, \ DREDGING \ MAINTENANCE
\ STRUCTURE \ UNSPECIFIED, \ STRUCTURE \ WATER CONTROL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ DEVELOPMENT \ INDUSTRIAL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ DEVELOPMENT \ RECREATIONAL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ ENERGY GENERATION \ OIL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ MITIGATION \ ENHANCEMENT
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ OTHER \ BANK STABILIZATION
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ STRUCTURE \ INTAKE/OUTFALL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ STRUCTURE \ MAINTENANCE
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ STRUCTURE \ WATER CONTROL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ TRANSPORTATION \ UTILITY \ ACCESS ROAD
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ TRANSPORTATION \ UTILITY \ AERIAL
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ TRANSPORTATION \ UTILITY \ BURIED
\ STRUCTURE \ UTILITY LINE OR STRUCTURE, \ TRANSPORTATION \ UTILITY \ MAINTENANCE
\ STRUCTURE \ WATER CONTROL
\ STRUCTURE \ WATER CONTROL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE
\ STRUCTURE \ WATER CONTROL, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY
\ STRUCTURE \ WATER CONTROL, \ DEVELOPMENT \ RECREATIONAL
\ STRUCTURE \ WATER CONTROL, \ DREDGING \ MAINTENANCE
\ STRUCTURE \ WATER CONTROL, \ MITIGATION \ CREATION
\ STRUCTURE \ WATER CONTROL, \ STRUCTURE \ INTAKE/OUTFALL
\ STRUCTURE \ WATER CONTROL, \ STRUCTURE \ MAINTENANCE
\ STRUCTURE \ WATER CONTROL, \ STRUCTURE \ WEIR

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \ STRUCTURE \ WATER CONTROL, \ TRANSPORTATION \ ROADS \ CULVERT **\STRUCTURE \WEIR \STRUCTURE \WEIR, \STRUCTURE \BULKHEAD** \TRANSPORTATION \ AIRPORT \ FACILITY \ TRANSPORTATION \ AIRPORT \ FACILITY. \ TRANSPORTATION \ AIRPORT \ MAINTENANCE \ TRANSPORTATION \ AIRPORT \ FACILITY, \ TRANSPORTATION \ ROADS \ CULVERT **\TRANSPORTATION \AIRPORT \MAINTENANCE** \TRANSPORTATION \ AIRPORT \ MAINTENANCE, \ STRUCTURE \ WATER CONTROL \ TRANSPORTATION \ AIRPORT \ RUNWAY \TRANSPORTATION \ AIRPORT \ RUNWAY, \ TRANSPORTATION \ AIRPORT \ FACILITY \TRANSPORTATION \ AIRPORT \ RUNWAY, \ TRANSPORTATION \ AIRPORT \ MAINTENANCE \TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW) \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW), \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW), \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW), \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW), \ TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ ROADS \ CULVERT **\TRANSPORTATION \ BRIDGE \ MAINTENANCE** \TRANSPORTATION \ BRIDGE \ MAINTENANCE, \ STRUCTURE \ BRIDGE/RELATED WORK \TRANSPORTATION \ BRIDGE \ MAINTENANCE, \ TRANSPORTATION \ BRIDGE \ PROTECTION **\TRANSPORTATION \ BRIDGE \ PROTECTION** \TRANSPORTATION \ BRIDGE \ REMOVAL, \ MITIGATION \ FISH/WILDLIFE \ ENHANCEMENT \ TRANSPORTATION \ BRIDGE \ REMOVAL, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \ TRANSPORTATION \ BRIDGE \ REMOVAL, \ TRANSPORTATION \ RAIL \ FACILITY \ TRANSPORTATION \ BRIDGE \ REMOVAL, \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ BRIDGE \ REPLACEMENT \ TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ OTHER \ BANK STABILIZATION \TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ STRUCTURE \ BRIDGE/RELATED WORK \ TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ STRUCTURE \ MAINTENANCE \TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ STRUCTURE \ UNSPECIFIED

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS **\TRANSPORTATION \ PIPELINE \ ACCESS ROAD** \TRANSPORTATION \ PIPELINE \ ACCESS ROAD, \ TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ PIPELINE \ STRUCTURE **\TRANSPORTATION \ PIPELINE \ AERIAL \TRANSPORTATION \ PIPELINE \ BURIED** \TRANSPORTATION \ PIPELINE \ BURIED, \ OTHER \ DAMS \ COFFER \TRANSPORTATION \ PIPELINE \ BURIED, \ STRUCTURE \ UTILITY LINE OR STRUCTURE \ TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ PIPELINE \ AERIAL \TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ UTILITY \ STRUCTURE **\TRANSPORTATION \ PIPELINE \ MAINTENANCE** \TRANSPORTATION \ PIPELINE \ MAINTENANCE, \ TRANSPORTATION \ PIPELINE \ BURIED **\TRANSPORTATION \ PIPELINE \ STRUCTURE** \TRANSPORTATION \ PIPELINE \ STRUCTURE, \ DREDGING \ GENERAL \TRANSPORTATION \ PIPELINE \ STRUCTURE, \ TRANSPORTATION \ PIPELINE \ BURIED, \ TRANSPORTATION \ PIPELINE \ ACCESS ROAD **\TRANSPORTATION \ PIPELINE \ SUBMERGED** \TRANSPORTATION \ RAIL \ BRIDGE \ TRANSPORTATION \ RAIL \ FACILITY \ TRANSPORTATION \ RAIL \ FACILITY, \ STRUCTURE \ WATER CONTROL \TRANSPORTATION \ RAIL \ MAINTENANCE \TRANSPORTATION \ RAIL \ MAINTENANCE, \ TRANSPORTATION \ RAIL \ BRIDGE \ TRANSPORTATION \ RAIL \ TRACK \TRANSPORTATION \ RAIL \ TRACK, \ TRANSPORTATION \ RAIL \ BRIDGE, \ TRANSPORTATION \ UTILITY \ SUBMERGED \TRANSPORTATION \ RAIL \ TRACK, \ TRANSPORTATION \ RAIL \ MAINTENANCE, \ DEVELOPMENT \ INDUSTRIAL \TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ DEVELOPMENT \ RECREATIONAL \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ MINING AND DRILLING \ MINING \ PHOSPHATE \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW) \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ BRIDGE \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ **IMPROVEMENTS** \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ ROADS \ CULVERT \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ **IMPROVEMENTS** \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ ROADS \ MAINTENANCE \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE), \ TRANSPORTATION \ UTILITY \ ACCESS ROAD \TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ ROADS \ CULVERT, \ DEVELOPMENT \ RESIDENTIAL \ SINGLE FAMILY \ TRANSPORTATION \ ROADS \ CULVERT, \ DREDGING \ MAINTENANCE \TRANSPORTATION \ ROADS \ CULVERT. \ MITIGATION \ ENHANCEMENT \TRANSPORTATION \ ROADS \ CULVERT, \ OTHER \ BANK STABILIZATION \TRANSPORTATION \ ROADS \ CULVERT, \ STRUCTURE \ MAINTENANCE \TRANSPORTATION \ ROADS \ CULVERT, \ STRUCTURE \ WATER CONTROL \TRANSPORTATION \ ROADS \ CULVERT, \ STRUCTURE \ WEIR \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW) \TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT, \ TRANSPORTATION \ ROADS \ MAINTENANCE \TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ MAINTENANCE \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS **\TRANSPORTATION \ ROADS \ IMPROVEMENTS** \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ DEVELOPMENT \ RECREATIONAL \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ OTHER \ BANK STABILIZATION \ TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ STRUCTURE \ INTAKE/OUTFALL \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ STRUCTURE \ RECREATIONAL \ TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ BRIDGE \ CONSTRUCTION (NEW) \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \ TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ ROADS \ CROSSING (NON BRIDGE) \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ MAINTENANCE \TRANSPORTATION \ ROADS \ IMPROVEMENTS, \ TRANSPORTATION \ ROADS \ MAINTENANCE \TRANSPORTATION \ ROADS \ LOGGING **\TRANSPORTATION \ ROADS \ MAINTENANCE** \TRANSPORTATION \ ROADS \ MAINTENANCE, \ DEVELOPMENT \ RECREATIONAL \TRANSPORTATION \ ROADS \ MAINTENANCE, \ OTHER \ BANK STABILIZATION \ TRANSPORTATION \ ROADS \ MAINTENANCE, \ OTHER \ BANK STABILIZATION, \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ ROADS \ MAINTENANCE. \ STRUCTURE \ BRIDGE/RELATED WORK \TRANSPORTATION \ ROADS \ MAINTENANCE, \ TRANSPORTATION \ BRIDGE \ REPLACEMENT \ TRANSPORTATION \ ROADS \ MAINTENANCE. \ TRANSPORTATION \ ROADS \ CULVERT \TRANSPORTATION \ ROADS \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ CULVERT, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \TRANSPORTATION \ ROADS \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS **\TRANSPORTATION \UTILITY \ACCESS ROAD** \TRANSPORTATION \ UTILITY \ ACCESS ROAD, \ DREDGING \ BOAT SLIP, \ STRUCTURE \ DOCK \ FLOATING \TRANSPORTATION \ UTILITY \ ACCESS ROAD, \ TRANSPORTATION \ UTILITY \ AERIAL \TRANSPORTATION \ UTILITY \ ACCESS ROAD. \ TRANSPORTATION \ UTILITY \ MAINTENANCE **\TRANSPORTATION \UTILITY \AERIAL** \ TRANSPORTATION \ UTILITY \ AERIAL, \ STRUCTURE \ UTILITY LINE OR STRUCTURE \TRANSPORTATION \ UTILITY \ AERIAL, \ TRANSPORTATION \ ROADS \ MAINTENANCE \TRANSPORTATION \ UTILITY \ AERIAL, \ TRANSPORTATION \ UTILITY \ ACCESS ROAD \ TRANSPORTATION \ UTILITY \ AERIAL. \ TRANSPORTATION \ UTILITY \ MAINTENANCE

Appendix E Dredge and Fill Activities as Defined in the Corps Database (WorkType.xlsx) \TRANSPORTATION \ UTILITY \ AERIAL, \ TRANSPORTATION \ UTILITY \ STRUCTURE \ TRANSPORTATION \ UTILITY \ AERIAL, \ TRANSPORTATION \ UTILITY \ SUBMERGED **\TRANSPORTATION \UTILITY \BURIED** \ TRANSPORTATION \ UTILITY \ BURIED, \ DEVELOPMENT \ ASSOCIATED INFRASTRUCTURE, \ DEVELOPMENT \ RESIDENTIAL \ MULTI- FAMILY \TRANSPORTATION \ UTILITY \ BURIED, \ DEVELOPMENT \ COMMERCIAL, \ TRANSPORTATION \ UTILITY \ STRUCTURE \ TRANSPORTATION \ UTILITY \ BURIED, \ STRUCTURE \ UTILITY LINE OR STRUCTURE \ TRANSPORTATION \ UTILITY \ BURIED, \ TRANSPORTATION \ PIPELINE \ BURIED \TRANSPORTATION \ UTILITY \ BURIED, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \ TRANSPORTATION \ UTILITY \ BURIED, \ TRANSPORTATION \ UTILITY \ AERIAL \TRANSPORTATION \ UTILITY \ BURIED, \ TRANSPORTATION \ UTILITY \ SUBMERGED **\TRANSPORTATION \UTILITY \MAINTENANCE** \TRANSPORTATION \ UTILITY \ MAINTENANCE, \ STRUCTURE \ UTILITY LINE OR STRUCTURE \TRANSPORTATION \ UTILITY \ MAINTENANCE, \ TRANSPORTATION \ ROADS \ MAINTENANCE \TRANSPORTATION \ UTILITY \ MAINTENANCE, \ TRANSPORTATION \ UTILITY \ BURIED \TRANSPORTATION \ UTILITY \ MAINTENANCE, \ TRANSPORTATION \ UTILITY \ STRUCTURE **\TRANSPORTATION \ UTILITY \ STRUCTURE** \TRANSPORTATION \ UTILITY \ STRUCTURE, \ STRUCTURE \ REMOVAL \TRANSPORTATION \ UTILITY \ STRUCTURE, \ STRUCTURE \ UTILITY LINE OR STRUCTURE \TRANSPORTATION \ UTILITY \ STRUCTURE, \ TRANSPORTATION \ PIPELINE \ BURIED \TRANSPORTATION \ UTILITY \ STRUCTURE, \ TRANSPORTATION \ ROADS \ IMPROVEMENTS \TRANSPORTATION \ UTILITY \ STRUCTURE, \ TRANSPORTATION \ UTILITY \ BURIED \ TRANSPORTATION \ UTILITY \ STRUCTURE, \ TRANSPORTATION \ UTILITY \ MAINTENANCE **\TRANSPORTATION \UTILITY \SUBMERGED** \TRANSPORTATION \ UTILITY \ SUBMERGED, \ STRUCTURE \ WEIR \TRANSPORTATION \ UTILITY \ SUBMERGED, \ TRANSPORTATION \ UTILITY \ AERIAL

# **Appendix F**

U.S. Army Corps of Engineers Jacksonville District Programmatic Biological Opinion



UNITED STATES DEPARTMENT OF COMMERCE

National Oceanic and Atmospheric Administration NATIONAL MARINE: FISHERIES SERVICE Southeast Regional Office 263 13th Avenue South St. Petersburg, Florida 33/701-5505 http://sero.nmfs.noaa.gov

> F/SER31: NMB SER-2015-17616

NOV 2 0 2017

Mr. Donald W. Kinard Chief, Regulatory Division U.S. Army Corps of Engineers P.O. Box 4970 Jacksonville, Florida 32232-0019

Ref.: U.S. Army Corps of Engineers Jacksonville District's Programmatic Biological Opinion (JAXBO)

Dear Mr. Kinard:

Enclosed is the National Marine Fisheries Service's (NMFS's) Programmatic Biological Opinion (Opinion) based on our review of the impacts associated with the U.S. Army Corps of Engineers (USACE's) Jacksonville District's authorization of 10 categories of minor in-water activities within Florida and the U.S. Caribbean (Puerto Rico and the U.S. Virgin Islands).

The Opinion analyzes the effects from 10 categories of minor in-water activities occurring in Florida and the U.S. Caribbean on sea turtles (loggerhead, leatherback, Kemp's ridley, hawksbill, and green); smalltooth sawfish; Nassau grouper; scalloped hammerhead shark, Johnson's seagrass; sturgeon (Gulf, shortnose, and Atlantic); corals (elkhorn, staghorn, boulder star, mountainous star, lobed star, rough cactus, and pillar); whales (North Atlantic right whale, sei, blue, fin, and sperm); and designated critical habitat for Johnson's seagrass; smalltooth sawfish; sturgeon (Gulf and Atlantic); sea turtles (green, hawksbill, leatherback, loggerhead); North Atlantic right whale; and elkhorn and staghorn corals in accordance with Section 7 of the Endangered Species Act. We also analyzed effects on the proposed Bryde's whale. We based our analysis on project-specific information provided by USACE, consultants, and NMFS's review of published literature. The Opinion concludes that the suite of activities evaluated within the Opinion is likely to adversely affect, but is not likely to jeopardize, the continued existence of Johnson's seagrass and is likely to adversely affect, but is not likely to destroy or adversely modify, critical habitat for smalltooth sawfish and Johnson's seagrass.



We look forward to further cooperation with you on other USACE projects to ensure the conservation and recovery of our threatened and endangered marine species. If you have any questions regarding this consultation, please contact Nicole Bonine, Consultation Biologist, at (727) 824-5336, or by email at Nicole.Bonine@noaa.gov.

NOV 2 0 2017

Sincerely,

Roy E. Crabtree, Ph.D. Regional Administrator

Enclosure File: 1514-22.F.4

## Endangered Species Act - Section 7 Consultation Biological Opinion

Agency:

United States Army Corps of Engineers, Jacksonville District

Applicant:

Activity:

United States Army Corps of Engineers, Jacksonville District

Authorization of Minor In-Water Activities throughout the Geographic Area of Jurisdiction of the U.S. Army Corps of Engineers Jacksonville District, including Florida and the U.S. Caribbean

**Consulting Agency:** 

National Marine Fisheries Service Southeast Regional Office Protected Resources Division (SER-2015-17616)

Date Issued:

Approved By:

Nov. 20, 2017 BELLE

Roy F. Crabtree, Ph.D. Regional Administrator

# **Table of Contents**

1 2		Consultation History Description of the Proposed Action and Potential Routes of Effect to Species and Cr Habitat	18 ritical 20
	2.1	Project Action Area	23
		2.1.1 Species Specific Restriction and Exclusion Zones within the Action Area	24
		2.1.2 USACE Effect Determinations for Listed Species and Critical Habitat	35
		2.1.3 Listed Species and Critical Habitat Occurring within the Action Area	38
	2.2	Activities Analyzed, Project Design Criteria, and Potential Routes of Effect	44
		2.2.1 Activity 1 (A1): Shoreline Stabilization	96
		2.2.2 Activity 2 (A2): Pile-Supported Structures and Anchored Buoys	109
		2.2.3 Activity 3 (A3): Maintenance, Minor, and Muck Dredging	134
		2.2.4 Activity 4 (A4): Water-Management Outfall Structures and Associated Endwa	alls 152
		2.2.5 Activity 5 (A5): Scientific Survey Devices	164
		2.2.6 Activity 6 (A6): Boat Ramps	171
		2.2.7 Activity 7 (A7): Aquatic Habitat Enhancement, Establishment, and Restoration Activities	on 181
		2.2.8 Activity 8 (A8): Transmission and Utility Lines	203
		2.2.9 Activity 9 (A9): Marine Debris Removal	216
		2.2.10 Activity 10 (A10): Temporary Platforms, Fill, and Cofferdams	220
		2.2.11 Summary and Cumulative Effect of Proposed Action to Listed Species and C Habitat	ritical 233
	2.3	Project-Specific Review	236
	2.4	Programmatic Review	240
3	3.1	Status of Listed Species and Critical Habitat Likely to be Adversely Affected Status of Species Likely to be Adversely Affected	242 242
		3.1.1 Johnson's Seagrass	242
	3.2	Status of Critical Habitat Likely to be Adversely Affected	252
		3.2.1 Johnson's Seagrass Critical Habitat	252
		3.2.2 Smalltooth Sawfish Critical Habitat	255
4		Environmental Baseline	265
	4.1	Factors Affecting Johnson's Seagrass and Johnson's Seagrass Critical Habitat in the Action Area	ne 265
		4.1.1 Federal and State Actions	265

		4.1.2 Conservation and Recovery Activities That May Benefit Johnson's Seagrass an Johnson's Seagrass Critical Habitat	d 267
	4.2	Factors Affecting Smalltooth Sawfish Critical Habitat within the Action Area	267
		4.2.1 Federal Actions	267
		4.2.2 State or Private Actions	267
		4.2.3 Other Potential Factors	268
5	5.1	Effects of the Action Effects to Johnson's Seagrass	268 268
	5.2	Effects to Johnson's Seagrass Critical Habitat	269
	5.3	Effects to Smalltooth Sawfish Critical Habitat	270
6	6.1	Cumulative Effects Cumulative Effects to Johnson's Seagrass and Johnson's Seagrass Critical Habitat	271 271
	6.2	Cumulative Effects to Smalltooth Sawfish Critical Habitat	272
7	7.1	Jeopardy Analysis Johnson's Seagrass	273 273
8	8.1	Destruction and Adverse Modification Analysis Johnson's Seagrass Critical Habitat	275 276
	8.2	Smalltooth Sawfish Critical Habitat	277
		8.2.1 Protect and Restore Smalltooth Sawfish Habitat (Recovery Objective #2)	278
		Shallow, Euryhaline Essential Feature Impacts	279
		Red Mangrove Essential Feature Impacts	281
		Summary of Impacts to the Essential Features	283
		8.2.2 Ensure Smalltooth Sawfish Abundance Increases (Recovery Objective #3)	284
9 10 11 12 12 12	) 2 3 1	Conclusion Incidental Take Statement Conservation Recommendations Reinitiation of Consultation Literature Cited Appendix A: Authorities under which an Action Will Be Conducted 14.1.1 Current Permits Used to Authorize Projects in Florida under NMFS Jurisdiction 303	284 285 286 287 287 303 n
15	5	Appendix B: In-water Construction Noise	308
	15.1	I Glossary of Acoustic Terms	308
	15.2	2 How in-water Noise effects ESA-listed Species Are Evaluated	310
		15.2.1 Exposure Criteria Used to Determine Potential Effects	312
	15.3	3 Calculations of Noise Thresholds for Pile Driving	314

15.3.1 Pile-Driving Noise Calculations	
15.3.2 Noise Control Measures to Reduce Injury	
15.3.3 Bubble Curtain Specifications for Pile Driving	
15.4 How to Calculate Noise Impacts	
15.4.1 Calculation Steps for Pile Driving	
15.4.2 Example Noise Calculation for a Single Family Dock	
16 Appendix C: North Atlantic Right Whale Information Form	
17 Appendix D: Frac-Out Plan Example	

# List of Figures

Figure 11. Pile-supported Structures. Example of a marginal dock on the left, boatlift with I- beam in the middle, and a chickee on the right. (The left and middle photos were provided by the Florida Marine Contractors Association and right photo from www.cnn.com.)	10
Figure 12. Sample Marina Images. The left image shows an example of an open water marina (provided by the Florida Marine Contractors Association), while the right image shows an example of marina contained within an upland cut basin (©2014 Google)	11
Figure 13. Canal dredging footprint drawing	35
Figure 14. Mechanical dredge (clamshell bucket and barge)	36
Figure 15. Cutterhead pipeline dredge schematic and representative close-up photographs (provided by USACE)	37
Figure 16. Sample drawing of an outfall pipe with a manatee grate. Manatee grates are typical attached to the pipe, but in this case, piles are placed in front of the pipe	lly 53
Figure 17. Example drawing of a boat ramp installed by placing prefabricated concrete slabs. Note that excavation for this example was minimal (removed approximately 10 cubic yards in this instance) and limited to shaping the slope of the boat ramp. This ramp extends to -3 ft MLW. Drawings prepared by Glen Boe and Associates, Inc. submitted to NMFS as part of a consultation request.	72
Figure 18. Oyster reefs. The left image shows oyster bags, the middle is an oyster mat, and th right is a barge filled with loose oyster cultch. All 3 images are from the Charlotte Harbor Habitat Restoration Plan (Boswell et al. 2012)	e 82
Figure 19. Living shoreline. This sample living shoreline cross-section is provided by NOAA Habitat Conservation and Restoration website (http://www.habitat.noaa.gov/restoration/techniques/lsimplementation.html)	's 82
Figure 20. Seagrass restoration. The left image is the placement of sediment to return a blowher to pre-injury elevation. The image on the right shows bird stakes placed in a restoration area (Both images are from www.darrp.noaa.gov).	ole 83
Figure 21. Artificial reef materials. The left image is an open top and bottom tetrahedron design the middle image is a series of discs mounted to a pile, and the right image is a reef ball. Left and middle images are from Reef Maker (http://www.reefmaker.com) and the right image is from the Reef Ball Foundation (www.reefball.org)	;n, 83
Figure 22. Sample image of horizontal directional drilling (Image from Underground Solution at http://www.undergroundsolutions.com)	s .04
Figure 23. Cathodic pile jacket (Image provided by USACE)	21
Figure 24. Structural cathodic pile jacket (Image provided by USACE)	22
Figure 25. Map of smalltooth sawfish critical habitat –CHEU and TTIEU	56

Figure 26. Diagram of a dredged channel	50
Figure 27. From left to right: current shoreline, + 3.5 in (+ 9 cm); + 18.5 in (+ 47 cm); and + 38.97 in (+ 99 cm) sea level rise by 2060	53
Figure 28. The relative effects to animals: distance to/from a noise source	1
Figure 29. Example of a confined space for a noise analysis. The left image shows a 150 ft radius in a channel resulting in the channel not being wide enough to ensure more than half the channel was free from cumulative injurious noise levels. The right image shows a portion of Sa Juan Harbor and a 260 ft radius that would not block movement of species if piles were installed at this location because species would be able to move along the left side of the image between the 2 structures.	տ d
Figure 30. An example of intermediate spreading loss where surface reflections result in non- uniform spreading of sound waves	25
Figure 31. Screenshot of the Spreading Loss Calculator. The dB loss over any range can be determined for 3 types spreading loss models (10 logR, 15 logR, and 20 logR). For example, at 50 m, there is a 25.5 dB reduction in noise from a pile due to intermediate transmission loss (15 logR).	28
Figure 32. An example graphic visualizing impact zones for a pile-driving project. Graphical representations of the impact zones are useful analytical tools in visualizing the area where project impacts will occur within the species' habitat	28
Figure 33. Screenshot of original Spreading Loss Calculator prior to any modifications	30
Figure 34. Screenshot of the Spreading Loss Calculator after the first column was modified 33	31

# List of Tables

Table 1. Limited Exclusion Zones in Smalltooth Sawfish Critical Habitat	. 25
Table 2. Gulf sturgeon critical habitat migratory restriction zones	. 28
Table 3. North Atlantic Right Whale Educational Sign Zone	. 32
Table 4. USACE Project Effects Determinations for Activities Occurring in Florida	. 37
Table 5. Listed Species NMFS Believes Are Likely to Occur in or near the Action Area	. 38
Table 6. Designated Critical Habitat NMFS Believes is In or Near the Action Area	. 39
Table 7. Essential Features/PCEs/PBFs of Each Critical Habitat Unit in Florida and the U.S.         Caribbean	. 40
Table 8. Projected Number of Authorizations Occurring in Certain Critical Habitat Units per      Activity for 5-year Period	. 46
Table 9. NMFS Effects Determinations for Johnson's Seagrass	. 54
--	-----------
Table 10. NMFS Effects Determinations for Nassau Grouper	. 58
Table 11. NMFS Effects Determinations for the Central and Southwest Atlantic DPS of         Scalloped Hammerhead Shark	. 60
Table 12. NMFS Effects Determinations for ESA-listed Whales	. 62
Table 13. NMFS Effects Determinations for NWA DPS of Loggerhead Sea Turtle Critical         Habitat	. 66
Table 14. NMFS Effects Determinations for Acropora Critical Habitat	. 67
Table 15. NMFS Effects Determination for North Atlantic Right Whale Critical Habitat	. 68
Table 16. NMFS Effects Determination for Green, Hawksbill, and Leatherback Sea Turtle         Critical Habitat	. 70
Table 17. NMFS Effects Determination for Atlantic Sturgeon Critical Habitat	. 71
Table 18. Impact Hammer Sound Source Levels and Impact Radius Distances	. 94
Table 19. Vibratory Hammer Sound Source Levels and Impact Radius Distances	. 95
Table 20. NMFS Effects Determinations for In-water Construction Noise Analyzed under this         Opinion to Sea Turtles and ESA-listed Fish	3 . 96
Table 21. USACE and NMFS Determinations on the Effects of Shoreline Stabilization         Activities (Activity 1) to ESA-listed Species listed in Table 5	102
Table 22. Estimated Amount of Waterbottom Affected By Shoreline Stabilization Projects per      5-Year Period	r 103
Table 23. USACE and NMFS Determinations on the Effects of Shoreline Stabilization         Activities (Activity 1) to Designated Critical Habitat	106
Table 24. Estimated Amount of Waterbottom Affected By Shoreline Stabilization Projects in Smalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat per 5-Year Period	1 106
Table 25. USACE and NMFS Determinations on the Effects of Pile-Supported Structures and         Anchored Buoys (Activity 2) to ESA-listed Species listed in Table 5	l 119
Table 26. Estimated Amount of Waterbottom Affected by Pile-Supported Structures andAnchored Buoys Occurring Outside of Johnson's Seagrass Critical per 5-Year Period	120
Table 27. Estimated Amount of Waterbottom Affected by Pile-Supported Structures and Anchored Buoys Occurring Within Johnson's Seagrass Critical Habitat per 5-Year Period	121

Table 28. USACE and NMFS Determinations on the Effects of Pile-Supported Structures andAnchored Buoys (Activity2) to Designated Critical Habitats126
Table 29. Estimated Amount of Waterbottom Affected by Pile-Supported Structures and Anchored Buoys Occurring Within Smalltooth Sawfish Critical Habitat per 5-Year Period 127
Table 30. Estimated Amount of Waterbottom Affected by Pile-Supported Structures andAnchored Buoys Occurring Within Gulf Sturgeon Critical Habitat per 5-Year Period
Table 31. USACE and NMFS Determinations on the Effects of Dredging Activities (Activity 3)to ESA-listed Species listed in Table 5
Table 32. Estimated Amount of Waterbottom Affected by Dredging Activities Per 5-Year      Period.      144
Table 33. USACE and NMFS Determinations on the Effect of Dredging Activities (Activity 3)on Designated Critical Habitat
Table 34. Total Waterbottom Affected by Dredging Projects within Smalltooth Sawfish, GulfSturgeon, and Johnson's Seagrass Critical Habitat.149
Table 35. USACE and NMFS Determinations on the Effects of Water-Management OutfallStructures and Associated Endwall Activities (Activity 4) to ESA-listed Species Listed in Table5
Table 36. Estimated Amount of Waterbottom and Shoreline Areas and Mangrove ShorelineAffected by Water-Management Outfall Structures and Associated Endwall Projects per 5-YearPeriod.157
Table 37. USACE and NMFS Determinations on the Effects of Water-Management andEndwall Activities (Activity 4) on Designated Critical Habitat160
Table 38. Estimated Amount of Waterbottom and Shoreline Areas Affected by Water- Management Outfall Structures and Associated Endwall Projects in Smalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat
Table 39. USACE and NMFS Determinations on the Effects of Scientific Survey DeviceActivities (Activity 5) to ESA-listed Species Listed in Table 5166
Table 40. Estimated Number of Scientific Survey Activities and the Associated TemporaryHabitat Impacts over a 5-Year Period.166
Table 41. USACE and NMFS Determinations on the Effects of Scientific Survey DeviceActivities (Activity 5) to Designated Critical Habitat169
Table 42. Estimated Number of Scientific Survey Device Projects and Associated TemporaryImpacts in Critical Habitat with Effects Determinations169
Table 43. USACE and NMFS Determinations on the Effects of Boat Ramp Activities (Activity6) on ESA-listed Species Listed in Table 5175
10

Table 60. Cofferdan	USACE and NMFS Determinations on the Effects of Temporary Platform, Fill, and n Activities under (Activity 10) to Designated Critical Habitat	28
Table 61. Cofferdan Seagrass (	Estimated Amount of Waterbottom Affected by Temporary Platform, Fill, and n Activities under (Activity 10) in Smalltooth Sawfish, Gulf sturgeon, and Johnson's Critical Habitat per 5-Year Period	229
Table 62.	Summary of NMFS Determination of Effects to Species by Activity Type 2	233
Table 63.	Summary of NMFS Determination of Effects to Critical Habitat by Activity Type. 2	234
Table 64.	Designated critical habitat units for Johnson's seagrass	253
Table 65.	Impacts to Johnson's Seagrass Anticipated per 5-year period2	:69
Table 66.	Impacts to Johnson's Seagrass Critical Habitat Anticipated per 5-year period2	270
Table 67.	Impacts to Smalltooth Sawfish Critical Habitat Anticipated per 5-year period2	271
Table 68.	Summary of Impacts to the Shallow, Euryhaline Habitat Essential Feature	.81
Table 69.	Summary of Impacts to the Red Mangrove Essential Feature	.83
Table 70.	NWPs used by the USACE Jacksonville District	04
Table 71. Species w	USACE General Permits in Florida and the U.S. Caribbean That May Affect vithin NMFS' Jursidiction	606
Table 72.	Hearing ranges of listed species	511
Table 73.	Impact pile-driving threshold noise levels for fish and sea turtles	12
Table 74. pile-drivi	Continuous noise threshold levels for fish and turtles from exposure to vibrato ng noise	<b>ry</b> 313
Table 75.	Impact Hammer sound source levels and impact radius distances	517
Table 76.	Vibratory Hammer sound source levels and impact radius distances	518
Table 77.	Effectiveness and cost of noise control measures for pile driving	520
Table 78.	Back-Calculation of Source Levels	529
Table 79.	Calculations of Threshold Exceedances	30
Table 80.	Calculations of Impact Zones Based on Source Levels for the Project	31

# Acronyms and Abbreviations

ADA	Americans with Disabilities Act
AIWW	Atlantic Intracoastal Waterway
ATON	Aid-to-Navigation

BMP	Best Management Practice		
CHEU	Charlotte Harbor Estuary Unit		
cSEL	Cumulative Sound Exposure Level		
CWA	Clean Water Act		
dB	decibels		
dB re 1 µPa	decibels relative to 1 micro-Pascal-squared second		
DPS	Distinct Population Segment		
EFH	Essential Fish Habitat		
EPA	U.S. Environmental Protection Agency		
ESA	Endangered Species Act		
FEMA	Federal Emergency Management Agency		
FDEP	Florida Department of Environmental Protection		
FKNMS	Florida Keys National Marine Sanctuary		
FPL	Florida Power and Light		
GIS	Geographic Information System		
GIWW	Gulf Intracoastal Waterway		
GP	General Permit		
GRBO	Gulf of Mexico Regional Biological Opinion		
ICW	Intracoastal Waterway		
IP	Individual Permit		
IPCC	Intergovernmental Panel on Climate Change		
JAXBO	Jacksonville District's Programmatic Biological Opinion		
LAA	Likely to Adversely Affect		
LOP	Letters of Permission		
MHW	Mean High Water		
MHWL	Mean High Water Line		
MLLW	Mean Lower Low Water		
MLW	Mean Low Water		
MMPA	Marine Mammal Protection Act		
Ν	Number of Projects		
N/A	Not Applicable		
NA DPS	North Atlantic Distinct Population Segment		
NE	No Effect		
NLAA	Not Likely to Adversely Affect		
NMFS	National Marine Fisheries Service		
NOAA	National Oceanic and Atmospheric Administration		
NP	Not Present		
NPS	National Park Service		
NWA DPS	Northwest Atlantic Distinct Population Segment		
NWP	Nationwide Permit		
Р	Present		
PATON	Private Aid to Navigation		
PBF	Physical and Biological Feature		
PCE	Primary Constituent Element		
PDC	Project Design Criteria		
PGP	Programmatic General Permit		
	- 12		

PRD	Protected Resources Division
PVC	Polyvinyl Chloride
RGP	Regional General Permit
RHA	Rivers and Harbors Act
RM	Red Mangrove
RMS	Root Mean Square
SAJ	South Atlantic Jacksonville
SARBO	South Atlantic Regional Biological Opinion
SH	Shallow, Euryhaline Habitat
SEL	Sound Exposure Level
SFWMD	South Florida Water Management District
SP	Standard Permit
SPGP	State Programmatic General Permit
sSEL	Single Strike Sound Exposure Level
SWPBO	Statewide Programmatic Biological Opinion
TNAP	Temporary Noise Attenuation Pile
TTIEU	Ten Thousand Islands/Everglades Unit
USACE	U.S. Army Corps of Engineers
USCG	U.S. Coast Guard
USFWS	U.S. Fish and Wildlife Service
WCIND	West Coast Inland Navigational District

# Units of Measurement

#### **Temperature**

<u>i cimperaturi c</u>	
°F	degrees Fahrenheit
°C	degrees Celsius

# Length and Area

ac	acre(s)
cm	centimeter(s)
ft	foot/feet
ft²	square foot/feet
in	inch(es)
km	kilometer(s)
lin ft	linear foot/feet
m	meter(s)
m <sup>2</sup>	square meter(s)
mm	millimeter(s)
nmi	nautical mile(s)
nmi²	square nautical mile(s)

#### Background

Section 7(a)(2) of the Endangered Species Act (ESA) of 1973, as amended (16 U.S.C. §1531 *et seq.*), requires that each federal agency ensure that any action authorized, funded, or carried out by the agency is not likely to jeopardize the continued existence of any endangered or threatened species or result in the destruction or adverse modification of critical habitat of those species. When the action of a federal agency may affect a protected species or its critical habitat, that agency is required to consult with either the National Marine Fisheries Service (NMFS) or the U.S. Fish and Wildlife Service (USFWS), depending upon the protected species or critical habitat that may be affected.

Consultations on most listed marine species and their designated critical habitat are conducted between the action agency and NMFS. Consultations are concluded after NMFS determines the action is not likely to adversely affect listed species or critical habitat or issues a Biological Opinion ("Opinion") that determines whether a proposed action is likely to jeopardize the continued existence of a federally listed species, or destroy or adversely modify federally designated critical habitat. The Opinion also states the amount or extent of listed species incidental take that may occur and develops non-discretionary measures that the action agency must take to reduce the effects of said anticipated/authorized take. The Opinion may also recommend discretionary conservation measures. No destruction or adverse modification of critical habitat may be authorized. The issuance of an Opinion detailing NMFS's findings concludes ESA Section 7 consultation.

This document represents NMFS's Opinion based on our review of impacts associated with 10 categories of in-water activities that the USACE's Jacksonville District authorizes under Section 10 of the Rivers and Harbors Act (RHA) of 1899 and Section 404 of the Clean Water Act (CWA) throughout Florida and the U.S. Caribbean, including:

- 1. Shoreline stabilization (e.g., installation, repair, and removal of structures)
- 2. Pile-supported structures and anchored buoys (e.g., installation, repair, and removal of structures)
- 3. Dredging including maintenance, minor, and muck dredging
- 4. Water-management outfall structures and associated endwalls<sup>1</sup> (e.g., installation, repair, and removal of water outfall structures)
- 5. Scientific survey devices (e.g., installation, repair, and removal of structures)
- 6. Boat ramps (e.g., installation, repair, and removal of structures)

<sup>&</sup>lt;sup>1</sup> Endwalls are retaining walls at the end of the outfall structure that protect the area surrounding the outfall pipe from scouring.

- 7. Aquatic habitat enhancement, establishment, and restoration activities (oyster reef and living shorelines, seagrass restoration, artificial reefs, fill to restore natural contours or improve water quality)
- 8. Transmission and utility lines (e.g., installation, repair, and removal of aquatic and subaqueous lines)
- 9. Marine debris removal
- 10. Temporary platforms, fill, and cofferdams (e.g., installation, repair, and removal of structures)

We analyze the effects of these 10 categories of activities in Florida and the U.S. Caribbean on the endangered (E) and threatened (T) species and critical habitat under our jurisdiction, in accordance with Section 7 of the ESA. This Opinion is based on information provided by USACE and the best scientific and commercial data available.

#### Programmatic Consultations

Programmatic consultations<sup>2</sup> allow for streamlined review of groups of frequently occurring or routine activities or Federal action agency policies, plans, regulations or programs that have well-understood and predictable effects on ESA-listed species and designated critical habitat.<sup>3</sup> Programmatic consultations can be used to evaluate the effects of authorizing certain categories of frequently occurring activities or of agency policy or programs, where specifics of any individual future project (either of the given category or type of activity, or occurring under the policy or program), such as the specific location, are not definitively known at the time of the programmatic consultation.

As is done in this Opinion, a Programmatic Consultation generally identifies project design criteria (PDCs), which are the specific criteria, including the technical and engineering specifications, indicating how an individual project must be sited, constructed, or otherwise carried out to avoid or minimize adverse effects to ESA-listed species or designated critical habitat. The PDCs serve 2 important purposes. First, they ensure that the actions under consultation are sufficiently similar that their effects can be analyzed together. Second, the PDCs help protect species and critical habitat, and ensure that the action agency is meeting its obligation under Section 7(a)(2); in designing the PDCs, NMFS and the action agency work to establish conditions that avoid adverse effects on listed species or designated critical habitat or,

<sup>&</sup>lt;sup>2</sup> It is important to note that the term "programmatic" is defined differently by NMFS when discussing a Programmatic Consultation or Programmatic Biological Opinion than it is by USACE when discussing a Programmatic *General Permit* (see Appendix A).

<sup>&</sup>lt;sup>3</sup> See, e.g., Joint Services memorandum, Alternative Approaches for Streamlining Section 7 Consultation on Hazardous Fuels Treatment Projects, https://www.fws.gov/endangered/esa-library/pdf/streamlining.pdf; 68 FR 1628 (January 13, 2003).

where the adverse effect cannot be avoided, to limit them to predictable levels that will not jeopardize the continued existence of listed species or destroy or adversely modify critical habitat either at the individual project level or in aggregate.

The Programmatic Consultation document evaluates the aggregate effect of categories of related actions or of the agency program. This includes the amount or extent of incidental take that is expected, if sufficient information exists to estimate take. Since programmatic consultations evaluate effects of expected future actions, the action agency must provide projections of the number of activities and the extent of expected effects from the proposed activities. The Programmatic Consultation document must demonstrate that, when the PDCs are followed, the aggregate expected effect of all projects is not likely to adversely affect listed species or their critical habitat(s), or will not jeopardize species or destroy or adversely modify their critical habitat(s), as applicable. At the project-specific consultation stage, each proposed action is reviewed to determine if it can be implemented according to the PDCs. For example, an action agency may certify that the expected effect of the project to be authorized is consistent with the PDCs and other conditions in the Programmatic Consultation. Adjustments to the project(s) may be necessary to bring them into compliance with the Programmatic Consultation document. Finally, the project-specific consultation procedures provide contingencies for proposed projects that cannot be implemented in accordance with the PDCs; for example, separate consultations may be performed on these projects if they are too dissimilar from those described in the Programmatic Consultation document. In addition, the Programmatic Consultation must provide a process for tracking the actual effects of the proposed activities, once implemented, and ensure that the number and scope of the effects do not exceed those analyzed in the consultation or otherwise require reinitiation. NMFS generally conducts a programmatic-level review of the actual effects and compliance with the Programmatic Consultation on an annual basis.

The following elements, which generally are included in a Programmatic Consultation to ensure its compliance with ESA Section 7 and its implementing regulations, have been included in this Opinion:

- Description of the proposed action, including the categories of similar projects under consultation (referred to as the Activities) and the PDCs applicable to all activities (general PDCs) and to the individual categories of activities (activity-specific PDCs) that are designed to avoid or minimize future adverse effects on listed species and critical habitat (Section 2.2).
- 2. Description of the manner in which the projects, when implemented consistent with the general and activity-specific PDCs, may affect listed species and critical habitat and evaluation of expected effects of the covered projects (Section 2.2). For each activity, we describe the process we used, in coordination with the action agency, to evaluate the expected effects, including the assumptions about the number and scope of each expected project. This Programmatic Opinion does not authorize any take.
- 3. Procedures for streamlined project-specific review, and the process for separate consultations for projects that do not meet the requirements of the Opinion (Section 2.3).

4. Procedures for monitoring projects and validating the predicted effects, and for the comprehensive review of the projects authorized in reliance on the Opinion as a whole (Section 2.4).

# 1 Consultation History

On December 14, 2015, the USACE requested a Programmatic Biological Opinion to address certain activities authorized under Section 10 of the RHA and Section 404 of the CWA in Florida. Consultation was initiated that day, though we continued to work with USACE to refine the proposed action. The original request was to address 11 categories of USACE-authorized activities. Through discussion with the USACE, the 11 categories of activities were reduced to 10 by combining 2 activities. In particular, we combined the reconfiguration and repair of existing docking facilities within a USACE-authorized marina with installation, repair, replacement, and removal of pile-supported structures and anchored buoys, since both categories primarily involve pile-supported structures. Additionally, this Opinion modifies some of the PDCs that the USACE included in its consultation request.

The USACE requested that this Opinion analyze activities authorized under its multiple existing and proposed general permits (GPs) in Florida. As explained below, the USACE had previously consulted with us on several of these GPs and this Opinion covers the activities authorized under 5 such permits, as noted below. Including the activities authorized under multiple GPs in this Opinion allowed us to review the majority of the minor in-water construction activities that the USACE permits in Florida under Section 10 of the RHA and Section 404 of the CWA, and to analyze the cumulative effects of these actions.

Once the USACE completes any process and other requirements (e.g., public notices or comment periods) necessary for full implementation, this Opinion will address the effects of activities authorized by the following renewals of existing GPs, and replaces and supersedes any prior consultations on the permits or prior iterations of these permits:

- Renewal of the Florida State Programmatic General Permit (SPGP). The USACE requested programmatic consultation on this renewal on August 26, 2015. We did not complete a separate programmatic consultation on that GP, but rather incorporated the actions authorized under that permit in this Opinion. The previous SPGP was evaluated through a Programmatic Opinion issued by NMFS on December 21, 2011 (SPGP IV-RI, SER-2011-05980),
- Renewal of the 12 USACE South Atlantic Jacksonville (SAJ) District GPs, which were previously evaluated through a Programmatic Opinion issued by NMFS on December 19, 2012 (12 USACE SAJ General Permits, SER 2011-01939).
- Renewal of the GP SAJ-82, which was previously evaluated in a Biological Opinion issued by NMFS on June 10, 2014 (SER-2008-02958).
- Renewal of the Programmatic General Permit (PGP) SAJ-42, which was previously evaluated in a Biological Opinion issued by NMFS on February 10, 2011 (SER-2008-01790).

Additionally, this Opinion will supersede the Florida Statewide Programmatic Biological Opinion (SWPBO) issued by NMFS on December 4, 2015 (SER-2013-12540).

This Programmatic Opinion will not address the following completed or in progress consultations:

- Renewal of SAJ-99 (SER-2014-13378) or SAJ-71 (SER-2015-17183) used for the deposition of aquaculture materials.
- Completion of the proposed SAJ-112 for coral propagation and nursery structures off the Florida, Puerto Rico, and U.S. Virgin Islands (SER-2014-15282).
- South Atlantic Regional Biological Opinion (SARBO) (NMFS 1997) and the Gulf of Mexico Regional Biological Opinion (GRBO) (NMFS 2007b), which both address federal actions associated with maintenance dredging.

In the SWPBO issued on December 4, 2015, we evaluated the USACE's authorization of 11 categories of minor in-water activities under Nationwide Permits (NWPs) or Individual Permits (IPs). Although this Opinion includes the same categories of activities as SWPBO,<sup>4</sup> it is broader than the SWPBO because it also considers those activities when authorized under USACE regional, general, and programmatic permits (i.e., it is not simply limited to authorization of those activities under USACE NWPs or IPs).

After working directly with the USACE to develop the updated PDCs for this Opinion, we sent a draft of the consultation history, descriptions of the actions, PDCs, and the assumptions that went into estimating the effects of the activities to the USACE on March 10, 2016. The USACE provided comments on March 17, 2016.

On November 14, 2016, the USACE requested programmatic consultation on 11 categories of activities associated with the proposed 2017 Nationwide Permits for work in Puerto Rico and the U.S. Virgin Islands.

On January 12 and 31, 2017, we met with the USACE to discuss expanding this consultation to include not only activities authorized under the proposed 2017 NWPs in Puerto Rico and the U.S. Virgin Islands, but also activities authorized under IPs within that area. Together with the USACE, we drafted and reviewed additional PDCs and we decided it was appropriate to expand the consultation. The USACE requested to consult on the same 10 categories of activities as

<sup>&</sup>lt;sup>4</sup> As we explained above, we have revised the description of the covered categories from 11 categories to 10 categories by combining "the reconfiguration and repair of existing docking facilities within a USACE-authorized marina" with "installation, repair, replacement, and removal of pile-supported structures and anchored buoys," since both categories primarily involve pile-supported structures. Thus, although the SWPBO covered 11 categories of activities, and this Opinion covers 10, the Opinions discuss the same categories of actions.

analyzed in this Opinion, and an additional 11<sup>th</sup> category, aquaculture. NMFS and the USACE decided to not consult on aquaculture in this Opinion.

At the USACE's request, we provided the USACE a draft of the entire document for review before we completed the consultation.

# 2 Description of the Proposed Action and Potential Routes of Effect to Species and Critical Habitat

This Programmatic Opinion evaluates 10 categories of the most common, minor in-water activities that the USACE regularly authorizes in the state of Florida and the U.S. Caribbean under Section 404 of the CWA and Section 10 of the RHA. The USACE authorizes these activities using a variety of mechanisms including regional general permits (RGPs), PGPs, NWPs, and IPs. This Opinion does not cover all types of activities or projects that the USACE permits under its CWA or RHA authority, but instead focuses on those minor projects that the USACE regularly authorizes that have predictable effects. This Opinion does not directly link each activity to a specific USACE CWA or RHA permit (by which we mean we are not consulting on a particular RGP, PGP, or NWP). Instead, the Opinion focuses on the categories of activities authorized under the USACE's CWA and RHA permitting authority and the effects those activities may have on species and critical habitat under NMFS's purview.

The goal of this Opinion is to streamline and consolidate ESA Section 7 consultation for the majority of minor in-water coastal development activities that the USACE Jacksonville District authorizes in Florida and the U.S. Caribbean. In this Opinion, we are able to provide a more comprehensive and cohesive review of the majority of the minor in-water construction projects in Florida and the U.S. Caribbean that are permitted by the USACE under Section 10 of the RHA and Section 404 of the CWA and to analyze the cumulative effects of these actions, than if we consulted on a project-by-project basis.

Below, we provide a description of:

- How the Opinion may be used (described below)
- The areas in Florida and the U.S. Caribbean in which the permitted activities covered under this Opinion occur, including areas where additional restrictions apply (Section 2.1).
- The 10 categories of activities covered by this Opinion, including the PDCs those activities must meet to be covered (Section 2.2).
- An estimate of the number of activities that the USACE (or an entity with delegated authority from the USACE) will permit over a 5-year period that are covered by this Opinion (Section 2.2).
- The project-specific review (Section 2.3) and programmatic review (Section 2.4) requirements necessary to ensure that the reliance on this Opinion is limited to those actions in Florida and the U.S. Caribbean that meet the PDCs in this Opinion and are consistent with this Opinion.

#### Use of the Opinion

As explained above, the USACE requested programmatic consultation on the effect of authorizing projects falling within 10 categories of activities occuring throughout the state of Florida and the U.S. Caribbean. The USACE determined that the 9 out of the 10 categories of activities may affect certain ESA-listed species or designated critical habitat within NMFS's purview (all but Activity 9, marine debris removal; see Section 2.1.2 for the USACE's effects determinations). NMFS believes that all of the activities described below, if carried out as described, will have effects on ESA-listed species and/or critical habitat as discussed herein. The USACE may rely on this Opinion to meet its ESA Section 7 consultation requirements when authorizing activities that meet the PDCs and other requirements of the Opinion. Nothing in this Opinion precludes the USACE from determining that a future project does not affect an ESA-listed species or designated critical habitat. NMFS and the USACE will continue to discuss the Opinion as it is applied, at the project-specific and programmatic reviews described in Sections 2.3 and 2.4, and may refine it in the future.

Two USACE Divisions, the USACE Regulatory Division and the USACE Civil Works Division, have responsibility for authorizing and/or implementing in-water projects under the CWA and the RHA and consulting on the effects of those projects under the ESA. Both Divisions can satisfy their ESA Section 7 requirements by relying on this Opinion for projects meeting the requirements of this Opinion.

In addition, entities to which the USACE has delegated certain CWA Section 404 and/or RHA Section 10 permitting authority—such as the Florida Department of Environmental Protection (FDEP), a water management district, or a local government with delegated authority under Section 373.441, F.S.—also may rely on this Opinion to satisfy their ESA Section 7 obligations, again, to the extent the authorized activities are consistent with the PDCs and the other requirements of the Opinion. Unless the USACE has negotiated with the entity otherwise, the delegated authority must submit information on its authorized projects under the project-specific review procedures set forth in Section 2.3 to both NMFS and the USACE, but NMFS will coordinate directly with the USACE on any issues raised during the project-specific review. The USACE remains responsible for the programmatic review, as set forth in Section 2.4.

The USACE also may share responsibility for authorizing the activities addressed in this Opinion with another federal action agency. When 2 or more federal agencies are involved in authorizing, funding, or carrying out an activity that may affect listed species or critical habitat, a single agency may be designated as the lead for consultation under Section 7 of the ESA, per the factors in 50 CFR 402.07, which include the "time sequence in which the agencies would become involved, the magnitude of their respective involvement, and their relative expertise with respect to the environmental effects of the action."<sup>5</sup> If a federal action agency other than the USACE has been designated the lead action agency for the Section 7 consultation on a project that fits within 1 or more of the activities covered under this Opinion, that agency may rely on this Opinion in meeting its ESA Section 7 requirements, as long as the USACE is part of the

<sup>&</sup>lt;sup>5</sup> 50 CFR § 402.07 requires the lead agency to provide written notification of the designation to NMFS.

consultation and the project under consultation meets the PDCs and all other requirements of the Opinion, as described in the examples below. The USACE remains responsible for meeting the project-specific review and programmatic review requirements, according to the procedures outlined in Section 2.3 and 2.4 of this Opinion. Since the non-USACE federal action agency may be aware of a proposed project first, we suggest the non-USACE federal action agency contact the USACE early in the process to ensure that both federal action agencies have the necessary information from the applicant to complete both the ESA Section 7 requirements and their own permitting, funding, or other requirements.

For example, the following projects may require ESA Section 7 consultation coordination between the USACE and another federal action agency:

- The National Park Service (NPS) may wish to install a new aids-to-navigation (ATON). In addition to requiring authorization from USACE under Section 10 of the RHA, the NPS would need authorization from the U.S. Coast Guard (USCG). Both federal authorizations (USACE and USCG) would require Section 7 consultation and either the USCG or the USACE could be the lead action agency for the consultation. If the USCG was the lead action agency, and the USACE was a co-consulting action agency, the USCG could use this Opinion in satisfying its ESA Section 7 requirements; as long as the ATON met the general and activity-specific PDCs of this Opinion (ATONs fall within Category 2, below).
- The Federal Emergency Management Agency (FEMA) or the National Oceanic and Atmospheric Administration (NOAA) Restoration Division may request ESA Section 7 consultation on its action to fund 1 of the 10 categories of in-water work covered under this Opinion. As long as FEMA or NOAA Restoration Center's action has a nexus to the Opinion—i.e., FEMA or NOAA Restoration is funding a project for which the applicant also must apply for a USACE permit for 1 of the 10 categories of activities covered under this Opinion—if FEMA or NOAA Restoration Center is designated as the lead federal action agency for Section 7 consultation, and the USACE was a co-consulting action agency, FEMA or NOAA Restoration Center may rely on the Opinion.

In the federal coordination examples above, the USACE remains responsible for meeting the project-specific review and programmatic review requirements, according to the procedures outlined in Section 2.3 and 2.4 of this Opinion. Throughout the Opinion, we use the term USACE to refer to the USACE or any entity that is taking the lead for or otherwise responsible for the Section 7 consultation as described above, unless otherwise noted.

This Opinion may not be used to consult on 1 portion of a larger project. In order to rely on this Programmatic Opinion to meet the ESA Section 7 consultation requirements, all of the activities within the project must be covered under this Opinion and must meet the PDCs of the Opinion, unless all aspects of the project have been considered and are covered under this Opinion and other programmatic opinions. For example, the USACE could rely on this Opinion for its decision to issue a permit allowing for construction of a dock and seawall and for minor dredging if all of the general PDCs and all of PDCs for those specific activities (the docks, seawalls, and minor dredging) were met. However, if 1 activity did not meet the PDCs, then the USACE would have to consult on the entire project with NMFS separately via an individual consultation.

For example, if the seawall did not meet PDC A.1.1.1 because it was greater than 500 ft in length, then the USACE would have to request an individual consultation that covers all components of the project. The USACE could not rely on this Opinion to consult on the minor dredging and the dock, and separately consult on the seawall. However, if all aspects of the project are covered in 2 or more programmatic opinions, the USACE may rely on those opinions and need not seek individual consultation on the project. NMFS expects this Opinion may be used in combination with the upcoming Programmatic Opinion on the Effects of Research, Restoration, and Relocation on Threatened Caribbean Corals (referred to as the 3Rs Programmatic Biological Opinion, NMFS tracking number SER-2016-18298). For example, if a project covered under this Opinion had coral onsite that was removed prior to starting construction, the coral relocation would be covered under the 3Rs Programmatic Biological Opinion the project would be covered under this Opinion. In this instance, the USACE would have to meet the project specific and programmatic review requirements of each opinion, but would not be required to separately consult on the coral relocation project as a whole.

In addition, if the project under consultation is broader than the covered activities, the USACE (or other entity that can rely on this Opinion) cannot rely on this Opinion and should consult on the broader project individually. For example, if, in the same grant, FEMA was proposing to fund both repairs to previously installed shoreline stabilization materials requiring USACE authorization (which is covered under Activity 1) and repairs to an adjacent fishing pier requiring USACE authorization (which is not covered under the Opinion), then either FEMA or the USACE, depending on which agency was the lead for Section 7 consultation, should consult on the entire project. Neither FEMA nor the USACE could rely on the Opinion with respect to Activity 1 and limit individual consultation to repairing the fishing pier. The one exception to this limitation pertains to the in-water disposal of material dredge via dredging that meets the requirements of this Opinion (Activity 3). If the USACE has separately consulted with NMFS on the in-water disposal, the USACE can rely on this Opinion to satisfy its consultation requirements associated with authorizing the dredging (Activity 3). If the project proposes to dispose of the material at an in-water disposal location not previously consulted on by NMFS, then the entire project (both dredging and disposal) should be consulted on separately. This exception is explained in more detail in Activity 3 (PDC A3.3).

The examples above might involve more than one federal authorization or federal action. NMFS has limited reliance on this Opinion to facilitate a comprehensive review of a given project.

#### 2.1 **Project Action Area**

The action area is defined by regulation as "all areas to be affected directly or indirectly by the federal action and not merely the immediate area involved in the action" (50 CFR 402.02). Indirect effects are those that are caused by the proposed action that occur later in time, but still are reasonably certain to occur. The Opinion applies to 10 categories of in-water activities that the USACE's Jacksonville District authorizes in Florida and the U.S. Caribbean (Puerto Rico and the U.S. Virgin Islands, including St. Croix, St. John, and St. Thomas). In this Opinion, the largest area where indirect effects are expected is the area associated with vessel traffic originating from docks and marinas (Activity 2) and boat ramps (Activity 6). The distance these vessels travel is related to the preferred destination for fishing, sailing, sightseeing, diving an

artificial reef, or other recreational activity. The distance and seasonal timing of trips also is variable by the vessel size and type, vessel origination point, and is seasonally variable. We believe vessels may travel out to 35 nautical miles (nmi) from shore, based on a recent study that documented recreational vessels traveling up to 35 nmi offshore in the northeast Florida to southeast Georgia region (Montes et al. 2016). Larger vessels such as ferries, tankers, and barges, which are excluded by PDC A2.1.1 in Section 2.2.2, may travel further offshore, but as these vessels are not included in the Opinion, we believe it is reasonable to limit the action area to the area within 35 nmi. Therefore, we will consider the action area to include waters within the Florida and the U.S. Caribbean, where the USACE authorizes the covered activities and NMFS manages species and critical habitat, including waters up to 35 nmi from shore.

## 2.1.1 Species Specific Restriction and Exclusion Zones within the Action Area

The Opinion covers 10 categories of actions occurring in the state of Florida and the U.S. Caribbean. To protect ESA resources, the Opinion establishes specific areas where additional restrictions apply (restriction zones) and areas where an otherwise eligible activity may not occur if seeking coverage under the Opinion (exclusion zones). These areas have specific value for ESA-listed resources, as described below. Maps of all exclusion and restriction zones described in this section are currently being developed as Geographic Information System (GIS) layers and will be provided to the USACE and will be available for download on our website upon completion at: <a href="http://sero.nmfs.noaa.gov/maps\_gis\_data/index.html">http://sero.nmfs.noaa.gov/maps\_gis\_data/index.html</a>.

# 2.1.1.1 Smalltooth Sawfish Critical Habitat Limited Exclusion Zones

This Opinion excludes many activities occurring in smalltooth sawfish critical habitat limited exclusion zones. As defined in the activity-specific PDCs in Section 2.2, shoreline stabilization (Activity 1), pile-supported structures (Activity 2), dredging (Activity 3), water-management outfall structures (Activity 4), boat ramps (Activity 6), aquatic enhancement (Activity 7), transmission and utility lines (Activity 8), and temporary platforms, fill, and cofferdams (Activity 10) do not qualify for coverage under this Opinion if they occur in areas identified as limited exclusion zones within smalltooth sawfish critical habitat (Table 1 and Figure 1). Because certain activities can take place in these areas (e.g., temporary placement of scientific survey devices [Activity 5] and marine debris removal [Activity 9]), we will refer to these areas as limited exclusion zones. The smalltooth sawfish critical habitat limited exclusion zones are based on studies that have identified certain areas as supporting higher levels of smalltooth sawfish pupping. Excluding activities occurring in these areas from coverage under this Opinion means that those projects will be subject to individual ESA Section 7 consultation and is aimed at protecting reproductive female smalltooth sawfish during pupping (see Section 3.2.2). The limited exclusion zones identified in this Opinion are based on current data. If we or the USACE determine that additional areas need protection or if the areas defined below require modification, we will discuss the necessary changes with the USACE at the programmatic review meetings between us and the USACE (see Section 2.4), and any revisions will be included in this Opinion by addendum.

Name	Latitude	Longitude
U.S. 41 Bridges (the area between the fol	lowing coordinates)	
U.S. 41 (northwest corner)	26.660413°N	81.885243°W
U.S. 41 (northeast corner)	26.666827°N	81.872966°W
U.S. 41 (southwest corner)	26.642991°N	81.873880°W
U.S. 41 (southeast corner)	26.649405°N	81.861605°W
Iona Cove (the area between the followin	g coordinates)	
Iona Cove (northwest corner)	26.521437°N	81.991586°W
Iona Cove (northeast corner)	26.521212°N	81.976191°W
Iona Cove (southwest corner)	26.511762°N	81.991762°W
Iona Cove (southeast corner)	26.511537°N	81.976368°W
Glover Bight (the area between the follow	wing coordinates)	
Glover Bight (northwest corner)	26.542971°N	81.997791°W
Glover Bight (northeast corner)	26.542678°N	81.977745°W
Glover Bight (southwest corner)	26.529478°N	81.998035°W
Glover Bight (southeast corner)	26.529185°N	81.977992°W
Cape Coral (the area between the follow	ing coordinates)	
Cape Coral (point 1)	26.551662°N	81.947412°W
Cape Coral (point 2)	26.551561°N	81.940683°W
Cape Coral (point 3)	26.539075°N	81.940916°W
Cape Coral (point 4)	26.539205°N	81.951049°W
Cape Coral (point 5)	26.542181°N	81.951047°W
Cape Coral (point 6)	26.542133°N	81.947776°W

Table 1. Limited Exclusion Zones in Smalltooth Sawfish Critical Habitat



Figure 1. Smalltooth sawfish limited exclusion zones.

## 2.1.1.2 Gulf Sturgeon Critical Habitat Migratory Restriction Zones

We have identified specific zones at the mouth of Gulf sturgeon spawning rivers and narrow inlets where additional restrictions apply to protect sturgeon that are migrating to and from spawning rivers (referred to as the "Gulf sturgeon critical habitat migratory restriction zones"). These restrictions limit noise generated during pile and sheet pile installation associated with shoreline stabilization (Activity 1), pile-supported structures (Activity 2), water-management outfall structures (Activity 4), boat ramps (Activity 6), aquatic enhancement (Activity 7), transmission and utility lines (Activity 8), and temporary platforms, fill, and cofferdams (Activity 10) activities in Gulf sturgeon critical habitat migratory restriction zones. They also limit placement of materials that cover large areas of sediment and could block movement through these narrow passes, including the placement of living shoreline, oyster reef, and artificial reef materials (Activity 7) and temporary platforms, fill, and cofferdams (Activity 10). In Florida, Gulf sturgeon spawning rivers include the Escambia River, Blackwater and Yellow Rivers, Choctawhatchee River, Apalachicola River, and Suwannee River (Figure 2). Limitations also apply to narrow inlets in Gulf sturgeon critical habitat, including narrow inlets in Apalachicola Bay (Indian Pass and Government Cut) and Destin Pass in Choctawhatchee Bay. Projects occurring in the estuarine Gulf sturgeon critical habitat units are under NMFS's jurisdiction, while the riverine critical habitat areas are under the jurisdiction of the USFWS. This delineation between the Services occurs at the mouth of each river. The mouth is defined as river kilometer 0/ river mile 0. Although the interface of fresh and saltwater, referred to as the saltwater wedge, occurs within the lower-most reach of a river, for ease in delineating critical habitat units, we are defining the boundary between the riverine and estuarine units as river kilometer 0/ river mile 0 (50 CFR 226.214; 68 FR 13370).

The following restrictions apply to projects proposed in the zones identified in Table 2. As is explained below, certain projects occurring in these zones are excluded from the Opinion and require separate consultation. We developed these restrictions and exclusions because in-water construction noise in these areas can discourage Gulf sturgeon from returning to spawning rivers from the open ocean and materials placed in these areas can block migration.

# PDCs Specific to the Gulf Sturgeon Critical Habitat Migratory Restriction Zones

- This Opinion does not apply to the placement of living shoreline, oyster reef, and artificial reef materials (Activity 7, PDC A7.26) and temporary platform, fill, and cofferdams (Activity 10, PDC A10.11) in Gulf sturgeon critical habitat migratory restriction zones.
- This Opinion does not apply to new transmission and utility line installation in the Gulf sturgeon critical habitat migratory restriction zones between September and March, when sturgeon are likely to be present in these areas. Emergency repair/replacement of transmission and utility lines may occur in these areas during this time frame if the work is conducted without the use of heavy in-water equipment (e.g., dredging equipment) (Activity 8, PDC A8.10).
- This Opinion does not apply to the installation of metal piles and metal sheet piles by impact hammer in the areas identified as Gulf sturgeon critical habitat migratory restriction zones.
- The allowable pile and sheet pile driving activities vary depending on the width of the project action area, as described below.
- <u>Areas that are 0-500 ft wide</u>: In areas up to 500 ft wide, the allowable pile or sheet pile driving activities within the Gulf sturgeon critical habitat migratory restriction zones are:
  - 1. Creating a pilot hole for any type of pile using an auger or drop punch
  - 2. Trenching a shoreline with mechanical equipment to create a space to install any type of sheet pile and backfilling behind it
  - 3. Installing any type of piles and sheet piles by jetting.
- <u>Areas that are 501-1,400 ft wide</u>: In areas over 500 ft wide, but less than 1,400 ft wide, the allowable pile or sheet pile driving activities within the Gulf sturgeon critical habitat migratory restriction zones are:

The activities described in 1-3 above, and

- 4. Installing any type of piles and sheet piles by vibratory hammer.
- <u>Areas over 1,401 ft wide</u>: In areas 1,401 ft wide or wider, the allowable pile or sheet pile driving activities within the Gulf sturgeon critical habitat migratory restriction zones are:

The activities described in 1-4 above, and

5. Installing wood, vinyl, and concrete piles and sheet piles by impact hammer.

If additional measures or areas are deemed necessary for protection, or if the areas defined below require modification, we will discuss these changes with the USACE at the programmatic review meetings (see Section 2.4), and any revisions will be included in this Opinion by addendum.

Water Body	Delineation Type	Point A	Point B	Point C	Point D
Ecoembia Divor		30.5146361°N	30.5323916°N8		
Escambia River		87.16093°W	7.13192°W		
Blackwater/	Line	30.5047°N	30.5047°N		
Yellow Rivers		87.0475°W	87.0196583°W		
Choctawhatchee	Line	30.385183°N	30.3814861°N		
Bay		86.515394°W	86.50684°W		
Choctawhatchee	Line	30.429794°N	30.37842°N		
River		86.147725°W	86.1252°W		
Apalachicola	Polygon	29.675561°N	29.6751°N	29.681216°N	29.684875°N
Bay		85.240283°W	85.2160583°W	85.2160583°W	85.221502°W
Apalachicola	Polygon	29.6308694°N	29.6223194°N	29.6267861°N	29.63268°N
Bay		85.1060027°W	85.097038°W	85.093172°W	85.09687°W
Apalachicola	Polygon	29.611361°N	29.611872°N	29.61736°N	29.6161583°N
Bay		84.958483°W	84.957338°W	84.95926°W	84.9626638°W
Apalachicola	Polygon	29.765272°N	29.77816°N	29.78695°N	29.7721°N
Bay		84.6916361°W	84.6669027°W	84.674269°W	84.695294°W
Apalachicola	Polygon	29.7131027°N	29.7120916°N	29.734772°N	29.731505°N
River		84.99772°W	84.9744472°W	84.9701027°W	84.9846027°W
Sumanaa Divar	Line	29.328483°N	29.291116°N		
Suwallee River		83.167525°W	83.1669694°W		
Sumanaa Divar	Line	29.291116°N,	29.2670194°N		
Suwallee Kiver		83.1669694°W	83.0946805°W		

 Table 2. Gulf sturgeon critical habitat migratory restriction zones

Lines (Points A and B) create a line marking the approximate mouth of the river. Projects on the marine side of the mouth of these rivers (i.e., areas under NMFS jurisdiction) must follow the migratory restrictions defined in this section.

Polygons (Points A-D) create an area between the points marking restricted sections of a bay or pass. Projects in these defined areas must follow the migratory restriction requirements defined in this section.



Figure 2. Gulf sturgeon critical habitat. Estuarine critical habitat units under NMFS's jurisdiction shaded and the rivers under USFWS's jurisdiction labeled (Images provided on the NMFS website at http://sero nmfs.noaa.gov/maps\_gis\_data/protected\_resources/critical\_habitat/index html).

# 2.1.1.3 Atlantic Sturgeon Critical Habitat Exclusion Zone (St. Marys River)

This Opinion does not cover activities proposed in the St. Marys River, which has been designated as Atlantic sturgeon critical habitat, as shown in Figure 3 below as SA7. Atlantic sturgeon riverine critical habitat is under NMFS's jurisdiction. The only Atlantic sturgeon critical habitat area occurring in within the action area is within the Florida portion of the St. Marys River (i.e., generally along the south side of the river). NMFS and the USACE jointly decided to exclude projects occurring in St. Marys River from coverage under this Opinion. The USACE does not expect to receive many applications for proposed projects in St. Marys River and will consult on those projects on an individual basis, as warranted.



Figure 3. Atlantic sturgeon critical habitat (Images provided on the NMFS website at http://sero nmfs.noaa.gov/protected resources/sturgeon/documents/critical habitat maps.pdf).

#### 2.1.1.4 North Atlantic Right Whales Educational Sign Zones

North Atlantic right whales are susceptible to vessel collisions given the time they spend at the surface. NMFS designated critical habitat for the North Atlantic right whale off the east coast of Florida to protect North Atlantic right whale calving grounds. NMFS recently expanded the critical habitat designation to include the area from Cape Fear, North Carolina south to approximately 27 nmi south of Cape Canaveral, Florida (81 FR 4837, January 27, 2016). To avoid and minimize the risk of vessel strikes with mothers and calves, especially during the known right whale core calving season of December through March, educational signs must be placed at public locations such as marinas, multi-family docking facilities, and boat ramps that are located within 11 nmi of an inlet that leads to areas within the known range of North Atlantic right whales (PDC A2.2 and PDC A6.3.2). Homeowners proposing to construct, repair, or replace a private dock within 11 nmi of an inlet that leads to areas within the known range of North Atlantic right whale, will be provided a handout with their USACE permit describing the presence of North Atlantic right whales in the area and the Federal regulations governing the approach to North Atlantic right whales (50 CFR 224.103(c)), (PDC A2.4 and Appendix C). The coordinates for the center points of these 11 nmi radii are provided in Table 3 below and images of the areas are shown in Figure 4. These zones are the North Atlantic right whale educational sign zones.

The 11 nmi radius was chosen based on a recent study on Offshore Recreational Boating Characterization in the Southeast U.S (Sea Grant 2016). In this study, 958 vessel owners responded to a questionnaire regarding vessel use. Based on the study results, we believe that the average recreational vessel owner is willing to travel 11 nmi to reach an inlet leading to open ocean, and thus we are requiring educational signs where the inlet is within 11 nmi of areas where the vessels might encounter North Atlantic right whales (i.e., within the known range for the species). Once recreational vessels reach open water, the questionnaire respondents indicated that they would be willing to travel up to 35 nmi offshore (as discussed in Section 2.1 defining the limits of the action area).

Speed restrictions already apply in specific locations, primarily at key port entrances, and in certain times in seasonal management areas. The restrictions apply to all vessels 65 ft and greater in length (73 FR 60173, October 10, 2008), including those that may be stored at structures authorized under this Opinion. NMFS also has already established a Dynamic Management Area program whereby vessels are requested, but not required, to either travel at 10 knots or less or route around locations when certain aggregations of right whales are detected outside seasonal management areas. NMFS also established regulations restricting approaches within 500 yards (460 m) of a right whale, whether by vessel of any size, aircraft or other means, to reduce disturbance and the potential for vessel interaction (62 FR 6729). The educational signs required by PDC A2.2 and PDC A6.3 (and described in the discussion on Activity 2 and Activity 6) are provided as additional educational outreach for vessel operators that transit the same general geographic region as the existing speed restriction and management areas.

Name	Latitude	Longitude
Cumberland Sound	30.719564°N	81.449467°W
Nassau Sound	30.516611°N	81.444278°W
St. John's River	30.408053°N	81.399467°W
St Augustine Inlet	29.918411°N	81.288117°W
Matanzas Inlet	29.713831°N	81.227000°W
Ponce Inlet	29.083056°N	80.916494°W
Port Canaveral	28.409306°N	80.586689°W
Sebastian Inlet	27.860833°N	80.446725°W
Fort Pierce Inlet	27.471711°N	80.290378°W
St. Lucie Inlet	27.165567°N	80.157236°W
Jupiter Inlet	26.943950°N	80.070908°W
Riviera Beach	26.772353°N	80.034508°W

Table 3. North Atlantic Right Whale Educational Sign Zone



Figure 4. North Atlantic right whale educational sign zones.

#### 2.1.1.5 U.S. Caribbean Sea Turtle Critical Habitat Restriction Zones

In sea turtle critical habitat in the U.S. Caribbean (Figure 5) (i.e., critical habitat for hawksbill, leatherback, and the North Atlantic distinct population segment of green sea turtles (NA DPS), this Opinion only applies to certain activities. This Opinion does not apply to watermanagement outfall structures (Activity 4, PDC A4.7) and aquatic enhancement (Activity 7, PDC A7.31) occurring in critical habitat for hawksbill, leatherback, and the NA DPS of green sea turtles. Within these areas, the Opinion is limited to the removal, repair, and replacement of existing structures including shoreline stabilization projects (Activity 1, PDC A1.12), boat ramps (Activity 6, PDC A6.11), and transmission and utility lines (Activity 8, PDC A8.11). Pile-supported structures (Activity 2), dredging (Activity 3), scientific survey devices (Activity 5), marine debris removal (Activity 9), and temporary cofferdams (Activity 10) can occur in these areas without limitation.

Sea turtle critical habitat within the U.S. Caribbean encompasses relatively small areas around specific islands or small areas of an island. For example, critical habitat for the NA DPS of green sea turtles encompasses just Culebra Island and the surrounding smaller islands,<sup>6</sup> many of which are under the management of the Culebra National Wildlife Refuge. Hawksbill sea turtle critical habitat includes only Mona Island, and the entire island is under the management of Mona Island Nature Preserve; therefore, common residential projects will not occur in this location. Leatherback sea turtle critical habitat includes only the southwest section of St. Croix Island. As stated above, this Opinion only applies to certain projects occurring in sea turtle critical habitat in the U.S. Caribbean due to the limited spatial size of these critical habitat units and the unique resources in these areas, including nesting beaches. The USACE does not expect to receive many applications for proposed projects in sea turtle critical habitat in the U.S. Caribbean compared to the number of projects addressed in other areas and will consult on those projects not covered under this Opinion separately, as necessary.

<sup>&</sup>lt;sup>6</sup> On April 6, 2016, NMFS published a final rule listing 11 DPSs of the green sea turtle, including the NA DPS. 81 FR 20058; April 6, 2016. NMFS will issue a rule designating critical habitat for the DPSs in a future rulemaking. In the interim, the existing critical habitat designation (i.e., waters surrounding Culebra Island, Puerto Rico; 63 FR 46693; Sept. 2, 1998) remains in effect for the NA DPS. 81 FR 20058; April 6, 2016.



Figure 5. Sea turtle critical habitat in the U.S. Caribbean. NA DPS of green sea turtle critical habitat (left image), hawksbill sea turtle critical habitat (center image), and leatherback sea turtle critical habitat (right image).

# 2.1.1.6 Bryde's Whale Exclusion Zone

According to the proposed listing rule for Bryde's whale (81 FR 88639, December 8, 2016), sightings of Bryde's whales in the Gulf of Mexico have been concentrated in the De Soto Canyon area, along the continental shelf break between 100 m and 300 m depth. Bryde's whales have been sighted in all seasons within the De Soto Canyon area (Maze-Foley and Mullin 2006; MMIQT 2015; Mullin 2007; Mullin and Hoggard 2000). Consequently, LaBrecque et al. (2015) considered this area, home to the small resident population of Bryde's whale in the northeastern Gulf of Mexico, as a Biologically Important Area (Figure 6). Biologically Important Areas are reproductive areas, feeding areas, migratory corridors, and areas in which small and resident populations are concentrated. Classifying an area as a Biologically Important Area does not have direct or immediate regulatory consequences. Rather, these classifications are intended to provide the best available science to help inform regulatory and management decisions, in order to minimize impacts from anthropogenic activities on marine mammals (LaBrecque et al. 2015). Due to the lack of information available on Bryde's whale and the proposed listing of the species as endangered, this Opinion excludes all activities occurring in the Bryde's whale exclusion zone, defined as the De Soto Canyon at depths from 100-300 m. The USACE will conference or consult on projects in this area on an individual basis, as necessary.



Figure 6. Bryde's whale exclusion zone.

## 2.1.2 USACE Effect Determinations for Listed Species and Critical Habitat

Based on an evaluation of the 10 categories of activities occurring within the action area, the USACE made the following determinations about the listed species and critical habitat that may be affected by each category of activity (Table 4). The effects determinations in Table 4 represent the USACE's final effects determinations provided for activities in Florida. The USACE initially provided different determinations with its request for consultation, but revised those determinations after completing the PDCs and re-evaluating the effects to species and critical habitat in light of the PDCs.

The USACE later provided effects determinations for the U.S. Caribbean in a request for consultation dated November 14, 2016. In that letter, the USACE stated that the activities on which they were seeking consultation may affect, but are not likely to adversely affect the following species and critical habitat in the U.S. Caribbean: sea turtles (green, Kemp's ridley, leatherback, loggerhead, hawksbill), Nassau grouper, coral (elkhorn, staghorn, mountainous star, lobed star, rough cactus, pillar), whales (blue, fin, humpback, sei, sperm); and critical habitat for sea turtles (green, hawksbill, leatherback, loggerhead) and coral (elkhorn and staghorn). The following species and critical habitat designations have changed since the request for consultation for this Opinion:

• The humpback whales that may occur within the action area are no longer listed under the ESA. On September 8, 2016, we revised the humpback whale listing to identify 14 distinct population segments (DPSs), 1 of which we determined is threatened, 4 of which we determined are endangered, and 9 of which did not warrant listing (81 FR 62259, Publication September 8, 2016). The whales found in the action area belong to the West Indies DPSs, which we found does not warrant listing, and therefore humpback whales are not considered

in this Opinion.

- Nassau Grouper, which occurs within the action area, was listed as threatened (81 FR 42268, Publication July 29, 2016) and is therefore included in this Opinion. The USACE sought consultation on the effect of the projects occurring in the U.S. Caribbean on Nassau grouper. Because the species' range includes portions of southern Florida, we are consulting on the effects to the species in both Florida and the U.S. Caribbean.
- Atlantic sturgeon critical habitat was proposed in June 2016 (81 FR 36077, June 3, 2016) and finalized in August 2017 (82 FR 39160, August 17, 2017). The action area includes areas designated as critical habitat, and therefore we are consulting on Atlantic sturgeon critical habitat in this Opinion.
- Bryde's whale, which occurs within the action area, was proposed for listing as an endangered species (81 FR 88639, December 8, 2016) and is therefore included in this Opinion.<sup>7</sup>
- In our listing decision, we explained that the Central and Southwest Atlantic DPSs of scalloped hammerhead shark were thought to occur within the U.S. Caribbean (79 FR 38213, July 3, 2014). In designating critical habitat for this species (80 FR 71774, November 17, 2015), however, we determined that there is no evidence that the scalloped hammerhead shark was, or is, present within the U.S. Caribbean. Since that time, in connection with our consultation on fishery independent monitoring in the U.S. Caribbean, we learned that the species may be present in the U.S. Caribbean, based on a single recorded interaction during hook-and-line fishery monitoring (SER-2009-07541). We have continued to collect information on the species presence in the U.S. Caribbean and are aware of a few additional interactions. Therefore, we believe the species may be present in the U.S. Caribbean and are aware of a few additional interaction area and are consulting on potential effects to the species in that area. The Central and Southwest Atlantic DPS for scalloped hammerhead shark is found in waters of the U.S. Caribbean, among other areas, but is not found in waters off of Florida. Therefore, we do not consult on this species in Florida.

 $<sup>^{7}</sup>$  NMFS has proposed to list Bryde's whale as endangered (81 FR 88639, December 8, 2016). Although not required by Section 7(a)(4) of the ESA, we are including a conference consultation on the effects of this action on Bryde's whale. This conference consultation will become the final consultation for this species if it is listed as proposed and if there are no changes to this proposed action that cause effects to Bryde's whale that were not considered in this conference consultation.

	Activity	Swimming sea turtles	Smalltooth sawfish	Gulf, shortnose, and Atlantic sturgeon	Johnson's seagrass	<i>Acropora</i> corals	Boulder star, mountainous star, lobed star, rough cactus, and pillar coral	North Atlantic right whale	North Atlantic right whale critical habitat	Smalltooth sawfish critical habitat	Gulf sturgeon critical habitat	Johnson's seagrass critical habitat	Loggerhead sea turtle critical habitat	<i>Acropora</i> critical habitat
1	Shoreline stabilization	NLAA	NLAA	NLAA	NE	NE	NE	NE	NE	LAA	NLAA	LAA	NE	NE
2	Pile-supported structures	NLAA	NLAA	NLAA	LAA	NE	NE	NLAA	NE	LAA	NLAA	LAA	NE	NE
3	Dredging	NLAA	NLAA	NLAA	LAA	NE	NE	NE	NE	NLAA	NLAA	LAA	NE	NE
4	Water-management outfall structures	NLAA	NLAA	NLAA	NE	NE	NE	NE	NE	LAA	NLAA	LAA	NE	NE
5	Scientific survey devices	NLAA	NLAA	NLAA	NE	NE	NE	NE	NE	LAA	NLAA	LAA	NE	LAA
6	Boat ramps	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NE	NLAA	LAA	NE	NE	NE
7	Aquatic enhancement	NLAA	NLAA	NLAA	NE	NE	NE	NE	NE	NE	NE	NE	NE	NE
8	Transmission/ utility lines	NLAA	NLAA	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	LAA	LAA	NE	NE
9	Marine debris removal	NE	NE	NE	NE	NE	NE	NE	NE	NE	NE	NE	NE	NE
10	Temporary platforms, fill, and cofferdams	NLAA	NLAA	NLAA	NLAA	NE	NE	NE	NE	LAA	LAA	NE	NE	NE
NE =	no effect; NLAA = may a	affect, not l	ikely to adve	ersely affect;	LAA = ma	y affect, lik	ely to adverse	ly affect	1			1		

 Table 4. USACE Project Effects Determinations for Activities Occurring in Florida

### 2.1.3 Listed Species and Critical Habitat Occurring within the Action Area

NMFS has determined that the following species (Table 5) and critical habitat (Table 6) occur within the action area and may be affected by activities analyzed under this Opinion.

Species	ESA Listing Status	Present in Florida?	Present in U.S. Caribbean?
Sea Turtles			
Green (NA DPS and South Atlantic DPS)	Т	Р	Р
Kemp's ridley	E	Р	NP
Leatherback	E	Р	Р
Loggerhead (Northwest Atlantic Ocean DPS [NWA DPS])	Т	Р	Р
Hawksbill	Е	Р	Р
Fish			
Smalltooth sawfish (U.S. DPS)	E	Р	NP
Gulf sturgeon (Atlantic sturgeon, Gulf subspecies)	Т	Р	NP
Shortnose sturgeon	Е	Р	NP
Atlantic sturgeon (All DPSs)	T/E <sup>8</sup>	Р	NP
Nassau Grouper	Т	Р	Р
Scalloped hammerhead (Central and Southwest Atlantic DPS)	Т	NP <sup>9</sup>	Р
Invertebrates and Marine P	lants		
Elkhorn coral (Acropora palmata)	Т	Р	Р
Staghorn coral (Acropora cervicornis)	Т	Р	Р
Boulder star coral (Orbicella franksi)	Т	Р	Р
Mountainous star coral (Orbicella faveolata)	Т	Р	Р
Lobed star coral (Orbicella annularis)	Т	Р	Р
Rough cactus coral (Mycetophyllia ferox)	T	Р	Р
Pillar coral (Dendrogyra cylindrus)	Т	Р	Р
Johnson's seagrass	Т	Р	NP

Table 5. Listed Species NMFS Believes Are Likely to Occur in or near the Action Area

<sup>&</sup>lt;sup>8</sup> Activities occurring within river and inshore habitats in the action area may affect Atlantic sturgeon from the South Atlantic DPS; however, Atlantic sturgeon from all DPSs may be affected in offshore waters within the action area. The New York Bight, Chesapeake Bay, Carolina, and South Atlantic DPSs are listed as endangered; the Gulf of Maine DPS is listed as threatened. Activities covered under this Opinion cannot occur in-the St Marys River, but can occur in the St. John's River and inland areas where the South Atlantic DPS may be present.

<sup>&</sup>lt;sup>9</sup> The Central and Southwest Atlantic DPS of scalloped hammerhead shark applies to scalloped hammerhead sharks originating from the Central and Southwest Atlantic Ocean, and includes species found within the waters of the Caribbean Sea, including the U.S. Virgin Islands and Puerto Rico. We identified a population of scalloped hammerhead shark that occurs in waters off of Florida, the Northwest Atlantic and Gulf of Mexico DPS for scalloped hammerhead shark, however, we determined that this DPS did not warrant listing (78 FR 20717, April 5, 2013; 79 FR 38213, July 3, 2014).

Marine Mammals			
North Atlantic right whale	E	Р	NP
Blue whale	E	Р	Р
Fin whale	E	Р	Р
Sei whale	E	Р	Р
Sperm whale	E	Р	Р
Bryde's whale (proposed)	E	Р	NP
E = endangered; T = threatened, P = Present, N	P = Not Present	1	

Table 6.	Designated	<b>Critical Habi</b>	itat NMFS B	elieves is I	n or Near	the Action Area
THEN TO A	The state of the state of the	CALCENT AATON	TABLE T ITLET OF TO	ATTA LAN TO T	AL VI LIVIII	the Lietton Lin en

Species	Unit in Florida	Unit in U.S. Caribbean
Smalltooth sawfish	<ul> <li>Charlotte Harbor Estuary (CHEU)</li> <li>Ten Thousand Islands/ Everglades (TTIEU)</li> </ul>	N/A
Gulf sturgeon	Units 9-14 <sup>10</sup>	N/A
Loggerhead sea turtle (NWA DPS)	<ul> <li>Nearshore Reproductive Habitat: Units LOGG-N-14 to 32</li> <li>Breeding Habitat: Units LOGG-N-17, 19</li> <li>Migratory Habitat: Units LOGG-N-17, 18, 19</li> <li>Sargassum Habitat: Unit LOGG-S-01</li> </ul>	N/A
Green sea turtle (NA DPS)	N/A	Culebra Island
Hawksbill sea turtle	N/A	Mona and Monita Island
Leatherback sea turtle	N/A	St Croix Island
Staghorn and elkhorn coral	Area 1: Florida	<ul> <li>Area 2: Puerto Rico and Associated Islands</li> <li>Area 3: St, John/St. Thomas, U.S. Virgin Islands</li> <li>Area 4: St. Croix, U.S. Virgin Islands</li> </ul>
Johnson's seagrass	Units A-J	N/A
North Atlantic right whale	Unit 2	N/A
Atlantic sturgeon	South Atlantic Unit 7 <sup>11</sup>	N/A
N/A = Not applicable	e	

Table 7 (below) provides a complete list of the essential features/primary constituent elements (PCEs) of each critical habitat unit that occurs in Florida and the U.S. Caribbean. Note that the table below refers to both essential features and PCEs of critical habitat. This duality of terms is

<sup>&</sup>lt;sup>10</sup> Gulf sturgeon critical habitat is under the joint jurisdiction of the USFWS and NMFS, with the USFWS managing riverine habitat and NMFS managing estuarine and marine habitats. Units 9-14 are the only areas under NMFS's jurisdiction that are found in the action area.

<sup>&</sup>lt;sup>11</sup> The South Atlantic Unit 7 (St. Marys Unit) includes the St. Marys River in (1) Camden and Charlton Counties in Georgia and (2) Baker and Nassau Counties in Florida.

because the USFWS uses the term "PCE" and NMFS uses "essential features" when describing critical habitat. When we develop a critical habitat rule jointly with USFWS, the term PCE is often used. Recent amendments to the Services' joint regulations implementing the ESA, however, removed reference to "primary constituent elements" (81 FR 7414, Feb. 11, 2016). As we explained in the final rule, removing this phrase is not intended to substantively alter anything about the designation of critical habitat, but to eliminate redundancy in how we describe the physical or biological features. New critical habitat rules will describe physical biological features (PBFs) to help identify habitat essential to the conservation of the species. In this Opinion, we refer to the features as they were described in the rule designating that critical habitat. For example, the Gulf sturgeon critical habitat rule refers to PCEs, and thus we have used that term in the table below. Critical habitat boundary maps are available at http://sero.nmfs.noaa.gov/maps\_gis\_data/protected\_resources/critical\_habitat/index.html.

Table 7.	Essential	Features	/PCEs/PBF	's of Each	Critical	l Habitat	Unit in	Florida	and the
U.S. Cari	ibbean								

Smalltooth sawfish (74 FR 45353, Sept. 2, 2009)	The physical and biological features essential to the conservation of the U.S. DPS of smalltooth sawfish, which provide nursery area functions are: red mangroves and shallow euryhaline habitats characterized by water depths between the Mean High Water line and 3 ft (0.9 m) measured at Mean Lower Low Water (MLLW). These features are included in critical habitat within the boundaries of the specific areas in paragraph (b) of this section, except where the features were not physically accessible to sawfish at the time of this designation (September 2009); for example, areas where existing water control structures prevent sawfish passage to habitats beyond the structure.
Gulf sturgeon (68 FR 13370, March 19, 2003)	<ul> <li>Based on the best available information, there are 7 PCEs essential for the conservation of the Gulf sturgeon. Only the following 4 are under NMFS's jurisdiction:</li> <li>1. Abundant prey items within estuarine and marine habitats and substrates for juvenile, subadult, and adult life stages;</li> <li>2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages;</li> <li>3. Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages;</li> <li>4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats (e.g., a river unobstructed by any permanent structure, or a dammed river that still allows for passage).</li> </ul>

turtle (79 FR habitat as a portion of the nearshore waters adjacent to nesting 1	beaches
39855, July 10, that are used by hatchlings to egress to the open-water environr	nent as
2014) well as by nesting females to transit between beach and open w	ater
during the nesting season. The following PCEs support this had	oitat: (i)
Nearshore waters directly off the highest density nesting beach	es and
their adjacent beaches, as identified in 50 CFR 17.95(c), to 1.6	kilometer
(km) offshore; (ii) Waters sufficiently free of obstructions or ar	tificial
lighting to allow transit through the surf zone and outward toward	ard open
water; and (iii) Waters with minimal man-made structures that	could
promote predators (i.e., nearshore predator concentration cause	d by
submerged and emergent offshore structures), disrupt wave pat	terns
necessary for orientation, and/or create excessive longshore cur	rents.
2. Winter areas: Florida does not contain any winter areas.	
3. Breeding areas: the PBF of concentrated breeding habitat as the	se sites
with high densities of both male and female adult individuals d	uring the
breeding season. PCEs that support this habitat are the following	1g: (i)
High densities of reproductive male and female loggerheads: (ii	i)
Proximity to primary Florida migratory corridor: and (iii) Proxi	mitv to
Florida nesting grounds.	- J
4. Constricted migratory habitat: the PBF of constricted migratory	habitat
as high use migratory corridors that are constricted (limited in y	vidth) by
land on one side and the edge of the continental shelf and Gulf	Stream on
the other side. PCEs that support this habitat are the following:	(i)
Constricted continental shelf area relative to nearby continental	shelf
waters that concentrate migratory pathways; and (ii) Passage co	onditions
to allow for migration to and from nesting, breeding, and/or for	aging
areas.	00
5. Sargassum habitat: the PBF of loggerhead Sargassum habitat as	5
developmental and foraging habitat for young loggerheads whe	re surface
waters form accumulations of floating material, especially Sarg	assum.
PCEs that support this habitat are the following: (i) Convergence	e zones.
surface-water downwelling areas, the margins of major boundar	rv
currents (Gulf Stream), and other locations where there are con-	centrated
components of the <i>Sargassum</i> community in water temperature	s suitable
for the optimal growth of <i>Sargassum</i> and inhabitance of logger	neads; (ii)
Sargassum in concentrations that support adequate prev abunda	nce and
cover: (iii) Available prev and other material associated with Sa	irgassum
habitat including, but not limited to, plants and evanobacteria a	nd
animals native to the <i>Sargassum</i> community such as hydroids a	nd
copepods; and (iv) Sufficient water depth and proximity to avai	lable
currents to ensure offshore transport (out of the surf zone) and	foraging
and cover requirements by <i>Sargassum</i> for post-hatchling logger	heads.
i.e., > 10-m depth.	,

Acropora (Staghorn and elkhorn coral) (73 FR 72210, Nov. 26, 2008)	The physical feature essential to the conservation of elkhorn and staghorn corals is: substrate of suitable quality and availability to support larval settlement and recruitment, and reattachment and recruitment of asexual fragments. "Substrate of suitable quality and availability" is defined as natural consolidated hard substrate or dead coral skeleton that is free from fleshy or turf macroalgae cover and sediment cover.
Johnson's seagrass (65 FR 17786, April 5, 2000)	Based on the best available information, general physical and biological features of the critical habitat areas include adequate water quality, salinity levels, water transparency, and stable, unconsolidated sediments that are free from physical disturbance.
North Atlantic right whale (81 FR 4837, Jan. 27, 2016)	Critical habitat includes 2 areas (Units) located in the Gulf of Maine and Georges Bank Region (Unit 1) and off the coast of North Carolina, South Carolina, Georgia and Florida (Unit 2). Only Unit 2 occurs within the action area.
	<ul> <li>The physical features essential to the conservation of the North Atlantic right whale, which provide calving area functions in Unit 2, are:</li> <li>1. Sea surface conditions associated with Force 4 or less on the Beaufort Scale</li> <li>2. Sea surface temperatures of 7°C to 17°C</li> <li>3. Water depths of 20-92 ft (6- 28 m), where these features simultaneously co-occur over contiguous areas of at least 231 squared nautical miles (nmi<sup>2</sup>) of ocean waters during the months of November through April. When these features are available, they are selected by right whale cows and calves in dynamic combinations that are suitable for calving, nursing, and rearing, and which vary, within the ranges specified, depending on factors such as weather and age of the calves.</li> </ul>
Atlantic sturgeon (82 FR 39160, August 17, 2017)	<ul> <li>The physical features essential for the conservation of Atlantic sturgeon belonging to the Carolina and South Atlantic DPSs are those habitat components that support successful reproduction and recruitment. These are:</li> <li>1. Hard bottom substrate (e.g., rock, cobble, gravel, limestone, boulder, etc.) in low salinity waters (i.e., 0.0-0.5 parts per thousand range) for settlement of fertilized eggs and refuge, growth, and development of early life stages;</li> <li>2. Aquatic habitat inclusive of waters with a gradual downstream gradient of 0.5 up to as high as 30 parts per thousand and soft substrate (e.g., sand, mud) between the river mouth and spawning sites for juvenile foraging and physiological development;</li> <li>3. Water of appropriate depth and absent physical barriers to passage (e.g., locks, dams, thermal plumes, turbidity, sound, reservoirs, gear, etc.) between the river mouth and spawning sites necessary to support: <ul> <li>(i) Unimpeded movement of adults to and from spawning sites;</li> <li>(ii) Seasonal and physiologically dependent movement of juvenile</li> </ul> </li> </ul>

	<ul> <li>Atlantic sturgeon to appropriate salinity zones within the river estuary; and</li> <li>(iii) Staging, resting, or holding of subadults or spawning condition adults. Water depths in main river channels must also be deep enough (at least 1.2 meters) to ensure continuous flow in the main channel at all times when any sturgeon life stage would be in the river;</li> <li>4. Water quality conditions, especially in the bottom meter of the water column, with temperature and oxygen values that support: <ul> <li>(i) Spawning;</li> <li>(ii) Annual and inter-annual adult, subadult, larval, and juvenile survival; and</li> <li>(iii) Larval, juvenile, and subadult growth, development, and recruitment. Appropriate temperature and oxygen values will vary interdependently, and depending on salinity in a particular habitat. For example, 6.0 mg/L dissolved oxygen or greater likely supports juvenile rearing habitat, whereas dissolved oxygen less than 5.0 mg/L for longer than 30 days is less likely to support rearing when water temperature is greater than 25°C. In temperatures greater than 26°C, dissolved oxygen greater than 4.3 mg/L is needed to protect survival and growth. Temperatures of 13 to 26 °C likely support spawning habitat.</li> </ul></li></ul>
Green sea turtle (63 FR 46693, Sept. 2,1998)	Critical habitat for the green sea turtle is designated in the waters surrounding the island of Culebra, Puerto Rico, from the mean high water line (MHWL) seaward to 3 nmi. These waters include Culebra's outlying Keys, including Cayo Norte, Cayo Ballena, Cayos Geniquí, Isla Culebrita, Arrecife Culebrita, Cayo de Luís Peña, Las Hermanas, El Mono, Cayo Lobo, Cayo Lobito, Cayo Botijuela, Alcarraza, Los Gemelos, and Piedra Steven. At the time of designation, essential features to critical habitat were not precisely defined; however, the critical habitat was designated to provide protection for important developmental and resting habitats. Juvenile and adult green sea turtles depend on seagrasses as the principal dietary component for foraging. In addition, coral reefs and other topographic features within the waters around Culebra Island and surrounding islands and cays provide green turtles with shelter during interforaging periods that serve as refuge from predators. On April 6, 2016, NMFS published a final rule listing 11 DPSs of the green sea turtle, including the NA DPS. 81 FR 20058; April 6, 2016. NMFS will issue a rule designating critical habitat for the DPSs in a future rulemaking. In the interim, the existing critical habitat designation described herein remains in effect for the NA DPS of green sea turtles.
Hawksbill sea turtles (63 FR 46693,	Critical habitat for the hawksbill sea turtle has been designated in the waters surrounding the islands of Mona and Monito, Puerto Rico, from the MHWL seaward to 3 nmi. At the time of designation, essential features to critical

Sept. 2, 1998)	habitat were not precisely defined; however, the critical habitat was designated to provide protection for important developmental and resting habitats. Hawksbill sea turtles depend on sponges as their principal dietary component and healthy coral reefs for foraging and shelter habitats.
Leatherback sea turtles (44 FR 8491, March 23, 1979)	Critical habitat for the leatherback sea turtle has been designated in the waters adjacent to Sandy Point on the southwest corner of St. Croix, U.S. Virgin Islands, in waters from the 100-fathom curve shoreward to the level of mean high tide, with boundaries at 17°42′12"N and 64°50′00″W. At the time of designation, essential features to critical habitat were not precisely defined; however, critical habitat for leatherback sea turtles was designated to provide protection to sea turtles using these waters for courting, breeding, and as access to and from nesting areas on Sandy Point Beach, St. Croix, U.S. Virgin Islands.

# 2.2 Activities Analyzed, Project Design Criteria, and Potential Routes of Effect

In this section of the Opinion, we describe the categories of activities under consultation, the PDCs that each activity must meet to be covered under this Opinion, and the expected effects of each category of activities on ESA-listed species and designated critical habitat. In particular, for each category of activity covered by this Opinion, we will provide the following information:

- 1. <u>Activity Description</u>: A general description of how the activity typically is implemented with sample photos and drawings. We are providing a general overview of the typical implementation for context; the installation materials, methods, and locations are limited by the PDCs.
- 2. <u>PDCs</u>: A description of the non-discretionary PDCs applicable to all projects covered under this Opinion. The general PDCs ensure that the covered activities meet certain thresholds designed to avoid or minimize impacts on ESA-listed species and critical habitat.

In addition to the general PDCs, each of the 10 categories of covered activities is subject to additional activity-specific PDCs. Like the general PDCs, activity-specific PDCs are non-discretionary requirements for coverage under the Opinion that avoid or minimize the potential effects of permitted activities on ESA-listed species and designated critical habitat.

All PDCs were developed based on information from the USACE's past permitting practices and review of consultations on USACE-authorized in-water construction activities in Florida and the U.S. Caribbean. The activity-specific PDCs are typical of measures used to protect ESA listed species and designated critical habitat and are substantially similar to the PDCs that NMFS included in other programmatic consultations with the USACE in the last 5 years including the SWPBO, 12 SAJ General Permit Programmatic, SAJ-42, SAJ-82, and SPGP IV-R1.

In addition, PDCs designed to avoid or minimize effects on critical habitat are provided at the end of each category of activity when additional protections, beyond the general and activity-specific PDCs, are required to avoid or minimize effects on a particular critical habitat unit.
Note: The critical habitat-protective PDCs are in addition to those applicable to the category of activity and, in some cases, supersede all other applicable PDCs, for example, if they are in conflict (i.e., more restrictive) than those otherwise applicable to that category of activity.

3. <u>Assumptions</u>: A description of the assumptions that the USACE used to estimate how each category of activity would affect ESA-listed species and designated critical habitat.

The USACE provided a series of assumptions regarding the location, number, magnitude, and other pertinent facts about the covered activities that could be used to estimate the effects of those activities on ESA-listed species or designated critical habitat. For example, the USACE estimated the number of projects expected to occur in critical habitat units as an activity's anticipated effects may vary if the activity occurs within a particular critical habitat unit. In developing these assumptions, the USACE looked to its past permits and previous analyses of how the permitted activities affected ESA-listed species and designated critical habitat.

The USACE also estimated the number of projects within each of the 10 categories of activities covered under this Opinion that it expects to authorize per 5-year period in both Florida and the U.S. Caribbean, based on number of similar projects that it authorized in the past in reliance on prior programmatic Opinions (and, for a small subset as applicable, authorizations associated with non-programmatic consultations). The USACE then applied an increase for Florida based on forecasts of population growth (assuming that this will result in authorizations for new development), and based on its assumption that it would receive increasing requests to improve and repair infrastructure as a result of past population growth. The USACE's estimates recognize the cyclic nature of when it will receive permit applications, and assume that economic recovery will continue. The USACE applied a similar increase for the U.S. Caribbean despite current economic conditions because of variable nature of development in that area and the great potential for increasing development. This may overestimate the amount of expected development in the U.S. Caribbean, but will ensure that we have evaluated the potential worst-case (highest development) scenario.

4. <u>Potential Routes of Effect</u>: A description of the ways in which each category of activity may potentially affect ESA-listed species and designated critical habitat.

In the sections entitled Potential Routes of Effect, we analyze the potential routes of effect from each category of activity covered under this Opinion on each of the listed species (Table 5) and critical habitat units (Table 6) likely to occur in or near the action area (see Section 2.1). For each category of activity, we provide the rationale for our effects analysis in a stepwise approach based on each species and/or critical habitat that may be affected. Activities that we believe are likely to adversely affect species or critical habitats listed in Tables 5 and 6 are discussed further in Section 3 and throughout the remainder of the Opinion.

	Category of Activity	Smalltooth sawfish critical habitat	Gulf sturgeon critical habitat	Johnson's seagrass critical habitat	<i>Acropora</i> critical habitat	North Atlantic right whale critical habitat	Loggerhead sea turtle critical habitat	Green sea turtle critical habitat	Hawksbill sea turtle critical habitat	Leatherback sea turtle critical habitat	Total outside of critical habitat	Total (in and outside of) critical habitat
1	Shoreline stabilization	1,012	165	306	116	0	135	4	0	2	5,363	7,102
2	Pile- supported structures	4,125	724	491	301	85	503	76	5	7	27,257	33,574
3	Dredging	150	51	31	64	7	124	2	0	2	890	1,320
4	Water- management outfall structures	19	24	7	28	0	50	0	0	0	2	129
5	Scientific survey devices	2	2	17	7	2	2	4	1	0	52	89
6	Boat ramps	22	30	2	145	7	138	4	1	1	359	708
7	Aquatic enhancement	46	91	7	68	0	7	0	0	0	48	267
8	Transmission/ utility lines	9	28	12	12	0	33	5	0	2	187	288
9	Marine debris removal	11	7	11	8	2	2	4	1	3	6	54
10	Temporary platforms, access fill, and cofferdams	31	12	12	1	0	3	4	0	2	51	116
	Total	5,426	1,134	894	749	102	997	103	8	19	34,215	43,646

 Table 8. Projected Number of Authorizations Occurring in Certain Critical Habitat Units

 per Activity for 5-year Period

# General PDCs Applicable to All Projects:

AP.1. The applicant must agree to adhere to PDCs for In-Water Activities (provided below).

- **AP.2.** All projects involving the installation of piles or sheet piles shall follow the PDCs for *In*-*Water Noise from Pile and Sheet Pile Installation* (Section 2.2). This Opinion does not cover projects that use seismic surveys, low frequency sonar, explosions, and seismic air guns.
- **AP.3.** All projects proposed in or near areas with mangroves, seagrasses, corals, or hard bottom habitat must refer to PDCs for *Mangroves, Seagrasses, Corals, and Hard Bottom for All Projects* (provided below) to determine whether the project is covered under the Opinion and, if it is covered, to ensure it is sited, designated, and implemented following all of the PDCs in that section.
- **AP.4.** For every project, the USACE must determine if the project is located within:
  - Smalltooth sawfish critical habitat limited exclusion zones (Section 2.1.1.1)
  - Gulf sturgeon critical habitat migratory restriction zones (Section 2.1.1.2)
  - Atlantic sturgeon critical habitat exclusion zone (St. Marys River) (Section 2.1.1.3)
  - North Atlantic right whale educational sign zones (Section 2.1.1.4)
  - U.S. Caribbean sea turtle critical habitat restriction zones (Section 2.1.1.5)
  - Bryde's whale exclusion zone (Section 2.1.1.6)

Where the activity is excluded from the Opinion within a particular zone, the application must be processed under a separate consultation. Where additional restrictions apply to activities within that zone, the USACE or other authorizing entity must ensure that the project meets the requirements for that zone.

**AP.5.** This Opinion only covers new construction (i.e., installation, repair, replacement) and does not apply to after-the-fact consultations or enforcement actions handled by the USACE.

**AP.6.** All activities must be completed during daylight hours.

# PDCs for In-Water Activities

For an activity to be covered under this Opinion, the USACE authorization must include the following conditions. Failure to comply with these conditions could result in enforcement action by the USACE and/or NMFS.

**AP.7.** <u>Education and Observation</u>: The permittee must ensure that all personnel associated with the project are instructed about the potential presence of species protected under the ESA and the Marine Mammal Protection Act (MMPA). All on-site project personnel are responsible for observing water-related activities for the presence of protected species. All personnel shall be advised that there are civil and criminal penalties for harming, harassing, or killing ESA-listed species or marine mammals. To determine which species may be found in the project area, please review the relevant Protected Species List at:

 $http://sero.nmfs.noaa.gov/protected\_resources/section\_7/threatened\_endangered/index.html$ 

- AP.8. <u>Reporting</u> of interactions with protected species:
  - a) Any collision(s) with and/or injury to any sea turtle, sawfish, whale, or sturgeon occurring during the construction of a project, shall be reported immediately to NMFS's Protected Resources Division (PRD) at (1-727-824-5312) or by email to takereport.nmfsser@noaa.gov and SAJ-RD-Enforcement@usace.army.mil.
  - b) Smalltooth sawfish: Report sightings to 1-844-SAWFISH or email Sawfish@MyFWC.com
  - c) Sturgeon: Report dead sturgeon to 1-844-STURG 911 (1-844-788-7491) or email nmfs.ser.sturgeonnetwork@noaa.gov
  - d) Sea turtles and marine mammals: Report stranded, injured, or dead animals to 1-877-WHALE HELP (1-877-942-5343).
  - e) North Atlantic right whale: Report injured, dead, or entangled right whales to the USCG via VHF Channel 16.
- **AP.9.** <u>Vessel Traffic and Construction Equipment</u>: All vessel operators must watch for and avoid collision with species protected under the ESA and MMPA. Vessel operators must avoid potential interactions with protected species and operate in accordance with the following protective measures:
  - a) Construction Equipment:
    - i) All vessels associated with the construction project shall operate at "Idle Speed/No Wake" at all times while operating in water depths where the draft of the vessel provides less than a 4-foot (ft) clearance from the bottom, and in all depths after a protected species has been observed in and has departed the area.
    - ii) All vessels will follow marked channels and/or routes using the maximum water depth whenever possible.
    - iii) Operation of any mechanical construction equipment, including vessels, shall cease immediately if a listed species is observed within a 50-ft radius of construction equipment and shall not resume until the species has departed the area of its own volition.

- iv) If the detection of species is not possible during certain weather conditions (e.g., fog, rain, wind), then in-water operations will cease until weather conditions improve and detection is again feasible.
- b) All Vessels:
  - i) Sea turtles: Maintain a minimum distance of 150 ft.
  - ii) North Atlantic right whale: Maintain a minimum 1,500-ft distance (500 yards).
  - iii) Vessels 65 ft in length or longer must comply with the Right Whale Ship Strike Reduction Rule (50 CFR 224.105) which includes reducing speeds to 10 knots or less in Seasonal Management Areas (http://www.fisheries.noaa.gov/pr/shipstrike/).
  - iv) Mariners shall check various communication media for general information regarding avoiding ship strikes and specific information regarding right whale sightings in the area. These include NOAA weather radio, USCG NAVTEX broadcasts, and Notices to Mariners.
  - v) Marine mammals (i.e., dolphins, whales [other than North Atlantic right whales], and porpoises): Maintain a minimum distance of 300 ft.
  - vi) When these animals are sighted while the vessel is underway (e.g., bow-riding), attempt to remain parallel to the animal's course. Avoid excessive speed or abrupt changes in direction until they have left the area.
  - vii) Reduce speed to 10 knots or less when mother/calf pairs or groups of marine mammals are observed, when safety permits.
- **AP.10.** <u>**Turbidity Control Measures during Construction:**</u> Turbidity must be monitored and controlled. Prior to initiating any of the work covered under this Opinion, the Permittee shall install turbidity curtains as described below. In some instances, the use of turbidity curtains may be waived by the USACE project manager if the project is deemed too minimal to generate turbidity (e.g., certain ATON installation, scientific survey device placement, marine debris removal) or if the current is too strong for the curtains to stay in place. Turbidity curtains specifications:</u>
  - a) Install floating turbidity barriers with weighted skirts that extend to within 1 ft of the bottom around all work areas that are in, or adjacent to, surface waters.
  - b) Use these turbidity barriers throughout construction to control erosion and siltation and ensure that turbidity levels within the project area do not exceed background conditions.
  - c) Position turbidity barriers in a way that does not block species' entry to or exit from designated critical habitat.
  - d) Monitor and maintain turbidity barriers in place until the authorized work has been completed and the water quality in the project area has returned to background conditions.
  - e) In the range of ESA-listed corals (St. Lucie Inlet, Martin County south to the Dry Tortugas and the U.S. Caribbean) and Johnson's seagrass (Turkey Creek/Palm Bay south to central Biscayne Bay in the lagoon systems on the east coast of Florida):
    - Projects that include upland earth moving (e.g., grading to install a building or parking lot associated with a dock and seawall project), must install sediment control barriers to prevent any upland sediments from reaching estuarine or marine waters.
    - The turbidity curtain requirement cannot be waived for any project that moves or removes sediment (e.g., dredging, auger to create a pile, trenching to install a cable

line). If turbidity curtains are not feasible in an area based on site conditions such as water current, high wave action, or stormy conditions, the project must undergo individual Section 7 consultation and is not covered under this Programmatic Opinion.

- **AP.11.** <u>Entanglement:</u> All turbidity curtains and other in-water equipment must be properly secured with materials that reduce the risk of entanglement of marine species (described below). Turbidity curtains likewise must be made of materials that reduce the risk of entanglement of marine species.
  - a) In-water lines (rope, chain, and cable, including the lines to secure turbidity curtains) must be stiff, taut, and non-looping. Examples of such lines are heavy metal chains or heavy cables that do not readily loop and tangle. Flexible in-water lines, such as nylon rope or any lines that could loop or tangle, must be enclosed in a plastic or rubber sleeve/tube to add rigidity and prevent the line from looping and tangling. In all instances, no excess line is allowed in the water.
  - b) Turbidity curtains and other in-water equipment must be placed in a manner that does not entrap species within the construction area or block access for them to navigate around the construction area.

# PDCs for Mangroves, Seagrasses, Corals and Hard Bottom for All Projects

Note: **For projects authorized in reliance on this Opinion only**, the PDCs below supercede any other guidance documents otherwise applicable to reduce or avoid impacts to mangroves, seagrasses, and corals. This includes the NMFS's *Construction Guidelines in Florida for Minor Piling-Supported Structures Constructed in or over Submerged Aquatic Vegetation, Marsh, or Mangrove Habitat* dated August 2001, and NMFS's *Key for Construction Conditions for Docks or Other Minor Structures Constructed in or over Johnson's Seagrass (Halophila johnsonii)*, dated October 2002. NMFS may still apply these guidance documents in other consultations, including consultations on Essential Fish Habitat under the Magnuson-Stevens Fishery Conservation and Management Act, as appropriate.

# AP.12. Mangroves

- To qualify for coverage under this Opinion, all projects must be sited and designed to avoid or minimize impacts to mangroves.
- Mangrove removal must be conducted in a manner that avoids any unnecessary removal and is limited to the following instances:
  - Removal to install up to a 4-ft-wide walkway for a dock.
  - Removal to install up to an 8-ft-wide walkway for public docks, where the walkway is necessary to address compliance with the Americans with Disability Act (ADA).
  - Removal to install culverts necessary to improve water quality or restore hydrology between 2 water bodies. Such mangrove removal is limited to a maximum of 20 linear feet (lin ft) of shoreline per culvert opening.
  - Removal of mangroves above mean high water (MHW) provided that the tree does not have any prop roots that extend into the water below the MHWL.

- Mangrove Trimming. Mangrove trimming is regulated by FDEP, Puerto Rico Department of Natural and Environmental Resources, and U.S. Virgin Islands Department of Planning and Natural Resources. Consistent with those authorities, when used in this Opinion, mangrove trimming refers to the removal (using hand equipment such as chain saws and/or machetes) of lateral branches (i.e., no alteration of the trunk of the tree) in a manner that ensures survival of the tree. This Opinion does not limit or supersede any restrictions on mangrove removal required under any federal, state, or local law.
  - This Opinion only covers projects with associated mangrove trimming occurring waterward of MHW if such trimming (1) occurs within the area where the authorized structures are placed or will be placed (e.g., removal of branches that overhang a dock), (2) is necessary to provide temporary construction access, and (3) is conducted in a manner that avoids any unnecessary trimming.
  - The Opinion does not apply to projects proposing to remove red mangrove props roots waterward of MHW, except for removal to install the dock walkways, as described above (up to a 4-ft walkway and up to a 8-ft ADA compliant walkway) and to install culverts necessary to improve water quality or restore hydrology between 2 water bodies.

# AP.13. Seagrass:

• Pile-supported structures must follow the PDCs for *Docks or Other Minor Structures* (PDC A2.17, Section 2.2.2)

# Johnson's seagrass:

- This Opinion does not apply to projects where Johnson's seagrass is found within the project footprint except for:
  - Installation of pile-supported structures that meet the PDCs for *Docks or Other Minor Structures* (PDC A2.17, Section 2.2.2).
  - Maintenance dredging of previously authorized areas. This is limited to the removal of no more than 0.1 acre (ac) (4,356 ft<sup>2</sup>) of Johnson's seagrass per year (Activity 3; see Section 2.2.3)
  - Transmission/utility line repairs within the same footprint of the lines being repaired (Activity 8; see Section 2.2.8).

# Non-listed seagrasses:

- All impacts to non-ESA listed native, non-invasive seagrasses should be avoided and minimized to the extent practicable.
- This Opinion does not apply to projects located within the geographic boundary of U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat identified in Section 2.1.1.5) if non-ESA listed, native, non-invasive seagrasses are found within the project footprint.

# AP.14. Coral and Hard Bottom Habitat

- This Opinion does not apply to projects that may affect, directly or indirectly, ESA-listed corals.
- Projects occurring within in the Florida Keys National Marine Sanctuary (FKNMS) may require separate consultation or authorization from NOAA's FKNMS. Projects authorized to occur in the FKNMS shall comply with any measures NOAA FKNMS has developed to avoid, minimize, and/or mitigate any effects on non-listed corals. For projects occurring outside of the FKNMS, if non-listed corals are found within the project footprint, we recommend relocating all non-listed corals, when possible, in a manner that is protective of the corals.
- This Opinion does not apply to projects where hard bottom habitat is found within the project footprint, except for the temporary placement (up to 24 months) of scientific survey devices (Activity 5) that have a footprint of less than 1 square foot (ft<sup>2</sup>) per device and are installed in a manner that does not permanently alter the hardbottom (e.g., the devices are not installed by drilling). For this Opinion, we define hard bottom in 2 ways:
  - Natural consolidated hard substrate that is suitable to support corals, coral larval settlement, reattachment and recruitment of asexual coral fragments. These areas of hard bottom or dead coral skeleton must be free from fleshy or turf macroalgae cover and sediment cover.
  - Nearshore and surf-zone, low-profile hard bottom outcroppings (e.g., worm-rock reef [sabellariid worm reefs] and eolianite, granodiorite). This habitat can be persistent or ephemeral, cycling through periods of exposure and cover by sand. The range of this hard bottom habitat extends along the southeastern coast of Florida from Cape Canaveral to Miami-Dade County and in the U.S. Caribbean. It is an important developmental habitat for juvenile hawksbill and green sea turtles, which use it for both foraging and refuge.

Given implementation of the general PDCs provided above, which apply to each of the 10 categories of activities covered under the Opinion, and some of the project-specific PDCs below, we are able to analyze how the covered activities may affect certain ESA-listed species and designated critical habitat. Since this analysis is completed up front, these species and critical habitats are not analyzed further in the Opinion unless stated otherwise.

## Determinations by Species for Specific Categories of Activities

Based on our knowledge of the species presence (or absence) within the action area and the general PDCs (as well as some activity-specific PDCs discussed later in the Opinion), we believe the following ESA-listed species either will not be affected or may be affected, but are not likely to be adversely affected by some or all of the activities analyzed under this Opinion.

# Corals

ESA-listed corals (elkhorn, staghorn, boulder star, mountainous star, lobed star, rough cactus, and pillar corals) may be found within the action area, from St. Lucie Inlet, Martin County south to the Dry Tortugas and in the U.S. Caribbean. In general, the types of activities evaluated in the

Opinion could affect ESA-listed corals through physical damage or removal of coral, or from turbidity, sedimentation, and changes in water quality. However, this Opinion does not apply to projects that could directly or indirectly affect ESA-listed corals (PDC AP.14). Therefore, if a project may affect ESA listed corals, a separate ESA consultation would be required. We note that the Opinion does not apply to dredging projects that penetrate or remove underlying hard substrate (e.g., bedrock, hardbottom) in *Acropora* critical habitat using any methods including blasting or fracturing since those methods may result in increased fine sediment that does not settle easily and could result in increased damage to corals (PDC A3.7).

#### Johnson's seagrass

Three activities covered under this Opinion may affect Johnson's seagrass through removal, burial, shading, or reduced water quality. In particular, we believe Johnson's seagrass may be adversely affected by: (1) installation of pile-supported structures (Activity 2); (2) maintenance dredging of previously authorized locations (Activity 3); and (3) utility line repairs (Activity 8). Effects to Johnson's seagrass will be evaluated in each of the sections describing those categories of activities.

We believe the remaining 7 categories of covered activities (Activities 1, 4, 5, 6, 7, 9, and 10) will not affect Johnson's seagrass by covering or removing the species since PDC AP.13 excludes projects from this Opinion if Johnson's seagrass occurs within the project footprint. Therefore, we believe none of these 7 categories of activities will remove or shade Johnson's seagrass.

We believe that turbidity and sedimentation generated during in-water construction associated with Activities 1, 4, 6, and 10 may affect, but is not likely to adversely affect Johnson's seagrass near the project but outside the project footprint by burying or covering it in sedimentation. The effects of such turbidity and sedimentation are discussed below in the turbidity section. For this reason, we note these activities are "not likely to adversely affect" Johnson's seagrass in the table below. Activities 5, 7, and 9 do not generate turbidity. Therefore, we believe these activities will not affect the species via turbidity or otherwise given PDC AP.13.

In addition, changes in water quality from water discharged from water-management outfall structures (Activity 4) may affect Johnson's seagrass. Specific PDCs associated with Activity 4 (Section 2.2.4) apply in Johnson's seagrass critical habitat for the installation or repair of water-management outfall structures. For example:

- PDC A4.3 states that all outfall structures covered under this Opinion must be designed and implemented to prevent erosion and scouring so that these outfall structures do not generate sedimentation and turbidity through erosion after the structure is placed and operational.
- PDC A4.2 limits coverage under this Opinion to activities associated with outfall structures when the effluent from the outfall is authorized, conditionally authorized, specifically exempted, or in compliance with regulations issued under the National Pollutant Discharge Elimination System Program (CWA Section 402). Since seagrasses are sensitive to pollution in the water, this PDC requires project compliance with water quality regulations for the water discharged from the outfall structures to limit pollution and make it extremely unlikely that corals will be affected by pollution from these outfalls.

• In Johnson's seagrass critical habitat, additional restrictions apply to the water discharged from the water-management outfall structure. Specifically, the PDC requires that the outfall also contain an in-line treatment structure to reduce water velocities, sedimentation, nutrients, and pollutants discharged from the outfall structure into marine waters. These treatment structures may include nutrient baffle structures, filters, natural bio filters, and low impact development such as infiltration basins and trenches or vegetative swales. The additional measures in PDC A4.3 reduce the risk of sedimentation and pollutants entering the marine environment from the water discharged from water-management outfall structures in areas that could support Johnson's seagrass (i.e., in Johnson's seagrass critical habitat).

Based on the PDC restrictions to the water discharged from water-management outfall structures, we believe that the risk of sedimentation, turbidity, and pollution harming Johnson's seagrass in or around new outfall structures is extremely low and the effect is discountable.

Finally, restoration of an area to support seagrasses, including filling blow holes and leveling sediments to the surrounding elevation (Activity 7), and marine debris removal (Activity 9) may have a beneficial effect on Johnson's seagrass by providing additional area for the species to recruit to. These benefits are not certain, and depend on the proximity of the species to the restored area.

Based on the restrictions in the general PDCs described above, as well as some activity-specific PDCs associated with Activity 4, we have made the following effects determinations, as summarized below in Table 9.

	Activity	NMFS Determination
1	Shoreline stabilization	NLAA
2	Pile-supported structures	LAA
3	Dredging	LAA
4	Water-management outfall structures	NLAA
5	Scientific survey devices	NE
6	Boat ramps	NLAA
7	Aquatic enhancement	NE
8	Transmission/ utility lines	LAA
9	Marine debris removal	NE
10	Temporary platforms, fill, and cofferdams	NLAA

 Table 9. NMFS Effects Determinations for Johnson's Seagrass

### Nassau Grouper

We believe the current functional range of Nassau Grouper is limited to the Florida Keys and the U.S. Caribbean. Below, we discuss the potential effects to Nassau grouper based on activities in each of these areas.

<u>In the U.S. Caribbean</u>: We expect all life stages of Nassau grouper may be present throughout Puerto Rico and the U.S. Virgin Islands. Nassau grouper spawn in specific aggregation sites in in the Caribbean. They start out as planktonic larvae that transition from oceanic open water

environments to bottom dwelling juveniles living in nearshore areas. The preferred habitat of juvenile Nassau grouper is believed to be macroalgal clumps (primarily *Laurencia*), seagrass beds, and corals (particularly *Porites* spp.), though they have also been found in mangrove lined lagoons or creeks and in several microhabitats, including empty queen conch shells, debris adjacent to seagrass, and rubble mounds. As Nassau grouper transition to larger juveniles and adults, they shift to nearshore reef habitats and then progressively move further offshore from fore reef habitat to deeper water reefs, crevices, caves, and ledges up to 130 m deep.

In order to determine the effects to Nassau grouper from the 10 categories of activities covered under this Opinion, we must consider the life stage of these fish and the habitat preferences based on sizes discussed above. We believe the activities analyzed in this Opinion that could affect small juvenile Nassau Grouper in the U.S. Caribbean are those activities that would place materials on or remove macroalgae (Activities 1-10), seagrasses (all activities except Activity 9), and mangroves (Activities 2 and 4). We believe these effects will be insignificant since Nassau grouper are unspecialized feeders with a wide array of foraging and refuge options and because they are mobile species that will be able to move away from these areas during construction and forage or seek refuge in other surrounding areas. Specifically,

- Macroalgae and seagrasses are prevalent foraging resources found throughout the U.S Caribbean. Macroalgae is not protected under the ESA or this Opinion and may be affected by the activities under this Opinion. However, the small losses of macroalgae from nearshore projects covered under this Opinion are not expected to significantly diminish the availability of this foraging habitat for Nassau Grouper, and thus we expect any effect on the species to be insignificant. All of the covered activities (except for marine debris removal under Activity 9) may affect seagrasses. However, PDC AP.13 limits seagrass impacts by recommending that all projects be positioned to avoid and minimize impacts to non-ESA listed, native, non-invasive seagrasses to the extent practicable. Based on this PDC, we do not expect seagrass resources to be substantially diminished, and we do not expect any impacts to these resources to have a measureable effect on the species. Thus, we expect any impacts to both microalgae and seagrasses to have an insignificant effect on the species.
- Pile-supported structures (Activity 2) and water-management outfall structures (Activity 4) may affect mangroves. However, under PDC AP.12, all projects must be sited and designed to avoid or minimize impacts to mangroves, and all removal of mangroves shall be conducted in a manner that avoids any unnecessary removal. The Opinion does not cover activities involving the removal of red mangrove props roots below MHW, except for removal to install the dock walkways defined above (4-ft walkway and 8-ft ADA compliant walkway) and to install culverts necessary to improve water quality or restore hydrology between 2 water bodies. The small losses of mangroves from nearshore projects covered under this Opinion are not expected to significantly diminish the availability of this refuge habitat for Nassau Grouper, and we do expect any impacts to this resource to have a measureable effect on the species, and any effect will be insignificant.

As small juvenile Nassau grouper transition to the larger juveniles and adult size class, their habitat preference shifts to primarily hardbottom, coral reefs, or ledges that offer foraging and

refuge habitat. These larger fish may also use nearshore available crevices for sheltering, including rock rubble habitat or crevices created in or around manmade structures such as docks and seawalls. As a result, the repair, replacement, or removal of shoreline stabilization projects (Activity 1), pile-supported structures (Activity 2), and marine debris removal (Activity 9) could affect sheltering habitat for larger juvenile and adult Nassau grouper. We believe these effects will be insignificant as we expect the species will move to nearby similar habitat including the extensive reefs habitat in the Caribbean and nearby structures such as seawalls and docks. Also, PDC AP.14 limits this Opinion to projects that do not directly or indirectly affect ESA-listed corals, and excludes projects if non-listed corals or hardbottom are found in the project footprint. Therefore, we do not expect any loss of reef or hardbottom refuge habitat used by large juvenile and adult Nassau grouper. The installation of living shorelines and artificial reefs (Activity 7) may result in a beneficial effect to large juveniles or adult Nassau grouper by providing additional refuge habitat in the U.S. Caribbean.

In the Florida Keys and Miami Dade County: We believe the current functional range is limited to the Florida Keys south of Government Cut on the north side of the Keys and up to the southern boundary of Everglades National Park on the south side of the Keys. The general absence of Nassau grouper outside of the Florida Keys is well documented by the lack of records in Florida Fish and Wildlife Conservation Commission's, Fisheries Independent Monitoring data as well as various surveys conducted by the NMFS's Southeast Fisheries Science Center. Adult and large juvenile Nassau grouper may be present in waters off of the Florida Keys and areas surrounding the Dry Tortugas, although available data suggests occurrence at low density. We do not expect any small juveniles in the Florida Keys and therefore do not expect Nassau grouper in the Keys to utilize macroalgae, seagrasses, or mangrove habitat. The larger juvenile and adult Nassau grouper in the Florida Keys may use hardbottom, reefs, ledges, rock rubble, and even existing docks and seawalls for refuge habitat if these areas provide adequate crevices for the species to hide. As we explained when discussing the potential effects on large juvenile and adult Nassau grouper in the U.S. Caribbean, we believe that the repair, replacement, or removal of shoreline stabilization projects (Activity 1), pile-supported structures (Activity 2), and marine debris removal (Activity 9) could affect sheltering habitat. We believe these effects will be insignificant as we expect Nassau grouper will move to nearby similar habitat, including the extensive reefs habitat found in the Florida Keys and nearby structures such as seawalls and dock found throughout this area. Also, PDC AP.14 limits this Opinion to projects that do not directly or indirectly affect ESA-listed corals, and excludes projects if non-listed corals or hardbottom are found in the project footprint. Therefore, we do not expect any loss of reef or hardbottom refuge habitat used by adult Nassau grouper. The installation of living shorelines and artificial reefs (Activity 7) may result in a beneficial effect to large juveniles or adult Nassau grouper by providing additional refuge habitat in the Florida Keys.

In addition to the effects described above, we believe mobile species, including Nassau grouper, may be affected by a number of what we are referring to as "common routes of effect," which are routes of effect common to all of the activities described in this Opinion. These routes of effects and our analysis of whether and how they affect ESA-listed species and/or critical habitat are described in a section below, entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion*. Nassau grouper may be affected, but is not likely to be

adversely affected by the following common routes of effect (the number corresponds to the section number in which the effect is discussed below):

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 5. Limiting Species' Movement and Access to Habitat
- 6. Vessel Strikes
- 7. Noise

Based on the restrictions in the general PDCs described above, and our determinations regarding the common routes of effect, we have made the following determinations regarding the effects of the activities on Nassau grouper, as summarized below in Table 10.

	Activity	NMFS Determination in Florida north of Government Cut, and north of the southern boundary of Everglades National Park	NMFS Determination in the Florida Keys south of Government Cut on the north side of the Keys and up to the southern boundary of Everglades National Park on the south side of the Keys (small juveniles)	NMFS Determination in the Florida Keys south of Government Cut on the north side of the Keys and up to the southern boundary of Everglades National Park on the south side of the Keys (large juveniles and adults)	NMFS Determination in the U.S. Caribbean (all life stages)
1	Shoreline stabilization	NP	NP	NLAA	NLAA
2	Pile-supported structures	NP	NP	NLAA	NLAA
3	Dredging	NP	NP	NLAA	NLAA
4	Water- management outfall structures	NP	NP	NLAA	NLAA
5	Scientific survey devices	NP	NP	NLAA	NLAA
6	Boat ramps	NP	NP	NLAA	NLAA
7	Aquatic enhancement	NP	NP	NLAA	NLAA
8	Transmission/ utility lines	NP	NP	NLAA	NLAA
9	Marine debris removal	NP	NP	NLAA	NLAA
10	Temporary platforms, access fill, and cofferdams	NP	NP	NLAA	NLAA

 Table 10. NMFS Effects Determinations for Nassau Grouper

#### **Scalloped Hammerhead**

At the time we listed the Central and Southwest Atlantic DPS of scalloped hammerhead shark, it was thought to occur within the U.S. Caribbean (79 FR 38213, July 3, 2014). In designating critical habitat for this species (80 FR 71774, November 17, 2015), we determined that there is no evidence that the scalloped hammerhead shark was present within the U.S. Caribbean. Since that time, in connection with our consultation on fishery independent monitoring in the U.S. Caribbean, we learned that the species may be present in the U.S. Caribbean, based on a recorded interaction during hook-and-line fishery monitoring SER-2009-7541). We have continued to collect information on the species presence in the U.S. Caribbean and are aware of a few additional interactions. The Central and Southwest Atlantic DPS for scalloped hammerhead

shark is found in waters of the U.S. Caribbean, among other areas, but is not found in waters off of Florida. Therefore, we do not consult on this species in Florida. We identified a population of scalloped hammerhead shark that occurs in waters off of Florida, the Northwest Atlantic and Gulf of Mexico DPS for scalloped hammerhead shark, however, we determined that this DPS did not warrant listing (78 FR 20717, April 5, 2013; 79 FR 38213, July 3, 2014).

Scalloped hammerhead shark are highly mobile and partly migratory, and are likely the most abundant of the hammerhead species (Maguire et al. 2006). Although scalloped hammerhead sharks are highly mobile, this species rarely crosses entire oceans (Diemer et al. 2011; Duncan and Holland 2006; Kohler and Turner 2001). The median distance between mark and recapture of 3,278 tagged adults along the eastern U.S. was less than 65 miles (100 km) (Kohler and Turner 2001). Tagging studies reveal the tendency for scalloped hammerhead shark to aggregate around and travel to and from core areas within locations (Bessudo et al. 2011; Duncan and Holland 2006; Hearn et al. 2010; Holland et al. 1993).

Both juvenile and adult scalloped hammerhead sharks occur as solitary individuals, pairs, or in schools (Compagno 1984). Adult aggregations are most common offshore over seamounts and near islands (Bessudo et al. 2011; CITES 2010; Compagno 1984; Hearn et al. 2010). Neonate and juvenile aggregations are more common in nearshore shallow water nursery habitats (Bejarano-Álvarez et al. 2011; Diemer et al. 2011; Duncan and Holland 2006).

The scalloped hammerhead shark is a high trophic level predator (Cortés 1999) and an opportunistic feeder with a diet that includes a wide variety of mobile species such as bony fish, octopi, cuttlefish, squid, crabs, lobsters, and rays (Bush 2003; Compagno 1984; Júnior et al. 2009; Noriega et al. 2011). In the listing rule, we identified commercial fishing as one of the primary threats to the Central and SW Atlantic DPS of scalloped hammerhead shark, and found it was unlikely that loss of habitat was contributing to the species' extinction risk (79 FR 38213). Therefore, we believe that any effects to offshore and nearshore potential foraging habitat caused by the minor construction projects analyzed in this Opinion will have an insignificant effect on the species.

We believe a mobile species, such as the Central and Southwest Atlantic DPS of scalloped hammerhead shark, may be affected by a number of what we are referring to as "common routes of effect," which are routes of effect common to all of the activities described in this Opinion. These routes of effects and our analysis of whether and how they affect ESA-listed species and/or critical habitat are described in a section below, entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion*. The Central and Southwest Atlantic DPS of scalloped hammerhead shark may be affected, but is not likely to be adversely affected by the following common routes of effect (the number corresponds to the section number in which the effect is discussed below):

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 5. Limiting Species' Movement and Access to Habitat

- 6. Vessel Strikes
- 7. Noise

Based on the restrictions in the general PDCs described above, and our determinations regarding the common routes of effect, we have made the following determinations regarding the effects of the activities on the Central and Southwest Atlantic DPS of scalloped hammerhead shark, as summarized below in Table 11.

Tabl	e 11.	NMFS Effects Determ	inations for the Central a	and Southwest A	Atlantic DPS of
Scall	oped	Hammerhead Shark			

	Activity	NMFS Determination in	NMFS Determination in the
		Florida <sup>12</sup>	U.S. Caribbean
1	Shoreline stabilization	N/A	NLAA
2	Pile-supported structures	N/A	NLAA
3	Dredging	N/A	NLAA
4	Water-management outfall	N/A	
4	structures	N/A	NLAA
5	Scientific survey devices	N/A	NLAA
6	Boat ramps	N/A	NLAA
7	Aquatic enhancement	N/A	NLAA
8	Transmission/ utility lines	N/A	NLAA
9	Marine debris removal	N/A	NLAA
10	Temporary platforms, access fill, and cofferdams	N/A	NLAA

## **ESA-listed Whales**

ESA-listed whales (North Atlantic right, blue, fin, sei, sperm, and Bryde's whale) may be found within the action area. Blue, fin, sei, sperm, and Bryde's whales generally occur in deeper waters while North Atlantic right whales frequently come much closer to shore.

The Bryde's whale, which we recently proposed to list as an endangered species, has limited distribution; it appears to be restricted to the De Soto Canyon area of the Gulf of Mexico at water depths of -100 to -300 m. This Opinion does not apply to activities within this area, defined as the Bryde's whale exclusion zone in Section 2.1.1.6, per PDC AP.4. Therefore, we believe that there will be no effect to Bryde's whale from any of the activities covered under this Opinion.

The only activities analyzed in this Opinion that occur in waters accessible to the other ESAlisted whales (North Atlantic right, blue, fin, sei, and sperm) are the installation of buoys (Activity 2), temporary scientific survey devices (Activity 5), artificial reef placement (Activity

<sup>&</sup>lt;sup>12</sup> The individuals in the Central and Southwest Atlantic DPS for scalloped hammerhead shark are not found within waters off of Florida. Therefore, we do not consult on the effects to this species from activities occurring in Florida. The Northwest Atlantic and Gulf of Mexico DPS for this species includes species originating from the Northwest Atlantic and Gulf of Mexico, which includes waters off of Florida, however, we determined that this DPS did not warrant listing (78 FR 20717, April 5, 2013; 79 FR 38213, July 3, 2014).

7), and marine debris removal (Activity 9). None of the remaining activities occur in waters deep enough for whales to be present (i.e., Activity 1, 3, 4, 6, 8, 10). For those activities that can occur in waters accessible to whales, we believe:

- <u>Activities 2 and 5</u>: ESA-listed whales can become entangled by encountering in-water lines, such as buoy lines that may be associated with anchored buoys (Activity 2) or scientific survey devices (Activity 5). However, we believe that ESA-listed whales will not become entangled in in-water lines associated with Activities 2 and 5 since PDC AP.11 of the PDCs for *In-Water Activities* require that in-water lines be made of materials and installed in a manner to minimize the risk of entanglement. Under that PDC, for an activity to be covered under this Opinion, it must use thick, heavy, and taut lines that do not readily loop or entangle and all in-water lines must be stiff, taut, non-looping, and cannot have excess line in the water. There are no reports of whales becoming entangled in thick, taut lines. Therefore, we believe there will be no effect to ESA-listed whales from entanglement from lines associated with Activities 2 and 5.
- <u>Activity 7</u>: We also believe that ESA-listed whales may be affected by the placement of artificial reefs (Activity 7) if the reef placement restricts their ability to transit in or to waters in the action area used for calving. In particular, North Atlantic right whales may be affected by the placement of artificial reefs (Activity 7, Section 2.2.7) if the reef placement restricts their ability to transit waters in the action area used for calving. Effects to North Atlantic right whales from the placement of artificial reefs are analyzed in Section 2.2.7. On the other hand, reefs placed in deeper water where blue, fin, sei, and sperm whales may occur would not restrict movement of these whales because of the available space above the placed reef material and therefore there would not affect these whale species.
- <u>Activity 9</u>: Marine debris removal (Activity 9) will have no effect on whales (North Atlantic right, blue, fin, sei, and sperm) since debris is removed by divers in the water. Even if marine debris removal could affect ESA-listed whales, divers would be able to identify and avoid interaction with whales in the area. Also, marine debris removal is typically completed in a given area in less than a day, thereby limiting the time in which whale interactions could occur.

In addition to the effects described above, we believe mobile species, including ESA-listed whales, may be affected by a number of "common routes of effect," which are routes of effect common to all of the activities described in this Opinion. These routes of effect, and our analysis of whether and how they affect ESA-listed species and/or critical habitat, are described in a section below, entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion*. In particular, we analyzed the potential for the following routes of effect to have an effect on ESA-listed whales, except for Bryde's whale, and concluded each was a "no effect" on the species:

- 1. Direct Physical Effects from Construction Activities
- 3. Potential Entanglement in Construction Materials
- 5. Limiting Species' Movement and Access to Habitat

- 6. Vessel Strikes
- 7. Noise

With respect to Bryde's whale, because their range is generally restricted to the DeSoto Canyon in the Gulf of Mexico and activities covered under this Opinion are not allowed to occur in that area (PDC AP.4, as defined in Section 2.1.1.6), we believe there will be no effect to Bryde's whale.

In addition, we believe that North Atlantic right whales may be affected by the placement of artificial reefs (Activity 7) in their designated critical habitat. As described in that Section 2.2.7, we believe placement of artificial reefs in critical habitat for the North Atlantic right whale may affect, but is not likely to adversely affect the species.

Based on the restrictions in the general PDCs described above, and our determinations regarding the common routes of effect, we have made the following effects determinations, as summarized below in Table 12.

	Activity	NMFS Determination for North Atlantic Right Whale	NMFS Determination for Blue, Fin, Sei, and Sperm Whales and the proposed for listing Bryde's Whale
1	Shoreline stabilization	NE	NE
2	Pile-supported structures	NE	NE
3	Dredging	NE	NE
4	Water- management outfall structures	NE	NE
5	Scientific survey devices	NE	NE
6	Boat ramps	NE	NE
7	Aquatic enhancement	NLAA	NE
8	Transmission/ utility lines	NE	NE
9	Marine debris removal	NE	NE
10	Temporary platforms, fill, and cofferdams	NE	NE

Table 12. NMFS Effects Determinations for ESA-listed Whales

## Determinations Regarding Effects to Critical Habitat from All Categories of Activities

Some of the areas that have been designated as critical habitat will not be affected by any of the activities analyzed in this Opinion. Below, we discuss the critical habitat that will not be affected by any activity and explain our no effect determinations. Effects to these critical habitats will not be discussed further in this Opinion. The essential features/ PCEs/ PBFs for each critical habitat unit are provided in Table 7, above.

# NWA DPS of Loggerhead critical habitat

We believe that there will be no effect to critical habitat for the NWA DPS of loggerhead sea turtles from any of the categories of activities analyzed under this Opinion. Below we analyze the potential effects to each of the loggerhead critical habitat types that occur in Florida.

- 1. Nearshore reproductive habitat: We believe that there will be no effect to the PCEs of nearshore reproductive habitat of loggerhead critical habitat from any of the 10 categories of activities analyzed under this Opinion. As explained in Table 7, the PBF of nearshore reproductive habitat is the portion of the nearshore waters adjacent to nesting beaches that are used by hatchlings to egress to the open-water environment as well as by nesting females to transit between beach and open water during the nesting season, which is supported by the following PCEs: (i) Nearshore waters directly off the highest density nesting beaches and their adjacent beaches, as identified in 50 CFR 17.95(c), to 1.6 km offshore; (ii) Waters sufficiently free of obstructions or artificial lighting to allow transit through the surf zone and outward toward open water; and (iii) Waters with minimal man-made structures that could promote predators (i.e., nearshore predator concentration caused by submerged and emergent offshore structures), disrupt wave patterns necessary for orientation, and/or create excessive longshore currents.
  - For example, under the activity-specific PDCs, shoreline stabilization (Activity 1, PDC A1.7), boat ramps (Activity 6, PDC A6.6), transmission/utility line (Activity 8, PDC A8.2), and temporary platform, fill, and cofferdams (Activity 10, PDC A10.7) activities cannot occur on or near beaches used for sea turtle nesting and would therefore not affect the PCEs of nearshore reproductive habitat for loggerhead critical habitat.
  - With respect to Activity 2, ATONs (pile-supported and anchored buoys) are the only pile-supported structures allowed in nearshore reproductive habitat (Activity 2, PDC A2.16). The placement of ATONs generally requires only a single pile or anchor line, neither of which is large enough in size to create obstructions, add lighting, aggregate predators, or disrupt wave patterns. Thus, they do not affect the PCEs that support this habitat type.
  - Dredging (Activity 3) does not create obstructions, aggregate predators, or disrupt wave patterns. In addition, PDC AP.6 requires all activities be completed during daylight hours so we do not expect any affects from lighting associated with dredging. Thus, dredging does not affect the PCEs that support this habitat type.
  - Stormwater outfall structures (Activities 4), temporary scientific survey devices (Activity 5), and marine debris removal (Activity 9), given how they are designed and implemented generally and under the PDCs of this Opinion, do not create obstructions, add lighting, aggregate predators, or disrupt wave patterns. Specifically, stormwater outfall structures occur along the shoreline and do not create an obstruction to movement, temporary scientific survey devices are small (average of 20 ft<sup>2</sup>) so they are not large enough to create an obstruction, and marine debris removal removes structures so they do not create an obstruction. None of these structures are large enough or placed anywhere that they would attract fish and larger predators that would increase predation or are large

enough or placed somewhere they would disrupt wave patterns. Also none of these structures have lighting. Thus, they will have no effect on the PCEs that support this habitat type.

• Artificial reefs (Activity 7) placed in nearshore reproductive habitat could result in an aggregation of predators attracted to reefs for increased prey availability; however, under the PDC A7.30, Section 2.2.7, structures cannot be placed in loggerhead critical habitat nearshore reproductive habitat and would therefore not affect the PCEs of nearshore reproductive habitat for loggerhead critical habitat.

Therefore, we believe that the 10 categories of activities analyzed herein will have no effect on the nearshore reproductive habitat of loggerhead critical habitat.

- 2. Concentrated breeding habitat: We believe that there will be no effect from any of the 10 categories of activities analyzed in this Opinion to any of the PCEs of concentrated breeding habitat of critical habitat of the NWA DPS of loggerhead. As we stated in Table 7, the PBF of concentrated breeding habitat is those sites with high densities of both male and female adult individuals during the breeding season. PCEs that support this habitat are the following: (i) High densities of reproductive male and female loggerheads; (ii) Proximity to primary Florida migratory corridor; and (iii) Proximity to Florida nesting grounds. None of the activities analyzed in this Opinion will result in the injury, death, or redistribution of loggerhead sea turtles, so there will be no effect to the density of reproductive male and female loggerhead turtles. Additionally, none of the activities analyzed in this Opinion will affect the proximity of the breeding habitat to the Florida migratory corridor or to the Florida nesting grounds. With respect to the nesting grounds, the only activities that could block access to nesting beaches are shoreline stabilization (Activity 1), boat ramps (Activity 6), and temporary platform, fill, and cofferdams (Activity 10). However, the PDCs for those specific activities preclude them from occurring on or near sea turtle nesting beaches (see PDC A1.7, PDC A6.6, and PDC A10.7). Therefore, we believe that the 10 categories of activities analyzed herein will have no effect on the concentrated breeding habitat of loggerhead critical habitat.
- 3. <u>Constricted migratory corridors</u>: We believe that there will be no effect from any of the 10 categories of activities analyzed in this Opinion to any of the PCEs of constricted migratory corridor habitat of NWA DPS of loggerhead critical habitat. As is set forth in Table 7, the PBF of constricted migratory habitat is high use migratory corridors that are constricted (limited in width) by land on one side and the edge of the continental shelf and Gulf Stream on the other side. PCEs that support this habitat are the following: (i) Constricted continental shelf area relative to nearby continental shelf waters that concentrate migratory pathways; and (ii) Passage conditions to allow for migration to and from nesting, breeding, and/or foraging areas. The constricted migratory corridor in Florida is approximately 4 miles wide at its narrowest point.

Only 4 of the covered activities may involve installation of structures within the constricted continental shelf area relative to nearby continental shelf waters that concentrate migratory pathways, including pile-supported structures and anchored buoys (Activity 2), scientific

survey devices (Activity 5), artificial reefs (Activity 7), and utility and transmission lines (Activity 8). Since objects or structures deployed in these areas are spatially limited, we do not expect them to affect the PCEs of this habitat type, namely the amount of constricted continental shelf area relative to nearby continental shelf waters that concentrate migratory pathways or passage conditions that allow migration to and from nesting, breeding, or foraging habitats. Given the width of the corridor, even artificial reefs (Activity 7) placed in these areas would not restrict movement of turtles. Specifically, PDC A7.14 in Section 2.2.7 states that new reef sections covered under this Opinion are limited to 1 reef section measuring  $\frac{1}{4}$ - by  $\frac{1}{4}$ -nmi area (40 ac) in size with a distance of 500 ft between each section. In addition, turtles would not only be able to navigate around these structures, but also swim over them since artificial reefs are required to maintain enough space between the top of the structure and the water's surface that they do not create navigational hazards, which will help ensure that the activities do not affect the passage conditions described above. Specifically, PDC A7.14 in Section 2.2.7 states that offshore reefs covered under this Opinion must maintain a minimum vertical clearance of twice the height of the structure from the top of the deployed material relative to the MLW at all times. Therefore, we believe that the 10 categories of activities analyzed herein will have no effect on the constricted migratory corridor habitat of the NWA DPS of loggerhead critical habitat.

- 4. <u>Sargassum habitat</u>: We believe that there will be no effect from any of the 10 categories of activities analyzed in this Opinion to any of the PCEs for Sargassum habitat of critical habitat for the NWA DPS of loggerhead sea turtles. As stated in Table 7 above, the PBF of loggerhead Sargassum habitat is developmental and foraging habitat for young loggerheads where surface waters form accumulations of floating material, especially Sargassum. PCEs that support this habitat are the following: (i) Convergence zones, surface-water downwelling areas, the margins of major boundary currents (Gulf Stream), and other locations where there are concentrated components of the Sargassum community in water temperatures suitable for the optimal growth of Sargassum and inhabitance of loggerheads; (ii) Sargassum in concentrations that support adequate prey abundance and cover; (iii) Available prey and other material associated with Sargassum habitat including, but not limited to, plants and cyanobacteria and animals native to the Sargassum community such as hydroids and copepods; and (iv) Sufficient water depth and proximity to available currents to ensure offshore transport (out of the surf zone), and foraging and cover requirements by Sargassum for post-hatchling loggerheads, i.e., > 10-m depth.
  - These floating algae mats on the ocean surface occur miles offshore beyond the limits of most of the categories of activities discussed in this Opinion, and therefore the majority of projects conducted under this Opinion (those associated with Activities 1, 3, 4, 6, 8, and 10) will not affect sargassum resources.
  - Placement of buoys (Activity 2), scientific devices (Activity 5), artificial reefs (Activity 7), and marine debris removal (Activity 9) could occur in *Sargassum* habitat; however, none of these activities has the potential to affect the concentration of floating *Sargassum* mats in amounts that support adequate prey abundance and cover or the availability of

prey PCEs of *Sargassum* habitat. Buoys and scientific devices are too small to affect large areas of floating mats, and artificial reefs are located below the water surface.

- Marine debris is typically removed by manual means, and PDC A9.2 applicable to marine debris removal requires visual confirmation that the debris can be removed without causing further damage to aquatic resources. PDC A9.5 also requires that the marine debris be lifted straight up rather than dragged, which will also prevent removal of *Sargassum*.
- None of the activities evaluated in this Opinion is large enough to affect ocean currents such as location of convergence zones, or the water depth or proximity to currents necessary for offshore transport to sargassum beds for foraging, and cover.

Therefore, we believe that the 10 categories of activities analyzed herein will not affect the *Sargassum* habitat of critical habitat of the NWA DPS of loggerhead sea turtles.

Based on the restrictions in the general PDCs discussed above, as well as some activity specific PDCs provided above, our effects determinations are summarized below in Table 13.

 Table 13. NMFS Effects Determinations for NWA DPS of Loggerhead Sea Turtle Critical

 Habitat

	Activity	NMFS Determination
1	Shoreline stabilization	NE
2	Pile-supported structures	NE
3	Dredging	NE
4	Water-management outfall structures	NE
5	Scientific survey devices	NE
6	Boat ramps	NE
7	Aquatic enhancement	NE
8	Transmission/ utility lines	NE
9	Marine debris removal	NE
10	Temporary platforms, fill, and cofferdams	NE

# Acropora critical habitat

We believe there will be no direct effect to the essential feature of *Acropora* critical habitat substrate of suitable quality and availability to support larval settlement and recruitment, and reattachment and recruitment of asexual fragments—from any of the categories of activities analyzed under this Opinion. Specifically under PDC AP.12 of the PDCs for *Mangroves*, *Seagrasses, Corals and Hard Bottom for All Projects*, hardbottom, including the hardbottom that meets the definition of the essential feature of *Acropora* critical habitat, cannot be within the project footprint. The only exception is for the temporary placement (up to 24 months) of scientific survey devices (Activity 5) that have a footprint of less than 1 ft<sup>2</sup> per device. We believe the temporary placement of scientific survey devices will not affect the hardbottom essential feature of *Acropora* critical habitat since it will not alter or change the feature. The devices are temporarily placed on the hardbottom habitat in a manner that does not damage the feature.

Activity-specific PDCs reiterate the general limitation in PDC AP.12, stating that the following activities are only allowed in Acropora critical habitat if the essential feature is not present: new and expanded shoreline stabilization projects (Activity 1, PDC A1.10); pile-supported structures (Activity 2, PDC A2.10); dredging (Activity 3, PDC A3.7); boat ramps (Activity 6, PDC A6.8); restoration and aquatic enhancement (Activity 7, PDC A7.28); and temporary platforms, fill, and cofferdams (Activity 10, PDC A10.9). For transmission and utility line projects (Activity 8), the activity-specific PDCs are broader for sub-categories of activities. Under those PDCs, the Opinion does not apply to the new installation of transmission and utility lines or to trenching and horizontal directional drilling within the geographic boundary of Acropora critical habitat, regardless of whether the essential feature is present (PDC A8.7). In addition, under that PDC, existing lines only may be repaired or replaced in areas where the essential feature is not present. Finally, with respect to marine debris removal (Activity 2), PDC A9.3 limits the Opinion to those removal activities that will not cause further damage to aquatic resources and PDC A9.5 excludes activities where debris is dragged through coral or hard bottom habitats from coverage. Therefore, we believe the activities covered under this Opinion will not physically damage or affect the essential feature of Acropora critical habitat through direct interaction with construction equipment or machinery.

For water-management outfall structures (Activity4), we do not expect any changes in water quality from the operation of water management outfall structures to affect *Acropora* critical habitat. PDC A4.3 states that all outfall structures covered under this Opinion must be designed and implemented to prevent erosion and scouring so that these outfall structures do not generate sedimentation and turbidity through erosion after the structure is placed and operational. Therefore, we believe that there will be no effect to *Acropora* critical habitat essential features from Activity 4.

Based on the restrictions in the general PDCs discussed above, as well as some activity specific PDCs provided above, our effects determinations are summarized below in Table 14.

I abr	Table 14. NWIPS Effects Determinations for Acropora Critical Habitat				
	Activity	NMFS Determination			
1	Shoreline stabilization	NE			
2	Pile-supported structures	NE			
3	Dredging	NE			
4	Water-management outfall structures	NE			
5	Scientific survey devices	NE			
6	Boat ramps	NE			
7	Aquatic enhancement	NE			
8	Transmission/ utility lines	NE			
9	Marine debris removal	NE			
10	Temporary platforms, fill, and cofferdams	NE			

Table 14. NMFS Effects Determinations for Acropora Critical Habitat

#### North Atlantic right whale critical habitat

We believe that there will be no effect to North Atlantic right whale critical habitat from any category of activity analyzed under this Opinion, except from the placement of artificial reef structures (Activity 7, Section 2.2.7), which we believe are not likely to adversely affect North Atlantic right whale critical habitat, as discussed in Section 2.2.7, below.

As provided in Table 7, the essential features which provide calving area functions in Unit 2, are (i) sea surface conditions associated with Force 4 or less on the Beaufort Scale, (ii) sea surface temperatures of 7°C to 17°C, and (iii) water depths of 20-92 ft (6- 28 m), where these features simultaneously co-occur over contiguous areas of at least 231 nmi<sup>2</sup> of ocean waters during the months of November through April. When these essential features are available, they are selected by right whale cows and calves in dynamic combinations that are suitable for calving, nursing, and rearing, and which vary, within the ranges specified, depending on factors such as weather and age of the calves.

The only activity that may affect these features is the placement of artificial reefs. None of the other minor activities analyzed under this Opinion are large enough in size or scale to change sea state conditions and none of the activities change water temperatures. The only activity that can change overall water depth is the placement of materials to create an artificial reef. The effects of artificial reefs on North Atlantic right whale critical habitat are discussed further in Section 2.2.7.

	Activity	NMFS Determination
1	Shoreline stabilization	NE
2	Pile-supported structures	NE
3	Dredging	NE
4	Water-management outfall structures	NE
5	Scientific survey devices	NE
6	Boat ramps	NE
7	Aquatic enhancement	NLAA
8	Transmission/ utility lines	NE
9	Marine debris removal	NE
10	Temporary platforms, fill, and cofferdams	NE

Our effects determinations are summarized below in Table 15.

Table 15. NMFS Effects Determination for North Atlantic Right Whale Critical Habitat

# U.S. Caribbean Sea Turtle Critical Habitat (Green, Hawksbill, and Leatherback Sea Turtle Critical Habitat)

We believe that there will be no effect to the essential features of critical habitat for hawksbill, leatherback, or the NA DPS of green sea turtles from Activity 1 or Activities 3-10 analyzed under this Opinion. Within the action area, all 3 of these sea turtle critical habitat units occur only within the U.S. Caribbean. At that time of critical habitat designation for green, hawksbill, and leatherback sea turtle critical habitat, the essential features of the critical habitat were not precisely defined. However, the critical habitat was designated primarily to provide protection

for important developmental and resting habitats for green and hawksbill sea turtles and for leatherbacks, to provide protection to sea turtles using the designated waters for courting and breeding. Specifically:

- Green sea turtles (both juvenile and adults) depend on seagrasses as the principal dietary component of foraging. In addition, coral reefs and other topographic features within the waters around Culebra Island and surrounding islands and cays provide green turtles with shelter during interforaging periods that serve as refuge from predators.
- Hawksbill sea turtles depend on sponges as their principal dietary component and rely on healthy coral reefs for foraging and shelter habitats.
- Leatherback sea turtles require access to and from nesting areas on Sandy Point Beach, St. Croix, U.S. Virgin Islands.

To avoid impacts to these important foraging and sheltering habitats, PDC AP.13 states that the Opinion does not apply to projects in the U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat identified in Section 2.1.1.5) if non-ESA listed, native, non-invasive seagrasses are found within the project footprint. Under PDC AP.14, the Opinion also does not apply to projects in areas where coral and/or hardbottom occur within the project footprint, unless the project is for the temporary placement (up to 24 months) of scientific survey devices (Activity 5) that have a footprint of less than 1 ft<sup>2</sup> per device. This Opinion further restricts the activities that can occur in sea turtle critical habitat in the U.S. Caribbean (hawksbill, leatherback, and the NA DPS of green sea turtles) so that they will not create an obstruction that would prevent access to and from nesting areas for leatherback sea turtles. Specifically:

- This Opinion does not apply to water-management outfall structures (Activity 4, PDC A4.7) and aquatic enhancement (Activity 7, PDC A7.31) projects occurring in sea turtle critical habitat in the U.S. Caribbean.
- Within sea turtle critical habitat in the U.S. Caribbean, some activities are limited to removal, repair, and replacement of existing structures, including shoreline stabilization projects (Activity 1, PDC A1.12), boat ramps (Activity 6, PDC A6.11), and transmission and utility lines (Activity 8, PDC A8.11). Since these projects will not expand the footprint of the existing structures, we believe they will not affect access to nesting beaches.
- Activity 2: Of the pile-supported structures and anchored buoys that are a part of this Opinion, only ATONs can be placed near sea turtle nesting beaches (PDC A2.16). No other pile-supported structures are allowed near sea turtle nesting beaches under this Opinion. The placement of ATONs is not expected to limit access to nesting beaches since ATONs are single small structures either mounted on a pile or anchored with a buoy.
- Activity 3: Though dredging may change the bottom contours in the area dredged, the removal of sediments does not create an obstruction to beach access.

- Activity 5: We do not believe that the temporary placement of scientific survey devices, which are small (estimated at 1-50 ft<sup>2</sup>), will create an obstruction to sea turtles accessing beaches.
- Activity 9: Marine debris removal only removes items that may have created an obstruction or blocked foraging or shelter access and does not create any new obstacles.
- Activity 10: Of the temporary platforms, fills, and cofferdams covered under this Opinion, only temporary cofferdams placed in conjunction with other construction activities that are covered under this Opinion can be placed in sea turtle critical habitat in the U.S. Caribbean. These structures cannot occur on or contiguous to ocean beaches that may be used by nesting sea turtles (PDC A10.12) so they would not obstruct access to nesting beaches.

Therefore, we believe there will be no effect to the essential features of critical habitat from any activity that can be permitted under this Opinion (Table 16).

 Table 16. NMFS Effects Determination for Green, Hawksbill, and Leatherback Sea Turtle

 Critical Habitat

	Activity	NMFS Determination
1	Shoreline stabilization	NE
2	Pile-supported structures	NE
3	Dredging	NE
4	Water-management outfall structures	NE
5	Scientific survey devices	NE
6	Boat ramps	NE
7	Aquatic enhancement	NE
8	Transmission/ utility lines	NE
9	Marine debris removal	NE
10	Temporary platforms, fill, and cofferdams	NE

# Atlantic sturgeon critical habitat

The only Atlantic sturgeon critical habitat in Florida is located in the St. Marys River. We believe that there will be no effect (Table 17) to Atlantic sturgeon critical habitat because this Opinion does not cover projects that occur in the St. Marys River as set forth in Section 2.1 (Action Area).

	Activity	NMFS Determination
1	Shoreline stabilization	NE
2	Pile-supported structures	NE
3	Dredging	NE
4	Water-management outfall structures	NE
5	Scientific survey devices	NE
6	Boat ramps	NE
7	Aquatic enhancement	NE
8	Transmission/ utility lines	NE
9	Marine debris removal	NE
10	Temporary platforms, fill, and cofferdams	NE

 Table 17. NMFS Effects Determination for Atlantic Sturgeon Critical Habitat

## Construction Related Effects for All Categories of Activities Analyzed under this Opinion

This Opinion covers 10 categories of in-water construction activities, each of which could affect the ESA-listed species listed in similar ways. For example, species could interact with construction equipment from the covered activities or could be affected by turbidity from the covered activities in the same way. The general construction-related routes of effect are limited by the general PDCs that are required for all projects and are discussed below.

# 1. Direct Physical Effects from Construction Activities

Potential effects to mobile species listed in Table 5 including whales (i.e., sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and whales) include the risk of interaction with construction equipment from land-based and barge-mounted equipment, small boats, and the placement or removal of materials associated with each of the 10 categories of activities covered under this Opinion. We believe the potential for these species to be adversely affected by interactions with equipment or materials used for any of the activities analyzed under this Opinion is extremely unlikely, and thus that the effect of such interactions is discountable. These species are mobile and are very likely to avoid the construction noise, moving equipment, and placement or removal of materials during construction, making interactions extremely unlikely. In addition:

- PDC AP.7 of the PDCs for *In-Water Activities* requires the permittee to instruct all personnel associated with the project about the potential presence of species protected under the ESA and the MMPA, and makes clear that all on-site project personnel are responsible for observing water-related activities for the presence of protected species.
- PDC AP.9 of the PDCs for *In-Water Activities* requires that all vessel operators watch for and avoid collisions with species protected under the ESA and MMPA. This PDC establishes speed restrictions and safe operating distances, and requires operation of mechanical construction equipment to cease when a listed species is observed within a 50-ft radius of construction equipment.

One activity, dredging (Activity 3), has the potential to physically harm mobile species listed in Table 5 except for whales (i.e., sea turtles, smalltooth sawfish, Nassau grouper, and sturgeon); however, the PDCs for Activity 3, and the general PDCs, render such interactions extremely unlikely. PDC A3.2 applicable to Activity 3 excludes projects using hopper dredging from coverage under the Opinion. Dredging covered under this Opinion is limited to minor dredging projects, maintenance dredging, muck dredging, and dredging/grading areas for the placement of seawalls and boat ramps. Hence, dredging will be completed using smaller equipment (e.g., backhoe, clamshell, hydraulic vacuum, cutterhead dredges [likely 18-in diameter or less]). NMFS has previously determined (NMFS 2007b) that, while ocean-going hopper-type dredges may lethally entrain protected species including sea turtles and sturgeon, non-hopper type dredging methods (e.g., mechanical such as clamshell, and bucket dredging; hydraulic [suction] cutterhead, and pipeline) are slower and extremely unlikely to adversely affect sea turtles and sturgeon. Despite rare reports of cold-stunned turtles (i.e., lethargic, dying, or previously dead) being taken by cutterhead dredges, in Laguna Madre, Texas (Robert Hauch, Galveston USACE, pers. comm. to Eric Hawk, NMFS PRD, March 6, 2012), NMFS has no new information that would change our conclusion that the effects of non-hopper type dredging methods are discountable. Due to these species' mobility, the use of non-hopper-type dredges, and adherence to the PDCs for *In-Water Activities*, it is extremely unlikely that these species' would be struck or otherwise adversely affected by the dredging equipment, therefore, the effects to these species from dredging is discountable.

Based on the restrictions in the general PDCs discussed above, as well as some activity specific PDCs provided above, we believe that effects to ESA-listed species from interaction with construction equipment from land-based and barge-mounted equipment, small boats, and the placement or removal of materials associated with any of the categories of activities covered under this Opinion will be discountable.

#### 2. Turbidity

Sedimentation and turbidity may be generated during sediment moving activities such as dredging to place a seawall (Activity 1); auguring to place a pile (Activity 2); maintenance, minor, or muck dredging (Activity 3); dredging to place an outfall structure (Activity 4) or boat ramp (Activity 6); trenching to place a subaqueous utility line (Activity 8), or placement of fill (Activity 10). To control and reduce turbidity, PDC AP.10 requires the use of turbidity curtains, which will be installed prior to and throughout all in-water construction. Turbidity curtains will remain in place post-construction until all turbidity and siltation subsides from in-water construction.

Turbidity curtains are designed to avoid and minimize the effects of turbidity and sedimentation outside of the curtains, though some turbidity and sedimentation may escape the curtains, either due to imperfections in the curtain material or installation, or in the event of an unexpected curtain failure. In a few, limited instances, the USACE project manager may waive the turbidity curtain requirement. For example, the project manager may waive the requirement for projects that are so small that turbidity is not expected (PDC AP.10). Similarly the requirement can be waived for projects that are not expected to affect water quality (PDC AP.10) such as for the placement of a single pile or removal of marine debris, which would not be expected to generate enough turbidity to affect water quality or to affect ESA-listed species. Another instance where

the turbidity curtains requirement may be waived is when the project is located in an area with high wave energy where securing turbidity curtains would not be feasible (PDC AP.10). If not properly secured, the curtains are at an increased risk of becoming loose and entangling animals or damaging nearby habitat. In high wave energy areas, turbidity would dissipate quickly, minimizing species' exposure to turbidity in the area.

We believe turbidity and sedimentation from in-water construction will have different effects to ESA-listed species based on whether they are mobile (i.e., sea turtles, smalltooth sawfish, sturgeon, and Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark) or stationary (Johnson's seagrass). Mobile species are able to swim through or avoid any potential turbidity without harm. These species are naturally exposed to turbidity or sedimentation throughout their environment in areas of naturally lower water clarity. Therefore, we believe any potential exposure to turbidity outside of the turbidity curtains will have no effect on mobile ESA-listed species.

ESA-listed stationary species (Johnson's seagrass) are unable to avoid turbidity and sedimentation that may be generated during in-water construction. Johnson's seagrass is susceptible to harm from being covered or buried by sedimentation generated by turbidity. For that reason, the following additional protective measures are required in the range of Johnson's seagrass:

- Turbidity curtains cannot be waived when the project is within the range of Johnson's seagrass, even if the project would be located in dynamic systems like high wave and current areas (PDC AP.10). If turbidity curtains are not feasible in these areas, the project cannot be covered under this Opinion and individual Section 7 consultation would be required.
- Dredging projects must ensure that the dewatering, transfer, and storage of dredge materials be contained and not result in sedimentation and turbidity that could cover and harm corals or Johnson's seagrass in the area (PDC A3.3.2).

Based on these PDC restrictions and protections, we believe that the risk of sedimentation and turbidity harming or affecting Johnson's seagrass is extremely low, and the effect of such sedimentation and turbidity is discountable.

## 3. Potential Entanglement in Construction Materials

The presence of flexible materials in the water, such as turbidity curtains (as required under PDC AP.11 for all activities generating turbidity) and in-water lines, such as those used to secure buoys covered in Activities 2 and temporary scientific survey devices covered under Activity 5, could create an entanglement risk to mobile species listed in Table 5 including whales (i.e., sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and whales). In particular, PDC AP.10 states that all activities covered under this Opinion must use turbidity curtains, unless the requirement has been waived as described in the PDC.

Although the turbidity curtains and in-water lines pose an entanglement risk to ESA-listed species in Table 5 including whales (i.e., sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and whales), PDC

AP.11, applicable to all projects, states that all in-water lines (e.g., rope, chain, and cable, including the lines to secure the turbidity curtains) must be stiff, taut, and non-looping to minimize the risk of entanglement. If flexible lines are used, they must be enclosed in plastic or rubber sleeves/tubes that add rigidity and prevent the line from looping and tangling. There are no reports of any listed species that have been entangled in turbidity curtains or stiff, taut, non-looping in-water lines or flexible lines enclosed in the plastic or rubber sleeves. In addition, PDC AP.11 requires all in-water lines, including the lines securing turbidity curtains, to be properly secured and for applicants to ensure that there is no excess line in the water. It also requires turbidity curtains and in-water equipment to be placed in a manner that does not entrap species within the construction area or block access for them to navigate around the construction area. Therefore, we believe that there will be no effect to these species from entanglement.

#### 4. Exclusion from Areas during Construction

Use of turbidity curtains (required under PDC AP.10) and the construction activities and related construction noise may preclude or deter mobile species listed in Table 5, except whales (i.e., sea turtles, smalltooth sawfish, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and sturgeon) from entering the project area. We believe the temporary exclusion of these species from a project area due to construction activities including related noise and presence of turbidity curtains will have insignificant effects on those species. Turbidity curtains will enclose the project site or portions of the project site at any given time and will be removed after construction. However, species excluded from the project area will be able to use surrounding areas during construction and return to the project site when construction is complete. Per PDCs AP.10 for *In-Water Activities*, turbidity curtains must be positioned in a way that does not block access to any designated critical habitat. Likewise, if an animal chooses to avoid the project site because of the activities and/or the related noise, it will be able to use surrounding areas and return to the project site when construction is complete. Therefore, we believe the temporary exclusion of areas from activities covered under this Opinion will have insignificant effects on these species.

#### 5. Limiting Species' Movement and Access to Habitat

Activities conducted under this Opinion could affect movement and access to habitat of mobile species listed in Table 5, as discussed below. In this section, we analyze if the placement of materials from any of the activities or if the presence of construction equipment in a specific area could potentially affect the species' ability to move freely in an area, resulting in its inability to access resources needed for essential life functions such as access to areas used for reproduction, foraging, or refuge, or otherwise interfere with their ability to use areas for reproduction. The loss of foraging or refuge habitat from the placement of the materials is analyzed separately for each activity type discussed in Sections 2.2.1-2.2.10.

#### Movement and access to foraging or refuge habitat

We believe the following activities will have no effect on mobile species listed in Table 5, including whales (i.e., sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and whales) from limiting their movement and ability to access foraging or refuge habitat, as described below:

- Shoreline stabilization (Activity 1); pile-supported structures (Activity 2, with the exception of ATONs, which are typically installed on a single pile or buoy in open waters); watermanagement outfall structures (Activity 4), and boat ramps (Activity 6) are all activities that occur along the shoreline. The placement of such materials along the shoreline would not create an obstruction for species to move around these features to access foraging and refuge habitat in surrounding areas. The placement of a single pile or buoy for an ATON also would not create an obstruction when placed in open water. Therefore, we believe these activities have no effect on species movement and access to foraging and refuge habitat.
- Maintenance, minor, and muck dredging (Activity 3) and marine debris removal (Activity 9) do not result in the placement of materials in the marine environment and would therefore not create an obstruction to movement for species in the area. As stated in the PDCs for Activity 3 (dredging), this Opinion does not analyze or otherwise operate as a consultation on the placement of dredged material in the marine environment, but is limited to the placement of dredged material above MHW beyond the limit of NMFS' jurisdiction, where we believe there is no risk of limiting species movement and access to foraging and refuge habitat.
- Scientific survey devices (Activity 5) and transmission and utility lines (Activity 8) are small in size and not expected to create an obstruction to movement of species in an area. Scientific survey devices only allows the temporary placement of materials (PDC A5.3), which the USACE assumes will be under 50 ft<sup>2</sup> each based on previously authorized devices. Transmission and utility lines are either installed subaqueous or are supported on single piles spaced apart to support aerial transmission lines.

Therefore, we believe that the placement of materials associated with Activities 1-6, 8, and 9 will have no effect on species by limiting their ability to access foraging and refuge habitat.

We believe the following activities could be designed to be large enough or placed in a manner that could obstruct movement of mobile species listed in Table 5, as described below. However, the PDCs are designed to minimize these structures to an extent as to minimize effects on movement of species in an area so that the effect to species from movement obstructions is insignificant, as described below:

• The installation of long, linear living shoreline features (Activity 7) could reduce or prevent mobile species listed in Table 5 with the exception of whales (i.e., sea turtles, smalltooth sawfish, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and sturgeon) from accessing an area between the shore and the living shoreline placed offshore. However, the PDC in Activity 7 for living shoreline projects (PDC A7.6) requires that intermittent gaps be included in the linear structures to ensure that the species can move through and around these areas to access habitat resources along the shore (e.g., mangroves habitat used by smalltooth sawfish) or in shallow waters along the shore (e.g., seagrasses, sponges, crustaceans used sea turtles; invertebrates for sturgeon; or bony fish, octopi, cuttlefish, squid, crabs, lobsters, and rays for Central and Southwest Atlantic DPS of scalloped hammerhead shark).

- The installation of artificial reefs (Activity 7) also could prevent movement of mobile species listed in Table 5, including whales (i.e., sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and whales); however, the PDC limitations (PDC A7.14) ensure that the size or placement is not so large as to obstruct species movement through an area. Since artificial reefs are placed offshore in deeper waters, mobile species can navigate over, between, or around these structures. Additional protective measures are required in North Atlantic right whale critical habitat to ensure that reef materials are placed and sized to allow adequate room for North Atlantic right whales to navigate above, between, and around these structures the (PDC A7.27). This Opinion also does not apply to artificial reefs if they are to be placed in Gulf sturgeon critical habitat (PDC A7.26). Therefore, we believe that the living shorelines and artificial reefs associated with Activity 7 are not likely to affect these species by impeding their movement or ability to access habitat.
- Activity 10 includes temporary platforms, fill, and cofferdams typically associated with bridge construction. Since these projects can occur in narrow areas spanned by bridges, PDCs (PDC A10.5) are in place to ensure the placement of these structured do not obstruct movement in confined areas such as a channel or river and that they must be placed so that they do not impede normal downstream flows or species movement in the area of mobile species listed in Table 5, except whales (i.e., sea turtles, smalltooth sawfish, Nassau grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and sturgeon).

Therefore, we believe the placement of structures under Activities 7 and 10 will have an insignificant effect mobile species from limiting their movement or ability to access foraging and refuge habitat.

## Limiting species movement for reproduction

In specific locations and during certain times of years, species migrate for reproductive purposes. Below is a discussion of the activities that could result in an obstruction to migrating species:

<u>Sea turtles</u>: Loggerhead, green, and leatherback sea turtles nest on Florida beaches and hawksbill, green, and leatherback sea turtles nest on beaches in the U.S. Caribbean. Female sea turtles migrate to nesting beaches to lay eggs and hatchlings migrate away from these beaches. The placement of materials or presence of equipment in front of (i.e., waterward of) nesting beaches could interfere with or obstruct sea turtles' ability to access or leave the beach. Also, the presence of lights in the area could disorient hatchlings leaving the beach that use the moon to navigate to sea. This Opinion includes PDCs designed to protect sea turtles so that there will be no effect to sea turtle migration or movement to, and use of, nesting beaches.

- PDCs preclude placement of materials on or near sea turtle nesting beaches, which prevents effects to sea turtles from reduced access to those resources from shoreline stabilization (Activity 1, PDC A1.7), boat ramps (Activity 6, PDC A6.6), and temporary platforms, fill, and cofferdams (Activity 10, PDC A10.7).
- Activities associated with pile-supported structures and ATONs (Activity 2), dredging (Activity 3), outfall structures (Activity 4), scientific survey devices (Activity 5), seagrass

restoration (Activity 7), utility and transmission lines (Activity 8), and marine debris removal (Activity 9) will not create obstructions, add lighting, or otherwise disrupt access to and from nesting beaches, or otherwise interfere with the use of these areas, by adult or hatchling sea turtles. These activities (1) are small in size such they will not obstruct movement, (2) are typically completed during the day when turtles are not approaching the beaches for nesting, and (3) do not create lighting, structures, or other potential disruptions to nesting beach access.

- Under the PDCs for Activity 2, docks installed within visible distance of on ocean beaches are required to use turtle-friendly lighting if lighting is necessary for the project (Activity 2, PDC A2.8). Turtle-friendly lighting is lighting that is installed in a manner that does not allow light to be seen from the water so that it does not disorient hatchlings leaving the beach.
- Finally, under the PDCs described below, this Opinion does not apply to projects occurring in sea turtle critical habitat in the U.S. Caribbean, including the critical habitat for green, hawksbill, and leatherback sea turtles

Therefore, we believe that none of the activities will result in obstructions to sea turtles accessing nesting beaches or create lighting that would be disorienting to hatchling sea turtles, and thus we believe the activities will not affect sea turtles by limiting their ability to access or use nesting beaches.

<u>Smalltooth sawfish</u>: Studies have identified certain areas that are suspected to support higher levels of pupping in smalltooth sawfish critical habitat. These areas are identified in Section 2.1.1.1 as smalltooth sawfish limited exclusion zones.

- Under the activity-specific PDCs, certain categories of activities that could otherwise result
  in effects to sawfish migration and movement to and from pupping areas will not affect this
  migration or movement since these activities cannot occur in smalltooth sawfish limited
  exclusion zones (areas identified as areas supporting higher levels of smalltooth sawfish
  pupping). In particular, shoreline stabilization (Activity 1, PDC A1.8.5), pile-supported
  structures and ATONs (Activity 2, PDC A2.12), dredging and disposal (Activity 3, PDC
  A3.5.4), outfall structures (Activity 4, PDC A4.5), boat ramps (Activity 6, PDC A6.5.4),
  aquatic restoration activities (Activity 7, PDC A7.25), utility and transmission lines (Activity
  8, PDC A8.8), and temporary platform, fill, and cofferdams (Activity 10, PDC A10.10)
  cannot occur in smalltooth sawfish limited exclusion zones. Because these activities cannot
  occur in sawfish limited exclusion zones, which include known hot spots and pupping areas,
  we believe those activities (Activities 1- 4, 6- 8, and 10) will not affect smalltooth sawfish by
  interfering with their migration to and use of areas for pupping.
- Scientific surveys (Activity 5) and marine debris removal (Activity 9) can occur in smalltooth sawfish limited exclusion zones; however, those activities do not block species movement or access to important areas. The PDCs specific to scientific surveys explicitly require that scientific survey devices do not block access of listed species to their habitat (PDC A5.2). Additionally, such devices can only be left in place for up to 24 months; thereafter, the site must be returned to pre-project conditions (PDC A5.3). Marine debris

removal under this Opinion is not of a scale that could create obstructions, involving coordination between divers and a small number of vessels, and is completed quickly. Because installation of scientific survey devices (Activity 5) and marine debris removal (Activity 9) are very low impact activities completed quickly, and to be covered under the Opinion, cannot block access to and from sawfish pupping habitat, we believe these activities will have no effect on smalltooth sawfish from interfering with migration to and use of areas for pupping.

Therefore, we believe that none of the activities will result in obstructions to smalltooth sawfish to accessing or using areas that support known higher levels of pupping in smalltooth sawfish critical habitat, and will have no effect on smalltooth sawfish.

<u>Sturgeon</u>: Shortnose, Atlantic, and Gulf sturgeon migrate between estuaries and spawning rivers. None of these species occur or migrate in the U.S. Caribbean.

- Shortnose sturgeon are not known to spawn in Florida rivers, and therefore there will be no effect to their spawning migration from activities analyzed in this Opinion.
- Atlantic sturgeon are believed to spawn in the St. Marys River in Florida, but the Atlantic sturgeon critical habitat exclusion zone, identified in Section 2.1.1.3, excludes activities occurring in that river from this Opinion. Given this exclusion, there will be no effect to Atlantic sturgeon from effects to spawning migrations from activities analyzed in this Opinion.
- Gulf sturgeon spawning rivers are under the jurisdiction of the USFWS while the estuarine and marine environment used by these species is under the jurisdiction on NMFS. The mouth of Gulf sturgeon spawning rivers and narrow inlets used by Gulf sturgeon to access spawning rivers are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones, where additional restrictions apply. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, activities that have the potential to disrupt or prevent migration and movement of Gulf sturgeon, including shoreline stabilization (Activity 1), pile-supported structures and ATONs (Activity 2), dredging and disposal activities (Activity 3), water-management outfall structures (Activity 4), boat ramps (Activity 6), aquatic enhancement (Activity 7), transmission and utility lines (Activity 8), and temporary platforms, fill, and cofferdams (Activity 10), are subject to activity-specific PDCs that ensure that these activities do not obstruct access to or deter Gulf sturgeon from entering the mouths of spawning rivers or narrow inlets between marine and estuarine areas. In particular, these activities are subject to PDCs that implement the additional noise restrictions listed in Section 2.1.1.2 for the Gulf sturgeon critical habitat migratory restriction zones (PDC A1.9, PDC A2.11; PDC A4.6; PDC A6.10), limit dredging (PDC A3.6) and the installation of utility and transmission lines (PDC A8.10) to outside spawning periods, restrict aquatic restoration (oyster reefs, living shorelines, and seagrass to less than -6 ft and no artificial reefs) in Gulf sturgeon critical habitat to areas shallower than the Gulf sturgeon preferred foraging depths (PDC A7.26), prohibits living shoreline, oyster reef, and artificial reef activities (Activity 7, PDC A7.26) and fill projects (Activity 19, PDC A10.11) in the Gulf sturgeon critical habitat migratory restriction zones. Since Activities 1-4, 6-8, and 10

are limited or cannot occur in Gulf sturgeon critical habitat migratory restriction zones by activity-specific PDCs, we believe these activities will have no effect on sturgeon migration and no effect on the species from interference with migration.

Scientific surveys (Activity 5) and marine debris removal (Activity 9can occur in the mouth of Gulf sturgeon spawning rivers and narrow inlets used by Gulf sturgeon to access spawning rivers that we identified as Gulf sturgeon critical habitat migratory restriction. However, the PDCs for scientific survey devices exclude installation of devices that have the potential to interfere with access of listed species to their habitat from the Opinion (PDC A5.2). Installation of scientific survey devices (Activity 5), which, based on information from the USACE, we assume will be less than 50 ft<sup>2</sup> in size, and marine debris removal activities (Activity 9) are very low impact activities that are completed quickly, often in less than a day. Thus, we believe that these small projects (in size, scope, and use of equipment) will not deter sturgeon from moving through narrow inlets or entering spawning rivers and therefore will have no effect on the migration of Gulf sturgeon.

Therefore, Activities 1-10 will have no effect on sturgeon (Atlantic, Gulf, or shortnose) from the placement of materials or presence of construction obstructing the species from accessing spawning rivers.

<u>Nassau grouper</u>: The only known spawning sites for Nassau grouper in the action area occur in the U.S. Caribbean. Nassau grouper spawn in known aggregation sites at night around a full moon from November to February. Since PDC AP.6 restricts all work to daytime hours, we believe that none of the activities will affect the ability for Nassau grouper to form aggregations for spawning.

<u>Central and Southwest Atlantic DPS of scalloped hammerhead shark</u>: At this time we are unaware of any specific reproductive areas for the Central and Southwest Atlantic DPS of scalloped hammerhead shark in the action area. Studies suggest that scalloped hammerhead sharks do not participate in natal homing, which would essentially restrict the species to a specific nursery ground, but instead have been found to use large estuaries as nursery habitats located 100 to 600 km from established nursery grounds (Duncan et al., 2006). Therefore, we believe that none of the activities are large enough to block access to an entire potential nursery area that the Central and Southwest Atlantic DPS of scalloped hammerhead shark may use for reproduction.

<u>Whales</u>: The only known reproductive area for whales in the action area is the area designated as North Atlantic right whale critical habitat. We designated these areas to protect the whales using the area for calving. The placement of artificial reefs in this area is the only type of activity analyzed in this Opinion that could be large enough in size or scale to obstruct movement of North Atlantic right whales calving. The effects of artificial reefs on North Atlantic right whale critical habitat are discussed further in Section 7.

## 6. Vessel Strikes

Vessel traffic, both recreational and commercial, has been documented in stranding reports to adversely affect protected marine mammals and sea turtles, but little information exists on interactions with smalltooth sawfish, Nassau Grouper, Central and Southwest Atlantic DPS of scalloped hammerhead shark, and sturgeon (Gulf sturgeon, shortnose, and Atlantic) as these species' primarily demersal habits would rarely put them at risk from vessels at the surface. No stranding reports indicate smalltooth sawfish or Nassau grouper have been struck by vessels. Limited stranding records show that sturgeon have been struck by large shipping vessels in narrow channels in the Northeast. These narrow channels combined with the deep drafts of the shipping vessels prevent sturgeon from being able to avoid interactions with the vessels. The action area does not contain narrow shipping channels that limit the species' ability to avoid the vessels as are found in the Northeast, so we do not expect such interactions here. In addition, independent of any of the PDCs in this Opinion, navigational markers alert both recreational and commercial boaters to shallow areas to prevent groundings. Because the vessels are likely to rely on these markers to avoid shallow areas for safety reasons (to prevent their boats from striking the bottom), there is little risk that the boats will be in these shallow waters and interact with species in these waters. Therefore, interactions with smalltooth sawfish, Nassau Grouper, and sturgeon are extremely unlike to occur and effects to those species from vessel strikes are discountable.

As stated above, vessel traffic can adversely affect listed whales (specifically the North Atlantic right whale) and sea turtles. Construction of pile-supported structures (Activity 2) and boat ramps (Activity 6) analyzed in this Opinion may indirectly result in increase in vessel use by introducing additional and increased shore-based transfer and support facilities. As such, the potential exists for adverse effects to these species from a possible increase in vessel usage. However, these activities are not expected to result in an increase in the number of vessels beyond the range identified in the following analysis.

The Florida Department of Highway Safety and Motor Vehicle website reports the total number of registered vessels per year in Florida over the past 16 years. The annual number of total registered vessels in Florida has varied up to 14% between the highest number of registered vessels in 2007 and the lowest number in 2000 and still has not returned to the highest number of registered vessels that occurred from 2005-2008. Even though it has fluctuated, the general trend has remained stable and unchanged. To confirm this, we conducted a statistical analysis of the data using a method called a one-way analysis of variance (also referred to as a one-way ANOVA) to access the potential differences in the total numbers of all registered vessels in 35 Florida coastal counties compared to the year (2005-2015). There was no significant difference in numbers of registered vessels in Florida coastal counties for the years 2005-2015 (based on the input values of alpha=0.05, F(10,374) = 0.177, p=0.99). We also compared various data sets to determine if distribution of registered vessels in Florida changed in a particular region. For this, we looked at the number of vessels registered per year in all Florida counties, in coastal counties only, in coastal counties that border Johnson's seagrass critical habitat only, in coastal counties that boarder North Atlantic right whale critical habitat only, and in coastal counties that boarder North Atlantic right whale critical habitat only where the vessels registered were classified as size Class 2 (26 ft to 39 ft 11 in) and Class 3 (40 ft to 64 ft 11 in). We reviewed the number of vessels registered in North Atlantic right whale critical habitat that were classified as
size Class 2 and 3 because the size and speed of a vessel that may strike a North Atlantic right whale can affect the survivability of the whale. As we noted in the final rule listing the species, right whales struck by vessels over 65 ft in length and travelling at a certain speed may suffer mortality. 73 FR 60173, October 10, 2008. For this reason, vessels over 65 ft in length must comply with speed restrictions that reduce the likelihood of fatal collisions with right whales. These speed restrictions apply in specific locations, primarily at key port entrances, and in seasonal management areas at certain times, although there is an exception when navigational safety requires a deviation. Thus, we did not isolate vessels of this size because the speed restrictions reduce the potential for vessel strike. In addition, these larger vessels are included in our analysis of vessels in and adjacent to North Atlantic right whale critical habitat. Although we limited our speed restrictions to vessels over 65 ft in length, in the final rule, we recognized that right whales had been struck and killed by vessels in the 40-65 ft class, although but death occurred in just 2 of the 8 cases studied. We also noted a single reported case of a 33.7 ft vessel that struck and killed a right whale calf in Australia in 2009. Under our Dynamic Management Area program, these smaller vessels are requested, but not required, to either travel at 10 knots or less or route around locations when certain aggregations of right whales are detected outside seasonal management areas, but since those measures are voluntary, we are not assuming that they will be followed. Thus, because it is possible that vessels 33.7 ft in length could strike and kill right whales, we isolated the number of registrations for Class 2 (26 ft to 39 ft 11 in) and Class 3 (40 ft to 64 ft 11 in) vessels registered in counties that border North Atlantic Right whale critical habitat.

The number of registered vessels per year in each of these regions, and of the various sizes as discussed above, is shown in Figure 7 below. None showed statistically significant fluctuations in the number of registered vessels. Any minor fluctuations in registered vessels are likely linked to external factors including economic conditions and gas prices, which will continue to fluctuate throughout time. Thus, even though new vessels continue to be sold throughout the state of Florida and new structures continue to be constructed to store those vessels, the total number of registered vessels in Florida has remained relatively static, as discussed above. We assume that this is because (1) most old vessels are ultimately replaced with new vessels; (2) many new docks and marinas likely also replace older structures so a permit to construct a new structure does not necessarily mean the new structure will support new vessels; and (3) vessels stored at 1 location may be relocated, but are not new to the state of Florida. Also, new residential docks may be built that do not support vessel dockage or storage, or that change the location of existing vessels. In fact, in a 5-year period from 2010-2014, the USACE estimates that it issued 11,412 permits for new docks and marinas in Florida, yet the total number of registered vessels in Florida dropped by 41,954 during the same time period. This indicates that the number of registered vessels in Florida is not directly related to the number of permitted new structures. With respect to relocation and distribution, we conducted the analysis above to determine whether the vessels were being relocated to a particular area, including areas sensitive to additional vessel traffic. We did not find a statistically significant change in the number of vessels in a particular area. We do not have data about the number of registered vessels in the U.S. Caribbean; however, we assume that this consistent level of vessel registrations would also apply in the U.S. Caribbean since we would expect the Caribbean to be affected by the same factors that keep overall vessel numbers constant despite increasing dock construction (new docks may just relocate vessels, or replace older structures).

Therefore, we believe installation, repair, and replacement of pile-supported structures (Activity 2) and boat ramps (Activity 6) do not contribute to an increase the number of registered vessels in the State of Florida or the U.S. Caribbean. Based on the discussion above, we believe that it is extremely unlikely that the proposed action will increase the number of registered vessels or the level of vessel traffic in the action area, and thus it is extremely unlikely that these projects will increase the incidence of vessel strike. Therefore, the potential effect of vessel strikes resulting from Activities 2 and 6 analyzed in this Opinion on whales and sea turtles is discountable.

Moreover, regarding the potential for vessel strike of the North Atlantic right whale, though the addition of structures that support vessels analyzed in this Opinion is unlikely to result in additional vessels in Florida generally or in areas where North Atlantic right whales occur specifically, we did provide additional measures in this Opinion for the protection of the species. In particular, within the North Atlantic right whale educational sign zone (see Section 2.1.1.4), PDC A2.2 requires all commercial, multi-family, or public docking facilities to post educational signs alerting boaters of the presence of North Atlantic right whales. In addition, PDC A2.4 provides that private home owners proposing to construct, repair, or replace a dock structure within 11 nmi of North Atlantic right whale critical habitat will be provided a handout with their USACE permit describing the presence of North Atlantic right whales in the area and the Federal regulations governing the approach to North Atlantic right whales (Appendix C). These measures further reduce the risk of interaction.



Figure 7. Number of registered vessels in Florida from 2005 to 2015 (Florida Department of Highway Safety and Motor Vehicle website: http://www.flhsmv.gov/dmv/vslfacts.html).

#### 7. Noise

We believe that the noise generated by the minor in-water activities analyzed in this Opinion, including the noise generated by the installation of piles and sheet piles associated with shoreline stabilization (Activity 1) and pile-supported structures (Activity 2), may affect ESA-listed sea turtles, smalltooth sawfish, sturgeon, Nassau grouper, and the Central and Southwest Atlantic DPS of scalloped hammerhead shark. Other activities addressed in this Opinion that may involve the installation of piles or sheet piles include outfall structures (Activity 4), scientific survey devices (Activity 5), boat ramps (Activity 6), aquatic enhancement (Activity 7), aerial transmission lines (Activity 8), and platforms and cofferdams (Activity 10); however, because the portion of these activities requiring pile and sheet pile installation must comply with the PDCs for Activities 1 and 2, we are not separately analyzing the noise from pile driving associated with those activities. Our effects determinations related to in-water construction noise relies on and updates the analysis that we completed for NMFS's Programmatic Biological Opinion on SAJ-82 for Monroe County, Florida (NMFS 2014b). This Opinion considers the effect of noise on all species under NMFS's jurisdiction that are found within the action area of this Opinion, including sturgeon and the recently listed Nassau grouper, along with an explanation of the calculations and thresholds used to make these effects determinations (Appendix B of this Opinion).

Installing piles and sheet piles generates sound waves that can result in physical injuries to sea turtles and fish (e.g., smalltooth sawfish, sturgeon, Nassau grouper, and the Central and Southwest Atlantic DPS of scalloped hammerhead shark) or change their behavior, as explained in more detail below. In order to determine the hydroacoustic effects on the species from the installation of piles and sheet piles (generally referred to as noise effects in this Opinion), we need to calculate the area in which the noise levels will be elevated (also known as the ensonified zone) as well as the portion of the ensonified zone in which the species may experience potentially adverse behavioral effects (the behavioral zone), potentially adverse physical injuries (physical injury zone), and potentially adverse physical injuries from repeated exposure to noise (cumulative injurious zone). The physical injury zone is the area where the species would experience injury from a single strike, or single loud noise event, and the cumulative injurious zone is a larger area where the species could experience physical injury if it were exposed to noise over time.

Calculating the size of these zones requires knowing the material that the pile or sheet piles are made of (e.g., wood, concrete, metal), the size of the pile or sheet pile (i.e., diameter, width, length), the equipment that will be used to install it (e.g., sledge hammer, vibratory hammer, impact hammer), and the number of strikes or vibratory time required to install the pile or sheet pile. The PDCs in this section limit the types and sizes of materials that can be used and the allowable installation methods.

Calculating the effects of in-water noise on marine species within these zones requires knowing where the pile or sheet pile will be installed. Section 2.1.1.2 states the type of pile driving activities that can occur in Gulf sturgeon critical habitat migratory restriction zones and these limitations are implemented in PDC A1.9 and PDC A2.11 for Activities 1 and 2 respectively. In addition, no pile driving activities can occur in the smalltooth sawfish critical habitat limited exclusion zones (PDC A2.12).

Another consideration in evaluating the effects noise may have on species is whether the pile driving, and thus the noise generated, will occur in open water or in confined spaces. This differentiation is important. If a project occurs in a confined space, the animal may not be able to avoid the physical injury, cumulative injurious, and/or behavioral zones or may become trapped in an area (e.g., the terminal end of a canal) because those zones block the only escape route. Conversely, in an open-water environment, the animal would be able to move freely away from the noise without being forced to move through or stay in areas where the noise levels over time could cause injury. For our noise analysis, we define a *confined space* as any area that has a solid object (e.g., shorelines or seawalls) that creates a constricted passage area such that species attempting to move through the area would be forced to pass through the cumulative injurious zone (generally, the largest of the zones). To allow species to move through the project areas without being exposed to noise at levels that could be injurious over time, the PDCs limit certain pile types and installation methods in these confined spaces to ensure that the cumulative injurious zone is limited to a maximum of half the width of the confined area.

In Florida, we consider the confined space to be any area that has a solid object (e.g., shorelines or seawalls) within 150 ft of the pile or sheet pile installation site. In the U.S. Caribbean, we consider confined space to be any area that has a solid object within 260 ft of the pile or sheet pile installation site. These confined space distances were calculated by doubling the cumulative injurious zone for the large fish found in Florida and the smaller fish found in the U.S. Caribbean (calculations provided in Tables 18 and 19, and an example of a confined space shown in Figure 8). As discussed in Appendix B, the effects of in-water noise on fish is different for small fish and large fish. Fish weighing less than 2 grams are more susceptible to injury from impulsive noises, like impact pile driving, than fish weighing more than 2 grams. Similarly, fish weighing less than 102 grams are more susceptible to non-impulsive noises, like vibratory pile driving, than fish weighing at least 102 grams. The only fish in the action area that we expect to be less than 102 grams and less than 2 grams are newly recruited and small juvenile Nassau grouper in the U.S. Caribbean, as discussed in the Nassau grouper section above. Thus, we have provided a larger confined space zone in the U.S. Caribbean to be protective of the smaller fish that might be found there. Because the cumulative injurious zone for large and small fish is the same or greater than the cumulative injurious zone for sea turtles, these zones are protective of both sea turtles and fish. In particular, they ensure that a fish or sea turtle would have at least half the width of the escape route free from cumulative injurious noise so that they could move through the area as needed without being exposed to potentially injurious noise levels over time. Docks or other pile-supported structures do not stop or reflect noise and do not create a confined space.



Figure 8. Example of a confined space (left) and an open-water environment (right) for a noise analysis. The left image shows a man-made residential canal and a 150 ft sound radius that would block movement of species if piles were installed at this location because the canal is not wide enough to ensure more than half the escape route is free from cumulative injurious noise levels. The right image shows a portion of San Juan Harbor and a 260 ft sound radius that would not block movement of species if piles were installed at this location because the canal is not wide enough to ensure more than half the escape route is free from cumulative injurious noise levels. The right image shows a portion of San Juan Harbor and a 260 ft sound radius that would not block movement of species if piles were installed at this location because species would be able to move freely away without passing through the area with cumulative injurious noise levels.

#### Project Design Criteria

Since the effects of in-water noise from pile and sheet pile installation are dependent on the type and size of material installed and the manner and location in which it is installed, the PDCs for pile and sheet pile installation are based on those factors, and are broken into different categories identified as A-E below. The tables below identify the PDC category for the type of activity, depending on whether it occurs in an open water environment or in a confined space. If the proposed project includes pile sizes outside the range of those described in the table below, then installation of those piles are not covered under this Opinion (e.g., installation of concrete piles larger than 24-in diameter are not covered under this Opinion).

## PDCs for In-Water Noise from Pile and Sheet Pile Installation

## **Open Water**

The letters A-E in the tables below specify the PDC category. Activities labeled A-D must follow the corresponding PDCs for labeled Category A-D below. Activities labeled E are excluded from this Opinion, as stated in Category E below.

	Trench and fill	Pilot hole (auger or drop punch)	Jetting	Vibratory	Impact hammer
Wood piles 14-inch (in) diameter or less when installed via impact hammer and 36-in or less for all other installation methods	А	А	А	А	В
Concrete pile 24-in diameter/width or less in open water	А	А	А	А	В
Metal pipe pile 36-in diameter or less	А	А	А	А	Е
2 metal boatlift I-beams	Α	А	А	А	В
Concrete slab wall- any size	Α	А	Α	А	В
Vinyl sheet pile- any size	Α	А	А	А	В
Metal sheet pile- any size	Α	А	А	А	Е

## **Confined Space**

In Florida, we consider the confined space to be any area that has a solid object (e.g., shorelines or seawalls) within 150 ft of the pile installation site and in the U.S. Caribbean we consider confined space to be any area that has a solid object within 260 ft of the pile installation site.

	Trench and fill	Pilot hole (auger or drop punch)	Jetting	Vibratory	Impact hammer
Wood pile 14-in diameter or less when installed via impact hammer and 36-in or less for all other installation methods	А	А	А	А	В
Concrete pile 24-in diameter/width or less (5 piles or less installed/day)	А	А	А	А	С
Concrete pile 24-in diameter/width or less (6-10 piles installed/day)	А	А	А	А	D
Metal pipe pile 36-in diameter or less	А	А	А	А	Е
2 metal boatlift I-beams	А	А	Α	А	В
Vinyl sheet pile – any size	А	А	Α	А	В
Concrete slab wall- any size (5 slabs or less installed/day)	А	А	А	А	С
Concrete slab wall- any size (6-10 slabs installed/day)	А	Α	А	А	D
Metal sheet pile- any size	A	А	A	A	E

# A. The Projects identified as <u>A</u> above must comply with PDCs identified for all projects in this Opinion. Specific PDCs related to noise include:

- 1. All work must occur during daylight hours only (PDC AP.6).
- 2. All construction personnel are responsible for observing water-related activities to detect the presence of these species and avoid them (PDC AP.7).
- B. The projects identified as <u>B</u> above must follow <u>all of the conditions under A</u>, above, <u>AND</u> also must limit the maximum number of piles installed per day to no more than 10 piles per day.
- C. The projects identified as <u>C</u> above must follow <u>all of the conditions under A</u>, above, <u>AND</u> also must limit the maximum number of piles installed per day to no more than 5 piles per day.
- **D.** The projects identified as <u>D</u> above must follow <u>all of the conditions under A and B</u>, above, <u>AND</u> also must abide by one of the noise abatement measures below, as chosen by the applicant:
  - 1. Bubble curtain: The bubble curtain design must adhere to the guidelines for unconfined and confined bubble curtains described in Appendix B.
  - 2. Temporary noise attenuation pile (TNAP) also known as a pile isolation casing: The TNAP design must be constructed of a double-walled tubular casing (a casing within a larger casing), with at least a 5-in-wide area between the casings that is dewatered to create a hollow space or 5-in wide area between the casings completely filled with closed-cell foam or other noise dampening material between the walls. The TNAP must be long enough to be seated firmly on the sea bottom, fit over the pile being driven, and extend at least 3 ft above the surface of the water.
  - **3.** The use of any other alternative noise control method must receive prior approval by NMFS and the USACE, as described in Section 2.3.
- E. The projects identified as  $\underline{E}$  are not covered under this Opinion.

Noise created by construction activities can physically injure or change animal behavior in the affected areas. Thus, we will evaluate 2 general categories of effects to listed species as a result of noise: physically injurious effects and behavioral effects. We discuss behavioral responses to noise first below because certain behavioral responses (i.e., avoidance) could affect whether the animal experiences physical injury because of noise.

## Behavioral responses from in-water noise associated with pile installation

Animals may have varied behavioral responses to noise, depending on the level of noise, season, location, habitat, and life stage exposed to the noise. Different species also have different life histories and sound-detection capabilities that could influence their response, in addition to a different set of potential behavioral responses that may be triggered by a given stimulus (e.g., some may try to hide in the area while others may swim away). Although the behavioral

response of noise avoidance is advantageous at preventing potential injury from the exposure to noise, avoidance behavior may disrupt or interfere with feeding, mating, migration, sheltering, or may increase the risk to individuals (e.g., via predation). Not all individuals are likely to have an avoidance response. Despite exposure to noise that may cause others to move away and abandon the area altogether, some individuals may be biologically motivated to remain in a habitat for feeding, sheltering, mating, or other biologically important reasons or because they are using the area as an established pathway between habitats.

To determine the potential effects from exposure to noise, it is important to understand the ways in which ESA-listed species in the action area may respond to the noise. Responses to pile or sheet pile driving from specific species in the action area are not well documented. However, we expect that species will avoid areas with construction activities that include the combination of the physical movement and presence of construction equipment, presence of turbidity curtains, general construction related noise and specifically louder noises generated from in-water pile and sheet pile installation. Specifically,

- Sea turtles: Studies have reported sea turtles responding to air gun noise and another loud impulsive noise source in the ocean by demonstrating alarm or avoidance behaviors (DeRuiter and Doukara 2010; McCauley et al. 2000a; McCauley et al. 2000b). Therefore, we would expect sea turtles would avoid in-water construction noise.
- Sturgeon: Krebs et al. (2012) reported that most tagged Atlantic sturgeon are likely to avoid areas during periods of high construction noise. As such, we would expect shortnose and Gulf sturgeon would also avoid in-water construction noise.
- Smalltooth sawfish and the Central and Southwest Atlantic DPS of scalloped hammerhead shark: No studies have evaluated how sawfish respond to elevated noise levels; however, Myrberg (2001) reported that sudden increases in sound beginning at 20 decibels above ambient caused shark (including hammerhead shark) to move away from a sound source. Since shark and sawfish are in the same Subclass (i.e., Elasmobranchs), we would expect sawfish to have a similar response to noise as shark.
- Juvenile fish: Smaller fish found in the action area (small juvenile Nassau grouper, juvenile smalltooth sawfish,<sup>13</sup> and juveniles of the Central and Southwest Atlantic DPS of scalloped hammerhead shark) may be less willing to move large distances than larger, adult fish due to

<sup>&</sup>lt;sup>13</sup> Juvenile sturgeon also are less than 2 grams when in riverine habitat. However, the action area does not contain rivers supporting juvenile sturgeon that are under NMFS jurisdiction. Gulf sturgeon are under the jurisdiction of the USFWS when in riverine habitat. Within Florida and the U.S. Caribbean, juvenile Atlantic and shortnose sturgeon are only known to occur in the St. Johns and St Marys River. Due to man-made structures and alterations, spawning areas in the St. Johns River are not accessible and therefore do not support a reproducing population (81 FR 36077). Also, PDC AP.4 does not allow projects to be covered under this Opinion in the St. Marys River, thereby excluding the St. Mary's River from the action area.

associated increase in predation risk. However, we believe even smaller fish will move at least short distances to avoid both the physical commotion and noise of in-water construction. Nassau grouper are the smallest species in our action area that could be exposed to noise from activities covered under this Opinion and the only species in the action area that could be small enough to be more susceptible to injurious noise effects. They are known to spawn in off-shore areas in the U.S. Caribbean so we would expect that juvenile Nassau grouper will recruit to nearshore areas in the U.S. Caribbean. Juvenile smalltooth sawfish may also be found throughout Florida, especially in areas of smalltooth sawfish critical habitat in southeast Florida. A few records of juvenile scalloped hammerhead shark likely from the Central and Southwest Atlantic DPS have been reported in the U.S. Caribbean through hook-and-line capture data. Smaller fish are more susceptible to predation than larger fish so they must determine if moving to avoid a potential threat outweighs the risk of staying in a preferred location such as nearshore seagrass and algae beds. Smaller fish are also biologically more susceptible to physical injury from sound exposure and may need to move further than larger fish to avoid noise that could cause physical injury. Still, we believe that all small fish in the action area (juvenile smalltooth sawfish or scalloped hammerhead shark and even the smallest, small juvenile Nassau grouper that have just recruited to nearshore areas), would move away from an injurious noise source. Though movement to avoid sound exposure could increase their risk of predation, it is likely that predatory species (e.g., larger fish) would also be moving away from the noise source instead of foraging on these smaller species in the area during construction, making predation unlikely.

These behavioral responses to noise inform our effects analysis below.

#### Potential effects from in-water noise associated with pile and sheet pile installation

As stated earlier, noise generated during the installation of pile and sheet piles can cause physical injury or result in behavioral effects. Physically injurious effects from noise can occur in 2 ways. First, effects can result if a single noise event occurs that exceeds the peak pressure threshold for direct physical injury to an animal. This would result in an immediate adverse effect on the animal. Second, effects can result from prolonged exposure to noise levels that exceed the daily cumulative sound exposure level (cSEL) threshold for the animal, which results in a physical injury to the animals if exposed to the noise levels for sufficient periods.

#### Physical Injury

Physical injury can range from minor physiological effects to physiological effects that could potentially result in mortality. Potential physiological effects to both fish and turtles are highly diverse, and range from very small ruptures of capillaries such as in fish fins (which are relatively minor physical responses) to severe hemorrhaging of major organ systems such as the liver, kidney, or brain (Hastings and Popper 2005) which may ultimately result in mortality. Other potential effects include rupture of the swim bladder, which is the air-filled organ in the abdominal cavity of most bony fish species that is involved in maintenance of buoyancy. Both sturgeon and Nassau grouper have swim bladders that are susceptible to rupture from noise. Both smalltooth sawfish and scalloped hammerhead shark, like all elasmobranchs, do not have swim bladders; their oversized liver provides buoyancy, and, as such, it may be susceptible to physical injury. ESA-listed species within the physical injury zone can suffer physical injury as described above and can also suffer from a reduction in their ability to hear (another form of physical injury), which can decrease their ability to detect and avoid predators, decrease their ability for passive listening to detect prey species, interfere with the detection of conspecifics (in vocal species), and inhibit their overall ability to detect cues in the acoustic soundscape. Permanent hearing loss (loss of hearing) or permanent threshold shift (change in hearing thresholds and frequencies), is the irreversible loss of hearing abilities (either total or partial). Permanent hearing loss and permanent threshold shift and other physical injuries could have negative consequences for an animal. Temporary hearing loss and temporary threshold shift are temporary hearing impairments from exposure to loud noises that are recoverable with time (hours to days). Neither is considered a permanent physical injury, but both could have some temporary effects to the auditory sense of an animal.

Physical injury can also result in adverse behavioral effects. An example of a behavioral effect resulting from a physical injury is if the temporary or permanent damage to hearing affected a species' ability to forage. Elasmobranch hearing sensitivity has been correlated with feeding behavior (Corwin 1977). Research has shown that pelagic, predatory sharks (such as the lemon shark (Nelson and Gruber 1963) and the bull shark (Kritzler and Wood 1961)) who feed on bony fishes have more sensitive low-frequency hearing than bottom-dwelling elasmobranchs (such as the little skate (Casper et al. 2003) and the horn shark (Kelly and Nelson 1975) that feed on other bottom-dwelling species (such as molluscs and crustaceans). The smalltooth sawfish is a mostly bottom-dwelling elasmobranch closely related to the little skate, and, like the little skate, would be expected to have less sensitive low frequency hearing and, therefore, rely more on senses that specialize in nearby detection such as electroreception, smell, and vision. Scalloped hammerhead shark would be expected to have more sensitive low-frequency hearing like other sharks. Currently, we do not understand precisely how sea turtles, smalltooth sawfish, scalloped hammerhead shark, sturgeon, and Nassau grouper use auditory cues in their environment and if temporary or permanent threshold shifts would alter their behavior or physiology in a way that was notable to the exposed individual.

We do not expect ESA-listed sea turtles and fish to suffer physical injury from the noise generated during pile and sheet pile installation covered under this Opinion. We also do not believe that it will cause temporary or permanent hearing loss or threshold shifts to these species that would cause physiological or behavioral changes in important life functions such as foraging.

As explained above and in Appendix B, physically injurious noise can occur from exposure to the noise generated during a single-strike (peak pressure) or the exposure to noise generated over the course of a day (as measured by cSEL).

• Physical injury (single-strike): For the types of pile and sheet pile installation analyzed in this Opinion, we would only expect a peak pressure injury from impact driving of metal piles or metal sheet piles; therefore, these are the only pile installation activities that are excluded from this Opinion, as described in the PDCs for *In-water Noise from Pile and Sheet Pile Installation* (provided above and listed as not covered, Category E) (see Appendix B for the

calculations supporting this conclusion). None of the activities covered under this Opinion (as restricted by the PDCs of this Opinion) will result in single-strike injury to any of the species in the action area.

• Physical injury (cumulative exposure): To determine if ESA-listed sea turtles or fish in the action area would be affected by noise exposure over the course of a full day of construction (cSEL), we rely on the information provided above regarding how the species in our action area respond to noise (section entitled "Behavioral responses from in-water noise associated with pile installation"). As discussed above, we believe that all of these species will avoid the cumulative exposure area by moving out of the way of the noise. Because we expect mobile species to move away from the sound exposure, we believe it is extremely unlikely that the species would be exposed to the noise for the amount of time required to cause injury and that the effect from noise would be discountable.

In addition, for areas within the action area that we consider to be biologically significant (e.g., known areas used for reproduction or migration), additional PDCs are provided to ensure that noise from in-water construction does not interfere with species in those biologically significant areas, as described by species below.

- Smalltooth sawfish: Pile driving activities cannot occur in areas known to support smalltooth sawfish pupping (smalltooth sawfish limited exclusion zones, defined in Section 2.1.1.1). If an otherwise covered pile driving activity is proposed in the smalltooth sawfish limited exclusion zone, the USACE must separately consult on that activity given the location.
- Gulf sturgeon: Only the certain pile installation types that we believe will not cause harm to Gulf sturgeon are allowed in the Gulf sturgeon critical habitat migratory restriction zones, defined in Section 2.1.1.2. The USACE must separately consult on all other activities.
- Sea turtles: The Opinion does not apply to construction on or near sea turtle nesting beaches under the activity-specific PDCs for shoreline stabilization (Activity 1, PDC A1.7). If an otherwise covered shoreline stabilization activity is proposed in those areas, the USACE must separately consult on that activity given the location. Activity 2 allows only the installation of ATONS near sea turtle nesting beaches (Activity 2, PDCs A2.15 and A2.16). We believe that installing ATONs, which are structures limited in size to only a single pile or buoy, does not restrict access to nesting beaches. All other pile supported structures in these areas require separate consultation.
- Nassau Grouper are soniferous fishes (i.e., noise producing), and males make specific sounds related to courtship activities (Appeldoorn et al. 2013). Nassau grouper spawn in aggregation sites in the U.S. Caribbean (as well as in other areas outside of the action area of this Opinion). Therefore, in-water construction in those areas could interfere with mating; however, spawning aggregation sites are located in off-shore areas, away from nearshore construction activities, such as construction of docks and seawalls that may

involve the noise-generating pile driving activities analyzed here. In addition, Nassau grouper spawn at night, and, under the PDCs for *In-Water Noise from Pile and Sheet Pile Installation*, all work must occur during daylight hours.

• The Central and Southwest Atlantic DPS of scalloped hammerhead sharks: At this time, we are unaware of any biologically important areas in the action area used by this species.

Therefore, we believe that the PDCs above will protect these species in biologically important areas, and thus that the noise generated during pile and sheet pile installation will have no effect on mobile species in biologically significant areas.

#### **Behavioral Effects**

Behavioral effects from noise can result if the species is exposed to noise at levels that could cause behavioral responses (within the behavioral effects zone), or the noise caused the species to change its behavior or interferes with biologically important behaviors. Species movement and migration is a factor when evaluating the behavioral effects of noise exposure in a specific project location. Certain project construction locations could impede the species' ability to move away from or around the sound source, and that is why we also consider if the project would occur in an open-water or confined location. In a confined location (described above), the species could be trapped and exposed to the noise throughout the course of the day's construction (construction is limited to daylight hours), resulting in exposure to sound that could cause behavioral responses (in the behavioral effects zone) or could interfere with foraging or other biologically important activities.

By limiting the work in these confined areas and limiting work to only daytime hours, species will be able to move to avoid any behavioral effects from the exposure to the noise in the behavioral effects zone by moving through or away from the area during quiet periods at night. The species are highly mobile, and we expect them to move away from the behavioral effects zone, and thus, such effects would be extremely unlikely to occur and discountable. In addition, since they will be able to leave or maneuver around the behavioral effects zone, they will be able to safely move to nearby similar habitat for foraging and refuge or move onto important biologically significant areas. Because we expect the species to be able to safely relocate to similar areas, the effect of being forced to move is insignificant. For the same reason, we believe species' movement away from the cumulative injury zone for physical injury (discussed above) will have an insignificant effect on the species.

In addition, as discussed in the preceding section, for areas within the action area that we consider to be biologically significant (e.g., known areas used for reproduction or migration), additional PDCs are provided to ensure that noise from in-water construction does not interfere with important biological functions and will have no effect on species by interfering with their ability to use these biologically significant areas as described by species.

#### Summary of Physically Injurious and Behavioral Effects

We believe that there will be no direct physical effects from exposure to peak pressure noise from pile or sheet pile driving allowed under this Opinion since the Opinion does not apply to activities that could result in peak pressure injury. Additionally, we believe the cumulative injury zones are limited in size and location to allow ESA-listed fish and sea turtles to move around or avoid the zones, thus limiting the species exposure. As described above, these species are not expected to stay within the cumulative noise zones during construction and therefore we believe it is extremely unlikely that they will suffer injury, and the effect is discountable (Table 20).

We also believe that noise from pile driving allowed under this Opinion is not likely to adversely affect ESA-listed fish and sea turtles from exposure to behavioral noise zones (Table 20). The species are highly mobile and can avoid these zones, making it extremely unlikely that they will experience behavioral impacts, and thus the effect is discountable. The noise may change the animal's behavior—causing it to move to avoid physical injury or behavioral responses from noise exposure. Triggering an avoidance behavior is an effect on the species, but we believe this effect is insignificant as well. Pile driving cannot occur in areas identified as biologically important to these species (Section 2.1.1), thus, we do not expect exposure to noise to cause the animals to alter any important life functions, and we believe these highly mobile species will be able to relocate to areas of similar habitat outside of the construction zones.

#### No Effect

Pile installation by jetting, using an auger or drop punch to create a pilot hole, or installing Ibeam boatlifts using a vibratory hammer will not result in physical injury (injury from exposure to peak pressure or cumulative sound exposure) or behavioral noise effects, because it will not create noise levels in excess of the respective thresholds for physical injury to, or behavioral responses in, sea turtles and ESA-listed fishes (Table 20). The methods used to determine the sound expected from these installation methods, and the final calculations, are provided in Appendix B.

## Calculations for Physically Injurious and Behavioral Effects from Pile and Sheet Pile Installation

Tables 18-19 below provide a summary of the radii at which ESA-listed sea turtles and fish (smalltooth sawfish, scalloped hammerhead shark, sturgeon, and Nassau grouper) will experience physical peak pressure injury, physical injury from cumulative exposure to noise, or behavioral effects at given noise thresholds from the pile and sheet pile installation analyzed under this Opinion. These calculations are explained in more detail in Appendix B. As noted at the outset, of the types of pile and sheet pile installation analyzed in this Opinion, we would only expect single strike (peak pressure) injury from impact driving of metal piles or metal sheet piles; therefore, these activities are excluded from this Opinion as described in the PDCs for *Inwater Noise from Pile and Sheet Pile Installation* (provided above and listed as not covered, category C).

	Source Level (dB re 1 µPa <sup>14</sup> )	Radius for Fish less than 2 grams	Radius for Fish over 2 grams	Radius for Sea Turtles
Calculate	d 10 pilos ins	14-in wood pile and viny	yl sheet yn nou pilo - 450 stuikae	non dou
	a to phes ms	talled per day with 45 strik	tes per plie – 450 strikes	s per day
(peak pressure)	Peak	0 m (1 ft)	0 m (1 ft)	0 m (1 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 4 m (12 ft) 5 piles = 11 m (36 ft) 10 piles = 17 m (56 ft)	1 pile = 2 m (7 ft) 5 piles = 6 m (20 ft) 10 piles = 9 m (30 ft)	1 pile = $2 m (7 ft)$ 5 piles = $6 m (20 ft)$ 10 piles = $9 m (30 ft)$
Behavior (Root Mean Square [RMS])	185 dB RMS	215 m (705 ft)	215 m (705 ft)	46 m (151 ft)
		24-in concrete pil	e	
Calculated	10 piles inst	alled per day with 160 strik	tes per pil <mark>e</mark> = 1,600 strik	es per day
Physical Injury (peak pressure)	200 dB Peak	0 m (1 ft)	0 m (1 ft)	0 m (1 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 9 m (28 ft) 5 piles = 25 m (83 ft) 10 piles = 40 m (131 ft)	1 pile = 5 m (16 ft) 5 piles = 14 m (46 ft) 10 piles = 22 m (72 ft)	1 pile = 5 m (16 ft) 5 piles = 14 m (46 ft) 10 piles = 22 m (72 ft)
Behavior (RMS)	185 dB	215 m (705 ft)	215 m (705 ft)	46 m (151 ft)
	Tw	o 12-in metal boat lift I-bea	am (H-pile) <sup>15</sup>	
Calculated	l 2 piles insta	lled per day with 660 strike	es per pile = 1,320 strike	es per day
Physical Injury (peak pressure)	205 dB Peak	1 m (3 ft)	1 m (3 ft)	1 m (3 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = $22 \text{ m} (72 \text{ ft})$ 2 piles = $35 \text{ m} (115 \text{ ft})$	1 pile = $12 \text{ m} (39 \text{ ft})$ 2 piles = $19 \text{ m} (62 \text{ ft})$	1 pile = $12 \text{ m} (39 \text{ ft})$ 2 piles = $19 \text{ m} (62 \text{ ft})$
Behavior (RMS)	190 dB RMS	465 m (1,526 ft)	465 m (1,526 ft)	100 m (328 ft)
		24-in metal sheet p	ile	-
Calculated 10	sheet piles in	nstalled per day with 660 st	rikes per pile = 6,600 st	rikes per day
Physical Injury (peak pressure)	220 dB Peak	9 m (30 ft)	9 m (30 ft)	9 m (30 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 410 m (1,345 ft) 10 piles = 858 m (2,815 ft)	1 pile = 223 m (732 ft) 10 piles = 858 m (2,815 ft)	1  pile = 223  m (732  ft) 10 piles = 858 m (2,815 ft)
Behavior (RMS)	204 dB RMS	858 m (2,8215 ft)	858 m (2,8215 ft)	185 m (607 ft)

## Table 18. Impact Hammer Sound Source Levels and Impact Radius Distances

 $<sup>^{14}\,</sup>dB$  re 1  $\mu Pa$  is a unit of measurement of sound in decibels relative to 1 micro-Pascal-squared second

<sup>&</sup>lt;sup>15</sup> Noise levels not believed to be accurate based on the installation method used. Boatlift I-beams only penetrate loose sediment until they reach the top of, or first few inches of, hard substrate to stabilize the structure on the hard substrate versus penetrating it.

Source Level (dB re 1 µPa)	Radius for Fish ≥102 grams	Radius for Sea Turtles
36-in wood, concrete, f piles for 8 hours per day (no	vinyl, or metal piles limit on the number of p	oiles per day)
195 dB Peak	0 m (0 ft)	0 m (0 ft)
cumulative	0 m (0 ft)	0 m (0 ft)
185 dB RMS	215 m (705 ft)	46 m (151 ft)
Two 12-in metal boat li ed per day for 30 minutes (1,80	ift I-beam (H-pile) <sup>16</sup> 00 seconds) per pile = 3,6	i00 seconds per day
165 dB Peak	0 m (0 ft)	0 m (0 ft)
cumulative	0 m (0 ft)	0 m (0 ft)
150 dB RMS	0 m (0 ft)	0 m (0 ft)
24-in metal sheet piles for 8 hours per da	sheet pile y (no limit on the numbe	r of sheet piles per day)
192 dB Peak	0 m (0 ft)	0 m (0 ft)
cumulative	0 m (0 ft)	0 m (0 ft)
178 dB RMS	74 m (243 ft)	16 m (52 ft)
	Source Level (dB re 1 μPa)         36-in wood, concrete,         f piles for 8 hours per day (no         195 dB Peak         cumulative         185 dB RMS         Two 12-in metal boat I         ed per day for 30 minutes (1,8)         165 dB Peak         cumulative         150 dB RMS         24-in metal         sheet piles for 8 hours per da         192 dB Peak         cumulative         192 dB Peak         193 dB RMS	Source Level (dB re 1 µPa)Radius for Fish ≥ 102 grams36-in wood, concrete, vinyl, or metal piles f piles for 8 hours per day (no limit on the number of p 195 dB Peak0 m (0 ft)195 dB Peak0 m (0 ft)cumulative0 m (0 ft)185 dB RMS215 m (705 ft)Two 12-in metal boat lift I-beam (H-pile) 16d per day for 30 minutes (1,800 seconds) per pile = 3,60165 dB Peak0 m (0 ft)cumulative0 m (0 ft)150 dB RMS0 m (0 ft)24-in metal sheet pilesheet piles for 8 hours per day (no limit on the number 0 m (0 ft)192 dB Peak0 m (0 ft)192 dB Peak0 m (0 ft)178 dB RMS74 m (243 ft)

Table 19. Vibratory Hammer Sound Source Levels and Impact Radius Distances

<sup>&</sup>lt;sup>16</sup> Noise levels not believed to be accurate based on the installation method used. Boatlift I-beams only penetrate loose sediment until they reach the top of, or first few inches of, hard substrate to stabilize the structure on the hard substrate versus penetrating it.

	Trench and fill	Pilot hole (auger or drop punch)	Jetting	Vibratory	Impact hammer
Wood pile 14-in diameter or less installed via impact hammer and 36-in or less for all other installation methods	NE	NE	NE	NLAA	NLAA
Concrete pile 24-in diameter/width or less (5 piles or less installed/day)	NE	NE	NE	NLAA	NLAA
Concrete pile 24-in diameter/width or less (6-10 piles installed/day)	NE	NE	NE	NLAA	NLAA
Metal pipe pile 36-in diameter or less	NE	NE	NE	NLAA	NC
2 metal boatlift I-beams	NE	NE	NE	NE	NC
Vinyl sheet pile – any size	NE	NE	NE	NLAA	NLAA
Concrete slab wall- any size (5 slabs or less installed/day)	NE	NE	NE	NLAA	NLAA
Concrete slab wall- any size (6-10 slabs installed/day)	NE	NE	NE	NLAA	NLAA
Metal sheet pile- any size	NE	NE	NE	NLAA	NC

 
 Table 20. NMFS Effects Determinations for In-water Construction Noise Analyzed under this Opinion to Sea Turtles and ESA-listed Fish

NE = no effect; NLAA = may affect, not likely to adversely affect

NC = not covered under this Opinion due to the potential effects to listed species

## 2.2.1 Activity 1 (A1): Shoreline Stabilization

## **General Description**

This category of activity covers activities related to shoreline stabilization, including the installation, repair (including all forms of maintenance), replacement, and removal of vertical seawalls and shoreline stabilization materials used to protect shorelines. These structures are typically installed from the land or from a shallow-draft barge. This Opinion does not cover the installation of jetties and groins or the placement of beach nourishment material. Living shoreline activities are analyzed separately under aquatic habitat enhancement, establishment, and restoration activities under Activity 7 in Section 2.2.7. Images of typical shoreline protection projects are provided below in Figure 9. For the purposes of this consultation, shoreline protection activities include installation and repair of vertical seawalls (including associated footers), and installation and repair of shoreline stabilization measures such as riprap, articulating blocks or mats, geotextile mats, and sand cement riprap. Vertical seawalls and shoreline stabilization measures are often combined to improve structural stability, such as placement of riprap material at the toe of a vertical seawall.

Vertical seawalls are constructed of vinyl, metal sheet pile, wood, or pre-fabricated concrete slabs. Vertical seawall installation can occur either by (1) using land-based equipment to trench,

grade, or shape the shoreline (i.e., dredge the area) and set the seawall pieces in place, or (2) using barge-mounted equipment to place, jet, or hammer the materials into position. The seawall may be supported by installing batter or king piles by vibratory or impact hammer and/or deadmen anchors that hook underground behind the seawall stabilizing them to the uplands. Vertical seawalls are often protected from erosion and scour by placing another structure (e.g., seawall footers) or shoreline stabilization materials (e.g., riprap) immediately waterward of the seawall. Seawall footers are short/low level structures placed directly in front of a seawall to protect the bottom from erosion and scouring. The seawall footer is typically less than half the height of the seawall and can be installed in place of riprap to stabilize the structure. This Opinion covers repair and replacement of seawalls up to 18-in waterward of most existing seawalls, as defined in the PDCs for Activity 1. This is consistent with FDEP's *Exempt Activities* (Section 62-330.051).

In addition to vertical seawalls, shorelines may be stabilized or protected using riprap, articulating blocks or mats, geotextile mats, and sand cement riprap. These materials are typically placed from the land and may require minor excavation of the shoreline to properly place the materials. Work may also be completed from the water by barge-mounted equipment. Riprap is placed by trenching the location (i.e., dredging), placing filter fabric, and then placing riprap on top of the fabric. Other materials may be approved for shoreline stabilization, if coordinated according to the standards in the Project Specific Review in Section 2.3. NMFS will review the projects that propose to use other materials to ensure the materials and methods do not present an entanglement or entrapment risk to protected species and to ensure that the effects of using those materials are consistent with the effects of those materials analyzed here.

Shoreline protection projects may be used in to stabilize only a targeted area or portion of shoreline. Small shoreline protection projects include stabilizing areas around outfall structures, boat ramps, culverts, and other structures analyzed under this Opinion.



Figure 9. Example images of shoreline armoring and stabilization. The left image is a concrete slab seawall, the middle is a riprap shoreline, and the right is a vinyl seawall. (Images provided by the Florida Marine Contractors Association in 2014).

# PDCs specific to Activity 1 for Shoreline Stabilization:

- A1.1. Activities covered by this Opinion include:
  - A1.1.1. New shoreline stabilization: New shoreline stabilization projects cannot exceed 500 ft in length. New seawalls and footers cannot extend any further waterward than 1.5 ft (18 in) from MHW, unless necessary to align a new seawall with 1 or more adjacent seawalls.

Repair or replacement of existing vertical seawalls: The repair, and replacement of seawalls and footers cannot extend any further waterward than 1.5 ft (18 in) from the wet face of the existing seawall or MHW, unless necessary to align with 1 or more adjacent seawalls. The repair or replacement of an existing seawall is not restricted to the 500 ft in length limit in PDC A1.1.1.

- A1.1.2. Shoreline stabilization materials may consist of riprap, articulating blocks or mats, and sand cement, geotextile/ filter fabric and mattresses. Installation of new shoreline stabilization materials where none previously existed may not extend more than 10 ft waterward of MHW (including the toe of the riprap). Riprap repair, and replacement may occur at its previous location, upland of, or within 1.5 ft (18 in) waterward of its previous location.
- A1.1.3. The Opinion does not cover removal of any length of seawall or other shoreline stabilization materials if such removal would result in an unstabilized shoreline.
- A1.1.4. The Opinion covers installation, repair, replacement, and removal of seawall footers.
- A1.1.5. This Opinion covers the removal/fill of upland cut boat ramps, slips, and basins to return the shoreline to the natural contour and/or to bring the shoreline into alignment with the adjacent property shorelines.

# The following PDCs apply to all the activities described in PDC A1.1 above:

- A1.2. Placement of backfill is limited to those situations where it is necessary to level the land behind seawalls or riprap. This includes backfill associated with installation of a seawall or riprap to remove/fill in an upland cut area (e.g., boat slip, boat ramp, boat basins) to return the shoreline to the original shape or to connect to adjacent seawalls to bring the shoreline into alignment with adjacent property shorelines.
- A1.3. Shoreline stabilization materials must be placed by hand around red mangrove prop roots.
- A1.4. Shoreline stabilization structures, other than vertical seawalls, shall be no steeper than a 2:1 Horizontal: Vertical slope for riprap, or the appropriate slope necessary to ensure shoreline stability while minimizing the total footprint when using materials other than riprap.

- A1.5. Installation and/or repairs to groins, jetties, or other structures placed perpendicular to shore, and beach nourishment/renourishment are not covered in this Opinion. Breakwaters/living shorelines are covered as described in Activity 7.
- A1.6. No placement of riprap below MHW is covered under this Opinion within the boundary of the FKNMS unless the FKNMS issues a NOAA permit or authorization that signifies the proposed activity is consistent with Title III of the Marine Protection, Research, and Sanctuaries Act of 1972, as amended. Proof of approval from the FKNMS is required as part of the project level review submission, described in Section 2.3, below.

A1.7. Shoreline protection shall not occur on ocean beaches used for sea turtle nesting.

## Additional PDCs for Activity 1 applicable in critical habitat:

In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat as described below.

- A1.8. Smalltooth sawfish critical habitat: Installation of new shoreline stabilization materials in smalltooth sawfish critical habitat under this Opinion is limited to:
  - A1.8.1. Placement of new shoreline stabilization materials (i.e. riprap, articulated concrete mats) in water depths deeper than -3 ft MLLW. No stabilization materials can be placed in waters between the MHWL and -3 ft MLLW.
  - A1.8.2. Installation of new or repair/replacement seawalls within 1.5 ft waterward of the existing seawall or MHW.
  - A1.8.3. Repair and replacement of shoreline stabilization materials (i.e., riprap, articulated concrete mats) within the same footprint of existing materials in depths between the MHWL and -3 ft MLLW. This means that these materials cannot result in the waterward extension or lateral expansion of materials beyond the previous footprint. Shoreline stabilization materials can be expanded in water depths deeper than -3 ft MLLW, as defined in PDC A1.1.3.
  - A1.8.4. Removal/filling of man-made upland cut areas (e.g., upland cut boat basins or boat ramps) to return the shoreline to its original contour are allowed, even if it removes the shallow, euryhaline essential feature.
  - A1.8.5. To be covered under this Opinion, activities cannot occur in areas identified as smalltooth sawfish limited exclusion zones, as defined in Section 2.1.1.1.
- A1.9. Gulf sturgeon critical habitat: All new shoreline stabilization materials (e.g., riprap, articulating concrete mats) can only be placed between the shoreline and where the water reaches a depth of -6 ft MHW. Additional noise restrictions are required for pile and sheet pile installation in the Gulf sturgeon critical habitat migratory restriction zones defined in Section 2.1.1.2.
- A1.10. *Acropora* critical habitat: To be covered under this Opinion, new or repair/replacement of shoreline protection cannot occur in *Acropora* critical habitat if the essential feature is

present. Repair and replacement of shoreline protection within *Acropora* critical habitat is covered if it is within the existing footprint.

- A1.11. Johnson's seagrass critical habitat: To be covered under this Opinion, installation of shoreline stabilization material (e.g., riprap and scour control materials, not vertical seawalls and footers) cannot occur if essential features are present. Repair and replacement of these materials (riprap and scour) is covered within the existing footprint. Vertical seawalls and footers can be installed, repaired, and replaced in Johnson's seagrass critical habitat even if the essential features are present, as long as the project is consistent with the applicable PDCs, including PDC A1.1.1 and A1.1.2 regarding overall length and waterward extension limit.
- A1.12. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): This Opinion does not apply to new structures in sea turtle critical habitat in the U.S. Caribbean. This Opinion does apply to repair and replacement of shoreline protection materials within U.S. Caribbean sea turtle critical habitat if the repair or replacement is within the existing footprint.

#### Assumptions

The USACE used the data collected from previous programmatic consultations to estimate how many shoreline stabilization projects it will authorize consistent with this Opinion over a 5-year period. Compared to previous Programmatic Opinions, the USACE was able to more accurately analyze data regarding shoreline stabilization activities and noted that the average size of the area impacted by shoreline stabilization projects differed between projects that occurred within smalltooth sawfish critical habitat (average seawall project resulted in 100.41 ft<sup>2</sup> of impacts to inwater areas) and outside of smalltooth sawfish critical habitat (average seawall project resulted in 291.59 ft<sup>2</sup> of impacts to in-water areas). We assume that shoreline stabilization projects in smalltooth sawfish have a smaller footprint because since we designated critical for the species in 2009, development in this area has been limited. For example, to avoid and minimize effects to the habitat, including the shallow, euryhaline water essential feature, seawalls may be designed to be smaller (not to extend out so far into the water) so as not to remove this feature entirely. In addition, placement of riprap along shorelines or in front of seawalls has been discouraged in these areas since the designation. Therefore, we will estimate the potential impacts of the expected shoreline stabilization activities within and outside of the geographic boundaries of smalltooth sawfish critical habitat using the data the USACE provided. The USACE also estimated that 27 projects would be covered in the next 5 years under this Opinion that would repair or replace a seawall that was longer than 500 lin ft. These 27 projects also may include riprap placement. All of these longer seawall repairs/replacements were estimated to occur outside of smalltooth sawfish critical habitat. Based on previously authorized seawall projects greater than 500 lin ft in length, the average repaired/replaced seawall length was 1,215 lin ft and covered 2,918.52 ft<sup>2</sup> of waterbottom (presumably from placement of the seawall waterward of the existing high water line or from the placement of riprap). We also assume that all of these 27 larger projects occur outside of critical habitat.

Thus, to analyze the effects of the shoreline stabilization projects, we looked the USACE's assumptions regarding the total number of shoreline stabilization activities to be covered under

this Opinion per 5-year period and whether the projects would occur within or outside of smalltooth sawfish critical habitat. Based on this information, we determined the potential effects to species (see Tables 21) and critical habitat (see Table 23).

USACE anticipates that it will authorize 7,102 shoreline stabilization activities per 5-year period that would qualify for coverage under this Opinion (i.e., that meet the PDCs and other conditions of this Opinion). A breakdown of how many of these will occur in each critical habitat unit is provided in Table 8.

## Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated the routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing shoreline stabilization involves all of the common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided above:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 5. Limiting Species' Movement and Access to Habitat
- 7. Noise

Please see Section 2.2 for our effects determinations.

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the installation, repair, replacement, and removal of shoreline stabilization structures beyond the common routes of effect considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 21. Sea turtles, smalltooth sawfish, and sturgeon may be affected by the permanent loss of habitat from shoreline stabilization activities associated with Activity 1. We quantified the potential extent of habitat impacts in Table 22 based on (1) the assumptions about the average area of impact per project, included above, (2) the number and location of projects (inside or outside of smalltooth sawfish critical habitat) estimated by USACE (Table 8), and (3) the construction limitations contained in the PDCs.

Listed Species	<b>USACE Determination</b>	NMFS Determination				
Sea Turtles	NLAA	NLAA				
Sawfish	NLAA	NLAA				
Sturgeon	NLAA	NLAA				
Effects I	Effects Determinations Explained in Section 2.2					
Johnson's seagrass	LAA	NE				
Corals	NE	N/A				
Whales	NE	NE				
Nassau grouper	Not provided	NLAA				

 Table 21. USACE and NMFS Determinations on the Effects of Shoreline Stabilization

 Activities (Activity 1) to ESA-listed Species listed in Table 5

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau Grouper for all covered activities, including Activity 1, was provided at the beginning of Section 2.2. The USACE's original determination for Johnson's seagrass was based on the assumption that shoreline stabilization activities would affect the species. Under the PDCs, this Opinion does not apply to shoreline stabilization projects where Johnson's seagrass is within the project footprint (PDC AP.13).

Type and Location of seawall	Number of projects (n <sup>17</sup> = 7,102)	Estimated Average Area Affected per Project (ft <sup>2</sup> )	Estimated Total Area Affected per 5-Year Period (ft <sup>2</sup> )	Estimated Total Area Affected per 5-Year Period (ac)
Smalltooth sawfish critical habitat	1,012	100.41	101,614.92	2.33
Outside of smalltooth sawfish critical habitat	6,063	291.59	1,767,910.17	40.59
Outside of smalltooth sawfish critical habitat- repair/replace longer than 500 lin ft	27	2,918.52	78,800.04	1.81
Total	7,102		1,948,325.13	44.73

 Table 22. Estimated Amount of Waterbottom Affected By Shoreline Stabilization Projects

 per 5-Year Period

Over a 5-year period, 44.73 ac of shallow nearshore habitat may be permanently covered or removed by the installation, repair, replacement, or removal of shoreline stabilization structures. This estimate includes projects in Florida and the U.S. Caribbean.

- Sea turtles: Sea turtles may be affected by the placement of shoreline stabilization materials on top of 44.73 ac of shallow nearshore areas that may be used for foraging. However, we believe that the effect to sea turtles from the potential loss of nearshore foraging habitat is insignificant.
  - The waterbottom affected by Activity 1 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes shoreline stabilization projects from the Opinion where Johnson's seagrass is present and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
  - Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
  - Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect listed corals, and excludes projects if non-listed

 $<sup>^{17}</sup>$  n = the total number of projects as provided in Table 8.

corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.

Shoreline stabilization activities may cover or remove nearshore areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition, the 44.73 ac of impact is very small compared to the extensive areas available throughout Florida and the U.S. Caribbean that support sea turtle prey species. Sea turtles can travel long distances to forage. The projects covered under this Opinion for Activity 1 will be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and throughout the U.S. Caribbean), so sea turtles will likely be able to forage in nearby areas outside of active project sites.

Given the above, we believe the effect to sea turtles from the potential loss of nearshore foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the shoreline stabilization activities under Activity 1 are not likely to adversely affect sea turtles.

Smalltooth sawfish: Smalltooth sawfish may be affected by placement of shoreline stabilization materials on top of 44.73 ac of shallow nearshore areas, which they could use for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat for foraging and red mangroves for refuge; these areas have been designated as critical habitat for the species. However, we believe that the effect on sawfish of the potential loss of nearshore foraging and refuge habitat associated with Activity 1 is insignificant. The PDCs limit this Opinion to shoreline protection activities occurring outside of smalltooth sawfish limited exclusion zones (PDC A1.8.1), which are areas research shows support higher levels of smalltooth sawfish pupping. Additionally, projects must be sited to avoid and minimize impacts to mangroves (AP.12), and mangrove removal for shoreline stabilization projects is not allowed under this Opinion for shoreline stabilization activities except for trimming and removal of mangroves above MHW that do not provide prop roots accessible to marine species (PDCs AP.12). Shoreline protection activities may also cover or remove nearshore areas inhabited by fish and crustaceans that serve as prey for smalltooth sawfish, and thus may affect areas used for smalltooth sawfish foraging. The area of impact (some amount less than 44.73 ac) is very small compared to the remaining large nearshore areas that support sawfish prey species. Sawfish can travel long distances to forage. The array of individual projects covered under this Opinion under Activity 1 that could affect smalltooth sawfish will be separated both temporally and spatially (along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites.

Therefore, the effect on juvenile sawfish from the loss of small areas of shallow water forage habitat is expected to be so small as to be undetectable, and thus is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the shoreline protection activities under Activity 1 are not likely to adversely affect smalltooth sawfish.

Sturgeon: Sturgeon may be affected by placement of shoreline stabilization materials on top of 44.73 ac of shallow nearshore areas that may be used for foraging. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. Shoreline stabilization projects may cover and bury nearshore bottom substrates containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. The effect on sturgeon of the potential loss of nearshore foraging habitat is insignificant. Sturgeon are opportunistic feeders that forage over large areas. Gulf sturgeon select foraging habitat based on substrate composition and depth, rather than prey density, abundance, or diversity. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) deep that are characterized by low-relief sand substrate (Fox et al. 2002). Hence, Gulf sturgeon, and likely shortnose and Atlantic sturgeon, often occupy waters deeper than those typically affected by shoreline protection activities (including both seawalls and riprap placement) that occur along shoreline (i.e., in the vicinity of the MHWL). Within important foraging areas (i.e., Gulf sturgeon critical habitat) depth limitations in PDC A1.9 exclude projects from coverage under this Opinion that would place riprap in the preferred foraging depth range of sturgeon. The area of impact (some amount less than 44.73 ac) is very small compared to the remaining large areas that support sturgeon prey species. The array of individual projects covered under this Opinion for Activity 1 will be separated both temporally and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites.

Therefore, the effect on sturgeon from the losses of small areas of shallow water forage habitat is expected to be so small as to be undetectable, and thus is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the shoreline stabilization activities under Activity 1 are not likely to adversely affect sturgeon.

#### Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from shoreline stabilization activities under Activity 1. The estimated impacts to each critical habitat unit and NMFS and USACE effects determinations for this category of activity are summarized in Table 23 with calculations of the estimated impacts provided in Table 24.

Designated Critical Habitat	<b>USACE</b> Determination	NMFS Determination				
Smalltooth sawfish	LAA	LAA				
Gulf sturgeon	NLAA	NLAA				
Johnson's seagrass	NLAA	LAA				
Loggerhead	NE	NE				
Effects Determinations Explained in Section 2.2						
Acropora NE NE						
North Atlantic right whale	North Atlantic right whale NE NE					
Atlantic sturgeon         Not provided         NE						
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.						

 Table 23. USACE and NMFS Determinations on the Effects of Shoreline Stabilization

 Activities (Activity 1) to Designated Critical Habitat

 Table 24. Estimated Amount of Waterbottom Affected By Shoreline Stabilization Projects

 in Smalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat per 5-Year

 Period

Type of seawall	Number of projects	Estimated Average Area Affected per Project (ft <sup>2</sup> )	Estimated Total Area Affected per 5-Year Period (ft <sup>2</sup> )	Estimated Total Area Affected per 5-Year Period (ac)
Smalltooth sawfish critical habitat	1,012	100.41	101,614.92	2.33
Gulf sturgeon critical habitat	165	291.59	48,112.35	1.10
Johnson's seagrass critical habitat	306	291.59	89,226.54	2.05

## Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

The USACE anticipates it may authorize 1,102 shoreline stabilization activities that qualify for coverage under this Opinion in a 5-year period in smalltooth sawfish critical habitat. We estimate that 101,614.92 ft<sup>2</sup> (2.33 ac) of waterbottom area within smalltooth sawfish critical habitat will be covered by the placement these structures (Table 24). This Opinion only covers shoreline protection activities under Activity 1 that occur outside of sawfish limited exclusion zones (PDC A1.8.5), and the estimate of the amount of waterbottom affected in smalltooth sawfish critical habitat accounts for that. Red mangroves are 1 of the essential features of sawfish critical habitat. PDC AP.12 limits red mangrove removal under the Opinion to specific instances not including removal of red mangrove for shoreline stabilization activities, and removal above the MHWL provided that red mangrove prop roots are accessible to marine species are not removed. PDC A1.3 also protects red mangroves by requiring hand-placement of

shoreline stabilization materials around red mangrove prop roots, avoiding harm to red mangroves from shoreline stabilization activities. Therefore, there will be no effect to the essential feature of red mangroves in sawfish critical habitat.

We believe shoreline stabilization activities are likely to adversely affect the shallow, euryhaline essential feature from the placement of materials in shallow, nearshore waters calculated in Table 24 above. PDC A1.8 limits impacts to the shallow, euryhaline essential feature by limiting the installation or repair of vertical seawalls to no more than 1.5 ft waterward of the existing seawall or the MHWL and prohibits new riprap and other stabilization materials from placement within shallow, euryhaline habitats (i.e., within from waters between the MHWL and -3 ft MLLW).

Combined, new and repair/replacement projects anticipated per 5-year period are estimated to permanently remove 101,614.92 ft<sup>2</sup> (2.33 ac) of shallow, euryhaline habitat within sawfish critical habitat (see Table 24). We believe the potential loss of shallow, euryhaline habitat due to shoreline stabilization projects under Activity 1 is likely to adversely affect critical habitat and is evaluated in Section 5.

#### Potential Routes of Effect to Gulf Sturgeon Critical Habitat

USACE anticipates that 165 shoreline stabilization activities may be covered under this Opinion per 5-year period within the geographic boundary of Gulf sturgeon critical habitat under this Opinion.

Combined, we anticipate that new and repair/replacement projects covered under this Opinion including the placement of seawalls and riprap will permanently remove or alter 48,112.35 ft<sup>2</sup> (1.1 ac) of estuarine habitats within Gulf sturgeon critical habitat per 5-year period (see Table 24). Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

1. Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages. Shoreline protection projects may cover and bury nearshore bottom substrates containing sturgeon prey species. But we expect the effects to this PCE will be insignificant since the estimated 1.1 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we estimate support sturgeon prey species. Further, not all of the 1.1 ac of habitat lost to shoreline stabilization support prey items or serve as preferred foraging habitat. Gulf sturgeon are suction feeders that tend to forage further offshore, in calm marine and estuarine waters that support their macroinvertebrate prey including brachiopods, mollusks, worms, and crustaceans (Mason and Clugston 1993). During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by lowrelief sand substrate (Fox et al. 2002). In addition, within Gulf sturgeon critical habitat, depth limitations in PDC A1.9 exclude projects involving the placement of riprap in the preferred foraging depth range of sturgeon from coverage under the Opinion (this Opinion only covers riprap placement projects in waters shallower than -6 MHW). Although other

shoreline stabilization activities, such as seawall installation or repair, may occur within Gulf sturgeon critical habitat, not all such projects will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from the installation, repair, replacement, or removal of shoreline stabilization structures; however, we believe that the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for *In-Water Activities* requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high-energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high-energy areas where turbidity will dissipate very quickly), changes in water quality is not a concern.
- 3. Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages. The placement of shoreline stabilization materials, such as seawalls, riprap, cement, and geotextile fabric, can affect sediment quality. The placement of shoreline stabilization materials converts sandy substrate, capable of supporting Gulf sturgeon prey, to hard man-made materials that do not support prey species. However, we believe the effects to this PCE will be insignificant since the estimated 1.1 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we estimate support sturgeon prey species and Gulf sturgeon foraging. Further, not all of the 1.1 ac of habitat lost to shoreline stabilization has the sediment quality needed to support Gulf sturgeon prey or serve as preferred foraging habitat. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Within Gulf sturgeon critical habitat, depth limitations in PDC A1.9 exclude projects involving the placement of riprap in the preferred foraging depth range of sturgeon (this Opinion only covers riprap placement projects in waters shallower than -6 MHW). Although other shoreline stabilization activities, such as seawall installation or repair, may occur within Gulf sturgeon critical habitat, not all will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats. Shoreline stabilization activities could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by shoreline stabilization activities in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging. However, we believe that there will be no effect to this PCE. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, PDC A1.9 limits construction activities in Gulf sturgeon critical habitat, requiring new shoreline stabilization materials to be placed between the shoreline and -6ft MHW, and requires compliance with the noise limitations in the Gulf sturgeon critical habitat migratory restriction zones. Shoreline stabilization projects occurring outside of Gulf sturgeon critical habitat migratory restriction zones. Shoreline stabilization projects occurring outside of Gulf sturgeon critical habitat migratory restriction zones. Shoreline stabilization projects occurring outside of Gulf sturgeon critical habitat migratory restriction zones. Shoreline stabilization projects occurring outside of Gulf sturgeon critical habitat migratory restriction zones. Shoreline stabilization projects occurring outside of Gulf sturgeon critical habitat migratory restriction zones.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and because we believe there will be no effect to safe and unobstructed migratory pathways, we believe that shoreline stabilization activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

#### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

The USACE anticipates that 306 shoreline stabilization activities may be covered under this Opinion per 5-year period in Johnson's seagrass critical habitat. The Opinion does not cover installation of new riprap, scour material, and materials other than vertical seawalls and footers occurring in Johnson's seagrass critical habitat if the essential features listed in Table 7 are present, but repair and replacement of these materials is covered within the existing footprint (PDC A1.11). This PDC limits the placement of new riprap, scour, and other stabilizing material; seawalls and footers may be installed, repaired, or removed in Johnson's seagrass critical habitat when the essential features are present, consistent with the other PDCs, including PDC A1.1.1 and A1.1.2). Combined, new and repair/replacement projects are anticipated to permanently remove 89,226.54 ft² (2.05 ac) of nearshore habitat (Table 24). We do not know if these areas contain the features of Johnson's seagrass critical habitat; however, for the purpose of this consultation, we will assume that the features may be present. Therefore, shoreline stabilization projects in Activity 1 are likely to adversely affect critical habitat and those effects are evaluated in Section 5.

#### 2.2.2 Activity 2 (A2): Pile-Supported Structures and Anchored Buoys

#### **General Description**

Pile-supported structures include docks, marinas, minor structures, chickees (i.e., over-water camping platforms), ATONs, and private aids-to-navigation (PATONs) including pile-supported signs or anchored floating buoys. Activity 2 includes the installation, repair (e.g., maintenance), replacement, and removal of these structures. Minor structures typically include mooring piles, dolphin piles (not to exceed a cluster of 4), boatlifts, hoists, davits, davit pads, fenders, fender piles, mooring whips and cleats. Pile supports may be made of wood, metal, concrete, and composite materials, and may be installed by jetting, auguring, vibratory hammer, and impact

hammer. Docks/piers can be designed in various configurations. T-shaped docks consist of a long walkway to a terminal platform(s) extending to either side, and L-shaped docks are similar but with a terminal extending to only one side. Marginal docks run parallel with the shoreline either directly attached to the shore or by constructing a short walkway perpendicular to the shore connecting to a longer dock constructed parallel with the shore. Floating vessel platforms include solid floating docks or platforms that are typically attached to permanent piles. Example images of these structures are shown below in Figures 10-12.



Figure 10. Examples of dock shapes, including T-shaped and L-shaped docks (©2014 Google).



Figure 11. Pile-supported Structures. Example of a marginal dock on the left, boatlift with I-beam in the middle, and a chickee on the right. (The left and middle photos were provided by the Florida Marine Contractors Association and right photo from www.cnn.com.).



Figure 12. Sample Marina Images. The left image shows an example of an open water marina (provided by the Florida Marine Contractors Association), while the right image shows an example of marina contained within an upland cut basin (©2014 Google).

Marinas occur in a variety of configurations and locations and can be designed for a variety of uses (e.g., commercial marina for boat sales, public marina, yacht club, fishing charters, commercial fishing) such as those shown in Figure 12. For the purposes of this Opinion, we also consider multi-family docking facilities as marinas (e.g., condo complexes, trailer parks, subdivisions that the home owners association owns the docks). Construction of new marina and multi-family facilities, reconfiguration and expansion of existing facilities, repair and replacement of deteriorating structures, and removal of existing structures are included in this activity.

Mooring fields are designated in-water areas where vessels can attach to anchored buoys. These areas are designed to allow boaters to moor in deeper water than is typically feasible at a marina. Vessel operators then access shore from dinghies.

## Project Design Criteria

PDCs Specific to Activity 2 - Pile-Supported Structures and Anchored Buoys:

- A2.1. Activities covered by this Opinion include the installation, repair, replacement, and removal of structures as described below:
  - A2.1.1. The pile-supported and anchored structures included in this Opinion are: docks and piers, boatlifts, mooring piles and dolphin piles associated with docks/piers; ATONs and PATONs; floating docks; pile-supported chickees (i.e., small, back-country, over-water, pile-supported, primitive camping shelters); boardwalks (as long as they are designed and clearly marked to prohibit fishing and vessel mooring); mooring fields and buoys; and other minor pile-supported structures. This does not include structures that support large commercial vessels including ferries, tankers, and cargo ships such as ferry terminals and large ports.
  - A2.1.2. Pile-supported docks/piers for a single-family residential lot are limited to 4 slips for motorized vessels. Slips for non-motorized vessels (e.g., kayak, canoe, and paddleboard) and associated launching areas do not count toward the total slip number.
  - A2.1.3. Pile-supported structures for marinas, multi-family facilities (e.g., condo complexes, trailer parks, subdivisions when the homeowners association owns and controls the in-water structures). Docks and piers for multi-family residential properties (e.g., condos, trailer parks, apartment complexes), and marinas are limited to a maximum of 50 total slips (i.e., combination of wet and dry slips for existing plus proposed slips).
  - A2.1.4. Anchored buoys and temporary pile-supported structures associated with marine events. Upon completion of the event, these structures must be removed and, to the maximum extent practical, the site must be restored to pre-construction elevations. Water depths in the area of marine events must be deep enough to support at least 5 ft of water depth under the keel of a vessel and between the keel of a vessel and ESA-listed coral colonies when transiting to the mooring areas. There is no limit on the number of vessel slips allowed for temporary structures associated marine events such as boat shows.
  - A2.1.5. Mooring fields are limited to a maximum of 50 motorized vessels (there is no limit on the number of non-motorized vessels).
  - A2.1.6. All pile-supported structures constructed must comply with PDC 2.17 for Docks or Other Minor Structures Constructed in Florida Under this Opinion (see below).

## The following PDCs apply to all the activities described in PDC A2.1 above:

A2.2. For commercial, multi-family, or public facilities, and marine events, signs must be posted in a visible location(s), alerting users of listed species in the area susceptible to vessel strikes and hook-and-line captures. The most current version of the signs that must be downloaded and sign installation guidance are available at:
(http://sero.nmfs.noaa.gov/protected\_resources/section\_7/protected\_species\_educational signs/index.html). The signs required to be posted by area are stated below:
A2.2.1. All projects in Florida shall use the Save Sea Turtle, Sawfish, and Dolphin sign. These signs shall include contact information to the sea turtle and marine

	4222	mammal stranding networks and smalltooth sawfish encounter database.
	A2.2.2.	defined in Section 2.1.1.4) shall nost the Help Protect North Atlantic Right
		Whales sign.
	A2.2.3.	On the east coast of Florida, projects located within the St. John's River and
		those occurring north of the St. Johns River to the Florida-Georgia line shall
		post the Report Sturgeon sign. On the west coast of Florida, projects occurring from the Cadar Kay, Elorida north to the Elorida Alabama line.
	A224	We are still developing the signs to be used in the U.S. Caribbean. Once
	112.2.1.	developed, those signs will be included at the website above.
A2.3.	For com	mercial, multi-family, or public facilities, monofilament recycling bins must be
	provided	at the docking facility to reduce the risk of turtle or sawfish entanglement in, or
	A231	Be constructed and labeled according to the instructions provided
	112.3.1.	at http://mrrp.myfwc.com.
	A2.3.2.	Be maintained in working order and emptied frequently (according to
		http://mrrp.myfwc.com standards) so that they do not overflow.
A2.4	For any o	dock project (new construction repair or replacement) at a private residence
112.1.	located v	within 11 nautical miles of North Atlantic right whale critical habitat (as
	measure	d in a radius from the center of the nearest inlet to open ocean and described in
	Section 2	2.1.1.4), the property owner will be provided a handout with their USACE
	permit de	escribing the presence of North Atlantic right whales in the area and the Federal
	regulatio	ons governing the approach to North Atlantic right whates (Appendix C).
A2.5.	ATONs	and PATONs must be approved by and installed in accordance with the
	requirem	nents of the USCG (see 33 CFR, chapter I, subchapter C, part 66 and RHA
	Section	10 and any other pertinent requirements).
A2.6.	Chickees	s must be less than 500 ft <sup>2</sup> and support no more than 2 slips.
A2.7	No activ	ities associated with municipal or commercial fishing piers are covered under
	this Opir	nion.
	<b>D</b> 1 ·	
A2.8.	Docks in turtlo fri	istalled within visible distance of ocean beaches are required to comply with andly lighting if lighting is pagessary to the project. Turtle friendly lighting is
	explaine	d and examples are provided on the Florida Fish and Wildlife Conservation
	Commis	sion website: http://myfwc.com/wildlifehabitats/managed/sea-turtles/lighting/
A29	Project c	construction will take place from uplands or from floating equipment (e.g.
A2.7.	barge); p	prop or wheel-washing is prohibited.
г	$\mathcal{O}^{-}\mathcal{O}^{-}$ r	
Addit	tional PDC	Cs for Activity 2 applicable in critical habitat:
L		
		113

In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat as described below.

- A2.10. *Acropora* critical habitat and the U.S. Caribbean: This Opinion does not cover new and expanded pile-supported structures in *Acropora* critical habitat where the essential features are present. The distance from ATONs to ESA-listed corals and *Acropora* critical habitat shall ensure there are no impacts to the corals or the essential feature of *Acropora* critical habitat from the movement of buoys and tackle. The appropriate distance shall be based on the size of the anchor chain or other tackle to be installed to secure the buoy to its anchor, particularly when the design of the ATON does not prohibit contact of tackle with the marine bottom. In all cases, buoy tackle will include flotation to ensure there is no contact between the anchor chain or line and the marine bottom.
- A2.11. Gulf sturgeon critical habitat: Additional noise restrictions are required for pile and sheet pile installation in the Gulf sturgeon critical habitat migratory restriction zones defined in Section 2.1.1.2. The noise restrictions are described in that section.
- A2.12. Smalltooth sawfish critical habitat: This Opinion does not cover activities occurring in areas identified as smalltooth sawfish limited exclusion zones defined in Section 2.1.1.1.
- A2.13. North Atlantic right whale critical habitat: This Opinion does not cover installation of anchored ATONs and permanent buoys in North Atlantic right whale critical habitat; temporary buoys for marine events are allowed in North Atlantic right whale critical habitat.
- A2.14. Johnson's seagrass critical habitat: This Opinion does not cover new marinas or multifamily facilities in Johnson's seagrass critical habitat. Repair, replacement, and reconfiguration of existing marinas or multi-family facilities may be covered if it (1) occurs within same overall footprint (out to the perimeter of the facility, including the outer limits of the structure and permitted mooring locations), (2) does not increase the total aerial extent (i.e., area of coverage from the dock structures) of the existing facility, and (3) does not affect Johnson's seagrass. Mooring fields are allowed in Johnson's seagrass critical habitat and within the range of Johnson's seagrass so long as they occur in waters deeper than -13 ft (-4 m).
- A2.15. NWA DPS of loggerhead sea turtle critical habitat: ATONs (pile-supported and anchored buoys) are allowed in nearshore reproductive habitat of the NWA DPS of loggerhead sea turtles under this Opinion. No other pile-supported structures are allowed in nearshore reproductive habitat under this Opinion.
- A2.16. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): ATONs (pile-supported and anchored buoys) are allowed near sea turtle nesting beaches under this Opinion. No other pile-supported structures are allowed near sea turtle nesting beaches under this Opinion.

#### A2.17. PDCs for Docks or Other Minor Structures

These PDCs address the anticipated dock construction scenarios expected within Florida and the U.S. Caribbean and provide NMFS PRD's construction guidelines for projects occurring (1) within Johnson's seagrass critical habitat; (2) within the range of Johnson's seagrass<sup>18</sup>, but outside of Johnson's seagrass critical habitat; and (3) outside of both the range and critical habitat for Johnson's seagrass. These scenarios consider whether a seagrass survey was conducted for projects within the range of Johnson's seagrass or located in Johnson's seagrass critical habitat. Surveys must be completed no earlier than 1 year before submitting the application to the action agency for project authorization. There is no seasonal restriction for Johnson's seagrass surveys; however, Johnson's seagrass is found within the range of other seagrass species that exhibit a seasonal pattern of growth and distribution. For comparison, NMFS Habitat Conservation Division's recommended sampling window for non-listed species is June 1 to September 30.

	Within Johnson's seagrass critical habitat	Within the Range of Johnson's seagrass (outside of critical habitat)	In the U.S. Caribbean and Florida (outside of the range and critical habitat of Johnson's seagrass)
Dock replacement in the exact previous dock with	ct footprint (i.e., s	ame location/configu	iration/ size) as the
No native seagrass under dock	А	А	А
Johnson's seagrass under dock	В	В	N/A
Native seagrass, other than Johnson's seagrass, under the dock	В	А	А
No current seagrass survey	В	В	A
New docks or dock expansion	ns with		
No native seagrasses within property limits	В	А	A
Johnson's seagrass within property limits	В	В	N/A
Native seagrass, other than Johnson's seagrass, within property limits	В	A	А
No current seagrass survey	В	В	A

**Dock Construction Scenarios** 

<sup>&</sup>lt;sup>18</sup> The range of Johnson's seagrass is defined as Turkey Creek/Palm Bay south to central Biscayne Bay in the lagoon systems on the east coast of Florida

B= Dock must meet PDCs below

N/A = not applicable; Johnson's seagrass could not occur under the dock because the dock project is outside the range of Johnson's seagrass

## Dock PDCs for Scenario B in the table above:

- 1. To avoid and minimize impacts to Johnson's seagrass and native, non-listed seagrasses to the maximum extent practicable:
  - The dock must be positioned to avoid and minimize effects to Johnson's seagrass
  - Over any area that contains Johnson's seagrass or native, non-listed seagrasses, the dock shall be oriented in a north-south orientation to the maximum extent that is practicable to allow maximum sunlight under the structure.
  - If practicable, terminal platforms shall be placed in deep water, waterward of Johnson's seagrass beds or native, non-listed seagrasses beds or in an area devoid of Johnson's seagrass or native, non-listed seagrasses.
  - Piles must be spaced a minimum of 10 ft apart in any area that contains Johnson's seagrass to minimize direct impacts.
  - Piles shall be installed in a manner that will not result in the formation of sedimentary deposits (e.g., donuts or halos) around the newly installed pilings.
  - No covered boat lifts are allowed over any Johnson's seagrass.
- 2. Decking options: Deck surfaces (parallel with the water) that are located waterward of the MHWL must be constructed of grated materials or plank construction or a combination of the both methods (e.g. plank decking on the walkway and grated decking on the terminal platform). These decking options are described below:

## Grated decking:

- Height requirement: The surface of the structure, including the dock walkway (the overwater narrow portion connecting the terminal platform to the shore and any over-water ramp required for access) and the dock, must be a minimum of 3 ft above MHW when constructed with grated decking.
- Size limitations: The dock walkway is limited to a width of 4 ft. The terminal platform is limited to a total area of 160 ft<sup>2</sup>. Marginal docks are limited to a width of 5 ft. The 5 ft width restriction is measured from wet side of the seawall. For example, if a seawall cap is 3 feet overwater then the dock would be limited to 2 feet.
- Material description: Decking materials shaped in the form of grids, grates, lattices, etc., to allow the passage of light through the open spaces. These materials must provide a minimum of 43% open space.

# Plank decking:

- Height requirement: The surface of the structure, including the dock walkway (the overwater narrow portion connecting the terminal platform to the shore and any over-water ramp required for access) and the dock, must be a minimum of 5 ft above MHW when constructed of plank decking.
- Size limitations: The dock walkway is limited to a width of 4 ft. The terminal platform is limited to a total area of 120 ft<sup>2</sup>. Marginal docks are limited to a width of 5 ft.
- Material description: Deck boards may be constructed of any material. Deck boards
## must be installed to provide a minimum of a 0.5-in gap between individual deck boards

## Assumptions

As with shoreline stabilization structures, the USACE reviewed projects covered under recent Programmatic Opinions to estimate impacts anticipated from pile-supported structures covered under this Opinion. The USACE provided the specifics regarding Activity 2, including:

- Number of minor structures (supporting 4 slips or less) and major structures projects (supporting 5-50 slips).
- Number of new construction and repair/replacement activities.
- Number and type (minor vs. major) projects located inside or outside of Johnson's seagrass critical habitat.
  - The distinction between projects occurring inside and outside of Johnson's seagrass critical habitat is important for 2 reasons. First, dock construction projects may adversely affect Johnson's seagrass critical habitat. Second, in projects meeting NMFS and USACE guidance entitled *Key for Construction Conditions for Docks or Other Minor Structures Constructed in or Over Johnson's Seagrass (Halophila johnsonii)*, the size and number of piles is smaller in Johnson's seagrass critical habitat.
- Assessment of how many dock projects are expected to occur within Johnson's seagrass critical habitat that will occur over Johnson's seagrass.
  - Approximately 10% of all dock projects are anticipated to occur within Johnson's seagrass critical habitat and over existing Johnson's seagrass. The average project of this type is estimated to shade 450 ft<sup>2</sup> of waterbottom that could contain Johnson's seagrass.
- Assessment of the number of dock projects anticipated to result in the removal of red mangroves within smalltooth sawfish critical habitat.
  - Although mangrove removal is allowed to install a dock walkway, the USACE is required to first avoid and minimize mangrove impacts. It is anticipated that very few docks built in smalltooth sawfish critical habitat will require the removal of red mangroves to place the dock due to the general requirement to avoid and minimize mangrove removal, either by siting the walkway to avoid mangroves entirely or to minimize the effects to only trimming. The USACE believes that only 7 dock projects in the last 5 years required mangrove removal and they anticipate mangrove removal will occur for less than 10 projects in the next 5 years.

USACE anticipates that 33,574 pile-supported structure and anchored buoy activities that qualify for coverage under this Opinion (i.e., that meet the PDCs and other conditions of this Opinion) will occur per 5-year period, 33,083 outside of Johnson's seagrass critical habitat and 491 in Johnson's seagrass critical habitat. A breakdown of how many of these will occur in each critical habitat unit is provided in Table 8.

Thus, to analyze the effects of the pile-supported structures and anchored buoys, we looked the USACE's assumptions regarding (1) the total number of shoreline protection activities to be covered under this Opinion per 5-year period; (2) whether the projects would be new construction or repair/replacement; and (3) whether the projects would occur within or outside of Johnson's seagrass critical habitat. Based on this information, we determined the potential effects to species (Table 25) and critical habitat (Table 28).

### Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing pile-supported structures would result in the following common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 6. Vessel Strikes
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the installation, repair, replacement, and removal of pile-supported structures and anchored buoys that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 25.

Sea turtles, smalltooth sawfish, sturgeon, and Johnson' seagrass may be affected by the permanent loss of habitat from the placement of piles and anchors associated with Activity 2. We quantified the potential extent of these direct impacts based on (1) the assumptions about the percent of projects that are new or repair/replacement and the number of piles required to support the structure, (2) the number and location of projects (inside and outside of Johnson's seagrass critical habitat) estimated by USACE (Tables 8), and (3) the construction limitations contained in the PDCs. In Tables 26 and 27 below, we determined the "total number of projects" by multiplying the total number of projects occurring outside of Johnson's seagrass critical habitat and within Johnson's seagrass critical habitat (Table 8) by the estimated percent of each project type to determine how many of each project type (e.g., new boatlifts, new large structures) is anticipated per 5-year period. The "total number of projects" was then multiplied by the "total number of new piles" and the "size of the piles" anticipated at that structure to determine the "total estimated amount of waterbottom affected." The impact from anchored buoys is expected to be small in size since anchors are typically not large and since we expect a small number of these projects as the use of anchored buoys are often limited to minor projects associated with marine events. Historically, the USACE has not tracked anchored projects and assumes the number of these projects is low, so we assume that the potential amount of waterbottom affected by anchored buoys is sufficiently captured in the estimated area we estimate will be affected by piles-supported structures.

For all of these calculations, we assumed that all piles used were square. We understand that piles may be square or round and that the area of impact from a round pile would be less than that of a square pile; however, since we are unsure which shape will be used, we will use the larger impact area of square piles acknowledging this may be a slight over estimate. We also assumed that all piles were 12-in by 12-in (average pile size) since some projects types did not state the pile size and some provided a range.

 Table 25. USACE and NMFS Determinations on the Effects of Pile-Supported Structures

 and Anchored Buoys (Activity 2) to ESA-listed Species listed in Table 5

Listed Species	USACE Determination	NMFS Determination			
Sea Turtles	NLAA	NLAA			
Sawfish	NLAA	NLAA			
Sturgeon	NLAA	NLAA			
Johnson's seagrass	LAA LAA				
Effects I	Determinations Explained in S	Section 2.2			
Corals	NE	N/A			
Whales	NE	NE			
Nassau GrouperNot providedNLAA					
NE= no effect; NLAA= may aff project may affect ESA-listed co	fect, not likely to adversely af orals, separate consultation is	fect; N/A= not applicable (if a required).			

The effects analysis for corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

Project Type	Percent	Total Number of projects (n = 33,083)	Total Number of New Piles/ Structures	Size of Piles (ft <sup>2</sup> )	Total Estimated Amount of Waterbottom Affected (ft <sup>2</sup> )	Total Estimated Amount of Waterbottom Affected <u>(ac)</u>
New boatlifts	38%	12,571.54 <sup>19</sup>	4	1	50,286.16	1.15
New minor dock	26.50%	8,767.00	35	1	306,844.83	7.04
New ATONS, anchored buoys, and single-pile structures	3.50%	1,157.91	1	1	1,157.91	0.03
Repair/replacement of minor structures	12%	3,969.96	0	N/A	0.00	0.00
New major structures (supporting up to 50 slips)	9%	2,977.47	388	1	1,155,258.36	26.52
Repair/replacement of major structures	11%	3,639.13	0	N/A	0.00	0.00
Total	100%	33,083*			1,513,547.25	34.75

 Table 26. Estimated Amount of Waterbottom Affected by Pile-Supported Structures and

 Anchored Buoys Occurring Outside of Johnson's Seagrass Critical per 5-Year Period

<sup>&</sup>lt;sup>19</sup> In this table and all future similar tables in the Opinion, we calculate the number of projects by multiplying the percent of projects by the total. We do not round the number of projects here because we are trying to estimate the total area affected based on a series of assumptions and rounding at each step may increase the final number. As we implement the Opinion, we will track the total number of projects and the total area affected.

Project Type	Percent	Total Number of projects (n = 491)	Total Number of New Piles/ Structure <u>s</u>	Size of Piles (ft²)	Total Estimated Amount of Waterbottom Affected (ft <sup>2</sup> )	Total Estimated Amount of Waterbottom Affected (ac)
New boatlifts	35.50%	174.31	4	1	696.65	0.02
New minor dock	36.30%	178.23	20	1	3,564.66	0.08
New ATONS, anchored buoys, and single-pile structures	5.20%	25.53	1	1	25.51	$0.00^{20}$
Repair/replacement of minor structures	14.00%	68.74	0	N/A	0	0.00
New major structures (supporting up to 50 slips)	Per PDC A2.14, the Opinion does not cover new marina or multi-family facilities (new major structures) in Johnson's seagrass critical habitat					facilities (new
Repair/replacement of major structures	9.00%	44.19	0	N/A	0	0.00
Total	100.00%	491			4,287.41	0.10

 Table 27. Estimated Amount of Waterbottom Affected by Pile-Supported Structures and

 Anchored Buoys Occurring Within Johnson's Seagrass Critical Habitat per 5-Year Period

Over a 5-year period, 34.85 ac of waterbottom area may be permanently covered or removed by the installation of new pile-supported structures and anchored buoys covered under this Opinion (34.75 ac [total from Table 26] + 0.1 ac [total from Table 27] = 34.85 ac). This estimate includes projects in Florida and the U.S. Caribbean. Repair and replacement of pile-supported structures are not expected to result in habitat losses because the number of piles is not expected to increase.

- <u>Sea Turtles:</u> Sea turtles may be affected by the placement of piles and anchors on top of 34.85 ac of waterbottom area that may be used as forage habitat for sea turtles. However, the effect on sea turtles of the potential loss of foraging habitat is insignificant.
  - The waterbottom affected by Activity 2 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes pile-supported structure projects from this Opinion where Johnson's seagrass is present, except for pile-supported structures that meet the PDCs for Docks or Other Minor Structures (PDC A2.17), and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable. Although some shading of seagrasses is allowed

<sup>&</sup>lt;sup>20</sup> 25.51 ft<sup>2</sup> is 0.0006 ac. We are rounding to 2 figures, hence we assume 0,00 ac will be affected.

from pile-supported structures as defined by the PDC A2.17, and shading may have small impacts on seagrass coverage, recruitment, and growth, we believe the minimal impacts to seagrass resources from shading will have an insignificant effect on sea turtles through reduction in forage resources.

- Limestone outcroppings and worm-rock reefs that are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the project footprint.
- Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect listed corals, and excludes projects if non-listed corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
- New pile-supported structures and anchored buoys may cover or remove areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition, the 34.84 ac of impact is very small compared to the extensive areas available throughout Florida and the U.S. Caribbean that support sea turtle prey species. Sea turtles can travel long distances to forage. The array of individual projects covered under Activity 2 will be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and throughout the U.S. Caribbean), so sea turtles will likely be able to forage in nearby areas outside of active project sites.

Given the above, effect to sea turtles from the potential loss of nearshore foraging habitat is insignificant. In total, based on the routes of effect analyzed here (insignificant effect on sea turtles from impacts to foraging habitat) and in Section 2.2, we determined that pile-supported structures and anchored buoys covered under this Opinion under Activity 2 are not likely to adversely affect sea turtles.

Smalltooth sawfish: Smalltooth sawfish may be affected by the placement of piles and anchors on 34.85 ac of waterbottom area, which they could use for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida, and in greater concentrations in smalltooth sawfish critical habitat. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for refuge; these areas have been designated as critical habitat for the species. However, we believe the effect on sawfish of the potential loss of foraging and refuge habitat associated with Activity 2 is insignificant. The PDCs limit activities in smalltooth sawfish critical habitat and exclude pile-supported structures and anchored buoys projects (activities under Activity 2) occurring smalltooth sawfish limited exclusion zones (PDC A2.12), which are areas research shows support higher levels of smalltooth sawfish pupping. In addition, under PDC AP.12, projects must be sited to avoid

and minimize impacts to mangroves, and mangrove removal is strictly limited. Pertinent to Activity 2, PDC AP.12 limits mangrove removal associated with dock installation to no more than 4 lin ft of mangrove shoreline to place a dock walkway or up to 8 lin ft for an ADA-compliant public dock. PDC AP.12 also allows removal above MHW provided that red mangrove prop roots that are accessible to marine species are not removed. The USACE believes that very few docks will require the removal of red mangroves because docks can typically be sited to avoid red mangrove impacts, as required by PDC AP.12.

USACE estimates that 10 docks of the 4.125 docks in smalltooth sawfish critical habitat (or 0.24%) could require red mangrove removal in in the next 5-year period. If we assumed that 1% (rounding up from 0.24%) of all pile-supported structures in smalltooth sawfish critical habitat would require red mangrove removal, approximately 412.5 projects would require red mangrove removal (4,125 total projects in smalltooth sawfish critical habitat from Table 8 x 1%= 412.5). If all of these projects would require 4-ft wide walkways, 1,650 lin ft (412.5 projects x 4 lin ft) of red mangroves would be removed. If all the docks were to require the largest walkway dimensions of 8-ft wide to accommodate ADA compliant walkways, a maximum of 3,300lin ft (412.5 projects x 8 lin ft) of red mangrove removal would occur. Therefore, we estimate that between 1,650 and 3,330 lin ft of mangrove shoreline in critical habitat may be removed for the placement of docks in Florida by activities that may be covered under this Opinion per 5-year period. Even at the upper limit associated with 8-ft walkways, the anticipated loss of mangrove shorelines is very small compared to the amount of mangrove shoreline remaining in critical habitat and throughout Florida. As is noted above, PDC A2.12 excludes pile-supported structures and anchored buoys projects occurring in smalltooth sawfish limited exclusion zones. Additionally, the array of individual projects covered under this Opinion for Activity 2 will likely be separated both temporally (over a 5vear period) and spatially (along the entire coast of Florida), and sawfish can forage and seek refuge in nearby areas outside of active project sites. Therefore, we believe the effect on smalltooth sawfish from the potential loss of foraging and refuge habitat is insignificant. Based on the routes of effect analyzed here and the routes of effect addressed in Section 2.2, we determined that pile-supported structures and anchored buoys covered under this Opinion for Activity 2 are not likely to adversely affect smalltooth sawfish.

• <u>Sturgeon:</u> Sturgeon may be affected by the placement of piles and anchors on 34.85 ac of waterbottom area that may be used as foraging habitat. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. Piles and anchors may cover and bury bottom substrates containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, the effect on sturgeon of the potential loss of foraging habitat is insignificant. Sturgeon are opportunistic feeders that forage over large areas. Gulf sturgeon select foraging habitat based on substrate composition and depth, rather than prey density, abundance, or diversity. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Hence, Gulf sturgeon, and likely shortnose and Atlantic sturgeon, often occupy waters deeper than those affected by at least smaller docks located closer to shore. Even still, the area of impact (some amount less than 34.85 ac) is very small compared to the areas approximately 1.5 million ac of available marine and estuarine critical habitat that we

estimate support sturgeon prey species. The array of individual projects covered under this Opinion for Activity 2 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and the routes of effect addressed in Section 2.2, we determined that pile-supported structures and anchored buoys analyzed under Activity 2 are not likely to adversely affect sturgeon.

Johnson's seagrass: We believe there will be no effect to Johnson's seagrass from mooring fields. Mooring fields in the range of Johnson's seagrass are limited to waters deeper than 13 ft (PDC A2.14). Studies show that Johnson's seagrass occurs in waters shallower than 10-13 ft (3-4 m) (NMFS 2007a). Water depths greater than 13 ft are not believed to provide the water transparency necessary for enough sunlight to reach the sea floor to support Johnson's seagrass growth. Therefore, mooring fields in waters too deep to support the growth of Johnson's seagrass are not expected to affect Johnson's seagrass.

We believe pile-supported structures in areas that support Johnson's seagrass may affect Johnson's seagrass via shading, however we believe the effect will be insignificant. All docks built throughout the range of the species (which includes all Johnson's seagrass critical habitat) will be required to adhere to the PDCs including PDC A2.17, which describes the allowable decking height and construction materials for overwater structures. These guidelines are designed to reduce the effects of shading by maximizing light transmission under the structures. They require new docks or dock expansions within Johnson's seagrass critical habitat to be constructed of either grated decking or plank decking meeting specific criteria. Within the range of the species, but outside of critical habitat, those new structures must be constructed of grated or plank decking unless there are no native seagrasses within the property limits or there is native seagrass, other than Johnson's seagrass, within those limits. With grated decking, the surface of the structure must be elevated to a minimum of 3 ft above MHW. Grated decking is decking constructed of a grid or other pattern that provides a minimum of 43% open space. In addition, the size of the dock walkway and the terminal platform, and the size of the marginal dock, are limited to reduce the overall size of the overwater structure and further reduce the area shaded. With plank decking, deck boards must be spaced  $\frac{1}{2}$  in apart and the surface of the structure must be elevated to a minimum of 5 ft above MHW. As with grated docks, the size of the walkway and the terminal platform, as well as the size of the marginal dock, also are limited.

We developed these PDCs based on studies that indicate both dock height and deck type influence seagrass survival. A 2008 study, found no statistical difference in density of Johnson's seagrass under grated decking compared to adjacent sites and reference areas (Landry et al. 2008). However, this pattern did not hold true for all seagrass species as most others were statistically reduced under grated decks when compared to the adjacent sites (Landry et al. 2008). Another study considered light availability and dock height and found "a significant positive correlation between dock height and light availability in the seagrass canopy in the shaded areas beneath the dock walkway" (i.e., the higher the dock, and the more space between the dock and MHW, the more light was able to reach the seagrass (Shafer et al. 2008)). Results showed a reduction in the frequency of occurrence of most seagrass species, including Johnson's seagrass, under docks that were not fully built to the

deck height requirements of the dock construction guidelines (Shafer et al. 2008). Yet, Shafer et al. (2008) stated that their data "...demonstrate that *H. johnsonii* is capable of growing under the low light conditions found in the shaded areas beneath dock walkways." Therefore, we believe new or expanded docks built to these guidelines will continue to support Johnson's seagrass and provide adequate light availability for its growth because Johnson's seagrass is more tolerant to low light conditions than other seagrasses in the area (NMFS 2002). As a result, we believe that docks built in accordance with PDC A2.17 will have only insignificant effects on Johnson's seagrass. Dock replacement in the same footprint and of the same size as the existing structure will not increase shading in the area and will have no effect on the species.

Likewise, any vessel moored at the pile-supported structures in areas supporting Johnson's seagrass could shade the waterbottom and affect the species. However, as noted above, we believe installation, repair, and replacement of pile-supported will not contribute to an increase the number of registered vessels in the State of Florida or in counties adjacent Johnson's seagrass critical habitat (see Section 2.2, *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* Item 6, Vessel Strikes), and that it is extremely unlikely that the proposed action will increase the number of vessels moored in areas supporting Johnson's seagrass. Thus it is extremely unlikely that these projects will affect Johnson's seagrass through vessel shading. In addition, the population of Johnson's seagrass has remained stable and even expanded north despite the continued repair, replacement, and construction of docks and the vessels associated with them in southeast Florida. Therefore, the effect of vessel shading is discountable.

In addition, we believe that placement of the piles in areas supporting Johnson's seagrass may affect the species. We evaluate the potential effects to Johnson's seagrass from the placement of piles in areas that support Johnson's seagrass based on information provided by USACE regarding the number of projects occurring within Johnson's seagrass critical habitat (491) and the percent of those projects that might occur in areas supporting the species. We recognize that Johnson's seagrass may occur throughout its range, which extends beyond the limits of its critical habitat. However, due to data limitations, we do not know how many docks have been constructed outside of critical habitat, but within the range of the species. Based on USACE's experience with permitting activities within the range of Johnson's seagrass, the USACE believes that 10% of the 491 projects that will occur in Johnson's seagrass critical habitat may be constructed in areas that support Johnson's seagrasses. As we calculated in Table 27, above, we believe the piles associated with the 491 structures expected in Johnson's seagrass critical habitat may be placed on up to 3,989.56 ft<sup>2</sup> (0.1 ac) of waterbottom. If we assume that 10% of these projects will be placed in areas that support Johnson's seagrass, and that placement in an area supporting Johnson's seagrass will affect the species, the projects may affect up to 398.96 ft<sup>2</sup> of Johnson's seagrass (3,989.56 ft<sup>2</sup> x 10% = 398.956 ft<sup>2</sup>). The effect to this amount of Johnson's seagrass from pile installation is analyzed in Section 5. Our approach could overestimate the effects to Johnson's seagrass. Although the USACE expects 491 projects in Johnson's seagrass critical habitat, 23% (or about 113) of those projects will be repair or replacement projects that we do not expect to affect the waterbottom or Johnson's seagrass because these projects generally do not involve additional piles in the water. However, we think including the repair and replacement

projects (a potential overestimate) will offset the potential underestimate from the data limitations that prevent us from quantifying the effects of those docks located outside critical habitat, but within the species' range, that may also affect the species.

The effects to the essential features of Johnson's seagrass critical habitat are analyzed separately below.

#### Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from installation, repair, replacement, and removal of pile-supported structures and anchored buoys covered under this Opinion. The estimated impacts to each critical habitat unit and NMFS and USACE effects determinations for this category of activity are summarized in Table 28.

## Table 28. USACE and NMFS Determinations on the Effects of Pile-Supported Structures and Anchored Buoys (Activity2) to Designated Critical Habitats

Project Location	Number of Projects	USACE Determination	NMFS Determination
Sawfish critical habitat	4,125	LAA	LAA
Gulf sturgeon critical habitat	724	NLAA	NLAA
Johnson's seagrass critical habitat	491	LAA	LAA
Loggerhead critical habitat	The effects analysis	NE	NE
Acropora critical habitat	for these critical habitat units was	NE	NE
North Atlantic right whale critical habitat	provided at the beginning of	NE	NE
Atlantic sturgeon critical habitat	Section 2.2	Not provided	NE

## Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

The USACE anticipates it may authorize 4,125 pile-supported structures that qualify for coverage under this Opinion in a 5-year period in smalltooth sawfish critical habitat. We estimate that 172,322 ft<sup>2</sup> (3.96 ac) of waterbottom area within smalltooth sawfish critical habitat will be covered by the placement of pile-supported structure and anchored buoy activities (Table 29).

Project Type	Percent	Number of project (n = 4,125)	Number of New Piles	Size of Piles (ft <sup>2</sup> )	Total Estimated Waterbottom Affected by Piles per 5- year (ft <sup>2</sup> )	Total Estimated Waterbottom Affected by Piles per 5- year (ac)
New boatlifts	38%	1,567.50	4	1	6,270	0.14
New minor dock	26.50%	1,093.13	20	1	21,863	0.50
New ATONS and single-pile structures	3.50%	144.38	1	1	144	0.00 <sup>21</sup>
Repair/ replacement of minor structures	12%	495.00	0 <sup>22</sup>	N/A	0	0
New major structures (supporting up to 50 slips)	9%	371.25	388	1	144,045	3.31
Repair/ replacement of major structures	11%	453.75	0	N/A	0	0
Total	100%	4,125			172,322	3.96
N/A = not applica	ble; these p	ojects are rep	pair/replace	ments tha	t do not have new	v impacts.

Table 29.	Estimated Amount of Waterbottom Affected by Pile-Supported Structures an	nd
Anchored	Buovs Occurring Within Smalltooth Sawfish Critical Habitat per 5-Year Per	riod

<sup>&</sup>lt;sup>21</sup> 144 ft<sup>2</sup> is approximately 0.003 ac, which rounded to 2 decimal places is 0.00 ac.

 $<sup>^{22}</sup>$  We assume repair and replacement structures are of the same size and do not result in additional piles on the waterbottom.

We believe the installation of the pile-supported structures and anchored buoys are not likely to adversely affect the shallow (less than -3 ft MLLW), euryhaline water essential feature of critical habitat. Installing piles or anchoring buoys will have no effect on the salinity (i.e., euryhaline component of the essential feature) of the surrounding waters. While some piles and anchored buoys will be installed within the shallow component of the essential feature, we believe any effects to the critical habitat will be insignificant. The placement of a few piles or anchored buoys will not change the overall depth in the area or restrict sawfish foraging, refuge, or movement in the area. The habitat will continue to provide for predator avoidance and habitat for prey, thus facilitating the recruitment of juveniles into the adult population. The Opinion does not cover projects under Activity 2 if they occur in sawfish limited exclusion zones (PDC A2.12). Therefore, we assume that all projects will occur outside of these sensitive areas. We believe the red mangrove essential feature will be adversely affected by this category of activity. While the PDCs for Mangroves, Seagrasses, Corals and Hardbottom for All Projects (PDC AP.12) require that pile-supported structures be located to minimize to the loss of mangroves, the PDCs do allow the removal of up to 4 lin ft of mangrove shoreline to accommodate the installation of walkways for residential docks and up to 8 lin ft for ADA compliant public dock walkways.

The USACE believes that only 10 dock projects covered under this Opinion will require the removal of red mangroves in smalltooth sawfish critical habitat in the next 5-year period. If we assume that all of these projects would be 8-lin ft wide to accommodate ADA compliant walkways (worst-case scenario), that would result in the potential removal of 80 lin ft of mangroves. The potential loss of 80-lin ft of the red mangrove essential feature may adversely affect critical habitat and the effects to smalltooth sawfish critical habitat are evaluated in Section 5.

#### Potential Routes of Effect to Gulf Sturgeon Critical Habitat

The USACE anticipates that 724 pile-supported structures may be covered under this Opinion per 5-year period in Gulf sturgeon critical habitat using this Opinion. We estimate that 30,245.1 ft<sup>2</sup> (0.7 ac) of waterbottom area in Gulf sturgeon critical habitat will be covered by pile and buoy placement (Table30). Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

Project Type	Percent	Number of projects (n = 724)	Number of New Piles	Size of Piles (ft²)	Total Estimated Amount of Waterbottom Affected by Piles per 5- year (ft <sup>2</sup> )	Total Estimated Amount of Waterbottom Affected by Piles per 5- year (ac)
New boatlifts	38%	275.12	4	1	1100.48	0.03
New minor dock	26.50%	191.86	20	1	3837.2	0.09
New ATONS and single-pile structures	3.50%	25.34	1	1	25.34	0.00 <sup>23</sup>
Repair/ replacement of minor structures	12%	86.88	0 <sup>24</sup>	N/A	0	0.00
New major structures (supporting up to 50 slips)	9%	65.16	388.00	1	25,282.08	0.58
Repair/ replacement of major structures	11%	79.64	0	N/A	0	0.00
Total	100%	724			30,245.10	0.70
N/A = not applic	able; these	projects are	repair/repla	acements the	hat do not have ne	w impacts.

 Table 30. Estimated Amount of Waterbottom Affected by Pile-Supported Structures and

 Anchored Buoys Occurring Within Gulf Sturgeon Critical Habitat per 5-Year Period

 $<sup>^{23}</sup>$  25.34 ft<sup>2</sup> is approximately 0.0006 ac, which rounded to 2 decimal places is 0.00.

<sup>&</sup>lt;sup>24</sup> We assume repair and replacement structures are of the same size and do not result in additional piles on the waterbottom.

- 1. Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages. Pile-supported structures and anchored buoys may cover and bury bottom substrates containing sturgeon prey species. However, we believe that the effects to this PCE will be insignificant since the estimated 0.7 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine habitat that we estimate support sturgeon prey species within Gulf sturgeon critical habitat. Further, not all of the 0.7 ac of habitat covered or buried may support prey items or serve as preferred foraging habitat. Gulf sturgeon are suction feeders that tend to forage further offshore, in calmer marine and estuarine waters that support their macroinvertebrate prey including brachiopods, mollusks, worms, and crustaceans (Mason and Clugston 1993). During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Though many of the smaller docks will occur within Gulf sturgeon critical habitat, not all will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prev species.
- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from the installation, repair, replacement, or removal of pile-supported structures and anchored buoys; however, we believe the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires turbidity to be monitored and controlled throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. Therefore, we believe that even minor or temporary turbidity generated will have insignificant effects on the water quality feature. Effects to temperate, salinity, pH, hardness, oxygen content and other water quality parameters are not expected to result from pile-supported structure installation, repair, and removal.
- 3. <u>Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages.</u> The placement of piles and buoys can affect sediment quality. The placement of these materials may remove or cover sandy substrate capable of supporting Gulf sturgeon prey. However, we believe the effects to this PCE will be insignificant because the removal of 0.7 ac of area that may contain suitable sediment quality is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we estimate has sediments that support sturgeon prey species

and Gulf sturgeon foraging. Further, not all of the 0.7 ac of habitat lost from piles and buoys placement may have the sediment quality needed to support Gulf sturgeon prey or serve as preferred foraging habitat. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Though some piles and buoys may occur within Gulf sturgeon critical habitat, not all will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats. Pile-supported structures and anchored buoys could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by these activities in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging. However, given the PDCs and exclusion zones, we believe there will be no effect to this PCE. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent interference with Gulf sturgeon entering or exiting a spawning river, PDC A2.11 contains noise limitations for construction activities in Gulf sturgeon critical habitat migratory restriction zones, pile-supported structures and anchored buoy will not create a barrier that would restrict the movement of sturgeon. Therefore, we believe there will be no effect to migratory pathways for spawning the provide structures and anchored buoy will be no effect to migratory pathways.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and there will be no effect to safe and unobstructed migratory pathways, we believe that pile-supported structure and anchored buoy activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

## Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe there will be no effect to Johnson's seagrass critical habitat from mooring fields. Mooring fields in the range of Johnson's seagrass are limited to waters deeper than -13 ft (PDC A2.14). Studies show that Johnson's seagrass occurs in waters shallower than -10 to -13 ft (-3 to -4 m) (NMFS 2007a). Water deeper than -13 ft is not believed to provide the water transparency necessary for enough sunlight to reach the sea floor to support Johnson's seagrass growth. Therefore, no effect is expected from mooring fields in deeper waters too deep to support the growth of Johnson's seagrass.

Johnson's seagrass critical habitat may be affected by pile-supported structures and anchored buoys (outside of mooring fields). Table 7 describes the Johnson's seagrass critical habitat essential features, and of those, the following may be affected (1) water quality, (2) stable, unconsolidated sediments that are free from physical disturbance, and (3) water transparency, as discussed below:

1. <u>Water quality</u>. We believe the effects to water quality will be insignificant since turbidity curtains will be used to contain turbidity during construction, and disturbed sediments are expected to settle out by the completion of the individual project. In a few instances, the

USACE project manager has the ability to waive the turbidity curtain requirement. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. Therefore, we believe that even minor or temporary turbidity generated will have insignificant effects on the water quality feature.

- 2. <u>Stable, unconsolidated sediments that are free from physical disturbance</u>. Based on the assumptions and calculations in Table 27, we estimate that 3,989.56 ft<sup>2</sup> (0.1 ac) of waterbottom area in Johnson's seagrass critical habitat will be affected (covered) from the placement of piles and buoys. The direct impact to 0.1 ac of habitat from pile placement is likely to adversely affect the stable, unconsolidated sediment essential feature. This effect on the stable, unconsolidated sediments essential feature is evaluated in Section 5.
- 3. Water transparency. The water transparency essential feature is described in the final rule designating Johnson's seagrass critical habitat (65 FR 17786). As described in the rule, the feature supports conservation of the species by ensuring it has sufficient light for photosynthesis. In particular, we explained that seagrass diminishes in turbid areas or in colored waters where reduced light limits photosynthesis, and thus sought to protect water of a sufficient transparency to allow sufficient light transmission to support species abundance and distribution. We noted a few potential causes of decreased water transparency, including suspended sediments, water color, and chlorophylls, and we explained that the essential features, including water transparency, may require special management (a factor in designating critical habitat) because of dock, marina, and bridge construction and shading from these structures. We also explained that docking facilities contribute to loss of Johnson's seagrass through shading. Putting this information together, NMFS has evaluated actions that could reduce the amount of sunlight reaching the bottom in Johnson's seagrass critical habitat as affecting the water transparency essential feature. Thus, below, we evaluate if shading from the placement of pile-supported structures and vessels moored at these structures reduces the amount of light able to reach the bottom to an extent that it affects the abundance and distribution of the species and affects the water transparency essential feature.
  - <u>New and expanded docks</u>: To reduce the effects of shading from docks, PDC A2.17 requires that all new docks or dock expansions built in Johnson's seagrass critical habitat be constructed of decking that allows light transmission under the dock. These guidelines are designed to reduce the effects of shading on seagrass growth. To maximize light transmission under the structure, the docks must either be constructed of grated decking that is elevated to a minimum of 3 ft above MHW and that provides a minimum of 43% open space, or of plank decking spaced ½ in apart and elevated to a minimum of 5 ft above MHW. As discussed above for Johnson's seagrass effects from shading, Johnson's seagrass was found to persist under docks constructed of grated decking with no statistical difference in density compared to reference sites (Landry et al. 2008). Another

study considered light availability and dock height and found "a significant positive correlation between dock height and light availability in the seagrass canopy in the shaded areas beneath the dock walkway" (i.e., the higher the dock, and the more space between the dock and MHW, the more light was able to reach the seagrass (Shafer et al. 2008)). Results showed a reduction in the frequency of occurrence of most seagrass species, including Johnson's seagrass, under docks that were not fully built to the deck height requirements of the dock construction guidelines (Shafer et al. 2008). Yet, Shafer et al. (2008) stated that their data "...demonstrate that *H. johnsonii* is capable of growing under the low light conditions found in the shaded areas beneath dock walkways." Therefore, we believe the addition of new or expanded docks built to these guidelines that provide either grated decking (allowing a minimum of 43% open space to allow light through the deck surface) or dock heights of at least 5 ft (that is elevated sufficiently to allow sunlight under the structure during the majority of the day) will continue to provide adequate light availability for Johnson's seagrass growth, and support the water transparency essential feature. As a result, we believe docks built in compliance with these guidelines will have an insignificant effect on the water transparency essential feature.

- <u>Repair and replacement of existing docks</u>: Under PDC A2.17, all repairs to existing docks rebuilt with the same size/area must be built with grated or plank decking that allows light transmission except for replacement docks of the same size/same footprint where no seagrasses (either Johnson's or non-ESA listed seagrasses) are identified under the existing structure. These replacements can use any type of decking material, deck spacing, and deck height and are not subject to the grated or plank decking requirements described above. We do not believe these replacement projects will result in any additional shading beyond the shading from the existing structures, and thus will not contribute additional effects to the water transparency essential feature. Therefore, we believe the replacement of docks in the same footprint will have no effect on the water transparency essential feature.
- <u>Vessels stored at docks</u>: Vessels moored at overwater structures can shade the waterbottom and thus can affect the water transparency essential feature of Johnson's seagrass critical habitat. In this Opinion, we reevaluated the number of vessels registered in Florida and specifically in counties adjacent to Johnson's seagrass critical habitat (Section 2.2, Vessel Strikes). Our analysis showed that the number of registered vessels in Johnson's seagrass critical habitat and throughout Florida has remained statistically unchanged over the last 10 years. Although the USACE continues to permit the installation, repair, and replacement of docks and marinas, the number of vessels registered in Florida and Johnson's seagrass critical habitat area has not changed. We assume that this is because (1) most old vessels are ultimately replaced with new vessels; (2) many new docks and marinas likely replace older structures so a permit to construct a new structure does not necessarily mean the new structure will support new vessels; and (3) vessels stored at 1 location may be relocated to a new located, but are not new vessels to the state of Florida or specific area like Johnson's seagrass critical habitat. Also, new residential docks may be built that do not support vessel dockage or storage. Although it

is possible that a vessel may be relocated from dry storage to a newly permitted wet slip, overall, we do not believe the amount of vessel shading is changing within Johnson's seagrass critical habitat. Therefore, we believe there will be no new or additional effect to the water transparency essential feature from shading from vessels moored at overwater structures authorized under Activity 2.

Therefore, of the essential features of Johnson's seagrass that may be affected by Activity 2, we conclude the water quality and water transparency essential features are not likely to be adversely affected. The stable, unconsolidated sediments essential feature is likely to be adversely affected and the effects to that feature are evaluated in Section 5.

## 2.2.3 Activity 3 (A3): Maintenance, Minor, and Muck Dredging

### **General Description**

This Opinion is limited to maintenance, minor, and muck dredging as described below.

#### Maintenance Dredging

For this Opinion, we define maintenance dredging as the dredging of an area previously dredged under a permit by the USACE at the site of the previously authorized dredging, to the same depth and within the same footprint, but under a new authorization. Maintenance dredging may be necessary around docks to moor vessels, around boat ramps, and around other structures necessary to maintain adequate depth for vessel movement. Dredged canals require regular maintenance dredging to maintain the original width and depth. Canals tend to shoal in overtime either slowly from daily water movement or suddenly from stochastic events such as hurricanes. We have classified canals into 3 categories for the purpose of evaluating maintenance dredging:

- Federal channels- Major waterways authorized and maintained by the federal government to accommodate vessel traffic, including the Intracoastal Waterway (ICW), Atlantic Intracoastal Waterway (AIWW), Gulf Intracoastal Waterway (GIWW), shipping channels, and main channels in harbor and ports. These federal channels are often maintained by larger equipment including hopper dredges. Maintenance dredging of federal channels is not covered under this Opinion because dredging of large navigational channels like the ICW involves large areas and volumes of dredged material that are beyond the scope of this Opinion. ESA consultation for many federally maintained navigational channels is addressed through other consultations such as SARBO (NMFS 1997) and GRBO (NMFS 2007b). Minor dredging in areas adjacent to federal channels is covered under this Opinion (e.g., dredging around a dock at a private residence located along the ICW), as long as the dredging is outside of the federally maintained center channel.
- 2. Open-water canals- These are mid-size channels that connect federal channels to confined channels, such as residential canals.
- 3. Confined channels- These are channels confined on both sides by land and include residential canals (e.g., Cape Coral canals) and smaller rivers that do not support significant vessel traffic.

Though the top width of open-water canals and confined channels varies by location, the width that canals are maintenance dredged is typically limited to 30 ft wide at the top of the cut and 20 ft wide at the bottom or the cut, as shown in Figure 13 below.



Figure 13. Canal dredging footprint drawing.

This diagram shows the standard 20-ft bottom dredging footprint and 5-ft side slopes for a total of a 30-ft-wide dredging footprint within a confined channel armored on both sides by seawalls. Note that the sides of the channels remain shallow and only the center of the canal is dredged for vessel navigation through the canal.

#### Minor Dredging

For this Opinion, we define minor dredging as any newly-authorized dredging (other than maintenance dredging) that is limited in size and depth according to the PDCs for Activity 3. Minor dredging may be required to accommodate vessel movement in an area or may be related to another activity such as for the installation of seawalls or outfall structures. Minor dredging may also include treasure hunting and salvage operations that use blowers, propeller deflectors, and suctioning devices.

#### Muck Dredging for Water Quality Enhancement

For this Opinion, we define muck dredging as any dredging that involves the removal of muck sediments alone. Muck sediments are defined as accumulated organic material typically found in areas with poor water quality. Equipment used for muck dredging can include hydraulic vacuum dredge, bucket dredge, or other similar dredging equipment. Muck dredging is used to improve the water quality or for restoration projects and is not intended to increase water depths to support vessel mooring. Muck dredging is limited to the PDCs listed below.

#### General Description of Dredging Methods

#### Mechanical Dredging

Mechanical dredges are characterized by the use of some form of bucket or clamshell that excavates material by scooping it from the bottom and then raising the bottom material and placing it onto a waiting barge or directly into a placement/disposal area (Figure 14). Mechanical dredges work best in consolidated, or hard-packed, substrate and can be used to clear rocks and debris. Dredging buckets have difficulty retaining loose, fine substrate, which can be washed from the bucket as it is raised through the water column. Special buckets have been designed for controlling the flow of water and material from buckets and are used when dredging contaminated sediments to minimize the spread of contamination. Mechanical dredges are rugged and can work in tightly confined areas. They vary in size from small equipment mounted on shallow-draft barges with limited bucket size (i.e., with capacities as small as 1 cubic yard) to larger equipment arrays mounted on a large barges with bucket capacities up to ten or more cubic yards), towed to the dredging site and secured in place by anchors or spuds. They are often used in harbors, around docks and piers, and in relatively protected channels, but are not suited for areas of high traffic or rough seas.

Dipper dredges and clamshell dredges, named for the scooping buckets they employ, are the 2 most common types of mechanical dredges (Figure 14). A bucket dredge begins the digging operation by dropping the bucket in an open position from a point above the sediment. The bucket falls through the water and penetrates into the bottom material. The sides of the bucket are then closed and material is sheared from the bottom and contained in the bucket compartment. The bucket is raised above the water surface, swung to a point over the barge, and then released into the barge by opening the sides of the bucket. Usually, 2 or more disposal barges are used in conjunction with the mechanical dredge. While 1 barge is being filled, another is being towed to the dumpsite by a tug and emptied. If a diked disposal area is used, the material must be unloaded using mechanical or hydraulic equipment. Using numerous barges, work can proceed continuously, only interrupted by changing dump barges or moving the dredge. This makes mechanical dredges particularly well suited for dredging projects where the disposal site is many miles away. The dipper dredge is essentially a power shovel mounted on a barge. It can dig hard materials and has all the advantages of the bucket dredge, except for its deep digging and sea state capabilities. Similar to the bucket dredge operation, the dipper dredge places material into a barge, which is towed to a disposal area (USACE 1993).



Figure 14. Mechanical dredge (clamshell bucket and barge).

## Hydraulic Dredging

Hydraulic dredging (also referred to as cutterhead or pipeline dredging) is characterized by the use of a pump to dredge sediment and the transportation of the dredged material slurry and water to identified discharge areas (Figure 15). The ratio of water to sediment within the slurry mixture is controlled to maximize efficiency. The main types of hydraulic dredges are pipeline and hopper dredges.

Pipeline dredges are designed to handle a wide range of materials including clay, hardpan, silts, sands, gravel, and some types of rock formations without blasting. They are used for new work and maintenance in projects where suitable placement/disposal areas are nearby and operate in an almost continuous dredging cycle resulting in maximum production, economy, and efficiency. Pipeline dredges are capable of dredging in shallow or deep water and have accurate bottom and side slope cutting capabilities. Limitations of pipeline dredges include relative lack of mobility,

long mobilization and demobilization times, inability to work in high wave action and currents, and they are impractical in high traffic areas.

Pipeline dredges are rarely self-propelled, and typically must be transported to and from the dredge site by barge or tow. Pipeline dredge size is based on the inside diameter of the discharge pipe which commonly ranges from 6-36 in. They require an extensive array of support equipment including the pipeline (floating, shore, and submerged), boats (crew, work, survey), barges, and pipe handling equipment. Most pipeline dredges have a cutterhead on the suction end. A cutterhead is a mechanical device that has rotating teeth to break up or loosen the bottom material so that it can be sucked through the dredge. Some cutterheads are rugged enough to break up rock for removal (Figure 15).



## Hydraulic Cutterhead Dredge



Figure 15. Cutterhead pipeline dredge schematic and representative close-up photographs (provided by USACE).

During the dredging operation, a cutterhead suction barge is held in position by 2 spuds at the stern of the dredge, only 1 of which can be on the bottom while the dredge swings. There are 2 swing anchors some distance from either side of the dredge, which are connected by wire rope to the swing winches. The dredge swings port and starboard alternately, passing the cutter through the bottom material until the proper depth is achieved. The dredge advances by "walking" itself forward on the spuds. This is accomplished by swinging the dredge to the port, using the port

spud an appropriate distance, then the starboard spud is dropped and the port spud is raised. The dredge is then swung an equal distance to the starboard, the port spud is dropped, and the starboard spud is raised.

Cutterhead pipeline dredges work best in large areas with deep shoals, where the cutterhead is buried in the bottom. A cutterhead removes dredged material through an intake pipe and then pushes it out the discharge pipeline directly to the placement/disposal site. Most, but not all, pipeline dredging operations involve upland placement/disposal of the dredged material. Therefore, the discharge end of the pipeline is connected to a shore pipe. When effective pumping distances to the placement/disposal site become too long, a booster pump is added to the pipeline to increase the efficiency of the dredging operation.

## Transportation Methodology

Dredged material is typically transported by barge and then transferred to a land-based dump truck for disposal in upland locations. In some instances, the material is barged to an approved water location or beneficial use site. Methods of transporting dredged material to disposal sites include self-propelled transport via barges or towing of loaded barges to disposal sites via tugboats. Tugboats may be used to move immobile equipment into place as well as tow loaded barges to the disposal sites. Dredged material may also be transported by pipeline as described above under hydraulic dredging.

## **Disposal Locations**

Dredged material can be disposed of in a USACE verified upland disposal sites, USACEpermitted beneficial reuse sites, existing/authorized Dredged Material Management Areas, or U.S. Environmental Protection Agency (EPA)-designated ocean dredged material disposal sites. Beneficial reuse sites are often areas of eroding shorelines and marshes or seagrass restoration areas where dredged material is used to return the area to a water depth that supports seagrasses. The disposal location is dependent of the type of material dredged, the proximity to the dredging locations, and permitting requirements. Beneficial reuse sites and ocean disposal sites used for projects covered under this Opinion must have undergone Section 7 consultation (either individual or programmatic) to evaluate the potential effects of disposal on ESA-listed species and critical habitat. Some beneficial reuse sites, such as living shorelines, are covered under Activity 7 of this Opinion. Beach renourishment placement is also considered a beneficial use of dredged material.

## Project Design Criteria

## PDCs specific to Activity 3 (maintenance, minor, and muck dredging):

A3.1. Activities covered by this Opinion include:

A3.1.1. Maintenance dredging of existing areas such as canals, channels, basins, berths, marinas, boat slips, and areas around intake and discharge structures. Maintenance dredging will be limited to the depth and width previously authorized by the USACE or other regulatory authority such as FDEP or water management districts. There is no size limit for maintenance dredging so long as it meets the previously authorized depth and width. If the previous authorized depth is unknown, dredging is limited to -7.0 ft MLW including any advanced maintenance or overdredge.

- A3.1.2. Minor dredging (non-maintenance) dredging as follows:
  - A3.1.2.1.For dredging to accommodate vessel mooring at boat slips around docks and marinas, the size is limited to the minimum necessary to accommodate vessel mooring, not to exceed 5,000 ft<sup>2</sup>. The maximum allowed dredging depth (including overdredge) cannot exceed the depth of the adjacent navigational channel (i.e., controlling depth) or a maximum depth of -7 ft MLW. For projects located adjacent to federal channels (not within the dredged navigational channel), the dredging depth can exceed -7 ft MLW so long as it does not exceed the controlling depth of the federal channel.
  - A3.1.2.2.For treasure hunting and salvage operations that use blowers, propeller deflectors, and suctioning devices, the size is limited to a total of 5,000 ft<sup>2</sup> and is limited to sandy areas only.
  - A3.1.2.3.Minor dredging does not include dredging to create new navigational channels.
  - A3.1.2.4. Minor dredging includes dredging to accommodate the placement of shoreline stabilization, outfall structures; boat ramps; upland cut boat ramps and basins; temporary platforms, fill, and cofferdams; and placement of erosion and scour control-measures. This type of dredging is usually required to embed materials and/or to allow smooth transition of the work to the natural surrounding elevation.
- A3.1.3. Muck Dredging, or removing accumulated organic to restore natural habitats and for water quality enhancement, as follows:
  - A3.1.3.1.Muck dredging cannot be used to increase water depths to support navigation, access, or vessel mooring.
  - A3.1.3.2.Dredging depths are limited to only that necessary to remove the muck layer down to natural sediments.

## The following PDCs apply to all the activities described in PDC A3.1 above:

- A3.2. This Opinion does not cover hopper dredging.
- A3.3. With respect to the dredged material, all dredged material must be placed in an USACEverified upland disposal site, EPA or USACE-designated open water disposal site, USACE Dredged Material Management Area, or USACE-approved beneficial use sites for mitigation or restoration, as long as it meets PDC A3.3.1-A3.3.3 below.
  - A3.3.1. The disposal sites shall employ erosion control measures such as upland erosion control, such as filtration or berms, or in-water turbidity curtains.
  - A3.3.2. Handling and storage of dredged material must be completed in a manner that prevents sedimentation, erosion, and turbidity during dewatering, overflow, transferring, and storage of the dredged material. For example, the overwater transfer of dredge material should either contain the dredged material and any water to prevent sedimentation or employ other methods, such as turbidity curtains in the marine environment, to ensure that any turbidity generated as the water is returned to the marine environment is contained. If the applicant conducts sediment testing voluntarily or in

compliance with other law, and such testing indicates high levels of contaminants in the sediments to be dredged, water from dewatering should not be released back into the marine environment.

- A3.3.3. This Opinion does not cover the use of in-water disposal sites (e.g., beneficial use sites or ocean disposal sites) unless the use of the in-water disposal sites has previously undergone ESA-Section 7 consultation with NMFS for disposal of material at these locations. If the applicant is seeking disposal in an in-water disposal location not previously consulted on by NMFS, then the entire project (both dredging and disposal) must be consulted on separately and is not covered under this Opinion.
- A3.3.4. This Opinion applies to upland disposal of beach quality sand on beaches if placed above the existing MHW, if the grain size analysis indicates that the dredged sand is compatible with the existing beach sand, and if the sand placement does not change the existing waterward extension of the beach. Placement of beach sand on nesting beaches above MHW that may affect ESA-listed species is under the jurisdiction of the USFWS.
- A3.4. This Opinion does not cover dredging within the mapped and authorized federal navigational channels (e.g., ICW, AIWW, GIWW, or harbors [e.g., Port Canaveral]). Dredging outside of the mapped channel in the surrounding waters is covered.

## Additional PDCs for Activity 3 applicable in critical habitat:

# In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat as described below.

- A3.5. Smalltooth sawfish critical habitat:
  - A3.5.1. Maintenance dredging of canals in smalltooth sawfish critical habitat is covered under this Opinion as long as it is within the previously authorized dredge footprint and to the previously USACE authorized depth.
  - A3.5.2. For minor dredging: If only the shallow euryhaline (MHWL to -3 ft MLLW) water essential feature is present (i.e., no red mangroves), dredged depths are limited to a maximum depth of -3 ft MLLW. If red mangroves are present, dredging, excavation, or disposal is not allowed within 5 ft of all red mangrove prop roots.
  - A3.5.3. Muck dredging, as defined in PDC A3.1.3, is not allowed in shallow, euryhaline habitat (MHWL to -3 ft MLLW)
  - A3.5.4. Dredging and disposal activities are not allowed in areas identified as smalltooth sawfish limited exclusion zones, defined in Section 2.1.1.1.
- A3.6. Gulf sturgeon critical habitat: No treasure hunting or muck dredging is allowed in Gulf sturgeon critical habitat at any time of year. No maintenance or minor dredging is allowed in Gulf sturgeon critical habitat between September and March, when sturgeon are likely to be present in these areas. When allowed, maintenance and minor dredging activities shall be conducted according to the PDCs above for all dredging activities.

- A3.7. *Acropora* critical habitat and the U.S. Caribbean: This Opinion does not cover dredging (maintenance, minor, or muck) and disposal in *Acropora* critical habitat where the essential features are present or within the U.S. Caribbean. This Opinion does not cover projects requiring penetrating or removing underlying hard substrate (e.g., bedrock, hardbottom) using any methods including blasting or fracturing. Treasure hunting is not allowed in *Acropora* critical habitat or the U.S. Caribbean.
- A3.8. Johnson's seagrass critical habitat: Treasure hunting is not allowed in waters less than 12 ft (4 m) deep. Muck dredging is not allowed if the essential features are present. All other dredging is covered if conducted according to the PDCs above for all dredging activities.
- A3.9. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat) and loggerhead sea turtle critical habitat: Dredging consistent with the PDCs above can occur in these critical habitat areas.

#### Assumptions

Maintenance dredging: Estimating the areal extent of areas affected by maintenance dredging is difficult because the USACE tracks dredging and disposal by volume, in cubic vards. Currently the USACE only has limited data on the area of impact. The USACE estimates that the average maintenance dredging is 5 ac per project. However, this could vary widely since maintenance dredging is not limited by area (meaning we have not set a limit on the amount of dredging in a particular location), but instead is limited to the previously authorized footprint and depth. Maintenance dredging in smalltooth sawfish critical habitat: Maintenance dredging of navigational canals within smalltooth sawfish critical habitat is performed by the West Coast Inland Navigational District (WCIND) as the local sponsor used by the USACE to perform and manage dredging in this area. WCIND provided NMFS with GIS data regarding all of the maintenance dredging of navigational canals within the smalltooth sawfish critical habitat for the CHEU. This data was based on studies under the Florida Sea Grant (Antonini et al. 2000; Fann et al. 2002; Swett et al. 2012; Swett et al. 2000a; Swett et al. 2000b; Swett et al. 2002; Swett et al. 2002 al. 2000c). According to the WCIND studies, if they deepened all of the canals that have areas that are currently less than -3 ft MLLW, this would result in the potential loss of 12,742,560 ft<sup>2</sup> (292.53 ac) of the shallow, euryhaline essential feature. Though it is unlikely that all of the canals in smalltooth sawfish critical habitat will be maintenance dredged per 5-year period, we consider what the cumulative effect would be from removing the essential feature in these areas.

Maintenance dredging in Johnson's seagrass critical habitat: According to the USACE, these projects can range from larger projects that maintenance dredge long stretches of navigational channels, which may affect up to 12,000 ft<sup>2</sup>, to smaller projects that maintenance dredge around a boat slips and smaller structures, which may affect only a couple hundred square feet. Based on their records, the USACE estimates that the average maintenance dredging project is approximately 1,200 ft<sup>2</sup>.

Maintenance dredging and Johnson's seagrass: The USACE has very few records of maintenance dredging projects that resulted in the loss of Johnson's seagrass. However, based

on their records, the USACE estimate maintenance dredging could result in a maximum loss of 4,356 ft<sup>2</sup> (0.1 ac) of Johnson's seagrass per year or 21,780 ft<sup>2</sup> (0.5 ac) per 5-year period.

Minor new dredging: The PDCs limit the size and depth of minor new dredging. The USACE estimates that the average minor dredging project will be 2,500 ft<sup>2</sup>. The difference for minor versus maintenance dredging is that maintenance dredging is by definition limited to previously dredged areas and will not result in impacts outside of previously disturbed areas.

Muck dredging: The size and depth of muck dredging is limited to the amount necessary to return the area to the natural depth contour by removing accumulated organic material. The USACE has not previously differentiated muck dredging from other types of dredging in their tracking systems so they do not know how many or how big muck dredging projects will be that are covered under this Opinion. Muck dredging was differentiated for this Opinion because more restoration type projects have been described as muck dredging lately. Because the dredging does not ultimately change the natural depth feature of an area, we classified it differently from maintenance and minor dredging and applied different PDC requirements. The USACE assumes that muck dredging makes up only a small fraction (5%) of all dredging projects because of its very specific nature to remove accumulated sediments that contribute to poor water quality that would only occur in specific areas with poor tidal flushing. These projects can be highly variable in size depending on site conditions. Projects could range from only the terminus of a channel that may be less than 1 ac or could be a large section of a stagnant channel or bay that could be 50 ac or more. For example, the Monroe County Canal Management Master Plan (AMEC Environment & Infrastructure 2013) estimates that muck dredging in canals in the Florida Keys will remove approximately 2.6 ac of material per canal while muck dredging projects in the Indian River Lagoon have been proposed that are approximately 1,000 ac. The USACE believes that larger projects will be rare and that on average, they will impact 25 ac. Due to the range of potential sizes of these types of projects, the PDCs have specific restrictions in critical habitat and sensitive areas.

Estimated areal extent (area) of maintenance, minor, and muck dredging activities is used, in combination with the estimated number of projects anticipated both in and outside of critical habitat (Table 8) to determine the potential effects from this category of activity. USACE anticipates that 1,320 dredging projects will be covered under this Opinion per 5-year period of which 74% will be maintenance dredging, 21% will be new minor dredging, and 5% will be muck dredging.

Thus, to analyze the effects of dredging activities, we looked at the USACE's assumptions regarding (1) the total number of maintenance, minor, and muck dredging activities to be covered under this Opinion per 5-year period; and (2) the anticipated size of the dredging areas based on dredging type and location. Based on this information, we determined the potential effects to species (Tables 31) and critical habitat (Table 33).

## Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing boat ramp projects would result in the following common routes of effect discussed above\_in the section entitled *Construction Related Effects for All Categories of* 

*Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from maintenance, minor, and muck dredging that was not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 31. Sea turtles, smalltooth sawfish, sturgeon, and Johnson's seagrass may be affected by the permanent loss of habitat from dredging. We quantified the potential extent of habitat impacts based on (1) the assumptions, (2) the number and type of projects estimated by USACE (Table 8), and (3) the limitations defined by the PDCs. Table 32 provides the estimated amount of waterbottom affected by dredging activities. This was calculated for each of the 3 "types of dredging" activities (i.e., maintenance, minor, and muck). The USAE provided the estimated total number of projects that will be covered per 5-year period (1,320 projects) and the percent of dredging expected to be in each of the 3 dredging categories. With this information, we calculated the "total number of projects." This was then multiplied by the "estimated average dredged area per project" to determine the "total area dredged."

Table 31. USACE and NMFS Determinations on the Effects of Dredging Activities (Activity3) to ESA-listed Species listed in Table 5

Listed Species	<b>USACE Determination</b>	NMFS Determination
Sea Turtles	NLAA	NLAA
Sawfish	NLAA	NLAA
Sturgeon	NLAA	NLAA
Johnson's seagrass	LAA	LAA
Effects	Determinations Explained in	Section 2.2
Corals	NE	N/A
Whales	NE	NE
Nassau grouper	Not provided	NE

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

Type of Dredging	Percent of Projects	Total Number of Projects (n = 1,320)	Estimated Average Area Dredged per Project (ac)	Total Area Dredged (ac)
Maintenance	74%	976.80	5	4,884.00
Minor	21%	277.20	0.06	16.63
Muck	5%	66.00	25	1,650.00
Total		1,320		6,550.63

 Table 32. Estimated Amount of Waterbottom Affected by Dredging Activities Per 5-Year

 Period.

Over a 5-year period, 6,550.63 ac of waterbottom area may be permanently altered or removed by the dredging activities. This estimate includes projects in both Florida and the U.S. Caribbean. For the 3 types of dredging, the total area dredged can have different effects on foraging and refuge habitat, as summarized below and explained in more detail for each species.

- Minor dredging: The 16.63 ac impacted by minor dredging represents new, permanent changes to waterbottom, resulting in the potential loss of foraging and refuge habitat.
- Maintenance dredging: The 4,884 ac of estimated maintenance dredging per 5-year period is limited to the same footprint as the already dredged areas, most of which are navigational channels. Previously dredged channels continue to shoal in due to natural wave action and require regular maintenance. Depending on the maintenance dredging interval/ timeframe, these areas can recruit with seagrasses and other benthic resources between dredging cycles. Due to the continued need to maintenance dredge channels to maintain navigational access, the process of them continuing to shoal in, and the continued recruiting of seagrasses and other prey resources back into the channel, we consider maintenance dredging to be a temporary impact.
- Muck dredging: The estimated 1,650 ac of muck dredging are limited to restoring areas to the previous depth in an attempt to improve water quality and likely habitat for species by removing accumulated sediments. We consider this to improve foraging and refuge habitat.

Also, the impacts to waterbottom area are unlikely to be concentrated as the array of individual projects covered under this Opinion under Activity 3 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S Caribbean).

• <u>Sea Turtles:</u> Sea turtles may be affected by the loss of 6,550.63 ac of waterbottom areas due to dredging as these areas could be used for foraging and potentially refuge. Of the 6,550.63 ac impacted, 1,650 ac will be muck dredged. The areas to be muck dredged are covered in accumulated organic material and thus would not support foraging resources due to the poor sediment and water quality in these areas. Therefore we do not believe muck dredging will

affect sea turtles by affecting their forage resources, except to improve the habitat in this area that may eventually recruit foraging resources for sea turtles in the future.

We believe the effect on sea turtles from the potential loss of foraging and refuge habitat from maintenance and minor dredging is insignificant. As discussed above, maintenance dredging will temporarily affect 4,884 ac of previously dredged areas that may not serve as high quality foraging and refuge habitat and resources that may have recruited back into the previously dredged channel will continue to be removed by maintenance dredging and reestablish again based on the dredging interval. Minor dredging 16.63 ac of new areas, may also affect foraging resources. The temporary effects from maintenance dredging and the permanent effects from minor dredging areas to sea turtle foraging resources are discussed below:

- The waterbottom affected by Activity 3 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 limits effects to Johnson's seagrass from maintenance dredging (PDC AP.13 limits to a maximum removal of 0.1 ac per year). In addition, PDC AP.13 recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
- Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
- Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly and indirectly affects listed corals, and excludes projects if non-listed corals and hardbottom habitat, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
- Dredging activities may remove areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These effects are temporary as seagrasses and macroalgae will likely regrow in these areas. In addition, these foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. The area of impact is small compared to the remaining large nearshore areas that support sea turtle prey species, and sea turtles can travel long distances to forage. Thus, we believe that loggerhead and Kemp's ridley sea turtles will be able to find foraging resources outside of the area affected by the projects.
- Sea turtles may also use channels to thermal regulate by entering deeper channels in the summer to avoid warmer surface waters and entering deeper water in the winter where waters may be warmer than winter surface temperatures. The inability to access these channels during dredging would be temporary and dredging would not occur throughout the entire reach of a channel at the same time allowing other channel areas to thermal regulate.

Given the above, we believe the effect to sea turtles from the potential loss of foraging and refuge habitat is insignificant. In total, based on the routes of effect analyzed here and in

Section 2.2, we determined that the dredging activities under Activity 3 are not likely to adversely affect sea turtles.

<u>Smalltooth sawfish:</u> Smalltooth sawfish may be affected by the loss of 6,550.63 ac of waterbottom area due to dredging, which could be used for foraging and refuge. However, of the estimated 6,550.63 ac of dredging, 1,650 ac of muck dredging will be in areas of accumulated organic material that also was not likely used by sawfish for forage and refuge activities. Since smalltooth sawfish are bottom dwelling species, movement through a layer of accumulated sediments would likely be difficult and the areas to be dredged likely would lack foraging resources because of the low sediment and water quality. Therefore, we believe muck dredging will not affect smalltooth sawfish by affecting their forage resources.

We believe the effect on sawfish of the potential loss of nearshore foraging and refuge habitat from maintenance and minor dredging is insignificant. Maintenance dredging will affect 4,884 ac of waterbottom. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. The areas to be dredged are not likely to provide quality habitat for smalltooth (e.g., maintenance dredging canals, marinas, around outfall structures). For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for refuge; these areas have been designated as critical habitat for the species. The PDCs limit activities in smalltooth sawfish critical habitat and prohibit activities in the limited exclusion zones (PDC A3.5). Smalltooth sawfish seek refuge and forage among red mangroves, and PDC AP.12 states that mangrove removal covered under this Opinion is limited to certain dock and outfall structure installation and does not include removal of mangroves for dredging activities. In addition, PDC AP.12 allows removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. Dredging may remove nearshore areas inhabited by fish and crustaceans that serve as prey for smalltooth sawfish. The area of impact (some amount less than 4,884) is small compared to the other nearshore areas that support sawfish prey species. Sawfish can travel long distances to forage. The array of individual projects covered under this Opinion for Activity 3 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites. Therefore, we believe the effect on smalltooth sawfish from impacts to foraging and refuge resources from minor and maintenance dredging will be insignificant. Based on the routes of effect analyzed here and in Section 2.2, we determined that dredging activities under Activity 3 are not likely to adversely affect smalltooth sawfish.

• <u>Sturgeon:</u> Sturgeon may be affected by dredging of 6,550.63 ac of waterbottom area that may be used for foraging. However, of the estimated 6,550.63 ac of waterbottom to be dredged, the 1,650 ac to be muck dredged will be in areas of accumulated organic material that also will not likely be used by sturgeon for foraging activities. Since sturgeon are bottom dwelling species, movement through a layer of accumulated sediments would likely be difficult and we expect these areas would lack foraging resources because of the low

sediment and water quality. Thus, we do not believe muck dredging will affect sturgeon by affecting their forage resources.

We believe the effect on sturgeon of the potential loss of foraging habitat from minor and maintenance dredging is insignificant. The 4,884 ac to be maintenance dredged are in areas that are not likely to provide high quality habitat for sturgeon (e.g., maintenance dredging canals, marinas, around outfall structures). As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. The PDCs allow dredging within areas utilized by Gulf sturgeon, including critical habitat. However, the PDCs preclude dredging in the estuaries and bays within Gulf sturgeon critical habitat between September and March, when sturgeon are likely to be present in these areas (PDC A3.6). When sturgeon return from spawning rivers, the dredging could affect the availability of foraging resources; however, we believe the effect on sturgeon will be insignificant because sturgeon are opportunistic feeders that forage over large areas and will be able to locate prey beyond the small minor dredging footprints and maintenance of existing channels. Also, effects foraging resources from dredging are temporary since benthic invertebrate populations in dredged areas have been observed to recover in 3-24 months after dredging (Culter and Mahadevan 1982; Saloman et al. 1982; Wilber et al. 2007).

Atlantic and shortnose sturgeon are limited in Florida to the St. Marys River and the St. Johns River. This Opinion does not apply to projects in the St. Marys River (Section 2.1.1.3), where Atlantic sturgeon spawn. Neither Atlantic nor shortnose sturgeon are known to spawn in the St. Johns River, so dredging in this river will not affect sturgeon spawning or migrating to foraging areas after leaving spawning rivers in Florida.

The array of individual projects covered under this Opinion for Activity 3 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Therefore, we believe the effect on sturgeon from impacts to foraging resources from minor and maintenance dredging will be insignificant. Based on the routes of effect analyzed here and in Section 2.2, we determined that dredging activities under Activity 3 are not likely to adversely affect sturgeon.

Johnson's seagrass: Johnson's seagrass may be affected by dredging activities evaluated in this Opinion. The USACE estimates that maintenance dredging could result in a maximum loss of 4,356 ft<sup>2</sup> (0.1 ac) of Johnson's seagrass per year or 21,780 ft<sup>2</sup> (0.5 ac) per 5-year period. PDC AP.13 limits removal of Johnson's seagrass to a maximum of 0.1 ac per year for all maintenance dredging activities that USACE may authorize under this Opinion. Although the impacts are limited by the PDCs, maintenance dredging activities evaluated in this Opinion is likely to adversely affect Johnson's seagrass. The loss of Johnson's seagrass from maintenance dredging is evaluated in Section 5.

PDC AP.13 does not allow impacts to Johnson's seagrass from muck dredging or minor dredging so we believe there will be no effect to the species from these types of dredging. In

particular, PDC AP.13 does not allow these activities to occur where Johnson's seagrass is within the project footprint.

## Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from dredging activities. Table 7 describes the specific features of critical habitat evaluated in this Opinion. The estimated impacts to each critical habitat unit (Table 34) and NMFS and USACE effects determinations for this category of activity are summarized in Table 33.

# Table 33. USACE and NMFS Determinations on the Effect of Dredging Activities(Activity 3) on Designated Critical Habitat

Project Location	USACE Determination	NMFS Determination			
Sawfish critical habitat	NLAA	NLAA			
Gulf sturgeon critical habitat	NLAA	NLAA			
Johnson's seagrass critical habitat	LAA	LAA			
Effects Determinations Explained in Section 2.2					
Loggerhead critical habitat	NE	NE			
Acropora critical habitat	NE	NE			
North Atlantic right whale critical habitat	NE	NE			
Atlantic sturgeon critical habitat Not provided NE					
The effects analysis for loggerhead, <i>Acropora</i> , North sturgeon critical habitat was provided at the beginning	h Atlantic right whale, ng of Section 2.2.	and Atlantic			

 Table 34. Total Waterbottom Affected by Dredging Projects within Smalltooth Sawfish,

 Gulf Sturgeon, and Johnson's Seagrass Critical Habitat

Type of Dredging	Percent of Projects	Total Number of Projects per Critical Habitat	Estimated Average Area Dredged per Project	Total Area Dredged (ft <sup>2</sup> )	Total Area Dredged (ac)
		Smalltooth sawfi	sh critical habitat	(n = 150)	
Maintenance	74%	111.00	WCIND determined all maintenance = 12,742,560 ft <sup>2</sup>	12,742,560.00	292.53
Minor	21%	31.50	2,500	78,750.00	1.81
Muck	5%	7.50	1,089,000	8,167,500.00	187.50
Total	100%	150		20,988,810.00	481.84
		Gulf sturgeon	critical habitat (i	n = 51)	
Maintenance	74%	37.74	217,800	8,219,772.00	188.70
Minor	26%	13.26	2,500	33,150.00	0.76
Muck	PDC.	A3.6 does not cove	er muck dredging	in Gulf sturgeon c	ritical habitat
Total	100%	51		8,252,922.00	189.46
		Johnson's seagra	ass critical habita	t (n = 31)	
Maintenance	74%	22.94	1,200.00	27,528.00	0.63
Minor	21%	6.51	2,500.00	16,275.00	0.37
Muck	5%	1.55	1,089,000	1,687,950.00	38.75
Total	100%	31		1,731,753.00	39.75

## Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that there will be no effect to the red mangrove essential feature of smalltooth sawfish critical habitat from any dredging covered under Activity 3. USACE anticipates that 150 dredging activities may be covered under this Opinion per 5-year period in smalltooth sawfish critical habitat. The PDCs for *Mangroves, Seagrasses, Corals and Hardbottom for All Projects* (PDC AP.12) prohibit the removal of mangroves, including red mangroves which are an essential feature of smalltooth sawfish critical habitat, from any dredging project (i.e., mangrove removal is limited to other, non-dredging projects, and removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed). In addition, PDC A3.5.2 states that for minor dredging, if red mangrove prop roots.

We believe there will be no effect to the shallow, euryhaline essential feature of smalltooth sawfish critical habitat from the estimated 292.53 ac of waterbottom area in smalltooth sawfish critical habitat to be maintenance dredged (Table 34). Maintenance dredging under this Opinion is limited to the areas in smalltooth sawfish critical habitat that have been previously dredged and is limited to the previously authorized depth and footprint (PDC A3.5.1). As stated in the Smalltooth Sawfish Critical Habitat Rule (NMFS 2009), maintained channels or marinas are not included in the designated critical habitat. Because the maintenance dredging to be covered under this Opinion only includes maintenance of previously dredged channels, we do not anticipate any effects to smalltooth sawfish critical habitat.

We believe minor dredging of 1.81 ac of waterbottom area in smalltooth sawfish critical habitat (Table 34) will have no effect on the shallow, euryhaline essential feature of smalltooth sawfish critical habitat. PDC A3.5.2 limits minor dredging to areas already deeper than -3 ft MLLW or areas shallower than -3 ft MLLW that are dredged to maintain the 0-3 ft shallow, euryhaline essential feature. For example, an area that is currently- 4 ft deep can be deepened to -6 ft deep or an area that is currently -1 ft deep can be deepened to -3 ft MLLW, without removing the shallow, euryhaline essential feature. Therefore, minor dredging will not remove the shallow, euryhaline essential feature, and changes in depth between the MHWL and -3 ft MLLW will not affect this essential feature.

Muck dredging is not allowed in shallow, euryhaline habitat (PDC A3.5.3) and PDC AP.12 does not allow removal of red mangroves in association with any dredging activities, including muck dredging. Therefore, the estimated 187.5 ac of muck dredging that could occur in smalltooth sawfish critical habitat will not affect the essential features of smalltooth sawfish critical habitat.

### Potential Routes of Effect to Gulf sturgeon Critical Habitat

We believe that maintenance and minor dredging activities may affect, but are not likely to adversely affect Gulf sturgeon critical habitat. PDC A3.6 prohibits muck dredging in Gulf sturgeon critical habitat; therefore, these types of dredging will have no effect on critical habitat. USACE anticipates that approximately 51 maintenance and minor dredging activities may be covered under this Opinion per 5-year period in Gulf sturgeon critical habitat. Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

<u>Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages</u>. Dredging may remove substrates containing sturgeon prey items. However, we believe the effects to this PCE will be insignificant since the estimated 189.46 ac (Table 34) of impact from the combination of maintenance (188.7 ac) and minor (0.76 ac) dredging is small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we estimate support sturgeon prey species</u>. Dredged material removal will temporarily affect the prey abundance (e.g., crustaceans on the benthic surface and infaunal polychaetes within the dredging footprint). As discussed above, these effects are primarily short-term in nature, consisting of a temporary loss of benthic

invertebrate populations in the dredged areas. Observed rates of benthic community recovery after dredging range from 3-24 months (Culter and Mahadevan 1982; Saloman et al. 1982; Wilber et al. 2007). The relatively species-poor benthic assemblages associated with low salinity estuarine sediments can recover in periods of time ranging from a few months to approximately 1 year, while the more diverse communities of high salinity estuarine sediments may require a year or longer. Therefore, we believe the effects to this PCE will be insignificant.

- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from dredging; however, the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement. An instance where turbidity curtains may be waived is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore effects from turbidity would be minimal, if any. Also, by limiting the dredging depths to no greater than the controlling depth of the adjacent channel for minor dredging and to the previously authorized depth for maintenance dredging, we believe these projects will not affect the hydrology and overall water quality in the area. Therefore, we believe the effects to this PCE will be insignificant.
- 3. <u>Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages.</u> Dredging can affect sediment quality; however, the effects to this PCE will be insignificant. The materials that will be dredged from a project area are likely to be the same as those remaining in the dredge footprint; therefore, no permanent alteration of habitat composition occurs within this area. Because similar habitat is expected to be present pre- and post-dredging, it is anticipated that the benthic biota in the dredging areas will have the ability to recover and re-colonize. Therefore, we believe the effects to this PCE will be insignificant.
- 4. <u>Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats</u>. Dredging activities could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by dredging activities in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging; however, we believe there will be no effect to this PCE. PDC A3.6 prohibits dredging anywhere in Gulf sturgeon critical habitat between September and March, when sturgeon are likely to be present in the migratory pathways.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and there will be no effect to safe and unobstructed migratory pathways, we believe that maintenance and minor dredging activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe minor dredging may affect the essential features of Johnson's seagrass critical habitat. USACE anticipates that 31 dredging activities may be covered under this Opinion per 5-year period in Johnson's seagrass critical habitat. We estimate that 39.75 ac of waterbottom may be dredged in Johnson's seagrass critical habitat.

We believe maintenance dredging 0.63 ac per 5-year period (Table 34) will have no effect to the essential features of Johnson's seagrass critical habitat. Maintenance dredging in areas that have been previously disturbed lack of the stable, unconsolidated sediment that is free from physical disturbance essential feature. Since Johnson's seagrass critical habitat must support all of the essential features to be considered functional critical habitat, these areas are not considered critical habitat. Therefore, maintenance dredging of these areas is expected to have no effect on Johnson's seagrass critical habitat.

We believe minor dredging of 0.37 ac is likely to adversely affect Johnson's seagrass critical habitat by permanently removing the essential features. The loss of 0.37 ac of Johnson's seagrass critical habitat from minor dredging (Table 34) is evaluated in Section 5.

We believe muck dredging of 38.75 ac will have no effect to the essential features of critical habitat. PDC A3.8 prohibits muck dredging where the essential features are present. Areas to be muck dredged within the geographic boundary of Johnson's seagrass have accumulated organic material that is so thick that it no long supports seagrass growth. Thus, it lacks the essential features of stable, unconsolidated sediments that are free from physical disturbance and/or adequate water quality. Removal of the accumulated material back to the original depth and sediments is intended to improve habitat by returning these areas back to their natural state that may then ultimately support seagrasses in the future, and thus could restore the feature.

## 2.2.4 Activity 4 (A4): Water-Management Outfall Structures and Associated Endwalls

#### **General Description**

This category of activity includes the installation, repair (including all forms of maintenance), replacement, and removal of water-management outfall structures. These structures are typically placed by trenching or excavating a conduit or discharge pipe from the stormwater system or mosquito ditch to an open water body for discharge or by installing a culvert to connect 2 water bodies to improve water circulation and water quality or restore hydrology. All work is typically completed using mechanical equipment from the uplands. Some discharge pipes or culverts are fitted with manatee grates or blocked by piles or bars (Figure 16), which are spaced a maximum of 8 in apart, depending on the size of the discharge pipe or culvert, to ensure that manatees do not enter these pipes or culverts and become trapped. All water discharged from outfall structures must meet all of the state and federal water current quality standards to protect the waterbody to which it is discharged. In addition, all outfall structures for stormwater-
management systems, including replacements, in *Acropora* critical habitat and Johnson's seagrass critical habitat must meet current state and federal water quality standards and contain an in-line treatment structure to reduce water velocities, sedimentation, nutrients, and pollutants discharged from the outfall structure into marine waters. Methods for meeting these standards may include nutrient baffle structures, filters, natural bio filters, and low impact development such as infiltration basins and trenches or vegetative swales. These additional water quality requirements for outfall structures in *Acropora* or Johnson's seagrass critical habitat do not apply to installation of manatee grates on existing culverts or maintenance of the head wall or other shoreline stabilization activities associated with the outfall.

Scour control measures are often used to prevent localized scour and erosion at discharge structures. These measures may include geotextile mats, riprap, or other materials to stabilize the immediate discharge area. The use of these materials is covered under the shoreline stabilization category of activity (Activity 1).



FRONT VIEW

SECTION

Figure 16. Sample drawing of an outfall pipe with a manatee grate. Manatee grates are typically attached to the pipe, but in this case, piles are placed in front of the pipe.

# Project Design Criteria

# PDCs specific to Activity for Water-Management Outfall Structures and Associated Endwalls:

- A4.1. Activities covered by this Opinion include:
  - 4.1.1 Installation, repair, replacement, extension, and removal of existing metal or concrete pipes, culverts, or other drainage conveyance structures that discharge storm water, surface water, or connect existing water bodies.
  - 4.1.2 Installation of new outfall structures to connect 2 existing water bodies to improve water flow and quality or restore hydrology.
  - 4.1.3 Installation of metal manatee grates. Grates are installed for manatee protection on all culverts that are between 8-in and 8-ft of diameter.

# The following PDCs apply to all the activities described in PDC A4.1 above:

- A4.2. This Opinion only covers water-management outfall structures when the effluent from the outfall is authorized, conditionally authorized, specifically exempted, or in compliance with the National Pollutant Discharge Elimination System Program (CWA section 402 or state water quality permit and any implementing regulations). The construction of intake structures is not covered unless it is directly associated with a USACE authorized outfall structure.
- A4.3. All outfall discharge shall be designed and implemented to prevent erosion and scour.

# Additional PDCs for Activity 4 applicable in critical habitat:

# In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat as described below.

- A4.4. *Acropora* critical habitat, Johnson's seagrass critical habitat, and throughout the U.S. Caribbean (whether in critical habitat units or not): All outfall structures for stormwater-management systems, including replacements and repairs, in these areas must be designed as follows:
  - A4.4.1. Structures that result in water discharge into nearshore waters must not be part of a combined-sewer system (sanitary and storm sewers that are connected).
  - A4.4.2. Meet current state and federal water quality standards.
  - A4.4.3. In addition to any requirements contained in state and federal water quality standards, outfall structures shall be designed to include a treatment structure that reduces water velocities, sedimentation, nutrients, and pollutants discharged from the outfall structure into marine waters to protect surrounding seagrasses or corals. These methods may include nutrient baffle structures, control structures with sediment forebays, filters, natural bio filters, velocity baffles at outfall, and low impact development such as infiltration basins, rain gardens, and trenches or vegetative swales. These requirements do not apply to installation of manatee grates on existing culverts or maintenance of the head wall or other shoreline stabilization activities associated with the outfall.

- A4.5. Smalltooth sawfish critical habitat: This Opinion does not cover activities in areas identified as smalltooth sawfish limited exclusion zones, as defined in Section 2.1.1.1.
- A4.6. Gulf sturgeon critical habitat: Additional noise restrictions are required for pile and sheet pile installation in the Gulf sturgeon critical habitat migratory restriction zones defined in Section 2.1.1.2.
- A4.7. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): This Opinion does not apply to water management outfall structures proposed in U.S. Caribbean sea turtle critical habitat.

## Assumptions

The USACE estimates that 100 ft<sup>2</sup> of habitat may be affected by each water-management outfall project. According to information provided in the *Monroe County Canal Management Plan* (AMEC Environment & Infrastructure 2013), culverts are anticipated to be used between manmade canals in the Florida Keys to improve water quality or restore hydrology. Due to the prevalence of mangroves throughout Florida, the placement of some of the culverts could result in the loss of mangroves along the shoreline even after designing the location to avoid and minimize the loss of mangroves. Therefore, the PDCs limit the removal of mangroves for when installing culverts to improve water quality or restore hydrology between 2 waterbodies to no more than 20 lin ft of mangroves along the shoreline per culvert opening (PDC AP.12). This is the amount that the *Monroe County Canal Management Plan* estimated may be required to be removed to install these culverts. The USACE estimates that no more than 10% of projects will result in the removal of mangroves.

Thus, to analyze the effects of the water-management outfall activities, we looked at the USACE's assumptions regarding (1) the total number of water-management outfall activities to be authorized per 5-year period that meet the requirements of the Opinion; and (2) if the installation of the structure would result in the removal of mangroves. Based on this information, we determined the potential effects to species (Table 35) and critical habitat (Table 37).

## Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing water-management outfall projects would result in 5 of the common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the installation, maintenance, repair, replacement, and removal of watermanagement outfall structures and associated endwalls that were not considered in Section 2.2. NMFS and USACE's effects determinations are summarized for this category of activity in Table 35. Sea turtles, smalltooth sawfish, and sturgeon may be affected by the permanent loss of habitat from the placement of materials along the shoreline. We quantified the potential extent of habitat impacts based on (1) the assumptions, (2) the number and location of projects estimated by USACE, and (3) the construction limitations defined by the PDCs. Estimated amount of waterbottom and shoreline area affected by water-management outfalls structures was calculated in Table 36 by multiplying the estimated "total number of projects" by the "average area of loss per project" to get the total amount of waterbottom and shoreline area affected by the placement of structures. Table 36 also provides the estimated total mangrove loss expected by multiplying the USACE's estimate of a 20 lin ft of mangrove loss per project involving the placement of culverts necessary to improve water quality or restore hydrology between 2 water bodies (also restricted to 20 lin ft by PDC AP.12) by the 10% of the 129 estimated project per year expected to have mangrove impacts.

Table 35. USACE and NMFS Determinations on the Effects of Water-Management OutfallStructures and Associated Endwall Activities (Activity 4) to ESA-listed Species Listed inTable 5

Listed Species	<b>USACE Determination</b>	NMFS Determination				
Sea Turtles	NLAA	NLAA				
Sawfish	NLAA	NLAA				
Sturgeon	NLAA NLAA					
Effects I	Effects Determinations Explained in Section 2.2					
Johnson's seagrass	NE	NE				
Corals	NE	N/A				
Whales	NE	NE				
Nassau grouper	Not provided	NLAA				
NE= no effect; NLAA= may aff	ect, not likely to adversely af	fect; N/A= not applicable (if a				

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

 Table 36. Estimated Amount of Waterbottom and Shoreline Areas and Mangrove Shoreline

 Affected by Water-Management Outfall Structures and Associated Endwall Projects per 5 

 Year Period.

Total Number of Projects	Average Area Affected Per Project	Total Area Affected
Waterbottom and shoreline area		
129	100 ft <sup>2</sup>	12,870 ft <sup>2</sup> (0.3 ac)
Mangrove Loss	1	
12.9 (10 % of the 129 projects)	20 lin ft	258 lin ft

#### Potential Routes of Effect to Sea Turtles, Smalltooth Sawfish, and Sturgeon

We calculated the combined estimated amount of both (1) waterbottom and shoreline areas and (2) mangrove shoreline area affected per project from all water-management outfall structures and endwall projects per 5-year period. We do not know how many projects will be new construction vs repair, replacement, and removal of water-management outfall structures and associated endwall projects. New projects will affect more waterbottom than repair or replacement projects, which will likely be in the same footprint and will not affect additional resources. Given the limited data available, to ensure that we have evaluated the potential worst-case scenario for impacts to these areas, we will assume that all of the projects are new and that each project will result in the average loss of 100 ft<sup>2</sup> of waterbottom and shoreline area (see assumptions above). Impacts to mangroves from the placement of culverts between 2 waterbodies are limited to no more than 20 lin ft of removal per project (PDC AP.12). In addition, the USACE believes that only 10% of projects will require the removal of mangroves since projects will first be sited to minimize and avoid mangroves (PDC AP.12).

Combined over a 5-year period, the placement of water-management outfall structures and endwall projects are estimated to cover and remove 12,870 ft<sup>2</sup> (0.3 ac) of waterbottom and shoreline areas and an estimated 258 lin ft of mangrove shoreline (Table 36). We acknowledge that that the estimated loss from water-management outfall structures is likely an overestimate since we assumed all projects are new construction (i.e., not repair, replace, or removal, which as noted above, we do not expect to affect additional resources). This loss is unlikely to be concentrated as the array of individual projects covered under Activity 4 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean).

- Sea Turtles: Sea turtles may be affected by the placement of structures on top of 0.3 ac of waterbottom and shoreline areas. However, the effect on sea turtles of the potential loss of nearshore foraging habitat is insignificant.
  - The area affected by Activity 4 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes all projects from the Opinion where Johnson's seagrass is present and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
  - Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the project footprint.

- Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect listed corals, and excludes projects if non-listed corals and hardbottom habitat, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
- Outfall structure activities may cover or remove nearshore areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition, the 0.3 ac of impact is very small compared to the remaining large nearshore areas that support sea turtle prey species, and sea turtles can travel long distances to forage. The projects covered under Activity 4 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites.

Given the above, the effect to sea turtles from the potential loss of foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the outfall structure and associated endwall activities under Activity 4 are not likely to adversely affect sea turtles.

Smalltooth sawfish: Smalltooth sawfish may be affected by the placement of structures on • top of 0.3 ac of waterbottom and shoreline areas and the loss of 202 lin ft of mangrove shoreline, both of which could be used for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. However, the effect on sawfish of the potential loss of nearshore foraging and refuge habitat is insignificant. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for refuge; these areas have been designated as critical habitat for the species. The PDCs limits this Opinion to outfall projects occurring outside of smalltooth sawfish limited exclusion zones (PDC A4.5), which are areas that research shows support higher levels of smalltooth sawfish pupping. Projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal associated with placing a culvert between 2 waterbodies to improve water quality or restore hydrology is limited to 20 lin ft per project by PDC AP.12 for a total estimated loss of 258 lin ft. This is a small area of mangrove shoreline reduction when compared to all the mangrove shoreline within the action area and likely undetectable to smalltooth sawfish. As is shown in Figure 16, installing the outfall structure and associated endwalls may also cover or bury nearshore areas by the placement of structures along the shoreline in areas potentially inhabited by fish and crustaceans that serve as prey for smalltooth sawfish. The area of impact (some amount less than 0.3 ac) is very small compared to the remaining large nearshore areas that support sawfish prey species. Sawfish can travel long distances to forage. The projects covered under Activity 4 will likely be separated both temporally (over a 5-year period) and spatially

(along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites. Thus, we believe the effect on sawfish of the potential loss of nearshore foraging and refuge habitat is insignificant. Based on the routes of effect analyzed here and in Section 2.2, we determined that the water-management outfall structures and associated endwall activities under Activity 4 are not likely to adversely affect smalltooth sawfish.

Sturgeon: Sturgeon may be affected by the placement of structures on top of 0.3 ac waterbottom and shoreline areas, which could be used for foraging. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. However, the effect on sturgeon of the potential loss of these nearshore foraging habitat is insignificant. As is shown in Figure 16, installing the outfall structure and associated endwall may cover and bury nearshore bottom substrates from the placement of these structures along the shoreline. The covered areas may contain sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, sturgeon are opportunistic feeders that forage over large areas. Gulf sturgeon select foraging habitat based on substrate composition and depth, rather than prev density, abundance, or diversity. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Hence, Gulf sturgeon, and likely shortnose and Atlantic sturgeon, often occupy waters deeper than those typically affected by water-management outfall structures and endwall projects occurring in the vicinity of the MHWL. The area of impact (some amount less than 0.3 ac) is very small compared to the remaining large areas that support sturgeon prey species. The projects covered under Activity 4 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that the water-management outfall structure and endwall projects under Activity 4 are not likely to adversely affect sturgeon.

### Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from water - management outfall structures and associated endwall projects covered under this Opinion. The NMFS and USACE effects determinations for this category of activity are summarized in Table 37.

Table 37. US	ACE and NMFS D	eterminations on the	Effects of Wate	er-Management and
Endwall Acti	vities (Activity 4) or	n Designated Critica	l Habitat	

Designated Critical Habitat	USACE Determination	NMFS Determination			
Smalltooth sawfish critical habitat	LAA	LAA			
Gulf sturgeon critical habitat	NLAA	NLAA			
Johnson's seagrass critical habitat	LAA	LAA			
Effects Determinations	Explained in Section 2.2				
Loggerhead critical habitat	NE	NE			
Acropora critical habitat	NE	NE			
North Atlantic right whale critical habitat	NE	NE			
Atlantic sturgeon critical habitat	Not provided	NE			
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.					

The estimated amount of waterbottom/ shoreline areas and mangrove shoreline affected by these structures is calculated in Table 38 by multiplying the estimated number of projects expected per 5-year period (total number of projects from Table 8) by the estimated 100 ft<sup>2</sup> of area waterbottom and of shoreline affected to determine the "total amount of waterbottom within critical habitat affected." Below, we discuss whether and how affecting this area may affect the essential features of critical habitat. In addition, in the section addressing effects to smalltooth sawfish critical habitat, we estimate red mangrove loss.

Table 38. Estimated Amount of Waterbottom and Shoreline Areas Affected by Water-Management Outfall Structures and Associated Endwall Projects in Smalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat

Project Location	Number of Projects (Table 8)	Total Waterbottom Area Within Critical Habitat Affected (ft²)	Total Waterbottom Area Within Critical Habitat Affected (ac)
Smalltooth sawfish critical habitat	19	1,900	0.04
Gulf sturgeon critical habitat	24	2,400	0.06
Johnson's seagrass critical habitat	7	700	0.02

## Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that water-management outfall activities may affect smalltooth sawfish critical habitat. USACE anticipates that it may authorize 19 water-management outfall activities that meet the requirements of this Opinion per 5-year period in smalltooth sawfish critical habitat. The USACE estimates that 10% of the all water-management outfall and associated endwall

projects (see assumptions above) will result in the removal of mangrove shoreline. Since we do not know if some or all of the mangrove removal for the placement of culverts necessary to improve water quality or restore hydrology between 2 water bodies will occur in our outside of smalltooth sawfish critical habitat, we calculated a minimum and maximum removal of this essential feature. If 10% of the 19 estimated projects in smalltooth sawfish critical habitat resulted in the 20 lin ft of mangrove removal, this would result in a total estimated loss of up to 38 lin ft (20 lin ft x 1.9 projects). This would be the minimum estimated removal of the red mangrove essential feature. However, if all of the projects involving mangrove removal occurred in smalltooth sawfish critical habitat (10% of all 129 projects, shown in Table 36 above), that would result in the removal of 258 lin ft of mangrove shoreline (20 lin ft x 12.9 projects = 258 lin ft). This would represent a maximum estimated removal and is likely an overestimate since it is unlikely all the mangrove removal projects would occur in smalltooth sawfish critical habitat. However, since red mangrove removal is likely to have the greatest effect in smalltooth sawfish critical habitat, we will analyze the effects of this activity if all mangrove removal occurred inside smalltooth sawfish critical habitat and affected red mangroves. We acknowledge this is likely an overestimate, but will err on the side of the species when evaluating the effects to this essential feature. This loss of the essential feature is likely to adversely affect critical habitat and is evaluated further in Section 5. In Table 38, we estimated that the placement of water-management outfall structures in

smalltooth sawfish critical habitat may affect up to 1,900 ft<sup>2</sup> (0.04 ac) of waterbottom and shoreline areas. We assume these affected areas are shallow, euryhaline habitat, which is likely to adversely affect the essential feature. Therefore, the potential loss of red mangroves and shallow, euryhaline habitat is likely to adversely affect critical habitat and evaluated in Section 5.

## Potential Routes of Effect to Gulf sturgeon Critical Habitat

We believe that water-management outfall activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat, as described below. USACE anticipates that 24 watermanagement outfall activities may be authorized per 5-year period in Gulf sturgeon critical habitat that rely on this Opinion, resulting in the placement of structures on top of 2,400 ft<sup>2</sup> (0.06 ac) of waterbottom and shoreline areas in Gulf sturgeon critical habitat (Table 38). Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

<u>Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages</u>. Installing water-management outfall and associated endwall structures, including the manatee grates, may cover and bury nearshore bottom substrates containing sturgeon prey species. However, we believe the effects to this PCE will be insignificant since the placement of structures on an estimated 0.06 ac of waterbottom and shoreline areas is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we estimate supports sturgeon prey species. Further, not all of the 0.06 ac of habitat lost is expected to support prey items or serve as preferred foraging habitat. Gulf sturgeon are suction feeders that tend to forage in calm marine and estuarine waters that support their macroinvertebrate prey including brachiopods,

mollusks, worms, and crustaceans (Mason and Clugston 1993). During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5 and 13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Since water-management and associated endwall projects occur along the shoreline, installation or repair of these structures may not occur in areas with the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from the installation, maintenance, repair, or removal of water-management outfall structures and associated endwall projects; however, the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement as described in PDC AP.10. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may be waived is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high wave energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high energy areas where turbidity will dissipate very quickly), we believe effects from turbidity on water quality would be insignificant.
- 3. Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages. The placement of water-management outfall structures and associated endwall projects, can affect sediment quality. The placement of these materials converts sandy substrate, capable of supporting Gulf sturgeon prey, to hard man-made materials that do not support prey species. However, we believe the effects to this PCE will be insignificant since the placement of structures on top of an estimated 0.06 ac waterbottom and shoreline areas is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we estimate supports sturgeon prey species and Gulf sturgeon foraging. Further, not all of the 0.06 ac of habitat lost will have the sediment quality needed to support Gulf sturgeon prey shoreline areas between 6.5 and13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Water-management structures and associated endwall projects are expected to occur along the shoreline in hence may not occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats. Water-management outfall structures and associated endwall activities could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging. However, we believe there would be no effect to this PCE. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, PDC A4.6 requires compliance with the noise restrictions for pile and sheet pile installation in Gulf sturgeon critical habitat and the Gulf sturgeon critical habitat migratory restriction zones. Outfall structure and associated endwall projects occurring in areas outside of Gulf sturgeon critical habitat migratory restriction zones. Will occur immediately contiguous with and parallel to shorelines and will not restrict the movement of sturgeon.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and there will be no effect to safe and unobstructed migratory pathways, we believe that water-management outfall structure and associated endwall projects activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

#### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe that water-management outfall activities may affect Johnson's seagrass critical habitat. USACE anticipates that it may authorize 7 water-management outfall activities that meet the requirements of this Opinion per 5-year period in Johnson's seagrass critical habitat (Table 38). We assume the affected areas contain the following essential features of Johnson's seagrass critical habitat, water quality, and stable, unconsolidated sediments that are free from physical disturbance, and assess the potential impacts to those features.

- Water quality. Water discharged from these water-management outfall structures may affect water quality. The PDCs state that all outfall structures for stormwater-management systems, including replacements, in Johnson's seagrass critical habitat must meet current state and federal water quality standards (PDC A4.2) and must also contain an in-line treatment structure to reduce water velocities, sedimentation, nutrients, and pollutants discharged from the outfall structure into marine waters (PDC A4.4). These methods may include nutrient baffle structures, filters, natural bio filters, and low impact development such as infiltration basins and trenches or vegetative swales. The additional protective measures are required in Johnson's seagrass critical habitat to reduce water quality impacts to seagrasses. Therefore, we believe this activity will result in insignificant effects to the Johnson's seagrass water quality essential feature.
- 2. Stable, unconsolidated sediments that are free from physical disturbance. Placement of these structures will permanently remove areas of the stable, unconsolidated sediment essential feature. The USACE estimates the placement of water-management outfall structure on top of 100 ft<sup>2</sup> or less of waterbottom and shoreline areas in Johnson's seagrass critical habitat per project (see assumptions above) and we assume that the USACE may authorize 7 of these activities per 5-year period, resulting the potential loss of a total of 700 ft<sup>2</sup> (0.02 ac) of

Johnson's seagrass critical habitat (Table 38). This loss is likely to adversely affect Johnson's seagrass critical habitat and is evaluated in Section 5.

# 2.2.5 Activity 5 (A5): Scientific Survey Devices

## **General Description**

This category of activity includes the installation, repair (including all forms of maintenance), replacement, and removal of scientific survey devices include scientific measuring devices such as staff gages, tide and current gages, meteorological stations, water recording and biological observation devices, water quality testing and improvement devices, and vibracore devices. Since these devices are used in scientific research, those using the devices may rely on new survey methods or new scientific experiments when deploying and using the devices that are not discussed here. It is likely that any new devices or methods of using them will have similar effects to those considered in this Opinion. Hence, the Opinion may cover other types of scientific survey devices if they are approved according to the procedures outlined in the Project Review requirements in Section 2.3. According to USACE, scientific survey devices are typically removed in less than 24 months. Many survey devices are installed with anchored buoys, vinyl poles, or single piles installed by hand or jetted in place from a barge. This type of installation can typically be completed in 1-2 days.

## Project Design Criteria

# PDCs specific to Activity 5 for Scientific Survey Devices:

- A5.1. This Opinion covers the installation, repair, and removal of scientific survey devices, including any related equipment and anchors, for up to 24 months if those devices are intended to measure and/or record scientific data in tidal waters, such as staff gages, weirs, tide and current gages, meteorological stations, water recording and biological observation devices, water quality testing and improvement devices, vibracore samplings, and similar structures.
- A5.2. The scientific survey device, including any related equipment and anchors, shall not block access of species to an area. For example, the structures shall not prevent movement in or out of a river or channel.
- A5.3. No later than 24 months after initial installation or upon completion of data acquisition, whichever comes first, the measuring device and any other structure or fills associated with that device (e.g., anchors, buoys, lines) must be removed and the site must be restored to pre-construction conditions.
- A5.4. The scientific survey device, including any related equipment and anchors, shall be inspected and any required maintenance performed at least twice a year and following storm events that may have moved or dislodged the structure to ensure that equipment and anchors are still in place and have not moved to areas containing ESA-listed corals.

## Assumptions

The USACE believes that activities in this category will be temporarily placed on 1 ft<sup>2</sup> to 50 ft<sup>2</sup> of waterbottom area, with an average of 20 ft<sup>2</sup> per project. The affected waterbottom includes

those areas upon which materials, anchors, lines, or any other equipment used for scientific surveys is placed. USACE anticipates that 89 temporary scientific survey activities will be covered under this Opinion per 5-year period (Table 8).

Thus, to analyze the effects of scientific survey projects, we looked at the USACE's assumptions regarding (1) the total number of scientific activities to be authorized per 5-year period; and the estimated size of the impacts from this activity. Based on this information, we determined the potential effects to species (Table 39) and critical habitat (Table 41).

#### Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing scientific survey device projects would result in the following common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the impacts associated with the temporary placement of scientific survey devices that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 39. Sea turtles, smalltooth sawfish, and sturgeon may be affected by the temporary placement of structures on top of waterbottom areas, which may serve as habitat, during the up to 24 months (PDC A5.3) that scientific survey devices are in place. We quantified the potential extent of impacts based on (1) the assumptions, (2) the number and location of projects estimated by USACE (Table 8), and (3) the construction limitations defined by the PDCs. Scientific survey devices can account for a large array of activities that scientists develop to test and track changes in the marine environment. Since we do not know the exact items that will be used, the PDCs are designed to limit effects to species and habitat by limiting the duration the devices may be used and by requiring the devices to be deployed in a manner that does not block species access to resources. The USACE assumes the materials placed for these activities will be small in size ranging from 1 ft<sup>2</sup> to 50 ft<sup>2</sup>, with an average impact of 20 ft<sup>2</sup> per project. In Table 40, we determined the minimum and maximum impacts possible from this temporary activity by multiplying the estimated number of projects by the minimum (1 ft<sup>2</sup>), maximum (50 ft<sup>2</sup>) and average (20 ft<sup>2</sup>) amount of expected impacts.

Listed Species	<b>USACE Determination</b>	NMFS Determination	
Sea Turtles	NLAA	NLAA	
Sawfish	NLAA	NLAA	
Sturgeon	NLAA	NLAA	
Effects I	Determinations Explained in S	Section 2.2	
Johnson's seagrass	NE	NE	
Corals	NE	N/A	
Whales	NE	NE	
Nassau Grouper	Not provided	NLAA	

 Table 39. USACE and NMFS Determinations on the Effects of Scientific Survey Device

 Activities (Activity 5) to ESA-listed Species Listed in Table 5

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau grouper was provided at the beginning of Section 2.2.

 Table 40. Estimated Number of Scientific Survey Activities and the Associated Temporary

 Habitat Impacts over a 5-Year Period.

Number of Projects	Minimum estimated impact of 1 ft <sup>2</sup> per project (ft <sup>2</sup> )	Maximum estimated impact of 50 ft <sup>2</sup> per project (ft <sup>2</sup> )	Average estimated impact of 20 ft <sup>2</sup> per project (ft <sup>2</sup> )	Average estimated impact (ac)
89	89	4,450	1,780	0.04

Over a 5-year period, sea turtles, sawfish, and sturgeon could be affected by the temporary placement of scientific survey devices on top of between 89 and 4,450 ft<sup>2</sup> (estimated average temporary loss of 1,780 ft<sup>2</sup> [0.04 ac])of waterbottom area, which may serve as foraging or refuge habitat. This estimate includes projects in both Florida and the U.S. Caribbean. This temporary placement of structures on waterbottom areas is unlikely to be concentrated as the array of individual projects covered under Activity 5 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean). Due to the limited time these devices will be in use (24 months as required by PDC A5.3) and the limited size of impact (based on the assumptions), scientific survey devices may be used in critical habitat units. In addition, certain types of scientific survey devices may be attached to a pile or anchor system. Those projects must follow the noise requirements for pile installation contained in the PDCs for *In-Water Noise from Pile and Sheet Pile Installation* (PDC AP.2), as applicable, as well as the requirements for anchor lines designed to prevent entanglement contained in the PDCs for *In-Water Activities*, specifically PDC AP.11.

• Sea turtles: Sea turtles may be affected by the temporary placement of structures on between 89 and 4,450 ft<sup>2</sup> of waterbottom area, which could be used as foraging habitat. However, the

effect on sea turtles of the potential loss of foraging habitat is insignificant since each of these scientific survey devices will only temporarily occupy a small space each (1-50 ft<sup>2</sup>). In addition,

- The waterbottom affected by Activity 5 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes projects scientific survey device projects where Johnson's seagrass is present within the project footprint and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
- Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the project footprint.
- Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect ESA-listed coral and excludes projects if nonlisted corals and hardbottom habitat, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
- Scientific survey device activities may temporarily cover areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition, the 89 to 4,450 ft<sup>2</sup>of impact is very small compared to the remaining large areas that support sea turtle prey species. Sea turtles can travel long distances to forage. The array of individual projects covered under Activity 4 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites.

Given the above, effect to sea turtles from the potential loss of foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that scientific survey device activities under Activity 5 are not likely to adversely affect sea turtles.

• Smalltooth sawfish: Smalltooth sawfish may be affected by the temporary placement of structures on top of between 89 and 4,450 ft<sup>2</sup> of waterbottom area, which may be used for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. However, the effect on sawfish of the potential loss of foraging and refuge habitat is insignificant. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat for the species. Projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal is strictly limited by PDC AP.12 to removal related to other covered activities and to removal above the MHWL provided that red mangrove prop roots that are accessible to marine

species are not removed. Therefore, the effect on juvenile sawfish of losses of small areas of shallow water refuge habitat is expected to be so small as to be undetectable. Scientific survey devices may temporarily cover small areas (1-50 ft<sup>2</sup> per project) of habitat that could be used as foraging habitat by smalltooth sawfish. Cumulatively, the area of impact (some amount less than 89-4,450 ft<sup>2</sup>) is very small compared to the remaining large nearshore areas that support sawfish prey species. In addition, the scientific survey devices may be placed outside of nearshore waters, further offshore, so the area affected may not always be the nearshore areas where the species seeks forage and refuge. In addition, sawfish can travel long distances to forage. The array of individual projects covered under Activity 5 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that the scientific survey device sawfish.

Sturgeon: Sturgeon may be affected by the temporary placement of structures on top of between 89 and 4,450 ft<sup>2</sup>ac of waterbottom area, which may be used for foraging. As is noted above, these estimates include projects in both Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. However, the effect on sturgeon of the potential loss of foraging habitat is insignificant. Scientific survey device projects may temporarily cover (24 months as per PDC A5.3) substrates containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, sturgeon are opportunistic feeders that forage over large areas. The area of impact is very small compared to the remaining large areas that support sturgeon prey species. The array of individual projects covered under Activity 5 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that the scientific survey device activities under Activity 5 are not likely to adversely affect sturgeon.

## Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from scientific survey device activities. The NMFS and USACE effects determinations for this category of activity are summarized in Table 41 and the estimated impacts to each critical habitat unit, based on the assumption that each project will affect an average of 20 ft<sup>2</sup> of waterbottom area (see assumptions above), are summarized in Table 42.

 Table 41. USACE and NMFS Determinations on the Effects of Scientific Survey Device

 Activities (Activity 5) to Designated Critical Habitat

Designated Critical Habitat	USACE Determination	NMFS Determination				
Smalltooth sawfish critical habitat	LAA	LAA				
Gulf sturgeon critical habitat	NLAA	NLAA				
Johnson's seagrass critical habitat	NLAA	LAA				
Effects Determinations	Effects Determinations Explained in Section 2.2					
Loggerhead critical habitat	NE	NE				
Acropora critical habitat	NE	NE				
North Atlantic right whale critical habitat	NE	NE				
Atlantic sturgeon critical habitatNot providedNE						
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.						

 Table 42. Estimated Number of Scientific Survey Device Projects and Associated

 Temporary Impacts in Critical Habitat with Effects Determinations

Project Location	Number of Projects (Table 8)	Minimum estimated impact of 1 ft <sup>2</sup> per project (ft <sup>2</sup> )	Maximum estimated impact of 50 ft <sup>2</sup> per project (ft <sup>2</sup> )	Average estimated impact of 20 ft <sup>2</sup> per project (ft <sup>2</sup> )	Average estimated impact (ac)
Sawfish critical habitat	2	2	100	40	0.00 <sup>25</sup>
Gulf sturgeon critical habitat	2	2	100	40	$0.00^{26}$
Johnson's seagrass critical habitat	17	17	850	340	0.01

 $<sup>^{25}</sup>$  40  $\mathrm{ft}^2$  is approximately 0.0009 ac, which rounded to 2 decimal places is 0.00 ac.

 $<sup>^{26}</sup>$  40  $\mathrm{ft}^2$  is approximately 0.0009 ac, which rounded to 2 decimal places is 0.00 ac.

### Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe the temporary placement of scientific survey devices will have no effect on the essential features of smalltooth sawfish critical habitat. USACE anticipates that 2 scientific survey activities meeting the requirements of this Opinion may be authorized per 5-year period in smalltooth sawfish critical habitat. The PDCs for *Mangroves, Seagrasses, Corals and Hardbottom for All Projects* (PDC AP.12) preclude the removal of mangroves for this activity and otherwise limit mangrove removal to removal that will not affect marine species (i.e., removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed). In addition, PDC A5.3 requires that when the survey devices are removed, the site is returned to pre-construction conditions. These 2 projects also may or may not even occur in areas that support the essential features of smalltooth sawfish critical habitat. Even if they did occur in an area with essential features, we believe that neither of the essential features will be affected by the temporary placement of 2 scientific survey devices with an estimated potential impact of 2-100 ft<sup>2</sup> (average 40 ft<sup>2</sup>), given that the devices must be installed in a way that does not result in the removal of red mangroves and that the installation will not change the depth of the shallow, euryhaline essential features.

### Potential Routes of Effect to Gulf Sturgeon Critical Habitat

We believe the effect of the temporary placement of scientific survey devices is not likely to adversely affect Gulf sturgeon critical habitat. USACE anticipates that 2 scientific survey activities meeting the requirements of this Opinion may be authorized per 5-year period in Gulf sturgeon critical habitat with the potential temporary impact of 2-100 ft<sup>2</sup> (average 40 ft<sup>2</sup>) of habitat (Table 42). Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

- <u>Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages</u>. The placement of 2 small scientific survey devices for up to 24 months (PDC A5.3) may affect 2-100 ft<sup>2</sup> (average 40 ft<sup>2</sup>) by temporarily covering bottom substrates containing sturgeon prey species; however, the effects to this PCE will be insignificant. The effect to this area is temporary, and the affected area is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments quality that we estimate supports sturgeon prey species and Gulf sturgeon foraging.
- 2. <u>Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages.</u> The placement of 2 of these projects is not expected to generate water quality issues from turbidity and therefore will have no effect on this essential feature.
- 3. <u>Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages.</u> The placement of 2 small scientific survey devices for up to 24 months (PDC A5.3) may affect 2-100 ft<sup>2</sup> (average 40 ft<sup>2</sup>) by temporarily

covering these areas, but will not change or affect sediment quality. Therefore, there will be no effect to this PCE.

4. <u>Safe and unobstructed migratory pathways necessary for passage within and between</u> <u>riverine, estuarine, and marine habitats</u>. The small size of these structures (1-50 ft<sup>2</sup> each) is not expected to impede the movement of species in the area or their use of habitat by Gulf sturgeon, therefore the migratory pathway PCE will not be affected. Also, PDC A5.2 requires that scientific survey devices be placed so as not to restrict species movement.

Because the effects to the PCEs of abundant prey items will be insignificant and there will be no effect to the water quality, sediment quality, and safe and unobstructed migratory pathways PCE, we believe that scientific survey activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

## Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe the temporary placement of scientific survey devices may affect Johnson's seagrass critical habitat. USACE anticipates that 5 scientific survey activities meeting the requirements of this Opinion may be authorized per 5-year period in Johnson's seagrass critical habitat. We believe that the temporary placement of between 17 and 850 ft<sup>2</sup> (average 340 ft<sup>2</sup>) of equipment in Johnson's seagrass critical habitat will not permanently alter Johnson's seagrass critical habitat essential features. The placement of the structure may temporarily affect water quality if turbidity is generated during placement, but this will settle quickly from a small structure. While the structure is in place, it may temporarily affect 850 ft<sup>2</sup> of the water transparency feature by blocking sunlight and covering the stable, unconsolidated essential feature. These effects will be discussed further in Section 5.

## 2.2.6 Activity 6 (A6): Boat Ramps

## **General Description**

This category of activity includes the installation, repair (including all forms of maintenance), replacement, and removal of boat ramps. Boat ramps are typically installed either in the uplands connecting to the water body or extending from the shore a short distance to provide the proper depth for vessels to safely enter the water. Most boat ramps require minor dredging/grading either to cut the upland location or to shape the slope of the ramp into the water. Following this site preparation, the pre-fabricated concrete slabs are typically placed, creating the ramp. Boat ramps are typically installed quickly (a day to a few days for smaller projects to a few weeks for larger ramps). Boat ramps can range from small private ramps for cances or boats at a single-family residence to public ramps supporting multiple entrances to the water. Many ramps also support a dock structure along the ramp to tie-up vessels during launching and to enter or exit the vessel. The construction of bulkheads and tie-up piers (i.e., docks) are analyzed separately under shoreline stabilization or construction of pile-supported structures. Below is a sample drawing of a boat ramp (Figure 17).



Figure 17. Example drawing of a boat ramp installed by placing prefabricated concrete slabs. Note that excavation for this example was minimal (removed approximately 10 cubic yards in this instance) and limited to shaping the slope of the boat ramp. This ramp extends to -3 ft MLW. Drawings prepared by Glen Boe and Associates, Inc. submitted to NMFS as part of a consultation request.

# Project Design Criteria

# PDCs Specific to Activity 6 for Boat Ramps:

A6.1. Activities covered by this Opinion include:

- A6.1.1. Removal or reconfiguration of existing boat ramps. The removal of upland-cut boat basins by walling off and filling them in is a type of shoreline stabilization covered under Activity 1, PDC A1.1, Section 2.2.1.
- A6.1.2. Installation of new boat ramps for motorized vessels (1) are limited in size to 40 ft wide, (2) can have up to 2 boat lanes, and (3) can be associated with no more than 50 trailered vehicle parking spaces.
- A6.1.3. New boat ramps for non-motorized vessels are limited in size to 60 ft wide.
- A6.1.4. Repair and replacement of existing boat ramps can occur within the same footprint of the existing ramp, even if the repaired or replaced boat ramp does not meet the size requirements in PDC A6.1.2 and A6.1.3. Reconfiguration of boat ramps for motorized vessels that propose to expand the footprint must meet the size limitations described in A6.1.2 (i.e., the reconfiguration must be no more than 40 ft wide and can have up to 2 boat lanes).

# The following PDCs apply to all the activities described in PDC A6.1 above:

- A6.2. Excavation is limited to the area necessary for site preparation. All excavated material shall be removed to an area that is not waters of the United States, as that term is defined and interpreted under the CWA, including wetlands.
- A6.3. All commercial or public boat ramps must have signs posted in a visible location(s) on the dock(s), alerting boaters of listed species in the area susceptible to vessel strikes or hook-and-line captures. These signs shall include contact information for the sea turtle and marine mammal stranding networks and smalltooth sawfish encounter database. Please visit our website (http://sero.nmfs.noaa.gov/protected\_resources/section\_7/protected\_species\_educational \_signs/index.html) to determine which signs are required for your area, for sign

installation guidance, and to download the most current version of the signs. In addition,

- A6.3.1. All projects shall use the Save Sea Turtle, Sawfish, and Dolphin sign.
- A6.3.2. Projects within the North Atlantic right whale educational sign zone (as defined in Section 2.1.1.4) shall post the *Help Protect North Atlantic Right Whales* sign.
- A6.3.3. Projects in the range of Gulf, Atlantic, or shortnose sturgeon also shall post the *Report Sturgeon* sign.
- A6.3.4. We are still developing the signs to be used in the U.S. Caribbean. Once developed, those signs will be included at the website above.
- A6.4. All commercial and public boat ramps also must install and maintain for the life of the facility monofilament recycling bins to reduce the risk of turtle or sawfish entanglement in or ingestion of marine debris. Monofilament recycling bins must:
  - A6.4.1. Be constructed and labeled according to the instructions provided at <u>http://mrrp.myfwc.com</u> or any equivalent guidance for the U.S. Caribbean.
  - A6.4.2. Be maintained for the life of the facility in working order and emptied frequently so that they do not overflow.
- A6.5. The length of new boat ramps and changes to existing boat ramps to make them longer should ensure a water depth at the end of the ramp is deep enough to minimize sediment resuspension associated with launching vessels in shallow water.
- A6.6. New boat ramps cannot be installed on ocean beaches that are used by nesting sea turtles.

# Additional PDCs for Activity 6 applicable in critical habitat:

# In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in critical habitat, as described below.

- A6.7. Smalltooth Sawfish critical habitat:
  - A6.7.1. New or expanded ramps located within smalltooth sawfish designated critical habitat cannot result in the loss of an essential feature of that critical habitat (red mangroves or shallow [MHWL to -3 ft MLLW], euryhaline water).
  - A6.7.2. Boat ramps can be constructed in waters between MHWL and -3 ft MLLW (shallow, euryhaline habitat essential feature), provided that the water depth is not increased to deeper than -3 ft MLLW.
  - A6.7.3. A boat ramp in smalltooth sawfish critical habitat may be repaired and replaced within the existing footprint.
  - A6.7.4. Boat ramp activities cannot occur in areas identified as smalltooth sawfish limited exclusion zone, defined in Section 2.1.1.1.
  - A6.7.5. Removal of upland cut boat ramps is allowed under this Opinion.
- A6.8. *Acropora* critical habitat: New or expanded boat ramps are not allowed in *Acropora* critical habitat where the essential features are present. Boat ramp repair and

replacement is allowed within Acropora critical habitat within the existing footprint.

- A6.9. Johnson's seagrass critical habitat: New or expanded boat ramps are not allowed in Johnson's seagrass critical habitat where the essential features are present. Boat ramp repair and replacement is allowed within the existing footprint.
- A6.10. Gulf sturgeon critical habitat: Additional noise restrictions are required for pile and sheet pile installation in the Gulf sturgeon critical habitat migratory restriction zones defined in Section 2.1.1.2.
- A6.11. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): This Opinion does not apply to new structures placed in sea turtle critical habitat in the U.S. Caribbean. Repair and replacement of shoreline protection within U.S. Caribbean sea turtle critical habitat is covered if it is within the existing footprint.

## Assumptions

USACE anticipates that 708 boat ramp activities meeting the requirements of this Opinion may be authorized per 5-year period. The USACE believes that the average boat ramp will cover or remove 2,592 ft<sup>2</sup> of waterbottom areas. The USACE estimates that 37% of the projects will be new construction and that 63% will be the repair, replacement, or removal of existing structures. Thus, to analyze the effects of boat ramp projects, we looked at the USACE's assumptions regarding (1) the total number of covered boat ramp activities to be authorized per 5-year period; (2) whether the projects would be new construction or repair/replacement; and (3) whether the projects would occur within or outside of critical habitat. Based on this information, we determined the potential effects to species (Table 43) and critical habitat (Table 45).

## Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing boat ramp projects would result in the following common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 6. Vessel Strikes
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the installation, repair, replacement, and removal of boat ramps structures that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 43. Sea turtles, smalltooth sawfish, and sturgeon may be

affected by the permanent placement of materials on waterbottom areas. We quantified the potential extent of impacts (Table 44) based on (1) the assumptions of the number expected to be new versus repair/replacement, (2) the total number projects estimated by USACE from Table 8, and (3) the construction limitations defined by the PDCs. The "total estimated amount of waterbottom affected" is calculated by multiplying the number of projects expected by the percent of each expected to be new (37%) compared to repair/replacements (63%) and then multiplying the "total number of projects" by the estimated impact per project of 2,592 ft<sup>2</sup>.

Listed Species	USACE Determination	NMFS Determination
Sea Turtles	NLAA	NLAA
Sawfish	NLAA	NLAA
Sturgeon	NLAA	NLAA
Effect	ts Determinations Explained in	Section 2.2
Johnson's seagrass	NE	NE
Corals	NE	N/A
Whales	NLAA	NLAA
Nassau Grouper	Not provided	NLAA
NE= no effect; NLAA= may project may affect ESA-listed	affect, not likely to adversely a d corals, separate consultation is	ffect; $N/A = not$ applicable (if a required).

 Table 43. USACE and NMFS Determinations on the Effects of Boat Ramp Activities (Activity 6) on ESA-listed Species Listed in Table 5

The effects analysis for Johnson's seagrass, corals, whales, and Nassau grouper was provided at the beginning of Section 2.2.

Table 44.	<b>Estimated Amount of</b>	Waterbottom	Affected by	from Boat	Ramp per	5-Year
Period						

Project Type	Percent of Projects	Total Number of Projects (n = 708)	Expected Loss Per Project	Total Estimated Waterbottom Affected (ft <sup>2</sup> )	Total Estimated Waterbottom Affected (ac)
New	37%	261.96	2,592	679,000.32	15.59
Repair/ replacement	63%	446.04	2,592	1,156,135.68	26.54
Total	100%	708		1,835,136.00	42.13

Over a 5-year period, an estimated 42.13 ac of waterbottom area may be altered by the installation of boat ramps. This estimate includes projects in both Florida and the U.S. Caribbean. Since 26.54 ac of this waterbottom area will be the repair or replacement of boat

ramps in the same footprint, only 15.59 ac of waterbottom area will be covered and changed by the placement of new boat ramps resulting in the loss of shallow, nearshore waters. The 15.59 ac loss of shallow, nearshore habitat from new boat ramp projects is unlikely to be concentrated as the array of individual projects covered under Activity 6 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean).

- <u>Sea Turtles</u>: Sea turtles may be affected by the loss of 15.59 ac of shallow, nearshore waters, which could be used for foraging, from new boat ramp projects. However, the effect on sea turtles of the potential loss of foraging habitat is insignificant.
  - The affected areas may include seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes boat ramp projects from the Opinion where Johnson's seagrass is found within the project footprint and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
  - Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
  - Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. However, PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affects listed corals and excludes projects if nonlisted corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
  - Boat ramp activities may cover or remove areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. The 15.59 ac of impact is very small compared to the remaining large areas that support sea turtle prey species. Sea turtles can travel long distances to forage. The new boat ramp projects covered under Activity 6 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites.

Given the above, the effect to sea turtles from the potential loss of foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the boat ramp activities under Activity 6 are not likely to adversely affect sea turtles.

• <u>Smalltooth sawfish</u>: Smalltooth sawfish may be affected by the loss of 15.59 ac of shallow nearshore habitat, which could be used for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. However, the effect on sawfish from the potential loss of foraging and refuge habitat is insignificant. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped,

typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for refuge; these areas have been designated as critical habitat for the species. The PDCs limit activities in smalltooth sawfish critical habitat and limits the Opinion to shoreline projection activities occurring outside of smalltooth sawfish limited exclusion zones (PDC A6.7.4). In addition, projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal is strictly limited by PDC AP.12 to removal associated with other covered activities and to removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. Moreover, under PDC A6.7.1, new or expanded ramps located within smalltooth sawfish designated critical habitat cannot result in the permanent loss of red mangroves. PDC A6.7.1 also prohibits the permanent removal of the depth feature of the shallow, euryhaline essential feature. Therefore, the effect on juvenile sawfish from the loss of small areas of shallow water refuge habitat is expected to be so small as to be undetectable. Boat ramp activities may also cover or remove nearshore areas inhabited by fish and crustaceans that serve as prey for smalltooth sawfish. The area of impact (some amount less than 15.59 ac) is very small compared to the remaining large nearshore areas that support sawfish prey species. Sawfish can travel long distances to forage. The placement of new boat ramps covered under Activity 6 that may affect smalltooth sawfish will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites. Therefore, we believe the effect to smalltooth sawfish from the potential loss of foraging and refuge habitat is insignificant. Based on the routes of effect analyzed here and in Section 2.2, we determined that the boat ramp activities under Activity 6 are not likely to adversely affect smalltooth sawfish

Sturgeon: Sturgeon may be affected by the loss of 15.59 ac of shallow habitat used for foraging. As is noted above, this figure includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. However, the effect on sturgeon of the potential loss of foraging habitat is insignificant. Boat ramp projects may cover and bury nearshore bottom substrates resulting in a loss of areas potentially containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, sturgeon are opportunistic feeders that forage over large areas. Gulf sturgeon select foraging habitat based on substrate composition and depth, rather than prev density, abundance, or diversity. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5 and 13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Hence, Gulf sturgeon, and likely shortnose and Atlantic sturgeon, often occupy waters deeper than those typically affected by boat ramp activities occurring from the shore and sloping out to a depth sufficient to launch a vessel. The area of impact (some amount less than 15.59 ac) is very small compared to the remaining large areas that support sturgeon prey species. The installation of new boat ramps covered under Activity 6 that may affect sturgeon will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Therefore, we believe the effect to sturgeon from the potential loss of foraging habitat is insignificant. Based on the routes of effect analyzed here and in Section

2.2, we determined that the boat ramp activities under Activity 6 are not likely to adversely affect sturgeon.

# Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from boat ramp activities. The NMFS and USACE effects determinations for this category of activity are summarized in Table 45. The estimated loss of critical habitat is summarized in Table 46 and calculated the same way we calculated the area of impacts in Table 44 above.

# Table 45. USACE and NMFS Determinations on the Effects of Boat Ramp Activities (Activity 6) to Designated Critical Habitat

Designated Critical Habitat	USACE Determination	NMFS Determination		
Smalltooth sawfish critical habitat	NLAA	NLAA		
Gulf sturgeon critical habitat	NLAA	NLAA		
Johnson's seagrass critical habitat	NE	NE		
Effects Determinations Explained in Section 2.2				
Loggerhead critical habitat	NE	NE		
Acropora critical habitat	NE	NE		
North Atlantic right whale critical habitat	NE	NE		
Atlantic sturgeon critical habitat	Not provided	NE		
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.				

Table 46. Estimated Amount of Waterbottom Affected by Boat Ramp Activities inSmalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat per 5-YearPeriod

Project Location	Total Number of Projects (Table 8)	Expected Loss Per Project (ft <sup>2</sup> )	Total Area Within Critical Habitat Affected (ft²)	Total Area Within Critical Habitat Affected (ac)
Smalltooth sawfish critical habitat	22	2,592	57,024	1.31
Gulf sturgeon critical habitat	30	2,592	77,760	1.79
Johnson's seagrass critical habitat	2	2,592	5,184	0.12

### Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that boat ramp activities may affect, but are not likely to adversely affect, smalltooth sawfish critical habitat. USACE anticipates that 22 boat ramps projects meeting the requirements of this Opinion may be authorized per 5-year period in smalltooth sawfish critical habitat. In total, the USACE expects that those projects will impact 1.31 ac of waterbottom area in smalltooth sawfish critical habitat. Under PDC A6.7.1, boat ramp activities cannot result in the loss of the essential features of smalltooth sawfish critical habitat (i.e., shallow, euryhaline habitat or red mangroves). Under PDC A6.7.2, boat ramps can only be constructed in waters between the MHWL and -3 ft MLLW if they do not change the shallow, euryhaline depth feature. In addition, under PDC A6.7.3, existing boat ramps can only be repaired or replaced within the same footprint, meaning repair and replacement projects will not result in any new impacts to the depth feature of the shallow, euryhaline essential feature. Boat ramp projects under this Opinion cannot occur in sawfish limited exclusion zones (PDC A6.7.4). The placement of new boat ramp slabs may alter the first few inches of depth through grading the shoreline and placement of the materials, but will not fill or remove the -3ft MLLW depth feature. Therefore, boat ramps will not remove the shallow, euryhaline essential feature, and changes in depth between the MHWL and -3 ft MLLW will not affect this essential feature.

#### Potential Routes of Effect to Gulf sturgeon Critical Habitat

We believe boat ramp activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat. USACE anticipates that 30 boat ramp projects meeting the requirements of this Opinion may be authorized per 5-year period in Gulf sturgeon critical habitat, affecting an estimated 1.79 ac of waterbottom area in Gulf sturgeon critical habitat. Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

1. Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages. Boat ramp projects may cover and bury waterbottom areas resulting in a loss of substrates potentially containing sturgeon prey species. However, we believe the effect to this PCE will be insignificant since the estimated 1.79 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we estimate support sturgeon prey species. Further, not all of the 1.79 ac of habitat affected by boat ramps support prey items or serve as preferred foraging habitat. Gulf sturgeon are suction feeders that tend to forage in calm marine and estuarine waters that support their macroinvertebrate prey including brachiopods, mollusks, worms, and crustaceans (Mason and Clugston 1993). During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5 and 13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Boat ramps occur from the shore and slope deeper to support vessel launching, meaning that not all will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from the installation, repair, or removal of boat ramps; however, the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement as described in PDC AP.10. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high energy areas where turbidity will dissipate very quickly), we believe effects from turbidity on water quality would be insignificant.
- 3. Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages. The installing new boat ramps will remove some waterbottom areas, which can affect sediment quality. The installed boat ramps convert sandy substrate, capable of supporting Gulf sturgeon prey, to hard, man-made ramps that do not support prey species. However, we believe the effects to this PCE will be insignificant since the estimated 1.79 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we believe support sturgeon prey species and Gulf sturgeon foraging. Further, not all of the 1.79 ac of habitat converted solid areas supporting boat ramp projects have the sediment quality needed to support Gulf sturgeon prey or serve as preferred foraging habitat. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5 and 13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Boat ramp projects occur from the shore and slope deeper to accommodate vessel launching, as such, these projects may not all occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.
- 4. <u>Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats</u>. New boat ramp projects could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways also could be obstructed by boat ramp activities in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging. However, effects to this PCE will be discountable. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, PDC A6.10 requires compliance with the noise restrictions for pile and sheet

pile installation in Gulf sturgeon critical habitat migratory restriction zones. Boat ramp projects occurring in areas outside of Gulf sturgeon critical habitat migratory restriction zones will occur along the shore and will not restrict the movement of sturgeon. Therefore, any effects are extremely unlikely to occur.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and effects to safe and unobstructed migratory pathways will be discountable, we believe that boat ramp activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe there will be no effect to Johnson's seagrass critical habitat from boat ramps because PDC A6.9 excludes new or expanded boat ramp projects in Johnson's seagrass critical habitat where the essential features are present, and repair and replacement can only occur within the same footprint. USACE anticipates that 2 boat ramp activities meeting the requirements of this Opinion may be authorized per 5-year period in Johnson's seagrass critical habitat. Given the PDCs, these projects can only occur in areas lacking the essential features or can only occur within the same footprint (preventing new or additional effects to the essential features).

## 2.2.7 Activity 7 (A7): Aquatic Habitat Enhancement, Establishment, and Restoration Activities

## **General Description**

As used in this Opinion, aquatic enhancement includes (1) constructing oyster reefs on unvegetated bottom in tidal waters, (2) constructing living shorelines, including using vegetative plantings and fill material to construct breakwaters parallel to the shore (3) enhancing or establishing submerged aquatic vegetation, (4) constructing artificial reefs, and (5) filling in areas to restore natural contours or improve water quality.

### **Oyster Reefs and Living Shorelines**

Oyster reefs and living shorelines are placed in shallow water environments. Both can be constructed to act as a natural breakwater along eroding shorelines. Living shorelines can be constructed out of different materials, including oysters or limestone boulders. The installation methods and materials used to create oyster reefs and living shorelines are described below. Oyster reefs can be configured in a number of different ways, including as a stand-alone reef or a series of reefs. When placed near shore, oyster reefs also can act as a breakwater/living shoreline (see below). Many oyster reefs are constructed of bags filled with oyster cultch (i.e., oyster shells placed to facilitate new oyster spat recruitment). Often, these bags are hand-placed. Sometimes, a perimeter is created with the oyster bags and the center is filled with loose cultch so that the loose material is contained. Loose material is often offloaded using barge-mounted mechanical equipment. Some oyster reefs are created by placing flat mats weighted to the seafloor with oyster cultch attached. All of these methods rely on natural recruitment of live oysters from the surrounding waters. Figure 18 provides images of different types of oyster reefs and equipment and materials used in oyster reef construction.



Figure 18. Oyster reefs. The left image shows oyster bags, the middle is an oyster mat, and the right is a barge filled with loose oyster cultch. All 3 images are from the Charlotte Harbor Habitat Restoration Plan (Boswell et al. 2012).

Living shorelines are created by placing a breakwater parallel to the shore. Breakwaters can be constructed of artificial materials such as prefabricated structures like reef balls or constructed of natural materials such as oysters, limestone boulders, mangrove island barriers, or biologs. Shoreline structures are aligned parallel with the shore as straight-line sections or shaped into crescent sections to reduce wave attenuation. Vegetation is often planted landward of the structures, between structure and the shoreline, to stabilize the shoreline. Below is a cross section diagram showing the transition from the living breakwater to the uplands (Figure 19). The PDCs require that all living shoreline structures must have breaks or gaps in the sections of living shoreline to allow for tidal flushing and species movement. We developed PDCs to set the living shoreline lengths and gap widths based on the practices and recommendations of FDEP and non-government organizations that specialize in living shorelines.



Figure 19. Living shoreline. This sample living shoreline cross-section is provided by NOAA's Habitat Conservation and Restoration website (http://www.habitat noaa.gov/restoration/techniques/lsimplementation.html).

### Seagrass Planting

Seagrass planting is often used to develop or restore seagrass beds for the purpose of aquatic habitat restoration. Sometimes, seagrass is planted on a site with no site preparation. In some cases, prior to planting, the site elevation must be restored, which is typically done by placing fill (either suitable loose sediment or bagged sediment) in blowholes/dredge holes or prop scars until

the holes are filled to an elevation level with the adjacent area. Loose material is often offloaded using barge-mounted mechanical equipment. Seagrass plants typically are obtained from laboratories, specialty nurseries, or from transplants from existing seagrass bed. Bird roosting stakes sometimes are used to speed seagrass recovery by taking advantage of a natural source of fertilizer (Figure 20). Bird roosting stakes normally are small wood or plastic stakes installed by hand.



Figure 20. Seagrass restoration. The left image is the placement of sediment to return a blowhole to pre-injury elevation. The image on the right shows bird stakes placed in a restoration area (Both images are from <u>www.darrp.noaa.gov</u>).

## Artificial Reefs

Artificial reefs can consist of a variety of materials. This Opinion is limited to reefs constructed of the materials described in the PDCs below. Materials are typically transported to the site by barge. Pre-fabricated reef modules are off-loaded by crane and loose concrete material is dumped over side onto a pre-surveyed, defined location. Pre-fabricated structures are available in a variety of shapes, including those pictured below (Figure 21).



Figure 21. Artificial reef materials. The left image is an open top and bottom tetrahedron design, the middle image is a series of discs mounted to a pile, and the right image is a reef ball. Left and middle images are from Reef Maker (http://www.reefmaker.com) and the right image is from the Reef Ball Foundation (www.reefball.org).

### Fill to Restore Natural Contours or Improve Water Quality

In areas where historic dredging has occurred or areas with ruts or scars from vessel groundings, fill is sometimes the best way to restore the natural contours of the area so that aquatic vegetation can recover or so that water quality can improve. With respect to water quality, if an area is significantly deeper than the surrounding habitat, it can create hypoxic conditions where oxygen levels in the deeper area are so low as to become deadly to aquatic life. Depending on the size and depth, these areas may be filled with hand held equipment or from mechanical equipment such as backhoes and bucket dredges from land or barge.

# Project Design Criteria

# PDCs Specific to Activity 7 for Aquatic Habitat Enhancement, Establishment, and Restoration Activities:

A7.1. Only native plant species can be planted.

Additional Conditions for living shoreline and oyster habitat on unvegetated bottom in tidal waters:

- A7.2. Oyster reef materials shall be placed and constructed in a manner that ensures that materials will remain stable and that prevents movement of materials to surrounding areas (e.g., oysters will be contained in bags or attached to mats and loose cultch must be surrounded by contained bagged oysters or another stabilizing feature).
- A7.3. Oyster reef materials must be placed in designated locations only (i.e., the materials shall not be indiscriminately or randomly dumped or allowed to spread outside of the reef structure).
- A7.4. Living shorelines can only be constructed in unvegetated, nearshore water along shorelines to create tidal marshes or mangrove habitat for the purpose of shoreline erosion control or aquatic habitat enhancement. Native plants can be placed along the shoreline or between the shoreline and the living shoreline structure.
- A7.5. Living shoreline structures and permanent wave attenuation structures can only be constructed out of the following materials: oyster breakwaters (described above in the project description and A7.2), clean limestone boulders or stone (sometimes contained in metal baskets or cages to contain the material), small mangrove islands, biologs, coir, rock sills, and pre-fabricated structures made of concrete and rebar that are designed in a manner so that they do not trap sea turtles, smalltooth sawfish, or sturgeon. Reef balls or similar structures that are not open on the bottom, open-bottom structures with a top opening of at least 4 ft, and reef discs stacked on a pile are pre-fabricated structures are designed in a manner so that they do not trap sea turtles. Other materials may be used for living shorelines if pre-approved by NMFS to ensure that they are stable and not an entanglement risk to listed species. The approval process to use other materials is described in the Section 2.3 (Project-Specific Review).
- A7.6. Both living shoreline and oyster reefs must have 5 ft gaps at least every 75 ft in length, as measured parallel to the shoreline and at the sea floor, to allow for tidal flushing and species movement.

Additional Conditions for the establishment or restoration of submerged aquatic vegetation:

A7.7. The placement of loose or bagged sediment suitable for the project site in blowholes/dredge holes or in prop scars, and berm redistribution or sod replacement in excavations, must be to an elevation level with or otherwise consistent with the adjacent area.

- A7.8. This Opinion covers leveling submerged spoil piles or berms if necessary to level the restoration area to match the elevation of adjacent seagrass beds.
- A7.9. Exclusion cages may be used around seagrass restoration areas if necessary to allow the seagrass beds to establish themselves to the point where they are sustainable after the cages are removed. Exclusion cages can only be used on a temporary basis, for a period not to exceed 4 months. Each exclusion cage must be securely fastened to the substrate so that it does not become detached. All cages must be constructed of firm, taut materials and cannot include any loose mesh, thin twistable wire, or rope that could twist or become entangled or present an entanglement risk to species.
- A7.10. Seagrass transplantation and harvesting from the donor site may occur only by hand. Donor sites could include (i) upland seagrass farms, (ii) areas with seagrasses that would be impacted by another project, or (iii) existing seagrass beds, as long as the seagrass is removed in a manner that is not detrimental to the existing seagrass bed. Transplantation methods may include, but are not limited to, plugging devices, manual transplant, peat pellets, peat pots, and coconut fiber mats. No in-water machinery (e.g., marsh buggies, track hoe) may be used in harvesting or transplanting the seagrasses. The selection of and harvesting from seagrass donor sites shall be coordinated with NMFS Habitat Conservation Division. This Opinion does not cover transplantation of the invasive seagrasses (e.g., *Halophila stipulacea*).
- A7.11. In Florida, this Opinion covers installation of stakes to attract birds, if necessary or appropriate for the project. Bird stakes should not be used in areas where additional nutrients may be detrimental to the seagrass. Bird stakes are not authorized in the U.S. Caribbean.
- A7.12. This Opinion covers installation of signage (supported on piles or anchored) if the signs are necessary to prevent motorized boats from entering the area and anchoring. Signs must be sized and placed in a manner that prevents the loss of native seagrasses from sign shading.

Additional conditions for the installation of artificial reefs from the placement of man-made materials:

- A7.13. Artificial reef materials shall be clean and free from asphalt, creosote, petroleum, other hydrocarbons and toxic residues, loose free-floating material, or other deleterious substances.
- A7.14. New reef sections are limited to 1 reef section measuring <sup>1</sup>/<sub>4</sub>- by <sup>1</sup>/<sub>4</sub>-nmi area (40 ac) in size with a distance of 500 ft between each section. Offshore reefs shall maintain a minimum vertical clearance of twice the height of the structure from the top of the deployed material relative to the MLW at all times.
- A7.15. Reauthorization of existing reefs is limited to the previously permitted size. Approved

materials defined in PDC A7.19 can be added to the existing reef area.

- A7.16. No artificial reef materials shall be deployed until a benthic assessment of the bottom conditions has been accomplished by diver or submersible video camera. The inspection of the deployment area may occur at the time of deployment but no more than 1 year prior to deployment. The permittee shall maintain a deployment buffer of at least 200 ft from any submerged aquatic resources, including seagrasses, macroalgae, hard or soft coral (including coral reefs), sponges, oysters, or hard bottom when placed in areas of sand. If materials are off-loaded from a barge or placed in areas that may generate turbidity (e.g., areas with fines or muck), a 500 ft buffer is required.
- A7.17. This Opinion does not cover the use of mid-water fish aggregation devices.
- A7.18. All reefs must be cleaned annually to remove marine debris and derelict fishing line in areas safely accessible to recreational SCUBA divers. Cleanup efforts shall follow the PDCs for Activity 9, marine debris removal, and all pertinent general PDCs.

# Additional conditions for reef materials:

- A7.19. Individual reef units or modules must weigh at least 500 pounds. Reef materials shall be clean and free from asphalt, petroleum, other hydrocarbons, and toxic residues, as well as loose, free-floating material, or other deleterious substances. All artificial reef materials and/or structures will be selected, designed, constructed, and deployed to create stable and durable marine habitat. Only the following reef materials may be used under this Opinion:
  - A7.19.1. Prefabricated artificial reef modules composed of ferrous and/or aluminumalloy metals, concrete, rock, or a combination of these materials.
  - A7.19.2. Natural rock boulders and pre-cast concrete material, such as culverts, stormwater junction boxes, power poles, railroad ties, jersey barriers, or other similar concrete material.
  - A7.19.3. Clean steel and concrete bridge or large building demolition materials such as slabs or piles with all steel reinforcement rods cut at the base of the concrete so no rebar or metal protrudes from the concrete.
- A7.20. Reef structures, materials, and installation methods shall be designed and deployed to prevent entanglement and entrapment of listed species. Open-bottom pre-fabricated reef modules may not be used unles the module also has a top opening sufficiently large to allow a turtle to escape. Approved ope-bottom modules include:
  - A7.20.1. Three-sided modules where each side of the tope opening is at least 36-in in length along its edge.
  - A7.20.2. Four or more sided modules where each side of the top opening is at least 40-in in length along its edge.
  - A7.20.3. Modules with a round opening with a diameter of at least 40-in (oval openings are not allowed unless a 40-in diameter circle space can fit within the oval).
  - A7.20.4. Modules that are approved by the FWS Artificial Reef Program as being turtle friendly. FWS is currently working on developing this list.

No open-bottom modules are allowed that include additional modules, discs, or othr materials stacked or placed on or immediately adjacent to the top opening, as they may prevent turtles from easily escaping.

- A7.21. This Opinion does not cover projects that use explosives to deploy reef material.
- A7.22. If pile placement is required in the construction of a reef, such placement must comply with the PDCs for Activity 2, pile-supported structures, and all applicable general PDCs.

Fill to restore natural contours or improve water quality:

- A7.23. Fill of scars or ruts caused by vessel groundings or similar activities must match the surrounding natural elevation.
- A7.24. This Opinion covers fill of deep holes or canal bottoms that are determined to be hypoxic (i.e., that have critically low dissolved oxygen levels).

# Additional PDCs for Activity 7 applicable in critical habitat:

# In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat, as described below.

- A7.25. Smalltooth sawfish critical habitat: Oyster reefs, living shorelines, and artificial reefs cannot be placed in waters containing the shallow, euryhaline essential feature. Fill to restore natural contours or improve water quality and seagrass restoration can occur in waters containing the shallow, euryhaline essential feature, as long as the activity meets the PDCs for Activity 7 and all pertinent general PDCs. No aquatic habitat enhancement, establishment, or restoration activities are allowed in areas identified as smalltooth sawfish limited exclusion zones (Section 2.1.1.1).
- A7.26. Gulf sturgeon critical habitat: Oyster reefs, living shorelines, and seagrass restoration in Gulf sturgeon critical habitat are restricted to areas that are in water depths shallower than -6 ft (-2 m) MHW (i.e., between the shoreline and -6 ft deep). Artificial reef structures cannot be placed in Gulf sturgeon critical habitat. Fill to restore natural contours or improve water quality can occur in Gulf sturgeon critical habitat, regardless of project depth. Living shorelines, oyster reefs, and artificial reefs cannot be placed in Gulf sturgeon critical natural reefs cannot be placed in Gulf sturgeon critical reefs cannot be placed in Gulf sturgeon critical natural contours or improve water quality can occur in Gulf sturgeon critical natural contours or improve water quality can occur in Gulf sturgeon critical natural contours of project depth. Living shorelines, oyster reefs, and artificial reefs cannot be placed in Gulf sturgeon critical habitat migratory restriction zones, defined in Section 2.1.1.2.
- A7.27. North Atlantic right whale critical habitat: All artificial reefs must meet specifications below. Oyster reefs, living shorelines, seagrass restoration, and fill to restore natural contours or improve water quality can occur in North Atlantic right whale critical habitat, as long as those activities meet the PDCs for Activity 7 and any pertinent general PDCs, as described above.

A7.27.1. No artificial reefs can be placed in water shallower than 30 ft deep A7.27.2. The maximum reef height off the sea floor is 20 ft

- A7.27.3. The maximum footprint of new reefs shall be  $1 \text{ nmi}^2$ . If a new reef is added to an existing artificial reef, the total footprint of the combined reefs must not exceed  $1 \text{ nmi}^2$ .
- A7.27.4. Density of newly permitted reefs shall not exceed 2 reefs (old or new) per 10 nmi<sup>2</sup>
- A7.27.5. All effort should be made to avoid placing reef material during North Atlantic right whale calving season (November 15 through April 15). If reef material has to be placed during North Atlantic right whale calving season, then the following additional measures are required:
  - The maximum speed for all vessels involved in placing the reef material is 10 knots.
  - Deployments cannot be conducted at any time when lighting or weather or sea conditions (e.g., darkness, rain, fog, sea state) prevent visual monitoring of the project area.
  - Deployment activities will not commence until the protected species observer reports that no marine mammals or sea turtles have been sighted for at least 60 minutes.
  - Deployment activities will cease immediately if sea turtles or marine mammals are sighted within the project area.
  - Deployment activities will not recommence until the protected species observer reports that no marine mammals or sea turtles have been sighted for at least 60 minutes.
- A7.28. *Acropora* critical habitat: This Opinion does not cover any aquatic habitat enhancement, establishment, or restoration activities in *Acropora* critical habitat where the essential feature is present.
- A7.29. Johnson's seagrass critical habitat: Living shorelines, oyster reefs, and artificial reefs cannot be placed in waters shallower than -13 ft MHW within the geographic boundaries of Johnson's seagrass critical habitat. Seagrass restoration and fill to restore natural contours or improve water quality can occur in Johnson's seagrass critical habitat regardless of depth, as long as those activities meet the PDCs for Activity 7 and any pertinent general PDCs, as described above.
- A7.30. Loggerhead critical habitat: Living shorelines, oyster reefs, and artificial reefs cannot be placed in nearshore reproductive habitat of loggerhead critical habitat. Seagrass restoration and fill to restore natural contours or improve water quality can occur in nearshore reproductive habitat of loggerhead critical habitat, as long as those activities meet the PDCs for Activity 7 and any pertinent general PDCs, as described above.
- A7.31. U.S. Caribbean Sea Turtle Critical Habitat (NA DPS of green, Hawksbill, and Leatherback Sea Turtle Critical Habitat): No aquatic enhancement activities (living shorelines, oyster reefs, artificial reefs, seagrass restoration, and fill to restore natural contours or improve water quality) can occur within sea turtle critical habitat in the U.S. Caribbean.
## Assumptions

The USACE anticipates 267 aquatic enhancement activities meeting the requirements of this Opinion may be authorized per 5-year period. They also provided estimates for the number of each type of aquatic enhancement project and the anticipated average waterbottom area that would be impacted per project, as shown in Table 47, below. For example, living shoreline, oyster reef, and artificial reef projects place materials on the waterbottom covering these areas (e.g., living shoreline and oyster reefs cover an average of 0.25 ac of waterbottom, and artificial reefs will cover 20 ac of waterbottom). For seagrass planting and fill to restore natural contours or improve water quality, the USACE provided the estimated area to be enhanced.

Project Type	Percent of Total Projects	Estimated Waterbottom Impacts Per Project (ac)
Living shorelines and oyster reefs	65%	0.25
Seagrass planting	15%	0.25
Artificial reefs	15%	20
Fill to restore natural contours or improve water quality	5%	1.25
Total	100%	

Table 47.	USACE's Est	timated Numb	er of Aquatic	Enhancement	Projects per 5	-year Per	iod
and the A	nticipated Im	pacts per Proj	ect.				

Information regarding the types, size, and numbers of projects is then combined with the breakdown of the estimated number of projects anticipated both within and outside of critical habitat units (see Table 8) to determine the potential critical habitat effects from this category of activity.

Thus, to analyze the effects of aquatic enhancement, we looked the USACE's assumptions regarding (1) the total number of covered aquatic enhancement activities by type to be covered per 5-year period (Table 8) and the number of these that would occur in smalltooth sawfish, Gulf sturgeon, and Johnson's seagrass critical habitat (percent estimates provided by USACE in Table 47). Based on this information, we determined the potential effects to species (Table 48) and critical habitat (Table 50).

# Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Aquatic enhancement projects would result in the following common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

1. Direct Physical Effects from Construction Activities

- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 5. Limiting Species' Movement and Access to Habitat
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, sturgeon, and whales from aquatic habitat enhancement, establishment, and restoration activities that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 48. Sea turtles, smalltooth sawfish, sturgeon, and whales may be affected by the placement of materials. We quantified the potential extent of impacts based on (1) the assumptions, (2) the number and location of projects estimated by USACE (Table 8), and (3) the construction limitations defined by the PDCs. In Table 49, the "total waterbottom impacts" was determined by multiplying the "total number of projects" by the "estimated waterbottom impacts per project."

 Table 48. USACE and NMFS Determinations of the Effects of Aquatic Enhancement,

 Enhancement (Activity 7) on ESA-listed Species Listed in Table 5

Listed Species	<b>USACE</b> Determination	NMFS Determination		
Sea Turtles	NLAA	NLAA		
Sawfish	NLAA	NLAA		
Sturgeon	NLAA	NLAA		
Whales	NE	NLAA		
Effects I	Determinations Explained in	Section 2.2		
Johnson's seagrass	NE	NE		
Corals	NE	N/A		
Nassau Grouper	Not provided	NLAA		
NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).				

The effects analysis for Johnson's seagrass, corals, and Nassau Grouper was provided at the beginning of Section 2.2.

Project Type	Percent of Total Projects	Total Number of Projects (n= 267)	Estimated Waterbottom Affected Per Project (ac)	Total Estimated Waterbottom Affected Per Project (ac)
Living shorelines and oyster reefs	65%	173.55	0.25	43.39
Seagrass planting	15%	40.05	0.25	10.01
Artificial reefs	15%	40.05	20	801.00
Fill to restore natural contours or improve water quality	5%	13.35	1.25	16.69
Total	100%	267		871.09

 

 Table 49. Estimated Amount of Waterbottom Affected by Aquatic Enhancement per 5-Year Period

Over a 5-year period, 871.09 ac of waterbottom may be affected by the aquatic enhancement activities (Table 49). This estimate includes projects in both Florida and the U.S. Caribbean. These waterbottom impacts are unlikely to be concentrated as the array of individual projects covered under this Opinion for Activity 7 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean).

- Sea turtles: Sea turtles may be affected by aquatic enhancement activities as these activities may remove foraging habitat (by placing structures on top of the habitat). Effects may also be beneficial, as these projects may enhance foraging or resource areas used by sea turtles.
  - Living shorelines, oyster reefs, and artificial reefs could be placed on foraging areas used by sea turtles. Seagrasses growing in nearshore waters are an important forage resource for green sea turtles. Living shoreline, oyster reefs, and nearshore artificial reef projects could be placed on of waterbottom that could contain seagrasses used by green turtles for foraging; however, under PDC AP.13, the Opinion does not apply to projects where Johnson's seagrass is found within the project footprint and recommends that effects to non-ESA listed, native seagrasses be avoided and minimized to the extent practicable. In addition, seagrass restoration projects are allowed to place temporary exclusion cages around new seagrass transplants for a maximum of 4 months to allow the seagrass beds to become established (PDC A7.9). These cages will prevent sea turtles from foraging in this area for a short period of time. Since sea turtles can travel long distances to forage, we believe the potential loss of small areas seagrass foraging areas from the placement of materials or the temporary exclusion of areas from seagrass transplant cages, will have insignificant effects on the ability of sea turtles to forage.
  - Living shoreline, oyster reef, and artificial reef projects may cover or remove areas that may support sea turtle prey species, including the crustaceans and mollusks that serve as

prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are common throughout Florida and the U.S. Caribbean. Since sea turtles can travel long distances to forage, we believe the potential loss of these foraging areas will have insignificant effects on the ability of sea turtles to forage.

- Seagrass restoration in areas in up to 10.01 ac of waterbottom has the potential to increase foraging habitat for green sea turtles. In addition, the filling deep holes to restore natural contours or to improve low water quality in up to 16.69 ac of waterbottom also has the potential to increase foraging areas by returning low quality waterbottom areas to the natural habitat that could recruit foraging resources such as (1) seagrasses used by green sea turtles; (2) crustaceans or mollusks that serve as prey for loggerhead; or (3) fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles.
- Limestone outcroppings and worm-rock reefs that are important developmental habitat for juvenile green turtles; however we believe there will no effect to this foraging habitat since, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
- Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates, which may be found in the waterbottoms affected by Activity 7. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect ESA-listed corals, and excludes projects if non-listed corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint. Activity 7 also allows the installation of artificial reefs and living shorelines on the waterbottom. The installation of these materials may enhance the areas that provide foraging habitat for hawksbills.

The array of individual projects covered under this Opinion for Activity 7 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites. Given the above, effect to sea turtles from interactions with foraging resources is insignificant or beneficial.

Sea turtles can become trapped in certain artificial reef structures. It is possible for a sea turtle to position itself under the edge of open-bottom reef structures and then become wedged or trapped inside the reef material when trying to extract itself. PDC A7.20 requires that reef structures, materials, and installation methods be designed and deployed to prevent entanglement and entrapment of listed species and specifically prohibits the use of open-bottom structures, unless the structure has at least a 4-ft opening at the top of the structure for turtles to escape (an example of an open top and bottom tetrahedron structure is shown in the left side of Figure 21 above). Based on these requirements, entrapment from artificial reefs is extremely unlikely to occur and thus the effect on sea turtles from potential entrapment is considered discountable.

Sea turtles can become entangled in fishing debris that accumulates on artificial reefs. In order to minimize the risk of fishing debris (e.g., broken fishing line and fishing gear), the

PDCs require underwater cleanup efforts annually at all artificial reefs (PDC A7.18). By regularly removing debris, we believe the risk of entanglement with sea turtles in discountable.

In total, based on the routes of effect analyzed here and in Section 2.2, we determined that aquatic restoration activities under Activity 7 are not likely to adversely affect sea turtles.

- Smalltooth sawfish: Smalltooth sawfish may be affected by oyster reef and living shoreline projects, which will be placed on 43.39 ac waterbottom, areas that the species may have used for foraging and refuge. As is noted above, this estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. However, we believe the effect on sawfish from the potential loss of nearshore foraging and refuge habitat is insignificant.
  - For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline waters and red mangroves for forage and refuge; these areas have been designated as critical habitat for the species. Projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal for projects is strictly limited by PDCs AP.12 to removal related to other covered activities and to removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. PDC A7.25 prohibits the placement of oyster reefs, living shorelines, and artificial reefs in areas where the shallow, euryhaline essential feature of smalltooth sawfish critical habitat is present, and prohibits activities in areas identified as smalltooth sawfish limited exclusion zones. During construction, smalltooth sawfish can use surrounding areas for forage and refuge and we expect them to return. Therefore, we believe the effect on sawfish of the potential loss of nearshore foraging and refuge habitat is insignificant.
  - The installation of artificial reefs also will cover areas of waterbottom. In particular, we anticipate that artificial reefs will impact 801 ac of waterbottom. However, because artificial reefs are typically placed in deeper waters than those used by juvenile smalltooth sawfish, none of these projects are expected to have effects to foraging or refuge habitat.

Seagrass restoration and the filling of scars and holes to return the area to natural contours and to improve water quality are allowed in shallow, euryhaline waters, but we do not expect such activities to affect how sawfish use these areas once construction is complete.

Smalltooth sawfish could become entangled in the cages placed around seagrass restoration areas. To minimize this risk, PDC A7.9 requires that these exclusion cages be securely fastened and made of firm, taut materials and cannot include any loose mesh, thin twistable wire, or rope that could twist or become entangled. The placement of cages is also limited to a maximum of 4 months (PDC A7.9). No aquatic restoration activities, including the seagrass restoration projects for which cages might be used, can be placed in smalltooth sawfish limited exclusion zones, as these areas may be used more frequently for sawfish pupping (PDC A7.25). Based on these PDCs, we believe it is extremely unlikely that the

placement of these cages will affect smalltooth sawfish, and thus the effect on the species from the cages is discountable.

The array of individual projects covered under this Opinion for Activity 7 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida). Based on the routes of effect analyzed here and in Section 2.2, we determined that aquatic restoration activities under Activity 7 are not likely to adversely affect smalltooth sawfish.

- Sturgeon: Sturgeon may be affected by projects placed on the waterbottom that they may have used for foraging habitat. In addition, installing living shorelines and nearshore oyster reefs and restoring holes to natural contours may benefit Gulf sturgeon by increasing foraging areas.
  - o Living shorelines, oyster reefs, artificial reefs, and seagrass restoration may cover and bury bottom substrates containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, sturgeon are opportunistic feeders that forage over large areas. Gulf sturgeon select foraging habitat based on substrate composition and depth, rather than prey density, abundance, or diversity. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5 and 13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002). Within Gulf sturgeon critical habitat, we do expect Gulf sturgeon to be affected by these projects since PDC A7.26 limits living shorelines, oyster reefs, and seagrass restoration projects in these areas to waters shallower than -6 ft (-2 m) MHW. In addition, PDC A7.26 prohibits artificial reef structures within the geographic boundary of Gulf sturgeon critical habitat since these projects often occupy large areas and would potentially remove large areas of Gulf sturgeon foraging habitat. Outside of Gulf sturgeon critical habitat, we expect effects to all sturgeon (Gulf, shortnose, and Atlantic) to be insignificant. As is noted above, sturgeon are opportunistic feeders and we expect they will be able to use other areas for foraging as the projects will be separated spatially (throughout Florida) and temporally (over a 5-year period). Thus, we believe the effect on sturgeon from the potential loss of nearshore foraging habitat is insignificant.
  - In addition, installing living shorelines and nearshore oyster reefs may provide an indirect benefit to Gulf sturgeon by enhancing the diversity of prey available to them. This may happen by the creation of a patchwork of oyster reefs that, over time, provide more diverse and structurally complex habitat for prey species (Boudreaux et al. 2006). As these prey species (e.g., macrofaunal species such as amphipods, polychaetes, gastropods, and bivalves) increase in abundance in the shallow nearshore project area, there will be a spill-over effect to neighboring areas that are deeper than 6 ft, where increased prey abundance will benefit Gulf sturgeon that may forage in those waters in the long-term. The use of oyster reefs as breakwaters to mitigate against coastal erosion also encourages nektonic production that could lead to greater prey availability in the immediate surroundings for Gulf sturgeon (Seitz et al. 2006).
  - The filling deep holes with low water quality in up to 16.69 ac of waterbottom also has the potential to increase foraging areas by returning low quality waterbottom areas to the natural habitat that could recruit Gulf sturgeon prey species.

Based on the routes of effect analyzed here and in Section 2.2, we determined that aquatic restoration activities under Activity 7 are not likely to adversely affect sturgeon.

Whales: Whales may be affected by the placement of 801 ac of artificial reefs in waters deep enough to be accessible to whale species, though not all 801 ac of artificial reef material will be placed in waters accessible to whales. Whales may have used these waters as a migratory pathway and as calving areas. North Atlantic right whales frequently come much closer to shore than other whale species, where the water is shallower. In North Atlantic right whale critical habitat, under PDC A7.27, artificial reef materials cannot be placed in waters shallower than 30 ft deep and cannot extend more than 20 ft off the sea floor. In addition, under the same PDC, the density of artificial reefs in North Atlantic right whale critical habitat is limited to no more than 2 reefs (including the new reef and any existing reefs) measuring 1 nmi<sup>2</sup> per 10 nmi<sup>2</sup> area (PDC A7.27). By limiting the vertical height of reef materials to 20 ft, and limiting reef placement in shallow waters (waters shallower than 30 ft deep), we believe that artificial reef will not create an obstacle for migration and movement within calving areas for North Atlantic right whales in their critical habitat. Outside of North Atlantic right whale critical habitat, the PDCs also limit the density of material placement to 1 reef section measuring <sup>1</sup>/<sub>4</sub>- by <sup>1</sup>/<sub>4</sub>-nmi area (40 ac) in size within a distance of 500 ft between each section, and states that offshore reefs shall maintain a minimum vertical clearance of twice the height of the structure from the top of the deployed material relative to the MLW at all times (PDC A7.14). These density limitations allow whales to easily move over and around these reefs and ensure that the reefs do not dramatically change the landscape in this habitat. Based on these restrictions, we believe that the change in depth and restrictions or limitations on species movement from the placement of materials will be so minimal that it will have insignificant effects on ESA-listed whales using the area for foraging or migrating.

#### Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from aquatic enhancement activities and to North Atlantic right whale critical habitat from the placement of artificial reefs. NMFS and USACE's effects determinations for this category of activity are summarized in Table 50 and the estimated impacts to each critical habitat unit are summarized in Table 51. In these tables, N/A denotes activities that are not allowed in a particular critical habitat unit based on the PDCs of this Opinion, as described below.

 Table 50. USACE and NMFS Determinations on the Effects of Aquatic Enhancement (Activity 7) to Designated Critical Habitat

	<b>USACE Determinations</b>			NM	<b>NMFS</b> Determinations		
Project Type	Sawfish Critical Habitat	Gulf sturgeon Critical Habitat	Johnson's seagrass Critical Habitat	Sawfish Critical Habitat	Gulf sturgeon Critical Habitat	Johnson's seagrass Critical Habitat	
Living shoreline and oyster reefs				NE	NLAA	NE	
Seagrass restoration				NE	NLAA	NLAA	
Artificial reefs	NE	NE	NE	NE	N/A <sup>27</sup>	NE	
Fill to restore natural contours or improve water quality				NE	NLAA	NE	
Effects Determinations Explained in Section 2.2							
		USACE Determinations		s N	MFS Determin	nations	
Loggerhead critica	ıl habitat		NE		NE		
Acropora critical h	nabitat	-	NE		NE		
North Atlantic rigl critical habitat	nt whale	NE		NE for oyster reefs, living shorelines, seagrass restoration, and fill to restore natural contours or water quality NLAA for artificial reefs, as discussed below		s, living storation, and contours or by l reefs, as ow	
Atlantic sturgeon of habitat	critical	Not provided			NE		
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.							

<sup>&</sup>lt;sup>27</sup> Under PDC A7.26, artificial reef structures cannot be placed in Gulf sturgeon critical habitat.

Since we do not know how many of each of these aquatic restoration methods will occur in or may affect each critical habitat unit, we based the calculations in Table 51 on the following assumptions:

<u>Fill to restore natural contour or improve water quality</u>: Since projects involving filling areas to restore the natural contour or to improve water quality are expected to be rare, the USACE estimates that these projects will not exceed 5% of all projects in critical habitat, and we assume each such project will affect 1.25 ac of waterbottom. Thus, we assumed that fill to restore the natural contour = (number of projects per critical habitat unit x 5% for fill projects) x 1.25 ac of impact per project.

<u>Artificial reefs</u>: Under the PDCs, artificial reefs cannot be placed in the critical habitat units (A7.26-Gulf sturgeon critical habitat) or in areas that support the essential features of critical habitat (A7.25-smalltooth sawfish critical habitat, see also PDC AP.12 regarding impacts to mangroves; A7.29-Johnson's seagrass critical habitat<sup>28</sup>), with the exception of North Atlantic right whale critical habitat. Therefore, we believe that artificial reefs either will not occur in or will not affect these critical habitat units. The effects of artificial reefs on North Atlantic right whale critical habitat are discussed in a separate section below.

Living shoreline and oyster reefs, and seagrass restoration: Since we know 5% of the projects occurring in critical habitat will be fill projects, and we know that no artificial reef projects will occur in designated critical habitat or in areas where they can affect the essential features of critical habitat, we can assume that the 95% of projects in areas supporting the essential features of critical habitat will be a combination of (1) living shoreline and oyster reef projects and (2) seagrass restoration projects, each of which result in an average of 0.25 ac of impact per project. Hence, we can calculate the estimated impacts these projects combined to determine the potential extent of impacts, as follows: Living shoreline and oyster reefs, and seagrass restoration = (number of projects per critical habitat unit x 95% living shoreline and oyster reefs, and seagrass restoration projects) x 0.25 ac of impact per project.

However, in certain circumstances, living shoreline and oyster reef projects cannot be placed in areas that support the essential features of critical habitat (A7.25-smalltooth sawfish critical habitat, see also PDC AP.12 regarding impacts to mangroves; A7.29-Johnson's seagrass critical habitat), (shown as N/A in Table 51 below). In those instances, we will assume that the remaining 95% of projects occurring in areas supporting the essential features of critical habitat are seagrass restoration projects and attribute impacts to those projects.

<sup>&</sup>lt;sup>28</sup> Under this PDC, artificial reefs cannot be placed in waters shallower than -13 ft MHW within the geographic boundaries of Johnson's seagrass critical habitat. We believe areas deeper than -13 ft MHW do not support enough light transmittance to function as critical habitat.

Table 51.	Maximum Area Supporting the Essential Features of Smalltooth Sawfish, Gulf
Sturgeon,	and Johnson's Seagrass Critical Habitat that May Be Affected by Aquatic
Restoratio	on Projects per 5-Year Period

Type of Dredging	Percent of Projects	Total Number of Projects per Critical Habitat	Estimated Average Area Supporting Essential Features Affected Per Project (ac)	Estimated Total Area Supporting Essential Features Affected (ac)
Smalltooth sawfish critical habita	nt (n = 46)			
Living shorelines and oyster reefs	0 <sup>29</sup>	N/A	N/A	0.00
Seagrass restoration	95%	43.70	0.25	10.93
Artificial Reefs	030	N/A	N/A	0.00
Fill to restore natural contours or improve water quality	5%	2.30	1.25	2.88
Total	L	46		13.81
Gulf sturgeon critical habitat (n =	= 91)			
Living shorelines and oyster reefs and seagrass restoration	95%	86.45	0.25	21.61
Artificial Reefs	031	N/A	N/A	0.00
Fill to restore natural contours or improve water quality	5%	4.55	1.25	5.69
Total	[	91	1	27.31
Johnson's seagrass critical habita	nt (n = 7)			
Living shorelines and oyster reefs	032	N/A	N/A	0.00
Seagrass restoration	95%	6.65	0.25	1.66
Artificial Reefs	033	N/A	N/A	0.00
Fill to restore natural contours or improve water quality	5%	0.35	1.25	0.44
Total		7		2.10

<sup>29</sup> Under PDC A7.25, living shorelines and oyster reefs cannot be placed in waters containing the shallow, euryhaline essential feature, and, under PDC AP.12, the Opinion does not apply to any such projects requiring removal of red mangroves, so we do not expect these projects to be in areas where they will affect the essential features of smalltooth sawfish critical habitat.

<sup>30</sup> Under PDC A7.25, artificial reefs cannot be placed in waters containing the shallow, euryhaline essential feature, and, under PDC AP.12, the Opinion does not apply to any such projects requiring removal of red mangroves, so we do not expect these projects to be in areas where they will affect the essential features of smalltooth sawfish critical habitat.

<sup>31</sup> Under PDC A7.26, artificial reefs cannot be placed in Gulf sturgeon critical habitat.

<sup>32</sup> Under PDC A7.29, living shorelines and oyster reefs cannot be placed in waters shallower than -13 ft MHW within the geographic boundaries of Johnson's seagrass critical habitat, and we think waters deeper than -13 ft MHW do not support the essential features of critical habitat.

<sup>33</sup> Under PDC A7.29, artificial reefs cannot be placed in waters shallower than -13 ft MHW within the geographic boundaries of Johnson's seagrass critical habitat, and we think waters deeper than -13 ft MHW do not support the essential features of critical habitat.

#### Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that aquatic enhancement projects will have no effect on the essential features of smalltooth sawfish critical habitat. USACE anticipates that 46 aquatic enhancement activities meeting the requirements of this Opinion may be authorized per 5-year period in smalltooth sawfish critical habitat. Mangroves are 1 of the essential features of smalltooth sawfish critical habitat, and PDC AP.12 does not allow removal of mangroves for these projects. Shallow, euryhaline water is the other essential feature of smalltooth sawfish critical habitat. As explained above, only seagrass restoration and fill to restore natural contours can occur in shallow, euryhaline habitat (PDC A7.25) and none of the activities can remove mangroves (PDC AP.12). Combined, seagrass restoration and fill to restore natural contours are expected to result 13.81 ac of impacts. Although we expect the seagrass restoration projects to affect 10.93 ac of waterbottom, we do not expect seagrass restoration to have any effect on the shallow, euryhaline essential feature since restoring seagrasses does not affect the depth or salinity of the water. In addition, we do not expect the 2.88 ac of restoration fill projects to have an effect on the shallow, euryhaline water essential feature. Filling in areas to restore their natural contour or to improve water quality returns the area to its natural depth contour. This actually could decrease project area depths (i.e., filling in deeper areas reduces the water depth). Thus, filling in the areas would not increase the depth of the water beyond the -3 ft MLLW essential feature depth. Therefore, no effects to the essential features are expected from any of these aquatic enhancement activities.

#### **Potential Routes of Effect to Gulf Sturgeon Critical Habitat**

We believe that aquatic enhancement projects may affect, but are not likely to adversely affect Gulf sturgeon critical habitat. USACE anticipates that 91 aquatic enhancement activities meeting the requirements of this Opinion may be authorized per 5-year period in Gulf sturgeon critical habitat. As explained above, given the PDC limitations, we assume that 5% of these projects will be activities to fill in areas to the natural depth contour or to improve water quality and the remaining 95% will be some combination of living shoreline/oyster reefs and seagrass restoration (PDC A7.26 prohibits artificial reef projects within the boundaries of Gulf sturgeon critical habitat). These nearshore aquatic enhancement projects (living shorelines and oyster reefs, seagrass restoration, fill in areas to the natural depth contour or fill in areas to restore water quality with low water quality area) could impact 27.31 ac of waterbottom in Gulf sturgeon critical habitat over a 5-year period. Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

<u>Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages</u>. We believe that living shorelines and oyster reefs, seagrass restoration, and placing fill to return areas to the natural contour or to improve water quality may cover and bury bottom substrates containing sturgeon prey species; however, the effects to this PCE will be insignificant. Restoration activities that fill in areas to the natural depth contour or fill in areas to restore water quality with low water quality would not affect the prey abundance PCE of Gulf sturgeon critical habitat as the areas being filled would not support prey abundance. Thus, only 21.61 ac of impacts associated with living shorelines

and oyster reefs, and seagrass restoration could affect the prey abundance PCE. We believe the effects to this PCE will be insignificant since the estimated 21.61 ac of impact is very small compared to the approximately1.5 million ac of available marine and estuarine critical habitat that we believe support sturgeon prey species.

Further, not all of the 21.61 ac impacted will support prey items or serve as preferred foraging habitat. PDC A7.26 restricts living shorelines and oyster reefs as well as and seagrass restoration to areas where water depths are shallower than -6 ft (-2 m) MHW. Gulf sturgeon are suction feeders that tend to forage in calm marine and estuarine waters that support their macroinvertebrate prey including brachiopods, mollusks, worms, and crustaceans (Mason and Clugston 1993). During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002), beyond the depth limit placed on these activities. Thus, although these activities may occur within Gulf sturgeon critical habitat, not all such projects will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

In addition, installing living shorelines and nearshore oyster reefs may provide an indirect benefit to this PCE by enhancing the number and diversity of prey available to them. This may happen by the creation of a patchwork of oyster reefs that, over time, provide more diverse and structurally complex habitat for prey species (Boudreaux et al. 2006). In addition, as these prey species (e.g., macrofaunal species such as amphipods, polychaetes, gastropods, and bivalves) increase in abundance in the shallow nearshore project area, there will be a spill-over effect to neighboring areas that are deeper than 6 ft, where increased prey abundance will benefit Gulf sturgeon that may forage in those waters in the long-term. The use of oyster reefs as breakwaters to mitigate against coastal erosion also encourages nektonic production that could lead to greater prey availability in the immediate surroundings for Gulf sturgeon (Seitz et al. 2006).

2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Living shorelines and ovster reefs, seagrass restoration, and placing fill to return areas to the natural contour or to improve water quality may generate turbidity that could cause localized and temporary reductions in water quality; however, we believe the effects to this PCE will be insignificant. Restoration activities that fill in areas to the natural depth contour or fill in areas to restore water quality with low water quality would not affect the water quality PCE of Gulf sturgeon critical habitat as these areas would not support water quality. Thus, only 21.61 ac of impacts associated with living shorelines and oyster reefs, and seagrass restoration could affect the water quality PCE, and we expect effects to this PCE from turbidity to be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and thus we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement as described in PDC AP.10. These instances include projects that are so

small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high energy areas where turbidity will dissipate very quickly), and thus the effects from turbidity on water quality would be insignificant.

- 3. Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages. Living shorelines and oyster reefs, seagrass restoration, and placing fill to return areas to the natural contour or to improve water quality may cover and bury nearshore bottom substrates and affect sediment quality. Restoration activities that fill in areas to the natural depth contour or fill in areas to restore water quality with low water quality would not affect the sediment quality PCE of Gulf sturgeon critical habitat as these areas would not support sediment quality. Thus, only 21.61 ac associated with living shorelines and oyster reefs, and seagrass restoration could affect the sediment quality PCE and we expect effects to this PCE to be insignificant since the estimated 21.61 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we believe support sturgeon prey species. Further, not all of the 21.61 ac of habitat affected by living shorelines and oyster reefs, and seagrass restoration are expected to have the sediment quality needed to support Gulf sturgeon prey or serve as preferred foraging habitat. PDC A7.26 restricts living shorelines and ovster reefs as well as seagrass restoration to areas that are in water depths shallower than -6 ft (-2 m) MHW. During foraging periods, Gulf sturgeon generally occupy shoreline areas between 6.5-13 ft (2-4 m) of depth characterized by low-relief sand substrate (Fox et al. 2002), in areas deeper than these materials are allowed to be placed. Thus, although these activities may occur within Gulf sturgeon critical habitat, not all such projects will occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prev species.
- 4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats. Living shorelines and oyster reefs, seagrass restoration, and placing fill to return areas to the natural contour or to improve water quality could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by the placement of materials in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging; however, we believe there will be no effect to this PCE. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, PDC A7.26 limits activities in Gulf sturgeon critical habitat migratory restriction zones. We believe that the placement of living shorelines and oyster reefs placed parallel with the shore will not affect migration of sturgeon, as they will create no impediment to the species passing along the shore.

Additionally, the PDCs require that living shorelines and oyster reefs provide a break in the reef to allow for movement of species (A7.6). Seagrass restoration and restoring areas by filling holes to the natural depth contour would not create obstacles to migration. Therefore, we believe aquatic restoration activities will have no effect on the migratory pathway PCE.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and there will be no effect to safe and unobstructed migratory pathways, we believe that shoreline stabilization activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

#### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

As we explained above, we expect 5% of the projects occurring in areas supporting the essential features of Johnson's seagrass critical habitat to be fill projects to restore natural contours or improve water quality and 95% to be seagrass restoration. We believe there will be no effect from fill projects to restore natural contours or improve water quality on Johnson's seagrass critical habitat. We also believe that seagrass restoration may affect, but is not likely to adversely affect, Johnson's seagrass critical habitat.

As stated above, living shorelines, oyster reefs, and artificial reefs cannot be placed in waters shallower than -13 ft MHW within the geographic boundaries of Johnson's seagrass critical habitat (PDC A7.29). Studies show that Johnson's seagrass occurs in waters shallower than 10-13 ft (3-4 m) (NMFS 2007a). Water depths greater than 13 ft are not believed to provide the water transparency necessary for enough sunlight to reach the sea floor to support Johnson's seagrass growth. Therefore, living shorelines and oyster reefs, and artificial reefs will occur in areas that do not support 1 of the essential features of Johnson's seagrass growth), and when 1 of the features is affected, the area ceases to function as critical habitat. For this reason, in Table 50 above, we did not assume that any of these projects would affect Johnson's seagrass critical habitat.

Fill to restore natural contours or improve water quality is allowed in Johnson's seagrass critical habitat and could result in impacts to 0.44 ac of critical habitat (Table 51). These areas are typically either (1) filled during seagrass restoration to restore the appropriate depth to support seagrasses or (2) involve filling holes or depressions often resulting from dredging that result in reduced water quality. The deep holes often have an accumulation of nutrients and chemicals that are harmful to seagrasses in the area if the area of low water quality is stirred up and distributed from events like hurricanes. By filling these areas with clean fill, it prevents potential seagrass die offs in the area and can restore areas that may then be able to support future seagrass recruitment. The areas to be filled would not be considered critical habitat since they would be unable to support seagrasses and therefore lack the essential features of Johnson's seagrass, including water quality. We believe this activity would have no effect on Johnson's seagrass critical habitat.

Seagrass restoration is allowed within Johnson's seagrass critical habitat and could result in impacts to 1.66 ac of critical habitat (Table 51). The water transparency essential feature of Johnson's seagrass critical habitat may be affected from the temporary turbidity that may occur

during seagrass restoration and filling activities. We believe that this effect will be temporary (likely 1 day or 2) and that turbidity will settle out quickly. Thus, we believe the effect on the critical habitat from turbidity will be insignificant. Restoration of an area to support seagrasses, including filling blow holes and leveling sediments to the surrounding elevation, may have a beneficial effect on Johnson's seagrass by providing additional area for the species to recruit to These benefits are not certain, and depend on the proximity of the species to the restored area. Restoration of areas within Johnson's seagrass critical habitat will have an insignificant effect on the essential features of critical habitat since restoring areas to be able to support seagrasses will not preclude Johnson's seagrass from recruiting into the newly restored areas, and will not affect the functionality of critical habitat.

#### Potential Routes of Effects to North Atlantic Right Whale Critical Habitat

We believe that aquatic enhancement activities may affect, but are not likely to adversely affect North Atlantic right whale critical habitat. The placement of artificial reefs in North Atlantic right whale critical habitat has the potential to affect both North Atlantic right whales and the essential features of their critical habitat. PDC A7.27 requires that materials be limited to placement in waters deeper than 30 ft deep, that materials not extend more than 20 ft off the sea floor, and that the density of materials in an area be limited to no more than 2 reefs per 10 nmi<sup>2</sup> area. Also any new reefs shall be limited to 1 nmi<sup>2</sup> in total size (PDC A7.27.3). In addition, if a new reef is added to an existing artificial reef, the total footprint of the combined reefs must not exceed 1 nmi<sup>2</sup>. The water depth essential feature includes areas -20-92 ft deep that are "selected by right whale cows and calves in dynamic combinations that are suitable for calving, nursing, and rearing, and which vary, within the ranges specified, depending on factors such as weather and age of the calves." By limiting the vertical height of reef materials to 20 ft, we believe that this will not create an obstacle to whales. The PDCs also limit the density of material placement in an area to be no more than 2 reefs per 10 nmi<sup>2</sup> (PDC A7.27.4) so that whales can easily move over and around these reefs without dramatically changing the landscape in this habitat. Based on these restrictions, we believe that the change in depth and placement of materials will be so minimal that it will have insignificant effects to North Atlantic right whales using the area and an insignificant effect to the essential features of critical habitat.

## 2.2.8 Activity 8 (A8): Transmission and Utility Lines

#### **General Description**

This category of activity includes the installation, repair (including all forms of maintenance), replacement, and removal of transmission and utility lines including aerial and subaqueous lines. Aerial transmission or utility lines are typically placed over smaller water bodies with the support piles or structures positioned on the uplands. When crossing larger bodies of water, support structures are placed in the water at set intervals. Support structures can reach approximately 1,500 ft<sup>2</sup> in size; however, such large structures are rare. Each support is typically constructed quickly (e.g., less than a day for a pile to a few weeks for larger footings). The specific requirements for size and placement are described in the PDCs below.

Subaqueous transmission or utility lines are installed on top of or under the bottom sediment. The same methods used to install new subaqueous transmission and utility lines often are used during repair/replacement activities because during repair/replacement, existing buried lines often are left in place and a new line or section of a line is installed alongside or on top of the existing line.

The placement of subaqueous transmission or utility lines under the bottom sediments can be accomplished by temporarily trenching the location for the line, placing the line, and backfilling the trench using barge-mounted equipment. The trench in which the line is buried can be created by sidecasting materials temporarily outside of the trench and then refilling it over the buried line. Other equipment and methods are being developed to expedite this process including the use of a jet plow. This machine simultaneously jets a trench, lays the cable, and backfills it as moves forward along the line path. When repairing or replacing subaqueous utility or transmission lines, riprap or concrete materials may be used to stabilize the line. These materials may also be used to stabilize new lines placed in high erosion areas.

Another option to place subaqueous transmission or utility lines under the bottom sediments is to use horizontal directional drilling. This method restricts all construction equipment work to uplands or within an area that is dewatered and contained in a cofferdam. The drill is set and bores a hole under the waterbody. Once the drill reaches the other side of the waterbody, it is attached to the line or pipe to be installed, and the new line is pulled back through the drilled hole with the drill as it is removed. Typically, the entire fused length of new line is pulled at one time with no interruptions. The horizontal directional drilling process requires the use of drilling fluid/mud (i.e., bentonite) to act as a lubricant and sealant. The drilling fluid is composed of naturally occurring bentonite clay and water. The drilling mud pressure and volume are monitored during drilling operations to assure there are no leakages due to fractures in the structure of the material being drilled through. If a fracture is present, it is possible for drilling mud to escape onto the surface or into the water. This rare event is called a "frac-out." An example Frac-out Contingency Plan is located in Appendix D. Below is a sample image of how directional drilling is performed (Figure 22).



Figure 22. Sample image of horizontal directional drilling (Image from Underground Solutions at <u>http://www.undergroundsolutions.com</u>)

# Project Design Criteria

# PDCs specific to Activity 8 for Transmission and Utility Line Activities:

A8.1. Activity 8 includes the installation, repair, replacement, and removal of support structures, footers, foundations, as well as the placement of riprap or concrete mat for pipeline protection. The USACE defines a "utility/transmission line" as any pipe or pipeline for the transportation of any gaseous, liquid, liquescent, or slurry substance, for any purpose, and any cable, line, wire or optical fiber for the transmission for any purpose of electrical energy, telephone, telegraph messages, digital signal, Internet, and radio or television communication.

- A8.2. Structures permanently placed on the waterbottom (e.g., foundations, piles, and footings) to support aerial transmission lines must total less than a 0.5 ac for all structures combined. Because permanent structures have the potential to interfere with or impede sea turtles from entering or exiting the beach, they cannot be placed on or near beaches used for sea turtle nesting.
- A8.3. Subaqueous utility and transmission lines may be installed (including as part of a repair/replacement project) using horizontal directional drilling, if the drilling originates and terminates on the uplands (i.e., no in-water work). For subaqueous transmission lines installed, repaired, or replaced using horizontal directional drilling, the applicant must provide and follow a frac-out contingency plan in Appendix D or another plan with at a minimum the same level of information as is provided in the plan contained in Appendix D.
- A8.4. Subaqueous utility and transmission lines may be installed (including as part of a repair/replacement project) by trenching. When excavating the trench, the bottom sediments may be temporarily sidecast into areas devoid of submerged aquatic vegetation and mangroves. Immediately upon completing the excavation and placing the transmission or utility line into the trench, the trench must be filled and the bottom contours must be restored to pre-construction conditions. The District Engineer may allow the trench to remain open and temporary sidecasting to continue after the excavation is complete, as long as the total time the trench is open and the material is sidecast during and after excavation does not exceed 180 days.
- A8.5. New subaqueous transmission and utility lines shall not be placed on the sea floor (i.e., pinned or anchored and not buried) under this Opinion. Sections of existing buried lines may be repaired or replaced above the sea floor by pinning or anchoring the new section of line in place to ensure that it does not move and damage surrounding seagrasses, hardbottom, coral, or coral reef habitat.
- A8.6. When repairing existing transmission or utility lines, riprap and articulated mats may be placed on subaqueous lines that are buried in trenches or on lines that are attached to the sea floor (in accordance with A8.5) to stabilize the line. Riprap and articulated mats may also be used to stabilize new subaqueous lines placed in high erosion areas. These stabilization materials are limited to the minimum amount necessary to stabilize and protect the lines existing lines (which may have been exposed by scouring) and cannot be placed on seagrasses, hardbottom, corals, or coral reef habitat.

# Additional PDCs for Activity 8 applicable in critical habitat:

In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat, as described below.

- A8.7. *Acropora* critical habitat: This Opinion does not apply to the new installation of transmission and utility lines within the geographic boundary of *Acropora* critical habitat. This Opinion covers the repair and replacement of transmission and utility lines in *Acropora* critical habitat, but only if the essential feature is not present, and only if the placement meets the measures described in PDC A8.5 to limit movement of the lines.
- A8.8. Smalltooth sawfish critical habitat: Structures supporting aerial transmission or utility lines, such as foundation towers and transmission line poles, cannot be placed in smalltooth sawfish critical habitat in areas where the essential features are present. Transmission or utility line projects are not allowed in areas identified as smalltooth sawfish limited exclusion zones, as defined in Section 2.1.1.1, above.
- A8.9. Johnson's seagrass critical habitat: All newly installed subaqueous transmission or utility lines must be placed using horizontal directional drilling from the uplands. Repair and replacement of existing subaqueous lines, whether the existing lines are buried within trenches or placed on the sea floor outside of trenches, is allowed in the same footprint as the existing line. Structures supporting aerial transmission or utility lines, such as foundation towers and transmission line poles, cannot be placed in Johnson's seagrass critical habitat in waters shallower than -13 ft deep.
- A8.10. Gulf sturgeon: No new transmission and utility line activities installation are allowed in the Gulf sturgeon critical habitat migratory restriction zones (defined in Section 2.1.1.2) between September and March, when sturgeon are likely to be present in these areas. Repair/replacement activities may occur in Gulf sturgeon critical habitat migratory restriction zones at any time of year as long as the repair or replacement is accomplished without the use of heavy in-water equipment (i.e., if the repair or replacement does not require trenching). Additional noise restrictions are required for pile and sheet pile installation in the Gulf sturgeon critical habitat migratory restriction zones defined in Section 2.1.1.2.
- A8.11. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): Under this Opinion, the only transmission and utility line projects that can occur in U.S. Caribbean sea turtle critical habitat are repair and replacement projects

## Assumptions

The USACE anticipates that 288 transmission and utility line activities meeting the requirements of this Opinion may be authorized per 5-year period (Table 8). The USACE provided the following estimated impacts per project:

- Subaqueous transmission or utility lines installed via trenching will temporarily affect (by trenching and refilling) an area of waterbottom that is approximately 2 ft wide and 1,000 ft long (2,000 ft<sup>2</sup> [0.05 ac]).
- Installation of aerial transmission or utility lines (i.e., placement of piles and footers to support the lines) will cover an average of 0.5 ac of waterbottom per project.
- Repair or replacement of subaqueous transmission and utility lines could result in the loss of 200 ft<sup>2</sup> of Johnson's seagrass per 5-year period based on information provided by Florida Power and Light (FPL) utility company. PDC AP.13 states that the Opinion does not apply to projects where Johnson's seagrass is within the project footprint, but makes an exception for these repair/replacement projects.

To analyze the effects of transmission and utility line projects, we looked the assumptions regarding (1) the total number of covered activities to be authorized a 5-year period and (2) the average size of the impacted waterbottom based on the assumptions and PDCs; and (3) the average amount of Johnson's seagrass lost per project based on the assumptions and PDCs. Based on this information, we determined the potential effects to species (Table 52) and critical habitat (Table 55).

# Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing new or repairing/replacing transmission and utility lines would result in 5 of the common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, sturgeon, and Johnson's seagrass from the installation, repair, replacement, and removal of aerial and subaqueous transmission and utility lines, and associated structures that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 52. Sea turtles, smalltooth sawfish, sturgeon, and Johnson's seagrass may be affected by the placement of transmission or utility lines and their support structures. We quantified the potential extent of impacts based on (1) the assumptions, (2) the USACE

estimated impacts per project based on the type of transmission and utility line project (Table 53), and (3) the construction limitations defined by the PDCs.

Table 52. USACE and NMFS Determinations on the Effects of Transmission and UtilityLine Projects on ESA-listed Species Listed in Table 5

Listed Species	<b>USACE</b> Determination	NMFS Determination		
Sea Turtles	NLAA	NLAA		
Sawfish	NLAA	NLAA		
Sturgeon	NLAA	NLAA		
Johnson's seagrass	LAA	LAA		
Effects Determinations Explained in Section 2.2				
Corals	NLAA	N/A		
Whales	NE	NE		
Nassau Grouper	Not provided	NLAA		
NE= no effect: NLAA= may affect not likely to adversely affect: N/A= not applicable (if a				

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

 Table 53. Estimated Waterbottom Covered by Transmission and Utility Line Projects Per

 Project

Project Type	Area Impacted per Project		
Installing subaqueous transmission or utility lines (new installation or repair/replacement) via horizontal directional drilling	0		
Installing subaqueous transmission or utility lines via trenching (new installation or repair/replacement)	2 ft wide x 1,000 ft long = 2,000 ft <sup>2</sup> $[0.05 \text{ ac}]^*$		
Installing foundations, piles, and footings to support aerial transmission or utility lines	0.5 ac		
*Impacts are temporary, given PDC A8.4, which requires trenches to be backfilled to natural			

contours immediately, or, pending District Engineer permission, after 180 days.

We assume that most transmission and utility line projects (whether new installation or repair/replacement) will be installed by trenching; however, we do not know how many of the estimated 288 utility projects will be installed by trenching versus other methods. Since we do not know the exact number of each project type and installation method, we calculate a minimum and maximum amount of waterbottom that will be altered or covered. As shown in Table 53, the minimum amount of impacts to waterbottom are from installing using horizontal directional drilling that will have no in-water impacts (i.e., 0 ac of waterbottom impacted). The maximum amount of waterbottom that may be altered by transmission and utility line projects would be

from the installation of foundations, piles, and footings to support aerial transmission or utility lines, which cover 0.5 ac of waterbottom per project. If we assume that all 288 projects could affect 0.5 ac of waterbottom per project, we could expect a maximum of 144 ac of waterbottom impacted (288 projects x 0.5 ac per project).

Horizontal directional drilling could result in a waterbottom alteration that would be greater than just the impacts from drilling if the drilling process resulted in a frac-out. We assume the potential for a frac-out is extremely rare and do not expect a frac-out from any of the projects authorized under this Opinion. If horizontal directional drilling is used as the installation method, the applicant must provide and follow a frac-out contingency plan that not only will require that turbidity barriers be in place, but that will further require that the exposure of bentonite be limited through isolation of the event and removal of the material in a timely manner. Additionally, the plan must require the directional drilling to stop immediately and must require a clean-up crew to be activated immediately upon a frac-out occurring. The plan also must require notification and documentation with state and federal agencies to provide additional assurances that the event will be contained. Further, horizontal directional drilling is not allowed within the geographic boundary of Acropora critical habitat since this area has a higher potential for subsurface sediments to fracture. Based on the extremely low risk of a frac out and the additional protective measures required if this method is used, we believe the risk to both species and critical habitat is discountable and will not be discussed further in this section. Because we do not expect a frac out from the horizontal directional drilling, we will continue to assume that such methods will not affect the waterbottom.

#### Potential Routes of Effect to Sea Turtles, Smalltooth Sawfish, and Sturgeon

Over a 5-year period, between 0 and 144 ac of waterbottom may be temporarily altered by trenching or permanently covered by the installation, repair, replacement, and removal of aerial and subaqueous utility and transmission lines, and the associated structures. This impacts to waterbottom is unlikely to be concentrated as the array of individual projects covered under this Opinion for Activity 8 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida).

- <u>Sea turtles</u>: Sea turtles may be affected by the loss of up to 144 ac of waterbottom, which the species may have used for foraging. However, the effect on sea turtles resulting from the loss of foraging habitat from transmission and utility line activities is insignificant.
  - The affected areas might include seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes transmission and utility line projects from the Opinion where Johnson's seagrass is present within the project footprint and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
  - Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles. Therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
  - Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. However, PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect listed corals, and excludes projects if non-

listed corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.

Utility line activities may cover or remove areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition, the maximum area of impact (144 ac) is very small compared to the remaining large areas that support sea turtle prey species. Sea turtles can travel long distances to forage. The array of individual projects covered under this Opinion for Activity 8 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and the U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites.

Given the above, we believe the effect to sea turtles from the potential loss of foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the utility line activities under Activity 8 are not likely to adversely affect sea turtles.

Smalltooth sawfish: Smalltooth sawfish may be affected by the loss of up to 144 ac of waterbottom, which the species may have used for foraging and refuge. This estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. However, the effect on sawfish of the potential loss of foraging and refuge habitat is insignificant. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for forage and refuge; these areas have been designated as critical habitat for the species. Projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal is strictly limited by PDCs AP.12 to removal for other covered activities and to removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. PDC A8.8 also prohibits installation of foundation and footers to support aerial transmission or utility lines in smalltooth sawfish critical habitat if such installation would result in the loss of any of the essential features. The PDC A8.8 also prohibits activities in the limited exclusion zones. Therefore, the effect on juvenile sawfish of the temporary losses of small areas of shallow water forage and refuge habitat is expected to be so small as to be undetectable. The array of individual projects covered under this Opinion for Activity 8 will likely be separated both temporally (over a 5year period) and spatially (along the entire coast of Florida), and sawfish can seek forage and refuge in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that utility line activities under Activity 8 are not likely to adversely affect smalltooth sawfish.

• <u>Sturgeon:</u> Sturgeon may be affected by the loss of up to 144 ac of waterbottom, which the species may have used for foraging. However, the effect on sturgeon of the potential loss of foraging habitat is insignificant. Utility line projects may cover and bury bottom substrates containing sturgeon prey species, such as benthic worms and insects, as well as crustaceans and mollusks. However, sturgeon are opportunistic feeders that forage over large areas. The estimated area of impact is very small compared to the remaining large areas that support sturgeon prey species. The array of individual projects covered under this Opinion for Activity 8 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that the utility line activities under Activity 8 are not likely to adversely affect sturgeon.

#### **Potential Effects to Johnson's Seagrass**

We believe that the repair, replacement, or removal of transmission lines may affect, but is not likely to adversely affect, Johnson's seagrass. The PDCs for *Mangroves, Seagrasses, Corals and Hardbottom for All Projects* (PDC AP.13) exclude transmission and utility line projects from the Opinion if Johnson's seagrass is found within the project footprint, with an exception for projects repairing existing lines within the same footprint. Depending on the location of the repair required, Johnson's seagrass may be lost during the repair. In the last 5 years, FPL reported the loss of 200 ft<sup>2</sup> of Johnson's seagrass during utility line repairs and they believe that similar losses may be experienced per 5-year period. We believe this amount of loss is likely to adversely affect Johnson's seagrass and the effect is further evaluated in Section 5.

## Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects on smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat transmission and utility line activities. The estimated impacts on each critical habitat unit and NMFS and USACE effects determinations for this category of activity are summarized in Tables 54 and 55.

Table 54. Estimated Potential Impacts to Areas Supporting the Essential Features ofSmalltooth Sawfish, Gulf Sturgeon, and Johnson's Seagrass Critical Habitat fromTransmission and Utility Line Projects Per 5-Year Period.

	Estimated	<i></i>		
Potential Loss of Essential Feature	Impact Per Project (Table 53)	Smalltooth Sawfish Critical Habitat (n = 9)	Gulf Sturgeon Critical Habitat (n = 28)	Johnson's Seagrass Critical Habitat (n = 12)
Installing subaqueous transmission or utility lines (new installation or repair/replacement) via horizontal directional drilling	0	N/A PDC A8.3 requires that horizontal directional drilling proceed from the uplands and does not involve any in-water work.		
Installing foundations, piles, and footings to support aerial transmission or utility lines	0.5 ac	N/A PDC A8.8 does not allow the placement of structures supporting aerial transmission or utility lines, such as foundation towers and transmission line poles, where the essential features are present.	Maximum of 14 ac if all projects were installed in areas supporting the essential features using this method (28 projects x 0.5 ac per project = 14 ac)	N/A PDC A8.9 placement of structures supporting aerial transmission or utility lines, such as foundation towers and transmission line poles is limited to waters deeper than -13 ft deep.
Installing subaqueous transmission or utility lines via trenching (new installation or repair/replacement)	0.05 ac	Maximum of 0.45 ac if all projects were installed in shallow, euryhaline waters using this method (9 projects x 0.05 ac per project = 0.45 ac)	Maximum of 1.4 ac if all projects were installed in areas supporting the essential features using this method (28 projects x 0.05 ac per project = 1.4 ac)	N/A PDC A8.9 limits trenching to only repair and replacement within the same footprint of the existing line. Areas previously altered by installing the existing line no longer function as critical habitat since they lack the stable, unconsolidated sediments that are free from physical disturbance essential feature.

Designated Critical Habitat	USACE Determination	NMFS Determination		
Smalltooth sawfish critical habitat	NLAA	NLAA		
Gulf sturgeon critical habitat	LAA	NLAA		
Johnson's seagrass critical habitat	LAA	NLAA		
Effects Determinations Explained in Section 2.2				
Loggerhead critical habitat	NE	NE		
Acropora critical habitat	NE	NE		
North Atlantic right whale critical habitat	NE	NE		
Atlantic sturgeon critical habitat	Not provided	NE		
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2				

 Table 55. USACE and NMFS Determinations on the Effects of Transmission and Utility

 Line Projects (Activity 8) on Designated Critical Habitat

## Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that the installation, repair, or replacement of transmission lines may affect, but is not likely to adversely affect, smalltooth sawfish critical habitat. USACE anticipates that 9 transmission line activities may be covered under this Opinion per 5-year period in smalltooth sawfish critical habitat. The PDCs limit activities in smalltooth sawfish critical habitat (PDC AP.12) and prohibit activities in exclusion zones (PDC A8.8). In particular, projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal is strictly limited by PDC AP.12 to removal for other covered activities and to removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. PDC A8.8 also protects against permanent impacts to the shallow, euryhaline habitat essential feature in smalltooth sawfish critical habitat by excluding the installation of foundation and footer type structures supporting aerial utility or transmission lines in areas where the shallow, euryhaline essential features is present. PDC A8.3 requires that all transmission and utility lines installed by horizontal directional drilling do not involve any in-water work so the shallow, euryhaline essential feature would not be affected. Therefore, we believe there will be no effect to the essential features of smalltooth sawfish critical habitat from aerial transmission line projects or lines installed by horizontal directional drilling.

The shallow, euryhaline essential feature could be temporarily affected by installing new or replacing existing subaqueous lines via trenching. As calculated in Table 54, if all projects were installed in shallow, euryhaline waters using trenching, a maximum of 0.45 ac of the shallow, euryhaline habitat may be affected. However, the trenches for these projects are generally shallow and we do not believe the trenches remove or modify the depth feature. Moreover, the effects of trenching are temporary as the trenches must be refilled within 180 days and any sidecast material can only be left on the waterbottom for up to 180 days based on PDC A8.4.

#### Potential Routes of Effect to Gulf Sturgeon Critical Habitat

We believe that the installation, repair, or replacement of transmission lines may affect, but is not likely to adversely affect, Gulf sturgeon critical habitat. USACE anticipates that 28 transmission line activities may be covered under this Opinion per 5-year period in Gulf sturgeon critical habitat. Since utility line projects in Gulf sturgeon critical habitat can be installed, repaired, or replaced by any of the methods discussed for Activity 8, the potential areas with the essential features could range from none if all of the projects relied upon horizontal directional drilling to up to 14 ac if all projects involved the installation of foundations, piles, and footings to support aerial transmission or utility lines (Table 54). Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

<u>Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages</u>. Permanent structures associated with utility lines (e.g., footings and foundations) may cover and bury bottom substrates containing sturgeon prey species; however, the effects to this PCE will be insignificant. Assuming all projects expected in Gulf sturgeon critical habitat involved aerial transmission lines, 14 ac of Gulf sturgeon critical habitat would be permanently covered or buried</u>. The estimated 14 ac of loss is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat that we believe support sturgeon prey species. Further, not all of the 14 ac of habitat lost will support prey items or serve as preferred foraging habitat.

If we assume all projects involve subaqueous lines requiring trenching, a total of 1.4 ac of waterbottom will be temporarily disturbed. Trenching up to 0.05 ac of habitat per project (28 projects total) for transmission or utility line placement will temporality displace prey in the linear trench where the line will be placed; however that area will be restored with the same bottom sediments within 180 days (PDC A8.4). Because similar habitat is expected to be present pre- and post-trenching, it is anticipated that the benthic biota in the areas will have the ability to recover and re-colonize. We expect that benthic prey availability will recover in 3-24 months (Culter and Mahadevan 1982; Wilber et al. 2007). Thus, we believe trenching will have an insignificant effect on this PCE.

2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from the installation, repair, or replacement of lines and associated structures; however, the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for *In-Water Activities* requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement as described in PDC AP.10. These instances include when

projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high energy areas where turbidity will dissipate very quickly), we believe effects from turbidity on water quality would be insignificant.

3. <u>Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages</u>. Permanent structures associated with aerial transmission or utility lines (e.g., footings and foundations) can affect sediment quality; however, the effects to this PCE will be insignificant. The placement of materials on the waterbottom converts sandy substrate capable of supporting Gulf sturgeon prey to hard manmade materials that do not support prey species. However, the estimated maximum of 14 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we believe support sturgeon prey species and Gulf sturgeon foraging.

Trenching up to 1.4 ac of habitat for transmission or utility line placement will temporality affect sediment quality in the area where the line will be placed; however this area will be restored with the same bottom sediments within 180 days (PDC A8.4). Because similar habitat is expected to be present pre- and post-trenching, it is anticipated that the benthic biota in the dredging areas will have the ability to recover and re-colonize. We expect that benthic prey availability will recover in 3-24 months (Culter and Mahadevan 1982; Wilber et al. 2007). Thus, we believe trenching will have an insignificant effect on this PCE.

4. Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats. Transmission or utility line activities could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by permanent structures (e.g., footings and foundations for aerial transmission line) in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging; however, we believe there will be no effect to this PCE. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from entering or exiting a spawning river, PDC A8.10 states that transmission or utility line projects cannot occur in the Gulf sturgeon critical habitat migratory restriction zones. Temporary trenching of transmission or utility lines and installation by horizontal directional drilling will not create obstacles to migration. Permanent structures built to support aerial transmission lines are typically spaced along the length of the aerial transmission and utility line to support it meaning that it does not create a barrier to species movement.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and there will be no effects to safe and unobstructed migratory pathways, we

believe that transmission or utility activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

#### Potential Routes of Effect to Johnson's Seagrass Critical Habitat

We believe that the installation, repair, or replacement of transmission or utility lines may affect, but is not likely to adversely affect, Johnson's seagrass critical habitat. USACE anticipates that 12 transmission line activities meeting the requirements of this Opinion may be authorized per 5-year period in Johnson's seagrass critical habitat. As shown in Table 54, none of the methods allowed under this Opinion for the installation, repair, or replacement of transmission and utility lines will affect the essential features of Johnson's seagrass critical habitat. Specifically:

- Transmission and utility lines installed by horizontal directional drilling requires that all drilling proceed from the uplands and does not involve any in-water work and therefore would not affect any of the essential features (PDC A8.3).
- Placement of structures supporting aerial transmission or utility lines, such as foundation towers and transmission line poles are limited to waters deeper than -13 ft deep (PDC A8.9). Studies show that Johnson's seagrass occurs in waters shallower than 10-13 ft (3-4 m) (NMFS 2007a). Water depths greater than 13 ft are not believed to provide the water transparency necessary for enough sunlight to reach the sea floor to support Johnson's seagrass growth. Therefore, these waters do not support the water transparency essential feature of Johnson's seagrass critical habitat, and the loss of 1 of the essential features results in a total loss in the conservation function of the critical habitat. Therefore, these structures will be placed in waters that do not function as critical habitat and will not affect the essential features of critical habitat.
- Transmission and utility line repair, or replacement involving trenching would be limited to only repair and replacement within the same footprint of the existing line (PDC A8.9). No new installation of transmission and utility lines is allowed by trenching in Johnson's' seagrass critical habitat. Areas previously altered by installing the existing line no longer function as critical habitat since they lack the stable, unconsolidated sediments that are free from physical disturbance essential feature. Therefore repairs occurring in these areas would have no effect to the essential features since the area already does not function as critical habitat.

#### 2.2.9 Activity 9 (A9): Marine Debris Removal

#### **General Description**

The need for marine debris removal is typically identified by divers, snorkelers, or aerial surveys. The debris to be removed can come from a multitude of sources and events, including storm damage, lost fishing gear, or illegal dumping. Removal is often done as part of a clean-up event in which a large number of divers and boats work cooperatively with an organization to clean-up a given area. For instance, in Florida, the Florida Fish and Wildlife Research Institute has 2 programs to remove lost and abandoned traps including "the spiny lobster, stone crab, and blue crab trap retrieval program" and the "derelict trap and trap debris removal program." The term derelict trap refers to traps left in the water following the date by which they must be

removed under the pertinent law. During clean-up, debris is typically collected by divers in dive bags. The bags can be tied to a rope or chain to be pulled back to the supporting vessel by hand or by winch. This Opinion covers marine debris removal by the methods and in the circumstances described in the PDCs below.

# Project Design Criteria

## PDCs specific to Activity 9 for Marine Debris Removal:

- A9.1. In-water activities are limited to the removal of marine debris that poses a threat to human health and safety and/or aquatic natural resources (flora, fauna, and their habitats) such as, but not limited to, large fishing nets, cables, crab traps, and derelict vessels.
- A9.2. Removal of marine debris shall require visual confirmation (e.g., divers, swimmers, and camera) that the item can be removed without causing further damage to aquatic natural resources.
- A9.3. If an item cannot be removed without causing harm to surrounding coral (ESA listed or non-listed), the item will be disassembled as much as practicable so that it no longer can accidentally harm or trap species.
- A9.4. Monofilament debris will be carefully cut loose from coral (ESA listed or non-listed) so as not to cause further harm. Under no circumstance will line be pulled through coral since this could cause breakage of coral.
- A9.5. Marine debris shall be lifted straight up and not be dragged through seagrass beds, coral reefs, coral, or hard bottom habitats. Trawling also cannot be used as a means of marine debris removal. Debris shall be properly disposed of in appropriate facilities in accordance with applicable federal and state requirements.
- A9.6. An absorbent blanket or boom shall be immediately deployed on the surface of the water around any derelict vessel to be removed if fuel, oil, or other free-floating pollutants are observed during the work.

## Assumptions

The USACE believes that there will be no effect any species listed in Table 5 or critical habitat listed in Table 6 from the removal of marine debris given the PDCs. USACE anticipates that 54 marine debris removal activities meeting the requirements of this Opinion may be authorized per 5-year period (Table 8). Of those, they estimate that 11 projects will occur in smalltooth sawfish critical habitat, 7 in Gulf sturgeon critical habitat, and 11 in Johnson's seagrass critical habitat. Thus, to analyze the effects of marine debris removal, we looked the USACE's assumptions the number of projects to be authorized per 5-year period and the amount of waterbottom each project will affect. Based on this information, we determined the potential effects to species (Table 56) and critical habitat (Table 57).

## Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Removal of marine debris would result in 3 of the common routes of effect discussed above\_in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The effects analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction

In this section of the Opinion, we evaluate the potential route of effects to sea turtles, smalltooth sawfish, and sturgeon from marine debris removal activities that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 56. In the previous categories of activities, we considered the amount of waterbottom that would be covered or buried by the placement of materials and calculated the amount of waterbottom that would be removed in a table. Since this activity is limited to carefully removing marine debris from the waterbottom, we do not expect this activity to cover any waterbottom, but rather to expose and clean up the waterbottom.

Listed Species	<b>USACE</b> Determination	<b>NMFS</b> Determination			
Sea Turtles	NE	NLAA			
Sawfish	NE	NLAA			
Sturgeon	NE	NLAA			
Effects Determinations Explained in Section 2.2					
Johnson's seagrass	NE	NE			
Corals	NE	N/A			
Whales	NE	NE			
Nassau Grouper	NE	NLAA			

 Table 56. USACE and NMFS Determinations on the Effects of Marine Debris Removal

 Activities (Activity 9) to ESA-listed Species listed in Table 5

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

#### Potential Routes of Effect to Sea Turtles, Smalltooth Sawfish, Sturgeon

We do not believe that the removal of marine debris will affect sea turtles, smalltooth sawfish, or sturgeon by affecting the species directly or by affecting their foraging or sheltering habitat. The removal of marine debris will be visually monitored and conducted so as to avoid harm to the species and surrounding environment during removal. Therefore, no effects are anticipated to

any of these species or their foraging or refuge habitat from the removal activity itself. In addition, the removal of these materials may allow habitat used by these species to recover and eventually provide foraging or refuge habitat, which could have a beneficial effect on the species.

Given the above, we believe there will be no effect to sea turtles, smalltooth sawfish, or sturgeon. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that marine debris removal activities under Activity 9 are not likely to adversely affect sea turtles.

#### Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from marine debris removal activities. NMFS and USACE effects determinations for this category of activity are summarized in Table 57. The USACE estimated 54 marine debris projects may be authorized per 5-year period. A breakdown of how many of these projects will occur within specific critical habitat unit is provided in Table 8.

Project Location	USACE Determination	NMFS Determination
Sawfish critical habitat	NE	NE
Gulf sturgeon critical habitat	NE	NE
Johnson's seagrass critical habitat	NE	NE
Effects Determinations Explained in Section 2.2		
Loggerhead critical habitat	NE	NE
Acropora critical habitat	NE	NE
North Atlantic right whale critical habitat	NE	NE
Atlantic sturgeon critical habitat	Not provided	NE
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2		

 Table 57. USACE and NMFS Determinations on the Effects of Marine Debris Removal (Activity 9) to Designated Critical Habitats

# Potential Routes of Effect to Critical Habitat (Smalltooth Sawfish, Gulf Sturgeon, or Johnson's Seagrass)

USACE anticipates that a maximum of 29 marine debris removal activities will occur in these smalltooth sawfish (11), Gulf sturgeon (7), or Johnson's seagrass (11) critical habitat units per 5-year period. The PDCs state that marine debris will be removed in a manner that does not cause it to be dragged or swung into any surrounding resources during removal (PDC A9.5). Divers, swimmers, or cameras will be used to assure the item(s) can be removed safely without causing further damage to aquatic resources (A9.2). The removal of debris may result in localized turbidity that is expected to be extremely small and temporary and not affect the water quality

essential features/PCEs of Johnson's seagrass or Gulf sturgeon critical habitat. In addition, the removal of debris could potentially result in beneficial effects to critical habitat by restoring the essential features such as exposing appropriate sediments for Johnson's seagrass (stable, unconsolidated sediments that are free from physical disturbance). With respect to smalltooth sawfish critical habitat, we do not expect removal of marine debris to change or modify the shallow, euryhaline essential feature, nor will such activity remove or affect red mangroves (PDC AP.12). Therefore, we believe there will be no effect to any of these designated critical habitats from the removal of marine debris.

## 2.2.10 Activity 10 (A10): Temporary Platforms, Fill, and Cofferdams

#### **General Description**

This activity category is limited to the (1) placement of temporary platforms and fill, (2) installation of pile jackets around piles to protect them (e.g., cathodic protection used for bridge supports), and (3) installation of cofferdams to dewater an area for construction. All of these activities are typically associated with construction of linear transportation projects and bridges. Cofferdams may also be used for other types of projects that require containment or a dewatered space to complete construction.

<u>Temporary platforms and fills</u>: Temporary platforms and fills may be required for new construction and to support bridge and causeway maintenance activities. Constructing temporary platforms and fills typically requires the use of barges, cranes, pumps, boats, front-end loaders, and track hoes.

Temporary platforms include space-frame structures (i.e., truss-like, lightweight, rigid structures constructed from interlocking struts in a geometric pattern) that provide high capacity working surfaces that are capable of spanning large decks or traversing the length of a bridge. Other temporary platforms include pontoons and work trestles, the latter of which are rigid frames used as support, especially referring to a bridge composed of a number of short spans supported by such frames.

Temporary fill includes roads of fill created in the waterbody near the bridge to transport equipment and materials or temporary islands of fill created to support equipment. Temporary fills covered by the Opinion must be removed in their entirety and the affected areas must be returned to pre-construction elevations.

<u>Pile Jacket Construction</u>: Pile jackets are a material or sleeve applied around a pile as protection. Types of equipment involved in pile jacket construction typically include barges, cranes, pumps, boats, etc. The equipment will be trucked, self-propelled, or barged to the site. Turbidity curtains, silt fences, sand bags, synthetic bales, or some combination of these items are used as directed by the project engineer to maintain State Water Quality Standards. Strict adherence to Section 104 of the Florida Department of Transportation Standard Specifications for Road and Bridge Construction is required to provide reasonable assurance that water quality standards will not be violated. Pile jackets typically include cathodic protection, cathodic protection with structural protection, and structural support jackets, as described below.

- Cathodic Protection Pile Jackets (see Figure 23) Cathodic protection is a technique used to control the corrosion of a metal surface by making it the cathode of an electrochemical cell. This pile jacket type provides galvanic cathodic protection to a pile to control corrosion but does not provide any additional structural strength to the pile. The jackets are a fiberglass form with pre-installed zinc mesh. The bottom of the pile jacket is always placed in the water, typically -6 in MLW with an anode installed below the jacket on a galvanized steel strap. The jacket contains negative and positive connection wires that are connected to the existing pile, the anode, zinc mesh and then to a terminal box. The jacket is then filled with an epoxy grout.
- 2. Structural Cathodic-Protection Pile Jackets (See Figure 24) These are the same type of system as the cathodic jacket but it also provides structural strengthening. The jacket is made wider to accommodate the new reinforcing steel and is filled with concrete.
- 3. Structural-Only Jackets These are purely for structural strengthening and do not provide cathodic protection. The jackets are not used in an environment where corrosion related damage can occur.

<u>Cofferdams</u>: Temporary metal or concrete boxes placed in the water to allow dewatering so construction can be completed in-the-dry inside the cofferdam. Cofferdams can be installed either by a vibratory hammer or via impact hammer; however, to be covered under this Opinion, cofferdams must be installed by vibratory method. This Opinion does not cover installation using an impact hammer.



Figure 23. Cathodic pile jacket (Image provided by USACE)



Figure 24. Structural cathodic pile jacket (Image provided by USACE)

## Project Design Criteria

# PDCs Specific to Activity 10 for Temporary Platform, Fill, and Cofferdam Activities:

- A10.1. This Opinion covers the installation, repair, replacement, and removal of pile jackets and cathodic protection; temporary platforms; temporary fill; and temporary cofferdams.
- A10.2. Temporary platforms and fills are limited to a total of 0.5 ac of clean fill per project; For platforms and fills, "temporary" is defined as fill that is in place for 120 days or less.
- A10.3. Before placing the temporary platform or fill, the applicant must place a geotextile barrier around the entire area to be filled to ensure that the fill will be removed completely at the end of construction.
- A10.4. Temporary fill materials must be placed in a manner that they will not be eroded by high water flows. Temporary fills and geotextile barriers must be removed in their entirety and the affected areas must be returned to pre-construction elevations.
- A10.5. Temporary platforms and fill placed in confined areas, such as a channel or river,

must be placed so that they do not impede normal downstream flows or species movement in the area.

- A10.6. Temporary cofferdams cannot be installed by impact hammer. For cofferdam installation, "temporary" is defined as cofferdams that are in place for 120 days or less.
- A10.7. Projects under Activity 10 are not allowed on or contiguous to ocean beaches that may be used by nesting sea turtles.

A10.8. The applicant must ensure that ESA-listed sea turtles and fish are not located in the cofferdam prior to dewatering the cofferdam to avoid entrapment.

# Additional PDCs for Activity 10 applicable in critical habitat:

In addition to the PDCs above, the project must be designed to meet the following PDCs if the project occurs in the critical habitat, as described below.

- A10.9. Projects under Activity 10 are not allowed in Johnson's seagrass critical habitat or *Acropora* critical habitat where the essential feature(s) of these critical habitats are present.
- A10.10. Smalltooth sawfish critical habitat: Projects under Activity 10 cannot occur in the smalltooth sawfish limited exclusion zones, defined in Section 2.1.1.1.
- A10.11. Gulf sturgeon critical habitat: No temporary platforms, fill, or cofferdam projects are allowed in the areas defined as Gulf sturgeon critical habitat migratory restriction zones, defined in Section 2.1.1.2.
- A10.12. U.S. Caribbean sea turtle critical habitat (hawksbill, leatherback, and the NA DPS of green sea turtle critical habitat): Only temporary cofferdams needed to complete other activities covered under this Opinion can occur in U.S. Caribbean sea turtle critical habitat. Cofferdams are not allowed on or contiguous to ocean beaches that may be used by nesting sea turtles.

## Assumptions

The USACE believes that the average pile jack installation, temporary platform, temporary fill, and temporary cofferdam project will cover or bury the following amount of waterbottom:

- Pile jacket installation: no additional waterbottom area will be covered or buried since these structures are placed on existing piles.
- Temporary fill: 0.5 ac of waterbottom area will be temporarily covered during construction.
- Temporary platforms: The piles supporting the temporary platforms will result in an average impact of 10 ft<sup>2</sup> (10 piles, each 1 ft<sup>2</sup>).
- Cofferdam: 250 ft<sup>2</sup> of waterbottom area will be temporarily covered during construction.

The USACE believes that they permit a ratio of 2 cofferdams to every 1 temporary platform or temporary fill. USACE anticipates that it will authorize 116 temporary platforms, temporary access fill, and temporary cofferdams meeting the requirements of the Opinion per 5-year period. The USACE was unable to estimate the number of pile jacket activities it expected to authorize. Since pile jackets are installed on existing piles and do not come in contact with the sea floor or any benthic resources and do not affect species using the area, we believe there will be no effect from pile jacket installation.

To analyze the effects of the other types of projects covered under Activity 10 (temporary fill, temporary platforms, and cofferdams), we looked the USACE's assumptions regarding the total number of covered activities per type to be authorized a 5-year period and the estimated size of the area affected. Based on this information, we determined the potential effects to species (Table 58) and critical habitat (Table 60).

# Potential Routes of Effect to Listed Species

In Section 2.2 above, we evaluated routes of effect common to the 10 categories of activities covered under this Opinion on ESA-listed species identified in Table 5. Installing temporary platforms, fill, and cofferdams would result in the following common routes of effect discussed above in the section entitled *Construction Related Effects for All Categories of Activities Analyzed under this Opinion* (the numbers below correspond to the section numbers). The analysis for each of these routes of effects is provided in those sections:

- 1. Direct Physical Effects from Construction Activities
- 2. Turbidity
- 3. Potential Entanglement in Construction Materials
- 4. Exclusion from Areas during Construction
- 5. Limiting Species' Movement and Access to Habitat
- 7. Noise

In this section of the Opinion, we evaluate the potential effects to sea turtles, smalltooth sawfish, and sturgeon from the installation and removal of temporary platforms, fill, and cofferdams
activities that were not considered in Section 2.2. NMFS and USACE effects determinations are summarized for this category of activity in Table 58.

In the previous categories of activities, we considered if placement of materials or activities could permanently remove, cover, or bury waterbottom and calculated the amount of waterbottom so affected in a table. For this category of activities, sea turtles, smalltooth sawfish, and sturgeon also may be affected by the temporary inability to use the area covered by the temporary platforms or temporary fill (limited to 120 days by PDC A10.2), or surrounded by the temporary cofferdam (PDC A10.6). We quantified the potential extent of impacts based on (1) the assumptions, (2) the number and estimated amount of waterbottom covered by each type of project based on estimates from the USACE (Table 59), and (3) the construction limitations defined by the PDCs.

 Table 58. USACE and NMFS Determinations on the Effects of Temporary Platforms, Fill, and Cofferdam Activities (Activity 10) on Sea Turtles, Smalltooth Sawfish, and Sturgeon

Listed Species	USACE Determination	NMFS Determination				
Sea Turtles	NLAA	NLAA				
Sawfish	NLAA	NLAA				
Sturgeon	NLAA	NLAA				
Effects Determinations Explained in Section 2.2						
Johnson's seagrass	NLAA	NE				
Corals	NE	N/A				
Whales	NE	NE				
Nassau grouper	Not provided	NLAA				

NE= no effect; NLAA= may affect, not likely to adversely affect; N/A= not applicable (if a project may affect ESA-listed corals, separate consultation is required).

The effects analysis for Johnson's seagrass, corals, whales, and Nassau Grouper was provided at the beginning of Section 2.2.

 Table 59. Estimated Amount of Waterbottom Affected by Temporary Platforms, Fill, and

 Cofferdam Activities in Florida and the U.S. Caribbean per 5-Year Period

Project Type (temporary)	Ratio of Projects	Number of Projects (n = 116)	Estimated Waterbottom Affected per Project (ft <sup>2</sup> )	Estimated Waterbottom Affected Per 5- Year Period (ft <sup>2</sup> )	Estimated Waterbottom Affected Per 5- Year Period (ac)
Platforms	1	19.33	10	193.33	0.00 <sup>34</sup>
Fill	1	18.33	21,780	399,300.00	9.17
Cofferdams	2	77.33	250	19,333.33	0.44
Total		116		418,826.67	9.61

- Sea turtles: Sea turtles may be affected by the temporary inability to access 9.61 ac of waterbottom area, which could be used as foraging habitat. The activities will temporarily cover foraging habitat for up to 120 days (PDC A10.2, A10.6) and areas affected by temporary platforms and fill would be returned to reconstruction elevations (PDC A10.4). However, the effect on sea turtles from the temporary loss of foraging habitat is insignificant.
  - The waterbottom affected by Activity 10 could contain seagrasses, which are an important forage resource for green sea turtles. However, PDC AP.13 excludes projects from the Opinion that affected ESA-listed Johnson's seagrass and recommends that impacts to native, non-listed seagrasses be avoided and minimized to the extent practicable.
  - Limestone outcroppings and worm-rock reefs are important developmental habitat for juvenile green turtles, therefore, under PDC AP.14, this Opinion does not apply to projects where hardbottom is found within the footprint.
  - Hawksbill sea turtles are most commonly associated with reef habitat and feed on sponges, algae, and other invertebrates. PDC AP.14 also limits this Opinion to projects that do not directly or indirectly affect listed corals, and excludes projects if non-listed corals and hardbottom habitats, which support sponges, algae, and other forage resources for hawksbill sea turtles, are within the project footprint.
  - Temporary platforms, fill, and cofferdams activates may cover or remove areas inhabited by sea turtle prey species, including the crustaceans and mollusks that serve as prey for loggerhead and the fish, jellyfish, shrimp, and mollusks that serve as prey for Kemp's ridley sea turtles. These foraging areas are larger and more common throughout Florida and the U.S. Caribbean than the specific habitat types like seagrass beds, hard bottom, limestone outcroppings, and reefs that must be avoided under this Opinion. In addition,

 $<sup>^{34}</sup>$  193.33 ft<sup>2</sup> is approximately 0.004 ac, which rounded to 2 decimal places is 0.00 ac.

impacts to 9.61 ac is very small compared to the remaining large areas that support sea turtle prey species and we would expect these resources to return and recolonize in the areas covered by temporary platforms and temporary fill post construction. Sea turtles can travel long distances to forage. The array of individual projects covered under this Opinion for Activity 10 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida and U.S. Caribbean), and sea turtles can forage in nearby areas outside of active project sites.

Given the above, effect to sea turtles from the potential loss of foraging habitat is insignificant. In total, based on the routes of effect analyzed here and in Section 2.2, we determined that the temporary platform, fill, and cofferdam activities under Activity 10 are not likely to adversely affect sea turtles.

- Smalltooth sawfish: Smalltooth sawfish may be affected by the temporary inability to access 9.61 ac of waterbottom area, which may be used a foraging and refuge habitat. This estimate includes projects in Florida and the U.S. Caribbean, but sawfish would only be affected by projects occurring in Florida. For the first several years of their lives, juvenile smalltooth sawfish exhibit site fidelity to the areas in which they are pupped, typically in very shallow, nearshore waters where they can avoid predation by coastal shark species. In South Florida, sawfish have established distinct nursery areas where they utilize shallow, euryhaline habitat and red mangroves for foraging and refuge; these areas have been designated as critical habitat for the species. However, the effect on sawfish from the potential loss of nearshore foraging and refuge habitat is insignificant. The PDCs prohibit activities in smalltooth sawfish limited exclusion zones (PDC A10.10), which are areas that research shows support higher levels of smalltooth sawfish pupping. Additionally, projects must be sited to avoid and minimize impacts to mangroves, and mangrove removal is not covered under this Opinion for this activity by PDCs AP.12. The only mangrove removal allowed that could be used in connection with this activity is removal above the MHWL provided that red mangrove prop roots that are accessible to marine species are not removed. Temporary platform, fill, and cofferdam activities may also cover or remove nearshore areas inhabited by fish and crustaceans that serve as prey for smalltooth sawfish. The area of impact (some amount less than 9.61 ac) is very small compared to the remaining large nearshore areas that support sawfish prey species, and will be temporary (120 days or less). We expect these resources to return and recolonize. In the meantime, sawfish can travel long distances to forage. The array of individual projects covered under this Opinion for Activity 10 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sawfish can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that that the temporary platform, fill, and cofferdam activities under Activity 10 are not likely to adversely affect smalltooth sawfish.
- Sturgeon: Sturgeon may be affected by the temporary loss of 9.61 ac of waterbottom, which may be used for foraging. This estimate includes projects in Florida and the U.S. Caribbean, but sturgeon would only be affected by projects occurring in Florida. Projects under Activity 10 may temporarily cover and bury substrates containing sturgeon prey species, such as

benthic worms and insects, as well as crustaceans and mollusks. However, the effect on sturgeon of the potential loss of foraging habitat is insignificant. These effects are primarily short-term in nature, consisting of a temporary loss of benthic invertebrate populations in the project area. Observed rates of benthic community recovery after dredging, range from 3-24 months (Culter and Mahadevan 1982; Saloman et al. 1982; Wilber et al. 2007). The relatively species-poor benthic assemblages associated with low salinity estuarine sediments can recover in periods of time ranging from a few months to approximately 1 year, while the more diverse communities of high salinity estuarine sediments may require a year or longer. Hence, we believe that these areas will ultimately recover after the platforms, fill, or cofferdams are removed. Also, sturgeon are opportunistic feeders that forage over large areas and the area of impact (some amount less than 9.61 ac) is small compared to the remaining areas that support sturgeon prey species. The array of individual projects covered under this Opinion for Activity 10 will likely be separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida), and sturgeon can forage in nearby areas outside of active project sites. Based on the routes of effect analyzed here and in Section 2.2, we determined that the temporary platform, fill, and cofferdam activities under Activity 10 are not likely to adversely affect sturgeon.

# Potential Routes of Effect to Critical Habitat

In this section of the Opinion, we evaluate the potential effects to smalltooth sawfish critical habitat, Gulf sturgeon critical habitat, and Johnson's seagrass critical habitat from the installation and removal of temporary platforms, fill, and cofferdams activities. The estimated temporary exclusion of water areas in each critical habitat unit and NMFS and USACE effects determinations for this category of activity are summarized in Table 60 and the calculated amount of waterbottom affected by these projects is calculated in Table 61.

	USACE	NIMES			
Designated Critical Habitat	Determination	<b>Determination</b>			
Smalltooth sawfish critical habitat	LAA	NLAA			
Gulf sturgeon critical habitat	LAA	NLAA			
Johnson's seagrass critical habitat	NE	NE			
Effects Determinations Explained in Section 2.2					
Loggerhead critical habitat	NE	NE			
Acropora critical habitat	NE	NE			
North Atlantic right whale critical habitat NE NE					
Atlantic sturgeon critical habitat	NE				
The effects analysis for loggerhead, <i>Acropora</i> , North Atlantic right whale, and Atlantic sturgeon critical habitat was provided at the beginning of Section 2.2.					

 Table 60. USACE and NMFS Determinations on the Effects of Temporary Platform, Fill,

 and Cofferdam Activities under (Activity 10) to Designated Critical Habitat

Table 61. Estimated Amount of Waterbottom Affected by Temporary Platform, Fill, and Cofferdam Activities under (Activity 10) in Smalltooth Sawfish, Gulf sturgeon, and Johnson's Seagrass Critical Habitat per 5-Year Period

Project Type (Temporary)	Ratio of Projects	Number of Projects	Estimated Waterbottom Affected per Project (ft <sup>2</sup> )	Estimated Waterbottom Affected Per 5- Year Period (ft <sup>2</sup> )	Estimated Waterbottom Affected Per 5- Year Period (ac)	
Smalltooth sav	vfish critica	l habitat (n	= 31)			
Platforms	1	5.17	10	51.67	0.00 <sup>35</sup>	
Fill	1	18.33	21,780	399,300.00	9.17	
Cofferdams	2	20.67	250	5,166.67	0.12	
Total	1	31		404,518.33	9.29	
Gulf sturgeon	critical hab	itat (n = 12)				
Platforms		2.00	10	20.00	0.00 <sup>36</sup>	
Fill	1	2.00	21,780	43,560.00	1.00	
Cofferdams	2	8.00	250	2,000.00	0.05	
Total		12		45,580.00	1.05	
Johnson's seag	rass critical	l habitat (n	= 12)			
Platforms		2.00	10	20.00	0.00 <sup>37</sup>	
Fill	1	2.00	21,780	43,560.00	1.00	
Cofferdams	2	8.00	250	2,000.00	0.05	
Total		12		45,580.00	1.05	

 $<sup>^{35}</sup>$  51.67 ft<sup>2</sup> is approximately 0.001 ac, which rounded to 2 decimal places is 0.00 ac.  $^{36}$  20 ft<sup>2</sup> is approximately 0.0005 ac, which rounded to 2 decimal places is 0.00 ac.  $^{37}$  20 ft<sup>2</sup> is approximately 0.0005 ac, which rounded to 2 decimal places is 0.00 ac.

# Potential Routes of Effect to Smalltooth Sawfish Critical Habitat

We believe that temporary platforms, fill, and cofferdam activities may affect, but are not likely to adversely affect, smalltooth sawfish critical habitat. USACE anticipates 31 temporary platforms, fill, and cofferdam activities meeting the requirements of this Opinion may be authorized per 5-year period in smalltooth sawfish critical habitat.

In Table 61, we determined that these activities will result in 9.29 ac of temporary effects to waterbottom areas in smalltooth sawfish critical habitat. Mangroves are 1 of the essential features of smalltooth sawfish critical habitat, and PDC AP.12 does not allow removal of mangroves for these projects. Shallow, euryhaline waters are the other essential feature of smalltooth sawfish critical habitat. Temporary platforms, fills, and cofferdam projects could temporarily affect the shallow, euryhaline essential feature. However, these projects must be completed in 120 days (PDC A10.2) and the areas where the structures were placed must be returned to pre-construction elevations (PDC 10.4).<sup>38</sup> Therefore, there will be no permanent change to either of the essential features and the temporary effects to critical habitat are expected to be insignificant.

#### **Potential Routes of Effect to Gulf Sturgeon Critical Habitat**

We believe that temporary platforms, fill, and cofferdam activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat. USACE anticipates 12 temporary platforms, fill, and cofferdam activities will be covered under this Opinion per 5-year period in Gulf sturgeon critical habitat. In Table 61, we determined that these activities will result in 1.05 ac of temporary effects to waterbottom areas in Gulf sturgeon critical habitat. Although we do not know to what extent these areas contain the PCEs of Gulf sturgeon critical habitat, we evaluate the potential effects to the PCEs below assuming these areas contain the first 3 PCEs (abundant prey items, water quality, sediment quality), and evaluate the effect to the fourth PCE (safe and unobstructed migratory pathways) based on best assumptions below.

 Abundant prey items, such as amphipods, lancelets, polychaetes, gastropods, ghost shrimp, isopods, mollusks and/or crustaceans, within estuarine and marine habitats and substrates for subadult and adult life stages. These activities may temporarily cover and bury nearshore bottom substrates containing sturgeon prey species; however, the effects to this PCE will be insignificant. These effects are primarily short-term in nature, consisting of a temporary loss of benthic invertebrate populations in the affected areas. Observed rates of benthic community recovery after dredging range from 3-24 months (Culter and Mahadevan 1982; Saloman et al. 1982; Wilber et al. 2007). The relatively species-poor benthic assemblages associated with low salinity estuarine sediments can recover in periods of time ranging from a few months to approximately 1 year, while the more diverse communities of high salinity estuarine sediments may require a year or longer. Hence, we believe that these areas will ultimately recover after the platforms, fill, or cofferdams are removed. The estimated 1.05 ac of impact is very small compared to the approximately 1.5 million ac of available marine and

<sup>&</sup>lt;sup>38</sup> This section addresses only the effects of placing the cofferdam, not the effects of the structure surrounded by the cofferdam. The cofferdam and the structure that it surrounds both must meet the PDCs of this Opinion.

estuarine critical habitat that we believe support sturgeon prey species and not all the activities are likely to occur in the preferred foraging depth range or in the sand substrate that supports Gulf sturgeon prey species.

- 2. Water quality, including temperature, salinity, pH, hardness, turbidity, oxygen content, and other chemical characteristics necessary for normal behavior, growth, and viability of all life stages. Localized and temporary reductions in water quality through increased turbidity may result from temporary platforms, fill, and cofferdam activities; however, the effects to this PCE will be insignificant. PDC AP.10 of the PDCs for In-Water Activities requires monitoring and controlling turbidity throughout the duration of all projects. Turbidity curtains will be required for most projects. When the curtains are deployed, turbidity will be contained within the active portion of the project site, and we expect any small amounts of turbidity that may escape to have an insignificant effect on water quality. In a few instances, the USACE project manager has the ability to waive the turbidity curtain requirement as described in PDC AP.10. These instances include projects that are so small that turbidity is expected to be minimal, such as the placement of a single pile, placement of a scientific survey device, or removal of marine debris. Another instance where turbidity curtains may not be used is in areas with high wave energy where securing turbidity curtains would not be feasible, thereby potentially increasing the risk of them becoming loose and entangling animals or damaging nearby habitat. In high energy areas, turbidity would dissipate quickly and would therefore not be a problem. In both of the instances where turbidity curtains will not be used (i.e., for projects that are so small turbidity is expected to be minimal and for high energy areas where turbidity will dissipate very quickly), these activities are expected to have insignificant effects on the water quality feature.
- 3. <u>Sediment quality, including texture and other chemical characteristics, necessary for normal behavior, growth, and viability of all life stages.</u> Temporary platform, fill, and cofferdam activities can affect sediment quality; however, the effects to this PCE will be insignificant. The placement of these materials will temporarily cover areas that support prey species, but not alter the physical or chemical properties of the sediment itself. In addition, the estimated 1.05 ac of impact is very small compared to the approximately 1.5 million ac of available marine and estuarine critical habitat with sediments that we believe support sturgeon prey species and Gulf sturgeon foraging. Further, the piles supporting the temporary platforms will be removed and we expect the areas temporarily affected by the fill will be restored to preconstruction elevations (PDC A10.4) so the sediment present in the area will be the same as preconstruction.
- 4. <u>Safe and unobstructed migratory pathways necessary for passage within and between riverine, estuarine, and marine habitats</u>. Temporary platforms, fill, and cofferdam activities could obstruct migratory pathways for spawning if they blocked areas between estuaries and rivers. Migratory pathways could also be obstructed by these activities in Gulf sturgeon critical habitat if they prevented movement within estuarine and marine areas used for foraging. However, effects to this PCE will be discountable. The mouth of Gulf sturgeon spawning rivers and narrow inlets are identified in Section 2.1.1.2 as Gulf sturgeon critical habitat migratory restriction zones. To prevent Gulf sturgeon from being deterred from

entering or exiting a spawning river, PDC A10.11 prohibits these temporary activities in Gulf sturgeon critical habitat migratory restriction zones.

Because the effects to the PCEs of abundant prey items, water quality, and sediment quality will be insignificant and effects to safe and unobstructed migratory pathways will be discountable, we believe that temporary platforms, fill, and cofferdam activities may affect, but are not likely to adversely affect, Gulf sturgeon critical habitat.

# Potential Routes of Effect to Johnson's Seagrass Critical Habitat

USACE anticipates 12 temporary platform, fill, and cofferdam activities will be covered under this Opinion per 5-year period in Johnson's seagrass critical.

We believe that temporary platforms, fill, and cofferdam activities will have no effect on Johnson's seagrass critical habitat. PDC A10.9 states that these activities cannot occur in areas where the essential features of critical habitat are present.

# 2.2.11 Summary and Cumulative Effect of Proposed Action to Listed Species and Critical Habitat

Sections 2.2.1 - 2.2.10 addressed the effect of each type of activity that can be covered under this Opinion and the potential that these activities are likely to adversely affect listed species or critical habitat. Our effects determinations for species and critical habitat are summarized in Tables 62-63.

	Activity	Sea turtles <sup>40</sup>	Smalltooth sawfish	Sturgeon <sup>41</sup>	Johnson's seagrass	Corals <sup>42</sup>	Whales <sup>43</sup>	Nassau Grouper	Scalloped Hammerhead Shark
1	Shoreline stabilization	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NLAA
2	Pile- supported structure	NLAA	NLAA	NLAA	LAA	NE	NLAA	NLAA	NLAA
3	Dredging	NLAA	NLAA	NLAA	LAA	NE	NE	NLAA	NLAA
4	Water- management outfall structures	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NLAA
5	Scientific survey devices	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NLAA
6	Boat ramps	NLAA	NLAA	NLAA	NE	NE	NLAA	NLAA	NLAA
7	Aquatic enhancement	NLAA	NLAA	NLAA	NE	NE	NLAA	NLAA	NLAA
8	Transmission/ utility lines	NLAA	NLAA	NLAA	LAA	NE	NE	NLAA	NLAA
9	Marine debris removal	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NLAA
10	Temporary platforms, fill, and cofferdams	NLAA	NLAA	NLAA	NE	NE	NE	NLAA	NLAA

Tahla 67	Summar	v of NMFS	Determination	of Effects to	Species by	Activity Type <sup>39</sup>
1 abie 02.	Summar	Y UL INIVILIS	Determination	of Effects to	species by	Activity Type

<sup>39</sup> Note that some of the effects determinations summarized in this table cover multiple types of activities (e.g., aquatic enhancement covers artificial reefs, living shorelines and oyster reefs). The effects determination summarized above is the worst-case scenario addressed in this Opinion and that the determination may be different for each activity type within the category of activities.

<sup>40</sup> Green, hawksbill, Kemp's ridely, leatherback, and loggerhead

<sup>41</sup> Atlantic, Gulf, and shortnose

<sup>42</sup> Elkhorn, staghorn, boulder star, lobed star, mountainous star, pillar, rough cactus

<sup>43</sup> North Atlantic right, blue, fin, sei, sperm

	Activity	Smalltooth sawfish critical habitat	Gulf sturgeon critical habitat	Johnson's seagrass critical habitat	Loggerhead sea turtle critical habitat	<i>Acropora</i> critical habitat	North Atlantic right whale critical habitat	U.S. Caribbean sea turtle critical habitat
1	Shoreline stabilization	LAA	NLAA	LAA	NE	NE	NE	NLAA
2	Pile- supported structure	LAA	NLAA	LAA	NE	NE	NE	NLAA
3	Dredging	NLAA	NLAA	LAA	NE	NE	NE	NLAA
4	Water- management outfall structures	LAA	NLAA	LAA	NE	NE	NE	NLAA
5	Scientific survey devices	NLAA	NLAA	NLAA	NE	NE	NE	NLAA
6	Boat ramps	NLAA	NLAA	NE	NE	NE	NE	NLAA
7	Aquatic enhancement	NE	NLAA	NLAA	NLAA	NE	NLAA	NLAA
8	Transmission / utility lines	NLAA	NLAA	NLAA	NE	NE	NE	NLAA
9	Marine debris removal	NE	NE	NE	NE	NE	NE	NLAA
1 0	Temporary platforms, fill, and cofferdams	NLAA	NLAA	NE	NE	NE	NE	NLAA

 Table 63. Summary of NMFS Determination of Effects to Critical Habitat by Activity

 Type<sup>44</sup>

In the previous sections we considered the effects of each activity individually and the cumulative effects of each category of activity to each species and critical habitat unit. We now consider the cumulative effects of all activities analyzed in this Opinion occurring throughout Florida. For those activities that we believe are likely to adversely affect species or critical habitat, we consider the cumulative effects further in Sections 5 and throughout the rest of the

<sup>&</sup>lt;sup>44</sup> Note that some of the effects determinations summarized in this table cover multiple types of activities (e.g., aquatic enhancement covers living shorelines, artificial reefs, and oyster reefs). The effects determination summarized above is the worst-case scenario addressed in this Opinion and that the determination may be different for each activity type within the category of activities.

NMFS published a final rule designating critical habitat for Atlantic sturgeon on August 17, 2017 (82 FR 39160). This Opinion does not provide an analysis for Atlantic sturgeon critical habitat effects, because the Opinion prohibits projects from occurring within the cricitial habitat (i.e., St Marys River).

Opinion. These include the effects to Johnson's seagrass, Johnson's seagrass critical habitat, and smalltooth sawfish critical habitat.

The cumulative effect to species from all in-water construction in Florida analyzed under this Opinion depends on the concentration of construction in an area at any given time. In other words, does the species have the ability to avoid construction by moving to a nearby area that is not currently under construction? Because we cannot know the exact location where each project will occur at a specific time in the future, we do not know the specific level of development that will occur within a given region or the distance between activities analyzed under this Opinion. Hence, we have to make assumptions based on past development in Florida and the U.S. Caribbean. We assume that projects are not likely to occur simultaneously in a small area. For instance, we assume that only 1 dock or seawall will be installed at a time within a given area (e.g., canal or stretch of shoreline). This assumption is made because the likelihood of multiple neighbors requiring construction at the same time, having the funds to perform the construction at the same time, and having multiple contract companies available to work in the same canal at the same time are extremely low. We also considered the cumulative effects if more than 1 project were to occur generally within the same time period within a region such as the same week or month. Since each of these projects is likely to be completed quickly (a couple of days to a couple of weeks depending on the type of activity), we believe that it is unlikely that multiple projects will occur simultaneously. Also, the areas identified as the most important to these species (e.g., pupping areas for smalltooth sawfish, migratory pathways to spawning rivers for sturgeon, access to nesting beaches for sea turtles) have specific protections both as general PDCs for the Opinion and activity-specific PDCs discussed throughout this document.

Based on this assumption that species will be able to move away from any particular construction location and avoid nearby construction and that sensitive areas are protected, we believe the same discussions for the cumulative effects of each category of activity discussed in the previous sections still applies when considering the cumulative effect to each species from all projects analyzed under this Opinion. Therefore, we believe the effects of construction including the risk of injury from construction, the addition of vessels to an area, the risk of blocking species movement, the temporary exclusion of areas during construction will still result in insignificant effects to these species when considering all construction in Florida addressed by this Opinion.

When considering the cumulative effects from the temporary or permanent impacts to foraging or refuge habitat from individual projects, we believe this will reduce the overall availability for sea turtles and smalltooth sawfish to find adequate foraging or refuge habitat throughout Florida. However, we believe this effect will be insignificant. The PDCs were designed to minimize the loss of these resources. Specifically, PDC AP.3 requires adherence to the PDCs for *Mangrove, Seagrasses, Corals, and Hardbottom for all Projects.* For example, this limits the loss of mangroves to only small areas for dock installation (i.e., to install a 4-8-ft wide walkway) and outfall structures (i.e., 20 lin ft of shoreline) and requires all projects avoid and minimize impacts to seagrasses. These individual and cumulative losses of resources were discussed for each category of activity. We believe these effects are still insignificant when considering the small losses allowed per individual project and that these projects are separated both temporally (over a 5-year period) and spatially (along the entire coast of Florida). Hence, mobile species like sea

turtles, smalltooth sawfish, and sturgeon will be able to forage and seek refuge in nearby areas outside of active project sites.

Effects to critical habitat from each category of activity were also limited by the PDCs to avoid or minimize impacts to essential features and to areas thought to be the most sensitive within those areas (identified as exclusion zones in Section 2.1). For smalltooth sawfish critical habitat and Johnson's seagrass critical habitat, we determined that some of the categories of activities would result in the loss of essential features. The cumulative effects of these losses are discussed further in Section 5 and throughout the remainder of the document. We also determined that there would no effect to *Acropora* critical habitat from any of the proposed activities. For North Atlantic right whale critical habitat and loggerhead critical habitat, the only category of activity that would affect the essential features were the placement of artificial reef materials under Activity 7. These individual and cumulative effects were analyzed in Section 2.2.7, and determined to be insignificant.

For Gulf sturgeon critical habitat, all of the categories of activities except marine debris removal were determined to have discountable or insignificant effects to the essential features. As discussed for each category of activity, impacts to Gulf sturgeon critical habitat were limited by the PDCs that apply to all projects and the activity-specific PDCs. These were limited in ways to ensure sensitive areas used for migration were avoided, projects were completed according to seasonal restrictions so as not to effect the features while sturgeon were present, and generally limited to areas already disturbed (maintenance dredging) or areas that are shallower than the preferred foraging areas used by Gulf sturgeon. Therefore, even though there were losses of essential features within critical habitat, they were limited in ways that are not likely to adversely affect the critical habitat or diminish the ecological function for which the critical habitat was designated.

We believe that none of the types of activities analyzed under this Opinion are likely to change the landscape of Florida's nearshore waters. These activities will allow the continued development of Florida, while the PDCs will help protect species and critical habitat from both individual projects and the combined impacts from the authorization of all projects estimated to occur per 5-year period in Florida. This assumption will be confirmed through the project-level and programmatic review process defined in this Opinion.

# 2.3 **Project-Specific Review**

Before USACE or its delegated authority (e.g., FDEP, Cape Coral, Miami-Dade Division of Environmental Protection, as set forth in Section 2 above) can authorize a covered activity and rely on this Opinion to fulfill its consultation obligations under Section 7 of the ESA, the USACE or entity with delegated authority must conduct a project-specific review to ensure that all of the PDCs are met (Section 2.2). If the PDCs are met, then the project qualifies for coverage under the Opinion. Thereafter, the USACE or the entity with delegated authority must attest to, and certify, compliance with the PDCs and the Opinion in a submission to NMFS, described below (please note there is a separate process for certain projects authorized under SPGP). The USACE's ability to evaluate projects for compliance is largely limited to field observation of structural dimensions as identified on as-built drawings and requirements that are observable post-construction, for example, whether educational signs are in place. PDCs related

to construction methods are not observable after construction. For example, the postconstruction inspections cannot confirm whether the applicant installed a particular number of piles per day or used turbidity curtains during pile installation. All pertinent PDCs will be included as a Special Condition of the Permit. The USACE remains responsible for conducting the programmatic review, discussed in Section 2.4. In addition, as stated in Section 2, where another action agency has been designated as the lead for Section 7 consultation, the USACE remains responsible for both the project-level and programmatic review.

<u>Submission to NMFS</u>: USACE or its delegated authority (e.g., FDEP, Cape Coral, and Miami-Dade Division of Environmental Protection) must email the following information in the bulleted list below to both the USACE and NMFS.

Email:

- 1. NMFS at nmfs.ser.statewideprogrammatic@noaa.gov, and
- 2. USACE at usace.ser.statewideprogrammatic@usace.army.mil

Required information:

- Certification that the activity meets all of the applicable general and activity-specific PDCs of this Opinion.
- A completed form (under development) that includes specific project details, information needed to assess impacts to resources (e.g., amount of waterbottom on which the projects will be placed, size of any piles to be used, pile installation methods), and information that confirms the PDCs are being met (e.g., where the PDCs limit covered projects to a particular size, the form will solicit the size). This form is being jointly developed by NMFS and USACE. NMFS and the USACE may jointly determine that modifications to the form are necessary to ensure that project-specific information is clearly and accurately conveyed and may jointly develop a modified form.
- Any other documentation necessary to support the conclusion that the authorized activity is consistent with the applicable PDCs and is in compliance with this Opinion. This could include project plans, site survey (e.g., benthic, seagrass, hard bottom), photos, environmental assessment, and any other relevant documentation.

NMFS will acknowledge receipt of the email submission through an automatic reply email.

<u>Projects authorized under SPGP</u>: FDEP uses an online, electronic self-certification process that allows individuals to certify that certain single-family docks and boatlifts are in compliance with the terms and conditions of the SPGP and thus covered under the SPGP. The online system requires the applicant to answer basic questions about the property, the proposed activity, and the specific location of the proposed project. The applicant is then required to locate the proposed project using the FDEP's online mapping component to confirm its location relative to protected areas like ESA-designated critical habitat. Finally, the applicant must certify that the information it has submitted is true and accurate. If, based on the answers, the self-certification program determines that the applicant has properly certified that its proposed project complies with the terms and conditions of the SPGP, then the applicant is issued a letter stating that the

proposed activity, as self-certified, is in compliance with the SPGP and may proceed. If, however, the program determines that the applicant has not properly certified that its proposed project complies with the terms and conditions of the SPGP, the applicant will be so informed and instructed to submit an application to the USACE.

As we noted at the outset, we previously consulted on effects of the SPGP and issued a biological opinion on the SPGP (SER-2011-05980). The self-certification program was developed at that time to ensure that only those projects having minor and predictable effects, discussed in the Opinion and consistent with the PDCs in the Opinion, would be immediately issued a letter confirming compliance with the SPGP. The USACE has requested that FDEP continue to be allowed to process applications for these pile-supported structures using the selfcertification process, and asked that the FDEP be exempt from the requirement above that it certify to the USACE and NMFS that the projects authorized under the SPGP self-certification meet the PDCs and are in compliance with this Opinion on a case-by-case basis, but rather to provide a blanket certification at this time. According to the USACE, neither FDEP nor the USACE's compliance inspections on projects authorized under the self-certification have identified problems with this system for compliance with the PDCs of SPGP. To confirm compliance, the USACE inspects 10% of the all USACE permits, including the self-certification permits under the SPGP, after the project has been constructed. Projects authorized under SPGP are also subject to programmatic review by the USACE and NMFS as described in Section 2.4. The self-certification program will continue to apply only to private single-family docks and boatlifts that comply with a list of criteria that either meet the PDCs of this Opinion or are more restrictive than the PDCs in this Opinion. For example, the SPGP self-certification currently only applies to pile-supported structures that are docks that are 1,000 ft<sup>2</sup> or less over water or 500 ft<sup>2</sup> or less over water when in an Aquatic Preserve or Outstanding Florida Water. Given these size limitations of docks in SPGP, these structures are more restrictive than those allowed under PDC A2.1 of this Opinion. By limiting the size, SPGP self-certification only applies to very minor structures, which we expect to have less effect than those allowed under PDC A2.1. In addition, the SPGP self-certification cannot be used for projects:

- On an unbridged, undeveloped coastal island or undeveloped coastal island segment or undeveloped coastal barrier island.
- On sandy beaches fronting the Gulf of Mexico or Atlantic coast shoreline, exclusive of bays, inlets, rivers, bayous, creeks, passes, and the like
- Within 50 ft of the MHWL at any riparian location
- Located in the coastal counties of Wakulla, Taylor, Dixie, Levy, Pasco, and Monroe.
- Located in Biscayne Bay Aquatic Preserve
- Located in the range of Johnson's seagrass
- In Federal Special Waters (Biscayne Bay National Park, Blackwater Creek, Faka Union Canal, Garfield Point, Loxahatchee River, Okeechobee Waterway, Rock Springs Run, St. Marys River, Tampa Bypass Canal, Timicuan Preserve, Wekiva River)
- Located in any of the restriction or exclusion zones identified in Section 2.1.1 of this Opinion
- Location in an area with non-ESA listed seagrasses and will result in any impacts or shading to these seagrasses.

Given the lack of problems with the system in the past, the additional restrictions limiting these projects to minor pile-supported structures, and the confirmation, via the mapping feature, that these pile-supported structures will not occur in sensitive resource areas like critical habitat or within the restriction and exclusion zones identified in this Opinion, the projects that are recognized to be in compliance with the SPGP under the self-certification are not required to undergo the project-specific review procedures described above. In particular, when a project is recognized as in compliance with the SPGP via the self-certification, the FDEP need not submit the project details to NMFS and the USACE and certify that the project meets the PDCs and is in compliance with the Opinion. For projects authorized under the self-certification, the applicant has certified that the project is of the type that would be in compliance with the PDCs, and the USACE is certifying now that the projects authorized via the self-certification will meet the PDCs of this Opinion. To ensure compliance with the terms of this Opinion, the permittee must submit the self-certification statement to FDEP, which will transmit the statement to the USACE, which will transmit the statement to NMFS in connection with the quarterly submissions. Receipt of the USACE's self-certification statement documents compliance with all conditions of the permit, including terms (i.e., PDCs) of this Opinion. Projects processed under the selfcertification are still subject to the programmatic review described in Section 2.4 of this Opinion. The FDEP and the USACE coordinate on FDEP's administration of the SPGP. The USACE may discuss any issues with the self-certified projects on its monthly calls with NMFS. The USACE also must include these projects in its quarterly reports to NMFS and in its annual review. If during any of these programmatic reviews (i.e., the monthly review, quarterly reports, and the annual review), NMFS or the USACE determines that this self-certification process is not working as anticipated, and projects are certified that do not meet the PDCs or other requirements of this Opinion, NMFS and the USACE will work together to take such action as appropriate to address the situation.

Superseding Process for Review and Inclusion of Substantially Similar Projects or Projects with Substantially Similar Effects: In a few instances, a project applicant may propose to use materials or installation methods that were not specifically considered in this Opinion, or the project may deviate from the PDCs in a minor fashion. For example, an applicant may propose to use a different living shoreline material than that considered in the Opinion (Activity 7) or may propose a new shoreline stabilization project that is more than 500 ft in length, and thus does not meet PDC A1.1.1 or involves more mangrove removal than allowed under PDC AP.12. In those instances, the USACE (or its delegated authority) must determine whether the effects of the modification on ESA-listed species or designated critical habitat are substantially similar to the effects considered in this Opinion. If the USACE (or its delegated authority) makes that preliminary determination, it must provide that rationale to NMFS and request permission to rely on the Opinion to satisfy its ESA Section 7 consultation obligations. If NMFS determines that the effect is substantially similar to the effects discussed and found in this Opinion, then NMFS may approve the modification, on that case-specific basis alone, and the project can be covered under this Opinion, and will be included in the programmatic review. If the USACE (or its delegated authority) seeks to authorize a project that proposes to use the same modification in the future, it still must seek permission for the modification. When requesting consideration for a new material, method, or modification, the USACE (or its delegated authority) must await written approval from NMFS before authorizing the project. This process supersedes the review process described above.

# 2.4 Programmatic Review

NMFS and the USACE will have monthly calls and conduct an annual programmatic review of the projects authorized in reliance on this Opinion to evaluate (1) whether the predicted nature and scale of the effects continues to be accurate; (2) whether the PDCs continue to avoid and minimize effects to species and critical habitat as designed or require modification; and (3) whether the project-specific review procedures are being followed and are effective at screening out projects that do not meet the PDCs or are not in compliance with the Opinion. In addition, the USACE will provide quarterly reports to NMFS regarding implementation of the Opinion.

NMFS and the USACE's Jacksonville District have dedicated project managers responsible for implementation, management, and administration of this and other Programmatic Opinions used by the Jacksonville District. The programmatic team leads from NMFS and the USACE, and other members as necessary, participate in monthly calls, quarterly reports, and the annual programmatic review to verify conclusions regarding the potential effects to ESA-listed species and critical habitat, review data on the cumulative effects of the combined projects from the previous year(s), and evaluate and suggest any procedural changes (e.g., modifications to the reporting form or clarification on a PDC) prompted by the review of data. If the monthly calls, quarterly reports, or the annual programmatic review show that the anticipated effects to listed species or critical habitat, as discussed in this Opinion, are different than the effects of the projects as implemented, reinitiation of consultation may be required (50 CFR 402.16). Reviews will be conducted in the following way:

<u>Monthly Call</u>: The USACE and NMFS will conduct a monthly call to discuss how this Opinion is being implemented. The monthly calls provide an opportunity to discuss issues as they arise and answer questions about the implementation of the program as a whole. They also provide a regular opportunity to evaluate whether the projects authorized in reliance on the Opinion are consistent with the Opinion. If both NMFS and USACE agree, the frequency of the calls can be decreased.

NMFS has previously issued programmatic opinions on groups of related USACE-permitted actions, most recently the Florida Statewide Programmatic Opinion, and the USACE and NMFS have had monthly calls to discuss implementation of these programmatic opinions. The monthly calls have proven to be an effective means of ensuring that the USACE and NMFS's understanding and interpretation of the Opinion are consistent, that both agencies are working together toward upholding the intent and integrity of the Opinion, and that the staff from both agencies who have been assigned to implement the programmatic are readily available to each other and to the project managers in both agencies to raise and answer questions as they arise.

<u>USACE Quarterly Reports</u>: The USACE shall provide NMFS with a completed spreadsheet of all activities authorized using this Opinion to satisfy the ESA Section 7 consultation requirements each quarter (i.e., January-March, April-June, July-September, and October-December). Before submitting the spreadsheet to NMFS, USACE shall check the spreadsheet for accuracy (e.g., properly formatted, completely filled out, no duplicates, latitude/longitude data is accurate and entered according to the formatting requirements provided). In addition, the USACE should review the data to determine whether the projects authorized in reliance on the

Opinion are consistent with the Opinion (e.g., the PDCs are being met) and to confirm that its assumptions about the number and location of the projects (and any other assumptions that formed the basis of the effects analysis in Section 2.2) were accurate. For example, this Opinion assumes that a specific number of activities will be completed over a 5-year period, identified in Sections 2.2.1-2.2.11. If the data shows that the number of projects is likely to exceed that expected number, the USACE should inform NMFS in the quarterly report. The USACE shall provide a short summary of their findings with their email submission of the spreadsheet to nmfs.ser.statewideprogrammatic@noaa.gov.

<u>Annual Review</u>: Each year, NMFS and the USACE will conduct an annual review of the projects authorized in reliance on this Opinion. The annual review will cover all projects authorized in the preceding year in reliance on this Opinion. The first annual review will cover projects authorized in the 12-month period starting from the date NMFS completes this Opinion. The second annual review will cover all projects authorized in the second year, and so on. The annual review consists of a USACE-data gathering and review component resulting in an annual, written report from the USACE to NMFS and a NMFS review and comment on the USACE annual review process is described below.

To complete the annual review, the USACE shall:

- 1. Provide an Excel spreadsheet with separate tables (sheets) for each activity type to NMFS. This data shall be reviewed by the USACE to confirm that this Opinion is being implemented properly. This includes confirming that:
  - a. The number of anticipated projects per activity type are consistent with (do not exceed) the anticipated number of projects from Table 8 in Section 2.2.
  - b. The loss of Johnson's seagrass, Johnson's seagrass critical habitat, and smalltooth sawfish critical habitat does not exceed the anticipated loss calculated in this Opinion in Section 7 and Section 8.
  - c. The PDCs were adhered to for each activity type (e.g., new seawalls do not exceed the length limit in PDC A1.1 or that docks were built according to the dock construction guidelines outlined in PDC A2.17 for length, height, and width).
- 2. Provide NMFS with a map of all activities authorized by activity type to confirm that they complied with the exclusion and restriction zones, did not occur in critical habitat, if required under the PDCs. NMFS will provide a map for this purpose that shows the exclusion zones, restriction zones, educational sign zones, and critical habitat boundaries to the USACE as a KMZ layer (i.e., a specific format of geographic information file known as a keyhole markup language). The USACE shall map the location of all projects authorized in reliance on the Opinion as an additional KMZ layer on this map.
- 3. Randomly select 10 projects from each of the 10 categories of activities for a total of 100 projects to be reviewed annually. If there are less than 10 projects in a given category, the USACE will review all of those projects. The USACE will review these projects to ensure that all information was reported accurately, all supporting documentation was provided with the project-specific review submission as outlined in Section 2.3, and that all of the PDCs were met. This review shall be documented and results provided to NMFS.

- 4. Provide a written report of their annual review to NMFS via email to nmfs.ser.statewideprogrammatic@noaa.gov. The report must include:
  - a. A discussion of the results of the data review in Item 1 above.
  - b. A discussion of the review of the mapped projects in Item 2 above.
  - c. A discussion of the results of the detailed project reviews in Item 3 above.
  - d. The results and summary of the USACE's pre- and post-construction compliance inspections completed during the previous year. The USACE performs compliance inspections on at least 10% of all of the projects that they permit. This information provides feedback on if the permitted actions are installed and implemented according to the observable permit conditions. Other entities that authorize projects under this Opinion under authority delegated from the USACE (e.g., FDEP, Cape Coral, Miami-Dade Division of Environmental Protection) shall also provide any compliance inspection data pertaining to projects permitted under the opinion to the USACE for inclusion in this annual report (e.g., FDEP compliance inspections of projects authorized under SPGP using this Opinion).
  - e. Any lessons learned or procedural changes the USACE believes are necessary to improve the program.

NMFS will review the data the USACE submits as well as the USACE's annual report and provide comments or set up a conference call to discuss the results. The NMFS and USACE programmatic team will then work to resolve any outstanding questions or concerns and the results of these discussions will be documented.

Following the annual review, NMFS and the USACE may jointly determine that revisions to the Opinion or the PDCs are necessary. Re-initiation of consultation may be required as appropriate as provided in 50 CFR Section 402.16.

# 3 Status of Listed Species and Critical Habitat Likely to be Adversely Affected

# 3.1 Status of Species Likely to be Adversely Affected

# 3.1.1 Johnson's Seagrass

NMFS listed Johnson's seagrass as threatened under the ESA on September 14, 1998. Kenworthy (1993; 1997; 2000) and NMFS (2002; 2007a) discuss the results of numerous field studies and summarize an extensive literature review regarding the status of Johnson's seagrass. In addition to the published literature, the Johnson's Seagrass Recovery Implementation Team (Recovery Team) is in the process of updating the 2002 Recovery Plan for Johnson's Seagrass. The updated Recovery Plan will contain the latest information concerning the status of this species and potential threats to its persistence and recovery. The following discussion summarizes those findings relevant to our evaluation of the proposed action.

# Life History and Population Biology

Based on the current knowledge of the species, Johnson's seagrass reproduction is believed to be entirely asexual, and dispersal is by vegetative fragmentation. Sexual reproduction in Johnson's seagrass has not been documented. Female flowers have been found; however, dedicated surveys in the Indian River Lagoon have not discovered male flowers, fertilized ovaries, fruits, or seeds, either in the field or under laboratory conditions (Hammerstrom and Kenworthy 2002; Jewett-Smith et al. 1997; NMFS 2007a). Searches throughout the range of Johnson's seagrass have produced the same results, suggesting either that the species does not reproduce sexually or that the male flowers are difficult to observe or describe, as noted for other *Halophila* species (Kenworthy 1997). Surveys to date indicate that the incidence of female flowers appears to be much higher near the inlets leading to the Atlantic Ocean.

Throughout its range, Johnson's seagrass occurs in dynamic and disjunctive patches. It spreads rapidly, growing horizontally from dense apical meristems with leaf pairs having short life spans (Kenworthy 1997). Kenworthy suggested that the observed horizontal spreading, rapid growth patterns, and high biomass turnover could explain the dynamic patches observed in distribution studies of this species. While patches may colonize quickly, they may also disappear rapidly. Sometimes they will disappear for several years and then re-establish, a process referred to as "pulsating patches" (Heidelbaugh et al. 2000; Virnstein and Hall 2009; Virnstein and Morris 2007). Mortality, or the disappearance of patches, can be caused by a number of processes, including burial from bioturbation and sediment deposition (Heidelbaugh et al. 2000), erosion, herbivory, desiccation, and turbidity. In the absence of sexual reproduction, one possible explanation for the pulsating patches is dispersal and re-establishment of vegetative fragments, a process that commonly occurs in aquatic plants and has been demonstrated in other seagrasses (Di Carlo et al. 2005; Philbrick and Les 1996), and was also confirmed by experimental mesocosm<sup>45</sup> studies with Johnson's seagrass (Hall et al. 2006).

Johnson's seagrass is a shallow-rooted species and vulnerable to uprooting by wind, waves, storm events, tidal currents, bioturbation, and motor vessels. It is also vulnerable to burial by sand movement and siltation (Heidelbaugh et al. 2000). Having a canopy of only 2 centimeter (cm) -5 cm, it may be easily covered by sediments transported during storms or redistributed by macrofaunal bioturbation during the feeding activities of benthic organisms. Mesocosm experiments indicate that clonal fragments can only survive burial for up to a period of 12 days (W.J. Kenworthy, NOAA's Center for Coastal Fisheries and Habitat Research, Beaufort, North Carolina, 1997 unpublished). Mechanisms capable of disturbing patches may create clonal fragments that become dispersed. Hall et al. (2006) showed that drifting fragments of Johnson's seagrass can remain viable for 4 to 8 days, during which time they can settle, root, and grow. The process of asexual fragmentation can occur year-round. Fragments could drift several kilometers under the influence of wind and tidally-driven circulation, providing potential recruits for dispersal and new patch formation. In the absence of sexual reproduction, these are likely to be the most common forms of dispersal and patch maintenance.

## Population Status and Distribution

Johnson's seagrass occurs in a variety of habitat types, including on intertidal wave-washed sandy shoals, on flood deltas near inlets, in deep water, in soft mud, and near the mouths of

<sup>&</sup>lt;sup>45</sup> A mesocosm is an experimental tool that brings a small part of the natural environment under controlled conditions.

canals and rivers, where presumably water quality is sometimes poor and where salinity fluctuates widely. It is an opportunistic plant that occurs in a patchy, disjunctive distribution from the intertidal zone to depths of approximately 2-3 m in a wide range of sediment types, salinities, and in variable water quality conditions (NMFS 2007a).

Johnson's seagrass exhibits a narrow geographical range of distribution and has only been found growing along approximately 200 km of coastline in southeastern Florida north of Sebastian Inlet, Indian River County, south to Virginia Key in northern Biscayne Bay, Miami-Dade County. This apparent endemism suggests that Johnson's seagrass has the most limited geographic distribution of any seagrass in the world. Kenworthy (Kenworthy 1997; Kenworthy 1999) confirmed its limited geographic distribution in patchy and vertically disjunctive areas throughout its range. Two survey programs have monitored the presence and abundance of Johnson's seagrass within this range. One program, conducted by the St. Johns River Water Management District since 1994, continues to survey the northern section of the species' geographic range between Sebastian Inlet and Jupiter Inlet (Virnstein and Hall 2009; Virnstein and Morris 2007). The second survey, initiated in 2006, monitored the southern range of the species between Jupiter Inlet and Virginia Key in Biscayne Bay (Kunzelman 2007). This survey is no longer conducted. Since the last status review (NMFS 2007a), there have not been any reported reductions in the geographic range of the species. In fact, the St. Johns River Water Management District observed Johnson's seagrass approximately 21 km north of the Sebastian Inlet mouth on the western shore of the Indian River Lagoon-a discovery that slightly extends the species' known northern range (Virnstein and Hall 2009).

Johnson's seagrass is a perennial species (meaning it lasts for greater than 2 growing seasons), showing no consistent seasonal or year-to-year pattern based on the northern transect surveys, but has exhibited some winter decline (NMFS 2007a). However, during exceptionally mild winters, Johnson's seagrass can maintain or even increase in abundance from summer to winter. In the surveys conducted between 1994 and 2007, it occurred in 7.1% of the l square meter (m<sup>2</sup>) quadrats in the northern range. Depth of occurrence within these surveys ranged from 0.03 to 2.5 m. Where it does occur, its distribution is patchy, both spatially and temporally. It frequently disappeared from transects only to reappear several months or several years later (NMFS 2007a).

Based on the results of the southern transect sampling, it appears there is a relatively continuous, although patchy, distribution of the species from Jupiter Inlet to Virginia Key (NMFS 2007a). The largest reported contiguous patch of Johnson's seagrass in the southern range was observed in Lake Worth Lagoon and was estimated to be 30 ac (Kenworthy 1997). Eiseman and McMillan (1980) documented Johnson's seagrass in the vicinity of Virginia Key (latitude 25.75°N); this location is considered the southern limit of the species' range. There have been no reports of this species further south of the currently known southern distribution. The presence of Johnson's seagrass in northern Biscayne Bay (north of Virginia Key) is well documented by various field experiences and observations of the area by federal, state, and county entities. Johnson's seagrass has been documented in various USACE and USCG permit applications reviewed by NMFS. Findings from the southern transect sampling (summer 2006 and winter 2007) show little difference in the species' frequency or abundance between the

summer and winter sampling period. The lower frequencies of Johnson's seagrass occurred at those sites where larger-bodied seagrasses (e.g., turtle grass, *Thalassia testudinum*, and manatee grass, *Syringodium filiforme*) were more abundant (NMFS 2007a). The southern range transect data support some of the conclusions drawn from previous studies and other surveys. This is a rare species; however, it can be found in relatively high abundance where it does occur. Based on the results of the southern transect sampling, it appears that, although it is disjunctively distributed and patchy, there is some continuity in the southern distribution, at least during periods of relatively good environmental conditions and no significant large-scale disturbances (NMFS 2007a).

Information on the species' distribution and results of limited experimental work suggest that Johnson's seagrass has a wider tolerance range for salinity, temperature, and optical water quality conditions than other species such as paddle grass, *Halophila decipiens* (Dawes et al. 1989) (Kenworthy and Haunert 1991); (Gallegos and Kenworthy 1996); (Durako et al. 2003; Kenworthy and Fonseca 1996; Torquemada et al. 2005). Johnson's seagrass has been observed near the mouths of freshwater discharge canals (Gallegos and Kenworthy 1996), in deeper turbid waters of the interior portion of the Indian River Lagoon (Kenworthy 2000; Virnstein and Morris 2007), and in clear water associated with the high energy environments and flood deltas inside ocean inlets (Heidelbaugh et al. 2000; Kenworthy 1993; Kenworthy 1997; Virnstein and Morris 2007; Virnstein et al. 1997). It can colonize and persist in high-tidal energy environments and has been observed where tidal velocities approach the threshold of motion for unconsolidated sediments (35-40 cm s<sup>-1</sup>). The persistent presence of high-density, elevated patches of Johnson's seagrass on flood tidal deltas near inlets suggests that it is capable of sediment stabilization. Intertidal populations of Johnson's seagrass may be completely exposed at low tides, suggesting high tolerance to desiccation and wide temperature tolerance.

In Virnstein's study areas within the Indian River Lagoon, Johnson's seagrass was found associated with other seagrass species or growing alone in the intertidal, and, more commonly, at the deep edge of some transects in water depths down to 180 cm. In areas in which long-term poor water and sediment quality have existed until recently, Johnson's seagrass appears to occur in relatively higher abundance, perhaps due to the inability of the larger species to thrive. Johnson's seagrass appears to be out-competed in seagrass habitats where environmental conditions permit the larger seagrass species to thrive (Kenworthy 1997; Virnstein et al. 1997). When the larger, canopy-forming species are absent, Johnson's seagrass can grow throughout the full seagrass depth range of the Indian River Lagoon (NMFS 2007a; Virnstein et al. 2009).

Observations by researchers have suggested that Johnson's seagrass exploits unstable environments or newly-created unvegetated patches by exhibiting fast growth and support for all local ramets in order to exploit areas in which it could not otherwise compete. It may quickly recruit to locally uninhabited patches through prolific lateral branching and fast horizontal growth. While these attributes may allow it to compete effectively in periodically disturbed areas, if the distribution of this species becomes limited to stable areas it may eventually be outcompeted by more stable-selected plants represented by the larger-bodied seagrasses (Durako et al. 2003). In addition, the physiological attributes of Johnson's seagrass may limit growth (i.e., spreading) over large areas of substrate if the substrate is somehow altered (e.g., dredged to a depth that would preclude future recruitment of Johnson's seagrass); therefore, its ability to recover from widespread habitat loss may be limited. The clonal and reproductive growth characteristics of Johnson's sea grass result in its distribution being patchy, non-contiguous, and temporally fluctuating. These attributes suggest that colonization between broadly disjunctive areas is likely difficult and that the species is vulnerable to becoming endangered if it is removed from large areas within its range by natural or anthropogenic means.

## Threats

The emerging consensus among seagrass experts on the Recovery Team is that the possibility of mortality due to reduced salinity over long periods of time is the most clearly identified threat to the species' long-term persistence. Some studies have shown that Johnson's seagrass has a wide tolerance for salinity. Conversely, short-term experiments have shown reduced photosynthesis and increased mortality at low salinities (<10 psu [practical salinity units, equivalent to parts per thousand]). Longer duration mesocosm experiments have resulted in 100% mortality of Johnson's seagrass after 10 days at salinities <10 psu (Kahn and Durako 2008). The Recovery Team has determined that the most significant threat to the species is the present or threatened destruction, modification, or curtailment of its habitat or range through water management practices and stochastic environmental factors that can alter the salinity of its habitat. Given that it is not uncommon for salinities to decline below 15 to 20 psu in its range (Steward et al. 2006), and that a number of natural and human-related factors can affect salinity throughout its range, the Recovery Team identified reduced salinity as a potential significant threat to the species because the potential for long-term mortality over a large scale could counteract the life history strategy the species uses to persist in the face of numerous, ongoing, environmental impacts. In previous reviews, including the critical habitat listing rule and the 2002 Recovery Plan, several additional factors were considered threats: (1) dredging and filling, (2) construction and shading from in-and over-water structures, (3) propeller scarring and anchor mooring, (4) trampling, (5) storms, and (6) siltation. In reviewing all information available since the original listing, the Recovery Team conducted assessments of each of these factors and has been unable to confirm that any of these pose a significant threat to the persistence and recovery of the species. A brief discussion of these factors follows.

Routine maintenance dredging associated with the constant movement of sediments in and around inlets may affect seagrasses by direct removal, light limitation due to turbidity, and burial from sedimentation. The disturbance of sediments can also destabilize the benthic community. Altering benthic topography or burying the plants may remove them from the photic zone. Permitted dredging of channels, basins, and other in- and on-water construction projects cause loss of Johnson's seagrass and its habitat through direct removal of the plants, fragmentation of habitat, shading, turbidity, and sedimentation. Although dredge-and-fill activities can and do adversely affect Johnson's seagrass and its designated critical habitat, these activities and the construction of in- and over-water structures are closely scrutinized through federal, state, and local permitting programs. The USACE, under Section 404 of the CWA and Section 10 of the RHA, has federal authority over the issuance of dredge-and-fill permits. These permits include language to protect and conserve seagrasses through field evaluations, consultations, and recommendations to avoid, minimize, and mitigate for impacts to seagrasses.

The USACE's SPGP authorizes permits for in-water construction activities such as shoreline stabilization projects; construction of boat ramps, boat launch areas, and structures associated

with such ramps or launch areas; docks, pier associated facilities, and other minor pilingsupported structures; and maintenance dredging of canals and channels. The previous SPGP (January 1, 2000 to March 31, 2009) was utilized 19,927 times, of which 52% was to issue permits associated with single-family docks (Stu Santos, USACE, pers. comm. to J. Cavanaugh, NMFS PRD, November 2012). The USACE requested consultation on the SPGP on October 30, 2009. NMFS completed a new biological opinion July 25, 2011 on SPGP that was valid through July 25, 2016.

The current SPGP does not allow construction in Johnson's seagrass critical habitat. For a dock to be authorized under the SPGP, the applicant must fully comply with the USACE's and NMFS's October 2002 *Key for Construction Conditions for Docks or Other Minor Structures Constructed in or Over Johnson's Seagrass (Halophila johnsonii)* and the associated August 2001 *Dock Construction Guidelines in Florida for Docks or Other Minor Structures Constructed in or over Submerged Aquatic Vegetation, Marsh, or Mangrove Habitat*. Additional PDCs apply to projects authorized under the SPGP (e.g., docks must be  $\leq 1,000$  ft<sup>2</sup>). The Recovery Team has worked with NMFS's PRD and Habitat Conservation Division staff to develop and improve guidelines for site monitoring methods (Greening and Holland 2003), dock construction guidelines (Shafer et al. 2008), and best management practices to minimize the impact of docks on Johnson's seagrass (Landry et al. 2008).

Shafer et al. (2008) emphasized avoidance of seagrasses as a first priority in their study evaluating the regulatory construction guidelines to minimize impacts to seagrasses from single-family residential dock structures in Florida and Puerto Rico. While most dock construction is subject to the construction guidelines (i.e., the USACE's and NMFS's jointly-developed October 2002 *Key for Construction Conditions for Docks or Other Minor Structures Constructed in or over Johnson's Seagrass* and the 2001 guidelines), some docks meeting certain provisions are exempt from state permitting<sup>46</sup> and contribute to loss of Johnson's seagrass through construction of docks, boat ramps, piers, maintenance dredging, and the construction of other minor over-water structures. The USACE is required to consult with NMFS under Section 7 of the ESA on the effects of implementing the SPGP; therefore, anticipated effects to Johnson's seagrass from implementation of the SPGP would be considered during ESA consultation between the USACE and NMFS. As is noted above, NMFS previously issued a Biological Opinion on the SPGP, which was effective until July 25, 2016. NMFS provides conservation recommendations in its Biological Opinions that, if implemented, would benefit Johnson's seagrass.

The Recovery Team has identified weaknesses in the oversight practices of state and federal agencies in the permitting process for some or all of the activities discussed above, due to budget, staffing, and technological limitations. The need for post-construction permit compliance and enforcement for dock structures in Florida and Puerto Rico has been discussed in Shafer et al. (2008). The Recovery Team also identified difficulties in monitoring Johnson's

<sup>&</sup>lt;sup>46</sup> <u>http://www.dep.state.fl.us/central/Home/SLERP/Docks/sfdock.pdf</u>

seagrass—a rare and patchily-distributed species—in single-event surveys associated with permit applications, and continues to work with collaborators to improve monitoring methods. While it is recognized that dredging and filling projects and construction and shading from in- and overwater structures can adversely affect Johnson's seagrass and its habitat, the Recovery Team determined that these activities are typically local and small-scale. The deficiencies in the permitting process were not presently a significant threat to the survival of Johnson's seagrass because they will not individually or cumulatively result in long-term, large-scale mortality of Johnson's seagrass, nor preclude the species from its strategy of recolonizing areas.

Propeller scarring and improper anchoring are known to adversely affect seagrasses (Kenworthy et al. 2002; Sargent et al. 1995). These activities can severely disrupt the benthic habitat by uprooting plants, severing rhizomes, destabilizing sediments, and significantly reducing the viability of the seagrass community. Propeller dredging and improper anchoring in shallow areas are major disturbances to even the most robust seagrasses. This destruction is expected to worsen with the predicted increase in boating activity within Florida. The Florida Department of Highway Safety and Motor Vehicles (http://www.flhsmv.gov/html/safety.html) reported 963,057 registered commercial and recreational vessels (including canoes) statewide in fiscal year 2007. Registrations declined to 787,780 in fiscal year 2012, likely due to the economic downturn. However, this number is likely to increase based on Florida's projected population growth from 18 million in 2006 to 25 million in 2025 (www.propertytaxreform.state.fl/docs/eo06141.pdf). An increase in the number of registered vessels will likely lead to an increase in adverse effects to seagrasses caused by propeller dredging/scarring.

Other indirect effects associated with motor vessels include turbidity from operating in shallow water, dock construction and repairs, marina expansion, and inlet maintenance dredging. These activities and impacts are also likely to increase (NMFS 2007a). Damage to seagrasses from propeller scarring and improper anchoring by motor vessels is recognized as a significant resource management problem in Florida (Sargent et al. 1995). A number of local, state, and federal statutes protect seagrasses from damage due to vessel impacts, and a number of conservation measures, including the designation of vessel control zones, signage, mooring fields, and public awareness campaigns, are directed at minimizing vessel damage to seagrasses. Despite these efforts, vessel damage can have significant local and small-scale (1 m<sup>2</sup> to 100 m<sup>2</sup>) impacts on seagrasses (Kirsch et al. 2005), but there is no direct evidence that these small-scale local effects are so widespread that they are a threat to the persistence and recovery of Johnson's seagrass.

Trampling of seagrass beds, a secondary effect of recreational boating, also disturbs seagrass habitat, but is a lesser concern. Trampling damages seagrasses by pushing leaves into the sediment and crushing or breaking the leaves and rhizomes. Since the designation of critical habitat; however, there have been no documented observations or reports of damage by trampling, and if there were, they would be small-scale and local. Therefore, the Recovery Team determined that trampling does not constitute a significant threat to the survival or recovery of Johnson's seagrass.

Large-scale weather events such as tropical storms and hurricanes, while often generating runoff conditions that decrease water quality, also produce conditions (wind setup and abrupt water

elevation changes) that can increase flushing rates. The effects of storms can be complex. There are several specifically documented storm effects on seagrasses: (1) scouring and erosion of sediments; (2) erosion of seeds and plants by waves, currents, and surge; (3) burial by shifting sand; (4) turbidity; and (5) discharge of freshwater, including inorganic and organic constituents in the effluents (Steward et al. 2006). Storm effects may be chronic, e.g., due to seasonal weather cycles, or acute, such as the effects of strong thunderstorms or tropical cyclones. Studies have demonstrated that healthy, intact seagrass meadows are generally resistant to physical degradation from severe storms, whereas damaged seagrass beds may not be as resilient (Fonseca et al. 2000; Whitfield et al. 2002). In the late summer and early fall of 2004, a total of 4 hurricanes passed directly over the northern range (with wind strengths at landfall from <39 to 120 miles per hour) of Johnson's seagrass in the Indian River Lagoon. A post-hurricane random survey in the area of the Indian River Lagoon affected by the 4 hurricanes indicated the presence of Johnson's seagrass was similar to that reported by the St. Johns River Water Management District transect surveys prior to the storms. This indicates that while the species may temporarily decline, under the right conditions it can return quickly (Virnstein and Morris 2007). Furthermore, despite evidence of longer-term reductions in salinity, increased water turbidity, and increased water color associated with higher than average precipitation in the spring of 2005, there was no evidence of long-term chronic impacts to seagrasses and no direct evidence of damage to Johnson's seagrass that could be considered a threat to the survival of the species (Steward et al. 2006).

Silt derived from adjacent land and shoreline erosion, river and canal discharges, inlets, and internally re-suspended materials can lead to the accumulation of material on plant leaves causing light deprivation. Deposition of silt can also lead to the burial of plants, accumulation of organic matter, and anoxic sediments. Johnson's seagrass grows in a wide range of environments, including those that are exposed to siltation from all the potential sources. Documentation of the direct effects of siltation on seagrasses is generally unavailable. The absence of seagrass has been associated with the formation of muck deposits; however, and localized areas of flocculent, anoxic sediments in isolated basins and segments of the Indian River Lagoon have been observed. Furthermore, sustained siltation experimentally simulated by complete burial for at least 12 days may cause mortality of Johnson's seagrass (W.J. Kenworthy, NOAA's Center for Coastal Fisheries and Habitat Reserach, Beaufort, North Carolina, unpublished data). In general, the effects of siltation are localized and not widespread and are not likely to threaten the survival of the species.

In addition to the 6 factors discussed above, we also consider the effects of altered water quality on Johnson's seagrass. Availability of light is one of the most significant environmental factors affecting the survival, growth, and distribution of seagrasses (Abal et al. 1994; Bulthuis 1983; Dennison 1987; Kenworthy and Fonseca 1996). Water quality and the penetration of light are affected by turbidity (suspended solids), color, nutrients, and chlorophyll, and are major factors controlling the distribution and abundance of sea grasses (Dennison 1987; Kenworthy and Fonseca 1996). Increases in color and turbidity values throughout the range of Johnson's seagrass generally are caused by high flows of freshwater discharged from water management canals, which can also reduce salinity. Wastewater and storm water discharges, as well as from land runoff and subterranean sources, are also causes of increased turbidity. Degradation of water quality due to increased land use and poor water

management practices continues to threaten the welfare of seagrass communities. Declines in water quality are likely to worsen, unless water management and land use practices can curb or eliminate freshwater discharges and minimize inputs of sediments and nutrients. A nutrient-rich environment caused by inorganic and organic nitrogen and phosphorous loading via urban and agricultural runoff stimulates increased algal growth that may smother or shade Johnson's seagrass, or shade rooted vegetation, and diminish the oxygen content of the water. Low oxygen conditions have a demonstrated negative impact on seagrasses and associated communities.

A long-term monitoring program implemented by the St. Johns River Water Management District assessed overall estuarine water quality in the northern and central region of Johnson's seagrass geographic range as mostly good (67%)(Winkler and Ceric 2006). Only 28% of the stations sampled had fair water quality, while 6% had poor quality. Fifty percent of the sampled estuarine sites were improving, while 6% were degrading, so many more sites were improving than were degrading. Forty-two percent of the lagoon sites had an insignificant trend while 3% had insufficient data to determine a trend. As water management experts have now become confident in the association between water quality and seagrass depth distribution, they have begun establishing water quality targets for the Indian River Lagoon based on seagrass as an indicator (Steward et al. 2005). There is a strong positive correlation between seagrass depth distribution and water quality, which enables managers to predict where seagrasses will grow based on water quality and the availability of light. Given that at least half of the sampling stations were indicating long-term improvements in water quality, it can be assumed that seagrass abundance should not be negatively impacted if water and land use management programs continue to be effective. For example, carefully controlling or reducing water flows from discharge canals will moderate salinity fluctuations and reduce turbidity, color, and light attenuation values.

There has not been a comprehensive assessment of water quality published or reported for the southern geographic range of Johnson's seagrass similar to the St. Johns River Water Management District study performed in the northern and central range. However, water quality experts at the South Florida Water Management District (SFWMD) confirm that efforts are underway to synthesize water quality information and to gain a more comprehensive understanding of the long-term status and trends of water quality in the southern range of Johnson's seagrass (Dan Crean, SFWMD, pers. comm. to Sarah Heberling, NMFS PRD, March 2011). Of particular concern is an assessment of the impacts of fluctuations in water quality corresponding with variation in climatology, especially "wet years" versus "dry years" variation. Future recovery efforts should include close coordination with the SFWMD and county environmental management agencies in Palm Beach and Dade Counties to evaluate the status and trends of water quality in these regions of the species' distribution.

## Climate Change Effects on Seagrasses

Here, we consider the possible effects of climate change (i.e., rising temperatures and sea levels) on seagrasses in general and on Johnson's seagrass in particular. The earth's climate is projected to warm between 2° and 4°C by 2100, and similar projections have been made for marine systems (Sheppard and Rioja-Nieto 2005). At the margins of temperate and tropical bioregions and within tidally-restricted areas where sea grasses are growing at their physiological limits, increased temperatures may result in losses of seagrasses and/or shifts in species composition (Short et al. 2007). The response of seagrasses to increased water temperatures will depend on the thermal tolerance of the different species and their optimum temperature for photosynthesis, respiration, and growth (Short and Neckles 1999). With future climate change and potentially warmer temperatures, there may be a 1 m-5 m rise in the seawater levels by 2100 when taking into account the thermal expansion of ocean water and melting of ocean glaciers. Rising sea levels may adversely impact seagrass communities due to increases in water depths above present meadows, reducing available light. Climate change may also reduce light by shifting weather patterns to cause increased cloudiness. Changing currents may cause erosion, increased turbidity and seawater intrusions higher up on land or into estuaries and rivers, which could increase landward seagrass colonization (Short and Neckles 1999). A landward migration of seagrasses with rising sea levels is a potential benefit, so long as suitable substrate is available for colonization.

It is uncertain how Johnson's seagrass will adapt to rising sea levels and temperatures. Much depends on how much and how quickly temperatures increase. For example, Johnson's seagrass that grows intertidally (e.g., in some parts of the Lake Worth Lagoon) may be affected by a slight change in temperature (since it may already be surviving under less than optimal conditions). However, this may be ameliorated with rising sea levels; assuming Johnson's seagrass would migrate landward with rising sea levels and assuming that suitable substrate would be available for a landward migration. However, rising sea levels could also adversely impact seagrass communities due to increases in water depths above existing meadows reducing available light.

Reduction in light availability may benefit some seagrass species (e.g., *Halophila* species) that require less light compared to the larger, canopy-forming species; therefore, much depends on the thermal tolerance of the different seagrass species and their optimum temperature for photosynthesis, respiration, and growth (Short and Neckles 1999). While sea level has changed many times during the evolutionary history of Johnson's seagrass, it is uncertain how this species will fare when considering the combined effects of rising temperatures and sea levels in conjunction with other stressors such as reduced salinity from freshwater runoff. It has been shown that evolutionary change in a species can occur within a few generations (Rice and Emery 2003), thus making it possible for seagrasses to cope if the changes occur at a rate slow enough to allow for adaptation.

# Status Summary

Based on the results of 14 years of monitoring in the species' northern range (1994-2007) and 3 years of monitoring in the species' southern range (2006-2009), there has been no significant change in the northern or southern range limits of Johnson's seagrass (NMFS 2007a). It appears that the populations in the northern range are stable and capable of sustaining themselves despite stochastic events related to severe storms (Steward et al. 2006) and fluctuating climatology. Longer-term monitoring data are needed to confirm the stability of the southern distribution of the species (NMFS 2007a). However, based on the results of the southern transect sampling, it appears there is a relatively continuous, although patchy, distribution of Johnson's seagrass from Jupiter Inlet to Virginia Key, at least during periods of relatively good environmental conditions and no significant large-scale disturbances. Larger seagrasses, predominantly turtle grass (*Thalassia testudinum*), begin to out-compete Johnson's seagrass in the southern range. While there has been a slight extension in the known northern range (Virnstein and Hall 2009), the limit

of the southern range in the vicinity of Virginia Key (latitude 25.75°N) appears to be stable. There have been no reports of this species further south of the currently known southern distribution.

As discussed in the *Threats* section, the Recovery Team has determined that the possibility of mortality due to reduced salinity over long periods of time is a potential significant threat to the species. The other potential threats discussed above (i.e., dredging/filling, construction and shading from in and over-water structures, propeller scarring and anchor mooring, trampling, storms, and siltation) were determined to be local and small-scale and are not considered threats to the persistence and recovery of the species. It is uncertain how Johnson's seagrass will be affected by the synergistic effects of rising temperatures and sea levels associated with climate change (in conjunction with other stressors such as reduced salinity from freshwater runoff). However, evolutionary change in a species can occur within a few generations (Rice and Emery 2003), thus making it possible for seagrasses to cope if the changes occur at a rate slow enough to allow for adaptation.

# 3.2 Status of Critical Habitat Likely to be Adversely Affected

# 3.2.1 Johnson's Seagrass Critical Habitat

NMFS designated Johnson's seagrass critical habitat on April 5, 2000 (65 FR 17786; see also, 50 CFR 226.213). The specific areas occupied by Johnson's seagrass and designated by NMFS as critical habitat are those with 1 or more of the following criteria:

- 1. Locations with populations that have persisted for 10 years
- 2. Locations with persistent flowering populations
- 3. Locations at the northern and southern range limits of the species
- 4. Locations with unique genetic diversity
- 5. Locations with a documented high abundance of Johnson's seagrass compared to other areas in the species' range

Ten areas (units) within the range of Johnson's seagrass (approximately 200 km of coastline from Sebastian Inlet to northern Biscayne Bay, Florida) are designated as Johnson's seagrass critical habitat (see Table 64). The total acreage of critical habitat for Johnson's seagrass range-wide is roughly 22,574 ac (NMFS 2002).

Unit A	A portion of the Indian River, Florida, north of the Sebastian Inlet Channel
Unit B	A portion of the Indian River, Florida, south of the Sebastian Inlet Channel
Unit C	A portion of the Indian River Lagoon, Florida, in the vicinity of the Fort Pierce Inlet
Unit D	A portion of the Indian River Lagoon, Florida, north of the St. Lucie Inlet
Unit E	A portion of Hobe Sound, Florida, excluding the federally-marked navigation channel of the ICW
Unit F	A portion of the south side of Jupiter Inlet, Florida
Unit G	A portion of Lake Worth, Florida, north of Bingham Island
Unit H	A portion of Lake Worth Lagoon, Florida, located just north of the Boynton Inlet
Unit I	A portion of northeast Lake Wyman, Boca Raton, Florida, excluding the federally- marked navigation channel of the ICW
Unit J	A portion of northern Biscayne Bay, Florida, including all parts of the Biscayne Bay Aquatic Preserve excluding the Oleta River, Miami River, and Little River beyond their mouths, the federally marked navigation channel of the Intracoastal Waterway, and all existing federally authorized navigation channels, basins, and berths at the Port of Miami to the currently documented southernmost range of Johnson's seagrass, Central Key Biscayne

Table 64. Designated critical habitat units for Johnson's seagrass

The physical habitat that supports Johnson's seagrass includes both shallow intertidal and deeper subtidal zones. The species thrives either in water that is clear and deep (2-5 m) or in water that is shallow and turbid. In tidal channels, it inhabits coarse sand substrates. The spread of the species into new areas is limited by its reproductive potential. Johnson's seagrass possesses only female flowers; thus vegetative propagation, most likely through asexual branching, appears to be its only means of reproduction and dispersal. If an established community is disturbed, regrowth and reestablishment are extremely unlikely. This species' method of reproduction impedes the ability to increase distribution as establishment of new vegetation requires considerable stability in environmental conditions and protection from human-induced disturbances.

## Essential Features of Critical Habitat

NMFS identified 4 habitat features essential for the conservation of Johnson's seagrass: (1) adequate water quality, defined as being free from nutrient over-enrichment by inorganic and organic nitrogen and phosphorous or other inputs that create low oxygen conditions; (2) adequate salinity levels, indicating a lack of very frequent or constant discharges of fresh or low-salinity waters; (3) adequate water transparency, which would allow sunlight necessary for photosynthesis; and (4) stable, unconsolidated sediments that are free from physical disturbance. All 4 essential features must be present in an area for it to function as critical habitat for Johnson's seagrass.

#### Status and Threats

A wide range of activities, many funded authorized or carried out by federal agencies, have and will continue to affect the essential habitat requirements of Johnson's seagrass. These are generally the same activities that may affect the species itself, and include: (1) vessel traffic and the resulting propeller dredging; (2) dredge and fill projects; (3) dock, marina, and bridge construction; (4) water pollution; and (5) land use practices (shoreline development, agriculture, and aquaculture).

Vessel traffic has the potential to affect Johnson's seagrass critical habitat by reducing water transparency. Operation of vessels in shallow water environments often leads to the suspension of sediments due to the spinning of propellers on or close to the bottom. Suspended sediments reduce water transparency and the depth to which sunlight penetrates the water column. Populations of Johnson's seagrass that inhabit shallow water and water close to inlets where vessel traffic is concentrated are likely to be most affected. This effect is expected to worsen with increases in boating activity.

The dredging of bottom sediments to maintain, or in some cases create, inlets, canals, and navigation channels can directly affect essential features of Johnson's seagrass critical habitat. Dredging results in turbidity through the suspension of sediments. As discussed previously, the suspension of sediments reduces water transparency and the depth to which sunlight can penetrate the water column. The suspension of sediments from dredging can also resuspend nutrients, which could result in over-enrichment and/or reduce dissolved oxygen levels. Further, dredging can destabilize sediments and alter both the shape and depth of the bottom within the dredged footprint. This may affect the ability of the critical habitat to function through the removal or modification of essential features.

Dock, marina, and bridge construction leads to loss of habitat via construction impacts (e.g., pile installation) and shading. Similar to dredging, installation of piles for docks or bridges can result in increased turbidity that can negatively impact water transparency over short durations. Additionally, installed piles also replace the stable, unconsolidated bottom sediments essential for the species. Completed structures can have long-term effects on critical habitat in the surrounding area because of the shade they produce. While shading does not affect water transparency directly, it does affect the amount and/or duration of sunlight that can reach the bottom. The threat posed by dock, marina, and bridge construction is especially apparent in coastal areas where Johnson's seagrass is found.

Other threats include inputs from adjacent land use. Johnson's seagrass critical habitat located in proximity to rivers, canal mouths, or other discharge structures is affected by land use within the watershed. Waters with low salinity that are highly colored and often polluted are discharged to the estuarine environment. This can impact salinity, water quality, and water transparency, all essential features of Johnson's seagrass critical habitat. Frequent pulses of freshwater discharge to an estuarine area may decrease salinity of the habitat and provoke physiological stress to the species. Nutrient over-enrichment, caused by inorganic and organic nitrogen and phosphorous loading via urban and agricultural land run-off, stimulates increased algal growth, decreased water transparency, and diminished oxygen content within the water. Low oxygen conditions have a demonstrated negative impact on seagrasses and associated communities. Discharges can also contain colored waters stained by upland vegetation or pollutants. Colored waters released

into these areas reduce the amount of sunlight available for photosynthesis by rapidly reducing the amount of shorter wavelength light that reaches the bottom. In general, threats from adjacent land use will be ongoing, randomly occurring events that follow storm events.

# 3.2.2 Smalltooth Sawfish Critical Habitat

## Smalltooth Sawfish Critical Habitat

The U.S. DPS of smalltooth sawfish was listed as endangered on April 1, 2003; however, at that time, NMFS was unable to determine critical habitat. After funding additional studies necessary for the identification of specific habitats and environmental features important for the conservation of the species, establishing a smalltooth sawfish recovery team, and reviewing the best scientific data available, NMFS issued a Final Rule (74 FR 45353; see also, 50 CFR § 226.218) to designate critical habitat for the U.S. DPS of smalltooth sawfish on September 2, 2009. The critical habitat consists of 2 units located along the southwestern coast of Florida: the CHEU, which is comprised of approximately 221,459 ac (346 mi<sup>2</sup>) of coastal habitat, and the TTIEU, which is comprised of approximately 619,013 ac (967 mi<sup>2</sup>) of coastal habitat.

## Critical Habitat Unit Impacted by these Actions

This consultation focuses on an activity occurring in the both CHEU and TTIEU. The CHEU encompasses portions of Charlotte and Lee Counties (Figure 25). The CHEU is comprised of Charlotte Harbor, Gasparilla Sound, Matlacha Pass, Pine Island Sound, San Carlos Bay, and Estero Bay. The unit is fed by the Myakka and Peace Rivers to the north and the Caloosahatchee River to the east. A series of passes between barrier islands connect the CHEU with the Gulf of Mexico. The CHEU is a relatively shallow estuary with large areas of submerged aquatic vegetation, oyster bars, saltwater marsh, freshwater wetlands, and mangroves. Freshwater flows from the Caloosahatchee River are controlled by the Franklin Lock and Dam, which periodically releases water, which thereby affects downstream salinity regimes. The CHEU unit boundaries are defined in detail in the Final Rule (74 FR 45353; see also 50 CFR § 226.218).

The TTIEU is located within Collier, Monroe, and Miami-Dade Counties (see Figure 25). The unit includes the waters of Everglades National Park, Florida Bay, Everglades City, Cape Romano-Ten Thousand Islands Aquatic Preserve, and the portion of Rookery Bay Aquatic Preserve south of state road 92. There are few man-made developments within the unit as the vast majority is federally protected. Developed areas include the areas of Goodland, Everglades City, Plantation, Chokoloskee, and Flamingo. The unit receives freshwater from a number of creeks and rivers found along the coast, including those associated with the Shark River Slough, which originates in and drains central Florida. The TTIEU is a relatively shallow nearshore environment with large areas of submerged aquatic vegetation, oyster bars, mud banks, and mangroves. The TTIEU boundaries are defined in detail in the final rule (74 FR 45353; see also 50 CFR § 226.218).



Figure 25. Map of smalltooth sawfish critical habitat -CHEU and TTIEU

## Essential Features of Critical Habitat

The recovery plan developed for the smalltooth sawfish, which represents NMFS's best judgment about the objectives and actions necessary for the species' recovery, identified a need to increase the number of juvenile smalltooth sawfish developing into adulthood by protecting or restoring nursery habitat (NMFS 2009). NMFS determined that without sufficient habitat, the population was unlikely to increase to a level associated with low extinction risk and de-listing. Therefore, NMFS identified 2 habitat features essential for the conservation of this species: (1) red mangroves, and (2) shallow, euryhaline habitats (shallow, euryhaline habitats) characterized by water depths between the MHWL and -3 ft (-0.9 m) MLLW (Final Rule, 74 FR 45353). These essential features of critical habitat provide juveniles refuge from predation and forage opportunities within their nursery habitat. One or both of these essential features must be present in an action area for it to function as critical habitat for smalltooth sawfish.

#### Habitat Use

Juvenile smalltooth sawfish, identified as those up to 3 years of age or approximately 8 ft (2.4 m) in length (Simpfendorfer et al. 2008), inhabit the shallow waters of estuaries and can be found in sheltered bays, dredged canals, along banks and sandbars, and in rivers (NMFS 2000). Juvenile smalltooth sawfish occur in euryhaline waters (i.e., waters with a wide range of salinities) and are often closely associated with muddy or sandy substrates, and shorelines containing red mangroves (Simpfendorfer 2001; 2003). The structural complexity of red mangrove prop roots

creates a unique habitat used by a variety of fish, invertebrates, and birds. Juvenile smalltooth sawfish, particularly young-of-the-year (measuring less than 39.4 in (100 cm) in length), use these areas as both refuge from predators and forage grounds, taking advantage of the large number of fish and invertebrates found there.

Tracking data from the Caloosahatchee River in Florida indicate very shallow depths and specific salinity ranges are important abiotic factors influencing juvenile smalltooth sawfish movement patterns, habitat use, and distribution (Simpfendorfer et al. 2011). An acoustic tagging study in a developed region of Charlotte Harbor, Florida, identified the importance of mangroves in close proximity to shallow-water habitat for juvenile smalltooth sawfish, stating that juveniles generally occur in shallow water within 328 ft (100 m) of mangrove shorelines (Simpfendorfer et al. 2010). Juvenile smalltooth sawfish spend the majority of their time in waters shallower than 13 ft (4 m) deep (Simpfendorfer et al. 2010) and are seldom found deeper than 32 ft (10 m) (Poulakis and Seitz 2004). Simpfendorfer et al. (2010) also indicated the following developmental differences in habitat use: the smallest young-of-the-year juveniles generally used water shallower than 1.6 ft (0.5 m), had small home ranges, and exhibited high levels of site fidelity. Although small juveniles exhibit high levels of site fidelity for specific nursery habitats for periods of time lasting up to 3 months (Wiley and Simpfendorfer 2007), they undergo small movements coinciding with changing tidal stages. These movements often involve moving from shallow sandbars at low tide and among red mangrove prop roots at higher tides (Simpfendorfer et al. 2010), behavior likely to reduce the risk of predation (Simpfendorfer 2006). As juveniles increase in size, they begin to expand their home ranges (Simpfendorfer et al. 2010; Simpfendorfer et al. 2011), eventually moving to more offshore habitats where they likely feed on larger prey and eventually reach sexual maturity.

Researchers have identified several areas within the Charlotte Harbor Estuary that are disproportionately more important to juvenile smalltooth sawfish, based on intra- or inter-annual capture rates during random sampling events within the estuary (Poulakis 2012; Poulakis et al. 2011). The areas, which were termed "hotspots," correspond with areas where public encounters are most frequently reported. Use of these hotspots can be variable within and among years based on the amount and timing of freshwater inflow. Smalltooth sawfish use hotspots further upriver during drought (i.e., high salinity) conditions and areas closer to the mouth of the Caloosahatchee River during times of high freshwater inflow (Poulakis et al. 2011). At this time, researchers are unsure what specific biotic (e.g., presence or absence of predators and prey) or abiotic factors (e.g., salinity) influence this habitat selection. Still, they believe a variety of conditions in addition to salinity, such as temperature, dissolved oxygen, water depth, shoreline vegetation, and food availability, may influence smalltooth sawfish habitat selection (Poulakis et al. 2011).

#### Status and Threats to Critical Habitat

Modification and loss of smalltooth sawfish critical habitat is an ongoing threat contributing to the current status of the species. Activities such as agricultural and urban development, commercial activities, dredge-and-fill operations, boating, erosion, and diversions of freshwater runoff contribute to these losses (SAFMC 1998). Large areas of coastal habitat were modified or lost between the mid-1970s and mid-1980s within the United States (Dahl and Johnson 1991; USFWS 1999). Since then, rates of loss have decreased even though habitat loss continues.

Between 1998 and 2004, approximately 2,450 ac (3.8 mi<sup>2</sup>) of intertidal wetlands consisting of mangroves or other estuarine shrubs were lost along the Atlantic and Gulf coasts of the United States (Stedman and Dahl 2008). In another study, Orlando et al. (1994) analyzed 18 major southeastern estuaries and recorded over 703 mile (1,131 km) of navigation channels and 9,844 mile (15,842 km) of shoreline with modifications. Additionally, changes to the natural freshwater flows into estuarine and marine waters through construction of canals and other water-control devices have altered the temperature, salinity, and nutrient regimes, reduced both wetlands and submerged aquatic vegetation coverage, and degraded vast areas of coastal habitat utilized by smalltooth sawfish (Gilmore 1995; Quigley and Flannery 2002; Reddering 1988; Whitfield and Bruton 1989). Juvenile sawfish and their critical habitat are particularly vulnerable to these kinds of habitat losses or alterations due to the juveniles' affinity for (and developmental need of) shallow, estuarine systems. Although many forms of habitat modification are currently regulated, some permitted direct and/or indirect damage to habitat from increased urbanization still occurs and is expected to continue in the future.

In Florida, coastal development often involves the removal of mangroves, the armoring of shorelines through seawall construction, and the dredging of canals. This is especially apparent in master plan communities such as Cape Coral and Punta Gorda which are located within the Charlotte Harbor Estuary. These communities were created through dredge-and-fill projects to increase the amount of waterfront property available for development, but in doing so, developers removed the majority of red mangrove habitat from the area. The canals created by these communities require periodic dredging for boat access, further affecting the shallow, euryhaline essential feature of critical habitat. Development continues along the shorelines of Charlotte Harbor in the form of docks, boat ramps, shoreline armoring, utility projects, and navigation channel dredging.

To protect critical habitat, federal agencies must ensure that their activities are not likely to result in the destruction or adverse modification of the physical and biological features that are essential to the conservation of sawfish, or the species' ability to access and use these features (ESA Section 7(a)(2); see also 50 CFR 424.12(b) [discussing essential features]). Therefore, proposed actions that may impact critical habitat require an analysis of potential impacts to each essential feature. As mentioned previously, there are 2 essential features of smalltooth sawfish critical habitat: (1) red mangroves; and (2) shallow, euryhaline habitats characterized by water depths between the MHWL and -3 ft (-0.9 m) measured at MLLW. The USACE oversee the permitting process for residential and commercial marine development in the CHEU. The FDEP and their designated authorities also regulate mangrove removal in Florida. All red mangrove removal permit requests within smalltooth sawfish critical habitat necessitate ESA Section 7 consultation. NMFS PRD tracks the loss of these essential features of smalltooth sawfish critical habitat.

#### Threats to Critical Habitat

#### Dock and Boat Ramp Construction

The USACE recommends that applicants construct docks in accordance with the NMFS-USACE *Dock Construction Guidelines in Florida for Docks or Other Minor Structures Constructed in or over Submerged Aquatic Vegetation (SAV), Marsh, or Mangrove Habitat* ("Dock Construction Guidelines") when possible. The current dock construction guidelines allow for some amount of

mangrove removal; however, it is typically restricted to either (1) trimming to facilitate a dock, or (2) complete removal up to the width of the dock extending toward open water, which the guidelines define as a width of 4 ft. Installation or replacement of boat ramps is often part of larger projects such as marinas, bridge approaches, and causeways where natural and previously created deepwater habitat access channels already exist. Boat ramps can result in the permanent loss of both the red mangrove and the shallow, euryhaline habitat features of critical habitat for smalltooth sawfish.

#### Marina Construction

Marinas have the potential to adversely affect aquatic habitats. Marinas are typically designed to be deeper than -3 ft MLLW to accommodate vessel traffic; therefore, most existing marinas lacking essential features are unlikely to function as critical habitat for smalltooth sawfish. The expansion of existing marinas and creation of new marinas can result in the permanent loss of large areas of this nursery habitat.

## Bulkhead and Seawall Construction

Bulkheads and other shoreline stabilization structures are used to protect adjacent shorelines from wave and current action and to enhance water access. These projects may adversely impact critical habitat for smalltooth sawfish by removal of the essential features through direct filling and dredging to construct vertical or riprap seawalls. Generally, vegetation plantings, sloping riprap, or gabions are environmentally-preferred shoreline stabilization methods instead of vertical seawalls because they provide better quality fish and wildlife habitat. Nevertheless, placement of riprap material removes more of the shallow euryhaline essential feature than a vertical seawall. Also, many seawalls built along unconsolidated shorelines require the removal of red mangroves to accommodate the seawalls.

## Cable, Pipeline, and Transmission Line Construction

While not as common as other activities, excavation of submerged lands is sometimes required for installing cables, pipelines, and transmission lines. Construction may also require temporary or permanent filling of submerged habitats. Open-cut trenching and installation of aerial transmission line footers are activities that have the ability to temporarily or permanently impact critical habitat for smalltooth sawfish.

## Transportation Infrastructure Construction

Potential adverse effects from federal transportation projects in smalltooth sawfish critical habitat (CHEU) include operations of the Federal Highway Administration, USACE, and FEMA. Construction of road improvement projects typically follow the existing alignments and expand to compensate for the increase in public use. Transportation projects may impact critical habitat for smalltooth sawfish through installation of bridge footers, fenders, piles, and abutment armoring, or through removal of existing bridge materials by blasting or mechanical efforts.

# Dredging

Riverine, nearshore, and offshore areas are dredged for navigation, construction of infrastructure, and marine mining. An analysis of 18 major southeastern estuaries conducted in 1993-1994 demonstrated that over 7,000 km of navigation channels have already been dredged (Orlando et al. 1994). Habitat effects of dredging include the loss of submerged habitats by disposal of

excavated materials, turbidity and siltation effects, contaminant release, alteration of hydrodynamic regimes, and fragmentation of physical habitats (GMFMC 1998; GMFMC 2005; SAFMC 1998). In the CHEU, dredging to maintain canals and channels constructed prior to the critical habitat designation, limits the amount of available shallow, euryhaline essential feature to the edges of waterways and these dredging activities can disturb juveniles that are using these areas. At the time of critical habitat designation, many previously dredged channels and canals existed within the boundaries of the critical habitat units; however, we are unsure which of those contained the shallow-water essential feature at that time. It is likely that many of these channels and canals were originally dredged deeper than -3 ft MLLW, but they have since shoaled in and now contain the essential feature of shallow, euryhaline habitat. Therefore, maintenance dredging impacts are counted as a loss to this essential feature, even though the areas may or may not have contained the essential feature at time of designation (see Figure 26 Diagrams A and B).





Diagram A depicts a cross section of a historically dredged channel/canal within the boundaries of the critical habitat units that has not been maintained. Diagram B depicts the typical cross section of a maintenance dredged channel/canal. Diagram C depicts a cross section of a maintained dredged channel/canal after sea level rise of > 1 ft.
Construction, Operations and Maintenance of Impoundments and Other Water Level Controls Federal agencies such as the USACE have historically been involved in large water control projects in Florida. Agencies sometimes propose impounding rivers and tributaries for such purposes as flood control, salt water intrusion prevention, or creation of industrial, municipal, and agricultural water supplies. Projects to repair or replace water control structures may affect smalltooth sawfish critical habitat by limiting sufficient freshwater discharge which could alter the salinity of estuaries. The ability of an estuary to function as a nursery depends upon the quantity, timing, and input location of freshwater inflows (Garmestani and Percival 2005; Norton et al. 2012; USEPA 1994). Estuarine ecosystems are vulnerable to the following man-made disturbances: (1) decreases in seasonal inflow caused by the removal of freshwater upstream for agricultural, industrial, and domestic purposes; (2) contamination by industrial and sewage discharges; (3) agricultural runoff carrying pesticides, herbicides, and other toxic pollutants; and (4) eutrophication (e.g., influx of nutrients such as nitrates and phosphates most often from fertilizer runoff and sewage) caused by excessive nutrient inputs from a variety of nonpoint and point sources. Additionally, rivers and their tributaries are susceptible to natural disturbances, such as floods and droughts, whose effects can be exacerbated by these man-made disturbances.

As stated above, smalltooth sawfish show an affinity for a particular salinity range, moving downriver during wetter months and upriver during drier months to remain within that range (Simpfendorfer et al. 2011). Therefore, water management decisions that affect salinity regimes may impact the functionality of critical habitat. This may result in smalltooth sawfish following specific salinity gradients into less advantageous habitats (e.g., areas with less shallow-water or red mangrove habitat). Furthermore, large changes in water flow over short durations would likely escalate movement patterns for smalltooth sawfish, thereby increasing predation risk and energy output. Researchers are currently looking into the effects of large-scale freshwater discharges on smalltooth sawfish and their designated critical habitat. The most vulnerable portion of the juvenile sawfish population to water-management outfall projects appears to be smalltooth sawfish in their first year of life. Newborn smalltooth sawfish remain in smaller areas irrespective of salinity, which potentially exposes them to greater osmotic stress (a sudden change in the solute concentration around a cell, causing a rapid change in the movement of water across its cell membrane), and impacts the nursery functions of sawfish critical habitat (Poulakis et al. 2013; Simpfendorfer et al. 2011).

#### **Climate Change Threats**

The Intergovernmental Panel on Climate Change (IPCC) has stated that global climate change is unequivocal and its impacts to coastal resources may be significant (IPCC 2007). There is a large and growing body of literature on past, present, and future impacts of global climate change induced by human activities (i.e., global warming mostly driven by the burning of fossil fuels). The latest report by the IPCC (2013) is more explicit, stating that, "science now shows with 95% certainty that human activity is the dominant cause of observed warming since the mid-twentieth century." Some of the anticipated outcomes are sea level rise, increased frequency of severe weather events, and changes in air and water temperatures. NOAA's climate change web portal provides information on the climate-related variability and changes that are exacerbated by human activities (http://www.climate.gov/#understandingClimate). The EPA's climate change webpage also provides basic background information on these and other measured or anticipated effects (http://www.epa.gov/climatechange/index.html).

Though the impacts on smalltooth sawfish cannot, for the most part, be predicted with any degree of certainty, we can project some effects to sawfish critical habitat. We know that both essential features (red mangroves and shallow, euryhaline waters less than -3 ft deep at MLLW) will be impacted by climate change. Sea level rise is expected to exceed 3.3 ft (1 m) globally by 2100, according to the most recent publications, exceeding the estimates of the Fourth Assessment of the IPCC (Meehl et al. 2007; Pfeffer et al. 2008; Rahmstorf et al. 2009). Mean sea level rise projections have increased since the Fourth Assessment because of the improved physical understanding of the components of sea level, the improved agreement of process-based models with observations, and the inclusion of ice-sheet dynamical changes (IPCC 2013). A 1-m sea level rise in the state of Florida is within the range of recent estimates by 2080 (Pfeffer et al. 2008; Rahmstorf et al. 2009).

Sea level increases would affect the shallow-water essential feature of smalltooth sawfish critical habitat within the CHEU. A 2010 climate change study by the Massachusetts Institute of Technology forecasted sea level rise in a study area with significant overlap with the CHEU (Vargas-Moreno and Flaxman 2010). The study investigated possible trajectories of future transformation in Florida's Greater Everglades landscape relative to 4 main drivers: climate change, shifts in planning approaches and regulations, population change, and variations in financial resources. Massachusetts Institute of Technology used (IPCC 2007) sea level modeling data to forecast a range of sea level rise trajectories from low, to moderate, to high predictions (Figure 27). The effects of sea level rise on available shallow-water habitat for smalltooth sawfish would be exacerbated in areas where there is shoreline armoring (e.g., seawalls). This is especially true in canals where the centerlines are maintenance-dredged deeper than 3 ft (0.9 m) for boat accessibility. In these areas, the areas that currently contain the essential feature depth (less than -3 ft at MLLW) will be reduced along the edges of the canals as sea level rises (see previous Figure 26, Diagram C).



Figure 27. From left to right: current shoreline, +3.5 in (+9 cm); +18.5 in (+47 cm); and +38.97 in (+99 cm) sea level rise by 2060.<sup>47</sup>

Along the Gulf Coast of Florida, and south Florida in particular, rises in sea level will impact mangrove resources. As sea levels rise, mangroves will be forced landward in order to remain at a preferred water inundation level and sediment surface elevation, which is necessary for successful growth. This retreat landward will not keep pace with conservative projected rates of elevation in sea level (Gilman et al. 2008). This forced landward progression poses the greatest threat to mangroves in areas where there is limited or no room for landward or lateral migration (Semeniuk 1994). Such is the case in areas of the CHEU where landward mangrove growth is restricted by shoreline armoring and coastal development. This man-made barrier will prohibit mangroves from moving landward and will result in the loss of the mangrove essential feature. Other threats to mangroves result from climate change: fluctuations in precipitation amounts and distribution, seawater temperature, carbon dioxide (CO<sub>2</sub>) levels, and damage to mangroves from increasingly severe storms and hurricanes (McLeod and Salm 2006). A 25% increase in precipitation globally is predicted by 2050 (McLeod and Salm 2006), but the specific geographic distribution will vary, leading to increases and decreases in precipitation at the regional level. Changes in precipitation patterns caused by climate change may adversely affect the growth of mangroves and their distribution (Field 1995; Snedaker 1995). Decreases in precipitation will increase salinity and inhibit mangrove productivity, growth, seedling survival, and spatial coverage (Burchett et al. 1984). Decreases in precipitation may also change mangrove species composition, favoring more salt-tolerant types (Ellison 2010). Increases in precipitation may benefit some species of mangroves, increasing spatial coverage and allowing them to outcompete other salt marsh vegetation (Harty 2004). Even so, potential mangrove expansion

<sup>&</sup>lt;sup>47</sup> Adapted from (Vargas-Moreno and Flaxman), M. Addressing the Challenges of Climate Change in the Greater Everglades Landscape. Project Sheet. November, 2010. Department of Urban Planning, Massachusetts Institute of Technology.

requires suitable habitat for mangroves to increase their range, which depends to a great extent on patterns and intensity of coastal development (i.e., bulkhead and seawall construction). Seawater temperature changes will have potential adverse effects on mangroves as well. Many species of mangroves show an optimal shoot density in sediment temperatures between 59°-77°F (15°-25°C) (Hutchings and Saenger 1987). Yet, at temperatures between 77°-95°F (25°-35°C), many species begin to show a decline in leaf structure and root and leaf formation rates (Saenger and Moverley 1985). Temperatures above 95°F lead to adverse effects on root structure and survivability of seedlings (UNESCO 1992) and temperatures above 100.4°F (38°C) lead to a cessation of photosynthesis and mangrove mortality (Andrews et al. 1984). Although impossible to forecast precisely, sea surface ocean temperatures are predicted to increase 1.8°-3.6°F (1°-2°C) by 2060 (Chapter 11 (IPCC 2013)), which will in turn impact underlying sediment temperatures along the coast. If mangroves shift pole-ward in response to temperature increases, they will at some point be limited by temperatures at the lower end of their optimal range and available recruitment area. This is especially true when considering already armored shorelines in residential communities such as those within and surrounding the CHEU of critical habitat for smalltooth sawfish.

As atmospheric CO<sub>2</sub> levels increase, mostly resulting from man-made causes (e.g., burning of fossil fuels), the world's oceans will absorb much of this CO<sub>2</sub>, causing potential increases in photosynthesis and mangrove growth rates. This increase in growth rate, however, would be limited by lower salinities expected from CO<sub>2</sub> absorption in the oceans (Ball et al. 1997), and by the availability of undeveloped coastline for mangroves to expand their range. A secondary effect of increased CO<sub>2</sub> concentrations in the oceans is the deleterious effect on coral reefs' ability to absorb calcium carbonate (Hoegh-Guldberg et al. 2007), and subsequent reef erosion. Eroded reefs may not be able to buffer mangrove habitats from waves, especially during storm/hurricane events, causing additional physical effects.

Finally, the anticipated increase in the severity of storms and hurricanes may also impact mangroves. Tropical storms are expected to increase in intensity and/or frequency, which will directly impact existing mangroves that are already adversely impacted by increased seawater temperatures, CO<sub>2</sub>, and changes in precipitation (Cahoon et al. 2003; Trenberth 2005). The combination of all of these factors may lead to reduced mangrove height (Ning et al. 2003). Further, intense storms could result in more severe storm surges and lead to potential changes in mangrove community composition, mortality, and recruitment (Gilman et al. 2006). Increased storms surges and flooding events could also affect mangroves' ability to photosynthesize (Gilman et al. 2006) and the oxygen concentrations in the mangrove lenticels (Ellison 2010).

## 4 Environmental Baseline

This section is an analysis of the effects of past and ongoing human and natural factors leading to the current status of the species, its habitat (including designated critical habitat), and the ecosystem, <u>within the action area</u>. By regulation, environmental baseline for Biological Opinions include the past and present impacts of all state, federal, or private actions and other human activities in the action area. We identify the anticipated impacts of all proposed federal projects in the specific action area of the consultation at issue that have already undergone formal or early Section 7 consultation as well as the impact of state or private actions which are contemporaneous with the consultation in process (50 CFR 402.02, emphasis added).

Focusing on the impacts of the activities in the action area specifically allows us to assess the prior experience and state (or condition) of the endangered and threatened individuals and areas of designated critical habitat that occur in an action area and that will be exposed to effects from the actions under consultation. This is important because, in some phenotypic states or life history stages, listed individuals will commonly exhibit, or be more susceptible to, adverse responses to stressors than they would be in other states, stages, or areas within their distributions. The same is true for localized populations of endangered and threatened species; the consequences of changes in the fitness or performance of individuals on a population's status depends on the prior state of the population. Designated critical habitat is not different; under some ecological conditions, the physical and biotic features of critical habitat will exhibit responses that they would not exhibit in other conditions.

# 4.1 Factors Affecting Johnson's Seagrass and Johnson's Seagrass Critical Habitat in the Action Area

# 4.1.1 Federal and State Actions

A wide range of activities funded, authorized, or carried out by federal agencies may affect Johnson's seagrass and the essential features of designated critical habitat for Johnson's seagrass. These include dredging, dock/marina construction, boat shows, bridge/highway construction, residential construction, shoreline stabilization, breakwaters, and the installation of subaqueous lines or pipelines. Other federal actions (or actions with a federal nexus) that may affect Johnson's seagrass and its designated critical habitat include actions by the EPA and the USACE to manage freshwater discharges into waterways; regulation of vessel traffic by the USCG; management of National Parks; management of vessel traffic (and other activities) by the U.S. Navy; and authorization of state coastal zone management plans by NOAA's National Ocean Service. Although these actions have probably removed Johnson's seagrass and affected its critical habitat, none of these past actions have jeopardized the continued existence of Johnson's seagrass, or destroyed or adversely modified its critical habitat.

Between April 1, 2008 and January 5, 2015, according to NMFS's Public Consultation Tracking System database, NMFS completed consultation under Section 7 of the ESA on 359 proposed activities with the potential to affect Johnson's seagrass and/or its designated habitat. Of these consultations, 90 were concluded formally (i.e., with issuance of a Biological Opinion), and the majority of these projects were single- or multi-family dock construction projects that each resulted in a few square feet to a few hundred square feet of impacts to Johnson's seagrass and/or its designated critical habitat. Other types of projects fall into 1 of the categories listed in the

previous paragraph and the majority of these projects resulted in impacts to less than 0.1 ac of Johnson's seagrass or its designated critical habitat. A few larger scale dredging projects resulted in more significant impacts.

#### Coastal Construction and Urban Development

Activities such as dock and seawall construction, dredging, and installation of smaller, less common structures like water outfall structures within the range of Johnson's seagrass and/or its critical habitat will continue, as the shoreline is highly prized for residential and commercial development. Since 2001, applicants have been encouraged to follow the NMFS and USACE's *Construction Guidelines for Minor Piling-Supported Structures in or over Submerged Aquatic Vegetation (SAV), Marsh or Mangrove Habitat.* Since 2002, they have been encouraged to follow the NMFS and USACE's *Key for Construction Conditions for Docks or Other Minor Structures Constructed in or Over Johnson's seagrass (Halophila johnsonii)* in order to minimize shading impacts to Johnson's seagrass and its critical habitat. Nevertheless, loss of Johnson's seagrass will continue due to shading and the installation of piles, even if docks are designed in full compliance with the dock construction guidelines.

Urban development since the 1960s has affected inshore water quality throughout the range of Johnson's seagrass and/or its critical habitat. However, Woodward-Clyde Consultants (1994) noted that improvements in erosion and sediment control in association with urban development in the 1980s and 1990s may have been responsible for reduced turbidity in those decades as compared to the previous 2 decades of development. Reductions in seagrasses were apparent in the 1970s, along with areas of highly turbid water. Increases in submerged aquatic vegetation were noted until coverage and density peaked in 1986, albeit at levels remaining below those observed in the decades prior to 1960. In association with upland development, water quality and transparency within the range of Johnson's seagrass are affected by storm water and agricultural runoff, wastewater discharges, and other point and nonpoint source discharges. The most clearly identified and manageable threat to the persistence and recovery of Johnson's seagrass is the possibility of mortality due to reduced salinity over long periods of time.

#### Recreational Vessel Traffic

Increasing recreational vessel traffic in the range of Johnson's seagrass resulting from marina and dock construction results in improper anchoring and propeller scarring. Propeller scarring and improper anchoring are known to adversely affect seagrasses (Kenworthy et al. 2002; Sargent et al. 1995) and are a major disturbance to even the most robust seagrasses in shallow waters. These activities can severely disrupt the benthic habitat by uprooting plants, severing rhizomes, destabilizing sediments, and significantly reducing the viability of the seagrass community. This destruction is expected to worsen with the predicted increase in boating activity within Florida. Damage to seagrasses from propeller scarring and improper anchoring by motor vessels is recognized as a significant resource management problem in Florida (Sargent et al. 1995). A number of local, state, and federal statutes prohibit damaging seagrasses through vessel impacts, and a number of conservation measures, including the designation of vessel control zones and mooring fields, the installation of signage and the implementation of public awareness campaigns, are directed at minimizing vessel damage to seagrasses. Despite these efforts, damage caused by vessels can have significant local and small-scale (1 m<sup>2</sup> to 100 m<sup>2</sup>)

impacts on seagrasses (Kirsch et al. 2005), but there is no direct evidence that these small-scale local effects are so widespread that they are a threat to the survival of Johnson's seagrass.

# 4.1.2 Conservation and Recovery Activities That May Benefit Johnson's Seagrass and Johnson's Seagrass Critical Habitat

State and federal conservation measures exist to protect Johnson's seagrass and its designated critical habitat under an umbrella of management and conservation programs that address seagrasses in general (Kenworthy et al. 2006). Johnson's seagrass habitat is also included in the designation of critical habitat for the Florida manatee and is therefore subject to ESA Section 7 consultation by the USFWS, which has ESA jurisdiction over that species. These conservation measures must be continually monitored and assessed to determine if they will ensure the long-term protection of the species and the maintenance of environmental conditions suitable for its continued existence throughout its geographic distribution.

# 4.2 Factors Affecting Smalltooth Sawfish Critical Habitat within the Action Area

## 4.2.1 Federal Actions

A wide range of activities funded, authorized, or carried out by federal agencies may affect the essential features of smalltooth sawfish critical habitat. These include dredging, dock/marina construction, residential construction, shoreline stabilization, the installation of breakwaters, and the installation of utility lines. Other federal actions (or actions with a federal nexus) that may beneficially affect smalltooth sawfish critical habitat include managing freshwater discharges consulted on under the Comprehensive Everglades Restoration Plan (NMFS tracking number SER-2013-11848) and management of Everglades National Park, where most of the smalltooth sawfish critical habitat for the TTIEU is located.

Between September 2009 and March 2017, NMFS has completed 122 consultations in CHEU of smalltooth sawfish critical habitat with the majority of these being minor residential development with each resulting a few hundred square feet or less of impacts to critical habitat. Because of the comparatively few number of projects within the TTIEU (a total of 11 projects), the focus of this consultation will be the CHEU.

In addition to activities that are consulted on a project-by-project basis by NMFS, activities are also authorized in smalltooth sawfish critical habitat under USACE GPs consulted on programmatically by NMFS (see Appendix A). Specifically, USACE GP SAJ-91 allows for the continued installation of docks and seawalls in the residential canals of Cape Coral (NMFS 2012). The individual and cumulative effects of these actions are monitored and tracked by both USACE and NMFS as part of the programmatic review process.

## 4.2.2 State or Private Actions

A number of non-federal activities that may adversely affect designated critical habitat for smalltooth sawfish in the action area including impacts from wastewater systems, aquaculture facilities, and residential shoreline stabilization activities that do not obtain federal permits (i.e., seawall, riprap). The direct and indirect impacts from some of these activities are difficult to quantify. However, where possible, conservation actions through the ESA Section 10 permitting, ESA Section 6 cooperative agreements, and state permitting programs are being implemented or investigated to monitor or study impacts from these sources. There are

numerous shoreline stabilization projects that have occurred and continue to occur within the smalltooth sawfish critical habitat that have completed the Section 7 consultation process.

# 4.2.3 Other Potential Factors

# Natural Disturbances

Stochastic (i.e., random) events, such as hurricanes, are common throughout the range of the smalltooth sawfish, especially in the current core of its range (i.e., south and southwest Florida). These events are by nature unpredictable, and their effect on designated critical habitat and the recovery of the species is unknown. However, they have the potential to impede recovery directly if animals die, or indirectly if important habitats are damaged as a result of these disturbances. In 2005, Hurricane Charley damaged habitat within smalltooth sawfish critical habitat, which has seemed to recover.

# Conservation and Recovery Actions Shaping the Environmental Baseline

Federal Essential Fish Habitat (EFH) consultation requirements pursuant to the Magnuson-Stevens Fishery Conservation and Management Act minimize and mitigate for losses of wetland and preserve valuable foraging and developmental habitat that is used by juvenile smalltooth sawfish. NMFS has designated mangrove and estuarine habitats as EFH as recommended by the Gulf of Mexico Fishery Management Council. Both essential features (shallow, euryhaline water less than/ shallower than -3 ft MLLW and red mangroves) are critical components of areas designated as EFH and receive a basic level of protection under the Magnuson-Stevens Act to the extent that the Act requires minimization of impacts to EFH resources.

# 5 Effects of the Action

# 5.1 Effects to Johnson's Seagrass

As discussed in Section 2.2 and summarized in Table 65, an estimated 22,378.96 ft<sup>2</sup> (0.51 ac) of Johnson's seagrass may be lost by pile-supported structures (Activity 2), dredging (Activity 3), and utility line repairs (Activity 8). In particular, we believe docks and other pile-supported structures could be placed on Johnson's seagrass, and dredging and utility line repairs could remove areas that support the species. NMFS believes these proposed actions are likely to adversely affect Johnson's seagrass, which is listed as threatened under the ESA. However, no incidental take statement or reasonable and prudent measures will be issued because the ESA does not require biological opinions to contain incidental take statements for threatened plants. Yet, because the actions will result in adverse effects to Johnson's seagrass, we must evaluate whether the actions are likely to jeopardize the continued existence of the species (Section 7).

	Category of Activity	Loss of Johnson's seagrass (ft²)	Loss of Johnson's seagrass (ac)
1	Shoreline stabilization	-	
2	Pile-supported structure	398.96	0.01
3	Dredging	21,780.00	0.50
4	Water-management outfall structures	-	
5	Scientific survey devices	-	
6	Boat ramps	-	
7	Aquatic enhancement	-	
8	Transmission/utility lines	200.00	$0.00^{48}$
9	Marine debris removal	-	
10	Temporary platforms, fill, and cofferdams	-	
	Total	22,378.96	0.51

Table 65. Impacts to Johnson's Seagrass Anticipated per 5-year period

# 5.2 Effects to Johnson's Seagrass Critical Habitat

As discussed in Section 2.2, Johnson's seagrass critical habitat may be affected by some activities authorized under this Opinion. Table 66 summarizes the estimated loss of critical habitat.

Direct impacts to areas that support Johnson's seagrass critical habitat essential features are likely to result from the placement of shoreline stabilization materials and structures (Activity 1), the placement of pile-supported structures (Activity 2), removing sediments via dredging (Activity 3), and the placement of water-management outfall structures (Activity 4). These activities are likely to have a permanent and direct adverse effect on the stable, consolidated sediments that are free from physical disturbance essential feature of Johnson's seagrass critical habitat by placing materials on the sediment or by removing the sediment by dredging. A permanent loss of any 1 of the essential features renders the area incapable of supporting Johnson's seagrass. The placement of scientific survey devices (Activity 5) also will have a direct and adverse effect on the stable, consolidated sediments essential feature. Since PDC A5.2 limits the placement of scientific survey devices to no more than 24 months, we believe the adverse effects will be temporary and that the feature will be restored once the device is removed. We estimate that direct losses of 111,041.1 ft<sup>2</sup> (2.56 ac) of critical habitat will result from activities covered under this Opinion (Table 66).

<sup>&</sup>lt;sup>48</sup> 200 ft<sup>2</sup> is approximately 0.005 ac, which rounded to 2 decimal places is 0.00 ac.

	Activity	Loss of Critical Habitat (ft <sup>2</sup> )	Loss of Critical Habitat (ac)
1	Shoreline stabilization	89,226.54	2.05
2	Pile-supported structure	3,989.56	0.10
3	Dredging	16,275.00	0.37
4	Water-management outfall structures	700.00	0.02
5	Scientific survey devices	850	0.02
6	Boat ramps	-	
7	Aquatic enhancement	-	
8	Transmission/ utility lines	-	
9	Marine debris removal	-	
10	Temporary platforms, access fill, and cofferdams		
		111,041.1	2.56

Table 66. Impacts to Johnson's Seagrass Critical Habitat Anticipated per 5-year period

# 5.3 Effects to Smalltooth Sawfish Critical Habitat

Both of the essential features to smalltooth sawfish critical habitat (red mangroves and shallow, euryhaline habitat) are likely to be adversely affected from the projects analyzed in this Opinion. As discussed in Section 2.2, we used the USACE's assumptions regarding about shoreline stabilization projects (Activity 1), pile-supported structures (Activity 2), and stormwater outfall structures (Activity 4) to calculate the anticipated loss of smalltooth sawfish critical habitat essential features (i.e., red mangroves and shallow, euryhaline habitat) from the placement of these structures. The total anticipated impacts provided in Table 67 are explained in Section 2.2.1 for shoreline stabilization, Section 2.2.2 for pile-supported structures, and Section 2.2.4 for water-management outfall structures.

Using remote sensing data acquired from the Fish and Wildlife Research Institute, we were able to compile information relating to the total area of these essential features within smalltooth sawfish critical habitat. While the available mangrove and shallow, euryhaline essential features will be diminished by the proposed activities in this Opinion, we believe that individual activities spread out throughout smalltooth sawfish critical habitat and in areas not defined as high density pupping areas (hot spots) will not sever or prevent access to alternate refuge or forage areas in the area. Still, some ecological function provided to juvenile smalltooth sawfish in terms of the shallow-water and red mangrove essential features will be lost, as discussed further in Section 8.2.

Category of Activity		Impacts to Red Mangroves (lin ft)	Impacts to Shallow, Euryhaline Habitat (ft²)	
1	Shoreline stabilization	-	101,614.92	
2	Pile-supported structures	80	-	
3	Dredging	-	-	
4	Water-management outfall structures	258	1,900	
5	Scientific survey devices	-	-	
6	Boat ramps	-	-	
7	Aquatic enhancement	-	-	
8	Transmission/ utility lines	-	-	
9	Marine debris removal	-	-	
10	Temporary platforms, fill, and cofferdams	-	-	
	Total	338	$103,514.92 \text{ ft}^2 (2.38 \text{ ac})$	

Table 67. Impacts to Smalltooth Sawfish Critical Habitat Anticipated per 5-year period

### 6 Cumulative Effects

Cumulative effects include the effects of future state, tribal, or local private actions that are reasonably certain to occur in the action area considered in this Opinion. Future federal actions that are unrelated to the proposed actions are not considered in this section because they require separate consultation pursuant to Section 7 of the ESA.

# 6.1 Cumulative Effects to Johnson's Seagrass and Johnson's Seagrass Critical Habitat

No categories of effects beyond those already described are expected in the action area. Dock and marina construction will likely continue at current rates, with associated loss and degradation of seagrass habitat, including Johnson's seagrass. However, these activities are subject to USACE permitting and thus the ESA Section 7 consultation requirement. Some of the dock construction (i.e., the construction that meets the PDCs in this Opinion) will be covered under this Opinion, and must meet the requirements of this Opinion to protect the species and critical habitat. As we explained above, under the PDCs for Activity 2, all new docks or dock expansions in Johnson's seagrass critical habitat must be constructed with grated or plank decking that allows light transmission. New docks or dock expansions within the range of the species but outside of Johnson's seagrass critical habitat must meet those conditions unless there are no seagrasses within the property limits or there is seagrass, other than Johnson's seagrass. Upland development and associated runoff will continue to degrade water quality and decrease water clarity necessary for growth of seagrasses. Flood control and imprudent water management practices will continue to result in freshwater inputs into estuarine systems, thereby degrading water quality and altering salinity. Long-term, large-scale reduction in salinity has been identified as a potentially significant threat to the persistence and recovery of Johnson's seagrass.

Increased recreational vessel traffic will continue to result in damage to Johnson's seagrass and its designated critical habitat by improper anchoring, propeller scarring, and accidental groundings. However, we expect that ongoing boater education programs and posted signage about the dangers to seagrass beds from these practices may reduce impacts to Johnson's seagrass and its designated critical habitat.

# 6.2 Cumulative Effects to Smalltooth Sawfish Critical Habitat

Smalltooth sawfish habitat has been degraded or modified throughout the southeastern United States from agriculture, urban development, commercial activities, channel dredging, boating activities, and the diversion of freshwater runoff. While the degradation and modification of habitat is not likely the primary reason for the decline of smalltooth sawfish abundance and their contracted distribution, it has likely been a contributing factor.

The smalltooth sawfish critical habitat units will likely continue to experience the same types of actions described in the status of critical habitat in Section 3.2.2. These actions include shoreline stabilization removing shallow, euryhaline habitat and mangroves, pile-supported structures removing mangroves, and outfall structures removing mangroves. The additive effect of these actions to the essential features of critical habitat will continue to be assessed by USACE to ensure that they either meet the PDCs in this Opinion or are reviewed by NMFS on a project-level basis through the Section 7 process.

Many threats to smalltooth sawfish critical habitat are expected to be exacerbated by the effects of global climate change (see Section 3.2.2, *Threats to Critical Habitat*). Potential increases in sea level may impact the availability of nursery habitat, particularly shallow euryhaline and red mangrove lined, low-lying coastal habitats (IPCC 2014; Wanless et al. 2005). Red mangroves could be negatively affected by increased temperatures, salinities, and acidification of coastal waters (Scavia et al. 2002; Snedaker 1995; Wanless et al. 2005), as well as increased runoff and erosion due to the expected increase in extreme storm events (IPCC 2014; Wanless et al. 2005). These alterations of the marine environment due to global climate change could ultimately affect the distribution, physiology, and growth rates of red mangroves, potentially eliminating them from particular areas. The magnitude of these effects on smalltooth sawfish critical habitat are difficult to predict, yet the cyclical loss of habitat from extreme storm events combined with sea level rise may result in a decrease in juvenile survival (Norton et al. 2012; Scavia et al. 2002). However, the activities analyzed in this Opinion are limited by the PDCs to be of such a small scale, scope, and limited time frame that is not very likely to contribute to, or be affected cumulatively by climate change.

Smalltooth sawfish habitat has been degraded or modified throughout the southeastern United States from agriculture, urban development, commercial activities, channel dredging, boating activities, and the diversion of freshwater runoff. While the degradation and modification of habitat is not likely the primary reason for the decline of smalltooth sawfish abundance and their contracted distribution, it has likely been a significant factor.

#### 7 Jeopardy Analysis

The analyses conducted in the previous sections of this Opinion serve to provide a basis to determine whether the proposed actions would be likely to jeopardize the continued existence of Johnson's seagrass. In Section 5, we outlined how the proposed actions can affect these species. Now we turn to an assessment of the species response to these impacts, in terms of overall population effects, and whether those effects of the proposed actions, when considered in the context of the status of the species (Section 3), the environmental baseline (Section 4), and the cumulative effects (Section 6), will jeopardize the continued existence of the affected species.

This section evaluates whether the proposed actions are likely to jeopardize the continued existence of is defined as "to engage in an action that reasonably would be expected, directly or indirectly, to reduce appreciably the likelihood of both the survival and recovery of a listed species in the wild by reducing the reproduction, numbers, or distribution of that species" (50 CFR 402.02). Thus in making this determination, NMFS must look at whether the proposed actions directly or indirectly reduce the reproduction, numbers, or distribution of a listed species. Then if there is a reduction in 1 or more of these elements, we evaluate whether it would be expected to cause an appreciable reduction in the likelihood of both the survival and the recovery of the species. Section 5 ("Effects of the Action") describes the effects of the proposed actions on species, and the extent of those effects in terms of an estimate of the area of impacts to Johnson's seagrass. In Section 5, we determined that 44,057 ft<sup>2</sup> (1.01 ac) of Johnson's seagrass could be impacted by activities authorized per 5-year period using this Opinion.

#### 7.1 Johnson's Seagrass

As noted in Section 5, we believe that up 22,378.96 ft<sup>2</sup> (0.51 ac) of Johnson's seagrass is likely to be adversely affected by the by pile-supported structures resulting in docks placed on or over Johnson's seagrass (Activity 2), dredging removing areas that support Johnson's seagrass (Activity 3), and utility line repairs removing areas that support Johnson's seagrass (Activity 8).projects authorized per 5-year period using this Opinion. We believe that Johnson's seagrass is likely to recolonize the impacted areas based on its life history strategy (i.e., it effectively outcompetes other seagrass species in periodically disturbed areas) (Durako et al. 2003). The loss of up to 22,378.96ft<sup>2</sup> (0.51 ac) of Johnson's seagrass is a reduction in numbers of the species; however, the Recovery Team has determined that effects of these types of activities are generally local and small-scale in nature and will not have larger, population scale effects. We do not consider such impacts threats to the survival of the species because these activities will not individually or cumulatively result in the long-term, large-scale mortality of Johnson's seagrass, particularly in light of its "pulsating patches" life history strategy, which allows the species to acclimate readily to disturbed areas. The loss of up to 22,378.96 ft<sup>2</sup> (0.51 ac) of Johnson's seagrass will not result in long-term mortality either in the immediate action area of each project or on a larger scale within the range of Johnson's seagrass.

Reproduction will be temporarily reduced at each project site that reduces Johnson's seagrass numbers, but NMFS does not believe that this reproductive loss appreciably reduces the likelihood of survival of Johnson's seagrass in the wild. Johnson's seagrass will continue to reproduce and spread because the proposed impacts are expected to be temporary (i.e., Johnson's seagrass is likely to recolonize the disturbed area and persist in other areas of the action area after the project is complete).

The proposed actions will not result in a complete reduction of Johnson's seagrass distribution or fragmentation of the range since we expect Johnson's seagrass will recolonize the disturbed areas and will continue to be capable of spreading via asexual fragmentation after the completion of pile-supported structures, dredging, and transmission/utility line projects covered under this Opinion since the location of these activities are expected to be spread out over all of Johnson's seagrass range and critical habitat. Therefore, the reproductive potential of the species in the action area will persist.

Recovery for Johnson's seagrass, as described in the recovery plan (NMFS 2002), will be achieved when the following recovery objectives are met: (1) the species' present geographic range remains stable for at least 10 years or increases; (2) self-sustaining populations are present throughout the range at distances less than or equal to the maximum dispersal distance to allow for stable vegetative recruitment and genetic diversity; and (3) populations and supporting habitat in its geographic range have long-term protection (through regulatory action or purchase acquisition). We review the project's expected impacts on critical habitat to determine whether it will be able to continue to provide its intended functions in achieving these recovery objectives.

The first recovery criterion for Johnson's seagrass is for its present range to remain stable for 10 years or to increase during that time. NMFS's 5-year review (2007a) of the status of the species concluded that the first recovery objective had been achieved as of 2007. In fact, the range had increased slightly northward at that time, and we have no information indicating range stability has decreased since then. We believe that the loss of up to 22,378.96ft<sup>2</sup> (0.51 ac) of Johnson's seagrass from pile-supported structures, dredging, and transmission/utility line projects covered under this Opinion will occur throughout the range of Johnson's seagrass and will not impede this recovery objective. These effects will not reduce or destabilize the present range of Johnson's seagrass. The loss of Johnson's seagrass from each of these activities is expected to be small and occur in individual, non-connected areas. In the case of the maintenance dredging and repair or replacement utility projects, if Johnson's is present, it is because it has regrown in this area since the last dredging or utility line installation and therefore is likely to recolonize this area again. The pile-supported projects will be required to provide decking that allows light attenuation and therefore is likely to allow at least some seagrass growth to persist or recruit to these areas. Hence, these projects will not reduce the range of the species.

The second recovery criterion for Johnson's seagrass requires that self-sustaining populations be present throughout the range at distances less than or equal to the maximum dispersal distance for the species. Drifting fragments of Johnson's seagrass can remain viable in the water column for 4-8 days (Hall et al. 2006), and can travel several kilometers under the influence of wind, tides, and waves. Because of this, we believe that the removal of seagrasses from pile-supported structures, dredging, and transmission/utility line projects covered under this Opinion will not break up self-sustaining populations and that seagrass fragments will be able to drift to and over these impacted project sites. Therefore, we believe the loss of Johnson's seagrass associated with the proposed actions will not impede the recovery criterion.

The final recovery criterion is for populations and supporting habitat in the geographic range of Johnson's seagrass to have long-term protection (through regulatory action or purchase acquisition). Though the affected project sites will not be available for the long-term, the total acreage of critical habitat for Johnson's seagrass range-wide is roughly 22,574 ac (NMFS 2002) and we assume the majority is still available for long-term protection, which would include areas surrounding the action area. Therefore, we conclude that the proposed actions' adverse effects on the essential features of Johnson's seagrass critical habitat will not impede achieving the recovery objectives listed above.

NMFS believes that the proposed actions will not appreciably reduce the likelihood of recovery of Johnson's seagrass in the wild. NMFS's 5-year review (2007a) of the status of the species concluded that the first recovery objective has been achieved. In fact, the range has increased slightly northward. The proposed actions will not impact the status of this objective. Self-sustaining populations are present throughout the range of the species. The species' overall reproductive capacity will be only minimally reduced by the reduction in Johnson's seagrass numbers and reproduction resulting from the proposed actions. The proposed actions will not lead to separation of self-sustaining Johnson's seagrass patches to an extent that might lead to adverse effects to 1 or more patches of the species. Similarly, the proposed actions are not likely to adversely affect the availability of suitable habitat in which the species can spread/flow in the future. While additional individual impacts may continue to occur, over the last decade the species has not demonstrated any declining trends. The proposed actions will not appreciably reduce the likelihood of recovery of Johnson's seagrass in the wild.

#### 8 Destruction and Adverse Modification Analysis

NMFS' regulations define *Destruction or adverse modification* to mean a direct or indirect alteration that appreciably diminishes the value of critical habitat for the conservation of a listed species. Such alterations may include, but are not limited to, those that alter the physical or biological features essential to the conservation of a species or that preclude or significantly delay development of such features. 50 CFR § 402.02. Other alterations that may destroy or adversely modify critical habitat may include impacts to the area itself, such as those that would impede access to or use of the essential features. We intend the phrase "significantly delay" in development of essential features to encompass a delay that interrupts the likely natural trajectory of the development of physical and biological features in the designated critical habitat to support the species' recovery. NMFS will generally conclude that a Federal action is likely to "destroy or adversely modify" designated critical habitat if the action results in an alteration of the quantity or quality of the essential physical or biological features of designated critical habitat, or that precludes or significantly delays the capacity of that habitat to develop those features over time, and if the effect of the alteration is to appreciably diminish the value of critical habitat for the conservation of the species.

This analysis takes into account the geographic and temporal scope of the proposed actions, recognizing that "functionality" of critical habitat necessarily means that it must now and must continue in the future to support the conservation of the species and progress toward recovery. Destruction or adverse modification does not depend strictly on the size or proportion of the area

adversely affected, but rather on the role the action area serves with regard to the function of the overall critical habitat, and how that role is affected by the action.

# 8.1 Johnson's Seagrass Critical Habitat

Ultimately, we seek to determine if, with the implementation of the proposed actions, critical habitat would remain functional (or retain the current ability for the essential features to become functional) to serve the intended conservation role for the species. This analysis takes into account the geographic and temporal scope of the proposed action, recognizing that "functionality" of critical habitat necessarily means that it must now and must continue in the future to support the conservation of the species and progress toward recovery. Recovery for Johnson's seagrass, as described in the recovery plan, will be achieved when the following recovery objectives are met: (1) the species' present geographic range remains stable for at least 10 years, or increases; (2) self-sustaining populations<sup>49</sup> are present throughout the range at distances less than or equal to the maximum dispersal distance to allow for stable vegetative recruitment and genetic diversity; and (3) populations and supporting habitat in its geographic range have long-term protection (through regulatory action or purchase acquisition).

The first recovery criterion for Johnson's seagrass is for its present range to remain stable for 10 years or to increase during that time. The loss of 111,041.1 ft<sup>2</sup> (2.56 ac) of designated critical habitat for Johnson's seagrass is not likely to impede the first recovery criterion. NMFS's 5-year review (2007a) of the status of the species concluded that the first recovery objective has been achieved. In fact, the range has increased slightly northward. The proposed action will not impact the status of this objective and will not impact the small area within the action area where Johnson's seagrass exists.

The second recovery criterion for Johnson's seagrass requires that self-sustaining populations be present throughout the range at distances less than or equal to the maximum dispersal distance for the species. Due to its asexual reproductive mode, self-sustaining populations are present throughout the range of the species. The loss of critical habitat from direct construction impacts will not significantly reduce the available critical habitat such that could significantly impact Johnson's seagrass self-sustaining populations by adversely affecting the availability of suitable habitat in which the species can spread/flow in the future. Drifting fragments of Johnson's seagrass can remain viable in the water column for 4-8 days (Hall et al. 2006), and can travel several kilometers under the influence of wind, tides, and waves. Because of this, we believe that the removal of Johnson's seagrass and Johnson's seagrass critical habitat from projects spread out of in both space and the time of construction will not break up self-sustaining populations. Therefore, we believe the loss of Johnson's seagrass critical habitat associated with the proposed actions will not impede the recovery criterion requiring that self-sustaining populations be present throughout the range at distances less than or equal to the maximum dispersal distance for the species.

<sup>&</sup>lt;sup>49</sup> Self-sustaining population is a population that has been documented to persist for at least 10 years.

The final recovery criterion is for populations and supporting habitat in the geographic range of Johnson's seagrass to have long-term protection (through regulatory action or purchase acquisition). Though the affected portions of the project sites will not be available for the long-term, thousands of acres of designated critical habitat are still available for long-term protection, which would include areas surrounding the action area. Therefore, we conclude that the proposed actions adverse effects on Johnson's seagrass critical habitat will not impede achieving the recovery objectives listed above and will therefore not appreciably reduce the species' likelihood of recovery in the wild.

In Section 5.2, we determined that the proposed project will result in the combined loss of 111,041.1 ft<sup>2</sup> (2.56 ac) of designated critical habitat for Johnson's seagrass from installation of shoreline stabilization (Activity 1), pile-supported structures (Activity 2), dredging (Activity 3), outfall structures (Activity 4), and scientific survey devices (Activity 5) structures authorized per 5-year period. We believe the placement of materials associated with Activities 1, 2, 4, and 5 on the sediment and the dredging of the sediment (Activity 3) will have a direct adverse effect to the stable, consolidated sediments that are free from physical disturbance essential feature. The loss associated with Activities 1-4 is permanent; however, as we explained above, the loss associated with the placement of scientific survey devices (850  $\text{ft}^2$ , 0.02 ac) is temporary because PDC A5.3 limits the placement of scientific survey devices to no more than 24 months and we believe the feature will be restored once the device is removed. We do not believe these losses will impede the third recovery objective. As discussed in Section 3.2.1, there are approximately 22,574 ac of Johnson's seagrass critical habitat. Most of the projects within critical habitat that we have consulted on have resulted in impacts to less than 1,000 ft<sup>2</sup> each. The loss of 111,041.1ft<sup>2</sup> (2.56 ac) of designated critical habitat for Johnson's seagrass from the activities covered under this Opinion would equate to a loss of 0.01% of Johnson's seagrass critical habitat (2.56 ac loss x100/22,574 total ac = 0.01%).

Therefore, we conclude that the proposed action's adverse effects on the essential features of Johnson's seagrass critical habitat will not impede achieving the recovery objectives listed above.

#### 8.2 Smalltooth Sawfish Critical Habitat

In designating critical habitat for the smalltooth sawfish, we explained that the key conservation objective for the species is to facilitate recruitment of juveniles into the adult population by protecting juvenile areas. We determined that the habitat features essential to achieving that conservation objective are (1) shallow, euryhaline habitats characterized by water depths between the MHWL and 3 ft (0.9 m) measured at MLLW and (2) red mangrove shorelines. These essential features are necessary to facilitate recruitment of juveniles into the adult population because they provide for predator avoidance and habitat for prey in the areas currently being used as juvenile nursery areas. Impacts to designated critical habitat, thus, have the potential to destabilize recovery efforts and impede chances for recovery. The critical habitat designation for smalltooth sawfish is divided into 2 units in southwest Florida where the physical features essential to the species' conservation can be protected from destruction or adverse modification: the CHEU and TTIEU. Activities covered under this Opinion can occur in both the CHEU and TTIEU.

The smalltooth sawfish recovery plan identifies 3 recovery objectives (NMFS 2009). Recovery Objective #1 is to minimize human interactions and associated injury and mortality; this objective is not relevant to critical habitat. Recovery Objective #2 is to protect and/or restore smalltooth sawfish habitats. Recovery Objective #3 is to ensure smalltooth sawfish abundance increases substantially and the species reoccupies areas from which it had previously been extirpated. Our analysis evaluates whether the anticipated impacts to critical habitat associated with the proposed action would interfere with the conservation objective behind the designated critical habitat—that is, facilitation of juvenile recruitment into a recovering adult population.

### 8.2.1 Protect and Restore Smalltooth Sawfish Habitat (Recovery Objective #2)

In establishing Recovery Objective #2, we recognized that recovery and conservation of smalltooth sawfish depends on the availability and quality of nursery habitats. Historically, juvenile sawfish were documented in mangrove and non-mangrove habitat in the southeastern United States. Due to the protections provided by the Ten Thousand Islands National Wildlife Refuge, Everglades National Park, and the FKNMS, much of the historic juvenile smalltooth sawfish habitat in southwest Florida has remained high-quality juvenile habitat. Recovery Regions G, H, and I in southwest Florida extend from the Manatee River on the west coast of Florida south through Everglades National Park and the Florida Keys to Caesar Creek on the southeast coast of Florida. The CHEU is in Recovery Region G and the TTIEU is in Recovery Unit H. Below we estimate the percent impact the proposed action will have on the shallow, euryhaline habitat and red mangrove shoreline essential features of critical habitat within the CHEU since the USACE was unable to provide an estimated number of projects it expects to occur in the TTIEU and since we assume very few, if any, projects will occur in the TTIEU. Most of the TTIEU consists of protected areas, including protected parks, in which development is not allowed. In addition, in the few areas within the TTIEU where development can occur (e.g., Goodland or Everglades City), USACE-approved projects have been small in scale and infrequent. We believe that it is more conservative to assume that all of the loss of critical habitat features is concentrated in CHEU since so few projects are likely to occur in the few communities in TTIEU that can be developed. In addition, since the CHEU has already experienced most of the loss of critical habitat essential features of both critical habitat units, analyzing additional continued loss of critical habitat essential features to this more degraded critical habitat unit is also more conservative.

The recovery plan states that for the three recovery regions with remaining high-quality habitats (i.e., Recovery Regions G, H, and I), juvenile habitats "must be maintained over the long term at or above 95% of the acreage available at the time of listing" (NMFS, 2009). To ensure that a proposed action will not impede Recovery Objective #2, we determine whether the critical habitat unit will be able to maintain 95% of its designated critical habitat after taking into account project impacts in the context of the status of the critical habitat, the environmental baseline, and cumulative effects. While the CHEU is only a part of the larger Recovery Region G, the 95% protection threshold applies across not just Recovery Region G, but also Recovery Regions H and I, we believe it is appropriate to consider whether 95% of each of the essential features of critical habitat in the CHEU is maintained. The 95% threshold seeks to protect nursery areas and the CHEU critical habitat unit contains the only known nursery areas within Recovery Region G. Functioning critical habitat contains either one or both of the essential features (shallow, euryhaline habitat and red mangroves). Therefore, below we estimate the

percent impact the proposed action will have on the shallow, euryhaline habitat and red mangrove essential features of critical habitat within the CHEU.

# Shallow, Euryhaline Essential Feature Impacts

NMFS estimated that that 84,480 ac of shallow, euryhaline habitat (abbreviated SH throughout this section) was available within the CHEU at the effective date of species listing (i.e., May 1, 2003) (Table 68, Line 1). As discussed above, we must determine whether a proposed action's impact will interfere with long-term maintenance of this essential feature at or above 95% of the acreage available at the time of listing; however, loss of critical habitat was not formally monitored until the effective date of critical habitat designation (i.e., October 2, 2009). Therefore, we must estimate habitat loss that occurred during the period between the effective date of species listing and the effective date of critical habitat designation (i.e., May 1, 2003 – October 2, 2009).

To do this, we use the dataset of completed Section 7 consultations (October 3, 2009 – September 30, 2016) to generate a rate of loss that can then be used as a proxy to back-calculate the loss of SH between the effective date of species listing and the effective date of critical habitat designation. We rely on this dataset because using approximately 7 years of information helps avoid over- or under-estimating the rate of habitat loss due to any potential inter-annual variability associated with economic growth and contraction that may have occurred in that time. From October 3, 2009, to September 30, 2016 (i.e., 84 months), NMFS completed a number of Section 7 consultations on proposed actions within the CHEU that resulted in the loss of 16.73 ac of SH.

Based on these losses, we estimate a monthly loss rate of SH in the CHEU using the following equation:

Monthly loss rate of SH (CHEU) = SH lost through federal agency actions  $\div$ 84 months Monthly loss rate of SH (CHEU) = 16.73 ac  $\div$  84 months Monthly loss rate of SH (CHEU) = 0.20 ac per month

Assuming the same monthly loss rates, we back-calculate the loss of SH in the 77 months between the effective date of species listing and the effective date of critical habitat designation (i.e., May 1, 2003 – October 2, 2009) for the CHEU using the following equation:

SH lost prior to critical habitat designation (CHEU) = 0.20 ac per month  $\times$  77 months SH lost prior to critical habitat designation (CHEU) = 15.40 ac (Table 68, Line 2)

Next, we determine the loss of SH since the effective date of critical habitat designation. From October 2, 2009, through September 30, 2017,<sup>50</sup> NMFS completed Section 7 consultations on projects within the CHEU that resulted in the loss of 19.72 ac of SH (Table 68, Line 3). While this number only takes into account projects with a federal nexus requiring ESA Section 7 consultation, there are very few projects without a federal nexus that could affect shallow, euryhaline habitat in the CHEU as most in-water construction projects require federal authorization.

Using this information, we calculate the SH currently available in CHEU using the following equation:

SH currently available (CHEU) = SH at time of species listing - (SH lost prior to critical habitat designation + SH lost since critical habitat designation) SH currently available (CHEU) = 84,480 ac - (15.40 ac + 19.72 ac) SH currently available (CHEU) = 84,444.88 ac (Table 68 Line 4)

We calculate the amount of SH that must be maintained in the CHEU using the following equation:

SH that must be maintained (CHEU) = SEH at time of species listing  $\times$  95% SH that must be maintained (CHEU) = 84,480 ac  $\times$  0.95 SH that must be maintained (CHEU) = 80,256 ac (Table 68 Line 5)

The activities anticipated to be covered under this Opinion per 5-year period would result in the permanent loss of 103,514.92 ft<sup>2</sup> (2.38 ac) (Table 68 Line 6). Using the above results, we estimate the impact of the proposed action in addition to the SH lost in CHEU since the effective date of species listing using the following equation:

% SH lost since species listing (CHEU)

= [(SH loss due to these projects

+ SH lost prior to critical habitat designation

+ SH lost since critical habitat designation)

 $\div$  Total SH at time of species listing]  $\times$  100

% SH lost since species listing (CHEU)

 $= [(2.38 ac + 15.40 ac + 19.72 ac) \div 84,480 ac] \times 100$ 

% SH lost since species listing (CHEU) =  $(37.50 ac \div 84, 480 ac) \times 100$ 

% SH lost since species listing (CHEU) = 0.04% (Table 68, Line 7)

<sup>&</sup>lt;sup>50</sup> Due to the small number of monthly projects affecting smalltooth sawfish critical habitat, NMFS updates shallow, euryhaline habitat losses quarterly based on the U.S. federal fiscal year (December 31, Mar 31, June 30, September 30).

Thus, we estimate the percent of SH remaining within the CHEU as:

% SH remaining (CHEU) = 100% – % SH lost since species listing (CHEU) % SH remaining (CHEU) = 100% – 0.04% (Table 68, Line 7) % SH remaining (CHEU) = 99.96% (Table 68, Line 8)

Sha	allow, Euryhaline Habitat in the CHEU	Acres		
1.	Available at the time of species listing	84,480		
2.	Losses prior to critical habitat designation	15.40		
3.	Losses since critical habitat designation (through federal agency actions)	19.72		
4.	Available as of September 30, 2017	84,444.88		
5.	Area that must be maintained per Recovery Plan	80,256 (95% of 84,480)		
6.	Affected by the proposed actions	2.38		
7.	Affected since species listing	37.50 (0.04% of 84,480)		
8.	Remaining	84,442.50 (99.96% of 84,480)		

Table 68. Summary	y of Impacts to	the Shallow, Euryhalin	ne Habitat Essential Feature
-------------------	-----------------	------------------------	------------------------------

#### Red Mangrove Essential Feature Impacts

Remote sensing data from FWRI indicated that approximately 5,512,320 lin feet of red mangrove shoreline (abbreviated RM throughout this section) was available in the CHEU at the effective date of species listing (i.e., May 1, 2003) (Table 69 Line 1). As described above, we must determine whether project impacts will interfere with long-term maintenance of this essential feature at or above 95% of the linear feet of habitat available at the time of listing; however, as described above, loss of critical habitat was not formally monitored until the effective date of critical habitat designation (i.e., October 2, 2009). Therefore, we must estimate habitat loss that occurred during the period between the effective date of species listing and the effective date of critical habitat designation (i.e., May 1, 2003 – October 2, 2009).

To do this, we use the dataset of completed Section 7 consultations (October 3, 2009 – September 30, 2016) to generate a rate of loss that can then be used as a proxy to back-calculate the loss of RM between the effective date of species listing and the effective date of critical habitat designation. We rely on this dataset because using approximately 7 years of information helps avoid over- or under-estimating the rate of habitat loss due to any potential inter-annual variability associated with economic growth and contraction that may have occurred in that time. From October 3, 2009, to September 30, 2016 (i.e., 84 months), NMFS completed a number of Section 7 consultations on proposed actions within the CHEU that resulted in the loss of 9,142.50 lin ft of red mangrove shoreline.

Based on these losses, we estimated a monthly loss rate of RM using the following equation:

Monthly loss rate of RM (CHEU) = RM lost through federal agency actions ÷ 84 months Monthly loss rate of RM (CHEU) = 9,142.50 lin ft ÷ 84 months Monthly loss rate of RM (CHEU) = 108.84 lin ft per month

Assuming the same monthly loss rates, we back-calculate the loss of RM in the 77 months between the effective date of species listing and the effective date of critical habitat designation (i.e., May 1, 2003 – October 2, 2009) in the CHEU using the following equation:

RM loss prior to critical habitat designation (CHEU) = 108.84 lin ft per month × 77 months RM loss prior to critical habitat designation (CHEU) = 8,380.68 lin ft (Table 69, Line 2)

Next, we determine the loss of RM since the effective date of critical habitat designation. From the October 2, 2009, through September 30, 2017,<sup>51</sup> NMFS completed Section 7 consultations on projects within the CHEU that resulted in the loss of 11,818.75 lin ft of red mangroves (Table 69, Line 3). While this number only takes into account projects with a federal nexus requiring ESA Section 7 consultation, there are very few projects without a federal nexus that could affect red mangrove shoreline in the CHEU as most in-water construction projects require federal authorization.

Using this information, we calculate the RM currently available in the CHEU using the following equation:

## RM currently available (CHEU)

= RM at time of species listing - (RM loss prior to critical habitat designation + RM loss since critical habitat designation) RM currently available (CHEU) = 5,512,320 lin ft - (8,380.68 lin ft + 11,818.75 lin ft) RM currently available (CHEU) = 5,492,120.57 lin ft (Table 69, Line 4)

We calculated the amount of RM that must be maintained in the CHEU using the following equation:

RM that must be maintained (CHEU) = RM at time of species listing  $\times$  95% RM that must be maintained (CHEU) = 5,512,320 lin ft  $\times$  0.95 RM that must be maintained (CHEU) = 5,236,704 lin ft (Table 69, Line 5)

<sup>&</sup>lt;sup>51</sup> Due to the small number of monthly projects affecting smalltooth sawfish critical habitat, NMFS updates red mangrove shoreline losses quarterly based on the U.S. federal fiscal year (December 31, March 31, June 30, September 30).

The activities anticipated to be covered under this Opinion per 5-year period would result in the loss of 338 lin ft of the estimated 5,512,320 lin ft of RM (Table 69, Line 6). Using the above results, we estimated the impact of the proposed action in addition to the RM lost in CHEU since the effective date of species listing using the following equation:

% RM lost in CHEU since species listing

= [(RM loss due to these projects

+ RM lost prior to critical habitat designation

+ RM lost since critical habitat designation)

÷ Total RM in CHEU at time of species listing] × 100

% RM lost in CHEU since species listing

 $= [(338 lin ft + 8,380.68 lin ft + 11,818.75 lin ft) \div 5,512,320 lin ft] \times 100$ 

% RM lost in CHEU since species listing =  $(20,537.43 \text{ lin } ft \div 5,512,320 \text{ lin } ft) \times 100$ % RM lost in CHEU since species listing = 0.37 % (Table 69, Line 7)

Thus, we estimate the percent of RM remaining within the CHEU as:

% RM remaining (CHEU) = 100% - % RM lost since species listing (CHEU) % RM remaining (CHEU) = 100% - 0.37% (Table 69, Line 7) % RM remaining (CHEU) = 99.63% (Table 69, Line 8)

Red Mangrove Shoreline in the CHEU		Linear Feet		
1.	Available at the time of species listing	5,512,320		
2.	Losses prior to critical habitat designation	8,380.68		
3.	Losses since critical habitat designation (through federal agency actions)	11,818.75		
4.	Available as of September 30, 2017	5,492,120.57		
5.	Linear feet that must be maintained per Recovery Plan	5,236,704 (95% of 5,512,320)		
6.	Affected by the proposed actions	338		
7.	Affected since species listing	20,537.43 (0.37% of 5,512,320)		
8.	Remaining	5,491,782.57 (99.63% of 5,512,320)		

Table 69.	Summar	of Im	pacts to	the R	led Ma	angrove	Essential	Feature

#### Summary of Impacts to the Essential Features

A very small percentage of the essential features of smalltooth sawfish designated critical habitat have been affected by in-water construction since the effective date of species listing. Including losses from this project, 99.96% of the SH essential feature (Table 68, Line 8) and 99.63% of the RM essential feature (Table 69, Line 8) available at the time of species listing remain in the CHEU. Thus, the loss of essential features associated with the proposed action, in combination with losses since we listed the species, does not provide any impediment to effectively protecting

95% of juvenile habitat in the CHEU available at the effective date of species listing, and therefore will not be an impediment to Recovery Objective #2.

# 8.2.2 Ensure Smalltooth Sawfish Abundance Increases (Recovery Objective #3)

In establishing Recovery Objective #3, we recognized that it was important that sufficient numbers of juvenile sawfish inhabit several nursery areas across a diverse geographic area to ensure survivorship and growth and to protect against the negative effects of stochastic events within parts of their range. To meet this objective, Recovery Region G (i.e., CHEU) and Recovery Region H (i.e., TTIEU) must support sufficiently large numbers of juvenile sawfish to ensure that the species is viable in the long-term and can maintain genetic diversity. The recovery objective requires that the relative abundance of small juvenile sawfish (< 200 cm) either increases at an average annual rate of at least 5% over a 27-year period, or juvenile abundance is at greater than 80% of the carrying capacity of the recovery region.

Assessing the effect of the proposed action on small juvenile abundance is made difficult by the state of available data. Since the designation of critical habitat and the release of the recovery plan in 2009, ongoing studies have been in place to monitor the U.S. DPS of smalltooth sawfish. The Florida Fish and Wildlife Research Institute is conducting a study in the CHEU that is supported primarily under funding provided by NMFS through the Section 6 Species Recovery Grants Program, while the NOAA Southeast Fisheries Science Center's Panama City Laboratory and Florida State University have focused studies in the TTIEU. The intent of these studies is to determine the abundance, distribution, habitat use, and movement of juvenile sawfish. Given the limited duration of the study in CHEU (September 2009-current), there is not yet enough data to discern the trend in juvenile abundance within that Unit. Early indications are that juvenile sawfish are at least stable and likely increasing in the CHEU, due in large part to ESA-listing of the species and designation of critical habitat. While it may be too early to state definitively that juveniles within CHEU are surviving to adulthood, researchers consistently capture newborn smalltooth sawfish, particularly within "hot spots," indicating adult smalltooth sawfish are pupping within Recovery Region G. Available data from the adjacent Recovery Region H (i.e., TTIEU) indicate that adult smalltooth sawfish are also reproducing within this recovery region and that the juvenile population trend is at least stable and possibly increasing-though variability is high (Carlson and Osborne 2012; Carlson et al. 2007). With no other data to consider, the abundance trend in TTIEU represents the best data available for assessing the population trends as a whole. Therefore, we do not believe the loss of habitat associated with activities covered under this Opinion, in combination with the losses to date, will impede the 5% annual growth objective for the juvenile population within Recovery Region G or H.

## 9 Conclusion

After reviewing the current status of Johnson's seagrass, the environmental baseline, the effects of the proposed actions, and the cumulative effects, it is our opinion that the authorization of activities analyzed under this Opinion and the removal of up to 22,378.96 ft<sup>2</sup> (0.51 ac) of Johnson's seagrass over a 5-year period will not reduce appreciably the likelihood of the species' survival or recovery, despite permanent adverse effects. Given the nature of the project and the information provided above, we conclude that the actions, as proposed, are likely to adversely affect but are not likely to jeopardize Johnson's seagrass.

After reviewing the current status of Johnson's seagrass critical habitat, the environmental baseline, the effects of the proposed actions, and the cumulative effects, it is our opinion that the authorization of activities analyzed under this Opinion and the removal of up to 111,041.1 ft<sup>2</sup> (2.56 ac), of which all but 850 ft<sup>2</sup> (0.02 ac) is permanent, of Johnson's seagrass critical habitat over a 5-year period will not diminish the value of the critical habitat for the conservation of the species, despite permanent adverse effects. Given the nature of the project and the information provided above, we conclude that the actions, as proposed, are likely to adversely affect but are not likely to destroy or adversely modify Johnson's seagrass critical habitat.

After reviewing the current status of smalltooth sawfish critical habitat, the environmental baseline, the effects of the proposed actions, and the cumulative effects, it is our opinion that the authorization of activities analyzed under this Opinion and the resulting removal of up to 103,514.92 ft<sup>2</sup> (2.38ac) of shallow, euryhaline habitat and 338 lin ft of red mangrove habitat essential features over a 5-year period will not interfere with achieving the relevant habitat-based Recovery Objectives for smalltooth sawfish. Therefore, we conclude the proposed action will not impede the critical habitat's ability to support the smalltooth sawfish's conservation, despite permanent adverse effects. Given the nature of the proposed action and the information provided above, we conclude that the action, as proposed, is likely to adversely affect, but is not likely to destroy or adversely modify, smalltooth sawfish critical habitat.

It is important to note that the conclusions drawn in this Opinion are based on a series of assumptions (see Section 2.2). Because a programmatic by nature covers future actions that have not been specifically identified, the analysis is based on the actions that occurred during the last 5 years that was used by the USACE to estimate the number of actions that may be authorized in the future using this Opinion analysis requirement. Our analysis was based on a 5-year period to account for annual variability; however, the projects covered under this Opinion will continue to be reviewed on an annual basis and every 5 years so long as the Programmatic Opinion continues to be accurate. A series of assumptions are made based on the best available data, PDCs are in place to define the limits of the proposed actions (see Section 2.2), and project-level review (Section 2.3) and programmatic review (Section 2.4) and reporting is required to evaluate that assumptions about the projects covered under this Opinion and the expected effects of those projects are consistent with the analysis in this Opinion. If the assumptions are inaccurate (for example, if the USACE authorized more projects than expected, or the projects are of a different scale) or the effects are different than those expected under this Opinion, then consultation may need to be reinitiated. This determination as to whether to reinitiate consultation will be made at the programmatic review between USACE and NMFS.

## 10 Incidental Take Statement

NMFS does not anticipate that the proposed actions will incidentally take any species and no take is authorized. Nonetheless, any take of ESA-listed species shall be immediately reported to takereport.nmfsser@noaa.gov. Refer to the present Biological Opinion by title (JAXBO), issuance date, NMFS tracking number (SER-2015-17616), and USACE permit number (SAJ-2015-02575). At that time, consultation reinitiation will be necessary.

## 11 Conservation Recommendations

Section 7(a)(1) of the ESA directs federal agencies to use their authorities to further the purposes of the ESA by carrying out conservation programs for the benefit of endangered and threatened species. Conservation recommendations are discretionary agency activities to minimize or avoid adverse effects of a proposed actions on listed species or critical habitat, to help implement recovery plans, or to develop information.

NMFS believes the following conservation recommendations are reasonable, necessary, and appropriate to conserve and recover Johnson's seagrass critical habitat and smalltooth sawfish critical habitat. NMFS strongly recommends that these measures be considered and adopted. In order for us to be kept informed of actions minimizing or avoiding adverse effects or benefiting listed species or their habitats, we request notification of the implementation of any conservation recommendations.

## Smalltooth Sawfish Critical Habitat

- 1. Continue public outreach and education on smalltooth sawfish and smalltooth sawfish critical habitat, in an effort to minimize interactions, injury, and mortality.
- 2. Provide funding to conduct directed research on smalltooth sawfish that will help further our understanding about the species. In other words, implement a relative abundance monitoring program which will help define how spatial and temporal variability in the physical and biological environment influence smalltooth sawfish, in an effort to predict long-term changes in smalltooth sawfish distribution, abundance, extent, and timing of movements.
- 3. Conduct or support surveys to help acquire detailed bathymetry and mangrove coverage within smalltooth sawfish critical habitat.
- 4. Continue to request the removal of existing riprap from future seawall restoration/replacement activities within smalltooth sawfish critical habitat.
- 5. Provide funding to conduct directed research in an effort to develop new technology to support vertical seawalls other than riprap (e.g., living seawalls that incorporate mangroves).

## Johnson's Seagrass Critical Habitat

- 6. NMFS recommends that USACE conduct and support monitoring to assess trends in the distribution and abundance of Johnson's seagrass. Data collected should be contributed to Florida Fish and Wildlife Conservation Commission to support ongoing GIS mapping of Johnson's and other seagrass distribution.
- 7. NMFS recommends that USACE, in coordination with seagrass researchers and industry, support ongoing research on light requirements and transplanting techniques to preserve and restore Johnson's seagrass.

## Johnson's seagrass

8. NMFS recommends that the USACE conduct a survey on the effects of dock and vessel shading on Johnson's seagrass for docks at varying heights and decking materials.

## 12 Reinitiation of Consultation

As provided in 50 CFR Section 402.16, reinitiation of formal consultation is required where discretionary federal agency involvement or control over the action has been retained (or is authorized by law) and if (1) take occurs, (2) new information reveals effects of the action that may affect listed species or critical habitat in a manner or to an extent not previously considered, (3) the identified action is subsequently modified in a manner that causes an effect to listed species or critical habitat that was not considered in the Biological Opinion, or (4) a new species is listed or critical habitat designated that may be affected by the identified action.

## 13 Literature Cited

- Abal, E. G., N. Loneragan, P. Bowen, C. J. Perry, J. W. Udy, and W. C. Dennison. 1994. Physiological and morphological responses of the seagrass Zostera capricorni Aschers, to light intensity. Journal of Experimental Marine Biology and Ecology 178(1):113-129.
- AMEC Environment & Infrastructure, I. 2013. Monroe County Canal Management Master Plan (CMMP).
- Andrews, T. J., B. F. Clough, and G. J. Muller. 1984. Photosynthetic gas exchange properties and carbon isotope ratios of some mangroves in North Queensland. Pages 15-23 *in* H. J. Teas, editor. Physiology and Management of Mangroves volume 9. Dr. W. Junk Publishers.
- Antonini, G., R. Swett, S. Schulte, and D. Fann. 2000. Regional waterway management system for south Sarasota County.
- Appeldoorn, R., M. Schärer-Umpierre, T. Rowell, and M. Nemeth. 2013. Measuring relative density of spawning red hind (Epinephelus guttatus) from sound production: consistency within and among sites. Pages 284-286 *in* Proc Gulf Caribb Fish Inst.
- Ball, M. C., M. J. Cochrane, and H. M. Rawson. 1997. Growth and water use of the mangroves *Rhizophora apiculata* and *R. stylosa* in response to salinity and humidity under ambient and elevated concentrations of atmospheric CO<sub>2</sub>. Plant, Cell & Environment 20(9):1158-1166.
- Bejarano-Álvarez, M., F. Galvan-Magana, and R. I. Ochoa-Baez. 2011. Reproductive biology of the scalloped hammerhead shark Sphyrna lewini (Chondrichthyes: Sphyrnidae) off south-west Mexico. International Journal of Ichthyology 17(1):11-23.
- Bessudo, S., G. A. Soler, A. P. Klimley, J. T. Ketchum, A. Hearn, and R. Arauz. 2011. Residency of the scalloped hammerhead shark (Sphyrna lewini) at Malpelo Island and evidence of migration to other islands in the Eastern Tropical Pacific. Environmental Biology of Fishes 91(2):165-176.

- Boswell, J., J. Ott, and A. Birch. 2012. Charlotte Harbor National Estuary Program Oyster Habitat Restoration Plan. Charlotte Harbor National Estuary Program.
- Boudreaux, M. L., J. L. Stiner, and L. J. Walters. 2006. Biodiversity of Sessile and Motile Macrofauna on Intertidal, Oyster Reefs in Mosquito Lagoon, Florida. Journal of Shellfish Research 25(3):1079-1089.
- Bulthuis, D. A. 1983. Effects of in situ light reduction on density and growth of the seagrass Heterozostera tasmanica (Martens ex Aschers.) den Hartog in Western Port, Victoria, Australia. Journal of Experimental Marine Biology and Ecology 67(1):91-103.
- Burchett, M. D., S. Meredith, A. Pulkownik, and S. Pulkownik. 1984. Short term influences affecting growth and distribution of mangrove communities in the Sydney region. Wetlands (Australia) 4(2):10.
- Bush, A. 2003. Diet and diel feeding periodicity of juvenile scalloped hammerhead sharks, *Sphyrna lewini*, in Kāne'ohe Bay, Ō'ahu, Hawai'i. Environmental Biology of Fishes 67(1):1-11.
- Cahoon, D. R., P. Hensel, J. Rybczyk, K. L. McKee, C. E. Proffitt, and B. C. Perez. 2003. Mass tree mortality leads to mangrove peat collapse at Bay Islands, Honduras after Hurricane Mitch. Journal of Ecology 91(6):1093-1105.
- CALTRANS. 2007. Compendium of Pile Driving Sound Data. Report prepared by Illinworth and Rodkin, Inc.
- CALTRANS. 2009. Technical guidance for assessment and mitigation of the hydroacoustic effects of pile driving on fish, Sacremento.
- Carlson, J. K., and J. Osborne. 2012. Relative abundance of smalltooth sawfish (*Pristis pectinata*) based on the Everglades National Park Creel Survey. NOAA Technical Memorandum NMFS-SEFSC-626. U.S. Department of Commerce, National Oceanic and Atmospheric Administration, National Marine Fisheries Service, Southeast Fisheries Science Center, Panama City Laboratory, NOAA Technical Memorandum NMFS-SEFSC-626, Panama City, Florida.
- Carlson, J. K., J. Osborne, and T. W. Schmidt. 2007. Monitoring the recovery of smalltooth sawfish, *Pristis pectinata*, using standardized relative indices of abundance. Biological Conservation 136(2):195-202.
- Casper, B., M. Halvorsen, and A. Popper. 2012. Are Sharks Even Bothered by a Noisy Environment? Pages 93-97 in A. Popper, and A. Hawkins, editors. The Effects of Noise on Aquatic Life, volume 730. Springer New York.
- Casper, B. M., P. S. Lobel, and H. Y. Yan. 2003. The Hearing Sensitivity of the Little Skate, (Raja erinacea): A Comparison of Two Methods. Environmental Biology of Fishes 68(4):371-379.

- CITES. 2010. Consideration of proposals for amendment of appendices I and II (CoP15 Prop. 15). 15th meeting of the Conference of the Parties, Doha (Qatar). CITES.
- Compagno, L. J. V. 1984. Part 2. Carcharhiniformes. Pages 251-655 *in* FAO Species Catalogue. Sharks of the World. An Annotated and Illustrated Catalogue of Sharks Species Known to Date, volume 4. FAO.
- Cortés, E. 1999. Standardized diet compositions and trophic levels of sharks. ICES Journal of Marine Science: Journal du Conseil 56(5):707-717.
- Corwin, J. T. 1977. Morphology of the macula neglecta in sharks of the genus *Carcharhinus*. Journal of Morphology 152(3):341-361.
- Culter, J. K., and S. Mahadevan. 1982. Long-Term Effects of Beach Nourishment on the Benthic Fauna of Panama City Beach, Florida. Miscellaneous Report No. 82-2. Mote Marine Laboratory, Mote Technical Report No. 45, Sarasota, Florida.
- Dahl, T. E., and C. E. Johnson. 1991. Status and trends of wetlands in the conterminous United States, mid-1970s to mid-1980s. U.S. Fish and Wildlife Service, Washington, D.C.
- Dawes, C. J., C. S. Lobban, and D. A. Tomasko. 1989. A comparison of the physiological ecology of the seagrasses Halophila decipiens ostenfeld and H. Johnsonii eiseman from Florida. Aquatic Botany 33(1–2):149-154.
- Dennison, W. C. 1987. Effects of light on seagrass photosynthesis, growth and depth distribution. Aquatic Botany 27(1):15-26.
- DeRuiter, S. L., and K. D. Doukara. 2010. Loggerhead turtles dive in response to airgun sound exposure. Journal of the Acoustical Society of America 127(3 Part 2):1726.
- Di Carlo, G., F. Badalamenti, A. C. Jensen, E. W. Koch, and S. Riggio. 2005. Colonisation process of vegetative fragments of Posidonia oceanica (L.) Delile on rubble mounds. Marine Biology 147(6):1261-1270.
- Dickerson, C., D. G. Clarke, R. M. Engler, and K. J. Reine. 2001. Characterization of Underwater Sounds Produced by Bucket Dredging Operations. Engineer Research and Development Center, Vicksburg, MS.
- Diemer, K. M., B. Q. Mann, and N. E. Hussey. 2011. Distribution and movement of scalloped hammerhead Sphryna lewini and smooth hammerhead Sphyrna zygaena sharks along the east coast of southern Africa. African Journal of Marine Science 33(2):229-238.
- Duncan, K. M., and K. N. Holland. 2006. Habitat use, growth rates and dispersal patterns of juvenile scalloped hammerhead sharks *Sphyrna lewini* in a nursery habitat. Marine Ecology Progress Series 312:211-221.

- Durako, M. J., J. I. Kunzelman, W. J. Kenworthy, and K. K. Hammerstrom. 2003. Depth-related variability in the photobiology of two populations of *Halophila johnsonii* and *Halophila decipiens*. Marine Biology 142(6):1219-1228.
- Eiseman, N. J., and C. McMillan. 1980. A new species of seagrass, *Halophila johnsonii*, from the Atlantic coast of Florida. Aquatic Botany 9:15-19.
- Ellison, J. 2010. Vulnerability of Fiji's mangroves and associated coral reefs to climate change. A review., Suva, Fiji, WWF South Pacific Office.
- Fann, D. A., R. A. Swett, L. Carlin-Alexander, and G. A. Antonini. 2002. Regional waterway management system for Lee County, Phase 3. TD-5. Gainesville, FL: Florida Sea Grant College Program.
- Fay, R. R., and A. N. Popper. 2000. Evolution of hearing in vertebrates: The inner ears and processing. Hearing Research 149(1-2):1-10.
- FHA. 2012. Technical Guidance for Assessment and Mitigation of the Hydroacoustic Effects of Pile Driving on Fish. Final, ICF 645.10, ICF International, Seattle, WA.
- Field, C. D. 1995. Impact of expected climate change on mangroves. Hydrobiologia 295(1-3):75-81.
- Fonseca, M. S., W. Kenworthy, and P. E. Whitfield. 2000. Temporal dynamics of seagrass landscapes: a preliminary comparison of chronic and extreme disturbance events.
   Proceedings of the Fourth International Seagrass Biology Workshop. Biología Marina Mediterranea 7:373-376.
- Fox, D. A., J. E. Hightower, and F. M. Parauka. 2002. Estuarine and Nearshore Marine Habitat Use by Gulf Sturgeon from the Choctawhatchee River System, Florida. Pages 111-126 in American Fisheries Society Symposium. American Fisheries Society.
- Gallegos, C. L., and W. J. Kenworthy. 1996. Seagrass depth limits in the Indian River Lagoon (Florida, USA): Application of the optical water quality model. Estuarine, Coastal, and Shelf Science 42:267-288.
- Garmestani, A. S., and H. F. Percival. 2005. Raccoon removal reduces sea turtle nest depredation in the ten thousand islands of Florida. Southeastern Naturalist 4(3):469-472.
- Gilman, E. L., J. Ellison, N. C. Duke, and C. Field. 2008. Threats to mangroves from climate change and adaptation options: A review. Aquatic Botany 89(2):237-250.
- Gilman, E. L., J. Ellison, V. Jungblut, H. Van Lavieren, L. Wilson, F. Areki, G. Brighouse, J. Bungitak, E. Dus, and M. Henry. 2006. Adapting to Pacific Island mangrove responses to sea level rise and climate change. Climate Research 32:161-176.

- Gilmore, G. R. 1995. Environmental and Biogeographic Factors Influencing Ichthyofaunal Diversity: Indian River Lagoon. Bulletin of Marine Science 57(1):153-170.
- GMFMC. 1998. Generic amendment for addressing essential fish habitat requirements in the following Fishery Management plans of the Gulf of Mexico: Shrimp Fishery of the Gulf of Mexico, United States waters; Red Drum Fishery of the Gulf of Mexico, Reef Fish Fishery of the Gulf of Mexico, Coastal Migratory Pelagic Resources (Mackerel) in the Gulf of Mexico and South Atlantic; Stone Crab Fishery of the Gulf of Mexico; Spiny Lobster Fishery of the Gulf of Mexico; Coral and Coral Reefs of the Gulf of Mexico. Gulf of Mexico Fishery Management Council, Tampa, Florida.
- GMFMC. 2005. Generic Amendment 3 for addressing EFH requirements, HAPCs, and adverse effects of fishing in the following FMPs of the Gulf of Mexico: Shrimp, Red Drum, Reef Fish, Stone Crab, Coral and Coral Reefs in the GOM and Spiny Lobster and the Coastal Migratory Pelagic resources of the GOM and South Atlantic. Gulf of Mexico Fishery Management Council, Tampa, FL.
- Greening, H., and N. Holland. 2003. Johnson's Seagrass (*Halophila johnsonii*) Monitoring Workshop. Johnson's Seagrass Recovery Team, Florida Marine Research Insistute, St. Petersburg, FL.
- Hall, L. M., M. D. Hanisak, and R. W. Virnstein. 2006. Fragments of the seagrasses *Halodule wrightii* and *Halophila johnsonii* as potential recruits in Indian River Lagoon, Florida. Marine Ecology Progress Series 310:109-117.
- Halvorsen, M. B., B. M. Casper, F. Matthews, T. J. Carlson, and A. N. Popper. 2012a. Effects of Exposure to Pile-Driving Sounds on the Lake Sturgeon, Nile Tilapia and Hogchoker. Proceedings of the Royal Society B: Biological Sciences 279(1748):4705-4714.
- Halvorsen, M. B., B. M. Casper, C. M. Woodley, T. J. Carlson, and A. N. Popper. 2012b. Threshold for Onset of Injury in Chinook Salmon from Exposure to Impulsive Pile Driving Sounds. PLoS ONE 7(6):e38968.
- Hammerstrom, K., and W. Kenworthy. 2002. Investigating the existence of a Halophila johnsonii sediment seed bank. . NCCOS, NOS, NOAA, Center for Coastal Fisheries and Habitat Research. , Beaufort, North Carolina.
- Harty, C. 2004. Planning strategies for mangrove and saltmarsh changes in southeast Australia. Coastal Management 32(4):405-415.
- Hastings, M. C. 2010. Recommendations for Interim Criteria for Vibratory Pile Driving. Submitted to ICF Jones & Stokes, Sacramento, CA. For Task order on vibratory pile driving, Caltrans Contract 43A0228.
- Hastings, M. C., and A. N. Popper. 2005. Effects of sound on fish. California Department of Transportation, Sacramento, California.

- Hearn, A., J. T. Ketchum, A. P. Klimley, E. Espinoza, and C. Penaherrera. 2010. Hotspots within hotspots? Hammerhead shark movements around Wolf Island, Galapagos Marine Reserve. Marine Biology 157:1899-1915.
- Heidelbaugh, W. S., L. M. Hall, W. J. Kenworthy, P. Whitfield, R. W. Virnstein, L. J. Morris, and M. D. Hanisak. 2000. Reciprocal transplanting of the threatened seagrass *Halophila johnsonii* (Johnson's Seagrass) in the Indian River Lagoon, Florida. Pages 197-210 in S. A. Bortone, editor. Seagrasses: Monitoring, Ecology, Physiology and Management. CRC Press, Boca Raton.
- Hoegh-Guldberg, O., P. J. Mumby, A. J. Hooten, R. S. Steneck, P. Greenfield, E. Gomez, C. D. Harvell, P. F. Sale, A. J. Edwards, K. Caldeira, N. Knowlton, C. M. Eakin, R. Iglesias-Prieto, N. Muthiga, R. H. Bradbury, A. Dubi, and M. E. Hatziolos. 2007. Coral reefs under rapid climate change and ocean acidification. Science 318(5857):1737-1742.
- Holland, K. N., B. M. Wetherbee, J. D. Peterson, and C. G. Lowe. 1993. Movements and distribution of hammerhead shark pups on their natal grounds. Copeia (2):495-502.
- Hutchings, P. A., and P. Saenger. 1987. Ecology of Mangroves. St. Lucia, Queensland, Australia; New York: University of Queensland Press.
- IPCC. 2007. Climate Change 2007: Climate Change Impacts, Adaptation and Vulnerability.
  Summary for Policymakers. S. Solomon, D. Quin, M. Manning, Z. Chen, M. Marquis, K.
  B. Averyt, M. Tignor, and H. L. Miller, editors. Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the IPPC (Intergovernmental Panel on Climate Change). Cambridge University Press, Cambridge, UK and New York, NY.
- IPCC. 2013. Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Pages 1535 in T. F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex, and P. M. Midgley, editors. Cambridge University Press, Cambridge, United Kingdom; New York, NY.
- IPCC. 2014. Climate change 2014: Impacts, adaptation, and vulnerability. IPCC Working Group II contribution to AR5. Intergovernmental Panel on Climate Change.
- Jewett-Smith, J., C. McMillan, W. J. Kenworthy, and K. Bird. 1997. Flowering and genetic banding patterns of *Halophila johnsonii* and conspecifics. Aquatic Botany 59(3-4):323-331.
- Júnior, T. V., C. M. Vooren, and R. P. Lessa. 2009. Feeding strategy of the night shark (*Carcharhinus signatus*) and scalloped hammerhead shark (*Sphyrna lewini*) near seamounts off northeastern Brazil. Brazilian Journal of Oceanography 57(2):97-104.

- Kahn, A. E., and M. J. Durako. 2008. Photophysiological responses of *Halophila johnsonii* to experimental hyposaline and hyper-CDOM conditions. Journal of Experimental Marine Biology and Ecology 367(2):230-235.
- Kelly, J. C., and D. R. Nelson. 1975. Hearing thresholds of the horn shark, Heterodontus francisci. The Journal of the Acoustical Society of America 58(4):905-909.
- Kenworthy, W., and M. Fonseca. 1996. Light requirements of seagrasses *Halodule wrightii* and *Syringodium filiforme* derived from the relationship between diffuse light attenuation and maximum depth distribution. Estuaries and Coasts 19(3):740-750.
- Kenworthy, W., and D. E. Haunert. 1991. The light requirements of seagrasses: proceedings of a workshop to examine the capability of water quality criteria, standards, and monitoring programs to protect seagrasses. NOAA Technical Memorandum NMFS-SEFSC-287.
- Kenworthy, W., S. Wyllie-Echeverria, R. Coles, G. Pergent, and C. Pergent-Martini. 2006. Seagrass Conservation Biology: An Interdisciplinary Science for Protection of the Seagrass Biome. Pages 595-623 in A. W. D. Larkum, R. J. Orth, and C. M. Duarte, editors. SEAGRASSES: BIOLOGY, ECOLOGYAND CONSERVATION. Springer Netherlands.
- Kenworthy, W. J. 1993. The distribution, abundance, and ecology of Halophila johnsonii Eiseman in the lower Indian River, Florida. National Marine Fisheries Service, Silver Spring, Maryland.
- Kenworthy, W. J. 1997. An updated biological status review and summary of the proceedings of a workshop to review the biological status of the seagrass Halophila johnsonii Eiseman. Southeast Fisheries Science Center, National Marine Fisheries Service, Beaufort, North Carolina.
- Kenworthy, W. J. 1999. Demography, population dynamics, and genetic variability of natural and transplanted populations of *Halophila johnsonii*, a threatened seagrass. Annual Progress Report, July 1999.
- Kenworthy, W. J. 2000. The role of sexual reproduction in maintaining populations of *Halophila decipiens*: implications for the biodiversity and conservation of tropical seagrass ecosystems. Pacific Conservation Biology 5(4):260-268.
- Kenworthy, W. J., M. S. Fonseca, P. E. Whitfield, and K. K. Hammerstrom. 2002. Analysis of Seagrass Recovery in Experimental Excavations and Propeller-Scar Disturbances in the Florida Keys National Marine Sanctuary. Journal of Coastal Research (ArticleType: research-article / Issue Title: SPECIAL ISSUE NO. 37. IMPACTS OF MOTORIZED WATERCRAFT ON SHALLOW ESTUARINE AND COASTAL MARINE ENVIRONMENTS / Full publication date: FALL 2002 / Copyright © 2002 Coastal Education & Research Foundation, Inc.):75-85.

- Ketten, D. R., and S. M. Bartol. 2006. Functional measures of sea turtle hearing. Office of Naval Research, Arlington, VA.
- Kirsch, K. D., K. A. Barry, M. S. Fonseca, P. E. Whitfield, S. R. Meehan, W. J. Kenworthy, and B. E. Julius. 2005. The Mini-312 Program—An Expedited Damage Assessment and Restoration Process for Seagrasses in the Florida Keys National Marine Sanctuary. Journal of Coastal Research (ArticleType: research-article / Issue Title: SPECIAL ISSUE NO. 40. Coastal Restoration: Where Have We Been, Where Are We Now, and Where Should We Be Going? / Full publication date: WINTER 2005 / Copyright © 2005 Coastal Education & Research Foundation, Inc.):109-119.
- Kohler, N. E., and P. A. Turner. 2001. Shark tagging: a review of conventional methods and studies. Pages 191-224 *in* The behavior and sensory biology of elasmobranch fishes: an anthology in memory of Donald Richard Nelson. Springer.
- Krebs, J., F. Jacobs, and A. N. Popper. 2012. Presence of Acoustic-Tagged Atlantic Sturgeon and Potential Avoidance of Pile-Driving Activities During the Pile Installation Demonstration Project (PIDP) for the Tappan Zee Hudson River Crossing Project. AKRF, Inc., Environmental Bioacoustics, LLC.
- Kritzler, H., and L. Wood. 1961. Provisional audiogram for the shark, *Carcharhinus leucas*. Science 133:1480-1482.
- Kunzelman, J. 2007. Southern Range, permanent transect implementation, summer sampling 2006. Report prepared for the Johnson's Seagrass Recovery Team. Florida Fish and Wildlife Conservation Commission, St. Petersburg, FL.
- LaBrecque, E., C. Curtice, J. Harrison, S. M. Van Parijs, and P. N. Halpin. 2015. 3. Biologically Important Areas for Cetaceans Within U.S. Waters – Gulf of Mexico Region. Aquatic Mammals 41(1):30-38.
- Landry, J. B., W. J. Kenworthy, and G. D. Carlo. 2008. The effects of docks on seagrasses, with particular emphasis on the threatened seagrass, *Halophila johnsonii*. Report submitted to NMFS Office of Protected Resources.
- Laughlin, J. 2010. UNDERWATER SOUND LEVELS ASSOCIATED WITH DRIVING STEEL PILES AT THE VASHON FERRY TERMINAL. Seattle, Washington: Washington State Department of Transportation.
- Lenhardt, M., S. E. Moein, and J. A. Musick. 1996. A method for determining hearing thresholds in marine turtles. Pages 160-161 *in* J. A. Keinath, D. E. Barnard, J. A. Musick, and B. A. Bell, editors. Fifteenth Annual Symposium on Sea Turtle Biology and Conservation.
- Lenhardt, M. L. 1994. Seismic and very low frequency sound induced behaviors in captive loggerhead marine turtles (*Caretta caretta*). Pages 238-241 *in* K. A. Bjorndal, A. B. Bolten, D. A. Johnson, and P. J. Eliazar, editors. Proceedings of the Fourteenth Annual Symposium on Sea Turtle Biology and Conservation.

- Lovell, J. M., M. M. Findlay, R. M. Moate, J. R. Nedwell, and M. A. Pegg. 2005. The Inner Ear Morphology and Hearing Abilities of the Paddlefish (*Polyodon spathula*) and the Lake Sturgeon (*Acipenser fulvescens*). Comparative Biochemistry and Physiology a-Molecular & Integrative Physiology 142(3):286-296.
- Maguire, J.-J., M. Sissenwine, J. Csirke, and R. Grainger. 2006. The state of the world highly migratory, straddling and other high seas fish stocks, and associated species. United Nations, Food and Agriculture Organization, Rome, Italy.
- Mason, W. T., and J. P. Clugston. 1993. Foods of the Gulf sturgeon (*Acipenser oxyrhynchus desotoi*) in the Suwannee River, Florida. Transactions of the American Fisheries Society 122(3):378-385.
- Maze-Foley, K., and K. D. Mullin. 2006. Cetaceans of the oceanic northern Gulf of Mexico: Distributions, group sizes and interspecific associations. Journal of Cetacean Research and Management 8(2):203-213.
- McCauley, R. D., J. Fewtrell, A. J. Duncan, C. Jenner, M.-N. Jenner, J. D. Penrose, R. I. T. Prince, A. Adhitya, J. Murdoch, and K. McCabe. 2000a. Marine seismic surveys - a study of environmental implications. APPEA Journal 40:692-708.
- McCauley, R. D., J. Fewtrell, A. J. Duncan, C. Jenner, M.-N. Jenner, J. D. Penrose, R. I. T. Prince, A. Adhitya, J. Murdoch, and K. McCabe. 2000b. Marine seismic surveys: Analysis and propagation of air-gun signals; and effects of air-gun exposure on humpback whales, sea turtles, fishes and squid Curtin University of Technology, Western Australia.
- McLeod, E., and R. V. Salm. 2006. Managing mangroves for resilience to climate change. IUCN, Gland, Switzerland.
- Meehl, G. A., T. F. Stocker, W. D. Collins, P. Friedlingstein, A. T. Gaye, J. M. Gregory, A. Kitoh, R. Knutti, J. M. Murphy, A. Noda, S. C. B. Raper, I. G. Watterson, A. J. Weaver, and Z. C. Zhao. 2007. Global climate projections. Pages 747-846 *in* S. Solomon, D. Qin, M. Manning, M. Marquis, K. Averyt, M. M. B. Tignor, J. H. L. Miller, and Z. Chen, editors. Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY.
- Meyer, M., D. T. T. Plachta, and A. N. Popper. 2003. When a "Primitive" Fish Listens to Tones: Encoding of Sound in the Auditory Periphery of the Shortnose Sturgeon, *Acipenser brevirostrum*. Pages 48 *in* Abstracts of the Twenty-Sixth Annual Midwinter Research Meeting of the Association for Research in Otolaryngology.
- Meyer, M., and A. N. Popper. 2002. Hearing in "primitive" fish: Brainstem responses to pure tone stimuli in the lake sturgeon, *Acipenser fulvescens*. Abstracts of the Association for Research in Otolaryngology 25:11-12.

- MMIQT, D. 2015. Models and analyses for the quantification of injury to the GOMX cetaceans from the Deepwater Horizon oil spill, DWH-AR0105866.
- Moein, S. E., J. A. Musick, and M. L. Lenhardt. 1994. Auditory behavior of the loggerhead sea turtle (Caretta caretta). Pages 89 *in* K. A. Bjorndal, A. B. Bolten, D. A. Johnson, and P. J. Eliazar, editors. Fourteenth Annual Symposium on Sea Turtle Biology and Conservation.
- Molvaer, O. I., and T. Gjestland. 1981. Hearing damage risk to divers operating noisy tools under water. Scandinavian Journal of Work, Environment & Health 7(4):263-270.
- Montes, N., R. Sweet, C. Sidman, and T. Fik. 2016. Offshore Recreational Boating Characterization in the Southeast US. Pages 77 *in*. University of Florida Sea Grant Program, Gainesville, Florida.
- Mullin, K. D. 2007. Abundance of cetaceans in the oceanic Gulf of Mexico based on 2003-2004 ship surveys. National Oceanic and Atmospheric Administration, National Marine Fisheries Service, Southeast Fisheries Science Center, Pascagoula, Mississippi.
- Mullin, K. D., and W. Hoggard. 2000. Visual surveys of cetaceans and sea turtles from aircraft and ships. Minerals Management Service.
- Myrberg, A. A. 2001. The Acoustical Biology of Elasmobranchs. Environmental Biology of Fishes 60(1):31-46.
- Nelson, D., and S. Gruber. 1963. Sharks: Attraction by low-frequency sounds. Science 142:975-977.
- Ning, Z. H., R. E. Turner, T. Doyle, and K. K. Abdollahi. 2003. Integrated assessment of the climate change impacts on the Gulf Coast Region: findings of the Gulf Coast Regional Assessment. LSU Graphic Services, Baton Rouge, Louisiana.
- NMFS. 1997. ESA Section 7 consultation on the continued hopper dredging of channels and borrow areas in the southeastern United States. Biological Opinion.
- NMFS. 2000. Smalltooth Sawfish Status Review. National Oceanic and Atmospheric Administration, National Marine Fisheries Service, Southeast Regional Office, Saint Petrsburg, FL.
- NMFS. 2002. Recovery plan for Johnson's seagrass (*Halophila johnsonii*). National Oceanic and Atmospheric Administration, National Marine Fisheries Service, Office of Protected Resources, Silver Spring, Maryland.
- NMFS. 2007a. Endangered Species Act 5-Year Review: Johnson's Seagrass (Halophila johnsonii, Eiseman). National Marine Fisheries Service, Silver Spring, Maryland.
- NMFS. 2007b. ESA Section 7 consultation on the dredging of Gulf of Mexico navigation channels and sand mining ("borrow") areas using hopper dredges by COE Galveston,
New Orleans, Mobile, and Jacksonville Districts. Second Revised Biological Opinion (November 19, 2003).

- NMFS. 2009. Smalltooth Sawfish Recovery Plan (*Pristis pectinata*). National Marine Fisheries Service, Silver Spring, Maryland.
- NMFS. 2012. 12 USACE SAJ General Permits Renewal.
- NMFS. 2014a. Biological Opinion for the Authorization of discharge of dredged and fill material or other structures or work into waters of the United States under the Corps' Nationwide Permit Program.
- NMFS. 2014b. Biological Opinion on Regional General Permit SAJ-82 (SAJ-2007-1590), Florida Keys, Monroe County, Florida. National Marine Fisheries Service, St Petersburg, FL.
- Noriega, R., J. M. Werry, W. Sumpton, D. Mayer, and S. Y. Lee. 2011. Trends in annual CPUE and evidence of sex and size segregation of *Sphyrna lewini*: Management implications in coastal waters of northeastern Australia. Fisheries Research 110(3):472-477.
- Norton, S. L., T. R. Wiley, J. K. Carlson, A. L. Frick, G. R. Poulakis, and C. A. Simpfendorfer. 2012. Designating critical habitat for juvenile endangered smalltooth sawfish in the United States. Marine and Coastal Fisheries 4(1):473-480.
- O'Hara, J., and J. R. Wilcox. 1990. Avoidance Responses of Loggerhead Turtles, Caretta caretta, to Low Frequency Sound. Copeia 1990(2):564-567.
- Orlando, S. P., Jr., P. H. Wendt, C. J. Klein, M. E. Patillo, K. C. Dennis, and H. G. Ward. 1994. Salinity Characteristics of South Atlantic Estuaries. NOAA, Office of Ocean Resources Conservation and Assessment, Silver Spring, MD.
- Pfeffer, W. T., J. T. Harper, and S. O'Neel. 2008. Kinematic Constraints on Glacier Contributions to 21st-Century Sea-Level Rise. Science 321(5894):1340-1343.
- Philbrick, C. T., and D. H. Les. 1996. Evolution of Aquatic Angiosperm Reproductive Systems. Bioscience 46(11):813-826.
- Poulakis, G. R. 2012. Distribution, Habitat Use, and Movements of Juvenile Smalltooth Sawfish, *Pristis pectinata*, in the Charlotte Harbor Estuarine System, Florida. Florida Institute of Technology, Melbourne, FL.
- Poulakis, G. R., and J. C. Seitz. 2004. Recent occurrence of the smalltooth sawfish, *Pristis pectinata* (Elasmobranchiomorphi: Pristidae), in Florida Bay and the Florida Keys, with comments on sawfish ecology. Florida Scientist 67(27):27-35.
- Poulakis, G. R., P. W. Stevens, A. A. Timmers, C. J. Stafford, and C. A. Simpfendorfer. 2013. Movements of juvenile endangered smalltooth sawfish, *Pristis pectinata*, in an estuarine

river system: use of non-main-stem river habitats and lagged responses to freshwater inflow-related changes. Environmental Biology of Fishes 96(6):763-778.

- Poulakis, G. R., P. W. Stevens, A. A. Timmers, T. R. Wiley, and C. A. Simpfendorfer. 2011. Abiotic affinities and spatiotemporal distribution of the endangered smalltooth sawfish, *Pristis pectinata*, in a south-western Florida nursery. Marine and Freshwater Research 62(10):1165-1177.
- Quigley, D. T. G., and K. Flannery. 2002. Leucoptic harbour porpoise *Phocoena phocoena* (L.). Irish Naturalists' Journal 27(4):170.
- Rahmstorf, S., A. Cazenave, J. A. Church, J. E. Hansen, R. F. Keeling, D. E. Parker, and R. C. J. Somerville. 2009. Recent climate observations compared to projections. Science 316(5825):709.
- Reddering, J. S. V. 1988. Prediction of the effects of reduced river discharge on estuaries of the south-eastern Cape Province, South Africa. South African Journal of Science 84:726-730.
- Rice, K. J., and N. C. Emery. 2003. Managing microevolution: restoration in the face of global change. Frontiers in Ecology and the Environment 1(9):469-478.
- Saenger, P., and J. Moverley. 1985. Vegetative phenology of mangroves along the Queensland coastline. Pages 9 in M. G. Ridpath, and L. K. Corbett, editors. Ecology of the wet-dry tropics: Proceedings of a joint symposium with the Australian Mammal Society in association with the Darwin Institute of Technology. Blackwell Scientific Book Distributors, Melbourne.
- SAFMC. 1998. Final Plan for the South Atlantic Region: Essential Fish Habitat Requirements for the Fishery Management Plan of the South Atlantic Fishery Management Council. South Atlantic Fishery Management Council, Charleston, SC.
- Saloman, C. H., S. P. Naughton, and J. L. Taylor. 1982. Benthic Community Response to Dredging Borrow Pits, Panama City Beach, Florida. Pages 141 in. National Marine Fisheries Service, Panama City Beach, Southeast Fisheries Center.
- Sargent, F. J., T. J. Leary, D. W. Crewz, and C. R. Kruer. 1995. Scarring of Florida's Seagrasses: assessment and management options. Florida Marine Research Institute, St. Petersburg, FL.
- Scavia, D., J. C. Field, D. F. Boesch, R. W. Buddemeier, V. Burkett, D. R. Cayan, M. Fogarty, M. A. Harwell, R. W. Howarth, C. Mason, D. J. Reed, T. C. Royer, A. H. Sallenger, and J. G. Titus. 2002. Climate change impacts on US coastal and marine ecosystems. Estuaries 25(2):149-164.
- Seitz, R., R. Lipcius, N. Olmstead, M. Seebo, and D. Lambert. 2006. Influence of shallow-water habitats and shoreline development on abundance, biomass, and diversity of benthic prey and predators in Chesapeake Bay. Marine Ecology Progress Series 326:11-27.

- Semeniuk, V. 1994. Predicting the effect of sea-level rise on mangroves in northwestern Australia. Journal of Coastal Research 10(4):1050-1076.
- Shafer, D. J., J. Karazsia, L. Carrubba, and C. Martin. 2008. Evaluation of regulatory guidelines to minimize impacts to seagrasses from single-family residential dock structures in Florida and Puerto Rico. U.S. Army Engineer Research and Development Center, Vicksburg, Mississippi.
- Sheppard, C., and R. Rioja-Nieto. 2005. Sea surface temperature 1871–2099 in 38 cells in the Caribbean region. Marine Environmental Research 60(3):389-396.
- Short, F., T. Carruthers, W. Dennison, and M. Waycott. 2007. Global seagrass distribution and diversity: A bioregional model. Journal of Experimental Marine Biology and Ecology 350(1-2):3-20.
- Short, F. T., and H. A. Neckles. 1999. The effects of global climate change on seagrasses. Aquatic Botany 63(3):169-196.
- Simpfendorfer, C. A. 2001. Essential habitat of the smalltooth sawfish (*Pristis pectinata*). Report to the National Fisheries Service's Protected Resources Division. Mote Marine Laboratory Technical Report.
- Simpfendorfer, C. A. 2003. Abundance, movement and habitat use of the smalltooth sawfish. Final Report. Mote Marine Laboratory Mote Technical Report No. 929, Sarasota, FL.
- Simpfendorfer, C. A. 2006. Movement and habitat use of smalltooth sawfish. Final Report. Mote Marine Laboratory, Mote Marine Laboratory Technical Report 1070, Sarasota, FL.
- Simpfendorfer, C. A., G. R. Poulakis, P. M. O'Donnell, and T. R. Wiley. 2008. Growth rates of juvenile smalltooth sawfish, *Pristis pectinata* (Latham), in the western Atlantic. Journal of Fish Biology 72(3):711-723.
- Simpfendorfer, C. A., T. R. Wiley, and B. G. Yeiser. 2010. Improving conservation planning for an endangered sawfish using data from acoustic telemetry. Biological Conservation 143:1460-1469.
- Simpfendorfer, C. A., B. G. Yeiser, T. R. Wiley, G. R. Poulakis, P. W. Stevens, and M. R. Heupel. 2011. Environmental Influences on the Spatial Ecology of Juvenile Smalltooth Sawfish (*Pristis pectinata*): Results from Acoustic Monitoring. PLoS ONE 6(2):e16918.
- Skalski, J. R., W. H. Pearson, and C. I. Malme. 1992. Effects of sounds from a geophysical survey device on catch-per-unit-effort in a hook-and-line fishery for rockfish (*Sebastes* spp.). Canadian Journal of Fisheries and Aquatic Sciences 49:1357-1365.
- Snedaker, S. 1995. Mangroves and climate change in the Florida and Caribbean region: scenarios and hypotheses. Hydrobiologia 295(1-3):43-49.

- Spence, J., R. Fischer, M. Bahtiarian, L. Boroditsky, N. Jones, and R. Dempsey. 2007. Review of Existing and Future Potential Treatments for Reducing Underwater Sound from Oil and Gas Industry Activities. Noise Control Engineering, Inc.
- Stedman, S., and T. E. Dahl. 2008. Status and trends of wetlands in the coastal watersheds of the Eastern United States 1998-2004. National Oceanic and Atmospheric Administration, National Marine Fisheries Service, and U.S. Department of the Interior, U.S. Fish and Wildlife Service.
- Steward, J., R. Virnstein, M. Lasi, L. Morris, J. Miller, L. Hall, and W. Tweedale. 2006. The impacts of the 2004 hurricanes on hydrology, water quality, and seagrass in the Central Indian River Lagoon, Florida. Estuaries and Coasts 29(6):954-965.
- Steward, J., R. Virnstein, L. Morris, and E. Lowe. 2005. Setting seagrass depth, coverage, and light targets for the Indian River Lagoon system, Florida. Estuaries and Coasts 28(6):923-935.
- Swett, R., D. Fann, B. Staugler, and F. S. Grant. 2012. Regional Waterway Management System for Charlotte County, Florida.
- Swett, R. A., G. A. Antonini, and S. Schulte. 2000a. Regional waterway management system for north Manatee County. Florida Sea Grant, TD-2, Gainesville, FL: Florida Sea Grant.
- Swett, R. A., D. Fann, G. Antonini, and L. Carlin-Alexander. 2000b. Regional waterway management system for Lee County, Phase 2. TD-3. Gainesville, FL: Florida Sea Grant College Program.
- Swett, R. A., D. A. Fann, and G. A. Antonini. 2002. Regional waterway management system for Manatee County: Bishop Harbor, Tidal Braden River, and lower reaches of the upper Manatee River. TD-6. Gainesville, FL: Florida Sea Grant College Program.
- Swett, R. A., D. A. Fann, G. A. Antonini, and L. C. Alexander. 2000c. Regional waterway management system for Lee County, Phase 1. Florida Sea Grant College Program Technical Document.
- Torquemada, Y., M. Durako, and J. Lizaso. 2005. Effects of salinity and possible interactions with temperature and pH on growth and photosynthesis of *Halophila johnsonii* Eiseman. Marine Biology 148(2):251-260.
- Trenberth, K. 2005. Uncertainty in Hurricanes and Global Warming. Science 308(5729):1753-1754.
- UNESCO. 1992. Coastal systems studies and sustainable development. Pages 276 *in* COMAR Interregional Scientific Conference. UNESCO, Paris, 21-25 May, 1991.
- USACE. 1993. Dredging Fundamentals Facilitator's Guide. U.S. Army Corps of Engineers, Huntsville Division.

- USEPA. 1994. Freshwater Inflow Action Agenda For The Gulf of Mexico; First Generation-Management Committee Report. U.S. Environmental Protection Agency.
- USFWS. 1999. South Florida Multi-Species Recovery Plan Atlanta, Georgia. 2172p.
- Vargas-Moreno, J. C., and M. Flaxman. 2010. Addressing the challenges of climate change in the greater everglades landscape. Massachusetts Institute of Technology, Cambridge, Massachusetts.
- Virnstein, R. W., and L. M. Hall. 2009. Northern range extension of the seagrasses *Halophila johnsonii* and *Halophila decipiens* along the east coast of Florida, USA. Aquatic Botany 90(1):89-92.
- Virnstein, R. W., L. C. Hayek, and L. J. Morris. 2009. Pulsating patches: a model for the spatial and temporal dynamics of the threatened seagrass Halophila johnsonii. Marine Ecology Progress Series 385:97-109.
- Virnstein, R. W., and L. J. Morris. 2007. Distribution and abundance of Halophila johnsonii in the Indian River Lagoon: An update. Technical Memorandum # 51. St. Johns River Water Management District, Palatka, Florida.
- Virnstein, R. W., L. J. Morris, J. D. Miller, and R. Miller-Myers. 1997. Distribution and abundance of Halophila johnsonii in the Indian River Lagoon. Technical Memorandum # 24. St. Johns River Water Management District, Palatka, Florida.
- Wanless, H. R., B. M. Vlaswinkel, and K. L. Jackson. 2005. Coastal Landscape and Channel Evolution Affecting Critical Habitats at Cape Sable, Everglades National Park, Florida. University of Miami.
- Whitfield, A. K., and M. N. Bruton. 1989. Some biological implications of reduced freshwater inflow into eastern Cape estuaries: a preliminary assessment. South African Journal of Science 85:691-694.
- Whitfield, P. E., W. J. Kenworthy, K. K. Hammerstrom, and M. S. Fonseca. 2002. The Role of a Hurricane in the Expansion of Disturbances Initiated by Motor Vessels on Seagrass Banks. Journal of Coastal Research (ArticleType: research-article / Issue Title: SPECIAL ISSUE NO. 37. IMPACTS OF MOTORIZED WATERCRAFT ON SHALLOW ESTUARINE AND COASTAL MARINE ENVIRONMENTS / Full publication date: FALL 2002 / Copyright © 2002 Coastal Education & Research Foundation, Inc):86-99.
- Wilber, D. H., D. G. Clarke, and S. I. Rees. 2007. Responses of benthic macroinvertebrates to thin-layer disposal of dredged material in Mississippi Sound, USA. Marine Pollution Bulletin 54(1):42-52.
- Wiley, T. R., and C. A. Simpfendorfer. 2007. The ecology of elasmobranchs occurring in the Everglades National Park, Florida: implications for conservation and management. Bulletin of Marine Science 80(1):171-189.

- Willis, M. R., M. Broudic, M. Bhurosah, and I. Masters. 2010. Noise Associated with Small Scale Drilling Operations. Paper submitted to the 3rd International Conference on Ocean Energy. Bilbao, Spain.
- Winkler, S., and A. Ceric. 2006. 2004 Status and trends in water quality at selected sites in the St. Johns River Water Management District. St. Johns River Water Management District, Palatka, FL.
- Woodward-Clyde Consultants. 1994. Biological resources of the Indian River Lagoon. Final Technical Report. Prepared for the Indian River Lagoon National Estuary Program.

#### 14 Appendix A: Authorities under which an Action Will Be Conducted

USACE uses multiple methods to authorize activities. Pursuant to Section 404 of the CWA (33 U.S.C. 1344) and Section 10 of the RHA of 1899 (33 U.S.C. 403), USACE has authority to issue GPs<sup>52</sup> (regional, programmatic, and nationwide) for any category of activities that are substantially similar in nature, and result in no more than minimal adverse effects on the environment, either individually or cumulatively. Section 10 of the RHA authorizes all structures and work in navigable waters of the United States while Section 404 of the CWA covers the discharge of dredged or fill materials in waters of the United States. USACE uses a combination of all 3 types of these GPs (regional, programmatic, and nationwide) when authorizing activities within the state of Florida and the U.S. Caribbean, provided it has been determined that the environmental consequences of the action(s) are individually and cumulatively minimal (see 33 CFR 325.2(e) and 33 CFR Part 330). PGPs are used to avoid unnecessary duplication of the regulatory control exercised by another federal, state, or local agency. All GPs are valid for a maximum of 5 years (33 CFR 325.2(e)(2)) and must be re-evaluated prior to reissuance.

As stated in the NWP Opinion (NMFS 2014a), a basic premise of the USACE's permitting program is that no discharges of dredged or fill material into waters of the United States shall be permitted if: (1) a practicable alternative exists that is less damaging to the aquatic environment, or (2) the discharge would cause the nation's waters to be significantly degraded. In order for a project to be permitted, it must be demonstrated that, to the extent practicable, steps have been taken to avoid impacts to wetlands and other aquatic resources, potential impacts have been minimized, and compensation will be provided for any remaining unavoidable impacts.

**14.1.1 Current Permits Used to Authorize Projects in Florida under NMFS Jurisdiction** Below is a description of permits used by USACE to authorize activities within the state of Florida. Table 70 provides a list of all of the RGPs and PGPs used to authorize activities under NMFS jurisdiction in the USACE Jacksonville District.

 Nationwide permits: NWPs are a type of GP issued for activities that occur throughout the United States. On January 6, 2017, the USACE issued its 2017 NWPs for work in streams and wetlands under Section 404 of the federal CWA and Section 10 of the RHA of 1899 (<u>82</u> <u>FR 1860</u>). The 2017 NWPs are effective March 19, 2017, for a period of 5 years, replacing the 2012 NWPs that expired on March 18, 2017. The USACE issued Decision Documents for each of the fifty-four 2017 NWPs. The Decision Documents state that the issuance of the 2017 NWPS will have "no effect" under the ESA. The USACE reached its "no effects" determination based on General Condition 18, which requires every applicant for a permit that "might affect" an ESA-listed species or its designed critical habitat to submit a pre-construction notification, so that the USACE can determine if the action requires ESA consultation.

<sup>&</sup>lt;sup>52</sup> The term "general permit" (GP) is defined at 33 CFR 322.2(f) and 33 CFR 323.2(h). Programmatic general permits are a type of general permit, and are defined at 33 CFR 325.5(c)(3).

NWP	Description
NWP-1	Aids to Navigation
NWP-2	Structures in Artificial Canals
NWP-3	Maintenance
NWP-4	Fish and Wildlife Harvesting, Enhancement, and Attraction Devices and Activities
NWP-5	Scientific Measurement Devices
NWP-6	Survey Activities
NWP-7	Outfall Structures and Associated Intake Structures
NWP-8	Oil and Gas Structures on the Outer Continental Shelf
NWP-9	Structures in Fleeting and Anchorage Areas
NWP-10	Mooring Buoys
NWP-11	Temporary Recreational Structures
NWP-12	Utility Line Activities, Utility lines, Utility line substations, Foundations for
	overhead utility line towers, poles, and anchors, Access roads
NWP-13	Bank Stabilization
NWP-14	Linear Transportation Projects
NWP-15	USCG Approved Bridges
NWP-16	Return Water From Upland Contained Disposal Areas
NWP-17	Hydropower Projects
NWP-18	Minor Discharges
NWP-19	Minor Dredging
NWP-20	Response Operations for Oil and Hazardous Substances
NWP-21	Surface Coal Mining Activities
NWP-22	Removal of Vessels
NWP-23	Approved Categorical Exclusions
NWP-24	Indian Tribe or State Administered Section 404 Program
NWP-25	Structural Discharges
NWP-27	Aquatic Habitat Restoration, Establishment, and Enhancement Activities
NWP-28	Modifications of Existing Marinas
NWP-29	Residential Developments
NWP-30	Moist Soil Management for Wildlife
NWP-31	Maintenance of Existing Flood Control Facilities
NWP-32	Completed Enforcement Actions
NWP-33	Temporary Construction, Access, and Dewatering
NWP-34	Cranberry Production Activities
NWP-35	Maintenance Dredging of Existing Basins
NWP-36	Boat Ramps
NWP-37	Emergency Watershed Protection and Rehabilitation
NWP-38	Cleanup of Hazardous and Toxic Waste
NWP-39	Commercial and Institutional Developments
NWP-40	Agricultural Activities
NWP-41	Reshaping Existing Drainage Ditches

Table 70. NWPs used by the USACE Jacksonville District

NWP-42	Recreational Facilities
NWP-43	Stormwater Management Facilities
NWP-44	Mining Activities
NWP-45	Repair of Uplands Damaged by Discrete Events
NWP-46	Discharges in Ditches
NWP-48	Existing Commercial Shellfish Aquaculture Activities
NWP-49	Coal Remaining Activities
NWP-50	Underground Coal Mining Activities
NWP-51	Land-Based Renewable Energy Generation Facilities
NWP-52	Water-Based Renewable Energy Generation Pilot Projects
NWP-53	Removal of Lowhead Dams
NWP-54	Living Shorelines

2. **Programmatic general permits:** PGPs are a type of GP issued by USACE that delegate authorization to other federal, tribal, state, or local regulatory authorities where there is duplication in their programs. The purpose of PGPs is to improve the regulatory process for applicants, enhance environmental protection, reduce unnecessary duplicative procedures and evaluations, and make more efficient use of limited resources. Each PGP is specifically defined and requires a written agreement between USACE and the entity receiving delegation authority. The agreement stipulates the review process and defines "kick-outs" (i.e., situations where the proposed activity would not meet the PGP Special Conditions and would come back to USACE for review).

In Florida, USACE provides delegated authorization to the following agencies to permit activities under NMFS jurisdiction for these listed permits: SAJ-91 provides administrative limited authority to the City of Cape Coral; SAJ-96 provides administrative limited authority to Pinellas County; SAJ-99 delegates authority to the Florida Department of Agriculture and Consumer Services for live rock and marine bivalve aquaculture; SAJ-42 provides limited administrative authority to Miami-Dade County, and the SPGP provides limited administrative authority to FDEP (see Table 71).

USACE retains the authority to modify, suspend, or revoke any PGP if the USACE believes that appropriate protection is not being afforded to the environment or other relevant aspects of the public interest, or when USACE concludes that adverse environmental effects are more than minimal, either individually or cumulatively. Additionally, USACE always retains its authority to require an individual permit for any particular project, even if the project otherwise meets all the requirements of the PGP. USACE exercises this authority when it concludes that the processing of an individual permit is necessary to protect the environment, public interest, or when individual or cumulative effects require a more rigorous review. Lastly, USACE retains the full range of its enforcement authority and options where it believes that a project does not comply with the terms or conditions of the PGP, regardless of whether the project has been permitted by the federal, tribal, state, or local regulatory authority.

USACE RGPs/ PGPs	Permit Description					
	USACE Permits in Florida					
SAJ-5	Maintenance Dredging in Residential Canals in Florida					
SAJ-13	Aerial Transmission Lines in Florida					
SAJ-14	Subaqueous Utility and Transmission Lines in Florida					
SAJ-17	Minor Structures in Florida					
SAJ-20	Private Single-Family Piers in Florida					
SAJ-33	Private Multi-Family or Government Piers in Florida					
SAJ-34	Private Commercial Piers in Florida					
SAJ-42 (PGP)	Minor Activities in Miami-Dade County					
SAJ-46	Bulkheads and Backfill in Residential Canals in Florida					
SAJ-72	Residential Docks in Citrus County					
SAJ-82 Single-Family Shoreline Stabilization, Marginal Docks, and Boat Ramps in M. County						
SAJ-91 (PGP)	Minor Activities in the Canal System of the City of Cape Coral					
SAJ-93	Maintenance Dredging Activities for the Atlantic Intracoastal Waterway, the Intracoastal Waterway, and the Okeechobee Waterway					
SAJ-96 (PGP)	Minor Activities in Pinellas County					
SAJ-99 (PGP)	Live Rock and Marine Bivalve Aquaculture in Florida					
SPGP (PGP)	State Programmatic General Permit for the State of Florida					
USACE Permits in the U.S. Caribbean						
SAJ-81	Maintenance Dredging the Mouths of Rivers, Creeks, Streams, Canals, and/or Storm Drainage Located in Navigable Waters of the United States, in the Commonwealth of Puerto Rico					
SAJ-85	Discharge of dredged or fill material in non-tidal waters for maintenance, enhancement or restoration of stream channels, conducted or supervised by the Puerto Rico Department of Natural and Environmental Resources					

 Table 71. USACE General Permits in Florida and the U.S. Caribbean That May Affect

 Species within NMFS' Jursidiction

- 3. **Regional general permits:** RGPs are a type of GP specific to a given region (in this case, Florida). Within the state of Florida, USACE staff individually review permit applications to determine if it meets the terms and conditions defined by an RGP. All RGPs require an applicant to submit a preconstruction notification and cannot begin construction until they have received a written verification from USACE that their project is authorized in accordance with the terms and conditions of the RGP.
- 4. Individual permits: If a project is not authorized by the USACE under an RGP, NWP, or PGP because the effects of the action will be more than minor in nature or if the project needs an additional level of review, then it is addressed as an individual permit. Individual permits are issued following a case-by-case evaluation by USACE in accordance with the procedures detailed in 33 CFR Part 325, and a determination that the proposed structure or work is in the public interest pursuant to 33 CFR Part 320. Individual permits require Section 7 consultation with NMFS (consultation) for projects involving in-water work that may affect ESA-listed species or critical habitat under our purview.

## **Types of Individual Permits:**

- 1. Letters of Permission: Letters of permission (LOP) are a type of individual permit issued in accordance with the abbreviated procedures located in 33 CFR 325.2(2). The procedures and standards for issuing LOPs are developed after coordination with federal and state fish and wildlife agencies, as required by the Fish and Wildlife Coordination Act, and a public interest evaluation. An LOP authorization can be issued without requesting public input. An LOP is appropriate for projects (1) subject to Section 10 of the RHA of 1899, (2) that are considered minor by the USACE District Engineer and would not have significant individual or cumulative effects on environmental values, and (3) should encounter no appreciable opposition. For example, marinas or multifamily piers with 20 or fewer boat slips may qualify for a Letter of Permission. LOPs may also be used in those cases subject to Section 404 of the CWA, after the USACE District Engineer (1) consults with federal and state agencies to develop a list of categories of activities proposed for authorization under LOP procedures, (2) issues a public notice advertising the proposed list and the LOP procedures, requesting comments and offering an opportunity for public hearing, (3) issues or waives a 401 certification, and, (4) issues a Coastal Zone Management Act consistency concurrence obtained or presumed either on a generic or individual basis. An example of a Section 404 LOP is an erosion control activity that does not to exceed 0.2 ac of fill.
- 2. Standard Permits (SPs): A project that does not qualify for GP or LOP authorization is reviewed through the standard permit process (see 33 CFR 325.5(b)(1)), which includes a public notice, public interest review, environmental documentation, and, if applicable, a Section 404(b)(1) Guidelines compliance analysis. SPs are used for projects that generally exceed 0.5 ac of impact to waters of the United States with more than a minimal impact on the environment.
- 3. **Modifications**: The construction window for individual permits is typically 5 years. If the permittee wishes to change the project or extend the construction window, a modification request must be submitted to the USACE. Any modification would still need to comply with the regulations discussed above for LOP and SPs. A new ESA consultation is only required for modifications if the project design has changed such that it increases the impact to listed species or critical habitat or if new species or critical habitat has been designated in the action area.

#### 15 Appendix B: In-water Construction Noise

#### 15.1 Glossary of Acoustic Terms

**Abandonment** – In reference to species' use of an area, the long-term discontinued resting, feeding, mating, or nursing areas, or other areas that are important to the species.

**Avoidance** – In reference to species' reaction to noise, individuals may avoid an area for the duration a noise is present.

**Behavioral Zone** – The distance from pile driving within which species might experience potentially adverse behavioral reactions.

**Cumulative Sound Exposure Level (cSEL)** – A measurement of the accumulated sound energy associated with a series of pile strikes occurring during a construction event. cSEL at a certain level can physically injure species. cSEL can be estimated from the single-strike (SEL) and the number of strikes that likely would be required to place the pile at its final depth by using the following equation:

 $cSEL = sSEL + 10 \log(number of pile strikes)$  for impact pile driving  $cSEL = sSEL + 10 \log(time in seconds)$  for vibratory pile driving.

**Cylindrical Spreading** – Sound that spreads away from a source in the shape of a cylinder. In some environments, sound will not propagate uniformly in all directions from a source. A simple approximation for spreading loss in a medium with upper and lower boundaries, such as the ocean, can be obtained by assuming that the sound is distributed uniformly over the surface of a cylinder having a radius equal to the range r and a height H equal to the distance between the upper and lower boundary (in the example of the ocean, the depth of the ocean). Beyond some range, the sound will hit the sea surface or sea floor, or other reflective surfaces.

**Dry Firing** – A method of raising and dropping the pile-driving hammer with no compression of the pistons, producing a lower-intensity sound than the full power of the hammer.

**Effective Quiet** – The level at which a sound becomes too quiet to contribute to hearing loss from cumulative sound exposure.

**Ensonified** – An area where sound is present. In the Opinion, the ensonified zone refers to the area of water where the sound from the noise-generating pile driving is present.

**Habituation** – Becoming accustomed to noise through repeated or prolonged exposure. Habituation can occur even when negative consequences result.

**Harassment Zone** – The distance from a noise within which behavioral reactions or temporary threshold shift may occur.

**Impact Zone** – The distance from the pile encompassing the effects of interest (i.e., the physical injury zone and/or the behavior zone).

**Onset of injury -** The point at which a permanent threshold shift or tissue damage occurs.

**Physical Injury Zone** – The distance from a noise source within which physical injury is expected.

**Masking** – Obscuring of softer sounds of interest by louder sounds at similar frequencies. Masking of the vocalizations of conspecifics, mates, predators, and other important signals may affect marine mammals. An analogous effect in humans would be the experience of having difficulty discerning an individual person's speech at loud parties.

Mortality Zone – The distance from a noise source within which mortality may occur.

**No response** – Some species may exhibit no apparent response to a noise while other species exhibit strong responses. No apparent response may indicate undetectable effects (e.g., habituation, permanent threshold shift, and stress).

**Peak Pressure** – The loudest sound impulse at any instant in time during a strike. Peak pressure is the maximum positive pressure between zero and the greatest pressure of signals in units of dB re 1  $\mu$ Pa peak or 0-peak for impulsive pile-driving noise. 0-p values are not to be confused with peak-to-peak measurements (p-p). Exposure to peak pressure above a certain level can result in physical injury.

**Permanent Threshold Shift** – Permanent shift in the auditory threshold from exposure to loud noise over a period of time. Permanent threshold shift is considered permanent hearing loss, and such a loss may affect foraging success, mate acquisition, and other biologically important activities.

**Physical Injury** – Some sounds, such as pressure waves or intense noise from a source, create pressure waves or high-energy sound waves that can result in tissue damage, including swim bladder damage. When we discuss physical injury from noise, we are referring to these types of injuries (tissue damage) and permanent threshold shift.

**Physiological Stress** – Noise is a potential environmental stressor and physiological stress is the term for the physiological response to that stressor. For example, noise can trigger immune system responses that can affect the health of animals.

**Ramp-Up** – A method that involves slowly increasing the power of the hammer, and thus the noise the hammer produces, over a period of time prior to operating at full power and beginning work.

**Root Mean Square (RMS)** – A type of average. In the context of measuring noise in an aquatic environment, the RMS represents the total sound energy in a single strike impulse. It is a decibel measure of the square root of the mean square pressure. For impulses, the average of the squared pressures over the time that comprise that portion of the waveform containing 90% of the sound energy of the impulse. Exposure to RMS above a certain level can result in behavior effects.

**Sensitization** – Increased behavioral response over time as animals learn that a repeating or ongoing noise has significant consequences.

**Single Strike Sound Exposure Level (sSEL)** – An index of the 90% of the sound energy in a single strike. Exposure to sSEL at certain levels can result in physical injury.

**Sound Exposure Level (SEL)** – Sound energy associated with a pile-driving pulse (sSEL), or series of pulses (cSEL), is characterized by the SEL. SEL is calculated by summing the cumulative pressure squared over the time of the event.

**Spherical Spreading** – Spherical spreading describes the decrease in level when a sound wave propagates away from a source uniformly in all directions from a sound source in the shape of a sphere.

**Temporary Threshold Shift** – Temporary hearing loss caused by a decreased sensitivity in hearing. The analogous effect in humans is ringing in the ears and loss of hearing experienced after a loud concert. Temporary threshold shift is generally fully recoverable within hours or days and results in short-term effects.

**Watch Zone** – A protective buffer zone to be monitored to detect animals that are heading towards the impacted area. The watch zone radius may vary depending on the type of project and species potentially occurring in the project area.

## 15.2 How in-water Noise Effects to ESA-listed Species Are Evaluated

Sea turtles, smalltooth sawfish, sturgeon, and grouper are low-frequency hearing generalists (Table 72) and are able to detect the vibrations and lower frequency sound components associated with construction noise. Our current noise analysis is grouped into 2 categories, 1 for effects to fish and the other for effects to sea turtles. When looking at the effects of in-water construction noise, smalltooth sawfish are categorized under the effects to fish along with more traditional fish such as sturgeon (Atlantic, shortnose, and Gulf) and Nassau grouper. Although several different types of activities produce underwater sound, the analysis shows that the loudest noises of concern that may harm listed species result from impact pile-driving activities. Many other underwater noises are produced by construction activities that increase ambient noise levels, but are generally not harmful to sea turtles and fish. Since pile-driving activities occur in shallow, nearshore waters, our analysis does not consider the effects of this action on ESA-listed whales, which we would expect to occur in deeper waters.

Species or Group	Hearing Range	References
sea turtles	100-2,000 Hz	Ketten and Bartol (2006); Lenhardt et al. (1996); Lenhardt (1994); McCauley et al. (2000a); McCauley et al. (2000b); Moein et al. (1994); O'Hara and Wilcox (1990)
smalltooth sawfish	<1,000 Hz	Hearing in the species has not been measured, but is based on assumed lower-frequency hearing for fish without swim bladders (e.g., (Casper et al. 2003).
sturgeon	100-2,000 Hz	(Fay and Popper 2000; Lovell et al. 2005; Meyer et al. 2003; Meyer and Popper 2002)
grouper	100-2,000 Hz	Hearing in this species has not been measured, but is based on assumed low-frequency hearing for fish with swim bladders similar to sturgeon provided above.

Table 72. Hearing ranges of listed species

During impact pile-driving, noise is produced when the energy from the hammer is transferred to the pile and released into the surrounding water and sediment. We have characterized these construction activities and associated noise levels using the best available information provided by the USACE, the Florida Marine Contractors Association, and published literature. Depending on the type and location of pile-driving activity, noise can result in a spectrum of responses in species, ranging from minor to those that can disturb or injure vulnerable animals (Figure 28).



Figure 28. The relative effects to animals: distance to/from a noise source.

#### 15.2.1 Exposure Criteria Used to Determine Potential Effects

NMFS formed a Fisheries Hydroacoustic Working Group in 2004 to evaluate the effects of inwater noise on species under NMFS's jurisdiction. This team consists of biologists from NMFS West Coast Region, USFWS, Federal Highway Administration, and the California, Washington and Oregon Department of Transportations, supported by national experts on sound propagation activities that affect fish and wildlife species of concern. In June 2008, the agencies signed a memorandum of understanding documenting interim criteria for assessing physiological effects of pile driving on fish, until more information was available to reassess these threshold criteria. The criteria were developed for the acoustic levels at which physiological effects to fish could be expected (the onset level), referred to as interim threshold standards below. It should be noted that these onset levels for physiological effects are not the levels at which fish are necessarily mortally damaged and that some physiological effects are minor and recoverable.

These interim threshold standards are used by all of the NMFS regional offices:

- Peak pressure from a single strike: 206 decibels relative to 1 micro-Pascal (dB re 1 μPa).
- cSEL: 187 decibels relative to 1 micro-Pascal-squared second (dB re 1μPa<sub>2</sub>-s) for fishes above 2 grams (0.07 ounces).
- cSEL: 183 dB re 1µPa2-s for fishes below 2 grams (0.07 ounces).

For purposes of assessing behavioral effects of pile driving at several West Coast projects, NMFS has employed a 150 dB re 1  $\mu$ Pa RMS sound pressure level threshold above which it estimates fish species will experience behavioral effects at several sites including the San Francisco-Oakland Bay Bridge and the Columbia River Crossings. This behavioral threshold is also currently used by the NMFS's Greater Atlantic Regional Office and the Southeast Regional Office. NMFS's Southeast Regional Office considers the threshold for behavioral noise effects to sea turtles to be 160 dB re 1  $\mu$ Pa RMS sound pressure level (Skalski et al. 1992). The thresholds used by NMFS's Southeast Regional Office are summarized in Tables 73 and 74 below.

Effect	Animal	Threshold Level (dB re 1 μPa) <sup>e</sup>
Physical Injury (peak pressure)	Fish <sup>a,b</sup> and Sea Turtles	206 (peak pressure)
Physical Injury (Cumulative exposure)	Fish less than 2 grams	183 cSEL
Dehavier	Fish <sup>c</sup>	150 (RMS)
Dellavioi	Sea Turtles <sup>d</sup>	160 (RMS)

 Table 73. Impact pile-driving threshold noise levels for fish and sea turtles

<sup>a</sup> FHA (2012). Fish are considered more sensitive to physical injury than sea turtles; therefore, fish thresholds are used for both fish and sea turtles as conservative interim criteria.

<sup>b</sup> (Halvorsen et al. 2012a; Halvorsen et al. 2012b)

<sup>c</sup>(McCauley et al. 2000b)

<sup>d</sup> Skalski et al. (1992)

<sup>e</sup>See glossary for definitions of different decibel (dB) levels.

Effect	Animal	Threshold Level (dB re 1 µPa)
Physical Injury (peak pressure) <sup>a</sup>	Sturgeon, Sawfish, and Sea Turtles	206 (peak pressure)
Physical Injury (Cumulative exposure)	Fish larger than 102 grams	234 cSEL
Behavior	Fish	150 (RMS)
	Sea Turtles	160 (RMS)

 Table 74. Continuous noise threshold levels for fish and turtles from exposure to vibratory pile-driving noise

<sup>a</sup> Injury criteria from Hastings (2010). There are no SEL criteria for sea turtles for continuous noises. Fish are considered more sensitive to physical injury than sea turtles; therefore, fish thresholds are used for sea turtles as conservative interim criteria

Research continues on the effects of in-water construction on fish, marine mammals, and other aquatic species. Studies by researchers indicate that the threshold of noise impacts to fish may be higher than the interim NMFS thresholds established by the Fisheries Hydroacoustic Working Group in 2008. NMFS is working to review continued research to determine if new guidance and thresholds are warranted. For instance, Halvorsen et al. (2012b) exposed 3 species of fish to pile driving noise in a laboratory setting. Following testing, fish were euthanized and examined for external and internal signs of barotrauma. Halvorsen et al. (2012b) classified the types of fish tested by differences in their anatomy that result in different physiological changes to these groups of fish from pile driving, as described below:

- Fish without swim bladders: Halvorsen et al. (2012b) tested hogchoker fish, a flat bodied fish, for this category of fish. In the southeast region, the only ESA-listed species without a swim bladder is the smalltooth sawfish.
- Physostomous (fish with open swim bladders): Halvorsen et al. (2012b) describes these fish as more evolutionarily ancestral than fish with a closed swim bladders (described below). Physostomous fish have a swim bladder that is connected to the gut via a pneaumatic duct that allows them to gulp air from the water surface and expel air quickly to adjust the volume of air within the swim bladder. They tested the lake sturgeon in their experiment for this category of fish. ESA-listed physostomous fish in the southeast region include Gulf, Atlantic, and shortnose sturgeon.
- Physoclistous (fish with closed swim bladders: Halvorsen et al. (2012b) describes these fish as recently evolved fish species. Physoclistous fish have a gas gland that provides gas exchange by diffusion between the swim bladder and blood. They tested Nile tilapia in their study for this category of fish. The only ESA-listed physostomous fish in the southeast region is the Nassau grouper.

All 3 categories of fish were exposed to a series of 5 trials beginning with a cSEL of 216 to a cSEL of 204 dB re 1uPa2-s (derived from 960 pile strikes and 186 dB re 1uPa2-s sSEL). In this and subsequent studies by Halvorsen et al. (2012a); Halvorsen et al. (2012b), injuries were categorized by a response weighted index to categorize injuries as mild, moderate, or mortal. The authors defined mild injuries response weighted index 1 as those that were non-life

threatening. Based on this and similar studies, the authors made recommendations for cumulative noise exposure thresholds to be raised from the current 187 dB cSEL to 210 dB cSEL, because only mild injuries were observed up to 210 dB cSEL. Because we consider even mild injuries to be physical injury and we are concerned about the potential starting point/onset of physical injury and not the mean of when only mild injuries were observed, NMFS will continue to use the injury thresholds summarized in Tables 73 and 74 above to be conservative and protective of ESA-listed species, while acknowledging that the cumulative sound exposure threshold may be adjusted as new research becomes available.

One potentially important result of the Halvorsen et al. (2012b) study is that the hogchoker (fish without a swim bladder) did not suffer visible external or internal injuries at lower cSEL levels tested, while the swim bladder fish still suffered mild internal injuries. The fish with open swim bladders also suffered fewer internal injuries than the fish with closed swim bladders. As more research continues, this may lead to policy changes for different species of ESA-listed fish such as the smalltooth sawfish that also lacks a swim bladder. Although, another consideration for bottom-dwelling elasmobranchs such as the smalltooth sawfish is that they are often in contact with the substrate (Casper et al. 2012). The vibrations within the sediment from pile driving could also be damaging, especially when considering the body shape of sawfish. Many of the organs of these dorsoventrally flattened fishes are in close proximity to the bottom surface of the body, providing little protection from pile-driving vibrations. It is unclear if the Halvorsen et al. (2012b) study took into consideration the secondary effects of noise from vibrations within the sediment.

#### 15.3 Calculations of Noise Thresholds for Pile Driving

#### Assumptions of the Analyses

The calculations in this document are based on a series of assumptions. These assumptions and resulting key points are provided in the list below and discussed further in this section:

- The number of piles necessary to complete construction varies by the size and intent of the project. For example, typical residential dock with an average of 15 piles requires approximately 10 hours of pile driving and can take 2 or more days to complete. Some larger residential docks can use up to 70 piles or more and noise could be produced over a period of 2 weeks or longer. Pile driving for common minor activities does not occur at night.
- Pile types considered include:
  - o vinyl piles and sheet piles
  - wood piles (round timber)
  - o concrete piles (in a variety of shapes including round and square)
  - o metal boat lift I-beams (steel or marine-grade aluminum)
  - o metal piles used for docks and seawalls
  - o metal sheet piles (steel or marine-grade aluminum).
- Noise spreads cylindrically in coastal waters and noise transmission is characterized by 15 logR spreading loss.

- Strike rates for dock construction with wood piles have been reported to occur once or twice per minute (CALTRANS 2007). The average number of strikes per pile is estimated as 45, calculated as an average of 1.5 times per minute for 30 minutes.
- Concrete pile installation is estimated to take 160 strikes per pile.
- Sheet piles and I-beams can be installed by impact hammer in no more than 12 to 15 minutes, with pile strikes about once every 1.4 seconds or 43 to 44 strikes per minute (660 strikes per pile) (CALTRANS 2007).
- Daily exposure limits are based on the installation of 10 wood or concrete piles per day.
- Vibratory pile driving may take up to 30 minutes per pile.
- Installing vinyl piles is expected to produce noise similar to that produced by the installation of wood piles. We will use the available measurements for wood piles to estimate vinyl pile noise until additional information and measures from the installation of vinyl piles is available.
- For this analysis, we considered an impact pile driver as a combustion driven device used to install piles. There are 2 main classes of impact hammers: external combustion and internal combustion. External combustion hammers use cables, steam, compressed air, or pressurized hydraulic fluid to raise the ram, which is then dropped by gravity (e.g., a drop hammer). Internal combustion hammers do not rely on gravity and force the ram into the pile (e.g., a diesel hammer). During impact pile driving, noise is produced when the energy from the hammer is transferred to the pile and released into the surrounding water and sediment as the pile makes contact with the area into which it is being driven.
- Hand installation of any type of pile was determined to not result in noise at levels that would cause physical injury from peak pressure or cumulative sound exposure or cause behavioral effects and does not require mitigation.
- The noise analysis also evaluates effects of projects occurring in both open-water and confined space settings. This differentiation is important because if a project occurs in a confined space, the animal may not be able to avoid the physical injury zone or behavioral zone or may become trapped in an area (e.g., the terminal end of a canal) due to those zones blocking the only escape route. For our noise analysis, a *confined space* is defined as any area that has a solid object (e.g., shorelines or seawalls) that creates a constricted passage area such that species attempting to move through the area would be forced to pass through the cumulative injurious zone, which is the zone in which species could be exposed to injurious noise levels if they remained in the area over a period of time. To allow species to move through the project areas without being exposed to noise at levels that could be injurious over time, the PDCs limit certain pile types and installation methods in these confined spaces to ensure that the cumulative injurious zone is limited to a maximum of half the width of the confined area.

In Florida, we consider the confined space to be any area that has a solid object (e.g., shorelines or seawalls) within 150 ft of the pile installation site and in the U.S. Caribbean we consider confined space to be any area that has a solid object within 260 ft of the pile installation site. These confined space distances were calculated by doubling the cumulative injurious zone for large fish found in Florida and smaller fish found in the U.S. Caribbean (calculations provided in Tables 75 and 76, and an example of a confined space shown in Figure 29). Again, this ensures that a fish or sea turtle would have at least half the width of the channel free from cumulative injurious noise so that they could move through the area as needed without being exposed. Docks or other pile-supported structures do not stop or reflect noise do not create confined spaces. Conversely, in an open-water environment, the animal would be able to move away from the noise without passing through an area with injurious noise levels.



Figure 29. Example of a confined space for a noise analysis. The left image shows a 150 ft radius in a channel resulting in the channel not being wide enough to ensure more than half the channel was free from cumulative injurious noise levels. The right image shows a portion of San Juan Harbor and a 260 ft radius that would not block movement of species if piles were installed at this location because species would be able to move along the left side of the image between the 2 structures.

#### **15.3.1** Pile-Driving Noise Calculations

Tables 75 and 76 below provide a summary of the noise calculation results for each of the pile types installed by each of the pile installation methods. All calculations were completed using the formulas and methods described further in this section. Pile-driving source noise data was derived from CALTRANS (2009). Source levels were back-calculated from the reported measurement distance to the pile using 15 logR cylindrical spreading loss. These calculations assume that single strike sSEL less than 150 dB does not accumulate to cause physical injury meaning the sound level has decreased sufficiently at a distance from the sound source and has reached the effective quiet (explained in more detail below).

	Source Level (dB re 1 µPa <sup>53</sup> )	Radius for Fish less than 2 grams	Radius for Fish over 2 grams	Radius for Sea Turtles
Calculate	d 10 pilos ins	14-in wood pile and viny	yl sheet zas pou pilo = 450 strikae	s nor day
Physical Injury (peak pressure)	195 dB Peak	0 m (1 ft)	0 m (1 ft)	0 m (1 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 4 m (12 ft) 5 piles = 11 m (36 ft) 10 piles = 17 m (56 ft)	1 pile = $2 m (7 ft)$ 5 piles = $6 m (20 ft)$ 10 piles = $9 m (30 ft)$	1 pile = 2 m (7 ft) 5 piles = 6 m (20 ft) 10 piles = 9 m (30 ft)
Behavior (Root Mean Square [RMS])	185 dB RMS	215 m (705 ft)	215 m (705 ft)	46 m (151 ft)
Coloristad	10 - 0 - 1	24-in concrete pil	e	
Calculated	10 piles insta	alled per day with 160 strik	tes per plie = 1,600 strik	es per day
(peak pressure)	Peak	0 m (1 ft)	0 m (1 ft)	0 m (1 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 9 m (28 ft) 5 piles = 25 m (83 ft) 10 piles = 40 m (131 ft)	1 pile = 5 m (16 ft) 5 piles = 14 m (46 ft) 10 piles = 22 m (72 ft)	1 pile = 5 m (16 ft) 5 piles = 14 m (46 ft) 10 piles = 22 m (72 ft)
Behavior (RMS)	185 dB	215 m (705 ft)	215 m (705 ft)	46 m (151 ft)
A	Tw	o 12-in metal boat lift I-bea	am (H-pile) <sup>54</sup>	
Calculated	l 2 piles insta	lled per day with 660 strike	es per pile = 1,320 strike	es per day
Physical Injury (peak pressure)	205 dB Peak	1 m (3 ft)	1 m (3 ft)	1 m (3 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 22 m (72 ft) 2 piles = 35 m (115 ft)	1 pile = $12 \text{ m} (39 \text{ ft})$ 2 piles = $19 \text{ m} (62 \text{ ft})$	1 pile = $12 \text{ m} (39 \text{ ft})$ 2 piles = $19 \text{ m} (62 \text{ ft})$
Behavior (RMS)	190 dB RMS	465 m (1,526 ft)	465 m (1,526 ft)	100 m (328 ft)
		24-in metal sheet p	ile	
Calculated 10	sheet piles in	istalled per day with 660 st	rikes per pile = 6,600 st	rikes per day
Physical Injury (peak pressure)	220 dB Peak	9 m (30 ft)	9 m (30 ft)	9 m (30 ft)
Physical Injury (Cumulative exposure)	cumulative	1 pile = 410 m (1,345 ft) 10 piles = 858 m (2,815 ft)	1 pile = 223 m (732 ft) 10 piles = 858 m (2,815 ft)	1 pile = 223 m (732 ft) 10 piles = 858 m (2,815 ft)
Behavior (RMS)	204 dB RMS	858 m (2,8215 ft)	858 m (2,8215 ft)	185 m (607 ft)

Table 75. Impact Hammer sound source levels and impact radius distances

 $<sup>^{53}</sup>$  dB re 1  $\mu Pa$  is a unit of measurement of sound in decibels relative to 1 micro-Pascal-squared second

<sup>&</sup>lt;sup>54</sup> Noise levels not believed to be accurate based on the installation method used. Boatlift I-beams only penetrate loose sediment until they reach the top of, or first few inches of, hard substrate to stabilize the structure on the hard substrate versus penetrating it.

	Source Level (dB re 1 µPa)	Radius for Fish ≥102 grams	Radius for Sea Turtles
Calculated installation of	36-in wood, concrete, piles for 8 hours per day (no	vinyl, or metal piles limit on the number of p	iles per day)
Physical Injury (peak pressure)	186 dB Peak	0 m (0 ft)	0 m (0 ft)
Physical Injury (Cumulative exposure)	cumulative	0 m (0 ft)	0 m (0 ft)
Behavior (RMS)	170 dB RMS	215 m (705 ft)	46 m (151 ft)
Calculated 2 piles installe	Two 12-in metal boat h ed per day for 30 minutes (1,8	ift I-beam (H-pile) <sup>55</sup> 00 seconds) per pile = 3,6	600 seconds per day
Physical Injury (peak pressure)	165 dB Peak	0 m (0 ft)	0 m (0 ft)
Physical Injury (Cumulative exposure)	cumulative	0 m (0 ft)	0 m (0 ft)
Behavior (RMS)	150 dB RMS	0 m (0 ft)	0 m (0 ft)
Calculated installation of	24-in metal sheet piles for 8 hours per da	sheet pile y (no limit on the numbe	er of sheet piles per day)
Physical Injury (peak pressure)	192 dB Peak	0 m (0 ft)	0 m (0 ft)
Physical Injury (Cumulative exposure)	cumulative	0 m (0 ft)	0 m (0 ft)
Behavior (RMS)	178 dB RMS	74 m (243 ft)	16 m (52 ft)

Table 76. Vibratory Hammer sound source levels and impact radius distances

Underwater Construction Noises below the Threshold Levels for Physical Injury or Behavioral Effects

During our calculations, we determined that auger, drop punch, jetting, installation by land-based equipment, and hand installation did not result in noise levels that would cause physical injury or behavioral effects on listed species. These activities can temporarily increase ambient noise levels in an area, but we do not expect them to result in physical injury or behavioral effects to listed species, and thus we believe they will have no effect on the species.

Auger: When installing piles into hard substrates, sometimes a pilot hole is created using an auger, or by drop punching. Noise levels from small-scale drilling operations that are representative of dock construction methods have been measured to be no more than 107 dB re 1 μPa (0-peak) at 7.5 m from the source (Willis et al. 2010). Our back-calculation resulted in an approximate source level no greater than 120 dB re 1 μPa (0-peak). Noise associated

<sup>&</sup>lt;sup>55</sup> Noise levels are not believed to be accurate based on the installation method used. The impact radii are estimated based on typical installation of a metal pile in hard substrate, which requires a number of strikes to penetrate the substrate. Boatlift I-beams only penetrate loose sediment until they reach the top of, or first few inches of, hard substrate to stabilize the structure on the hard substrate. Thus, installing boatlift I-beams by impact hammer is likely to result in less noise than installing other metal piles by impact hammer in hard substrate.

with augering is below the behavioral and injury thresholds used in this analysis, and is discountable.

- 2. Drop punching is a method that uses a 12- to 24-in-diameter steel punch dropped repeatedly from a barge-mounted crane. After the pilot hole is created, the pile is inserted then driven to resistance using an impact hammer. Noise generated during drop punching has either not been measured or is unreported in the available literature. The best available information on construction equipment striking the sea bottom comes from measurements of bucket dredge noise. The noise produced from the heavy bucket dropped onto the channel bottom was measured to be 124 dB re 1 μPa (RMS) at 150 m from the work site (Dickerson et al. 2001). Back-calculating the noise attenuation 150 m results in a potential source level of 156 dB re 1 μPa (RMS), 6 dB above the behavioral threshold for fish. However, drop punch noise falls below 150 dB re 1 μPa (RMS) within a few feet of the installation site, and is well below the potential injury thresholds used in this analysis. In addition, the PDCs in this Opinion require crews to shut down construction equipment if protected species are observed within 50 ft of the project.
- 3. Jetting uses high-pressure water sprayed beneath the pile to excavate sediment and sand layers, and is often used in conjunction with other pile-driving methods to assist penetration of the pile into the substrate. Jetting results in much lower noise levels than either impact or vibratory pile driving alone and minimizes the amount of hammering necessary. Noise measurements taken with water jetting turned on or off during pile driving resulted in no additional noise recorded above that of the pile-driving noise (CALTRANS 2007). For complete pile replacement in the existing footprint, with no new piles driven, the source levels for jetting could be up to 170.5 dB re 1 μPa (peak) (Molvaer and Gjestland 1981) or about 160 dB (RMS), which would result in small behavioral zone for smalltooth sawfish and sturgeon (about 10-15 m) and no behavioral disturbance for sea turtles. Water jetting noise has an associated behavioral zone up to 15 m for fish, but this zone is within the same distance that construction is likely occurring and is too small to have any significant effect on fish behavior or habitat and is insignificant.
- 4. Land-based equipment does not generate noise in the marine environment that reach levels high enough to exceed the behavioral and physical injury thresholds used in this analysis. Because the air-water interface is an almost perfect reflector of acoustic waves, the noise generated by land-based mechanical excavators, generators, or other machinery will reflect off the surface and will not be transmitted into the water at levels expected to affect the species.
- 5. **Hand installation** of any type of pile was determined to not result in injurious or behavioral noise impacts and does not require mitigation.

#### 15.3.2 Noise Control Measures to Reduce Injury

There has been a fair amount of work performed in reducing noise from pile driving, which is summarized in Table 77. For coastal waters, 'bubble curtains' have been the primary focus of noise control efforts based on their sound attenuation capabilities (when properly designed) and cost effectiveness. Recently, confined bubble curtains and TNAPs (also referred to as pile isolation casings) have shown consistently good results as noise-reducing measures. This section evaluates the effectiveness of these noise abatement measures and others.

Sound	Description	Effe	ctiveness	Cost
Treatment		Reduction	Metric	
Bubble curtain or bubble tree	Air bubbles used to block sound	5-20+ dB	RMS, Peak, SEL	\$50-200
Confined bubble curtain	A fabric, solid, or tubular curtain is used to confine bubbles	9-22 dB	RMS, Peak, Particle velocity	\$100-200
Pile caps	Micarta caps used between the impact piling head and the pile to reduce noise	1-8 dB	RMS, Peak, SEL	Low material cost. May increase time to install pile.
Wood pile cushions	A block of wood used between the pile head and pile to reduce noise (often used with a pile cap).	11-26 dB	RMS, Peak, SEL	Low material cost. May increase time to install pile.
Temporary Noise Attenuation Pile	A physical barrier lined with foam or other materials	8-14 dB	RMS, Peak, SEL	Unknown.
Dewatered cofferdam	Removal of water around pile	15 dB 3-35 dB	RMS, Peak	Unknown. Assumed more than bubble curtains
Vibratory hammers	Alternative to impact hammers	10-20+ dB	RMS, Peak, SEL	2-3 times cost of impact hammers
Suction piles	Replacement for existing techniques	Very large reduction	All	Potential cost savings
Press-in piles	Piles are pressed into place	Very large reduction	All	Unknown

Table 77. Effe	ctiveness and	cost of	noise control	measures fo	or pile	driving
----------------	---------------	---------	---------------	-------------	---------	---------

Table modified from Pile Driving Treatments table found in Spence et al. (2007) and updated with data by Laughlin (2010).

TNAPs or pile casings consist of a steel casing lined with noise insulating foam that is placed over the pile during installment. For smaller piles associated with docks, use of polyvinyl chloride (PVC) piles may be a viable alternative to steel piles, but they have not yet been tested.

A TNAP design is a hollow walled (air-filled) metal pile casing or foam-lined metal casing placed around the pile being driven. This method is best applied to vertical, non-interlocking piles. Noise levels with a TNAP can be reduced by an average of 11 dB (8 to 14 dB) (Laughlin 2010), but have been reported in previous studies to have an even greater capacity to reduce noise. In the latest report (Laughlin 2010), a double-wall TNAP was constructed using 2 concentric pipes with outside diameters of 60 in and 48 in with and a wall thickness of 1 in. The 5-in space between the inner and outer steel tubes was partially filled with a 4-in-thick, sound-absorbing material. Tests were conducted both with and without bubbles between the pile and the hollow tube via a bubble ring at the bottom of the TNAP. Bubbles resulted in a slightly greater reduction of 1-3 dB than TNAPs without bubbles. Laughlin (2010) recommended that TNAPs could be used as an alternative to the bubble curtain as an underwater noise mitigation device since the average noise reduction is about the same. New designs of TNAPs and confined bubble curtain designs are still being investigated to improve noise reduction capabilities.

Because of their shape, TNAPs cannot be used on sheet piles so a confined bubble curtain may be used to reduce noise levels. Confined bubble curtains consist of some type of aquatic barrier around a pile that is filled with bubbles. The bubbles are created by forcing compressed air through small holes drilled in PVC or steel pipe. The bubble curtain disrupts the sound waves as they passes through, thereby reducing noise levels transmitted beyond the curtain. In order to be effective, the bubble curtain must fully enclose the pile through the entire water column; the ring emitting the bubbles must be seated properly on the sea floor to avoid suspension or sinking of the PVC pipe. Reductions in peak pressure, RMS pressure, and energy are typically on the order of 5-20 dB or more (though we consider an average reduction of 11 dB for bubble curtains). Although the effectiveness of confined bubble curtains is more variable than TNAPs, the average noise reduction is about the same. Based on the best case scenario of a 20 dB reduction in noise, our estimates of the amount of noise reduction required to reduce the injury distance to less than 50 ft indicate that steel sheet piles may still have an associated injury zone even when confined bubble curtains are used. Yet, the actual noise levels will be largely dependent upon the type of sheet pile, the power of the impact hammer, the substrate hardness, and water depth.

The installation of metal sheet pile with an impact hammer requires an individual analysis for each project to determine the appropriate noise abatement needed to adequately reduce injurious and behavioral noise levels; therefore, activities involving metal sheet pile installation with an impact hammer are not covered under this Programmatic Opinion. These activities will require separate Section 7 consultation to address the increased risk of injury from noise. Projects requiring sheet piles that produce injurious noise levels will likely need to employ confined bubble curtain designs, as well as additional mitigation measures when possible, such as working at low tide, to reduce the amount of noise transmitted into the water.

It is important to note that vibratory piling-driving methods may be a reasonable solution for reducing the peak noise levels and removing impulsive sounds in certain applications. Vibratory methods have been measured to be significantly quieter than impact pile driving methods. In many cases, the reduction in noise levels can eliminate any risk of injury and minimize the area of behavioral disturbance. The Florida Marine Contractor's Association has indicated that vibratory hammers are a viable option for the installation of I-beam used for in some boatlift

designs; however, flexibility is needed for using a variety of pile driver types depending on the location and sediment layers encountered during pile driving.

# 15.3.3 Bubble Curtain Specifications for Pile Driving

When using an impact hammer to drive or proof concrete or metal piles, use 1 of the following sound attenuation methods:

- 1) If water velocity is equal to or less than 1.6 ft per second (1.1 miles per hour) for the entire installation period, surround the pile being driven by a confined or unconfined bubble curtain that will distribute small air bubbles around 100% of the pile perimeter for the full depth of the water column.
  - a) General An unconfined bubble curtain is composed of an air compressor(s), supply lines to deliver the air, distribution manifolds or headers, perforated aeration pipe, and a frame. The frame facilitates transport and placement of the system, keeps the aeration pipes stable, and provides ballast to counteract the buoyancy of the aeration pipes in operation.
  - b) The aeration pipe system shall consist of multiple layers of perforated pipe rings, stacked vertically in accordance with the following:

Water Depth (m)	Number of Layers
0 to less than 5	2
5 to less than 10	4
10 to less than 15	7
15 to less than 20	10
20 to less than 25	13

- c) The pipes in all layers shall be arranged in a geometric pattern which shall allow for the pile being driven to be completely enclosed by bubbles for the full depth of the water column and with a radial dimension such that the rings are no more than 0.5 m from the outside surface of the pile.
  - i. The lowest layer of perforated aeration pipe shall be designed to ensure contact with the substrate without burial and shall accommodate sloped conditions.
  - ii. Air holes shall be 1.6 millimeter (mm) (1/16-in) in diameter and shall be spaced approximately 20 mm (3/4 in) apart. Air holes with this size and spacing shall be placed in 4 adjacent rows along the pipe to provide uniform bubble flux.
  - iii. The system shall provide a bubble flux 3.0 m<sup>3</sup> per minute per linear meter of pipe in each layer (32.91 ft<sup>3</sup> per minute per lin ft of pipe in each layer). The total volume of air per layer is the product of the bubble flux and the circumference of the ring:

 $V_{t} = 3.0$  cubic meter/minute/meter x Circumference of the aeration ring in m

or  $V_{t} = 32.91$  cubic feet/minute/feet x Circumference of the aeration ring in ft

iv. Meters shall be provided as follows:

- Pressure meters shall be installed at all inlets to aeration pipelines and at points of lowest pressure in each branch of the aeration pipeline.
- Flow meters shall be installed in the main line at each compressor and at each branch of the aeration pipelines at each inlet. In applications where the feed line from the compressor is continuous from the compressor to the aeration pipe inlet, the flow meter at the compressor can be eliminated.
- Flow meters shall be installed according to the manufactures recommendation based on either laminar flow or non-laminar flow.
- 2) If water velocity is greater than 1.6 ft per second (1.1 miles per hour) at any point during installation or you are constructing a seawall, surround the pile or area being driven by a confined bubble curtain (e.g., a bubble ring surrounded by a fabric or non-metallic sleeve). The confined bubble curtain will distribute air bubbles around 100% of the pile perimeter for the full depth of the water column, according to specifications below.
  - a) General A confined bubble curtain is composed of an air compressor(s), supply lines to deliver the air, distribution manifolds or headers, perforated aeration pipe(s), and a means of confining the bubbles.
  - b) The confinement shall extend from the substrate to a sufficient elevation above the maximum water level expected during pile installation such that when the air delivery system is adjusted properly, the bubble curtain does not act as a water pump (i.e., little or no water should be pumped out of the top of the confinement system).
  - c) The confinement shall contain resilient pile guides that prevent the pile and the confinement from coming into contact with each other and do not transmit vibrations to the confinement sleeve and into the water column (e.g., rubber spacers, air filled cushions).
  - d) In water less than 15 m deep, the system shall have a single aeration ring at the substrate level. In waters greater than 15 m deep, the system shall have at least 2 rings: 1 at the substrate level and the other at mid-depth.
  - e) The lowest layer of perforated aeration pipe shall be designed to ensure contact with the substrate without sinking into the substrate and shall accommodate for sloped conditions.
  - f) Air holes shall be 1.6 mm (1/16-in) in diameter and shall be spaced approximately 20 mm (3/4 in) apart. Air holes with this size and spacing shall be placed in 4 adjacent rows along the pipe to provide uniform bubble flux.
  - g) The system shall provide a bubble flux of 2.0 m<sup>3</sup> per minute per linear meter of pipe in each layer (21.53 ft<sup>3</sup> per minute per lin ft of pipe in each layer). The total volume of air per layer is the product of the bubble flux and the circumference of the ring: Vt = 2.0 cubic meter/minute/meter x Circumference of the aeration ring in m

or

Vt = 21.53 cubic feet/minute/feet x Circumference of the aeration ring in ft

- (h) Flow meters shall be provided as follows:
  - i. Pressure meters shall be installed at all inlets to aeration pipelines and at points of lowest pressure in each branch of the aeration pipeline.

- ii. Flow meters shall be installed in the main line at each compressor and at each branch of the aeration pipelines at each inlet. In applications where the feed line from the compressor is continuous from the compressor to the aeration pipe inlet, the flow meter at the compressor can be eliminated.
- iii. Flow meters shall be installed according to the manufacturer's recommendation based on either laminar flow or non-laminar flow.

# 15.4 How to Calculate Noise Impacts

# **15.4.1** Calculation Steps for Pile Driving

*a. Review the Project for Needed Information* The basic information on the pile driving activity required to conduct an effects analysis is:

- the material composition of the piles (vinyl, wood, concrete, metal)
- the type of piles (e.g., sheet, H, tubular, square, etc.)
- the diameter of the piles
- the number of piles driven
- the number of hammer strikes per pile
- the duration to drive a single pile
- the number of piles driven per day
- the time of year of the activity
- the type of pile-driving method (e.g., hydraulic, diesel, vibratory hammer)
- other pile driving methods (e.g., drilling, jetting)
- the total duration of the project
- depth, bottom, type, and habitat characteristics
- a map of the project area

## b. Choose a Spreading Loss Model

The decrease in noise level with distance from the source (also called attenuation or spreading loss) can be estimated using a spreading loss model. A general equation to be used for planning and assessment purposes to predict noise at some distance from a pile is:

 $TL(R) = SL - N \log R$ ; where

TL is the threshold level in Table 73 at a distance R from the pile in meters,

SL is the source level,

N is a coefficient for geometric spreading (e.g., spherical or cylindrical), and R is a distance from the source.

For pile-driving projects, geometric spreading (N) can range between 10 and 20, but usually takes the form of 2 equations based on water depth.

Spherical spreading in deep water is expressed as:  $TL(R) = SL - 20 \log R$ Intermediate spreading in shallower water is expressed as:  $TL(R) = SL - 15 \log R$  A general assessment rule in determining which spreading loss model to use is to compare your impact zone distance to the water depth in the project area. Use spherical spreading (20 logR) if your impact zone is shorter than the water depth, and cylindrical (10 logR) or intermediate spreading (15 logR) if your impact zone is longer than the water depth. This is explained by the fact a sound wave will not generally travel further than the depth of the water column before being reflected. In deep water, surface reflection does not occur as quickly. Pile driving in deeper water is best modeled using spherical spreading where there are few reflections of the sound waves off hard surfaces such as the sea bottom. In shallow water, surface reflections result in non-uniform or cylindrical spreading of the sound waves (see Figure 30).



Figure 30. An example of intermediate spreading loss where surface reflections result in non-uniform spreading of sound waves

Sound propagation can range between 10 logR and 15 logR in shallow water. For planning purposes, the use of the 15 logR spreading loss model is recommended unless project-specific data are available. Aside from offshore energy projects, most pile driving occurs in shallow, coastal areas so intermediate spreading loss is the most common model used for coastal areas. To find the distance of the threshold level TL(R) to determine your impact zone use the Spreading Loss Calculator explained below in Step f.

#### c. Determine the Noise Reference Levels

Noise levels produced from pile-driving noise can be estimated from similar projects reported in technical papers and peer reviewed literature. Typically, the pile size, type, and pile-driving method are used to characterize noise levels. A particularly useful reference is CALTRANS (2009). The source level will need to be determined on a project-by-project basis through information provided by the applicant or through reference levels reported in the literature. Report the noise levels in the effects analysis. The noise levels used in the effects analysis should be tabulated for easy reference.

It is important to note the distance of the reported noise level. Many reference levels are reported at 10 m from the pile. We can back-calculate noise levels from 10 m to the pile by adding 10 dB for 10 logR cylindrical loss, 15 dB for 15 log intermediate spreading loss, and 20 dB for 20 logR spherical spreading loss. Other reference level distances can easily be back-calculated by determining the dB loss for the distance using the Spreading Loss Calculator.

#### d. Determine Source Level: Cumulative Sound Exposure Level

Cumulative sound exposure is based on the amount of time an animal may be exposed to noise from repeated strikes of impact hammers (or the amount of time for vibratory piling). For any given set of conditions (source level, type of transmission loss, strikes/pile) over some period of time, cumulative sound exposure (cSEL) may result in some risk of hearing loss even if the sSEL (single exposure level) is below the threshold for physical injury. This calculation is important if animals may be expected to be repeatedly exposed to noise over time (e.g., nursery or developmental areas, preferred feeding or resting areas, or semi-enclosed areas in which animals may remain).

NOTE: cSEL assumes constant exposure, and does not account for the movement of fish and sea turtles. Movements must be monitored during the activity, modeled, or considered qualitatively in the analysis.

For dock and seawall construction, which only occurs during daylight hours in residential areas, the cSEL can be calculated on a daily basis:

Daily cSEL Source Level = sSEL Source Level + 10 log(number of strikes/ pile)(number of piles/day)

As a general guideline, consider the cumulative effects of noise exposure over a 24-hr period, as long as there is sufficient "quiet" recovery time between exposures. We believe that limiting construction to only daylight hours provides sufficient quiet time for activities covered under this Opinion. The effects of repeated daily exposures over days, weeks, or months may be considered qualitatively or quantitatively if the different noise sources and exposure levels are present over time.

Another important consideration in calculating cSEL in the context of pile driving is the "effective quiet" level. For fish, the effective quiet level has been set at 150 dB (CALTRANS 2009). For sea turtles, we are applying the same level. For animals exposed to levels at or less than the effective quiet level, noise impacts will not accumulate to cause injury. Therefore, we

need to calculate the distance from the source at which noise levels will attenuate below this effective quiet level. Only within this range will potentially injurious cSEL accumulate.

For example, if a pile has a 180 dB sSEL source level, the maximum cumulative sound exposure injury range is at 100 m from the pile. This is determined by finding the difference between the single-strike source level and effective quiet (180 dB - 150 dB = 30 dB). A 30 dB loss occurs at 100 m from a pile using the 15 logR spreading loss. Therefore, animals beyond 100 m would not accumulate potentially injurious cSEL and 100 m would be the limit of the physical injury zone from cumulative sound exposure.

# e. Determine the Impact Zones by Calculating Threshold Distances: Using the Spreading Loss Calculator

In previous steps, you will have already calculated the source level for both a single strike and for cumulative daily strikes. In Step 4, you have chosen the spreading loss model appropriate for a project. A quick and effective method to calculate impact zone distances with the model is to first calculate the difference in dB (-dB) between the source level and threshold level, then determine what distance that dB difference occurs with the Spreading Loss Calculator.

For example, to determine the distance of the daily cumulative sound exposure level of injury, first subtract the threshold levels for each animal group in Table 73 from the cSEL source level.

<u>Calculate the Difference (-dB) Between Source Level and Injury Threshold Levels</u> fish  $\geq 2$  grams and sea turtles = Source Level (cSEL) – 187 dB fish < 2 grams = Source Level (cSEL) – 183 dB

<u>Calculate the Difference (-dB)</u> Between Source Level and Behavioral Threshold Levels for all fish sizes = Source Level (RMS) - 150 dBsea turtles = Source Level (RMS) - 160 dB

After determining the dB difference between source level and threshold level, use the Spreading Loss Calculator to input different ranges in the first column (Range) to find the distance that the -dB difference would occur. The calculator uses 3 spreading loss formulas to allow for quick calculations of several ranges (see Figure 31). The equations solve for any range input by the user by automatically calculating noise reduction at those distances from a pile (-dB) using 3 spreading loss equations for any range input by the user.

A graphical representation of the impact zones is provided below to help visualize the area where project impacts will occur within the species' habitat (Figure 32).

	Spherical (20 logR) and Cylindrical (10 and 15 logR) Spreading Loss						
	Instructions: Input range from source to obtain spherical and cylindrical spreading loss (- dB)						
Range (m)	Range (m) log (R) 20 logR Spherical Spreading Loss (- dB) 10 log R Cylindrical Spreading Loss (- dB) 15 log R Cylindrical Spreading Loss (- d						
1	0	0	0	0			
2	0.301029996	6.020599913	3.010299957	4.515449935			
4	0.602059991	12.04119983	6.020599913	9.03089987			
8	0.903089987	18.06179974	9.03089987	13.5463498			
10	1	20	10	15			
25	1.397940009	27.95880017	13.97940009	20.96910013			
50	1.698970004	33.97940009	16.98970004	25.48455007			
100	2	40	20	30			
1000	3	60	30	45			
2000	3.301029996	66 02059991	33.01029996	49.51544993			
10000	4	80	40	60			
100000	5	100	50	75			
500000	5.698970004	113.9794001	56.98970004	85.48455007			
1000000	6	120	60	90			

Figure 31. Screenshot of the Spreading Loss Calculator. The dB loss over any range can be determined for 3 types spreading loss models (10 logR, 15 logR, and 20 logR). For example, at 50 m, there is a 25.5 dB reduction in noise from a pile due to intermediate transmission loss (15 logR).



Figure 32. An example graphic visualizing impact zones for a pile-driving project. Graphical representations of the impact zones are useful analytical tools in visualizing the area where project impacts will occur within the species' habitat.

### 15.4.2 Example Noise Calculation for a Single Family Dock

#### Step 1. Gather project details.

In the following example, a federal agency is proposing to issue a permit for the construction of a single family dock in Florida.

The applicant provided project details including the following:

- 16 piles will be driven by impact hammer
- all piles are 12-in-diameter wood piles
- each pile takes 30 minutes to install
- plan to install 10 piles per day
- pile installation is continuous, but only during daylight hours
- the hammer strikes at an average rate of 1.5 strikes per minute (45 strikes per pile)
- water depth ranges from 0-5 m
- sea turtles and smalltooth sawfish may be in the project area

### Step 2. Determine noise reference levels and choose a spreading loss model.

Referencing the noise levels reported for a 12-in wood pile in CALTRANS (2009), the source level is estimated to be 180 dB Peak Pressure, 160 dB sSEL, and 170 dB RMS at a distance of 10 m from the pile. Because the project is in shallow water, the 15 logR intermediate spreading loss model will be used. To back-calculate the source level from the reported level measured 10 m from the pile, we added 15 dB to each of the literature values (Table 78).

Type of Noise Impact	Measured Source Level (10 m from Source)	Back-Calculation to Source (15 logR)	Final Source Level
Peak Pressure	180	15	195
Single strike (sSEL)	160	15	175
RMS (behavioral)	170	15	185

#### Table 78. Back-Calculation of Source Levels

#### Step 3. Calculate the cumulative exposure level (cSEL).

To address the sound exposure level over the course of a day, the SEL from exposure to a single pile strike (sSEL) was converted to cSEL for exposure to the total pile strikes per day. This is calculated using the following formula:

Daily cSEL Source Level = sSEL Source Level + 10 log(number of strikes/ pile)(number of piles/day)

$$= 175 + 10 \log(45)(10)$$
  
= 175 + 26.5  
= 201.5 (rounded to 202)

**Step 4. Calculate the difference between project noise levels and threshold values.** To determine if noise associated with pile installation reaches a level loud enough to disturb or injure protected species, we compare project source levels to the literature threshold values (Table 79).

Effect	Animal	Threshold Level (dB re 1 µPa)	Project Levels (dB re 1 μPa)	Difference in dB
Physical Injury	All fish and turtles	206 (peak pressure)	195 (peak pressure)	Not exceeded
	Fish > 2 grams	187 (SEL)	175 (sSEL)	Not exceeded
	$1^{1}\sin \geq 2$ grains	107 (SEL)	202 (cSEL)	15
Behavior	Fish	150 (RMS)	185 (RMS)	35
	Sea turtles	160 (RMS)	185 (RMS)	25

 Table 79. Calculations of Threshold Exceedances

# Step 5. Use the Spreading Loss Calculator to determine the zone of impact in cases where source levels exceed threshold values.

Change the values in column 1 (Range) of the Spreading Loss Calculator (Figure 33) to calculate noise attenuation over specific distances. Alter the values in column 1 as necessary to find dB levels in the last column (15 logR model) that most closely match those calculated in Step 4 above (Table 79 "Difference in dB"). We demonstrate this in Figure 34 below where the arrows represent the ranges that we modified in column 1 and the resulting changes in dB loss calculated in the last column. By changing the values in column 1 (10, 46, and 215 m) we were able to match the dB of loss calculated in Table 79 above (15, 25, and 35 dB respectively). The ranges (i.e., distances) associated with the arrows correspond with the radii to which noise source levels exceed threshold levels at the project site (impact radius, Table 80).

Spherical (20 logR) and Cylindrical (10 and 15 logR) Spreading Loss						
Instructions: Input range from source to obtain spherical and cylindrical spreading loss (- dB)						
Range (m)	Range (m) 📃 log (R) 🔤 20 logR Spherical Spreading Loss (- dB) 10 log R Cylindrical Spreading Loss (- dB) 15 log R Cylindrical Spreading Loss (-					
1	0	0	0	0		
2	0.301029996	6.020599913	3.010299957	4.515449935		
4	0.602059991	12.04119983	6.020599913	9.03089987		
8	0.903089987	18.06179974	9.03089987	13.5463498		
10	1	20	10	15		
25	1.397940009	27.95880017	13.97940009	20.96910013		
50	1.698970004	33.97940009	16.98970004	25.48455007		
100	2	40	20	30		
1000	3	60	30	45		
2000	3.301029996	66 02059991	33.01029996	49.51544993		
10000	4	80	40	60		
100000	5	100	50	75		
500000	5.698970004	113.9794001	56.98970004	85.48455007		
1000000	6	120	60	90		

Figure 33. Screenshot of original Spreading Loss Calculator prior to any modifications

Spherical (20 logR) and Cylindrical (10 and 15 logR) Spreading Loss					
Instructions: Input range from source to obtain spherical and cylindrical spreading loss (- dB)					
Range (m) log (R) 20 logR Spherical Spreading Loss (- dB) 10 log R Cylindrical Spreading Loss (- dB) 15 log R Cylindrical Spreading Loss (- dB)					
1	0	0	0	0	
2	0.301029996	6.020599913	3.010299957	4.515449935	
4	0.602059991	12.04119983	6.020599913	9.03089987	
8	0.903089987	18.06179974	9.03089987	13.5463498	
10	1	20	10	15	
25	1.397940009	27.95880017	13.97940009	20.96910013	
46	1.662757832	33.25515663	16.62757832	24.94136748	
100	2	40	20	30	
215	2.33243846	46.6487692	23 3243846	34.9865769	
2000	3.301029996	66.02059991	33.01029996	49.51544993	
10000	4	80	40	60	
100000	5	100	50	75	
500000	5.698970004	113.9794001	56.98970004	85.48455007	
1000000	6	120	60	90	

Figure 34. Screenshot of the Spreading Loss Calculator after the first column was modified

Effect	Animal	Threshold Level (dB re 1 μPa)	Project Levels (dB re 1 μPa)	Difference in dB	Impact Zone Radius (m)
Physical Injury	All fish and turtles	206 (peak pressure)	195 (peak pressure)	Not exceeded	0
	Fish > 2 grams	197 (SEI )	175 (sSEL)	Not exceeded	0
	$\Gamma ISH \leq 2$ granns	107 (SEL)	202 (cSEL)	15	10
Behavior	Fish	150 (RMS)	185 (RMS)	35	215
	Sea turtles	160 (RMS)	185 (RMS)	25	46

# Table 80. Calculations of Impact Zones Based on Source Levels for the Project

# Step 6. Compare the calculated cSEL impact zone to the maximum impact zone limited by effective quiet.

Now we need to determine whether the amount of accumulated noise exposure and the corresponding cSEL impact zone would be limited by the noise reaching effective quiet. First, calculate the difference between the source level (sSEL) and the effective quiet level (150 dB sSEL).

In this example: 175 dB sSEL (source level) – 150 dB sSEL (effective quiet level) = 25 dB

Next, consult the Spreading Loss Calculator to find the distance over which sound would attenuate by that amount.

25 dB spreading loss occurs at 46 m using 15 logR (see Figure 34)46 m is the maximum range of the physical injury zone from cumulative sound exposure before reaching effective quiet

Last, compare the calculated cSEL impact zone radius from Step 5 (10 m) to the maximum impact zone limited by effective quiet, calculated in this step (46 m). Use the smaller of the 2 values as the cumulative sound exposure impact radius. In this case, the originally calculated 10 m is the impact zone radius for daily cumulative sound exposure.

# Summary

Interpretations of the noise impact zones associated with the project shown in Table 80 are as follows:

- Potential injury for fish and sea turtles from single pile strikes (sSEL) is not measurable as the source levels do not exceed the threshold values for injury.
- Potential injury to fish from cumulative sound exposure (cSEL) each day is possible within 10 m from the pile. For the purpose of this example, fish includes sturgeon and smalltooth sawfish. Since threshold values for sea turtles are not currently known we assume the thresholds used for fish also apply to sea turtles. Therefore we assume the 10 m cumulative sound exposure injury zone also applies to sea turtles.
- Behavioral responses of sea turtles may occur up to 46 m (150 ft) from the project.
- Behavioral responses of fish may occur up to 215 m (705 ft) from the project.

# 16 Appendix C: North Atlantic Right Whale Information Form

- Federal Regulations Governing the Approach to North Atlantic Right Whales 1. Federal regulations governing the approach to North Atlantic right whales can be found at 50 CFR 224.103(c). It is illegal to approach and remain within 500 yards of right whales; 500 yards is equal to the distance of 5 football fields. Prohibitions on approaching right whales are as follows (Excerpts from 50 CFR 224.103(c), available at www.ecfr.gov): Unless otherwise lawfully allowed or unless doing so would create an imminent and serious threat to a person or vessel, it is unlawful to: 500 yards *(i)* Approach (including by interception) within 500 yards (460 m) of a right whale by vessel *Fail to undertake required right whale avoidance* (ii) *measures. If underway, a vessel must steer a course* away from the right whale and immediately leave the area at a slow safe speed. 2. Updates can be downloaded from:
  - a. <u>http://www.nmfs.noaa.gov/pr/species/mammals/cetaceans/rightwhale\_northatlantic.htm</u>, or b. <u>www.ecfr.gov</u>
## 17 Appendix D: Frac-Out Plan Example

Proposed Methods for Protection of Water Quality for Directional Bored Water Crossings (best management practices [BMPs] and Frac-out Plan)

### BMPs

[The APPLICANT] and [the Contractor] will implement the following BMPs to minimize the potential for adverse environmental impacts during horizontal directional drilling activities:

• BMPs for erosion control within the staging area shall be implemented and maintained at all times during the drilling and back-reaming operations to prevent siltation and turbid discharges in excess of State Water Quality Standards pursuant to Rule 62-302, F.A.C. Methods shall include, but are not limited to the immediate placement of turbidity containment devices such as turbidity screen, silt containment fence, hay bales, and earthen berms, etc. to contain the drilling mud. Earthen berms shall not be utilized as to impact wetlands or other surface waters.

## **Frac-out Plan**

To provide an additional level of resource protection, the following measures shall be taken to monitor any potential releases of drilling fluid:

- Measures used to prevent frac-out during the drilling operation include maintaining the proper depth for the soil conditions along the drilling route as well as proper management of drilling fluids circulation pressure. Under the waterway, the minimum distance between the pipe and the bottom of the waterway will be \_\_[#]\_\_ ft as shown on the cross section. This is expected to be sufficient to prevent frac-out when drilling under the waterway.
- Non-toxic fluorescent dyes will be added to the drilling lubricant as a method for monitoring bentonite releases in the underwater portions of this drilling. Details of the fluorometry monitoring method shall be submitted to the USACE prior to the pre-construction meeting.
- The volume of bentonite in the drill string will be monitored at all times during the directional drilling operation. Should a drop in volume of bentonite occur, immediately conduct a visual inspection of both terrestrial and subaqueous portions of the horizontal directional drilling corridor.
- Should the detection of dye or a drop in volume of bentonite occur, the Contractor will follow the Release Procedures outlined below.
- The Contractor will identify prior to commencement of construction an environmental scientist/biologist with experience in-water quality monitoring and habitat protection to be used in the event of a frac-out. The biologist will supervise the implementation of the Frac-Out Plan, Release Procedure, and Containment Plan outlined below. Divers shall be present during drilling operations in order to respond to a potential frac-out release.
- All drilling fluids associated with the horizontal directional drilling operation will be contained on site. The volume of the drilling fluids recirculation/solids settlement pit will be determined by the Contractor at the Pre-Construction meeting. Periodically during the drilling process settled solids will be removed from the pit by a backhoe and disposed of

at a site of the Contractor's choice in accordance with applicable regulations. At the conclusion of drilling operations, drilling fluid remaining in the pit will be settled and hauled to a disposal site of the Contractor's choice in accordance with applicable regulations. After back-reaming, drilling materials will be removed from the inside of the pipeline by pigging it from the exit point towards the rig area.

- At all times, adequate protection will be taken to avoid impacts to the Aquatic Preserve/Outstanding Florida Waters and contiguous wetlands. This shall include, but is not limited to halting of construction/drilling and/or placement of turbidity containment devices.
- A Vactor Truck shall be onsite and available at all times.
- A Spill Kit (i.e., absorbent pads/brooms, goggles, gloves) shall be on-site and available at all times.

# **Release Procedure:**

- If a frac-out is confirmed, all construction activity contributing to the frac-out shall cease immediately.
- If the return drilling mud/fluid is less than the projected amount to be recovered, divers shall begin their search for the missing material within 1 hour of potential release. Once the drilling mud and frac-out is located, then the drilling mud containment plan shall be immediately implemented.
- If a frac-out has occurred during construction activities, the permittee shall notify the USACE of Engineers, Palm Beach Gardens Regulatory office, within 24 hours of the occurrence. The notification shall include the time of the frac-out, the response time of the underwater diver, and the environmental conditions of the affected area.

# **Drilling Mud Containment Plan:**

- Should the release of drilling materials occur on land, a sediment fence shall be constructed around the site and the material shall be removed by vacuum truck.
- Should the release of drilling materials occur in-water, clean-up with a vacuum system shall commence within 24 hours.
- The scientist/biologist underwater divers will guide the suction hose of the pump to minimize both the removal of natural bottom material and the disturbance of any existing vegetation.
- Any escaped drilling lubricant must be pumped into filter bags or directly into a vactor truck.
- A barge company will be contacted to transport a vactor truck should it be needed to respond "in-water."
- Once the spill is contained, the escaped drilling lubricant shall be properly disposed of in an approved upland disposal site.
- Clean-up with a vacuum system shall commence within 24 hours.
- After containment/recovery of the drilling material/resources, a detailed written report shall be submitted to the USACE, within 10 business days, indicating the location of the frac-out, amount of drilling material discharged and the amount of drilling mud recovered, the process in which the drilling mud was recovered, and the area that was affected by the drilling discharge.